MILLER PETROLEUM INC Form 10-K July 28, 2010

#### **UNITED STATES**

#### SECURITIES AND EXCHANGE COMMISSION

#### Washington, D.C. 20549

#### **FORM 10-K**

(Mark One)

ü

#### ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the fiscal year ended: April 30, 2010 OR

### TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from: \_\_\_\_\_\_ to \_\_\_\_\_

#### **MILLER PETROLEUM, INC.**

(Exact name of registrant as specified in its charter)

(State or Other Jurisdiction of Incorporation or Organization)

Tennessee

#### 001-34732

(Commission File Number) 62-1028629 (I.R.S. Employer Identification No.)

# 3651 Baker Highway, Huntsville, TN 37756

(Address of Principal Executive Office) (Zip Code)

#### (423) 663-9457

(Registrant s telephone number, including area code)

#### Securities registered pursuant to Section 12(b) of the Act:

Common Stock, par value \$0.0001 per share

Title of each class

#### Securities registered pursuant to Section 12(g) of the Act:

None (Title of Class)

Indicate by check mark if the registrant is a well-known seasoned is	ssuer, as define Yes	d in Rule 405 of the ü	Securities Act. No
Indicate by check mark if the registrant is not required to file report Act.	ts pursuant to S	ection 13 or Section	15(d) of the
	Yes	ü	No
Indicate by check mark whether the registrant (1) has filed all report Securities Exchange Act of 1934 during the preceding 12 months (or required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.		•	. ,
Indicate by check mark whether the registrant has submitted electro	onically and pos	sted on its Corporate	Website if

Indicate by check mark whether the registrant has submitted electronically and posted on its Corporate Website, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§229.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files).

Yes No

Name of each exchange on which registered NASDAQ Global Market Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229.405 of this chapter) is not contained herein, and will not be contained, to the best of registrant s knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. ü

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company.

Large accelerated filer Non-accelerated filer		Accelerated filer Smaller reporting company		
Indicate by check mark whether the registrant is a shell comp Rule 12b-2 of the Act).	pany (as defined in	Yes	ü	No

The aggregate market value of the outstanding common stock, other than shares held by persons who may be deemed affiliates of the registrant, computed by reference to the closing sales price for the registrant s common stock on October 30, 2009 (the last business day of the registrant s most recently completed second quarter), as reported on the OTC Bulletin Board, was approximately \$6,200,000. As of July 22, 2010, there were 33,389,383 shares of common stock of the registrant outstanding.

#### DOCUMENTS INCORPORATED BY REFERENCE

List hereunder the following documents if incorporated by reference and the Part of the Form 10-K (e.g., Part I, Part II, etc.) into which the document is incorporated: (1) Any annual report to security holders; (2) Any proxy or information statement; and (3) Any prospectus filed pursuant to Rule 424(b) or (c) under the Securities Act of 1933. The listed documents should be clearly described for identification purposes (e.g., annual report to security holders for fiscal year ended December 24, 1980). None.

# MILLER PETROLEUM, INC.

# TABLE OF CONTENTS

PART I	<u>Page No.</u>
Business	
	1
Item 1A.	
Risk Factors	
	<u>20</u>
Item 1B.	
Unresolved Staff Comments	
	<u>26</u>
<u>Item 2.</u>	
Properties	
	<u>26</u>
<u>Item 3.</u>	
Legal Proceedings	
	<u>26</u>
<u>Item 4.</u>	
(Removed And Reserved)	
	27

#### PART II

<u>Item 5.</u>

Market for Registrant s Common Equity, Related Stockholder Matters and Issuer Purchases of

# Equity Securities

<u>Item 10.</u>

Directors, Executive Officers and Corporate Governance

	<u>43</u>
<u>Item 11.</u>	
Executive Compensation.	
	<u>48</u>
<u>Item 12.</u>	
Security Ownership of Certain Beneficial Owners and Management and Related	
Stockholder Matters	
	<u>52</u>
<u>Item 13.</u>	
Certain Relationships and Related Transactions, and Director Independence.	
	<u>54</u>
<u>Item 14.</u>	
Principal Accounting Fees and Services	
	<u>55</u>
PART IV	
<u>Item 15.</u>	
Exhibits, Financial Statement Schedules.	

<u>56</u>

#### CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING INFORMATION

Various statements in this annual report contain or may contain forward-looking statements that are subject to known and unknown risks, uncertainties, and other factors which may cause actual results, performance or achievements to be materially different from any future results, performance or achievements expressed or implied by such forward-looking statements. These forward-looking statements were based on various factors and were derived from utilizing numerous assumptions and other factors that could cause our actual results to differ materially from those in the forward-looking statements. These factors include, but are not limited to the following:

the capital intensive nature of oil and gas development and exploration operations and our ability to raise adequate capital to fully develop our operations and assets,

our ability to perform under the terms of the Assignment Oversight Agreement with the Alaska DNR, including meeting the funding commitments of that agreement,

fluctuating oil and gas prices and the impact on our results of operations,

our ability to secure an extension of the Susitna Basin Exploration License,

the impact of the global economic crisis on our business,

the impact of natural disasters on our Cook Inlet Basin operations,

the imprecise nature of our reserve estimates,

•

.

our ability to recover proved undeveloped reserves and convert probable and possible reserves to proved reserves,

the possibility that present value of future net cash flows will not be the same as the market value,

the costs and impact associated federal and state regulations,

changes in existing federal and state regulations,
.
our dependence on third party transportation facilities,
.
insufficient insurance coverage,
.
conflicts of interest related to our dealings with MEI,
.
cashless exercise provisions of outstanding warrants,

market overhang related to restricted securities and outstanding options, warrants and convertible notes,

adverse impacts on the market price of our common stock from sales by the selling security holders.

Most of these factors are difficult to predict accurately and are generally beyond our control. You should consider the areas of risk described in connection with any forward-looking statements that may be made herein. Readers are cautioned not to place undue reliance on these forward-looking statements, and readers should carefully review this annual report in its entirety, including the risks described in Item 1A. Risk Factors. Except for our ongoing obligations to disclose material information under the Federal securities laws, we undertake no obligation to release publicly any revisions to any forward-looking statements, to report events or to report the occurrence of unanticipated events. These forward-looking statements speak only as of the date of this annual report, and you should not rely on these statements without also considering the risks and uncertainties associated with these statements and our business.

# **OTHER PERTINENT INFORMATION**

We maintain our web site at www. millerenergyresources.com. Information on this web site is not a part of this prospectus.

Unless specifically set forth to the contrary, when used in this prospectus the terms Miller," "we", "us", "our", and similar terms refer to Miller Petroleum, Inc., a Tennessee corporation doing business as Miller Energy Resources, and its subsidiaries.

# PART I

ITEM 1.

**BUSINESS.** 

#### Overview

We are an independent exploration and production company that utilizes seismic data and other technologies for geophysical exploration and development of oil and gas wells in the Appalachian region of eastern Tennessee and the Cook Inlet Basin in south central Alaska. In addition to our engineering and geological capabilities, we provide land drilling services on a contract basis to customers primarily engaged in natural gas exploration and production.

During 2010, we have significantly expanded our operations through the December 2009 acquisition of oil and gas assets from Pacific Energy Resources through a bankruptcy proceeding in which we acquired onshore and offshore production and processing facilities, the Osprey offshore energy platform, over 600,000 net lease acres of land with hundreds of miles of 2-D and 3-D geologic seismic data, miscellaneous roads, pads and facilities. Our current strategy focuses the majority of our efforts on growing our company, including the following:

increasing our overall oil and gas production through maintenance and repairs of nonperforming or underperforming wells located in Alaska,

organically growing production through drilling for our own benefit on existing leases and acreage in the exploration license with a view towards retaining the majority of working interest in the new wells, and

expanding our contract drilling and service capabilities and revenues, including drilling contracts with third parties.

#### Our exploration and production activities

Historically, we focused our exploration, development, and production efforts in the Appalachian region of eastern Tennessee. During 2010 we significantly increased our operations through the acquisitions of KTO and ETC in our Appalachian region and the assets in Alaska, which comprise our Cook Inlet operations. As of April 30, 2010, we had approximately 662,992 acres of gross oil and gas leases and exploration license rights (616,550 net acres), which includes 471,474 acres under the Susitna Basin Exploration License.

# **Cook Inlet Basin**

We own approximately 131,190 gross acres of leasehold interests, the exploration license rights to an additional 471,474 acres and 10 crude oil and five natural gas wells in which we own an interest. Cook Inlet stretches 180 miles

from the Gulf of Alaska to Anchorage in south-central Alaska. The Cook Inlet Basin contains large oil and gas deposits including several offshore fields. There are also numerous oil and gas pipelines running around and under the Cook Inlet.

At the time the Alaskan assets were acquired by us, all of the operated wells were shut-in, a term used in the oil and gas industry which means the wells were closed off so that they could not produce oil, with the exception of the WF-2 natural gas well. As of June 30, 2010, four of the oil wells had been returned to production. In addition, Cook Inlet Energy owns a 30% working interest in two gas wells operated by Aurora Gas, Three Mile Creek 1 and Three Mile Creek 2, which have been operated continuously.

Oil wells drilled in this area range from 9,000 vertical feet to 10,000 vertical feet in depth while gas wells have a vertical depth of 8,000 feet to 9,000 feet. Wells that are deviated (continue on from the vertical depth either diagonally or horizontally) will have a longer measured depth of approximately 5,000 feet giving total measured depth of 14,000 feet to 15,000 feet. Well spacing is quite variable, as there are large parts of Cook Inlet which are completely undeveloped, and others, that are more mature. Our fields have approximately 60 to 80 acre spacing. The Cook Inlet basin contains a thick section of terrestrial Tertiary rocks which includes shales, sandstones, and coals. The primary targets in the area are crude oil reserves.

In January 2010 we entered into a Master Services Agreement with Fairweather E&P Services, Inc., a company based in Anchorage, Alaska which provides a wide range of support services for the oil and gas industries, whereby it acts as an independent contractor for us in the development and/or refurbishment of the wells in Cook Inlet Basin. The agreement it provides us with engineering, logistics, field and project management support for the

well and facility work in Cook Inlet Basin which are anticipated to be completed on or before December 31, 2012. We pay the contractor for all costs associated with these services, including any services that Fairweather E&P may subcontract to third party providers, at its cost plus 15%. Fairweather E&P is required to maintain certain minimum levels of insurance coverage and the agreement contains customary cross-indemnification provisions. We may terminate the agreement at any time without reason.

#### Susitna Basin Exploration License

Included in the Alaskan assets we acquired is a 100% interest in an Exploration License granted by the State of Alaska in October 2005 covering approximately 471,474 acres in the Susitna basin area north of Anchorage, Alaska. Under the terms of the Exploration License, the licensee was granted a five year exclusive license to explore for oil and gas on the specified lands, and upon fulfillment of the work commitment, the license for all or any part of the land could be converted into oil and gas leases. The original work commitment of approximately \$3.5 million was fulfilled, and we have the right at any time to covert the license for all or any portion of the acreage into oil and gas leases at any time. Once the exploration license is converted into oil and gas leases, we are required to pay a per acre fee to the state and commence drilling operations within specified timeframes. In an effort to control the timing of the development of this acreage, in April 2010 we requested a three year extension of the exploration license for a work commitment of \$750,000. While we reasonably believe the state will grant our request for extension, there are no assurances we are correct, or that if granted, that the terms and conditions of the extension will be satisfactory to us. If we are unable to negotiate an extension, it is likely we will convert the license for only a portion of the land into oil and gas leases.

#### **Osprey Platform**

Included in the assets acquired from Pacific Energy was the Osprey platform which is located in the Redoubt Unit approximately 1.8 miles southeast of the West Foreland in central Cook Inlet at a water depth of approximately 45 feet. The Osprey platform, which produces from the Redoubt Unit is connected to our Kustatan Production Facility by two eight-inch and one six-inch pipeline and one power cable. It relies on our Kustatan Production Facility, which is currently inactive, and our West McArthur River Unit Production Facility to provide all of its electricity and gas, and the Kustatan Production Facility to process all of Osprey's produced fluids. The platform has 21 slots, eight of which are currently used, and an attached 40 man camp. The platform is currently inactive.

The Osprey platform was placed on site during June 2000 and it initially conducted exploration drilling operations between January 2001 and July 2002. Eight wells were drilled, which in their present configuration consist of one water flood well, one Class I injection well, and six oil wells. The oil wells were equipped with electrical submersible pumps which were necessary to bring the oil to surface. In 2005, the third-party drilling rig was removed from the platform after a contract dispute. The removal of the rig crippled the ability to maintain and repair the platform s wells or to expand production. Shortly afterwards, a series of mechanical problems took much of the platform s production offline and these problems could not be corrected without a rig present. Reduced production was temporarily restored by deploying jet pumps, which can provide artificial lift, but do not require a rig; however, production continued to fall, and the Osprey platform was shut-in in the spring of 2009.

In order to restore production from the Redoubt Unit, it will be necessary to mobilize a drilling rig to the Osprey platform and repair six wells. We believe that past experience suggests that a rig should be permanently located on the platform. Two of these wells require only the replacement of the electrical submersible pumps, and the other four wells require re-drilling in sections. We estimated that the total cost of restoring production, including the purchase of a drill rig, is approximately \$35 million.

#### **Assignment Oversight Agreement**

On November 5, 2009, Cook Inlet Energy, LLC entered into an Assignment Oversight Agreement with the Alaska DNR which set out certain terms under which the Alaska DNR would approve the assignment of certain specified

state oil and gas leases from Pacific Energy Resources to Cook Inlet Energy. This agreement remains in place following our acquisition of Cook Inlet Energy in December 2009. Generally, the agreement requires Cook Inlet Energy to provide the Alaska DNR with additional information and oversight authority to ensure that Cook Inlet Energy is acting diligently to develop the oil and gas from Redoubt Shoal, West McArthur River Field and West Foreland Field. Under the terms of the agreement, until the Alaska DNR determines, in its sole discretion, that Cook Inlet Energy has completed its development and operation obligations under the assigned leases, Cook Inlet Energy agreed to the following:

file a monthly summary of expenditures by oil and gas filed, tied to objectives in Cook Inlet Energy s business plan and plan of development previously presented to the Alaska DNR,

.

meet monthly with the Alaska DNR to provide an update on operations and progress towards meeting these objectives,

notify the Alaska DNR 10 days prior to commitment when Cook Inlet Energy is preparing to spend funds on a purchase, project or item of more than \$100,000 during the first 12 months, more than \$1 million during the second 12 months and more than \$5 million thereafter, and

submit a new plan of development and plan of operations for the Alaska DNR s approval on or before December 15, 2009 and submit a plan of development annually thereafter on or before February 1, 2010. Cook Inlet Energy timely met both of these deadlines.

The agreement required Cook Inlet Energy to obtain financing in the minimum amount of \$5,150,000 to provide funds to be used for expenditures approved by the Alaska DNR as part of Cook Inlet Energy s plan of development. The funds are to be used for workover and repair of the wells, repair of the physical infrastructure, construction of a grind and inject plant at the West McArthur River facility, normal operating expenses associated with the leases and infrastructure and other capital project which are to be pre-approved by the Alaska DNR. The agreement also required Cook Inlet Energy to demonstrate funding commitments to support restoration of the base production at the Redoubt Unit, including bringing a number of the shut-in wells back on line, which was estimated at \$31 million in the agreement but which we have internally increased to \$35 million to accommodate the purchase of a drilling right. These funding commitments necessary under the agreement were provided to us under the terms of the Vulcan Capital Corporation, LLC letter agreement described earlier in this annual report. We have subsequently provided these funds through for the West McArthur River facility using a portion of the proceeds of our capital raising efforts described elsewhere herein, and intend to seek alternative sources of funding for the balance of the necessary capital.

Cook Inlet Energy is prohibited from using any of the approximate \$36.15 million or any proceeds from the operations under the assigned leases of the funding commitments for non-core oil and gas activities under the assigned leases, or any activities outside the assigned leases, without the prior written approval of the Alaska DNR until the parties mutually agree that the full dismantlement obligation under the assigned leases is funded. The assigned leases will be subject to default and termination should Cook Inlet Energy fail to submit the information required under the agreement and expenditure of funds for items or activities do not support core oil and gas activities, as reasonably determined by the Alaska DNR.

# **Recent Developments**

Cook Inlet Energy, LLC was one of nine successful bidders in State of Alaska s Division of Oil & Gas Cook Inlet Areawide 2010 Competitive Oil and Gas Lease Sale. There were 38 bids for 36 tracts covering an estimated 144,640 acres of State of Alaska oil and gas acreage. Cook Inlet Energy bid on seven tracts and was the successful high bidder on each of those tracts which cover an estimated 27,520 acres. Cook Inlet Energy s winning bid for these seven tracts was \$908,800. Cook Inlet Energy paid a deposit of \$181,767 at the time of the auction and the balance will be due once the title work is complete which we presently anticipate to be in January 2012. All of Cook Inlet Energy s bids completed acreage positions covering prospects acquired in its purchase of a portfolio of Pacific Energy Alaska assets late last year.

# Membership in Cook Inlet Spill Prevention and Response, Inc.

Cook Inlet Energy is a Class G member of the Cook Inlet Spill Prevention and Response, Inc., which we refer to as CISPRI. CISPRI is a non-profit corporation formed in 1990 to provide oil spill prevention and response capabilities in Cook Inlet. CISPRI has been designated as a Class "E" Oil Spill Removal Organization by the U.S. Coast Guard, which is the highest level of designation based on spill containment and removal equipment requirements for offshore/ocean response. CISPRI's response zone includes the entire Cook Inlet Region, stretching from Palmer to the Barren Islands and out into the Gulf of Alaska. At each annual meeting of CISPRI members adopt a budget for the coming year which includes funds for day to day operational activities of CISPRI, investments in capital equipment and materials to be used in connection with the cleanup activities and research and development and training. The budget is funded though payment of dues by the members and the amount of dues is calculated in accordance with a participation formula. Class G members pay an annual fee of \$10,000 together with additional fees based upon the amount of oil we transport.

If a spill is identified as originating from facilities owned or operations conducted by one or more of the members, CISPRI will act to control and clean up the spill of crude oil/synthetic crude oil or refined petroleum products arising from those operations without any future action by the members. Any member that utilizes or receives the benefit of these activities must reimburse CISPRI for all expenses of control and clean up, including costs of equipment, materials and personnel. Each member is required to execute a response action contract providing terms and conditions under which response and cleanup activities will be undertaken. Cook Inlet Energy is a party to such an agreement which, in part, requires Cook Inlet Energy to maintain worker s compensation insurance, employers liability insurance, comprehensive general and automotive liability insurance covering injury or death or persons and property damage of at least \$10 million. Cook Inlet Energy is in compliance with this insurance requirement. All members accept responsibility for spills which result from their operations or facilities and have indemnified CISPRI and all other members for all liabilities arising for a spill. This indemnification is not limited by the amount of insurance coverage.

Cook Inlet Energy may resign its membership in CISPRI upon 30 days written notice. At the effective date of the resignation, Cook Inlet Energy is obligated to pay all unpaid dues and assessments levied prior to the notice of resignation. Cook Inlet Energy s membership may be terminated by the Board of Directors of CISPRI upon 60 days notice if its determined Cook Inlet Energy is no longer eligible for membership. Cook Inlet Energy would not be entitled to a refund of any monies paid to CISPRI.

#### **Appalachian Region**

We own approximately 54,506 gross acres of leasehold interests with 185 producing oil wells and 334 producing gas wells in which we own an interest. Wells drilled in this area range from 1,800 to 4,200 feet in depth and the well spacing is generally from 20 to 40 acres per well and are predominately in Fort Payne formation.

The following table provides information on our reserves at April 30, 2010.

	Net Reserves at April 30,							
	201	10	20	09	20	2008		
		Natural		Natural		Natural		
	Oil	Gas	Oil	Gas	Oil	Gas		
<b>Reserves category</b>	(MBbls)	(MMcf)	(MBbls)	(MMcf)	(MBbls)	(MMcf)		
PROVED								
Developed producing								
Cook Inlet	1,695	1,085						
Appalachian region	108	619	43	563	63	511		
Developed - non								
producing								
Cook Inlet	856							
Appalachian region	6	33	10	30	11	1,341		
Undeveloped								
Cook Inlet	7,679	3,722						
Appalachian region				1,271				
Total Proved	10,344	5,459	53	1,864	74	1,852		
PROBABLE								
Non-producing								
Cook Inlet		5,567						
Appalachian region								
Undeveloped								
Cook Inlet	6,173							

Appalachian region		3,695
Total Probable	6,173	9,262
POSSIBLE		
Undeveloped		
Cook Inlet	1,078	5,203
Appalachian region	39	
Total Possible	1,117	5,203

When used in this table, MBbls means million barrels of oil and MMcf means million metric cubic feet. We also use a number of terms when describing our reserves. Proved reserves are the quantities of oil and gas that, by analysis of geosciences and engineering data, can be estimated with reasonable certainty to be economically producible. We provide information on two types of proved reserves - developed and undeveloped. Proved developed reserves are reserves that can be expected to be recovered through existing wells with existing equipment and operating methods and proved undeveloped reserves are reasonably certain reserves in drilling units immediately adjacent to the drilling unit containing a producing well as well as areas beyond one offsetting drilling unit from a producing well.

Under recent SEC rules we are now also permitted to provide information about probable and possible reserves. As set forth above, prior to 2010 our reserve reports did not contain any estimates on probable or possible reserves. Probable reserves are additional reserves that are less certain to be recovered than proved reserves but which, in sum with proved reserves, are as likely as not to be recovered. Possible reserves are additional reserves that are less certain to be recovered than probable reserves that are less certain to be recovered than probable reserves are additional reserves. The various reserve categories have different risks associated with them. Proved reserves are more likely to be produced than probable reserves and probable reserves are more likely to be produced than possible reserves. Because of these risks, the different reserve categories should not be considered to be directly additive.

Our reserve estimates for oil and natural gas at April 30, 2010 for our Cook Inlet assets were prepared by Ralph E. Davis Associates, Inc., an independent engineering firm, and our reserve estimates for oil and gas at April 30, 2010 for our Appalachian region assets were prepared by Lee Keeling and Associates, Inc., an independent engineering firm. Both of these reserve reports which are filed as exhibits to this annual report, were prepared in accordance with the generally accepted petroleum engineering and evaluation principles and most recent definitions and guidelines established by the SEC. All reserve definitions comply with the applicable definitions of the rules of the SEC. The reserves were estimated using engineering and geological methods widely accepted in our industry. The accuracy of the reserve estimates is dependent upon the quality of available data and upon independent geological and engineering interpretation of that data. For proved developed producing, the estimates considered to be definitive, using performance methods that utilize extrapolations of various historical data including oil, gas and water production and pressure history. For other than proved producing, proved undeveloped reserves and probable and possible reserve estimates were made using volumetric methods.

Our policies regarding internal controls over reserve estimates require reserves to be in compliance with the SEC definitions and guidance and for reserves to be prepared by an independent engineering firm under the supervision of our Chief Financial Officer. We provide the engineering firm with estimate preparation material such as property interests, production, current operation costs, current production prices and other information. This information is reviewed by our Chief Executive Officer and our Chief Financial Officer to ensure accuracy and completeness of the data prior to submission to our third party engineering firm. A letter which identifies the professional qualifications of each of the independent engineering firms who prepared the reserve reports are included in those reserve reports which are filed as exhibits to this annual report. There was no conversion of undeveloped reserves to proved reserves during the fiscal year ended April 30, 2010.

Each of the engineering reports also projected future net income (FNI) from our net reserves and the present value, discounted at 10% per annum, of that future net income FNI @ 10% as summarized in the following table. Future net income is based upon gross income from future production, less direct operating expenses and taxes. Estimated future capital for development costs was also deducted from gross income at the time it will be expended. No allowance was made for depletion, depreciation, income taxes or administrative expense. In the following table, the price per barrel of oil was \$73.01 and the price per MMcf of natural gas was \$4.84 for the Cook Inlet reserves and \$71.85 per barrel of oil and \$5.15 per MMcf of natural gas for the Appalachian region reserves. In each instance these prices are computed in accordance with the SEC s rule and represent the average fiscal year prices.

	Producing	No	n-Producing	τ	J <b>ndeveloped</b>	Expenses	Total
Proved - Cook Inlet	_		_		_	-	
FNI	\$ 108,169,312	\$	45,505,746	\$	422,335,438	\$ (94,233,000)	\$ 481,777,496
FNI @ 10%	\$ 75,596,359	\$	26,222,301	\$	267,256,594	\$ (57,103,397)	\$ 311,971,859
Probable - Cook Inlet							
FNI		\$	24,160,285	\$	313,927,312		\$ 338,087,597
FNI @ 10%		\$	17,047,756	\$	174,810,344		\$ 191,858,100
Possible - Cook Inlet							
FNI				\$	71,892,688		\$ 71,892,688
FNI @ 10%				\$	39,039,215		\$ 39,039,215
Proved - Appalachian							
region							
FNI	\$ 6,700,649	\$	507,591				\$ 7,208,240
FNI @ 10%	\$ 3,483,407	\$	223,913				\$ 3,707,320
Probable -							
Appalachian region							
FNI							
FNI @ 10%							
Possible -							
Appalachian region							
FNI				\$	1,513,434		\$ 1,513,434
FNI @ 10%				\$	1,059,364		\$ 1,059,364

At April 30, 2010 our standardized measure of discounted future net cash flows for proved reserves was \$315,679,195. The present value of future net pre-tax cash flows attributable to estimated net proved reserves, discounted at 10% per annum, ( PV-10 ) is a computation of the standardized measure of discounted future net cash flows on a pre-tax basis. The table below provides a reconciliation of PV-10 to the standardized measure of discounted future net cash flows at April 30, 2010. PV-10 may be considered a non-GAAP financial measure under the SEC s regulations. We believe PV-10 to be an important measure for evaluating the relative significance of our natural gas and oil properties. PV-10 is computed on the same basis as the standardized measure of discounted future net cash flows but without deducting income taxes. We further believe investors may utilize our PV-10 as a basis for comparison of the relative size and value of our reserves to other companies. However, PV-10 is not a substitute for the standardized measure. Our PV-10 measure and the standardized measure of discounted future net cash flows do not purport to present the fair value of our natural gas and oil reserves.

	April 30, 2010
Net present value of future cash flows, before income taxes for proved reserves	\$ 315,679,195

Future income taxes, discounted at 10% Standardized measure of discounted future net cash flows 137,320,450 178,358,745

\$

The following table presents our producing wells by operating area at April 30, 2010.

			Producing	g Wells			
		Gross (a)		Net (b)			
Location	Oil	Gas	Total	Oil	Gas	Total	
Cook Inlet	3	3	6	3	3	6	
Appalachian							
region	185	334	519	170	142	312	
Total	188	337	525	173	145	318	

(a)

The number of gross wells is the total number of wells in which a working interest is owned.

(b)

The number of net wells is the sum of fractional working interests we own in gross wells expressed as whole numbers and fractions thereof.

Our staff of professional geologists is responsible for identifying areas with potential for economic production of natural gas and oil. They utilize results from logs, seismic data and other tools to evaluate existing wells and to predict the location of economically attractive new natural gas and oil reserves. To further this process, we have collected and continue to collect logs, core data, production information and other raw data available from state and private agencies and other companies and individuals actively drilling in the regions being evaluated. From this information, the geologists develop models of the subsurface structures and formations that are used to predict areas for prospective economic development.

On the basis of these models, we obtain available natural gas and oil leaseholds, farm-outs and other development rights in these prospective areas. In most cases, to secure a lease, we pay a lease bonus and an annual rental payment, converting to a royalty upon initial production. In addition, overriding royalty payments may be granted to third parties in conjunction with the acquisition of drilling rights initially leased by others.

We believe that we hold good and defensible title to our developed properties, in accordance with standards generally accepted in the industry. As is customary in the industry, a preliminary title examination is conducted at the time the undeveloped properties are acquired. Prior to the commencement of drilling operations, a title examination is conducted and remedial work is performed with respect to discovered defects which we deem to be significant. Title examinations have been performed with respect to substantially all of our producing properties.

Certain of the properties we own are subject to royalty, overriding royalty and other outstanding interests customary to the industry. The properties may also be subject to additional burdens, liens or encumbrances customary to the industry, including items such as operating agreements, current taxes, development obligations under natural gas and oil leases, farm-out agreements and other restrictions. We do not believe that any of these burdens will materially interfere with the use of the properties.

The following table presents, by operating area, leased acres or acreage subject to the Susitna Basin Exploration License as of April 30, 2010.

	Developed	<b>Developed Acres</b>		ed Acres	Total Acres		
Project	Gross	Net	Gross	Net	Gross	Net	
Cook Inlet	34,996	32,800	529,825	523,422	564,821	556,222	

Appalachian						
region	13,680	9,408	51,594	37,916	65,274	47,324
Total acreage	48,676	42,208	581,419	561,338	630,095	603,546

,	-
	1
	1

The following table presents the net undeveloped acres that we control under fee leases and the Susitna Basin Exploration License and the period the leases and exploration license are scheduled to expire, absent pre-expiration drilling and production which extends the term of the lease(s) or the fulfillment of the exploration license terms which permits us to convert all or any portion of the exploration license into oil and gas leases. The expiration dates of the leases are subject to one year automatic renewals so long as we are producing oil and/or gas on the lease. The term of the Susitna Basin Exploration License expires in October 2010, subject to extension as described earlier in this annual report.

	Net Undeveloped Acres	
Lease/Exploration License	Year of Expiration	<b>Total Acres</b>
Cook Inlet		
MHT 9300062 - Olsen Creek	2010	4,857
MHT 9300063 - Olsen Creek	2010	3,906
ADL 390551 - Stingray	2012	520
ADL 390735 - Stingray	2013	2,047
ADL 390578 - N Ivan	2012	5,705
ADL 390585 - N Alexander	2012	5,689
ADL 390571 - Pretty Creek	2012	1,160
ADL 390749 - Otter	2013	2,522
ADL 390579 - Otter	2012	5,760
ADL 390370 - Raptor	2010	1,280
ADL 390379 - Raptor	2010	2,536
ADL 391108 - Raptor	2014	1,271
ADL 17595 - Raptor	1967, Held by Unit	1,235
ADL 17602 - Sabre	1967, Held by Unit	896
ADL 18758 - Sabre	1967, Held by Unit	280
ADL 390078 - Susitna Basin Exploration License	2010	471,474
ADL 390555 - Tutna	2012	1,280
ADL 390556 - Tutna	2012	2,522
ADL 390557 - Tazlina	2012	2,529
ADL 390549 - Cherryville	2012	2,560
ADL 17594	1967, Held by Unit	80
ADL 17597	1967, Held by Unit	2,280
ADL 18730	1967, Held by Unit	480
ADL 18777	1967, Held by Unit	553
Total		523,422
Appalachian region		
Lindsay		1,535
Edwards-Fowler, Gann		81
Butler et al		24
Gunsight		1,335
Phillips et al from Gunsight acreage		901
KTO acreage and wells		19,128
ETC acreage and wells		3,507
Baker-Senior lease farm out		3,220
Other Undeveloped, net		8,185
Total		37,916
Total acreage		561,338

The following table presents our development and exploratory drilling activities during the past three years. There is no correlation between the number of productive wells completed during any period and the aggregate reserves to those wells. Productive wells consist of producing wells capable of commercial production.

	Drilling Activities 2010 2009 2008						
	Gross	Net	Gross	Net	Gross	2000	Net
Development:							
Producing							
Cook Inlet							
Appalachian region						2	1
Total producing						2	1
Non-Producing							
Cook Inlet							
Appalachian region							
Total non-producing							
Injection Cook Inlet							
Appalachian region							
Total injection							
Dry							
Cook Inlet							
Appalachian region						1	5
Total dry							
Total development						3	6
Exploratory:							
Productive							
Cook Inlet							
Appalachian region							
Total productive							
Dry Carla Inlat							
Cook Inlet Appalachian region							
Total dry							
Pending determination							
Total exploratory							
Total drilling activity						3	6
Our current efforts are focused	on reworl	king certain of the	wells in Cook Ir	nlet and we de	o not curren	tly have	any
ongoing drilling operations in e							
Alaska as described elsewhere					-	-	, wells,
including fracturing, logging an							
specializing in those operations							
services used in the developme	·	<b>v</b> .	•••	• •			•
negotiate rates and costs for ser	vices and	supplies when con	unions muicale	mai such an	approach 18	warrante	<del>.</del>

#### Principal markets and principal customers

The existing markets for natural gas production in south central Alaska are the Tesoro Nikiski Refinery, utility companies, petrochemical manufacturing, the production of liquefied natural gas (LNG) for export to Alaskan or

Asian markets, and the production of synthetic crude oil ( syncrude ). Presently, the sole market for our crude oil produced at our Alaskan operations is the Tesoro Nikiski Refinery. Crude oil is shipped by pipeline and tanker vessel to the Tesoro Nikiski refinery, operated by Tesoro Alaska Petroleum Company. The main export pipeline is operated by the Cook Inlet Pipeline Company, which is operated by Chevron Pipelines and tanker vessels operate under contract to Tesoro.

As a result of the acquisition of the assets in December 2009, Cook Inlet is a successor to the September 2003 contract with Tesoro Refining and Marketing Company. Under the terms of this agreement, Tesoro has agreed to purchase all of the Alaskan Cook Inlet crude oil available at the Drift River Terminal which is produced from leases on the west side of Cook Inlet for a maximum annual capacity of the lesser of the average proportionate share of the Alaskan Cook Inlet crude oil produced of 40,000 barrels per day. The per barrel pricing is based upon the simple arithmetic average of the published daily New York Mercantile Exchange (NYMEX) settlement prices for light sweet crude oil less certain adjustments and deductions. This pricing may be modified upon the mutual agreement of the parties if the volume falls below 9,000 barrels per day or exceeds 24,000 per day. The initial term of the agreement was to December 31, 2008 and thereafter it has automatically renewed in additional one year terms. The agreement may be terminated by either party upon notice 60 days prior to the automatic renewal. All of our present and planned future oil production is from the west side of Cook Inlet, and would be covered by this contract. Sales to Tesoro Refining and Marking Company under this agreement represented approximately 82% of our total revenue in 2010.

As of July 25, 2010 all natural gas produced by Cook Inlet Energy was used by it to generate heat and power at its production facilities. At such time as gas production exceeds Cook Inlet Energy s internal needs, it can sell the excess production as all of Cook Inlet Energy s gas wells are connected to the south central Alaska Railbelt pipeline network through the Cook Inlet Gas Gathering System and/or the Beluga Pipeline, both of which are operated by Marathon Pipelines.

The principal markets for our crude oil and natural gas produced in the Appalachian region are refining companies, utility companies and private industry end users. Crude oil is stored in tanks at the well site until the purchaser retrieves it by tank truck. Direct purchases of our crude oil are made statewide at our well sites by Barrett Oil Purchasing Company. Our natural gas has multiple markets throughout the eastern United States through gas transmission lines. Access to these markets is presently provided by three companies in northeastern Tennessee, including Cumberland Valley Resources, NAMI Resources Company, and Tengasco. Local markets in Tennessee are served by Citizens Gas Utility District and the Powell Clinch Utility District. Natural gas is delivered to the purchaser via gathering lines into the main gas transmission line. Surplus gas is placed in storage facilities or transported to East Tennessee Natural Gas which serves Tennessee and Virginia. In 2010 and 2009, sales to Barrett Oil Purchasing and Sunoco, collectively, represented approximately 9% and approximately 24%, respectively, of our total revenues and sales to Cumberland Valley Resources, which purchases natural gas produced from a joint venture with Delta Producers, Inc., accounted for approximately 21% of our total revenue for 2009 and approximately4% of our total revenue in 2010.

The following table presents information regarding production volumes and revenues, average sales prices and costs, after deducting royalties and interests of others, with respect to oil and gas production attributable to our interest for the last three years. In the following table, average production cost are costs incurred to operate and maintain the wells and equipment and to pay the production costs, which does not include ad valoreum and severance taxes per unit of production, and is exclusive of work-over costs.

	Year Ended April 30,					
		2010		2009		2008
Oil production (Bbls)						
Cook Inlet						
Production		46,445				
Average sales price	\$	78.76				
Average production cost	\$	43.54				
Appalachian region						
Production		2,945		4,580		4,984
Average sales price	\$	71.33	\$	68.77	\$	77.25
Average production cost	\$	54.64	\$	52.49	\$	21.73

We must currently pay a \$14.57 per barrel Cook Inlet Pipe Line tariff to the Drift River Terminal and a minimum of 5% in royalties to the State of Alaska from any oil or gas sold from the West McArthur River Unit and the Redoubt Unit, although with increased production at the West McArthur River Unit these escalate to a maximum of 12.5%. The Redoubt Unit is scheduled to have an increased royalty rate of 12.5% in 2012, unless we can negotiate a reduced rate. We are currently in royalty discussions with the State of Alaska, Division of Natural Resources and hope there will be a reduction. We have also protested the increase in the Cook Inlet Pipe Line tariff, that was increased from \$4.06 per barrel to the current rate following the 2009 eruption of Mount Redoubt that

disrupted shipments though the Drift River Terminal. There are no assurances we will be successful in either appeal. We are also obligated to pay Cook Inlet Region, Inc. a 12.5% royalty on any gas sold from the portion of the West Foreland Gas Field located outside of the West McArthur River Unit. Finally, there are overriding royalty interests totaling approximately 12.4% for West McArthur River Unit, 4% for Redoubt Unit, and 5% for the portion of the West Foreland Gas Field located outside the West McArthur River Unit.

#### Our contract drilling operations

We provide land drilling services on a contract basis in the domestic market to customers that are primarily engaged in oil and natural gas exploration and production. The market that we serve is primarily the Appalachian Basin, which has unconventional natural gas bearing formations. Natural gas production from unconventional formations, including tight sands, shales and coalbed methane, is both the largest and fastest growing component of U.S. natural gas production. In addition to vertical drilling, we anticipate the need for horizontal drilling, which increases exposure of the wellbore to gas-bearing formations and provides better drainage. The horizontal drilling rigs contemplated for this are specially equipped for this type of work, as they typically require air circulation systems for penetrating through hard rock and enhanced fluid circulation systems for drilling horizontally into natural gas bearing formations. We plan to contract out the horizontal drilling operations in order to be able to provide this service.

Our services range from contract drilling by the foot or day rate to offering turnkey services to our customers. Our services are typically limited to the drilling portion of oil and gas extraction. Thus, when offering turnkey solutions, we will contract out the non-drilling functions such as possibly horizontal drilling and fracturing to non-affiliated third parties. We are responsible for the costs of rig refurbishment.

During calendar year 2010, in addition to providing drilling services for our other subsidiaries, we expect our wholly owned subsidiary Miller Drilling TN, LLC to drill, as needed, under contract in connection with our agreement to satisfy the two year drilling contract that it has with Atlas Energy. In addition, through our relationship with Atlas Energy, Miller Drilling has the opportunity to bid on other drilling or service work that Atlas Energy bids on in the State of Tennessee.

We are dependent on local customers for drilling revenues. Five customers, Atlas Energy Resources, LLC, Tri-Global Holdings, LLC, Montello Resources, LLC, Delta Producers Inc. and Herman Gettelfinger, a member of our Board of Directors, accounted for approximately 35% and approximately 94%, respectively, of our service and drilling revenue for 2010 and 2009.

#### Competition

Our oil and gas exploration activities in Alaska and Tennessee are undertaken in a highly competitive and speculative business environment. In seeking any other suitable oil and gas properties for acquisition, we compete with a number of other companies doing business in Alaska, Tennessee and elsewhere, including large oil and gas companies and other independent operators, many with greater financial resources than we have.

At the local level, as we seek to expand our lease holdings, we compete with several companies who are also seeking to acquire leases in the areas of the acreage which we have under lease. In Alaska, we have eight significant competitors including Aurora Gas, Bucaneer Alaska, Chevron, ConocoPhillips, Escopeta Oil, XTO, Linc Energy, and Marathon. However, we believe we have a competitive edge because we already have existing oil and gas production, facilities, infrastructure, and pipelines that connect us to the oil and gas markets as well as some of the lowest operating cost in the area. We believe that our existing Alaska oil and gas reserves and current leases with large acreage positions, enhance our competitors. In the Appalachian region, we have six significant competitors including Atlas Energy Resources, LLC, Consol Energy, Inc., Can Argo Energy Corporation, Champ Oil, John Henry Oil and Tengasco. These companies are in competition with us for oil and gas leases in known producing areas in

which we currently operate, as well as other potential areas of interest. We believe we can effectively complete for leases, however, as in the Appalachian region we have name recognition over 40 years, we are the largest operator of oil and gas wells in Tennessee and we have a staff of experienced, proven petroleum geologists and engineers that allows us to exploit the potential the Appalachian region provides.

1	1
I	I

Substantially most of our competitors have more capital, longer operating histories and significantly greater financial and operating resources than we do. Given the relative size of our operations and our limited capital there is no assurance we will ever effectively compete in the area of obtaining the most leases in our target areas.

#### **Government Regulation**

While the prices of oil and natural gas are set by the market, other aspects of our business and the industry in general are heavily regulated. The availability of a ready market for oil production and natural gas depends on several factors beyond our control. These factors include regulation of production, federal and state regulations governing environmental quality and pollution control, the amount of oil and natural gas available for sale, the availability of adequate pipeline and other transportation and processing facilities and the marketing of competitive fuels. State and federal regulations generally are intended to protect consumers from unfair treatment and oppressive control, to reduce the risk to the public and workers from the drilling, completion, production and transportation of oil and natural gas, to prevent waste of oil and natural gas, to protect rights among owners in a common reservoir and to control contamination of the environment. Pipelines are subject to the jurisdiction of various federal, state and local agencies.

Our exploration and production business is subject to various federal, state and local laws and regulations on the taxation of natural gas and oil, the development, production and marketing of natural gas and oil and environmental and safety matters. Many laws and regulations require drilling permits and govern the spacing of wells, rates of production, water discharge, prevention of waste and other matters. Prior to commencing drilling activities for a well, we must procure permits and/or approvals for the various stages of the drilling process from the applicable state and local agencies in the state in which the area to be drilled is located. The permits and approvals include those for the drilling of wells. Additionally, other regulated matters include the following:

bond requirements in order to drill or operate wells;

the location of wells;

.

•

the method of drilling and casing wells;

the surface use and restoration of well properties;

the plugging and abandoning of wells; and

the disposal of fluids.

The Regulatory Commission of Alaska regulates the intrastate pipeline tariffs and encompasses all pipelines Cook Inlet Energy ships through including the CIPL, CIGGS, and Beluga lines. The Regulatory Commission of Alaska must also review and approve most major gas sales contracts, and through this mechanism plays the dominant role in

determining gas pricing, since Alaska has no spot market for gas.

Alaska law requires that we obtain state permits for the drilling of wells and to post a \$200,000 bond with the Alaska Oil and Gas Conservation Commission. Injection wells are regulated by the Alaska Oil and Gas Conservation Commission and the United States Environmental Protection Agency (EPA). A \$490,000 abandonment escrow is established for two Class 1 non-hazardous injection wells for benefit of EPA. These funds are held by the First National Bank of Alaska in escrow accounts currently on deposit. There are also many additional obligations to landholders, which are the Alaska DNR, Cook Inlet Region, Inc. ( CIRI ), Salamatof Native Association and the United States Bureau of Land Management ( BLM ). The Alaska DNR requires \$600,000 in bonding to operate of Alaskan oil and gas leases, and as a condition of the bankruptcy sale has bound Cook Inlet Energy to funding an abandonment escrow over time based on production, not to exceed \$50 million, which includes abandonment expenses associated with onshore assets, principally the wells associated with the West McArthur River Unit, is not to exceed \$10 million, and the escrow for the offshore assets, principally for the closure of wells on the Osprey platform and the physical removal of the platform, is not to exceed \$40 million. We are presently negotiating the term and amount of these escrow accounts in an effort to significantly reduce the amounts to levels we believe are more reasonable with the actual costs. Cook Inlet Energy is also obligated to establish an abandonment escrow of \$1.5 million to cover future abandonment expenses related to the three West Foreland gas wells for benefit of the BLM and CIRI, \$500,000 of which has already been funded, and \$750,000 to establish an abandonment escrow for future abandonment expenses related to surface facilities and pipelines for benefit of CIRI and Salamatof Native Association. The BLM requires a bond of \$25,000. \$500,000 is due to Cook Inlet Region, Inc. in each of December 2010 and 2011 and \$250,000 is payable to Salamatof Native in May each of 2010, 2011 and 2012.

Under the Oil Pollution Act of 1990 Cook Inlet Energy is required to fund a citizens advisory group, the Cook Inlet Regional Citizen s Advisory Council, under which its commitment is approximately \$55,000 per year.

Tennessee law requires that we obtain state permits for the drilling of oil and gas wells and to post a bond with the Tennessee Gas and Oil Board to ensure that each well is reclaimed and properly plugged when it is abandoned. The reclamation bonds cost \$1,500 per well. The cost for the plugging bonds are \$2,000 per well or \$10,000 for ten wells. Currently, we have several of the \$10,000 plugging bonds. For most of the reclamation bonds, we have deposited a \$1,500 Certificate of Deposit with the Tennessee Gas and Oil Board.

Sales of natural gas in Tennessee are affected by intrastate and interstate gas transportation regulation. Beginning in 1985, the Federal Energy Regulatory Commission ("FERC"), which sets the rates and charges for transportation and sale of natural gas, adopted regulatory changes that have significantly altered the transportation and marketing of natural gas. The stated purpose of FERC's changes is to promote competition among the various sectors of the natural gas industry. In 1995, FERC implemented regulations generally grandfathering all previously approved interstate transportation rates and establishing an indexing system for those rates by which adjustments are made annually based on the rate of inflation, subject to certain conditions and limitations. These regulations may tend to increase the cost of transporting oil and natural gas by pipeline. Every five years, FERC will examine the relationship between the change in the applicable index and the actual cost changes experienced by the industry. We are not able to predict with certainty what effect, if any, these regulations will have on us.

The state and regulatory burden on the oil and natural gas industry generally increases our cost of doing business and affects our profitability. While we believe we are presently in compliance with all applicable federal, state and local laws, rules and regulations, continued compliance (or failure to comply) and future legislation may have an adverse impact on our present and contemplated business operations. Because such federal and state regulation are amended or reinterpreted frequently, we are unable to predict with certainty the future cost or impact of complying with these laws.

We are subject to various federal, state and local laws and regulations governing the protection of the environment, such as the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended (CERCLA), the Resource Conservation and Recovery Act (RCRA), the Clean Air Act and the Federal Water Pollution Control Act of 1972 (the "Clean Water Act"), which affect our operations and costs. In particular, our exploration, development and production operations, our activities in connection with storage and transportation of oil and other hydrocarbons and our use of facilities for treating, processing or otherwise handling hydrocarbons and related wastes may be subject to regulation under these and similar state legislation. These laws and regulations:

restrict the types, quantities and concentration of various substances that can be released into the environment in connection with drilling and production activities;

limit or prohibit drilling activities on certain lands lying within wilderness, wetlands and other protected areas; and

impose substantial liabilities for pollution resulting from our operations.

.

CERCLA, also known as "Superfund," imposes liability for response costs and damages to natural resources, without regard to fault or the legality of the original act, on some classes of persons that contributed to the release of a "hazardous substance" into the environment. These persons include the "owner" or "operator" of a disposal site and

entities that disposed or arranged for the disposal of the hazardous substances found at the site. CERCLA also authorizes the EPA and, in some instances, third parties to act in response to threats to the public health or the environment and to seek to recover from the responsible classes of persons the costs they incur. It is not uncommon for neighboring landowners and other third parties to file claims for personal injury and property damage allegedly caused by the hazardous substances released into the environment. In the course of our ordinary operations, we may generate waste that may fall within CERCLA's definition of a "hazardous substance." We may be jointly and severally liable under CERCLA or comparable state statutes for all or part of the costs required to clean up sites at which these wastes have been disposed.

We currently lease properties that for many years have been used for the exploration and production of oil and natural gas. Although we and our predecessors have used operating and disposal practices that were standard in the industry at the time, hydrocarbons or other wastes may have been disposed or released on, under or from the properties owned or leased by us or on, under or from other locations where these wastes have been taken for disposal. In addition, many of these properties have been operated by third parties whose actions with respect to the

treatment and disposal or release of hydrocarbons or other wastes were not under our control. These properties and wastes disposed on these properties may be subject to CERCLA and analogous state laws. Under these laws, we could be required to do the following:

remove or remediate previously disposed wastes, including wastes disposed or released by prior owners or operators,

•

clean up contaminated property, including contaminated groundwater; or to perform remedial operations to prevent future contamination, and/or

clean up contaminated property, including contaminated groundwater; or to perform remedial operations to prevent future contamination.

At this time, we do not believe that we are associated with any Superfund site and we have not been notified of any claim, liability or damages under CERCLA.

The RCRA is the principal federal statute governing the treatment, storage and disposal of hazardous wastes. RCRA imposes stringent operating requirements and liability for failure to meet such requirements on a person who is either a "generator" or "transporter" of hazardous waste or an "owner" or "operator" of a hazardous waste treatment, storage or disposal facility. At present, RCRA includes a statutory exemption that allows most oil and natural gas exploration and production waste to be classified as nonhazardous waste. A similar exemption is contained in many of the state counterparts to RCRA. As a result, we are not required to comply with a substantial portion of RCRA's requirements because our operations generate minimal quantities of hazardous wastes. At various times in the past, proposals have been made to amend RCRA to rescind the exemption that excludes oil and natural gas exploration wastes from regulation as hazardous waste. Repeal or modification of the exemption by administrative, legislative or judicial process, or modification of similar exemptions in applicable state statutes, would increase the volume of hazardous waste we are required to manage and dispose of and would cause us to incur increased operating expenses.

The Clean Water Act imposes restrictions and controls on the discharge of produced waters and other wastes into navigable waters. Permits must be obtained to discharge pollutants into state and federal waters and to conduct construction activities in waters and wetlands. The Clean Water Act requires us to construct a fresh water containment barrier between the surface of each drilling site and the underlying water table. This involves the insertion of a seven-inch diameter steel casing into each well, with cement on the outside of the casing. The cost of compliance with this environmental regulation is approximately \$10,000 per well. Certain state regulations and the general permits issued under the Federal National Pollutant Discharge Elimination System program prohibit the discharge of produced waters and sand, drilling fluids, drill cuttings and certain other substances related to the oil and natural gas industry into certain coastal and offshore waters. Further, the EPA has adopted regulations requiring certain oil and natural gas exploration and production facilities to obtain permits for storm water discharges. Costs may be associated with the treatment of wastewater or developing and implementing storm water pollution prevention plans.

The Clean Water Act and comparable state statutes provide for civil, criminal and administrative penalties for unauthorized discharges for oil and other pollutants and impose liability on parties responsible for those discharges for the costs of cleaning up any environmental damage caused by the release and for natural resource damages resulting from the release. We believe that our operations comply in all material respects with the requirements of the Clean Water Act and state statutes enacted to control water pollution.

Our operations are also subject to laws and regulations requiring removal and cleanup of environmental damages under certain circumstances. Laws and regulations protecting the environment have generally become more stringent in recent years, and may in certain circumstances impose "strict liability," rendering a corporation liable for environmental damages without regard to negligence or fault on the part of such corporation. Such laws and regulations may expose us to liability for the conduct of operations or conditions caused by others, or for acts which may have been in compliance with all applicable laws at the time such acts were performed. The modification of existing laws or regulations or the adoption of new laws or regulations relating to environmental matters could have a material adverse effect on our operations.

In addition, our existing and proposed operations could result in liability for fires, blowouts, oil spills, discharge of hazardous materials into surface and subsurface aquifers and other environmental damage, any one of which could result in personal injury, loss of life, property damage or destruction or suspension of operations. We

have an Emergency Action and Environmental Response Policy Program in place. This program details the appropriate response to any emergency that management believes to be possible in our area of operations. We believe we are presently in compliance with all applicable federal and state environmental laws, rules and regulations; however, continued compliance (or failure to comply) and future legislation may have an adverse impact on our present and contemplated business operations.

### Consultants

In February 2010, we entered into a one year consulting agreement with Tyler Energy Consulting Group to provide certain investor relation services to us. Under the terms of this agreement issued 250,000 shares of our common stock valued at \$487,500 as compensation. The agreement also provided that at our discretion we could issue up to an additional 250,000 shares of our common stock to the firm as additional compensation. On March 29, 2010 we issued the consultant an additional 70,000 shares of our common stock valued at \$462,000 and on April 5, 2010 we issued the consultant an additional 110,000 shares of our common stock valued at \$660,000 under the terms of this agreement. We granted Tyler Energy Consulting Group piggy back registration rights covering these shares and agreed to reimburse the firm for business related expenses. The agreement contains customary indemnification provisions.

We have entered into two agreements with Bristol Capital, LLC, an affiliate of Bristol Capital Advisors, LLC which is the investment advisor to Bristol Investment Fund, Ltd., an investor in the 2010 Offering, including the following:

On March 12, 2010 we entered into a one year consulting agreement with Bristol Capital, LLC under which it agreed to assist us with strategic planning, management and business operations, introductions to further our business goals, advice and services regarding our growth initiative and other similar services we might request. As compensation for these services we granted Bristol Capital, LLC a five year warrant to purchase 300,000 shares of our common stock at an exercise price of \$2.50 and five year options to purchase an additional 300,000 shares of our common stock at an exercise price of \$2.50 per share. However, Bristol Capital, LLC will only be entitled to exercise the warrant or the option, not both, which will result in a total possible issuance of 300,000 shares of our common stock. These securities were valued at \$1,169,293. In the event we issue securities at a price per share which is lower than \$2.50 we are required to increase the number of shares underlying the warrant and option. Under the terms of this agreement, we agreed to include the shares underlying the warrant or option in the next registration statement we filed with the SEC, depending upon the form of registration statement is utilized. Upon the effectiveness of such registration statement, the option will automatically terminate. In the event we fail to file a registration statement covering the shares of common stock underlying the warrant by September 10, 2010, Bristol Capital, LLC may elect to terminate either security in its discretion. In any event, Bristol Capital, LLC shall only retain either the warrant or the option, but not both. Bristol Capital, LLC is contractually limited under the terms of this consulting agreement so that its beneficial ownership of our common stock cannot exceed 9.9% of our outstanding shares. It may waive this limitation upon 61 days notice to us.

We agreed to pay all out-of-pocket expenses incurred by Bristol Capital, LLC under this agreement, subject to our prior approval. The agreement also contains customary indemnification and confidentially provisions.

On April 26, 2010 we entered into a finder s agreement with Bristol Capital, LLC pursuant to which on our behalf it has commenced preliminary discussions with two parties regarding our possible acquisition of certain specified assets. We agreed that if within two years from the date of the agreement we should enter into a definitive agreement with either named party or any of their affiliates for the acquisition of these assets, we will pay Bristol Capital, LLC a finder s fee of either an assignment of 5% of the interest in the assets or shares of our common stock valued at 5% of

the aggregate purchase price at its election. We also agreed to pay a fee equal to 5% of the transaction value. If the efforts by Bristol Capital, LLC on our behalf should result in a joint venture or similar partnership related to these assets, we agreed to pay it a finder s fee of 5% of the anticipated economic value of such an agreement.

### Employees

At July 25, 2010 we had 45 full time and two part-time employees.

## **Our history**

We were incorporated in the State of Delaware in November 1985 originally under the name Longhorn Development Company, Inc. for the purpose of searching out and acquiring or participating in a business or business opportunity. In August 1988 we changed our name to Single Chip Systems International, Inc. In August 1988 we acquired all of the issued and outstanding securities of Single Chip Systems, Inc., a California corporation, in exchange for shares of our common stock. Our then current officers and directors resigned and the officers and directors of Single Chip Systems, Inc. were appointed officers and directors of our company. Prior thereto, on July 1, 1988, Single Chip Systems, Inc. had entered into a technology utilization license agreement with Ramtron International Corporation which granted Single Chip Systems, Inc. the royalty-bearing, non-exclusive licenses to use the ferroelectric technologies and the certain trademarks in production, manufacture and sales of Single Chip Systems, Inc. products. We failed to receive any economic benefit related to the license agreement and we recorded a \$100,000 loss on the license agreement in the period ended December 31, 1988.

Thereafter, we had no business or operations until the transaction in January 1997 as hereinafter described. In May 1996 we changed our name to Triple Chip Systems, Inc.

Mr. Deloy Miller formed Miller Petroleum, Inc. ( pre-merger Miller ), a company which is the basis of our current operations, in January 1978. In January 1997, we closed an Agreement and Plan of Reorganization with pre-merger Miller whereby we issued 5,582,535 shares of our common stock in exchange for all of the outstanding common stock of per-merger Miller. The acquisition was accounted for as a recapitalization of our company because the shareholders of pre-merger Miller controlled the company after the acquisition. Following the transaction, in January 1997, pre-merger Miller was merged into our company and we changed our name to Miller Petroleum, Inc. in conjunction with the redomestication of our company into the State of Tennessee.

Effective as of June 13, 2008 we entered into an agreement with Atlas Energy Resources, LLC under which we assigned it:

an unencumbered, undivided 100% working interest and an 80% net revenue interest in and to the oil and gas lease comprising 27,620 acres known as Koppers North and Koppers South and located in Campbell County, Tennessee; and an unencumbered, undivided 100% interest and an 82.5% net revenue interest (net of a 5% overriding royalty interest to us) in and to the oil and gas lease comprising 1,952 acres adjacent to Koppers North and Koppers South and located in Campbell County, Tennessee; and

.

.

an unencumbered, undivided 100% working interest and an 80% net revenue interest in eight gas wells on Koppers South. We have the option to repurchase the wells within one year form the closing date or within 30 days after the pipeline to be built by Atlas Energy has been completed and is ready to accept gas for transport.

The transaction was subject to unwinding pursuant to a pending litigation between our company and CNX Gas Company LLC as disclosed in Item 3. Legal Proceedings. Transferring any of the leases or any interest thereon was also subject to a 60-day standstill period which has since expired. The aggregate consideration for the assignment of the leases and wells to Atlas Energy was \$19,625,000, \$9,025,000 of which was paid us and the remaining \$10,600,000 of which was paid directly to Wind City Oil & Gas, LLC in consideration of a settlement of claims between Wind City and our company described below.

As part of the transaction, we also agreed to contract with Atlas Energy for two rigs for two years to drill wells, commencing a significant commitment to contract drilling. To give Atlas Energy the level of service required, during

the first quarter of fiscal 2009 we acquired a 2007 COPCO Model RD III drilling rig and related equipment drilling rig from Atlas to assist in drilling the wells. For two years after the closing date, Atlas Energy granted us the opportunity to bid on any other drilling or service work that Atlas Energy bids on in the State of Tennessee. In addition, we entered into:

•

a natural gas transportation agreement with Atlas Energy which provides us access to the Atlas Volunteer Pipeline, to the extent that capacity is available, on substantially the same terms as those offered to the producers delivering into the system; and

a natural gas processing agreement pursuant to which Atlas Energy will provide gas processing services to us on substantially the same terms as those services are provided to other producers delivering gas into the Atlas Volunteer Pipeline and deliver back to us gas with a heating value of 1,100 BTUs per cubic foot.

Effective as of June 13, 2008, we also settled all issues and controversies with Wind City Oil & Gas, LLC, Wind Mill Oil & Gas, LLC and Wind City Oil & Gas Management, LLC. Under the terms of the settlement, we paid Wind City \$10,600,000 for the re-purchase of 2,900,000 shares of our common stock and reacquisition of all leases previously assigned by us to Wind City or the related parties, all wells and equipment associated with these leases, all pipeline rights and rights of way, all contract rights, and all other equipment, property and real property rights. As set forth above, we used a portion of the proceeds from the Atlas Energy transaction to pay the settlement amounts.

On June 8, 2009 we acquired certain assets from Ky-Tenn Oil, Inc., a Kentucky corporation ("KTO"), an unrelated third party, including KTO's:

undivided interest in approximately 170 oil and gas wells in Morgan, Scott and Fentress counties in Tennessee, together with all property, fixtures and improvements, leasehold interest and contract rights related to these wells;

undivided interest in approximately 35,325 acres of oil and gas leases in Scott and Morgan counties, Tennessee;

interest in an operating agreement with the Tennessee State Energy Development Partnership;

interest in a gas gathering pipeline system; and

other rights related to these assets, including royalty and working interests, licenses, permits, and similar incidental rights.

As consideration for these assets we issued KTO 1,000,000 shares of our common stock valued at \$320,000 and we granted the seller piggy-back registration rights covering these shares. Pursuant to this FASB guidance, we originally valued these assets at \$252,455 and recorded a loss on the transaction of \$67,545. Subsequently, we completed the determination of the value of all undeveloped reserves for this acreage during the quarter ended October 31, 2009 and accordingly we recorded an additional gain of \$1,057,564 on this transaction.

On June 18, 2009 we acquired 100% of the stock of East Tennessee Consultants, Inc., a Tennessee corporation ("ETC") and 100% of the membership interests in East Tennessee Consultants II, LLC, a Tennessee limited liability company ("LLC") from the owners of these entities. Pursuant to FASB ASC 805-10, we have valued these companies at \$1,862,369 and have recorded a gain on the transaction of \$828,745. As consideration for these companies we issued the sellers, who were unrelated third parties, 1,000,000 shares of our common stock valued at \$250,000. We granted the sellers registration rights covering these shares.

Under the terms of the stock purchase agreement, the sellers agreed not to engage in oil and gas operations for a period of three years following the closing date. We also agreed that each of the sellers, Messrs. Eugene D. Lockyear, Douglas G. Melton and Jerry G. Southwood, would continue their employment with the acquired companies for at least three years from the closing date of the transaction at their same compensation and benefit levels to which they were entitled in May 2009. In addition, Mr. Lockyear was appointed Vice President of Operations of our company. We also agreed that if any or all of the sellers incur any income tax liability as a result of the receipt of the above shares as consideration for the stock purchase, we agreed to pay a bonus to such seller equal to the amount of his tax liability within 30 days from the request of the sellers.

On December 10, 2009, Cook Inlet Energy acquired former Alaskan assets of Pacific Energy Resources valued at more than \$479 million through a Delaware Chapter 11 bankruptcy proceeding, including oil and gas assets which include onshore and offshore production facilities, \$215 million in proven energy reserves, \$122 million in probable energy reserves and \$31 million in possible energy reserves, providing total reserves of \$368 million. The purchased assets included the West McArthur River oil field, the West Foreland natural gas field, and the Redoubt unit with the Osprey offshore platform, all located along the west side of the Cook Inlet. Also included in the asset purchased were 602,000 acres of oil and gas leases which includes 471,474 acres under the Susitna Basin Exploration License as well as completed 3D seismic geology and other production facilities, together with:

all easements, wells and tangible assets,

.

all oil and gas or proceeds from the sale of oil and gas produced in connection with the acquired assets from the closing date,

all contracts, unitization, communization and pooling declarations, orders and agreements,

all permits, records, royalty interests, partnership and joint venture interests,

to the extent assignable, all rights to indemnities,

all leases for real property used by the seller in connection with the operation of the acquired assets,

escrow accounts and bonds deposited with government entities with respect to the acquired assets,

all surety bonds, plugging bonds, abandonment bonds, standby trust agreements, escrow accounts for plugging, abandonment, decommissioning, removal and restoration obligations, together with security deposits,

all imbalances owed to the sellers by a third party at the closing, as well as all oil and gas in pipelines and tanks or held by or for the account of the sellers related to the assets acquired, and

•

.

•

•

the portion of the seller s Royalty Distributors Inc. s account relating to post-petition suspended royalties.

In this transaction, Cook Inlet Energy assumed certain liabilities related to the acquired assets, including:

•

all liabilities associated with or arising out of the ownership of, or operation of, the assets after the closing date,

all environmental liabilities with respect to the acquired assets,

all accounts payable from the closing date,

.

.

.

all royalty obligations associated with or related to the acquired assets after the closing date,

all claims arising out of the ownership or operation of the acquired assets after the closing date,

all plugging, abandonment, decommissioning, removal and/or restoration liabilities associated with or arising from the acquired assets with respect to all periods prior to, on or after the closing date,

permitted encumbrances and imbalances owed by the sellers to third parties, and

post-petition suspended royalties maintained by Royalty Distributors Inc.

At closing we paid Pacific Energy Resources \$2.25 million and provided \$2.22 million for bonds, contract cure payments and other federal and State of Alaska requirements to operate the facilities. Under the terms of the purchase agreement, Donkel Oil & Gas, LLC was granted a one-half of 1% overriding royalty interest in the oil and gas leases acquired by Cook Inlet Energy, a 1% overriding royalty interest on Pacific Energy s working interest in all exploration oil and gas leases acquired by Cook Inlet Energy in the transaction, and a 5/10th of 1% overriding royalty interest owned by Pacific Energy on the leases that comprise the Cosmopolitan Unit and Falls Creek. In addition, Donkel Oil & Gas, LLC received a 1% overriding royalty interest on Cook Inlet Energy s working interest in any oil and gas lease which arises from certain properties included in an exploration license, which includes one lease, together with a 1% overriding royalty on our working interest in two additional oil and gas leases.

On December 10, 2009, we acquired 100% of the membership interests in Cook Inlet Energy, an Alaska limited liability company from the owners of this entity. As consideration for these companies we issued the sellers, who were unrelated third parties, four year stock warrants to purchase 3,500,000 shares of our common stock at exercise prices ranging from \$0.01 to \$2.00 per share. In addition, we are required to deliver \$250,000 in cash to satisfy certain expenses as well as reimbursement for reasonable out of pocket expenses; such payment has not yet been made. Under the terms of the stock purchase agreement, the sellers agreed not to engage in non-company related oil and gas operations for a period of three years following the closing date. We also agreed that each of the sellers, Messrs. David M. Hall, Walter J. Wilcox II and Troy Stafford, would continue their employment with the acquired company for at least three years from the closing date of the transaction at their specifically defined compensation and benefit levels. In addition, Mr. Hall was appointed as a member of our Board of Directors and as Chief Executive Officer of Cook Inlet Energy. Subsequent to April 30, 2010, Mr. Stafford has left the employ of Cook Inlet Energy.

### Miller Energy Income 2009-A, L.P. Offering

In 2009 we formed both Miller Energy GP and MEI. MEI was organized to provide the capital required to invest in various types of oil and gas ventures including the acquisition of oil and gas leases, royalty interests, overriding royalty interests, working interests, mineral interests, real estate, producing and non-producing wells, reserves, oil and gas related equipment including transportation lines and potential investments in entities that invest in such assets except for other investment partnerships sponsored by affiliates of MEI.

Between August 2009 and April 2010 MEI sold 61.35 units of securities to 23 accredited investors in transactions exempt from registration under the Securities Act of 1933 in reliance on exemptions provided by Section 4(2) and Regulation D of that act. Each unit consisted of a \$50,000 limited partnership interest in MEI, together with 25,000 shares of our common stock and a five year warrant to purchase an additional 25,000 shares of our common stock with an exercise price of \$1.00 per share. In order to receive our securities as part of the offering, investors in the MEI were required to purchase at least one unit. We issued a total of 1,329,250 shares of our common stock and common stock purchase warrants to purchase an additional 1,329,250 shares of our common stock.

MEI received \$3,067,500 in proceeds from this offering. It paid selling commissions of 7% on the sale of certain units (an aggregate of \$115,780) and a non-accountable marketing and due diligence allowance of 1% of the proceeds received from certain units sold in the offering (an aggregate of \$16,540), to Dimirak Securities Corporation, a related party, and Newbridge Securities Corporation, both broker-dealers and members of FINRA. Mr. Scott M. Boruff, our CEO, is a director and 49% owner of Dimirak Securities Corporation. MEI also paid a wholesaling commission based on 2% of the gross proceeds (an aggregate of \$33,080) to several firms including Newbridge Securities Corporation, Empire Securities Corporation and Arque Capital Ltd. MEI paid a dealer management fee of 1% of the gross offering proceeds (an aggregate of \$30,540) to several parties including Dimirak Securities Corporation, its affiliate Dimirak Financial Corporation and Empire Securities Corporation. Finally, Dimirak Securities Corporation received an additional \$30,675 in other fees which was based on 1% of investor capital and two individuals received a total of \$81,000 for referral fees. Purchasers of these securities have been granted piggy back registration rights covering the shares of our common stock.

## ITEM 1A.

#### **RISK FACTORS.**

An investment in our common stock involves a significant degree of risk. You should not invest in our common stock unless you can afford to lose your entire investment. You should consider carefully the following risk factors and other information in this annual report before deciding to invest in our common stock.

#### **Risks Related to Our Business**

# We have a history of operating losses and our net income in 2010 and 2009 are both the result of one-time acquisition gains. Our revenues are not currently sufficient to fund our operating expense and there are no assurances we will develop profitable operations.

We reported an operating loss of approximately \$11.1 million in 2010 and approximately \$3.2 million in 2009. Our net income of approximately \$249.5 million in 2010 is attributable to the approximate \$461.1 million gains on the acquisitions of the Alaska and Tennessee assets. Our net income of approximately \$8.4 million in 2009 is attributable to the approximate \$11.7 million one-time gain on the sale of assets in Tennessee. While our revenues increased approximately \$4.3 million in 2010 from 2009, our operating expenses increased by approximately \$12.2 million from year to year. Approximately 31% of this increase in 2010 is associated with one-time non cash expenses and 19% is associated with ongoing non cash expenses. In addition, we did not begin reporting revenues from our Alaskan assets until January 2010. However, as a result of the significant expansion of our business during 2010 our operating expenses presently far exceed our revenues. We anticipate that our operating expenses will continue to increase as we fully develop our operations following the acquisition of the Alaskan assets and expect an increase in our revenues as well. However, until such time as we are able to significantly increase our revenues, we will continue depleting our cash resources to fund our operating expenses. We believe our present working capital is sufficient to fund our operations for the foreseeable future, but we may have to reduce or cease our expansion efforts if we have not seen an increase in revenues in the next few months. While we believe that we will successful in increasing our revenues to a level which will pay our operating expenses, if we are not successful we will need to raise additional capital until such time as our revenues are sufficiently increased. As described below, our ability to raise additional capital could be adversely impacted by the terms of our March 2010 unit offering. If we are unable to raise additional capital as necessary to fund our operating expenses, our ability to continue to grow our company would be hampered and we could be forced to curtail development of some or all of our assets until such time, if ever, that we are able to raise the needed capital.

# Our development and exploration operations require substantial capital, and we may be unable to obtain needed capital or financing on satisfactory terms, which would prevent us from fully developing our business and substantially increasing our revenues.

The oil and gas industry is capital intensive and we anticipate in addition to funds necessary to fund our operating expenses while we continue to develop the Alaskan assets, that we will need to raise between \$75 million and \$100 million to meet our funding commitments under the Assignment Oversight Agreement with the State of Alaska DNR and to fully develop our Tennessee and Alaskan reserves. We intend to seek this additional capital through the sale of equity or debt instruments; however, the terms of the March 2010 unit offering contain restrictive covenants that prevent us from raising any additional equity capital for six months following the effective date of a registration statement we are obligated to file with the SEC registering the securities sold in this offering and which contain additional covenants which may impede our capital raising activities in future periods. We may not be able to obtain this necessary equity or debt financing on terms favorable to us, if at all. If we are unable to raise the capital as needed, the State of Alaska could terminate all of the Cook Inlet leases which would be a material adverse event to our company and we would be unable to fully develop our Alaskan reserves which would materially impact our ability to increase our revenues in future periods. To the extent such funds are not available from any of those sources,

our operations and activities will be limited to those operations and activities we can afford with the funds then available to us. The failure to obtain additional financing could also result in a curtailment of our operations relating to exploration and development of our prospects.

# We have failed to timely file a registration statement related to our March 2010 unit offering and are subject to registration rights penalties which are payable in cash.

Under the terms of our March 2010 unit offering we were required to file a registration statement with the SEC registering for resale the shares of common stock sold in the offering, including those underlying the warrants included in the units, by April 15, 2010. We also agreed to use our best efforts to cause the registration statement to

be declared effective by the SEC within 90 days from the filing date or 120 days if the registration statement should be selected for a full review by the staff of the SEC. The registration rights agreement provides that if we failed to timely file the registration statement, or if it should not be declared effective within the prescribed time, we are subject to liquidated damages payable in cash equal to 2% of the aggregate purchase price of the securities up to a maximum of 12% of the total proceeds of the offering. We have yet to file the registration statement. Because we did not timely file the registration statement, during the fourth quarter of 2010 we accrued registration rights penalties of \$602,040 which is payable in cash to the investors in that offering.

### Cook Inlet Energy s operations are subject to oversight by the Alaska DNR and the Cook Inlet Energy oil and gas leases could be terminated if it fails to uphold the terms of the Assignment Oversight Agreement. If these leases were terminated, we would be unable to continue our operations as they are presently conducted and could be liable for significant additional costs associated with decommissioning the Osprey platform.

Cook Inlet Energy is a party to an Assignment Oversight Agreement with the Alaska DNR that was entered into in November 2009 as a condition of the sale of the oil and gas leases from Pacific Energy Resources to Cook Inlet Energy. The agreement states that its intent is to ensure that there were sufficient funds, and that those funds are only spent, to fulfill Cook Inlet Energy s initial development, operation, and dismantlement obligations under the assigned leases, applicable statutes and regulations. Those commitments include approximately \$5.15 million to be used by Cook Inlet Energy to restore base production at the West McArthur River Unit Facility and approximately \$31 million to support restoration of base production at the Redoubt Unit. Internally we have increased that requirement to \$35 million to accommodate the purchase of a drilling rig for the Osprey platform. During the interim period, Cook Inlet Energy has fulfilled its commitment to restore base production from the West McArthur River Unit. However, we remain obligated to restore base production from the Redoubt Unit. We will need to either close the funding commitment from Vulcan Capital Corporation or raise the necessary capital from third parties. As a result of the restrictive covenants of the March 2010 unit offering, our ability to raise additional capital is limited. If we are unable to raise the capital as necessary to fully fund the commitment made in connection with the Assignment Oversight Agreement, the State of Alaska could terminate all of the Cook Inlet leases which would be a material adverse event to our company. Our operations would be limited to the Appalachian region and it is unlikely we would be able to continue our operations as they are presently conducted. In addition, we would then be obligated to fund the decommissioning and abandonment of the Osprey platform, which could cost as much as \$40 million, of which only approximately \$6.6 million has been escrowed with Alaska.

# Oil and gas prices fluctuate due to a number of uncontrollable factors, creating a component of uncertainty in our development plans and overall operations. Declines in prices adversely affect our financial results and rate of growth in proved reserves and production.

Oil and gas markets are very volatile, and we cannot predict future oil and natural gas prices. The prices we receive for our oil and natural gas production heavily influence our revenue, profitability, access to capital and future rate of growth. The prices we receive for our production depend on numerous factors beyond our control. These factors include, but are not limited to, changes in global supply and demand for oil and gas, the actions of the Organization of Petroleum Exporting Countries, the level of global oil and gas exploration and production activity, weather conditions, technological advances affecting energy consumption, domestic and foreign governmental regulations and tax policies, proximity and capacity of oil and gas pipelines and other transportation facilities and transportation costs and the price and technological advancement of alternative fuels.

The downward pressure in oil and natural gas prices that began in the last half of 2008 continued in 2010. The average realized gas price per thousand standard cubic feet (Mscf) for 2010 decreased 40% from 2009 and the average realized oil price per barrel for 2010 decreased 9% from 2009. The decrease in prices significantly decreased the amount available to invest in exploration and development drilling and the present value of our proved reserves. Our proved oil and gas reserves and production volumes decrease in quantity unless we successfully replace the reserves we produce with new discoveries or acquisitions. For the foreseeable future, we expect to make substantial capital

investments for the exploration and development of new oil and gas reserves to replace the reserves we produce, to increase our daily production and to increase our total proved reserves. It will be necessary for us to raise additional capital to fund these expenditures. Low prices also reduce the amount of oil and gas that we can economically produce and may cause us to curtail, delay or defer certain exploration and development projects.

# The global economic crisis may have impacts on our business and financial condition that we currently cannot predict.

The continued credit crisis and related turmoil in the global financial system may have an impact on our business and our financial condition, and we may face challenges if conditions in the financial markets do not improve. Our ability to access the capital markets may be restricted at a time when we would like, or need, to raise financing, which could have an impact on our flexibility to react to changing economic and business conditions. The economic situation could have an impact on potential lenders, purchasers of our oil and gas production and working interest owners in properties we operate, causing them to fail to meet their obligations to us.

# Estimates of oil and natural gas reserves are inherently imprecise. Any material inaccuracies in these reserve estimates or underlying assumptions will affect materially the quantities and present value of our reserves.

Estimates of proved oil and natural gas reserves and the future net cash flows attributable to those reserves are prepared by independent petroleum engineers and geologists. There are numerous uncertainties inherent in estimating quantities of proved oil and natural gas reserves and cash flows attributable to such reserves, including factors beyond our control and that of our engineers. Reserve engineering is a subjective process of estimating underground accumulations of oil and natural gas that cannot be measured in an exact manner. Different reserve engineers may make different estimates of reserves and cash flows based on the same available data. The accuracy of an estimate of quantities of reserves, or of cash flows attributable to such reserves, is a function of the available data, assumptions regarding future oil and natural gas prices and expenditures for future development drilling and exploration activities, and of engineering and geological interpretation and judgment. Additionally, reserves and future cash flows may be subject to material downward or upward revisions, based upon production history, development drilling and exploration activities and prices of oil and natural gas. Actual future production, revenue, taxes, development drilling expenditures, operating expenses, underlying information, quantities of recoverable reserves and the value of cash flows from such reserves may vary significantly from the assumptions and underlying information set forth herein.

# Approximately 74% of our total estimated proved reserves at April 30, 2010 were proved undeveloped reserves. In addition, there are no assurances that probable and possible reserves will be converted to proved reserves.

Recovery of proved undeveloped reserves requires significant capital expenditures and successful drilling operations. The reserve data included in the reserve engineer reports assumes that substantial capital expenditures are required to develop such reserves. Although cost and reserve estimates attributable to our natural gas and crude oil reserves have been prepared in accordance with industry standards, we cannot be sure that the estimated costs are accurate, that development will occur as scheduled or that the results of such development will be as estimated. We also have a significant amount of probable and possible reserves are more likely to be produced than probable reserves and probable reserves are more likely to be produced than possible reserves. There are no assurances that we can develop probable or possible reserves into probable reserves, or that if developed, probable reserves will become producing reserves to the level of the estimates.

#### There are no assurances that we will be able to extend the Susitna Basin Exploration License.

Included in the Alaskan assets we acquired is the Susitna Basin Exploration License granted by the State of Alaska covering approximately 471,474 acres which expires in October 2010. The acreage which is the subject of this exploration license represents approximately 84% of our net undeveloped acreage at April 30, 2010. Under the terms of the exploration license, providing that the work commitment of approximately \$3.5 million was fulfilled, during the exploration license term the license has the right to convert the license for all or a portion of the acreage into oil and gas leases. This original work commitment was met by the prior licensee and we presently have the right to convert the license into leases. Once the exploration license is converted into oil and gas leases, we are required to pay a per

acre fee to the state and commence drilling operations within specified timeframes. In an effort to control the timing of the development of this acreage, in April 2010 we requested a three year extension of the exploration license for a work commitment of \$750,000. While we reasonably believe the state will grant our request for extension, there are no assurances we are correct, or that if granted, that the terms and conditions of the extension will be satisfactory to us. If we are unable to negotiate an extension, it is likely we will convert the license for only a

portion of the land into oil and gas leases. The loss of the remaining rights would reduce the acreage which we could develop into proved, producing reserves.

# The present value of future net cash flows from our proved reserves will not necessarily be the same as the current market value of our estimated natural gas, crude oil and natural gas liquids reserves.

You should not assume that the present value of future net revenues from our proved reserves referred to in this annual report is the current market value of our estimated natural gas, crude oil and natural gas liquids reserves. In accordance with SEC requirements, the estimated discounted future net cash flows from our proved reserves are based on prices and costs on the date of the estimate, held flat for the life of the properties. Actual future prices and costs may differ materially from those used in the present value estimate. Actual future net cash flows will also be affected by increases or decreases in consumption by oil and gas purchasers and changes in governmental regulations or taxation. The timing of both the production and the incurrence of expenses in connection with the development and production of oil and gas properties affects the timing of actual future net cash flows from proved reserves. In addition, the 10% discount factor, which is required by the SEC to be used in calculating discounted future net cash flows for reporting purposes, is not necessarily the most appropriate discount factor. The effective interest rate at various times and the risks associated with our business or the oil and gas industry in general will affect the accuracy of the 10% discount factor.

### Our industry is subject to extensive environmental regulation that may limit our operations and negatively impact our production. As a result of increased enforcement of existing regulations and potential new regulations following the Gulf oil spill, the costs for complying with government regulation could increase.

Extensive Federal, state, and local environmental laws and regulations in the United States affect all of our operations. Environmental laws to which we are subject in the U.S. include, but are not limited to, the Clean Air Act and comparable state laws that impose obligations related to air emissions, the Resource Conservation and Recovery Act of 1976 (RCRA), and comparable state laws that impose requirements for the handling, storage, treatment or disposal of solid and hazardous waste from our facilities, the Comprehensive Environmental Response, Compensation and Liability Act (CERCLA) and comparable state laws that regulate the cleanup of hazardous substances that may have been released at properties currently or previously owned or operated by us or at locations to which our hazardous substances have been transported for disposal, and the Clean Water Act, and comparable state laws that regulate discharges of wastewater from our facilities to state and federal waters. Failure to comply with these laws and regulations or newly adopted laws or regulations may trigger a variety of administrative, civil and criminal enforcement measures, including the assessment of monetary penalties, the imposition of remedial requirements, and the issuance of orders enjoining future operations or imposing additional compliance requirements on such operations. Certain environmental laws, including CERCLA and analogous state laws, impose strict, joint and several liability for costs required to clean up and restore sites where hazardous substances or hydrocarbons have been disposed or otherwise released. Moreover, it is not uncommon for neighboring landowners and other third parties to file claims for personal injury and property damage allegedly caused by the release of hazardous substances, hydrocarbons or other waste products into the environment. Environmental legislation may require that we do the following:

acquire permits before commencing drilling;

restrict spills, releases or emissions of various substances produced in association with our operations;

limit or prohibit drilling activities on protected areas such as wetlands or wilderness areas;

take reclamation measures to prevent pollution from former operations;

.

.

.

take remedial measures to mitigate pollution from former operations, such as plugging abandoned wells and remedying contaminated soil and groundwater; and

take remedial measures with respect to property designated as a contaminated site.

There is inherent risk of incurring environmental costs and liabilities in connection with our operations due to our handling of natural gas and other petroleum products, air emissions and water discharges related to our operations, and historical industry operations and waste disposal practices. The costs of any of these liabilities are presently unknown but could be significant. We may not be able to recover all or any of these costs from insurance. In addition, we are unable to predict what impact the Gulf oil spill will have on independent oil and gas companies

such as our company. For instance, companies such as ours currently pay an \$0.08 per barrel tax on all oil produced in the U.S. which is contributed to the Oil Spill Liability Trust Fund. There are pending proposals to raise this tax to \$0.18 to \$0.25 per barrel. It is also probable that there will be increased enforcement of existing regulations and adoption of new regulations which will also increase our cost of doing business which would reduce our operating profits in future periods.

#### The effects of future environmental legislation on our business are unknown but could be substantial.

Environmental legislation is evolving in a manner expected to result in stricter standards and enforcement, larger fines and liability and potentially increased capital expenditures and operating costs. Changes in, or enforcement of, environmental laws may result in a curtailment of our production activities, or a material increase in the costs of production, development drilling or exploration, any of which could have a material adverse effect on our financial condition and results of operations or prospects. In addition, many countries, as well as several states in the United States have agreed to regulate emissions of greenhouse gases. Methane, a primary component of natural gas, and carbon dioxide, a byproduct of burning natural gas, are greenhouse gases. Regulation of greenhouse gases could adversely impact some of our operations and demand for products in the future.

# Should we fail to comply with all applicable FERC administered statutes, rules, regulations and orders, we could be subject to substantial penalties and fines.

Under the Energy Policy Act of 2005, the Federal Energy Regulatory Commission, or FERC, has authority to impose penalties for violations of the Natural Gas Act, up to \$1 million per day for each violation and disgorgement of profits associated with any violation. FERC has recently proposed and adopted regulations that may subject our facilities to reporting and posting requirements. Additional rules and legislation pertaining to these and other matters may be considered or adopted by FERC from time to time. Failure to comply with FERC regulations could subject us to civil penalties.

### Our business depends on oil and natural gas transportation facilities, most of which are owned by others.

The marketability of our oil and natural gas production depends in large part on the availability, proximity and capacity of pipeline systems owned by third parties. The lack of available capacity on these systems and facilities could result in the shut-in of producing wells or the delay or discontinuance of drilling plans for properties. The lack of availability of these facilities for an extended period of time could negatively affect our revenues. Federal and state regulation of oil and natural gas production and transportation, tax and energy policies, changes in supply and demand, pipeline pressures, damage to or destruction of pipelines and general economic conditions could adversely affect our ability to produce, gather and transport oil and natural gas.

# Our business involves many operating risks that may result in substantial losses for which insurance may be unavailable or inadequate.

Our operations are subject to hazards and risks inherent in drilling for oil and gas, such as fires, natural disasters, explosions, formations with abnormal pressures, casing collapses, uncontrollable flows of underground gas, blowouts, surface cratering, pipeline ruptures or cement failures, and environmental hazards such as natural gas leaks, oil spills and discharges of toxic gases. Any of these risks can cause substantial losses resulting from injury or loss of life, damage to or destruction of property, natural resources and equipment, pollution and other environmental damages, regulatory investigations and penalties, suspension of our operations and repair and remediation costs. In addition, our liability for environmental hazards may include conditions created by the previous owners of properties that we purchase or lease. We maintain insurance coverage against some, but not all, potential losses. We do not believe that insurance coverage for all environmental damages that could occur is available at a reasonable cost. Losses could occur for uninsurable or uninsured risks, or in amounts in excess of existing insurance coverage. The occurrence of an event that is not fully covered by insurance could harm our financial condition and results of operation.

# Our Cook Inlet Basin leases and our Osprey Platform are located in a region of active volcanoes and we could be subject to the adverse impacts of natural disasters.

The Cook Inlet region contains active volcanoes, including Augustine Volcano, Mount Spurr and Mount Redoubt, and volcanic eruptions in this region have been associated with earthquakes and tsunamis and debris avalanches have also resulted in tsunamis. In 2009 the Cook Inlet Pipeline Co. suspended operations on several

occasions as a result of the spring 2009 major eruption of Mount Redoubt which also resulted in a shutdown of the Drift River Oil Terminal. Our operations in this area are subject to all of the inherent risks associated with operations in a geographical region which is subject to natural disasters and we are susceptible to the risk of damage to our operations and assets located in the Cook Inlet Basin. While our facilities are engineered to withstand seismic activity, and the current tight line configuration should allow us to continue shipments through an active volcanic period without much interruption, we do not maintain business interruption insurance which could adversely impact our results of operations as the result of lost revenues in future periods.

# We may be subject to certain conflicts of interest related to Miller Energy Income 2009 -A, LP which may not be resolved in a manner favorable to our company.

A wholly owned subsidiary of our company is the general partner of Miller Energy Income 2009-A, LP ("MEI") and Messrs. Miller, Boruff and Boyd are officers of MEI. In November and December 2009 we borrowed an aggregate of approximately \$2.7 million from MEI under the terms of four year secured notes. In the event there should be a dispute under this loan, there are no assurances that it will be resolved in our favor. We asked a third party to hold in escrow titles to substantial company drilling equipment to serve as collateral for these loans. In the event of a dispute under this loan, we could lose ownership of this equipment which we need to perform drilling and drilling services. In that event, we would be unable to continue our operations as they are presently conducted which would have a material adverse impact on our results of operations in future periods. There are no assurances that decisions Messrs. Miller, Boruff and/or Boyd make in matters related to MEI will be beneficial to us.

# Certain of our outstanding warrants contain cashless exercise provisions which means we will not receive any cash proceeds upon their exercise.

At July 22, 2010 we have common stock warrants outstanding to purchase an aggregate of 2,910,000 shares of our common stock with an average exercise price of \$1.21 per share which are exercisable on a cashless basis. This means that the holders, rather than paying the exercise price in cash, may surrender a number of warrants equal to the exercise price of the warrants being exercised. It is possible that the warrant holders will utilize the cashless exercise feature which will deprive us of additional capital which might otherwise be obtained if the warrants did not contain a cashless feature.

# A large portion of our outstanding common shares are restricted securities and we have outstanding options, warrants and purchase rights to purchase approximately 47% of our currently outstanding common stock.

At July 22, 2010 we had 33,389,383 shares of common stock outstanding together with outstanding options and warrants to purchase an aggregate of 11,976,955 shares of common stock at exercise prices of between \$0.01 and \$6.53 per share and \$1,285,000 principal amount of convertible notes which are convertible into 2,336,364 shares of common stock at a conversion price of \$0.55 per share. Of our outstanding shares of common stock at July 22, 2010, approximately 16,426,863 shares are "restricted securities." Future sales of restricted common stock under Rule 144 or otherwise could negatively impact the market price of our common stock. In addition, in the event of the exercise of the warrants and options and the conversion of the notes, the number of our outstanding common stock will increase by approximately 32%, which will have a dilutive effect on our existing shareholders.

# The impacts of non-cash gains and losses from derivative liability accounting in future periods could materially impact our financial results.

As a result of the terms of the March 2010 unit offering, as well as the terms of other recent financing transactions we have entered into in 2010, our financial statements for the year ended April 30, 2010 were impacted by the accounting effect of the application of derivative accounting. We anticipate that our financial statements in future periods will also be impacted. The application of EITF 07-05 *Determining Whether an Instrument (or Embedded Feature) is Indexed to a Company's Own Stock,* which was effective on January 1, 2009 will significantly affect the application of

ASC Topic 815 and ASC Topic 815-40 for both freestanding and embedded derivative financial instruments in our financial statements. Generally, warrants, conversion features in debt, and similar terms that include full-ratchet or reset provisions, which mean that the exercise or conversion price adjusts to pricing in subsequent sales or issuances, no longer meet the definition of indexed to a company's own stock and are not exemption for equity classification provided in ASC Topic 815-15. This means that instruments that were previously classified in equity will require reclassification to liabilities and ongoing measurement under ASC Topic 815. As a result of the application of this accounting principle, we recorded a current derivative liability of \$720,840

and a long-term derivative liability of \$16,708,947 at April 30, 2010. Beginning in the first quarter of 2011, we expect to record either a gain or loss based upon the market price of our common stock. The amount of quarterly non-cash gains or losses we will record in future periods is unknown at this time as the measurement is based upon the fair market value of our common stock on the measurement date. It is likely, however, that these non-cash gains or losses could have a material impact on our financial results in future periods.

## ITEM 1B.

### UNRESOLVED STAFF COMMENTS.

None.

# ITEM 2.

.

### **PROPERTIES.**

Our executive offices presently comprise approximately 4,968 square feet for the main office building and 6,600 square feet for the shop building on 14.05 acres of land in Huntsville, Tennessee that we own. We also own or rent facilities in the following locations:

Tennessee: Knoxville, Huntsville and Sunbright.

Anchorage, Alaska

### Production facilities

Cook Inlet Energy operates two onshore production facilities and one offshore platform located on or near the West Foreland, which is a small peninsula located in a remote area on the west side of Cook Inlet. The West Foreland is accessible by barge and fixed wing aircraft, and is not tied in with the Alaskan road system or electrical grid. Cook Inlet Energy maintains its own 10-mile road system and local electrical system.

The West McArthur River Unit Production Facility is one of our two onshore production facilities. It is located 3.5 miles south of Chevron's Trading Bay Production Facility, which is near the site of the local airstrip and barge landing. The West McArthur River Unit Production Facility can process 5,000 barrels of oil and seven MMscf of natural gas per day, generate three megawatts of electricity and contains 10,000 barrels of on-site tankage. The West McArthur River Unit Production Facility also includes our onshore camp, which provides housing and life support facilities sufficient for 65 people.

The Kustatan Production Facility is our other onshore production facility. This facility can process 30,000 barrels of oil a day, generate 16 megawatts of electricity, treat up to 20 MMscf of natural gas and contains 50,000 barrels of on-site tankage. The facility, which is located five miles south of The West McArthur River Unit Facility, is currently inactive.

#### Oil and gas properties

Information on our oil and gas properties appears earlier in this annual report.

# **ITEM 3.**

### LEGAL PROCEEDINGS.

CNX Gas Company, LLC commenced litigation on June 11, 2008 in the Chancery Court of Campbell County, State of Tennessee in a case style CNX Gas Company, LLC vs. Miller Petroleum Inc., Civil Action No. 08-071, to enjoin us from assigning or conveying certain leases described in the Letter of Intent signed by CNX and our company on May 30, 2008, to compel us to specifically perform the assignments as described in the Letter of Intent; and for damages. A Notice of Lien Lis Pendens was issued June 11, 2008. We moved for entry of summary judgment dismissing the claims asserted against us by CNX and on January 30, 2009 the court found that CNX s claims had no merit. The court granted our motion and dismissed all claims asserted by CNX in that action. CNX has appealed the ruling, and briefs have been submitted to the Court of Appeals of Tennessee. Oral arguments were held on May 18, 2010, and an opinion from the Court of Appeals is expected sometime in the fall of 2010.

On May 20, 2009 Gunsight Holdings, LLC, a Florida limited liability company, filed a complaint in the United States District Court for the Eastern District of Tennessee, Northern Division, against us styled Gunsight Holdings, LLC, Plaintiff, v Miller Petroleum, Inc. and Ky-Tenn Oil, Inc., Defendants, Case No. 3-09-CV-221. The litigation surrounds certain rights related to approximately 6,800 acres in Scott County, Tennessee which KTO purportedly acquired under a lease assignment from an unrelated party in August 2004. In September 2008, KTO

assigned us 75% of its interest in the subject lease and the working interest in all the wells on the leased land, retaining a 25% interest in the wells consisting of landowner's royalty and overriding royalty. On June 8, 2009 we acquired certain assets from KTO including KTO's undivided interest in approximately 170 oil and gas wells in Morgan, Scott and Fentress counties in Tennessee, together with all property, fixtures and improvements, leasehold interest and contract rights related to these wells and undivided interest in approximately 35,325 acres of oil and gas leases in Scott and Morgan counties, Tennessee. The lease which is the subject of the litigation was included in the assets purchased by us from KTO. The plaintiff is alleging that our company and KTO have failed or refused to pay royalties due to the plaintiff s predecessors and have breached the implied duty of further exploration by failing to drill required wells, failing to reasonably develop or explore the property, failing to maintain an active interest in further development of the property and otherwise failing to act as a prudent operator of the property thereby causing damages to the plaintiff exceeding \$75,000. The plaintiff is seeking a declaratory judgment of its allegations, removal of our company and KTO from the property, a full accounting of activities related to the property and all monies received from those activities, damages and costs of action. We have filed an answer denying the various claims and asserting affirmative defenses including that there has been continuous production from the subject lease. We are currently in discovery.

On October 8, 2009 we filed an action styled Miller Petroleum, Inc. v. Maynard, Civil Action No. 9992 in the Chancery Court for Scott County, Tennessee, seeking a declaratory judgment that there has been continuing commercial production of oil, and oil and gas lease owned by us is still in full force and effect. The defendant filed an Answer and Counterclaim, seeking in the Counterclaim a declaration that the oil and gas lease has expired. Although no compensatory monetary damages have been sought against us, the Counterclaim does seek attorney fees, expenses and costs. There has been no discovery to date and a trial date has not been assigned. Plaintiff s attorney left his firm, and the case has been delayed as Plaintiff s new counsel (with the same firm) becomes familiar with the case.

On March 26, 2010, Petro Capital III, LP filed an action styled Petro Capital III, LP v. Miller Petroleum, Inc., Civil Action No. 3:10-cv-00606P in the United States District Court for the Northern District of Texas, Dallas Division, seeking damages for breach of contract; damages for alleged negligent misrepresentation; a declaratory judgment regarding the proper number of and exercise price of the original warrants to which Petro Capital is entitled under a warrant and registration rights agreement, the number of penalty warrants to which Petro Capital may be entitled under a warrant and registration rights agreement, as well as the proper exercise price thereof, and damages resulting from the alleged breach of contract; and attorney s fees. On April 6, 2010, Petro Capital filed an Amended Complaint that did not include additional causes of action. We filed an Answer and Counterclaim on April 28, 2010. The Counterclaim seeks a declaratory judgment to declare void the issuance of any penalty warrants after May 4, 2007 (the latest date upon which the shares underlying the warrants would become freely tradable under Rule 144). The Counterclaim further seeks a declaratory judgment as to the number of shares and proper exercise price for the original warrant.

We are also party to various routine legal proceedings arising in the ordinary course of our business. Management believes that none of these actions, individually or in the aggregate, will have a material adverse effect on our financial condition or results of operations.

# ITEM 4.

### (REMOVED AND RESERVED).

### PART II

### ITEM 5.

# MARKET FOR REGISTRANT S COMMON EQUITY, RELATED STOCKHOLDER MATTERS AND ISSUER PURCHASES OF EQUITY SECURITIES.

Since May 6, 2010 our common stock has been listed on the NASDAQ Global Market under the symbol MILL. Previously, our common stock was quoted on the OTC Bulletin Board and in the over the counter market on the Pink Sheets. The reported high and low sales prices for the common stock as reported on the various markets on which our stock was quoted during the periods indicated are shown below. The quotations reflect inter-dealer prices, without retail mark-up, markdown or commission, and may not represent actual transactions.

	High		Low	
2008				
First quarter ended July 31, 2007	\$	0.49	\$	0.25
Second quarter ended October 31, 2007	\$	0.34	\$	0.02
Third quarter ended January 31, 2008	\$	0.10	\$	0.04
Fourth quarter ended April 30, 2008	\$	0.22	\$	0.09
2009				
First quarter ended July 31, 2008	\$	0.54	\$	0.10
Second quarter ended October 31, 2008	\$	0.51	\$	0.15
Third quarter ended January 31, 2009	\$	0.40	\$	0.15
Fourth quarter ended April 30, 2009	\$	0.40	\$	0.15
2010				
First quarter ended July 31, 2009	\$	0.38	\$	0.22
Second quarter ended October 31, 2009	\$	0.70	\$	0.28
Third quarter ended January 31, 2010	\$	2.95	\$	0.60
Fourth quarter ended April 30, 2010	\$	6.60	\$	1.95

On July 22, 2010, the last sale price of our common stock as reported on the NASDAQ Global Market was \$5.66. As of July 22, 2010, there were approximately 400 record owners of our common stock.

### **Dividend Policy**

We have never paid cash dividends on our common stock and we do not anticipate that we will declare or pay dividends in the foreseeable future. Payment of dividends, if any, is within the sole discretion of our Board of Directors and will depend, among other factors, upon our earnings, capital requirements and our operating and financial condition. In addition under Tennessee law, we may not pay a dividend if, after giving effect, we would be unable to pay our debts as they become due in the usual course of business or if our total assets would be less than the sum of our total liabilities plus the amount that would be needed if we were to be dissolved at the time of the payment of the dividend to satisfy the preferential rights upon dissolution of shareholders whose preferential rights

### ITEM 6.

# SELECTED FINANCIAL DATA.

Not applicable to a smaller reporting company.

# ITEM 7.

# MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

#### Overview

We are an independent exploration and production company that utilizes seismic data, and other technologies for geophysical exploration and development of oil and gas wells in the Appalachian region of eastern Tennessee and the Cook Inlet Basin in south central Alaska. In addition to our engineering and geological capabilities, we provide land drilling services on a contract basis to customers primarily engaged in natural gas exploration and production.

During fiscal 2009 and 2010, we completed several transactions which we believe had both a positive impact on our balance sheet and will assist us in our continued growth. These transactions, which are described in detail earlier in this annual report under Item 1. Business - Our history included the following:

sale of leases and wells to Atlas Energy Recourses, LLC,

settlement of Wind City litigation,

acquisition of assets from Ky-Tenn Oil, Inc.,

acquisition of East Tennessee Consultants,

acquisition of Cook Inlet Energy LLC in Alaska, and

acquisition of Alaskan assets of Pacific Energy Resources.

As a result of the aforedescribed acquisitions, we presently have approximately 662,992 acres of gross oil and gas leases and exploration license rights (616,550 net acres), which includes 471,474 acres under the Susitna Basin Exploration License. The terms of these new leases have a net revenue interest ranging from 0.1% to 100.0% and run from three to five years. We are presently reviewing these leases, as well as our other existing leases, to determine the capital requirements and timing for drilling additional wells. We plan to drill five new wells in the next 12 months. As a part of our fiscal 2008 sale to Atlas Energy, we retained a 5% royalty interest on a 1,930 acre tract that we expect to be the subject of Atlas Energy drilling.

With the closing of these transactions, our management is now able to focus the majority of its efforts on growing our company. In addition to raising capital we are also continuing to focus our short-term efforts on three distinct areas, including the following:

Increase our overall oil and gas production through maintenance and repairs of nonperforming or underperforming wells,

Organically growing production through drilling for our own benefit on existing leases and under license rights, leveraging our 100,000 plus well log database and over 660,000 acres which are either under lease or part of the Susitna Basin Exploration License, with a view towards retaining the majority of working interest in the new wells, and

Expanding our contract drilling and service capabilities and revenues, including drilling and service contracts with third parties.

.

Our ability, however, to implement one or more of these goals is dependent upon the availability of additional capital. To fully expand our operations as set forth above, we will need up to \$75 million to \$100 million to fund the balance of our expansion plans, including up to approximately \$67.4 million associated with obligations arising from our purchase of the Alaskan assets to provide the required capital, we are seeking to leverage our existing assets as well as raise additional capital through the sale of equity and/or debt securities. We do not have any firm commitments for the additional capital we need to fully fund our operations and there are no assurances the capital will be available to us upon terms acceptable to us, if at all. While we are actively seeking to secure the additional capital, terms of the Securities Purchase Agreement for our March 2010 unit offering contain restrictive covenants which may adversely impact our ability to raise additional capital during the 12 months following the effective date of a registration statement we are obligated to file with the SEC. If we are not able to raise the capital as required, we will be unable to fully implement our expanded business model, and the State of Alaska could terminate the leases which comprise substantially all of our Cook Inlet Basis assets.

#### **Results of Operations**

Our fiscal year end is April 30. The year ended April 30, 2010 is referred to as fiscal 2010, the year ended April 30, 2009 is referred to as fiscal 2009 and the year ending April 30, 20011 is referred to as fiscal 2011. When used in the following tables, NM means not meaningful.

#### Fiscal 2010 as compared to fiscal 2009.

The following table shows the components of our revenues for fiscal 2010 and 2009, together with their percentages of total revenue in each year and percentage change on a year-over-year basis.

For the Year Ended April 30,					
	% of			% of	
	2010 (\$)	Revenue	2009 (\$)	Revenue	Change
Revenues					
Oil and gas revenue	4,437,215	76%	640,094	41%	+593%
Service and drilling					
revenue	1,429,789	24%	927,210	59%	+54%
Total revenues	5,867,004	100%	1,567,304	100%	+274%

Oil and gas revenue represents revenues generated from the sale of oil and natural gas produced from the wells in which we have a partial ownership interest. Oil and gas revenue is recognized as income as production is extracted and sold. We reported a 593% increase in oil and gas revenues for fiscal 2010 over 2009. The increase was primarily due to the addition of the Alaskan oil well production during fiscal 2010 which accounted for revenues of approximately \$3,621,881 for year then ended. In addition, we produced 122,015 Mcf of gas in fiscal 2010 in Alaska but we did not sell this as substantially all was used in our Alaska oil production.

Our increase in oil and gas revenue from fiscal 2009 to 2010 was primarily due to production and sales from the Alaska acquistion as well as increased oil and gas prices. For example, at April 30, 2010 oil was priced at \$86.07 per barrel versus \$50.35 at April 30, 2009 and at April 30, 2010 natural gas was \$3.92 Mcf as compared to \$3.37 per Mcf at April 30, 2009. In addition, we had 188 producing oil wells and 337 producing gas wells on April 30, 2010 compared to 20 producing oil wells and 32 producing gas wells on April 30, 2009. For fiscal 2010 we produced 63,002 barrels of oil and 154,291Mcf of natural gas as compared to 4,580 barrels of oil and 50,073 Mcf of natural gas during fiscal 2009.

Service and drilling revenue represents revenues generated from drilling, maintenance and repair of third party wells. Service and drilling income is recognized at the time it is both earned and we have a contractual right to receive the revenue. Our service and drilling revenue increased 54% for fiscal 2010 as compared to fiscal 2009. During fiscal 2010 we drilled 10 wells for Atlas Energy Resources, LLC, which compares to six wells drilled for them for fiscal 2009.

#### **Direct Expenses**

The following table shows the components of our direct expenses for fiscal 2010 and 2009. Percentages listed in the table reflect margins for each component of direct expenses and percentages of total revenue for each component of other expenses.

	For the Year Ended April 30,			
	2010 (\$)	Margin	2009 (\$)	Margin
Direct Expenses				
Oil and gas	2,583,384	42%	240,389	62 %
Service and drilling	1,342,509	6%	1,184,901	(28)%

Depletion expense	1,741,150	NM	221,465	NM	
Total direct expenses	5,667,043	3%	1,646,755	(5)%	
The cost of oil and gas revenues were \$2,583,384 for fiscal 2010 which reflected a gross profit of \$1,853,831 and a					
gross profit margin of 42% as compared to a gross profit of \$399,705 and a gross profit margin of 62% for fiscal					
2009. We follow the successful efforts method of accounting for our oil and gas activities. Accordingly, costs					
associated with the acquisition, drilling and equipping of successful exploratory wells are capitalized. During fiscal					
2010 we capitalized approximately \$376,216,621 of costs primarily associated with the recent Alaska acquisition as					
previously discussed, but also with drilling and equipping of wells as compared to \$975,992 during fiscal 2009.					
During fiscal 2009 we acquired leases for 5,007 acres for approximately \$666,000 and					

spent \$270,644 on drilling and equipping three existing wells. However, geological and geophysical costs, delay and surface rentals and drilling costs of unsuccessful exploratory wells are charged to expense as incurred and are included in the cost of service and drilling revenue. Finally, costs of drilling development wells are capitalized. Upon the sale or retirement of oil and gas properties, the cost thereof and the accumulated depreciation or depletion are removed from the accounts and any gain or loss is credited or charged to operations.

The cost of service and drilling revenue represents direct labor costs of employees associated with these services, as well as costs associated with equipment, parts and repairs. Fiscal 2010 showed \$1,342,509 for this component, up 13% from \$1,184,901 in fiscal 2009. This represented a profit of \$87,280 or 6% for fiscal 2010 compared to a loss of \$257,691 or 28% loss for fiscal 2009. As previously discussed, while drilled 10 wells for Atlas Energy during fiscal 2010 and only six wells during fiscal 2009, in preparation for the Atlas Energy drilling contract we spent significant time and expense maintaining and repairing our drilling equipment in fiscal 2009 which contributed to the loss for that year.

Depletion of capitalized costs of proved oil and gas properties is provided on a pooled basis using the units-of-production method based upon proved reserves. Acquisition costs of proved properties are amortized by using total estimated units of proved reserves as the denominator. All other costs are amortized using total estimated units of proved developed reserves. During fiscal 2010 depletion expense was \$1,741,150 or 30% of total revenue, as compared to \$221,465 or 14% of total revenue for fiscal 2009. The primary reason for the increase in depletion expense for fiscal 2010 was the addition of wells as a result of the acquisitions. As a result of these components, total direct expenses were \$5,667,043, which reflected a profit of \$199,961 and a profit margin of 3% for fiscal 2010. This compares to a loss of \$79,451 which was a negative profit margin of 5% experienced for fiscal 2009.

### Total Costs and Expenses and Total Other Income (Expense)

The following table shows the components of certain of our costs and expenses and certain of our total other income (expenses) for fiscal 2010 and 2009. Percentages listed in the table reflect percentages of total revenue for each component of other expenses.

	For the Year Ended April 30,			
	2010 (\$)	% of Revenue	2009(\$)	% of Revenue
Selling, general and				
administrative	10,345,217	176%	2,712,943	173%
Depreciation and amortization	968,158	17%	427,605	27%
Interest expense, net of interest				
income	501,739	9%	24,785	2%
Loss on derivative securities	15,861,007	NM		
Loan fees and costs	741,309	13%	124,085	8%
Loss (gain) on sale of				
equipment	9,755	NM	(10,450)	NM
Gain on sale of oil and gas				
properties			(11,715,570)	747%
Gain on acquisitions	(461,111,924)	>1,000%		
Total other expense (revenues)	(432,684,739)	>1,000%	(8,436,602)	538%

Selling, general and administrative expense includes salaries, general overhead expenses, insurance costs, professional fees and consulting fees. The increase of \$7.6 million for fiscal 2010 as compared to fiscal 2009 primarily reflects costs associated with the addition of our acquisitions during fiscal 2010 in Alaska and Tennessee, which included an increases in fund raising expenses of \$2.7 million, employee related expenses of \$0.8 million, office related expenses of \$0.7 million, taxes other than income of \$0.3 million and travel related expenses of \$0.2 million. In addition, an additional \$1.7 million was recorded as compensation expense during fiscal 2010, which reflected the cost of options

and warrants issued to various employees and directors. Also, during fiscal 2010, we wrote off \$666,476 of prepaid offering costs associated with Miller Rig & Equipment, LLC and Miller Energy Drilling 2009-A, LP. Both offerings associated with these companies ended in December, 2009 and no funds were raised.

Depreciation and amortization expenses reflect the usage of our fixed assets over time. The increase in depreciation and amortization for fiscal 2010 as compared to fiscal 2009 reflects an increase in the amount of depreciation due to the Alaskan assets purchased. These non-cash expenses will continue at this higher level as the Alaska assets are being depreciated over a range of 30 to 40 years.

We recorded a non-cash loss on derivative securities for fiscal 2010 of \$15,861,007 relating to the change in fair value of derivative instruments during fiscal 2010. This was comprised of three transactions, 3,000,000 warrants issued in the current and past years, which are subject to an ongoing litigation matter, 716,715 warrants issued in an equity financing in March 2010 and 300,000 warrants issued pursuant to a consulting arrangement in March 2010. We utilized the Black-Scholes pricing model to calculate the expense. The fair value of the warrants issued and outstanding at May 1, 2009, attributed to this derivative liability has been determined to be immaterial due to the low stock price in comparison to the exercise price, hence there was no adjustment to make for fiscal 2009.

Loan fees and costs of \$741,309 for fiscal 2010, primarily represent non-cash expenses related to the fair value of warrants issued to new investors as an incentive to invest in the MEI partnership as well as expenses related to the fair value of warrants owned in connection with a prior financing transaction.

During fiscal 2009 we recorded a one-time gain of \$11,715,570 on the sale of the oil and gas leases to Atlas Energy and the concurrent settlement of the Wind City litigation as described elsewhere herein. As a result of the one-time settlement transaction, we reported net income of \$8,356,373 for fiscal 2009.

During fiscal 2010, we recorded a gain on acquisitions of \$461,111,924. This was primarily due from the Alaskan acquisition as previously discussed. As a result of this non-cash gain, for fiscal 2010 we recorded net income of \$249,453,180, an increase of \$241,096,807 over fiscal 2009.

We do not anticipate recording similar gains on acquisitions in future periods.

# Liquidity and capital resources

Liquidity is the ability of a company to generate adequate amounts of cash to meet the enterprise's needs for cash. At April 30, 2010 we had a working capital surplus of \$338,110 as compared to a working capital deficit of \$313,565 at April 30, 2009. This increase in capital surplus is primarily due to increased cash provided from financing activities, while partially offset by cash used by operating activities.

From April 30, 2009 to April 30, 2010, cash increased from \$46,566 to \$2,750,841. This increase was primarily due from the cash raised through net equity sales of \$9,646,478 and proceeds of borrowings of \$5,926,444 which were raised during fiscal 2010, partially offset by the funds required for the Alaska oil and gas assets. Other asset categories increased significantly due to the Alaska transaction. Fixed assets increased \$111.1 million from April 30, 2009 to April 30, 2010 as new assets booked for Alaska were \$110.5 million. Oil and gas properties increased \$374.4 million during this time as well, as \$368.0 million of the increase was due to the addition of Alaska reserves. Deferred income taxes payable rose to \$184.5 million on April 30, 2010 from \$778 on April 30, 2009, primarily due to the recording of the Alaska transaction. The Alaska transaction also contained a one-time gain of \$277.2 million which is reflected in the \$275.7 million stockholders equity recorded at April 30, 2010 as compared to \$7.2 million recorded on April 30, 2009.

We do not presently have any commitment for capital expenditures other than related to the Osprey platform and onshore assets as described below. However, as set forth earlier in this section we require a substantial amount of capital to fund our other obligations associated with the acquisition of the Alaskan assets.

Under the terms of the purchase agreement for the Alaskan assets and the Assignment Oversight Agreement, Cook Inlet Energy assumed all liabilities related to the plugging, abandonment, decommissioning, removal and/or restoration liabilities associated with or arising from the acquired assets with respect to all periods prior to, on or after the closing date. Under the terms of the purchase agreement for the Alaskan assets, these assumed liabilities include approximately \$10 million for the onshore assets and approximately \$40 million associated with a retirement liability for the Osprey platform, of which approximately \$6.6 million is presently on deposit in an escrow fund with the State of Alaska. We are presently in discussion with the State of Alaska to reduce these amounts to levels we believe are

more realistic. During the fourth quarter of 2010 we accrued approximately \$15.0 million for these liabilities, which includes approximately \$3.5 million for the onshore assets and approximately \$10.0 million for the Osprey platform. We are also seeking to obtain confirmation from the State of Alaska that the \$6.6 million, currently in the escrow account is specifically allocated to the Osprey platform.

#### Cash flows

.

Net cash used by operating activities for fiscal 2010 was \$2,160,152. This primarily reflects the cash paid for the costs of revenues and selling, general and administrative expense in excess of revenues received for the period, which included the gain from the Alaska transaction, but partially offset by the issuance of equity for services, compensation and financing costs of \$3,892,886.

Net cash used by operating activities in fiscal 2009 was \$1,721,122. This primarily reflects the cash paid for the costs of revenues and selling, general and administrative expense in excess of revenues received for the period, which included the gain from the sale of oil and gas properties, but partially offset by the issuance of equity for services, compensation and financing costs of \$1,605,994.

Net cash used by investing activities for fiscal 2010 of \$10,476,830 is primarily due to the cash we paid for the Alaska assets of \$4,541,252 and the purchase of oil and gas properties of \$5,600,843, which were primarily costs associated with well start ups.

Net cash provided by investing activities of \$6,760,273 in fiscal 2009 reflects the net cash we received from the Atlas Energy transaction of \$12,519,713, partially offset by the purchase of additional drilling equipment and vehicles of \$4,408,998 and funds used for the purchase of a lease and capitalized costs associated with the purchase of oil and gas properties of \$1,268,942.

Net cash provided by financing activities of \$15,341,257 for the fiscal 2010 primarily reflects the net cash received from the sale of stock of \$9,646,478, proceeds received from borrowings of \$5,926,444, a \$1,856,488 decrease in restricted cash due to payoff of a bank financing, and cash acquired through acquisitions of \$203,993, which was partially offset by payments on notes payable of \$2,309,205.

Net cash used in financing activities of \$5,035,021 for fiscal 2009 primarily reflects the repurchase of 2,900,000 shares of our common stock from Wind Mill for \$4,350,000 due to the settlement of Wind Mill litigation as discussed elsewhere here. In addition, we used cash to pay off certain notes payable of \$726,630 during fiscal 2009.

### Loan Commitment from Vulcan Capital Corporation, LLC

On November 5, 2009 we entered into a letter agreement with Vulcan Capital Corporation, LLC which memorialized the terms of our agreement with Vulcan Capital Corporation to provide us with a financial debt package for the specific requirements of our acquisition of Cook Inlet Energy and the Alaskan assets of Pacific Energy Resources. Mr. Ford Graham, the President of Vulcan Capital Corporation, LLC, is a former executive officer and director of our company. This November 2009 agreement superseded an earlier October 2009 letter agreement and provided for an increased debt commitment from \$5.5 million to \$36.5 million. Under the terms of this letter agreement, Vulcan Capital Corporation has agreed to lend us not less than \$36.5 million at the closing of these transactions. Both of these transactions, which closed in December 2009, are described earlier in this annual report under Item 1. Business - Our history. The terms of the proposed debt included the following:

a senior secured status on all Cook Inlet Energy assets acquired by us,

the use of proceeds from the loan is limited to the Assignment Oversight Agreement between Cook Inlet Energy and the State of Alaska

a three year term, with a prepayment penalty, and 10% per annum interest

interest only for 24 months, principal due as a bullet,

.

•

•

•

.

•

there will be no sinking fund established, we will be unable to pay dividends before the debt is repaid and there will be certain unspecified negative covenants,

we will grant Vulcan Capital Corporation warrants exercisable at \$0.01 per share equal to 10% of the total debt package,

Vulcan Capital Corporation will have a right of first refusal for all other of our debt and/or equity requirements with regard to these acquired assets and any other of our Alaskan business activities,

Vulcan Capital Corporation or its assignee will be provided with one Board seat during the term of the loan,

the documents will contain customary representations and warranties and we will indemnify Vulcan Capital Corporation; and

we will pay all fees and expenses at closing.

The assets identified in the letter agreement include all properties and operations regarding specific State of Alaska oil and gas leases, wells, all associated infrastructures, including the Osprey platform and its subsea pipelines, and all associates agreements held by Pacific Energy Resources including the \$6.7 million Redoubt escrow fund. The letter agreement provides that Vulcan Capital Corporation will provide us this debt through it newly created special purpose vehicle, Vulcan Miller Alaska Energy, LLC. In preparation of the transaction, Vulcan Capital Corporation placed cash or cash equivalents of over \$5 million into Vulcan Miller Alaska Energy s separate account and agreed to leave the account and its assets unencumbered until closing of the transaction, at which time it is to deliver the funds to us for our sole and exclusive use for the assigned leases. The balance of the funds are to be provided to us upon demand. Under the terms of the letter agreement, upon acceptance by us Vulcan Capital Corporation was to deliver to us senior loan documents for execution to close this transaction. We accepted the letter agreement on November 5, 2009, however we have not proceeded towards a closing of the financing arrangement. We have subsequently been advised by Vulcan Capital Corporation that the amount previously deposited in the Vulcan Miller Alaska Energy, LLC account has been transferred from the account and is no longer available. We have utilized a portion of the funds we raised in private placements to fund approximately \$5.15 million of the anticipated use of proceeds of the Vulcan Capital Corporation commitment as we believed the terms of those financings were more advantageous to us. As we have not yet begun working on the Redoubt Unit which represents the balance use of proceeds from the Vulcan Capital Corporation commitment, while the commitment remains available to us, we intend to seek to obtain alternative financing upon terms which are more advantageous to us than the Vulcan Capital Corporation commitment. Our ability however to obtain this financing is limited by the terms of the March 2010 unit offering as described elsewhere herein.

### **Recent Financing Transactions**

In order to finance the expansion of our operations into Alaska and to provide capital to us for our other operations, we entered into the following financing transactions:

.

On November 1, 2009 we borrowed \$2,365,174 from Miller Energy Income 2009-A, LP ("MEI ), a limited partnership of which our wholly-owned subsidiary, Miller Energy GP, LLC, is the general partner. Under the four year secured promissory note we issued MEI to evidence this loan, interest is payable at the rate of 12% per annum, with interest only payments due monthly. On December 15, 2009 we borrowed an additional \$356,270 from MEI and issued it a second, four year secured promissory note which also pays interest at the rate of 12% per annum with interest only payments due monthly. In connection with these loans, we granted MEI a first priority security interest in oil and gas drilling equipment owned by us. Pursuant to the terms of an escrow agreement, a third-party escrow agent has been retained to hold the certificates of title for the collateral to which title is evidenced by a certificate. The remaining equipment is subject to a financing statement that has been filed with the Tennessee Secretary of State. We used the proceeds from these loans for general corporate purposes including reducing outstanding debt and to partially fund the Alaska transaction.

In December 2009 we issued \$2,855,000 principal amount 6% convertible secured promissory notes to provide funds for the Alaskan asset transaction. Included in the sales of these notes was an aggregate of \$500,000 purchased by Messrs. Scott Boruff, our Chief Executive Officer and a member of our Board of Directors, and Mr. Deloy Miller, members of our Board of Directors. We paid a finder s fee of \$20,000. Interest on the notes is paid quarterly and the principal is due December 4, 2016. Holders of \$1,150,000 principal amount of these notes, including Messrs. Boruff and Miller, have subsequently converted those notes into an aggregate of 2,090,909 shares of our common stock. As of June 30, 2010, notes in the aggregate principal amount of \$1,375,000 remain outstanding. The notes contain a convertible feature which the note holder has the right, but not the obligation, at the holder's option, at any time prior to payment in full of the principal balance of the note, to convert the unpaid principal amount of \$0.55 per share. We granted the note holders a lien on and security interest in property, assets and rights including, but not limited to, all of our mineral rights and oil and gas assets and all proceeds from those assets in the 35,325 leased acres located in

#### Morgan and Scott Counties on the Chattanooga Shale and the 173 natural gas and oil producing wells.

.

Between December 2009 and January 2010 we sold 6,015,000 shares of our common stock in private transactions to accredited investors for \$1.00 per share. This was a discount of 16.67% from market value on the date of determination. We received \$5,657,000 in net cash proceeds from this offering, after payment of offering costs, commissions and finder s fees, which was used for general corporate purposes, including reducing debt and partially financing the Alaska asset acquisition. We paid Sutter Securities Incorporated, a FINRA member firm, cash compensation of \$200,000 as well as the non- accountable sum of \$10,000 for its legal fees and expenses and issued it five-year warrants to purchase an aggregate of 280,000 shares of our common stock at exercise prices ranging from \$1.35 to \$1.815 per share. We also paid finder s fees of \$123,000 and issued five-year warrants to purchase an aggregate of 52,500 shares of our common stock at exercise price of \$1.35 per share. In addition, we paid Seaside 88 Advisors, LLC, the general partner of one of the purchasers of the shares, the non-accountable sum of \$25,000. The warrants are exercisable on a cashless basis. If we make any subsequent sales of our securities within one year to any purchaser introduced to us by Sutter Securities Incorporated, we are obligated to pay that firm a finder s fee on those sales. Under the terms of the Securities Purchase Agreements we agreed that until 12 months from the closing date, if in connection with a Subsequent Financing (as defined in the Securities Purchase Agreement), either our company or any of our subsidiaries should issue any common stock or common stock equivalents entitling any person or entity to acquire shares of common stock at an effective price per share less than the per share purchase price of \$1.00 (subject to reverse and forward stock splits and the like), that we will issue to the purchaser of this current stock sale, a number of additional shares of common stock to the aforementioned purchasers to prevent the follow-on investment from being a dilutive issuance (see the attachment for more specific details). If shares are issued for a consideration other than cash, the per share selling price shall be the fair value of such consideration as determined in good faith by the Board of Directors. We also granted the purchasers of stock certain piggy back registration rights until such time as the purchasers are able to resell the shares of common stock purchased in the offering pursuant to Rule 144 of the Securities Act until the requirement for adequate public information on our company is no longer applicable.

On March 26, 2010, we executed a Securities Purchase Agreement pursuant to which at closing we agreed to sell units of our securities, including 1,433,432 shares of our common stock at a purchase price of \$3.50 per share and five year warrants to purchase an additional 716,716 shares of common stock with an exercise price of \$5.28 per share to 14 accredited and/or institutional purchasers. This offering closed on April 1, 2010. We received gross proceeds \$5,017,002. Sutter Securities Incorporated, a broker-dealer and member of FINRA, acted as finder for us in this unit offering. Under the terms of a Finder s Agreement with the firm, we paid Sutter Securities Incorporated a fee of \$346,190 and issued the firm five-year common stock purchase warrants to purchase an aggregate of 100,339 shares of our common stock at an exercise price of \$5.28 per share. In addition, we paid a finder s fee of \$5,000 to Viriathus Capital LLC and paid the attorney for Sutter Securities Incorporated legal expenses totaling \$10,000 incurred in the preparation of the various transactional documents. We are using the net proceeds of this offering for general working capital.

The Securities Purchase Agreement for the March 2010 unit offering provides that until September 26, 2010 any securities sold in the offering are subject to a per share price protection. In the event we were to issue any shares of common stock, or securities convertible into or exercisable for shares of common stock, to any third party purchaser at a purchase price or exercise price per share which is less than \$3.50 per share, or less than the exercise price of \$5.28 per warrant share (collectively, the Discounted Per Share Purchase Price ), we will automatically issue additional shares of our common stock to the purchasers in the March 2010 unit offering without the payment of any additional consideration by those purchasers. The number of shares we may be obligated to issue will be equal product of:

the fraction obtained by dividing (A) the sum of the number of initial shares and any additional shares we may have already issued the purchasers under the terms of the Securities Purchase Agreement then held by the purchasers on the date of the dilutive issuance by (B) the sum of the number of initial shares issued to the purchasers on the closing date and all additional shares issued to the purchasers after the closing date,

#### multiplied by

the difference between the aggregate number of shares of common stock that would have been issued to the purchasers at the closing if the subscription amount of \$3.50 per share was divided by the Discounted Per Share Purchase Price minus the aggregate number of shares of common stock equal to the sum of the initial shares, plus, to the extent there has been a previous issuance of additional shares to the purchasers, the number of additional shares previously issued to the purchasers.

In implementing this per share price protection, to the extent that an issuance of additional shares would result in a purchaser or any of its affiliates beneficially owning in excess of 4.99% of our common stock, then we will initially issue only a number of additional shares that would result in a purchaser (together with the purchaser s affiliates) beneficially owning 4.99% of our common stock. After this initial issuance, and until all additional shares which otherwise would have been issued under this per share price protection would have been issued, from time to time we will issue a number of the unissued additional shares so that the purchaser (together with the purchaser s affiliates) will beneficially own only 4.99% of our common stock.

We agreed under the terms of the Securities Purchase Agreement that we would not offer or sell any shares of common stock until six months from the effective date of the registration statement we are obligated to file as described below. In addition, we agreed that so long as the purchasers own any of our securities purchased in the March 2010 unit offering, we would not enter into any agreement for the issuance or sale by us or any of our subsidiaries of any common stock or common stock equivalent for cash in a variable rate transaction nor would we enter into any form of equity line of credit. Generally, a variable rate transaction means a transaction in which we sell securities which are convertible or exercisable into shares of our common stock at a price that varies based upon the market price of our common stock or contains a price reset provision.

Finally, under the terms of the Securities Purchase Agreement we agreed that until one year after the effective date of the registration statement we are obligated to file:

that if we or any of our subsidiaries issue any common stock or common stock equivalents for cash, debt or a combination of cash and debt, purchasers in the offering would have a right to participate in this subsequent financing in an amount equal to 100% of any subsequent financing, and upon the same terms and conditions and at a price as may be contemplated by this subsequent financing; and

we would not undertake a forward or reverse stock split or a reclassification of our common stock without the prior written consent of the purchasers holding a majority in interest of the shares sold in the offering.

The terms of the warrants issued to the purchasers in the offering, as well as the compensatory warrants issued to Sutter Securities Incorporated, are identical and provide that the number of shares issuable upon the exercise of the warrants, as well as the exercise price of the warrants, is subject to proportional adjustment in the event of stock splits, stock dividends, recapitalizations and similar corporate events. The warrants are exercisable on a cashless basis. The warrants are not exercisable to the extent that (i) the number of shares of our common stock beneficially owned by the holder and (ii) the number of shares of our common stock issuable upon the exercise of the warrants would result in the beneficial ownership by holder of more than 4.99% of our then outstanding common stock. This provision may be waived upon 61 days notice to us; provided, however, that the beneficial ownership limitation can in no event exceed 9.99% of the number of shares of the common stock outstanding immediately after giving effect to the issuance of

shares of common stock upon exercise of the warrant.

So long as the warrants are outstanding, if we should issue or sell any common stock or common stock equivalent, or grant any option to purchase any of our common stock or common stock equivalents, at a price less than the then exercise price, we will automatically reduce the exercise price of the warrants and the number of shares of common stock issuable upon the exercise of the warrants will be automatically increased so that the aggregate exercise price payable upon the exercise of the warrant, after taking into account the decrease in the exercise price, will be equal to the aggregate exercise price prior to the adjustment. In addition, while the warrant is outstanding, if we should issue or sell any common stock or common stock equivalent, or grant any option to purchase any of our common stock or common stock equivalents, at a price less than the daily volume weighted average of our common stock (the VWAP ) on the record date of the proposed transaction, then we will automatically adjust the exercise price of the warrants by multiplying it by a fraction, with the denominator being the number of shares of the common stock outstanding on the date of issuance of such rights, options or warrants

plus the number of additional shares of common stock offered for subscription or purchase, and the numerator being be the number of shares of the common stock outstanding on the date of issuance of such rights, options or warrants plus the number of shares which the aggregate offering price of the total number of shares offered would purchase at the VWAP.

Under the terms of the Registration Rights Agreement entered into with the purchasers in the March 2010 unit offering, we were obligated to file a registration statement with the SEC covering the shares of common stock issued and sold in the offering, as well as the shares of common stock underlying the warrants, on or before April 15, 2010 so as to permit the public resale thereof. We have not yet filed the registration statement. We agreed to use our best efforts to cause the registration statement to be declared effective by the SEC within 90 days from the filing date or 120 days if the registration statement should be selected for a full review by the staff of the SEC. The registration rights agreement provides that if we failed to timely file the registration statement, or if it should not be declared effective within the prescribed time, we are subject to liquidated damages payable in cash equal to 2% of the aggregate purchase price of the securities up to a maximum of 12% of the total proceeds of the offering. Because we did not timely file the registration statement, we began accruing liquidated damages during the fourth quarter of 2010.

### Modernization of Oil and Gas Reporting

.

In December 2008, the SEC announced that it had approved revisions to its oil and gas reporting disclosures by adopting amendments to Rule 4-10 of Regulation S-X and Items 201, 801, and 802 of Regulation S-K. These new disclosure requirements are referred to as "Modernization of Oil and Gas Reporting" and include provisions that do the following:

introduce a new definition of oil and gas producing activities. This new definition allows companies to include in their reserve base volumes from unconventional resources. Such unconventional resources include bitumen extracted from oil sands and oil and gas extracted from coal beds and shale formations.

report oil and gas reserves using an un-weighted average price using the prior 12-month period, based on the closing prices on the first day of each month, rather than year-end pricing. This should maximize the comparability of reserve estimates among companies and mitigate the distortion of the estimates that arises when using a single pricing date.

permit companies to disclose their probable and possible reserves on a voluntary basis. Prior rules limited disclosure to only proved reserves.

update and revise reserve definitions to reflect changes in the oil and gas industry and new technologies. New updated definitions include "by geographic area" and "reasonable certainty."

permit the use of new technologies to determine proved reserves if those technologies have been demonstrated empirically to lead to reliable conclusions about reserves volumes.

require additional disclosures regarding the qualifications of the chief technical person who oversees its overall reserve estimation process. Additionally, disclosures are required related to internal controls over reserve estimation, as well as a report addressing the independence and qualifications of a company's reserves preparer or auditor based on Society of Petroleum Engineers criteria.

We began complying with the disclosure requirements in this annual report on Form 10-K.

#### **Off Balance Sheet Arrangements**

We do not have any off-balance sheet arrangements that we are required to disclose pursuant to these regulations. In the ordinary course of business, we enter into operating lease commitments, purchase commitments and other contractual obligations. These transactions are recognized in our financial statements in accordance with generally accepted accounting principles in the United States.

# **Critical Accounting Policies**

## General

The preparation of financial statements requires management to utilize estimates and make judgments that affect the reported amounts of assets, liabilities, revenues and expenses and related disclosure of contingent assets and liabilities. These estimates are based on historical experience and on various other assumptions that management believes to be reasonable under the circumstances. The estimates are evaluated by management on an ongoing basis, and the results of these evaluations form a basis for making decisions about the carrying value of assets and liabilities that are not readily apparent from other sources. Although actual results may differ from these estimates under different assumptions or conditions, management believes that the estimates used in the preparation of our financial statements are reasonable. The critical accounting policies affecting our financial reporting are summarized in Note 1 to the consolidated financial statements included in this annual report. Policies involving the most significant judgments and estimates are summarized below.

### Impact of Derivative Accounting

As a result of recent financing transactions we have entered into, our financial statements for the year ended April 30, 2010 and future periods have and will be impacted by the accounting effect of the application of derivative accounting. The application of EITF 07-05 *Determining Whether an Instrument (or Embedded Feature) is Indexed to a Company's Own Stock,* which was effective on January 1, 2009 will significantly affect the application of ASC Topic 815 and ASC Topic 815-40 for both freestanding and embedded derivative financial instruments in our financial statements. Generally, warrants, conversion features in debt, and similar terms that include full-ratchet or reset provisions, which mean that the exercise or conversion price adjusts to pricing in subsequent sales or issuances, no longer meet the definition of indexed to a company's own stock and are not exemption for equity classification provided in ASC Topic 815-15. This means that instruments that were previously classified in equity are reclassified to liabilities and ongoing measurement under ASC Topic 815. The amount of quarterly non-cash gains or losses we will record in future periods will be based upon the fair market value of our common stock on the measurement date.

### Estimates of Proved Reserves and Future Net Cash Flows

Estimates of our proved oil and gas reserves and related future net cash flows are used in impairment tests of goodwill and other long-lived assets. These estimates are prepared as of year-end by independent petroleum engineers and are updated internally at mid-year. There are many uncertainties inherent in estimating quantities of proved reserves and in projecting future rates of production and timing of development expenditures. The accuracy of any reserve estimate is dependent on the quality of available data and is subject to engineering and geological interpretation and judgment. Results of our drilling, testing and production after the date of these estimates may require future revisions, and actual results could differ materially from the estimates.

### Impairment of Long-Lived Assets

Our long-lived assets include property, equipment and goodwill. Long-lived assets with an indefinite life are reviewed at least annually for impairment, and all long-lived assets are reviewed whenever events or changes in circumstances indicate that their carrying values may not be recoverable.

# Oil and Gas Activities

We follow the successful efforts method of accounting for our oil and gas activities. Accordingly, costs associated with the acquisition, drilling and equipping of successful exploratory wells are capitalized. Geological and geophysical costs, delay and surface rentals and drilling costs of unsuccessful exploratory wells are charged to expense as incurred. Costs of drilling development wells are capitalized. Upon the sale or retirement of oil and gas

properties, the cost thereof and the accumulated depreciation or depletion are removed from the accounts and any gain or loss is credited or charged to operations.

#### Depreciation, Depletion and Amortization

Depreciation, depletion and amortization of capitalized costs of proved oil and gas properties is provided on a pooled basis using the units-of-production method based upon proved reserves. Acquisition costs of proved properties are amortized by using total estimated units of proved reserves as the denominator. All other costs are amortized using total estimated units of proved reserves.

#### Fair Value of Financial Instruments

Effective May 1, 2008, we adopted guidance issued by the FASB on "Fair Value Measurements" for assets and liabilities measured at fair value on a recurring basis. This guidance establishes a common definition for fair value to be applied to existing generally accepted accounting principles that require the use of fair value measurements, establishes a framework for measuring fair value, and expands disclosure about such fair value measurements. The adoption of this guidance did not have an impact on our financial position or operating results, but did expand certain disclosures.

The FASB defines fair value as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. Additionally, the FASB requires the use of valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs.

These inputs are prioritized below:

- Level 1: Observable inputs such as quoted market prices in active markets for identical assets or liabilities.
- Level 2: Observable market-based inputs or unobservable inputs that are corroborated by market data
- Level 3: Unobservable inputs for which there is little or no market data, which require the use of the reporting entity's own assumptions.

Cash and cash equivalents include money market securities and commercial paper and marketable securities representing certificates of deposits maturing in less than one year that are considered to be highly liquid and easily tradable. These securities are valued using inputs observable in active markets for identical securities and are therefore classified as Level 1 within the fair value hierarchy.

In addition, the FASB issued, "The Fair Value Option for Financial Assets and Financial Liabilities," effective for May 1, 2008. This guidance expands opportunities to use fair value measurements in financial reporting and permits entities to choose to measure many financial instruments and certain other items at fair value. We did not elect the fair value option for any of our qualifying financial instruments, other than those subject to recent acquisitions.

### Equity-Based Compensation

The computation of the expense associated with stock-based compensation requires the use of a valuation model. The FASB issued accounting guidance requires significant judgment and the use of estimates, particularly surrounding Black-Scholes assumptions such as stock price volatility, expected option lives, and expected option forfeiture rates, to value equity-based compensation. We currently use a Black-Scholes option pricing model to calculate the fair value of our stock options. We primarily use historical data to determine the assumptions to be used in the Black-Scholes model and have no reason to believe that future data is likely to differ materially from historical data. However, changes in the assumptions to reflect future stock price volatility and future stock award exercise experience could result in a change in the assumptions used to value awards in the future and may result in a material change to the fair value calculation of stock-based awards. This accounting guidance requires the recognition of the fair value of stock compensation in net income. Although every effort is made to ensure the accuracy of our estimates and assumptions, significant unanticipated changes in those estimates, interpretations and assumptions may result in recording stock option expense that may materially impact our financial statements for each respective reporting period.

#### **Recent Accounting Pronouncements**

On January 1, 2009, we adopted the FASB guidance for Business Combinations, which replaces SFAS No. 141, Business Combinations ("SFAS 141R" FASB ASC 805-10), and requires an acquirer to recognize the assets acquired, the liabilities assumed, and any non-controlling interest in the acquiree at the acquisition date, measured at their fair values as of that date, with limited exceptions. This Statement also requires the acquirer in a

2	n
3	9

business combination achieved in stages to recognize the identifiable assets and liabilities, as well as the non-controlling interest in the acquiree, at the full amounts of their fair values. Additionally, this Statement requires acquisition-related costs to be expensed in the period in which the costs were incurred and the services are received instead of including such costs as part of the acquisition price. This guidance makes various other amendments to authoritative literature intended to provide additional guidance or to conform the guidance in that literature to that provided in this Statement. Our acquisition of the Ky-Tenn Oil, Inc and Cook Inlet assets and the stock and membership interests of East Tennessee Consultants, Inc. and East Tennessee Consultants II, LLC were recorded in accordance with this guidance.

In April 2009, the FASB issued FASB ASC 805-20 (formerly FSP SFAS No. 141R-1, Accounting for Assets Acquired and Liabilities Assumed in a Business Combination That Arise from Contingencies). FASB ASC 805-20 amends the guidance in FASB ASC 805 (formerly SFAS 141R) relating to the initial recognition and measurement, subsequent measurement and accounting and disclosures of assets and liabilities arising from contingencies in a business combination. FASB ASC 805 (formerly FSP SFAS 141R) is effective for fiscal years beginning after December 15, 2008. We adopted FASB ASC 805 (formerly FSP SFAS 141R) as of the beginning of fiscal 2009. We will apply the requirements of FASB ASC 805-20 (formerly FSP FAS 141R-1) prospectively to any future acquisitions.

In December 2009, the FASB issued guidance for Consolidations - Improvements to Financial Reporting by Enterprises Involved with Variable Interest Entities (Topic 810). The amendments in this update are a result of incorporating the provisions of SFAS No. 167, Amendments to FASB Interpretation No. 46(R). The provisions of such Statements are effective for fiscal years, and interim periods within those fiscal years, beginning on or after November 15, 2009. Earlier adoption is not permitted. The presentation and disclosure requirements shall be applied prospectively for all periods after the effective date. Management believes this Statement will not have a material impact on our financial statements once adopted.

We determined that all other issued, but not yet effective accounting pronouncements are inapplicable or insignificant to us and once adopted are not expected to have a material impact on our financial position.

# ITEM 7A.

# QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK.

Not applicable to a smaller reporting company.

### ITEM 8.

### FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA.

Our financial statements are contained in pages F-1 through F-30, which appear at the end of this annual report.

### ITEM 9.

# CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE.

None.

# ITEM 9A(T).

#### **CONTROLS AND PROCEDURES.**

#### **Disclosure Controls and Procedures.**

Our Chief Executive Officer and Chief Financial Officer are responsible for establishing and maintaining disclosure controls and procedures for us. Disclosure controls and procedures are controls and procedures designed to reasonably assure that information required to be disclosed in our reports filed under the Securities Exchange Act of 1934, such as this report, is recorded, processed, summarized and reported within the time periods prescribed by SEC rules and regulations, and to reasonably assure that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, to allow timely decisions regarding required disclosure.

Our management does not expect that our disclosure controls or our internal controls will prevent all error and fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. In addition, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs.

Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within a company have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty, and that breakdowns can occur because of simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people or by management override of the control. The design of any systems of controls also is based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Over time, controls may become inadequate because of changes in conditions, or the degree of compliance with the policies or procedures may deteriorate. Because of these inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected.

As required by Rule 13a-15 under the Securities Exchange Act of 1934, as of April 30, 2010, the end of the period covered by this report, our management concluded its evaluation of the effectiveness of the design and operation of our disclosure controls and procedures. As of the evaluation date, our Chief Executive Officer and Chief Financial Officer, concluded that we do not maintain disclosure controls and procedures that are effective in providing reasonable assurance that information required to be disclosed in our reports under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods prescribed by SEC rules and regulations, and that such information is accumulated and communicated to our management to allow timely decisions regarding required disclosure as a result of material weaknesses in our disclosure controls and procedures. During fiscal 2010 we failed to timely file with the SEC several Current Reports on Form 8-K. In an effort to remediate these weaknesses, during the fourth quarter of 2010 we hired a General Counsel. She has developed systems which should ensure that the information required to be disclosed in our reports under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods prescribed by SEC rules and procedures. During fiscal 2010 we failed to timely file with the SEC several Current Reports on Form 8-K. In an effort to remediate these weaknesses, during the fourth quarter of 2010 we hired a General Counsel. She has developed systems which should ensure that the information required to be disclosed in our reports under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods prescribed by SEC rules and regulations, and that such information is accumulated and communicated to our management to allow timely decisions regarding required disclosure.

# Management s Report on Internal Control over Financial Reporting

Our management is responsible for establishing and maintaining adequate internal control over financial reporting as defined in Rule 13a-15(f) under the Securities Exchange Act of 1934. Our management assessed the effectiveness of our internal control over financial reporting as of April 30, 2010. In making this assessment, management used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission ("COSO") in Internal Control-Integrated Framework. Based on the assessment using those criteria, our management concluded that the internal control over financial reporting was not effective at April 30, 2010.

While we have designed a system of internal controls to supplement our existing controls during our implementation of Section 404 of the Sarbanes-Oxley Act of 2002 ("SOX 404"), we have been unable to complete testing of these controls and accordingly lack the documented evidence that we believe is necessary to support an assessment that our internal control over financial reporting is effective. Without such testing, we cannot conclude that there are any significant deficiencies or material weaknesses, nor can we appropriately remediate any such deficiencies that might have been detected. In addition, during the analysis of our internal controls in connection with our implementation of SOX 404, we did identify a number of controls weaknesses, the remediation of these controls of which are material to our internal control environment and critical to providing reasonable assurance that any potential errors could be detected. Those identified controls weaknesses include:

We do not maintain a sufficient complement of personnel with an appropriate level of accounting knowledge, experience and training in the selection and application of U.S. GAAP and SEC reporting requirements commensurate with our financial reporting requirements at our newly acquired subsidiaries and / or the corporate office to accommodate the newly acquired subsidiaries financial reporting needs.

Our newly acquired subsidiaries have yet to integrate a uniform accounting reporting system.

.

•

Due to the nature of these material weaknesses in our internal control over financial reporting, there is more than a remote likelihood that misstatements which could be material to our annual or interim financial statements could occur that would not be prevented or detected. To remediate these weaknesses we, during fiscal year, will need to:

Hire additional accounting personnel, both in Tennessee and Alaska, who are sufficiently experienced in the application of GAAP and SEC reporting requirements; and

Implement a uniform accounting reporting system within our newly acquired subsidiaries.

We anticipate that we will be able to complete these remediation efforts by April 30, 2011

*Changes in Internal Control over Financial Reporting.* There have been no changes in our internal control over financial reporting during our fourth fiscal quarter that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

## ITEM 9B.

## **OTHER INFORMATION**

None.

.

#### PART III

#### **ITEM 10.**

#### DIRECTORS, EXECUTIVE OFFICERS AND CORPORATE GOVERNANCE.

#### **Directors and Executive Officers**

Name	Age	Position
		Chairman of the Board of Directors and Chief Operating
Deloy Miller	63	Officer
Scott M. Boruff	47	Chief Executive Officer, President and director
David M. Hall	40	Chief Executive Officer of Cook Inlet Energy and director
Paul W. Boyd	51	Chief Financial Officer
Charles M. Stivers <sup>1</sup>	47	Director
Herman E. Gettelfinger <sup>2,3</sup>	77	Director
General Merrill A. McPeak		
1,	74	Director
Jonathan S. Gross <sup>3</sup>	51	Director
David J. Voyticky <sup>1,2,3</sup>	41	Director

1

Member of the Audit Committee.

2

Member of the Compensation Committee.

3

Member of the Nominating and Corporate Governance Committee.

**Deloy Miller**. Mr. Miller has been Chairman of the Board of Directors since December 1996, and was Chief Executive Officer from December 1997 to August 2008. Since then, Mr. Miller has been our Chief Operating Officer. From 1967 to 1997, Mr. Miller was the founder and Chief Executive Officer of our company. He is a seasoned gas and oil professional with more than 40 years of experience in the drilling and production business in the Appalachian basin. During his years as a drilling contractor, he acquired extensive geological knowledge of Tennessee and Kentucky and received training in the reading of well logs. Mr. Miller served two terms as president of the Tennessee Oil & Gas Association and in 1978 the organization named him the Tennessee Oil Man of the Year. He continues to serve on the board of that organization. Mr. Miller was appointed in 1978 by the Governor of Tennessee to be the petroleum industry s representative on the Tennessee Oil & Gas Board, the state agency that regulates gas and oil operations in the state. Mr. Miller is the father-in-law of Mr. Boruff.

*Scott M. Boruff*. Mr. Boruff has served as a director and our Chief Executive Officer since August 2008 and as our President since June 2010. Prior to joining our company, Mr. Boruff has been a licensed investment banker and was a director from 2006 to 2007 with Cresta Capital Strategies, LLC a New York investment banking firm that was responsible for closing transactions in the \$150 to \$200 million category. Mr. Boruff specialized in investment banking consulting services that included structuring of direct financings, recapitalizations, mergers and acquisitions

and strategic planning with an emphasis in the gas and oil field. As a commercial real estate broker for over 20 years Mr. Boruff developed condominium projects, hotels, convention centers, golf courses, apartments and residential subdivisions. As a consultant to us, Mr. Boruff led the last three major financial transactions completed by the company. Since April 2009, Mr. Boruff has also been a director and 49% owner of Dimirak Securities Corporation, a broker-dealer and member of FINRA. See Item 13. Certain Relationships and Related Transactions and Director Independence appearing later in this annual report. Mr. Boruff holds a Bachelor of Science in Business Administration from East Tennessee State University. Mr. Boruff is the son-in-law of Mr. Miller.

*David M. Hall*. Mr. Hall has served as Chief Executive Officer of our Cook Inlet Energy subsidiary and member of our Board of Directors since December 2009. Mr. Hall was the former Vice President and General Manager of Alaska Operations, Pacific Energy Resources Ltd. from January 2008 to December 2009. Before that time, from 2000 to 2008, he served as the Production Foreman and Lead Operator in Alaska for Forest Oil Corp, rising to Production Manager for all of Alaska operation for Forest Oil.

*Paul W. Boyd.* Mr. Boyd has served as our Chief Financial Officer since September 2008. Prior to joining our company, from 2001 until August 2008 Mr. Boyd was Chief Financial Officer and Treasurer of IdleAire Technologies Corporation, a Knoxville, Tennessee company which provides a patented system that enables long haul truck drivers to park their trucks for extended periods of time while still using the heat, air conditioning and many other amenities. From 1999 to 2000 Mr. Boyd was Chief Financial Officer of United States Internet, Inc., a Knoxville, Tennessee company which was a subsidiary of Earthlink Company. From 1996 to 1999 he was Treasurer

of Clayton Homes, Inc., a manufacturer of manufactured housing which is a subsidiary of Berkshire Hathaway, Inc. Mr. Boyd received a B.B.A. in Accounting from the University of Houston and is a certified public accountant.

*Charles M. Stivers*. Mr. Stivers has been a member of our Board of Directors since 2004. He also served as our Chief Financial Officer from 2004 until January 2006. Mr. Stivers has over 18 years accounting experience and over 12 years of experience within the energy industry. He owns and operates Charles M. Stivers, C.P.A., which specializes in the oil and gas industry and has clients located in eight different states. Mr. Stivers served as Treasurer and Chief Financial Officer for Clay Resource Company and Senior Tax and Audit Specialist for Gallaher and Company. He received a Bachelor of Science degree in accounting from Eastern Kentucky University.

*Herman E. Gettelfinger*. Mr. Gettelfinger has been a member of our Board of Directors since 1997. Mr. Gettelfinger, who has been active in the gas and oil drilling and exploration business for more than 35 years, is a co-owner and President of Kelso Oil Company, Knoxville Tennessee. Kelso is one of eastern Tennessee s largest distributors of motor oils, fuels and lubricants to the industrial and commercial market.

General Merrill A. McPeak (USAF, retired). General McPeak has been a member of our Board of Directors since April 2010. General McPeak has served as President of McPeak and Associates, a 15-year old management consulting firm, since its founding in 1995. From October 1990 until October 1994, he was Chief of Staff of the U.S. Air Force and a member of the Joint Chiefs of Staff. During this period, he was the senior officer responsible for organization, training and equipage of a combined active duty, National Guard, Reserve and civilian work force of over 850,000 people serving at 1,300 locations in the United States and abroad. As a member of the Joint Chiefs of Staff, he and the other service chiefs were military advisors to the Secretary of Defense, the National Security Council and the President of the United States. Following retirement from active service, General McPeak began a second career in business. He is Chairman of Ethicspoint, a privately-held, Portland-based provider of risk management and compliance software-as-a-service, including secure, anonymous reporting of ethical violations in the workplace. General McPeak has also served as a member of the Board of Director for Del Global Technologies Corp. (since 2005), Point Blank Solutions, Inc. (since 2008), Tektronix, Inc. (1995 to 2006); Ouintessence Photonics Corp. (2006) to 2009), Blacklight Power Inc. (2003 to present), Health Sciences Group, Inc. (2005 to 2007), MathStar, Inc. (2005 to February 2010), Gigabeam Corp. (2004 to 2009), TWA (1997 to 2002), ECC International (1997 to 2003), Praegitzer Industries (1997 to 1999) and Western Power and Equipment (1998 to 2000). In 1992, San Diego State University honored General McPeak with its first ever Lifetime Achievement Award. In 1995, George Washington University gave him its Distinguished Alumni Award, the George. He was among the initial seven inductees to the Oregon Aviation Hall of Honor. He is a member of the Council on Foreign Relations, New York City, and in 2008 and 2009 was a national co-chairman of Obama for President.

*Jonathan S. Gross*. Mr. Gross has been a member of our Board of Directors since April 2010. Mr. Gross has 29 years of experience in domestic and international oil and gas exploration and currently serves as Senior Vice President - Geosciences for Energy Partners, Ltd. (NYSE: EPL). From June 2009 to May 2010, he served as President of Jexco, LLC, a Houston, Texas geological and geophysical consulting company. During his career, Mr. Gross has served as the Chief Operating Officer of Houston Exploration Services, Inc., a Houston, Texas based subsidiary of Kuwait Energy Company from July 2008 to May 2009, Senior Vice President of Exploration and Technology Manager of Cheniere Energy, Inc. from June 1999 to July 2008, and has also worked for Zydeco Energy, Inc. from January 1998 to May 1999. He has domestic and international experience in both onshore and offshore basins in several parts of the world including the U.S., Trinidad, West Africa, North Africa, the Middle East and Eurasia. Mr. Gross received his B.A. in Geophysical Sciences from the University of Chicago in 1981 and is a member of the American Association of Petroleum Geologists, the Society of Exploration Geophysicists and the Houston Geological Society.

*David J. Voyticky*. Mr. Voyticky has been a member of our Board of Directors since April 2010. Mr. Voyticky is has over 15 years of domestic and international mergers and acquisitions, restructuring and financing experience. Since August 2005, Mr. Voyticky has been an independent consultant to companies in the middle market on value maximization strategies. As part of this focus, Mr. Voyticky provides strategic and capital markets advice to high

growth businesses He served as a vice president with Goldman, Sachs & Co. from June 2000 to May 2002, a vice president of Houlihan Lokey Howard & Zukin Capital, Inc. in Los Angeles from July 2002 to January 2005, and an associate with J.P. Morgan in London and New York from June 1996 to May 2000. During that period, he advised public and private domestic and multinational corporations and financial sponsors on mergers, acquisitions, divestitures, joint ventures, cross-border transactions, anti-raid (defense) preparation and capital-raising activities. Mr. Voyticky designed and was a founding partner of Red Mountain Capital Partners.

From December 2005 through June 2006, Mr. Voyticky was a partner in the \$300 million re-launch of Chapman Capital L.L.C., an activist hedge fund focused on the publicly traded middle market companies. Since January 2010 he has been a member of the Board of Directors of Best Energy Services, Inc. Mr. Voyticky received a J.D. and a M.B.A degree from the University of Michigan and a Masters in International Policy and Economics from the Ford School at the University of Michigan. He also received a Bachelor of Arts in Philosophy from Pomona College.

There are no family relationships between any of the executive officers and directors, except as set forth above. Each director is elected at our annual meeting of shareholders and holds office until the next annual meeting of shareholders, or until his successor is elected and qualified. As a term of the acquisition of Cook Inlet, we agreed Cook Inlet s owners prior to the acquisition would be represented by a seat on our Board of Directors for a period of three years from December 11, 2009. Mr. Hall has initially been designated as the director representing Cook Inlet s prior owners. In the event Mr. Hall should die or otherwise become incapacitated or unavailable to act as director, Mr. Wilcox will be designated as the successor directors and thereafter Mr. Stafford. Under the terms of the November 2009 letter agreement with Vulcan Capital Corporation, LLC, upon closing of the loan agreement Vulcan Capital Corporation will have the right to designate a member of our Board of Directors during the term of the loan.

# **Director Qualification**

The following is a discussion for each director of the specific experience, qualifications, attributes or skills that led to our conclusion that such person should be serving as a member of our Board of Directors as of the date of this annual report in light of our business and structure. In addition to their individual skills and backgrounds which are focused on our industry as well as financial and managerial experience, we believe that the collectively skills and experience of our Board members are well suited to guide us as we continue to grow our company.

*Deloy Miller* Mr. Miller has extensive experience as a seasoned gas and oil professional. our company. Mr. Miller has more than 40 years of experience in the drilling and production business in the Appalachian basin, extensive geological knowledge of Tennessee and Kentucky, training in the reading of well logs, and particular familiarity with our operations as our founder, former Chief Executive Officer, and current Chief Operating Officer.

*Scott M. Boruff* Mr. Boruff has experience in the financial industry, specializing in investment banking consulting services that included structuring of direct financings, recapitalizations, mergers and acquisitions and strategic planning with an emphasis in the gas and oil field.

*David M. Hall* Mr. Hall has a comprehensive knowledge of our Alaskan operations, with nearly 20 years of experience with our Alaskan assets, together with engineering expertise in which he trained as both an electrical engineer and industrial engineer.

*Herman E. Gettelfinger* Mr. Gettelfinger has over 35 years of experience in the gas and oil drilling and exploration business including as co-owner and President of Kelso Oil Company, one of East Tennessee s largest distributors of motor oils, fuels and lubricants to the industrial and commercial market.

*Jonathan S. Gross* Mr. Gross has 29 years of experience in domestic and international oil and gas exploration and education and is trained as a geologist. Mr. Gross has experience in both onshore and offshore basins in several parts of the world, has held various positions in several energy companies and is a geologist.

*Merrill A. McPeak* General McPeak has extensive experience in management consulting and a successful military career, including his position as Chief of Staff of the U.S. Air Force and a member of the Joint Chiefs of Staff, during which time he was the senior officer responsible for organization, training and equipage of a combined active duty, National Guard, Reserve and civilian work force of over 850,000 people serving at 1,300 locations in the United States and abroad and advised the Secretary of Defense, the National Security Council and the President of the United States. General McPeak currently serves or has served in the past on the Board of Directors of a number of publicly

traded companies.

*Charles M. Stivers* Mr. Stivers, a certified public accountant, has over 18 years of experience in accounting and over 12 years of experience within the energy industry. Mr. Stivers owns and operates an accounting firm that specializes in the oil and gas industry with clients in eight different states.

*David J. Voyticky* Mr. Voyticky has over 15 years of domestic and international mergers and acquisitions, restructuring and financing experience and education and training, with experience as an independent consultant to companies in the middle market on value maximization strategies, providing strategic and capital markets advice to high growth businesses.

# **Director Compensation**

We have not established standard compensation arrangements for our directors and the compensation payable to each individual for their service on our Board is determined from time to time by our Board of Directors based upon the amount of time expended by each of the directors on our behalf. Currently, executive officers of our company who are also members of the Board of Directors do not receive any compensation specifically for their services as directors. Mr. David M. Hall, an employee of our company, also serves on our Board of Directors granted options under our stock option plan exercisable at fair market value on the date of grant as compensation to our non-employee directors for their services to us. Our Board of Directors intends to adopt a policy regarding compensation of non-employee directors prior to the end of our current fiscal year.

The following table provides information about compensation paid to our non-employee directors during the 2010 for their services as directors. The value of the securities issued reflects the aggregate grant date fair value computed in accordance with ASC Topic 718. While options were granted to these individuals as described below, because none of these options have vested and the grant is subject to continued Board service, under generally accepted accounting principles we will recognize compensation expense for these grants over the vesting period.

	Fees Paid or Earned				Change in Pension Value and		
	in Cash	Stock	Option	Non-Equity Incentive Plan	Nonqualified Deferred Compensation	All Other Compensation	Total
Name	(\$)	Awards (\$)	Awards (\$)	Compensation (\$)	Earnings (\$)	(\$)	(\$)
(a)	<b>(b)</b>	(c)	( <b>d</b> )	(e)	<b>(f)</b>	<b>(g)</b>	( <b>h</b> )
Charles M. Stivers <sup>1</sup> Herman E.	0	0	0	0	0	0	0
Gettelfinger <sup>2</sup> Merrill A.	0	0	0	0	0	0	0
McPeak <sup>3</sup>	0	0	0	0	0	0	0
Jonathan S. Gross <sup>4</sup> David J. Voyticky	0	0	0	0	0	0	0
5	0	0	0	0	0	0	0

1

Mr. Stivers was granted options to purchase an aggregate of 100,000 shares of our common stock at an exercise price of \$5.94 per share, of which options to purchase 33,333 shares vest on April 27, 2011, options to purchase an additional 33,333 shares vest on April 27, 2012 and options to purchase the remaining 33,334 shares vest on April 27, 2013.

2

Mr. Gettelfinger was granted options to purchase an aggregate of 100,000 shares of our common stock at an exercise price of \$5.94 per share, of which options to purchase 33,333 shares vest on April 27, 2011, options to purchase an additional 33,333 shares vest on April 27, 2012 and options to purchase the remaining 33,334 shares vest on April 27, 2013.

3

General McPeak was granted options to purchase an aggregate of 200,000 shares of our common stock at an exercise price of \$5.94 per share, of which options to purchase 66,667 shares vest on April 27, 2011, options to purchase an additional 66,667 shares vest on April 27, 2012 and options to purchase the remaining 66,666 shares vest on April 27, 2013.

4

Mr. Gross was granted options to purchase an aggregate of 200,000 shares of our common stock at an exercise price of \$5.94 per share, of which options to purchase 66,667 shares vest on April 27, 2011, options to purchase an additional 66,667 shares vest on April 27, 2012 and options to purchase the remaining 66,666 shares vest on April 27, 2013.

5

Mr. Voyticky was granted options to purchase an aggregate of 200,000 shares of our common stock at an exercise price of \$5.94 per share, of which options to purchase 66,667 shares vest on April 27, 2011, options to purchase an additional 66,667 shares vest on April 27, 2012 and options to purchase the remaining 66,666 shares vest on April 27, 2013.

## **Code of Business Conduct and Ethics**

We have adopted a Code of Business Conduct and Ethics that applies to our President, Chief Executive Officer, Chief Financial Officer Chief Accounting Officer or Controller and any other persons performing similar functions. This Code provides written standards that we believe are reasonably designed to deter wrongdoing and promote honest and ethical conduct, including the ethical handling of actual or apparent conflicts of interest between personal and professional relationships, and full, fair, accurate, timely and understandable disclosure in reports we file with the Securities Exchange Commission. A copy of our Code of Business Conduct and Ethics is available on our website site. It has also been filed with the Securities and Exchange Commission as an exhibit to this annual report.

## **Committees of the Board of Directors**

Our Board of Directors has established an Audit Committee, a Compensation Committee and a Nominating and Corporate Governance Committee. While our Board of Directors established the Audit Committee in 2004, the Compensation Committee and Nominating and Corporate Governance Committee were only recently established in conjunction with the listing of our common stock on the NASDAQ Stock Market.

*Audit Committee*. The Audit Committee assists the Board in fulfilling its oversight responsibility relating to the integrity of our financial statements, our compliance with legal and regulatory requirements and the qualifications and independence of our independent registered public accountants. The Audit Committee is composed of three directors, Messrs. Stivers (Chairman) and Voyticky and General McPeak, all of whom have been determined by the Board of Directors to be independent, as defined by the Marketplace Rules of the NASDAQ Stock Market. The Board has determined that Mr. Stivers, the Chairman of the Audit Committee, qualifies as an audit committee financial expert as defined by the SEC.

*Compensation Committee*. The Compensation Committee is responsible for overseeing our compensation programs and practices, including our executive compensation plans and incentive compensation plans. The Chief Executive Officer will provide input to the Compensation Committee with respect to the individual performance and compensation recommendations for the other executive officers. The Compensation Committee is composed of three directors, Messrs. Voyticky (Chairman), Stivers and Gettelfinger, all of whom have been determined by the Board of Directors to be independent, as defined by the Marketplace Rules of the NASDAQ Stock Market.

*Nominating and Corporate Governance Committee.* The Nominating and Corporate Governance Committee will recommend the slate of director nominees for election to our Board of Directors, identify and recommend candidates to fill vacancies occurring between annual stockholder meetings, review the composition of Board committees and monitor compliance with, review, and recommend changes to our various corporate governance policies and guidelines. The committee will also prepare and supervise the Board s annual review of director independence and the Board s annual self-evaluation. The Nominating and Corporate Governance Committee is composed of three directors, Messrs. Gettelfinger (Chairman), Voyticky and Gross, all of which have been determined by the Board of Directors to be independent, as defined by the Marketplace Rules of the NASDAQ Stock Market.

The Nominating and Corporate Governance Committee will consider all qualified candidates for our Board of Directors identified by members of the committee, by other members of the Board of Directors, by senior management and by our stockholders. The committee will review each candidate, including each candidate s independence, skills and expertise based on a variety of factors, including the person s experience or background in management, finance, regulatory matters and corporate governance. Further, when identifying nominees to serve as director, the Nominating and Corporate Governance Committee will seek to create a Board that is strong in its collective knowledge and has a diversity of skills and experience with respect to accounting and finance, management and leadership, vision and strategy, business operations, business judgment, industry knowledge and corporate governance. In addition, prior to nominating an existing director for re-election to the Board of Directors, the Nominating and Corporate Governance Committee will consider and review an existing director s Board and

committee attendance and performance, length of Board service, experience, skills and contributions that the existing director brings to the Board, equity ownership in our company and independence.

The Nominating and Corporate Governance Committee will follow the same process and use the same criteria for evaluating candidates proposed by stockholders, members of the Board of Directors and members of senior management. Based on its assessment of each candidate, the committee will recommend candidates to the

Board. However, there is no assurance that there will be any vacancy on the Board at the time of any submission or that the committee will recommend any candidate for the Board.

# **ITEM 11.**

.

•

# EXECUTIVE COMPENSATION.

The following table summarizes all compensation recorded by us in fiscal 2010 for the following:

our principal executive officer or other individual serving in a similar capacity,

our two most highly compensated executive officers other than our principal executive officer who were serving as executive officers at April 30, 2010 as that term is defined under Rule 3b-7 of the Securities Exchange Act of 1934, and

up to two additional individuals for whom disclosure would have been required but for the fact that the individual was not serving as an executive officer at April 30, 2010.

For definitional purposes, these individuals are sometimes referred to as the named executive officers. The value attributable to any option awards in the following table is computed in accordance with ASC Topic 718. The value of the securities issued reflects the aggregate grant date fair value computed in accordance with ASC Topic 718 assuming the following weighted averages:

Expected life (in years)	3.0
Volatility	221.97%
Discount rate - bond equivalent rate	1.55%
Annual rate of quarterly dividends	0.00

### **Summary Compensation Table**

### **NON-EQUITY**

						INCENTIVE	NONQUALIFIED	ALL	
				STOCK	OPTION	PLAN	DEFERRED	OTHER	
ME AND		SALARY	BONUS	AWARDS	AWARDS	COMPENSATION	COMPENSATION	COMPENSATION	тот
NCIPAL SITION	YEAR	(\$)	(\$)	(\$)	(\$)	(\$)	EARNINGS (\$)	(\$)	(\$
	<b>(B)</b>	( <b>C</b> )	<b>(D</b> )	(E)	<b>(F)</b>	( <b>G</b> )	<b>(H</b> )	<b>(I</b> )	(J

2010	260,228	576,547	0	0	0	0	12,640	849,
2009	182,755	283,000	20,625	0	0	0	9,059	495,
2010	203,846	0	0	0	0	0	1,161	205,
2009	200,000	0	0	0	0	0	2,244	202,
2010	73,077	200,000	0	314,936	0	0	0	588,
	2009 2010 2009	2009         182,755           2010         203,846           2009         200,000	2009         182,755         283,000           2010         203,846         0           2009         200,000         0	2009         182,755         283,000         20,625           2010         203,846         0         0           2009         200,000         0         0	2009         182,755         283,000         20,625         0           2010         203,846         0         0         0           2009         200,000         0         0         0	2009         182,755         283,000         20,625         0         0           2010         203,846         0         0         0         0           2009         200,000         0         0         0         0	2009       182,755       283,000       20,625       0       0       0         2010       203,846       0       0       0       0       0         2009       200,000       0       0       0       0       0	2009       182,755       283,000       20,625       0       0       0       9,059         2010       203,846       0       0       0       0       0       1,161         2009       200,000       0       0       0       0       2,244

1

Mr. Boruff has served as our Chief Executive Officer since August 2008 and the terms of his compensation are set forth in his employment agreement which is described later in this section. Mr. Boruff is entitled to a bonus each year equal to 100% of his base salary and 100,000 shares of our common stock if we meet certain revenue and EBITDA milestones. These milestones were met in each of 2010 and 2009. Mr. Boruff s bonus in 2010 and 2009 included \$319,500 and \$33,000, respectively, attributable to the value of 100,000 shares of our common stock issued to him. The value of stock awards and option awards in each of 2010 and 2009 represents the value of restricted stock awards and option grants made to him in each of those years under the terms of his employment agreement. All other compensation for both 2010 and 2009 included an auto allowance of \$1,000 per month plus \$640 and \$59, respectively, of compensation derived from personal use of a company vehicle. The amount of Mr. Boruff s compensation excludes fees paid to Dimirak Securities Corporation, a broker-dealer and member of FINRA, and its related parties under the terms of a Marketing Agreement. Mr. Boruff is a director and owns 49% of Dimirak Securities Corporation. See Item 14. Certain Relationships and Related Transactions and Director Independence appearing later in this annual report. Mr. Boruff s compensation for 2010 excludes the value of options granted to him in April 2010 which have not yet vested.

#### 2

Mr. Miller served as our Chief Executive Officer from December 1997 to August 2008 and is currently our Chief Operating Officer. All other compensation included \$1,161 and \$2,244 of compensation derived from personal use of a company vehicle in 2010 and 2009, respectively. Mr. Miller s compensation for 2010 excludes the value of options granted to him in April 2010 which have not yet vested.

3

Mr. Graham served as our President and Vice-Chairman of our Board of Directors from December 2009 until June 2010. Option awards in 2010 represent the value of 10 year warrants which were immediately exercisable to purchase an aggregate of 1,000,000 shares of our common stock, with warrants to purchase 400,000 shares exercisable at \$0.01 per share, warrants to purchase 200,000 shares exercisable at \$0.69 per share, warrants to purchase 200,000 shares exercisable at \$0.69 per share, warrants to purchase 200,000 shares exercisable at \$2.00 per share. Mr. Graham has transferred the \$0.01, \$1.00 and \$2.00 warrants to Vulcan Capital Corporation, a company of which he is a control person. Mr. Graham s compensation for 2010 excludes the value of options granted to him in April 2010 which have not yet vested. Because Mr. Graham resigned prior to the vesting of any of these options, these options have lapsed and the shares underlying them have been returned to the pool available for future grants under the terms of our Stock Plan.

# **Employment Agreement with Mr. Boruff**

Effective August 1, 2008, we entered into an employment agreement, as amended in September 2008, with Mr. Scott M. Boruff pursuant to which Mr. Boruff will serve as our Chief Executive Officer for an initial term of five years, subject to additional one-year renewal periods. Under the terms of the agreement, as amended, Mr. Boruff s compensation consists of the following:

a base salary of \$250,000 per annum, with provision for cost-of-living increases,

•

.

10 year options to purchase 250,000 shares of our common stock at an exercise price per share of \$0.33, with vesting in equal annual installments over a period of four years from the grant date, or immediately upon a change of control of our company as described in the agreement, and

•

a restricted stock grant of 250,000 shares of common stock, with vesting in equal annual installments over a period of four years from the issuance date, or on an accelerated basis in the event of a change of control of our company also as described in the agreement.

Mr. Boruff is also entitled to receive certain incentive compensation in the form of cash and shares of our common stock based upon, and subject to, two performance benchmarks, gross revenue and earnings before income taxes, depreciation and amortization (EBITDA), as follows:

•

100% of his base salary and 100,000 shares of our common stock in the event that our gross revenues for fiscal 2009 (annualized beginning on the date of the agreement) were not less than \$2,000,000 and EBITDA for such period was not less than \$200,000,

100% of his base salary and 100,000 shares of our common stock in the event that our gross revenues for fiscal 2010 are not less than \$4,000,000 and EBITDA for such period was not less than \$400,000,

100% of his base salary and 100,000 shares of our common stock in the event that our gross revenues for fiscal 2011 are not less than \$8,000,000 and EBITDA for such period was not less than \$800,000,

100% of his base salary and 100,000 shares of our common stock in the event that our gross revenues for fiscal 2012 are not less than \$16,000,000 and EBITDA for such period was not less than \$1,600,000, and

100% of his base salary and 100,000 shares of our common stock in the event that our gross revenues for fiscal 2013 are not less than \$30,000,000 and EBITDA for such period was not less than \$3,000,000.

One half of each element of incentive compensation is earned if the gross revenue benchmark is achieved, and the other half of each element is earned if the EBITDA benchmark is achieved.

Mr. Boruff is also entitled to a \$1,000 per month automobile allowance. The employment agreement also provides that Mr. Boruff is entitled to participate in the employee benefit plans, programs and arrangements we have in effect during the employment term which are generally available to our senior executives. The agreement also contains indemnification, confidentiality and non-solicitation clauses.

The agreement may be terminated by us for cause, as defined in the agreement, or upon his death or disability, or for no cause. In the event the agreement is terminated for either reason, if Mr. Boruff should terminate the agreement for any reason or if the agreement is not renewed, he is only entitled to receive his base salary through the date of termination. We may also terminate the agreement without cause, in which event Mr. Boruff will be entitled to his base salary through the date of termination and, should we terminate the agreement during the initial term, as severance, his base salary for one year. If we should terminate the agreement as a result of a change of control as defined in the agreement, he is entitled to a lump sum payment equal to 2.99 times Mr. Boruff s then base salary.

In addition to the compensation payable to him under the terms of his employment agreement, in April 2010 the Compensation Committee of the Board of Directors granted Mr. Boruff options under our stock option plan to purchase 500,000 shares of our common stock with an exercise price of \$5.94 per share as additional compensation. These options vest over three years in arrears commencing on April 27, 2011, and are subject to continued employment. The Compensation Committee determined to make this award of additional compensation to Mr. Boruff in recognition of his past performance and a desire to retain him throughout the three year vesting period.

#### How Mr. Miller's Compensation was Determined

Mr. Miller, who served as our principal executive officer until December 1997 to August 1, 2008, is not a party to an employment agreement with our company. His compensation is determined by the Chief Executive Officer. The Board considered a number of factors in determining Mr. Miller's compensation including the scope of his duties and responsibilities to our company and the time he devotes to our business. The Board of Directors did not consult with any experts or other third parties in fixing the amount of Mr. Miller's compensation. During each of 2010 and 2009, Mr. Miller's compensation package included a base salary of \$200,000. We also provide him with a company vehicle. In addition, in April 2010 the Compensation Committee of the Board of Directors granted Mr. Miller options under our stock option plan to purchase 350,000 shares of our common stock with an exercise price of \$5.94 per share as additional compensation. These options vest over three years in arrears commencing April 27, 2011, and are subject to continued employment. The Compensation Committee determined to make this award of additional compensation to Mr. Miller in recognition of his past performance and a desire to retain him throughout the three year vesting period.

### How Mr. Graham s Compensation was Determined

Mr. Graham served as our President from December 2009 to June 2010. We were not a party to an employment agreement with him, but at the time of his hiring our Board approved annual base compensation of \$200,000, a one-time signing bonus of \$200,000, a seat on our Board of Directors as Vice-Chairman and other compensation which is customarily paid to our other senior officers. In addition, on December 10, 2009 we granted him 10 year warrants which were immediately exercisable to purchase an aggregate of 1,000,000 shares of our common stock valued at \$314,936 as additional compensation. Mr. Graham has transferred certain of the warrants to Vulcan Capital Corporation, a company of which he is a control person. The Board considered a number of factors in determining Mr. Graham s compensation including the scope of his duties and responsibilities to our company. The Board of Directors did not consult with any experts or other third parties in fixing the amount of Mr. Graham's compensation. In addition, in April 2010 the Compensation Committee of the Board of Directors granted Mr. Graham options under our stock option plan to purchase 100,000 shares of our common stock with an exercise price of \$5.94 per share as additional compensation. These options vest over three years in arrears commencing April 27, 2011, and are subject to continued employment. The Compensation Committee determined to make this award of additional compensation to Mr. Graham in recognition of his past performance and a desire to retain him throughout the three year vesting period. Because Mr. Graham resigned prior to the vesting of any of these options, these options have lapsed and the shares underlying them have been returned to the pool available for future grants under the terms of our Stock Plan.

#### Severance Agreement with Mr. Graham

In June 2010 Mr. Graham resigned as an executive officer and director of our company. Following his resignation we entered into a Separation Agreement and General Release with Mr. Graham pursuant to which he will receive six month s salary as severance and he will be entitled to retain the warrants to purchase 1,000,000 shares of our common stock with exercise prices ranging from \$0.01 per share to \$2.00 per share granted to him when he joined our company in December 2009. The options to purchase an additional 100,000 shares of our common stock at an exercise price of \$5.94 per share granted in April 2010 were contingent upon continued service

and will, accordingly, terminate. The Separation Agreement and General Release contains continuing indemnification and confidentially provisions and a general release to us from Mr. Graham.

#### How Mr. Boyd s Compensation was Determined

Mr. Boyd, who has served as our Chief Financial Officer since September 2008, is not a party to an employment agreement with our company. His compensation is determined by our Chief Executive Officer who considered a number of factors in determining Mr. Boyd's compensation including the scope of his duties and responsibilities to our company and the time he devotes to our business. We did not consult with any experts or other third parties in fixing the amount of Mr. Boyd's compensation. During each of 2010 and 2009, Mr. Boyd s compensation package included a base salary and an automobile allowance of \$500 per month. At the time he joined our company we granted Mr. Boyd two year options to purchase 250,000 shares of our common stock at an exercise price of \$0.40 per share, of which options to purchase 125,000 shares vested on the date of grant and the remaining options vested on March 31, 2010. In addition, in February 2010 our Board granted Mr. Boyd five year options to purchase 25,000 shares of our common stock at an exercise price of \$2.52 per share which vested on May 19, 2010. In addition, in April 2010 the Compensation Committee of the Board of Directors granted Mr. Boyd options under our stock option plan to purchase 350,000 shares of our common stock with an exercise price of \$5.94 per share as additional compensation. These options vest over three years in arrears commencing April 27, 2011, and are subject to continued employment. The Compensation Committee determined to make this award of additional compensation to Mr. Boyd in recognition of his past performance and a desire to retain him throughout the three year vesting period. Mr. Boyd s total compensation for 2010 was \$165,368.

### **Outstanding Equity Awards at Fiscal Year-End**

The following table provides information concerning unexercised options, stock that has not vested and equity incentive plan awards for each named executive officer outstanding as of April 30, 2010:

<b>OPTION</b> A	WARDS			STOCK AWARDS				
	Equity							
	Incentive						Equity incentive	
	plan awards:			Number		Equity	plan awards:	
Number of	Number of			of shares	Market	incentive plan	Market or	
Securities	Securities			or units	value of shares or	awards: Number	payout value of	
Underlying	Underlying			of stock	units of	of unearned	unearned shares,	
Unexercised	Unexercised	Option		that have	stock that	shares, units	units or other	
options	Unearned	Exercise	Option	not	have not	or other rights	rights that have	
(#)	options	price	Expiration	vested	vested	that have not	not vested	
unexercisable	(#)	(\$)	date	(#)	(\$) <sup>1</sup>	vested (#)	( <b>\$</b> ) <sup>1</sup>	
( <b>c</b> )	( <b>d</b> )	<b>(e)</b>	( <b>f</b> )	<b>(g)</b>	( <b>h</b> )	(i)	( <b>j</b> )	

### Edgar Filing: MILLER PETROLEUM INC - Form 10-K

	187,500		0.33	8/1/2018		1,083,750 1,734,000		50,000	289,000	
					300,000	1,734,000		50,000	289,000	
25.04)	\$	2.32	\$		(16.12)	\$	4.63		\$	5.
9,971		292,669,971		292	.,669,971	2	292,669,971			292,669,9
2,707	\$	3,570	\$		4,980	\$	4,905		\$	5,4
2,667)		(3,258)			(4,699)		(5,126)			(5,6
							123			

	As of September 30,		As	of December	31.	
	2012	2011	2010	2009 illions)	2008	2007
Consolidated Balance Sheet Data:						
Current assets	\$ 5,343	\$ 6,602	\$ 5,311	\$ 5,845	\$ 5,951	\$ 6,553
Property and equipment, net	12,535	12,703	13,213	13,192	12,600	11,258
Goodwill, spectrum licenses and other intangible						
assets, net	14,403	21,009	27,439	27,440	27,477	25,393
Other assets	332	295	328	297	262	155
Total assets	32,613	40,609	46,291	46,774	46,290	43,359
Current liabilities	4,862	4,504	4,455	8,149	5,978	5,297
Long-term payables to affiliates	13,620	15,049	15,854	9,682	13,850	6,712
Other long-term liabilities	5,655	5,271	5,490	4,693	3,679	2,537
Stockholder s equity	8,476	15,785	20,492	24,250	22,783	28,813

#### COMPARATIVE PER SHARE DATA

The following table sets forth selected historical per share information of T-Mobile and MetroPCS and unaudited pro forma per share information after giving effect to the transaction, assuming that 1.8368 shares of MetroPCS common stock had been issued in exchange for each outstanding share of T-Mobile common stock.

You should read this information in conjunction with the selected historical financial information, the unaudited pro forma condensed combined financial information and the separate historical financial statements of T-Mobile and MetroPCS and the notes thereto included elsewhere in or incorporated by reference in this proxy statement. The historical per share information as of and for the nine-month period ended September 30, 2012 is derived from unaudited condensed consolidated financial statements of T-Mobile and MetroPCS, respectively, as of and for the nine months ended September 30, 2012. The historical per share information for the year ended December 31, 2011 is derived from audited consolidated financial statements of T-Mobile and December 31, 2011. The unaudited pro forma condensed combined financial statements and the pro forma per share information are not necessarily indicative of the operating results or financial position that would have been achieved had the transaction been completed at the beginning of the period presented and should not be construed as representative of future operations. Neither T-Mobile nor MetroPCS declared any dividends related to their respective common stock during the periods presented.

	Year Ended December 31, 2011				
	MetroPCS T-Mobile			obile	
				Pro Forma	
				Equivalent	
				of one	
				MetroPCS	
	Historical	Pro Forma	Historical	Share (1)	
Basic net income (loss) per common share	\$ 0.83	\$ (6.50)	\$ (16.12)	\$ (11.94)	
Diluted net income (loss) per common share	\$ 0.82	\$ (6.50)	\$ (16.12)	\$ (11.94)	

	Nine Months Ended September 30, 2012			
	MetroPCS T-Mobile			Iobile
				Pro Forma
				Equivalent
				of one
				MetroPCS
	Historical	Pro Forma	Historical	Share (1)
Basic net income (loss) per common share	\$ 0.99	\$ (9.91)	\$ (25.04)	\$ (18.20)
Diluted net income (loss) per common share	\$ 0.99	\$ (9.91)	\$ (25.04)	\$ (18.20)
Book value per share (2)	\$ 9.13	\$ 11.08	\$ 28.96	\$ 20.35

(1) These amounts were calculated by applying an assumed exchange ratio of 1.8368 times the unaudited pro forma financial information.

(2) The historical book value per common share is computed by dividing total stockholders equity by the number of shares of common stock outstanding as of September 30, 2012. The pro forma book value per share is computed by dividing pro forma stockholders equity by the pro forma number of shares of common stock outstanding as of September 30, 2012.

#### COMPARATIVE PER SHARE MARKET PRICE AND DIVIDEND INFORMATION

Shares of MetroPCS are currently listed and principally traded on the NYSE under the symbol PCS. T-Mobile is not publicly traded and is a wholly-owned indirect subsidiary of Deutsche Telekom. The following table sets forth, for the periods indicated, the high and low sales price per share of MetroPCS common stock as reported on the NYSE:

Fiscal Year Ending December 31, 2012	High	Low
Fourth Quarter (through November 12, 2012)	13.57	9.83
Third Quarter	11.95	6.23
Second Quarter	9.21	5.59
First Quarter	12.01	8.01
Fiscal Year Ending December 31, 2011		
Fourth Quarter	9.73	7.51
Third Quarter	17.77	8.71
Second Quarter	18.69	15.94
First Quarter	16.32	12.53
Fiscal Year Ending December 31, 2010		
Fourth Quarter	12.74	10.29
Third Quarter	10.49	8.30
Second Quarter	9.15	7.15
First Quarter	7.99	5.53

The table below sets forth the closing price of MetroPCS on October 2, 2012, the last trading date prior to the public announcement of the transaction and as of November 12, 2012, the most recent practicable trading day prior to the date of this proxy statement. The market prices of MetroPCS common stock likely will fluctuate between the date of this proxy statement and the time of the special meeting and the completion of the transaction. No assurance can be given concerning the market prices of MetroPCS common stock before the completion of the transaction or the market price of the combined company s common stock after the completion of the transaction. Immediately following the transaction, Holding or its designee will own 74% of the fully-diluted shares of MetroPCS common stock outstanding immediately following the cash payment and MetroPCS equityholders immediately prior to the transaction collectively will own the remaining 26% of MetroPCS common stock on a fully-diluted basis. These percentages are fixed in the business combination agreement and will not be adjusted for changes in the business, financial condition or operating results of MetroPCS or T-Mobile or changes in the market price of MetroPCS common stock. As a result, the stock price of the combined company s common stock that MetroPCS stockholders will receive in the transaction may vary significantly from the prices shown in the table below.

		MetroPCS	
		Com	mon Stock
October 2, 2012		\$	13.57
November 12, 2012		\$	10.05
	<u>0 · 1 · 1 · 1 / 1 / 1 / 1 / 1 / 1 / 1 / 1</u>		1

MetroPCS stockholders should obtain current market prices for shares of MetroPCS in deciding whether to vote for the approval of the proposals.

#### Holders

As of November 12, 2012, the latest practicable date prior to the filing of this proxy statement, there were 364,148,299 shares of MetroPCS common stock outstanding and approximately 30,000 holders of record of MetroPCS common stock.

#### Dividends

MetroPCS has never declared or paid cash dividends on its capital stock and does not expect to pay any cash dividends in the foreseeable future.

#### SUMMARY UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

The following unaudited pro forma condensed combined financial information gives effect to the transaction between MetroPCS and T-Mobile which is to be accounted for as a reverse acquisition with T-Mobile treated as the accounting acquirer. The unaudited pro forma condensed combined statements of operations data for the year ended December 31, 2011 and for the nine months ended September 30, 2012 reflects the transaction as if it occurred on January 1, 2011. The unaudited pro forma condensed combined balance sheet data as of September 30, 2012 reflects the transaction as if it had occurred on September 30, 2012. The pro forma adjustments are based on the information available at the time of the preparation of this proxy statement. See the section entitled

T-Mobile and MetroPCS Unaudited Pro Forma Condensed Combined Financial Information beginning on page 196.

The unaudited pro forma condensed combined financial information that follows is presented for informational purposes only and is not intended to represent or be indicative of the combined results of operations or financial position that would have been reported had the transaction been completed as of January 1, 2011 or September 30, 2012, and should not be taken as representative of the future consolidated results of operations or financial position of the combined company. In connection with the unaudited pro forma condensed combined financial information, T-Mobile allocated the preliminary purchase price to the acquired assets and liabilities based upon their estimated fair value. These estimates are based on financial information available at the time of the preparation of this proxy statement. Based on the timing of the closing of the transaction and other factors, we and T-Mobile cannot assure that the actual adjustments will not differ materially from the pro forma adjustments reflected in the unaudited pro forma condensed combined financial information. It is expected that, following the transaction, the combined company will incur expenses associated with the transaction and integration of the operations of the two companies. These transaction and integration costs are not reflected in this unaudited pro forma condensed combined financial information. The unaudited pro forma condensed combined financial information. The unaudited pro forma condensed combined financial information also does not give effect to the potential impact of any anticipated synergies, operating efficiencies or cost savings that may result from the transaction.

	Nine Months Ended Year Ended September 30, 2012 December 31, 2 (in millions except shares and per share amounts)			nber 31, 2011
Statement of Operations Data				
Total revenues	\$	18,728	\$	25,600
Operating loss		(5,714)		(3,291)
Loss before income taxes		(6,785)		(4,873)
Net loss		(7,141)		(4,679)
Net loss per common share				
Basic and Diluted	\$	(9.91)	\$	(6.50)
Weighted average shares				
Basic and Diluted	72	0,929,874		719,539,741

	As of September 30, 2012 (in millions)
Balance Sheet Data	
Cash and cash equivalents	\$ 2,053
Current assets	7,657
Total assets	42,036
Current liabilities	4,168
Long-term payables to affiliates	15,000
Long-term debt, net	8,200
Stockholders equity	7,990

A c of

#### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This proxy statement and the documents referred to or incorporated by reference into this proxy statement contain forward-looking statements within the meaning of the safe harbor provisions of the Private Securities Litigation Reform Act of 1995, Section 27A of the Securities Act of 1933, as amended, which we refer to as the Securities Act, and Section 21E of the Exchange Act. Forward-looking statements are statements that are not statements of historical fact, including statements about beliefs, opinions and expectations. Forward-looking statements are based on, and include statements about, MetroPCS, Deutsche Telekom s, T-Mobile s and the combined company s plans, prospects, expected future financial condition, results of operations, cash flows, dividends and dividend plans, objectives, beliefs, financing plans, business strategies, budgets, goals, future events, future revenues or performance, financing needs, outcomes of litigation, projected costs, operating metrics, capital expenditures, competitive positions, acquisitions, investment opportunities, integration, cost savings, capital expenditures, synergies, growth opportunities, dispositions, plans and objectives of management for future operations and any other information that is not historical information. These statements generally can be identified by the fact that they do not relate strictly to historical or current facts and include, without limitation, words such as may. will. could. should. would. expect, plan, project. forecast. intend. anticipate. believe. estimate. potential, continue and similar expressions and variations. These statements are based on Deutsche Telekom s, T-Mobile s and pursue, target, MetroPCS intent, beliefs, opinions, expectations, and projections at the time such statements are made, and are not guarantees of future performance and are made only as of the date of this proxy statement or the applicable documents incorporated herein by reference. These statements involve risks, uncertainties, assumptions and other factors that are difficult to predict and that could cause actual results to differ materially from those expressed in them or indicated by them. Factors that could cause these differences include the factors identified under Risk Factors, beginning on page 33, as well as the following, among others:

the inability to have developed or to obtain handsets, equipment or software that our customers want, demand and expect or to have handsets, equipment or software serviced, updated, revised or maintained in a timely and cost-effective manner;

MetroPCS and T-Mobile s and each of their competitors current and planned promotions and advertising, marketing, sales and other initiatives, including pricing decisions, entry into consolidation and alliance activities, and MetroPCS and T-Mobile s ability to respond to and support them;

MetroPCS and T-Mobile s ability to manage their networks to deliver the services, content, applications, service quality and speed their customers expect and demand and to maintain and increase the capacity of their networks and business systems to satisfy the demands of their customers and the demands placed by devices on their networks;

the highly competitive nature of the wireless broadband mobile industry and changes in the competitive landscape;

the effects of the transaction on dealers, retailers, vendors, suppliers, customers, content and application providers, MetroPCS equity and debt holders and MetroPCS and T-Mobile employees;

the ability of MetroPCS and T-Mobile to operate their respective businesses in light of the transaction and the covenants contained in the business combination agreement;

the diversion of management s time and attention while the transaction is pending;

each of T-Mobile s and MetroPCS plans and expectations relating to, without limitation, (i) its growth opportunities and competitive position; (ii) its products and services; (iii) its customer experience; (iv) its results of operations, including expected synergies, earnings and cash flows; (v) the impact of the transaction on its credit ratings; and (vi) integration matters;

#### Edgar Filing: MILLER PETROLEUM INC - Form 10-K

the federal income tax consequences of the transaction and the enactment of additional state, federal, and/or foreign regulatory and tax laws and regulations;

expectations, intentions and outcomes relating to outstanding litigation, including securities, class action, derivative, patent and product safety claims, by or against third parties;

the possibility that the transaction is delayed or does not close, including due to the failure to receive the required MetroPCS stockholder approvals or required approvals from governmental authorities necessary to satisfy the closing conditions, along with satisfaction or waiver of other closing conditions, pursuant to the business combination agreement;

alternative acquisition proposals that could delay completion of the transaction or divert management s time and attention from the transaction;

T-Mobile s and MetroPCS ability to successfully integrate their businesses and realize the expected spectrum, cost and capital expenditure savings and synergies and other benefits from the transaction;

changes in economic, business, competitive, technological and/or regulatory factors, including the passage of legislation or action by governmental or regulatory entities to block the transaction;

any changes in the regulatory environment in which MetroPCS or T-Mobile operates, including any change or increase in restrictions on MetroPCS or T-Mobile s ability to operate its networks;

terminations of, or limitations imposed on, MetroPCS or T-Mobile s business by, contracts entered into by either MetroPCS or T-Mobile, or the effect of provisions with respect to change in control, exclusivity, commitments or minimum purchase amounts contained in such contracts;

the impact of economic conditions on MetroPCS and T-Mobile s business plans, strategies and stock prices;

the impact on MetroPCS and T-Mobile s networks and businesses from major equipment failures and security breaches related to the network or customer information;

the ability to obtain financing on terms favorable to MetroPCS and T-Mobile;

the impact of public and private regulations;

possible disruptions or intrusions of MetroPCS or T-Mobile s network, billing, operational support and customer care systems that may limit or disrupt their ability to provide service, or which may cause disclosure or improper use of customers information and associated harm to MetroPCS or T-Mobile s customers, systems, reputation and goodwill;

MetroPCS and T-Mobile s continued ability to offer a diverse portfolio of wireless devices, some on an exclusive basis;

MetroPCS and T-Mobile s ability to obtain and continue to obtain roaming on terms that are reasonable;

#### Edgar Filing: MILLER PETROLEUM INC - Form 10-K

severe weather conditions, natural disasters, energy shortages, wars or terrorist attacks, and any resulting financial impact not covered by insurance;

disruptions of MetroPCS and/or T-Mobile s key suppliers provisioning of products, services, content or applications;

fluctuations in interest and exchange rates;

significant increases in benefit plan costs or lower investment returns on plan assets;

material adverse changes in labor matters, including labor negotiations or additional organizing activity, and any resulting financial and/or operational impact;

the diversion of management s time and attention to litigation relating to the transaction;

write-offs in connection with the transaction, or changes in MetroPCS and/or T-Mobile s accounting assumptions that regulatory agencies, including the SEC, may require or that result from changes in the accounting rules or their application, which could result in an impact on earnings;

the significant capital commitments of MetroPCS and T-Mobile;

MetroPCS and T-Mobile s ability to remain focused and keep all employees focused on the business during the pendency of the transaction;

the current economic environment in the United States; disruptions to the credit and financial markets in the United States; and the impact of the economy on consumer demand and fluctuations in consumer demand generally for the products and services to be provided as a result of the transaction;

MetroPCS and T-Mobile s ability to manage their respective growth, achieve planned growth, manage churn rates, maintain their respective cost structures and achieve additional economies of scale;

MetroPCS and T-Mobile s ability to negotiate and maintain acceptable agreements with their respective suppliers and vendors, including obtaining roaming on reasonable terms;

the seasonality of MetroPCS and T-Mobile s respective businesses and any failure to have strong customer growth in the first and fourth quarters;

the rates, nature, collectability and applicability of taxes and regulatory fees on the services MetroPCS and T-Mobile provide and increases or changes in taxes and regulatory fees or the services to, or the manner in, which such taxes and fees are applied, calculated, or collected;

the rapid technological changes in our industry and MetroPCS and T-Mobile s ability to adapt, respond and deploy new technologies and successfully offer new services using such new technology;

MetroPCS and T-Mobile s ability to fulfill the demands and expectations of their customers, provide the customer care such customers want, expect, or demand and secure the products, services, applications, content and network infrastructure equipment each of MetroPCS and T-Mobile needs, or which their customers or their potential customers want, expect or demand;

the availability of additional spectrum and MetroPCS and T-Mobile s ability to secure additional spectrum, or secure it at acceptable prices, when they need it;

MetroPCS and T-Mobile s ability to adequately defend against suits filed by others and to enforce or protect their intellectual property rights;

MetroPCS and T-Mobile s, as well as the combined company s, capital structure, including the indebtedness amounts of each, the limitations imposed by the covenants in the documents governing the indebtedness of each and the maintenance of the financial and disclosure controls and procedures of each;

MetroPCS and T-Mobile s ability to attract and retain key members of management and train personnel;

#### Edgar Filing: MILLER PETROLEUM INC - Form 10-K

MetroPCS and T-Mobile s reliance on third parties to provide distribution, products, software content and services that are integral, used in or sold by their respective businesses and the ability of their respective suppliers to perform, develop and timely provide each of them with technological developments, products and services that each of them needs to remain competitive; and

governmental regulation affecting MetroPCS and T-Mobile s services and changes in government regulation, and the costs of compliance and the failure to comply with such regulations.

MetroPCS cautions that the foregoing list of factors is not exclusive. Additional information concerning these and other risk factors is contained in MetroPCS Annual Report on Form 10-K for the year ended December 31, 2011 and Quarterly Report on Form 10-Q for the fiscal quarter ended September 30, 2012. All

<sup>31</sup> 

subsequent written and oral forward-looking statements concerning MetroPCS, Deutsche Telekom, T-Mobile, the transaction or other matters attributable to MetroPCS, Deutsche Telekom, or T-Mobile or any person acting on their behalf are expressly qualified in their entirety by the cautionary statements above. Forward-looking statements herein or in documents incorporated herein by reference speak only as of the date of this proxy statement or the applicable document incorporated herein by reference (or such earlier date as may be specified therein), as applicable, are based on current assumptions and expectations or assumptions and expectations as of the date of the document incorporated herein by reference, and are subject to the factors above, among other things, and involve risks, uncertainties, events, circumstances, uncertainties and assumptions, many of which are beyond our ability to control or predict. You should not place undue reliance on these forward-looking statements. MetroPCS does not intend to, and does not undertake an obligation to, update these forward-looking statements in the future to reflect future events or circumstances, except as required by applicable securities laws and regulations. For more information, see the section entitled Where You Can Find More Information beginning on page 210. The results presented for any period may not be reflective of results for any subsequent period.

You should carefully read and consider the cautionary statements contained or referred to in this section in connection with any subsequent written or oral forward-looking statements that may be issued by us or persons acting on our behalf, and all future written and oral forward-looking statements attributable to MetroPCS, Deutsche Telekom, T-Mobile, the combined company, the transaction or any other matters, are expressly qualified in their entirety by the foregoing cautionary statements.

#### **RISK FACTORS**

In addition to the other information included in and incorporated by reference into this proxy statement, including the matters addressed in the section titled Cautionary Note Regarding Forward-Looking Statements beginning on page 29, you should carefully consider the following risks before deciding whether to vote for the proposals. In addition, you should read and consider the risks associated with each of the businesses of MetroPCS and T-Mobile because these risks also will remain for the combined company; these risks with respect to MetroPCS can be found in MetroPCS most recent Annual Report on Form 10-K, as updated by subsequent Quarterly Reports on Form 10-Q, all of which are filed with the SEC and incorporated by reference into this proxy statement. Many of these risks, and the events that cause these risks, are beyond our ability to control or predict. If any of the these risks were to occur, the respective businesses, financial results, financial condition, operating results or stock prices, as applicable, of MetroPCS, T-Mobile or the combined company could be materially adversely affected. Although we incorporate by reference and describe below and elsewhere in this proxy statement the risks we consider to be the most material to the transaction and our business, assets, liabilities, prospects, outlook, financial condition and results of operations, there may be other known, unknown or unpredictable economic, business, competitive, regulatory or other risks or factors that also could have a material adverse effect on the transaction or our business, assets, liabilities, prospects, outlook, financial condition and results of operations in the future. In addition, past performance may not be a reliable indicator of future performance and historical trends should not be used to anticipate results or trends in future periods. Any information about our intentions, projections or forward-looking statements in this proxy statement or in documents incorporated by reference into this proxy statement is a statement of our intentions, projections and forward-looking statements as of the date of this proxy statement or the documents incorporated by reference into this proxy statement and is based upon, among other things, the regulatory, industry, competitive, economic and market conditions as of such date, as well as various of our assumptions at such time. We may change our intentions, projections or forward-looking statements at any time and without notice, based upon any changes in such conditions, in our assumptions or otherwise. You should also read and consider the other information in this proxy statement and the other documents incorporated by reference into this proxy statement. See the section entitled Where You Can Find More Information beginning on page 210.

#### **Risk Factors Relating to the Transaction**

# The amount of MetroPCS common stock and the cash payment to be issued or paid in the transaction is fixed and will not be adjusted for changes in the business, assets, liabilities, prospects, outlook, financial condition or results of operations of MetroPCS or T-Mobile or in the event of any change in MetroPCS stock price.

Immediately following the transaction, Holding or its designee will own 74% of the fully-diluted shares of MetroPCS common stock outstanding and MetroPCS equityholders immediately prior to the transaction collectively will own the remaining 26% of fully-diluted MetroPCS common stock and will receive their pro rata share of the cash payment. These percentages and the cash payment are fixed in the business combination agreement and will not be adjusted for changes in the business, assets, liabilities, prospects, outlook, financial condition or results of operations of MetroPCS or T-Mobile, or changes in the market price of, analyst estimates of, or projections relating, to MetroPCS common stock. For example, if T-Mobile were to experience a decline in its business, assets, liabilities, prospects, outlook, financial condition or results of operations prior to the completion of the transaction, while MetroPCS experienced an improvement in its business, assets, liabilities, prospects, outlook, financial condition and results of operations prior to the completion of the transaction, there would be no adjustment in the overall ownership percentages in the combined company or the amount of the cash payment as described above.

# The transaction is conditioned on the receipt of approvals from various governmental entities, which may not approve the transaction, may delay the approvals, or may impose conditions or restrictions on, jeopardize or delay completion of, or reduce the anticipated benefits of, the transaction.

Completion of the transaction is conditioned upon filings with, and, in certain cases, the receipt of governmental authorizations, consents, orders or other approvals from, governmental entities, including the FTC,

the Antitrust Division, the FCC, the CFIUS and, if applicable, state public utility or service commissions and foreign authorities. The parties have made or plan to make initial filings with each of these governmental entities where required; however, the applicable waiting period under U.S. antitrust laws has not yet expired or been terminated and the other governmental entities have not concluded their review and/or yet provided the requisite authorizations, consents, orders or other approvals.

There is no assurance that all of these required authorizations, consents, orders and other approvals will be obtained, and, if they are obtained, they may not be obtained before you vote on the proposals relating to the transaction. Moreover, if they are obtained, they may require actions or impose restrictions, limitations or conditions on the assets, businesses, licenses or product lines of MetroPCS or T-Mobile. The business combination agreement requires the parties to satisfy any actions, or to agree to any restrictions, limitations or conditions, in each case with respect to any of the assets, businesses, licenses or product lines of MetroPCS, Deutsche Telekom, T-Mobile, any of their respective subsidiaries, or any combination thereof, unless it would have a material adverse effect on the business, assets, liabilities, prospects, outlook, financial condition or results of operations of MetroPCS, T-Mobile and their respective subsidiaries, taken as a whole. It is possible that such actions, restrictions, limitations or conditions may have an adverse effect on the business, assets, liabilities, prospects, outlook, financial condition or results of operations of MetroPCS or T-Mobile, but not qualify as a material adverse effect under the business combination agreement. These required actions, restrictions, limitations and conditions also may jeopardize or delay completion of the transaction, reduce the anticipated benefits of the transaction or allow the parties to terminate the transaction.

## Failure to complete the transaction, or a delay in completing the transaction, could negatively impact the stock price of MetroPCS and the future business, assets, liabilities, prospects, outlook, financial condition and results of operations of MetroPCS, T-Mobile and/or the combined company.

If the transaction is not completed, MetroPCS ongoing business may be adversely affected and the market price of our common stock may decline, particularly to the extent that the current market price reflects a market assumption that the transaction will be completed. If closing is delayed, including by a delay in receipt of necessary governmental approvals or by the receipt of a competing proposal, the ongoing businesses, financial condition and results of operations of MetroPCS and T-Mobile may be adversely affected. Additionally, if the transaction is not completed, MetroPCS, under certain circumstances described in Summary of the Business Combination Agreement Termination and Effect of Termination, beginning on pages 151 and 152, respectively, may be required to pay Deutsche Telekom \$150 million. Any of the foregoing, or other risks arising in connection with the failure of or delay in completing the transaction, including the diversion of management attention from pursuing other opportunities and operating the ongoing business during the pendency of the transaction, may have an adverse effect on the business, assets, liabilities, prospects, outlook, financial condition or results of operations of MetroPCS and/or T-Mobile.

## The business combination agreement contains provisions that could affect whether a potential competing acquirer of MetroPCS makes a competing proposal or that could delay the completion of the transaction.

The business combination agreement contains no shop provisions that, subject to limited exceptions, restrict MetroPCS ability to solicit, encourage, facilitate or discuss competing third-party proposals to acquire stock or assets of MetroPCS. Further, while the MetroPCS board is permitted to make a recommendation change to the stockholders with respect to the transaction under certain circumstances, unless Deutsche Telekom terminates the business combination agreement, MetroPCS nonetheless will be required to submit the proposals to a stockholder vote at the special meeting. This requirement, which is often called a force the vote provision, means that MetroPCS does not have the right before the stockholder vote to terminate the business combination agreement to accept a superior proposal. In addition, Deutsche Telekom generally has an opportunity to offer to modify the terms of its proposal in response to a competing superior acquisition proposal, before the MetroPCS board makes a recommendation change with respect to the transaction. This requirement is commonly called a match right. In some circumstances, upon

termination of the business combination agreement MetroPCS will be required to pay a termination fee of \$150 million to Deutsche Telekom. See the sections entitled Summary of the Business Combination Agreement No Solicitation of Alternative Proposals, Change in the MetroPCS Board s Recommendation, Termination and Effect of Termination beginning on pages 142, 144, 151 and 152, respectively.

These provisions may affect whether a potential competing acquirer will make a proposal to acquire MetroPCS, even if it were prepared to pay consideration with a higher per share value than the market value proposed to be received or realized in the transaction. These provisions also might result in a potential competing acquirer proposing to pay a lower price than it might otherwise have proposed to pay because of the added expense of the \$150 million termination fee that may become payable in certain circumstances or the possible delay in executing such transaction as a result of having to wait until after the MetroPCS stockholder vote on the proposal contained in this proxy statement at the special meeting. Further, because T-Mobile is a private company, a third party may have difficulty establishing that a competing proposal constitutes a higher value proposal than the current transaction and could have an adverse effect on MetroPCS business, assets, liabilities, prospects, outlook, financial condition or results of operations. Any delay in completing the transaction also could reduce the benefits, cost and capital expenditure savings and synergies from combining the businesses of MetroPCS and T-Mobile.

If the business combination agreement is terminated and MetroPCS determines to seek another business combination, it may not be able to negotiate a transaction with another party on terms comparable to or better than the terms of the transaction.

## Some of MetroPCS and T-Mobile s existing agreements contain change in control or early termination rights that may be implicated by the transaction, and exclusivity, commitment or minimum purchase provisions in some of MetroPCS existing agreement may limit the combined company s business.

Some of MetroPCS and T-Mobile s existing contracts contain provisions that allow the counterparty to terminate the agreement after a specified period following a change of control of MetroPCS and/or T-Mobile. If a third party has such a right and were to exercise its right to terminate an agreement as a result of the completion of the transaction, such termination could disrupt MetroPCS and/or T-Mobile s existing operations and adversely affect MetroPCS business, assets, liabilities, prospects, outlook, financial conditions and results of operations. For example, renewal of MetroPCS roaming agreements allow the counterparty to terminate or limit the benefits of the roaming agreements in connection with a change in control. If such counterparty were to do so, it could reduce the area in which MetroPCS customers can receive service outside of MetroPCS network coverage area. In addition, some of MetroPCS existing contracts contain exclusivity, commitment or minimum purchase provisions that will continue to apply to the combined company after the completion of the transaction. These exclusivity, commitment or minimum purchase provisions could limit the combined company s ability take advantage of certain opportunities in specified geographic areas which could reduce some of the anticipated benefits from the transaction or its ability to achieve the planned synergies. The combined company may be required to enter into agreements for similar services with other third parties or renegotiate its existing agreements to avoid such a disruption or restrictions.

# MetroPCS and T-Mobile are subject to various uncertainties and contractual restrictions while the transaction is pending that could disrupt their potential businesses and could adversely affect their businesses, assets, liabilities, prospects, outlooks, financial conditions and results of operations.

Uncertainty about the effect of the transaction on employees, customers, suppliers, vendors, distributors, dealers and retailers may have an adverse effect on MetroPCS and/or T-Mobile. These uncertainties may impair MetroPCS and/or T-Mobile s ability to attract, retain and motivate key personnel, dealers and retailers until the transaction is completed and for a period of time thereafter, as employees and prospective employees may experience uncertainty about their future roles with the combined company. Additionally, these uncertainties

could cause customers, suppliers, distributors, dealers, retailers and others who deal with MetroPCS or T-Mobile to seek to change existing business relationships with MetroPCS or T-Mobile or fail to extend an existing relationship with MetroPCS or T-Mobile. Suppliers, distributors and content and application providers may also delay or cease developing new products for MetroPCS or T-Mobile that are necessary for the operations of their respective businesses due to the uncertainty created by the transaction. In particular, because the transaction contemplates a discontinuance of MetroPCS existing network technology over time, current suppliers, including handset suppliers, may choose not to invest in, may delay, or may cease developing or supporting, new handsets for MetroPCS that are necessary for MetroPCS to compete effectively or to meet the demands, requirements or expectations of its customers. In addition, existing competitors may target MetroPCS or T-Mobile s existing customers by highlighting potential uncertainties and integration difficulties that may result from the transaction.

MetroPCS has a small number of key personnel. The pursuit of the transaction and the preparation for the integration may place a burden on management and internal resources. Any significant diversion of management attention away from ongoing business concerns and any difficulties encountered in the transition and integration process could affect MetroPCS and/or T-Mobile s financial results.

In addition, the business combination agreement restricts each of MetroPCS and T-Mobile, without the other s consent, from taking certain actions outside of the ordinary course of business while the transaction is pending. These restrictions may prevent MetroPCS and/or T-Mobile from pursuing otherwise attractive business opportunities and making other changes to their respective businesses prior to completion of the transaction or termination of the business combination agreement. See the section entitled Summary of the Business Combination Agreement Conduct of Business beginning on page 140.

## Multiple lawsuits have been filed against the parties challenging the transaction, and an adverse ruling may delay or prevent the transaction from being completed.

Since the announcement on October 3, 2012 of the execution of the business combination agreement, MetroPCS, Deutsche Telekom, Global, Holding, T-Mobile and the members of the MetroPCS board including an officer, have been named as defendants in multiple stockholder derivative and class action complaints challenging the transaction. The lawsuits generally allege, among other things, that the transaction fails to properly value MetroPCS and that the individual defendants breached their fiduciary duties in approving the business combination agreement and, in some of the lawsuits, that those breaches were aided and abetted by Deutsche Telekom, Global, Holding and T-Mobile. The lawsuits seek, among other things, injunctive relief enjoining the defendants from completing the transaction on the agreed-upon terms, monetary relief, punitive damages, and attorneys fees and costs.

One of the conditions to the closing of the transaction is that no governmental entity has enacted, issued, promulgated, enforced or entered any law, statute, ordinance, rule, regulation, judgment, injunction, decree or other order (whether temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits completion of the transaction. Consequently, if the plaintiffs secure injunctive or other relief prohibiting, delaying, or otherwise adversely affecting the defendants ability to complete the transaction, then such injunctive or other relief may prevent the transaction from becoming effective within the expected time frame or at all. If completion of the transaction is prevented or delayed, it could result in substantial costs to the parties. In addition, the parties could incur significant costs in connection with the lawsuits, including costs associated with the defense and indemnification of MetroPCS directors and officers. Finally, any delay could allow the other parties to the transaction to terminate the business combination agreement if the closing is delayed beyond the outside date provided therein.

## Directors and executive officers of MetroPCS have interests in the transaction that may be different from, or in addition to, those of other stockholders of MetroPCS, which could have influenced their decisions to support or approve the transaction.

In considering whether to approve the proposals at the special meeting, you should recognize that all of the directors and executive officers of MetroPCS have interests in the transaction that may differ from, or that are in

addition to, your interests as a stockholder of MetroPCS. These interests include, among others, continued service as a director or an executive officer of the combined company, employment or consulting arrangements, acceleration of awards under MetroPCS equity incentive compensation plans, elections to cash out stock options in certain circumstances, arrangements that provide for severance benefits if certain executive officers employment is terminated under certain circumstances following the completion of the transaction and rights to indemnification and directors and officers liability insurance that will survive the completion of the transaction. The MetroPCS board and the special committee were aware of these interests during the time that the business combination was being negotiated and at the time they approved the transaction. These interests may cause MetroPCS directors and executive officers to view the transaction differently than you may view it as a stockholder. See the section entitled Interests of MetroPCS Directors and Officers in the Transaction beginning on page 118.

## There are risks associated with the reverse stock split, including that the reverse stock split may not result in a proportionate increase in the per share price of the MetroPCS common stock.

If MetroPCS completes the transaction, MetroPCS will effect a 1-for-2 reverse stock split as of the effective time and make the cash payment. MetroPCS cannot predict whether or to what extent the reverse stock split will proportionately increase the market price of MetroPCS common stock in comparison to the reduction in the number of shares of MetroPCS common stock issued and outstanding before the reverse stock split or what effect the cash payment will have on the price of MetroPCS common stock. The market price of MetroPCS common stock may decrease following the transaction for various reasons, including due to the reverse stock split or to account for other factors, including the combined company s performance, businesses, assets, liabilities, prospects, outlooks, financial conditions, results of operations, analyst reports and projections, and broader market conditions that are unrelated to the number of shares of MetroPCS common stock outstanding or the cash payment. As a result, there can be no assurance that the reverse stock split will proportionately increase the market price of MetroPCS stock, that the market price of MetroPCS common stock will reflect the cash payment, or that the market price of MetroPCS common stock will remain at a higher value for any significant length of time after completion of the transaction.

## A significant stockholder of MetroPCS executed a voting and support agreement in connection with the transaction, and that voting and support agreement is binding on such stockholder, even if the MetroPCS board changes its recommendation to the MetroPCS stockholders.

Concurrent with the execution of the business combination agreement, Madison Dearborn Capital Partners IV, L.P., which we refer to as Madison Dearborn, an approximate 8.32% holder of MetroPCS common stock, entered into a voting and support agreement with Deutsche Telekom, which we refer to as the support agreement, pursuant to which Madison Dearborn agreed to vote all shares of its MetroPCS common stock (i) in favor of the required MetroPCS stockholder approvals and (ii) against certain third party proposals to acquire MetroPCS and against any other actions that could reasonably be expected to materially impede, interfere with, delay, postpone, discourage or adversely affect the transaction or any other transactions contemplated by the business combination agreement. The support agreement is binding on Madison Dearborn, even if the MetroPCS board changes its recommendation to the MetroPCS stockholders with respect to the required MetroPCS stockholder approvals; provided that, the support agreement will terminate in the event Deutsche Telekom elects to terminate the business combination agreement due to a change in the recommendation of the MetroPCS board. See the section entitled Summary of Ancillary Agreements Voting and Support Agreement beginning on page 158.

## If the reverse stock split and cash payment do not qualify as a recapitalization within the meaning of Section 368(a)(1)(E) of the Code, the MetroPCS stockholders may be required to pay additional U.S. federal income taxes.

For U.S. federal income tax purposes, MetroPCS believes the reverse stock split and cash payment should be integrated and treated as a recapitalization within the meaning of Section 368(a)(1)(E) of the Code. There can

be no assurance, however, that the Internal Revenue Service, which we refer to as the IRS, or the courts will agree with such treatment. If the IRS or a court were to determine that the reverse stock split and cash payment should not be treated as a recapitalization within the meaning of Section 368(a)(1)(E) of the Code, the MetroPCS stockholders might be required to treat the cash payment as ordinary income. See Material U.S. Federal Income Tax Consequences beginning on page 130.

In connection with the transaction, MetroPCS, T-Mobile and/or the combined company may be required to take write-downs or write-offs, restructuring and impairment or other charges that could negatively impact the business, assets, liabilities, prospects, outlook, financial condition and results of operations of MetroPCS, T-Mobile and/or the combined company.

Although MetroPCS and T-Mobile have conducted extensive due diligence in connection with the transaction, they cannot assure you that this diligence revealed all material issues that may be present, that it would be possible to uncover all material issues through a customary amount of due diligence, or that factors outside of MetroPCS and T-Mobile s control will not later arise. Even if MetroPCS and T-Mobile s due diligence successfully identifies certain risks, unexpected risks may arise and previously known risks may materialize in a manner not consistent with MetroPCS and T-Mobile s preliminary risk analysis. Further, as a result of the transaction, purchase accounting, and the proposed operation of the combined company going forward, MetroPCS, T-Mobile and/or the combined company may be required to make write-offs or write-downs, restructuring and impairment or other charges. As a result, MetroPCS, T-Mobile and/or the combined company may be forced to write-down or write-off assets, restructure its operations, or incur impairment or other charges that could negatively impact the business, assets, liabilities, prospects, outlook, financial condition and results of operations of MetroPCS, T-Mobile and/or the combined company.

#### **Risk Factors Relating to the Combined Company Following the Transaction**

## Even following the completion of the transaction, the combined company will continue to face intense competition from other competitors, some of which have greater resources than the combined company, and such competition may intensify in the future.

The U.S. wireless telecommunications industry is highly competitive with a range of diversified competitors. After the completion of the transaction, the combined company will have increased spectrum assets, network coverage and capacity, and improved marketing and purchasing scale. However, following the transaction, the combined company will remain the fourth largest wireless carrier in the United States, and the combined company s relative competitive position will present risks associated with scale, brand positioning, and perception of network coverage and quality. Several of the combined company s competitors are larger companies that have better-developed brands and brand awareness; greater spectrum and capital resources; better networks and support systems; substantially greater financial, technical, personnel and marketing resources and bargaining power; greater access to capital on better terms, cash reserves, and spectrum holdings; larger geographic coverage areas, roaming territories, and third-party distribution networks; better in-building coverage; unique intellectual property; greater numbers of customers; exclusive distributorship arrangements for certain popular brands of handsets, applications and content; and more advanced technology. The combined company will face a long-term challenge to compete effectively in terms of pricing, products, coverage, and new technologies and services. Some competitors operate using alternative business models that have the potential to affect negatively the combined company s ability to attract and retain customers. All of the foregoing may affect the combined company s ability to compete successfully.

## The failure to integrate successfully the businesses of MetroPCS and T-Mobile in the expected time frame could adversely affect the combined company s future results following the transaction.

The success of the transaction will depend, in large part, on the ability of the combined company to realize the anticipated benefits, including projected synergies and cost savings, from combining the businesses of MetroPCS and T-Mobile. To realize these anticipated benefits, the businesses of MetroPCS and T-Mobile must

be successfully integrated. This integration will be complex and time-consuming and may divert management s time and attention from the business. The failure to integrate successfully and to manage successfully the challenges presented by the integration process may result in the combined company not achieving the anticipated benefits of the transaction.

The parties may encounter numerous potential difficulties in the integration process, including the following:

being unable to integrate successfully the businesses of MetroPCS and T-Mobile in a manner that permits the combined company to achieve the cost savings anticipated to result from the transaction;

migrating MetroPCS customers to the combined company s T-Mobile based global system for mobile communications, which we refer to as GSM, evolved high speed packet access, which we refer to as HSPA+, and LTE networks;

integrating MetroPCS and T-Mobile s existing information and billing systems, cell sites, customer service programs and distributed antenna systems;

decommissioning MetroPCS networks;

integrating and adding T-Mobile s technology to MetroPCS cell sites and distributed antenna systems;

combining MetroPCS and T-Mobile s product and service offerings, subscriber plans, customer service, and sales and marketing approaches;

preserving subscriber, supplier, vendor, content provider, dealer, retailer, and other important relationships;

resolving complexities associated with managing the larger combined company;

addressing the potential effect of the transaction on MetroPCS and T-Mobile s business and relationships with employees, customers, suppliers, vendors, content providers, distributors, dealers, retailers, regulators and the communities in which they operate;

addressing the potential difficulty in coordinating geographically dispersed organizations and business headquarters;

addressing possible differences in corporate cultures and management philosophies;

integrating personnel from the two companies while maintaining focus on providing consistent, high quality products and services;

retaining key employees and members of management of MetroPCS and T-Mobile;

#### Edgar Filing: MILLER PETROLEUM INC - Form 10-K

encountering difficulties in consolidating and preparing the combined company s financial statements, or having to restate the financial statements of the combined company;

addressing the potential difficulty in maintaining cost controls during the integration process;

discharging and otherwise addressing potential unknown liabilities and unforeseen expenses, delays or regulatory conditions associated with the transaction; and

experiencing performance shortfalls at one or both of MetroPCS and T-Mobile as a result of the diversion of management s attention caused by completing the transaction and integrating their operations.

## Many of the anticipated synergies from the transaction may not be realized for a significant period of time after the completion of the transaction, if at all, and will require substantial capital expenditures to be fully realized.

The success of the transaction will depend, in part, on the ability of the combined company to realize the anticipated synergies as a result of the transaction. Many of the anticipated synergies are not expected to occur for a significant time period following the completion of the transaction and will require substantial capital expenditures in the near term to be fully realized. Many of the anticipated synergies will only occur, if at all, after these substantial capital expenditures or expenses have been incurred. Even if the combined company is able to integrate the two companies successfully, this integration may not result in the realization of the full benefits of the synergies that are currently expected or the achievement of these benefits within the anticipated time frame or at all.

## The combined company s future results could suffer if it does not effectively manage its expanded business, operations and employee base following the transaction.

The size of the combined company s business, operations and employee base following the transaction will be greater than the stand-alone size of either MetroPCS or T-Mobile s business, operations and employee base prior to the transaction and will include two separate business units headquartered in different cities. The combined company s future success depends, in part, upon its ability to manage this expanded business, operations and employee base, which will pose substantial challenges for management, including challenges related to the management and monitoring of new operations and associated increased costs and complexity. No assurances can be given that the combined company will successfully manage its expanded business, operations and employee base following the transaction.

## The combined company s results of operations after the transaction may be affected by factors different than those currently affecting MetroPCS and T-Mobile s separate results of operations.

T-Mobile s business differs from MetroPCS business. For example, MetroPCS provides wireless services based on a no-annual contract model, while T-Mobile provides wireless services largely based on an annual contract model, in addition to no-contract plans. The combined company s results of operations, as well as the price of the combined company s common stock after the transaction, may be affected by factors different than those currently affecting MetroPCS or T-Mobile s results of operations and MetroPCS stock price. The price of the combined company s common stock may fluctuate significantly following the completion of the transaction, including as a result of factors over which MetroPCS and T-Mobile have no control. In addition, T-Mobile is a privately-held corporation and, therefore, there is no public valuation of T-Mobile s business. Due in part to this lack of existing public valuation, we cannot predict the price at which the combined company s common stock may trade after the completion of the transaction.

## The unaudited pro forma financial statements included in this proxy statement are presented for illustrative purposes only and may not be an indication of the combined company s financial condition or results of operations following the transaction.

The unaudited pro forma financial statements contained in this proxy statement are presented for illustrative purposes only, are based on various adjustments, assumptions and preliminary estimates and may not be an indication of the combined company s financial condition or results of operations following the transaction. The unaudited pro forma financial statements have been derived from MetroPCS and T-Mobile s historical financial statements and certain adjustments and assumptions have been made regarding the combined company. The assumptions used in preparing the unaudited pro forma financial information may not prove to be accurate, and other factors may affect the combined company s financial condition or results of operations following the transaction. Moreover, the unaudited pro forma financial statements do not reflect all costs that are expected to be incurred in connection with the transaction. As a result, the combined company s actual financial condition and results of operations may differ significantly from the unaudited pro forma financial statements.

#### The combined company is expected to incur substantial expenses related to the transaction and the integration of MetroPCS and T-Mobile.

The combined company is expected to incur substantial expenses in connection with the transaction and the integration of MetroPCS and T-Mobile. There are a large number of processes, policies, procedures, operations, staff, functions, technologies and systems that must be integrated, including purchasing, accounting and finance, legal, regulatory, sales, inventory forecasting, billing, payroll, marketing, customer service and benefits. While MetroPCS and T-Mobile have assumed that a certain level of these expenses would be incurred and that such integration could be accomplished within a certain period of time, there are many factors beyond their control that could affect the total amount or the timing of the expenses. Moreover, many of these expenses are, by their nature, difficult to estimate accurately. These expenses could, particularly in the near term, exceed the savings that the parties expect to achieve from the elimination of duplicative expenses and the realization of economies of scale. These integration expenses likely will result in the combined company taking significant charges against earnings following the completion of the transaction. The amount and timing of such charges are uncertain at present.

# Following the completion of the transaction, the combined company will be controlled by Deutsche Telekom, and Deutsche Telekom will continue to have approval rights over certain actions taken by the combined company as long as it beneficially owns 30% or more of the combined company s common stock. The interests of Deutsche Telekom may differ from the interests of other stockholders of the combined company.

Immediately following the completion of the transaction, Deutsche Telekom will beneficially own and possess voting power over approximately 74% of the fully diluted shares of the combined company s common stock. Under the stockholder s agreement to be entered into as provided in the business combination agreement, which we refer to as the stockholder s agreement, Deutsche Telekom may acquire additional shares of the combined company s common stock up to an aggregate of 80.1% of the combined company s common stock without the combined company board s approval.

Through its control of the combined company s voting power and the provisions set forth in the new certificate of incorporation and the stockholder s agreement, Deutsche Telekom will have the ability to elect a majority of the combined company s directors, hire or fire the Chief Executive Officer, and control all other matters requiring the approval of the combined company s stockholders. As a result, the combined company will be a controlled company as defined in the NYSE listing rules and will, therefore, not be subject to NYSE requirements that would otherwise require the combined company to have (i) a majority of independent directors, (ii) a nominating committee composed solely of independent directors, (iii) compensation of its executive officers determined by a majority of the independent directors or a compensation committee composed solely of independent directors, and (iv) director nominees selected, or recommended for the board s selection, either by a majority of the independent directors.

In addition, the new certificate of incorporation and the stockholder s agreement will restrict the combined company s ability to take certain actions without Deutsche Telekom s prior written consent as long as Deutsche Telekom beneficially owns 30% or more of the outstanding shares of the combined company s common stock, including the ability to enter into acquisitions of any business, debt or equity interests, operations or assets of any person for consideration in excess of \$1 billion, sale of any division, business, operations or equity interests of the combined company or any of its subsidiaries for consideration in excess of \$1 billion or hire or fire the Chief Executive Officer. These restrictions could prevent the combined company from taking actions that the combined company s board determines are in the best interests of the combined company and its stockholders or that the other non-Deutsche Telekom stockholders determine are in their best interest. See the section entitled Summary of Ancillary Agreements Stockholder s Agreement beginning on page 154.

Deutsche Telekom will also have control over all matters submitted to our stockholders for approval, including the election or removal of directors, changes in our capital structure, transactions requiring stockholder approval under Delaware law and corporate governance. As a holder of a substantial amount of equity and debt

of the combined company, Deutsche Telekom may have different interests than other holders of MetroPCS common stock and may make decisions adverse to your interests. For example, a substantial amount of the combined company s cash flow will be used to pay the interest and make principle repayments on the indebtedness held by Deutsche Telekom. In addition, Deutsche Telekom s interests may shift if it transfers a substantial amount of its equity holdings in the combined company while retaining its large credit position. For example, if the combined company encounters financial difficulties, the interests of Deutsche Telekom as a creditor of the combined company might conflict with stockholders interests. This concentrated control and these approval rights could delay, defer, or prevent a change in control, merger, consolidation, or sale of all or substantially all of the combined companies assets that the combined company s other stockholders do not support. In addition, this concentrated control and these approval rights could discourage a potential investor from seeking to acquire the combined company s common stock and, as a result, might harm the market price of the combined company s common stock.

# Following the completion of the transaction, Deutsche Telekom will be permitted to transfer shares of the combined company s common stock in any transaction that would result in the transferee owning 30% or less of the outstanding shares of the combined company s common stock, and any control or other purchase price premium Deutsche Telekom receives from such transfer need not be shared with the rest of the combined company s stockholders.

Pursuant to the stockholder s agreement, Deutsche Telekom will be prohibited from transferring any shares of the combined company s common stock in any transaction that would result in the transferee owning more than 30% of the outstanding shares of the combined company s common stock unless such transferee offers to acquire all of the then outstanding shares of the combined company s common stock at the same price and on the same terms and conditions as the proposed transfer from Deutsche Telekom. However, Deutsche Telekom will be permitted to transfer shares of the combined company s common stock, without requiring the transferee to acquire all such other outstanding shares, in any transaction that would result in the transferee owning 30% or less of the outstanding shares of the combined company s common stock. In the event Deutsche Telekom receives a control or other purchase price premium from such transfer of its shares only, it will not be obligated to share such premium with the rest of the combined company s stockholders. See the section entitled Summary of Ancillary Agreements Stockholder s Agreement beginning on page 154.

# Following the completion of the transaction, Deutsche Telekom will be subject to a six month lock-up period with respect to its shares of the combined company s common stock, after which, subject to limited restrictions, it will be permitted to transfer freely its shares of the combined company s common stock, which could have a negative impact on the combined company s stock price.

Following the completion of the transaction, Deutsche Telekom will be prohibited from transferring any shares of the combined company s common stock for six months. However, following such six-month period, Deutsche Telekom will be permitted, subject to limited restrictions described in the section entitled Summary of Ancillary Agreements Stockholder s Agreement beginning on page 154, to transfer freely its shares of the combined company s common stock without notice. Any such transfer could significantly increase the number of the combined company s shares available in the market, which could cause a decrease in the combined company s stock price. In addition, even if Deutsche Telekom does not transfer a large number of its shares into the market, its right to transfer a large number of shares into the market may depress the combined company s stock price.

## The combined company does not have a contractual right to make indemnification claims against Deutsche Telekom for the breach of any representations, warranties or covenants made by Deutsche Telekom or its subsidiaries in the business combination agreement.

Under the business combination agreement, the combined company does not have a right to make contractual indemnification claims against Deutsche Telekom after the closing, including for a breach by

Deutsche Telekom or its subsidiaries (including T-Mobile or its subsidiaries) of the representations and warranties made to MetroPCS or for a violation by Deutsche Telekom or its subsidiaries (including T-Mobile or its subsidiaries) of certain covenants and agreements in the business combination agreement.

This limitation does not affect any other entitlement, remedy or recourse permitted by law that the combined company may have against Deutsche Telekom or its subsidiaries, including the right to specific performance set forth in the business combination agreement. Such action may be brought by the combined company at the direction of a majority of the directors of the combined company not affiliated with Deutsche Telekom. However, because Deutsche Telekom will appoint a majority of the directors, including some directors not affiliated with Deutsche Telekom, such Deutsche Telekom appointees that are not affiliated with Deutsche Telekom can vote against such suit. Because such directors may constitute a majority of the unaffiliated directors, the combined company may not bring suit even though it may be in the best interests of the stockholders other than Deutsche Telekom. See the section entitled Summary of the Business Combination Agreement Specific Performance beginning on page 153.

#### The combined company will incur substantial indebtedness in connection with the transaction.

The parties intend to finance the transaction, and refinance MetroPCS and T-Mobile s existing indebtedness, with up to \$18.5 billion in senior unsecured debt financing through the issuance of debt securities to Deutsche Telekom and third-party investors. As a result, the combined company is expected to have long-term indebtedness that will be substantially greater than either MetroPCS or T-Mobile s long-term indebtedness prior to the transaction and related refinancing. This new indebtedness will increase the related risks MetroPCS now faces with its current indebtedness, described in detail in MetroPCS most recent Annual Report on Form 10-K, as updated by subsequent Quarterly Reports on Form 10-Q.

## A substantial majority of the indebtedness that would be incurred in connection with the transaction will be subject to significant limitations on redemption, which may impede the refinancing of such indebtedness.

The \$15.0 billion notes to be purchased by Deutsche Telekom, as well as the \$2.5 billion notes and \$1.0 billion notes (if purchased by Deutsche Telekom or issued by Wireless on or prior to the closing date of the transactions), may only be redeemed, prior to certain specified dates for each series, if the combined company or T-Mobile pays a customary make whole premium at the time of redemption. Such make whole premium may be significant, and may make it financially prohibitive for the combined company to refinance such series prior to such dates to take advantage of lower interest rates which may become available for the combined company s debt or in connection with pursuing other business opportunities.

## A substantial portion of the indebtedness that would be incurred in connection with the transaction is subject to a pricing reset that may materially increase the interest rate applicable to that indebtedness.

\$7.5 billion of the \$15.0 billion notes to be purchased by Deutsche Telekom, and 50% of any of the \$2.5 billion notes and \$1.0 billion notes purchased by Deutsche Telekom, will be subject to a pricing reset that will cause the interest rate of the relevant debt securities to be recalculated according to a formula which depends in part upon designated indices and other benchmark debt securities, only a portion of which is calculated based on the trading prices of MetroPCS indebtedness. This reset will occur, for each series of the debt securities, either two, two and a half or three years after the issue date of the debt securities. As a result of the pricing reset feature, as of the reset date for each applicable series of debt securities, a material increase in the combined company s interest expense could result with respect to such series as of the reset date, which would increase its interest and financing costs, decrease its net income and further increase the related risks MetroPCS now faces with its current indebtedness, described in detail in MetroPCS most recent Annual Report on Form 10-K, as updated by subsequent Quarterly Reports on Form 10-Q.

Interest rates for such benchmark indices and debt securities are highly sensitive to many factors, including domestic and international economic and political conditions, policies of governmental and regulatory agencies,

developments affecting the results or prospects of the issuer of the benchmark securities or of securities referenced in the benchmark indices, and other factors beyond the combined company s control. As a result, a significant increase in these interest rates at the time that the relevant debt securities are recalculated could have an adverse effect on the combined company s financial position and results of operations.

#### The combined company s working capital revolving credit facility subjects the combined company to interest rate changes.

The entire amount of indebtedness outstanding from time to time under the combined company s \$500 million working capital revolving credit facility will bear interest at floating rates. While the combined company will be permitted to hedge against higher interest rates, there is no assurance it will be able to do so or will do so on commercially reasonable terms. Interest rates are highly sensitive to many factors, including domestic and international economic and political conditions, policies of governmental and regulatory agencies and other factors beyond the combined company s control. An increase in benchmark interest rates could cause a material increase in the combined company s interest expense, which would increase its financing costs, decrease its net income and further increase the related risks MetroPCS now faces with its current indebtedness, described in detail in MetroPCS most recent Annual Report on Form 10-K, as updated by subsequent Quarterly Reports on Form 10-Q. See Summary of the Financing \$500 Million Working Capital Revolving Credit Facility beginning on page 116.

## Upon a change in control triggering event, the combined company may be required to offer to repurchase all of the debt securities to be issued in connection with the transaction; the combined company may not have the ability to finance such repurchase.

Upon a change in control triggering event, as described in Exhibit G to the business combination agreement, the combined company may be required to offer to repurchase all of the debt securities to be issued in connection with the transaction and the Wireless existing notes, in an aggregate principal amount of up to \$20.5 billion, at a price equal to 101% of their principal amount, plus accrued and unpaid interest, if any, to the date of repurchase. In addition, pursuant to a noteholder agreement between the combined company and Deutsche Telekom, in the event of a change in control that is not caused by Deutsche Telekom, Deutsche Telekom will have the right to require the combined company to redeem any of the notes held by Deutsche Telekom, even if a change in control triggering event has not occurred. If a change in control triggering event, or an event requiring it to purchase notes held by Deutsche Telekom, were to occur, the combined company may not have sufficient funds to pay the change in control purchase price and may be required to obtain third-party financing in order to do so. However, the combined company may not be able to obtain such financing on commercially reasonable terms, or at all.

The combined company s failure following a change in control triggering event to make or consummate an offer to purchase the notes would constitute an event of default under the documentation governing the notes. In such an event, the trustee or the holders of at least 25% in aggregate principal amount of the outstanding notes may accelerate the maturity of all of the notes. In addition, any such event of default would likely trigger an event of default on other outstanding or future indebtedness of the combined company.

## The agreements governing the combined company s indebtedness will include restrictive covenants that limit the combined company s operating flexibility.

The agreements governing the combined company s indebtedness will impose material operating and financial restrictions on the combined company. These restrictions, subject in certain cases to customary baskets, exceptions and incurrence-based ratio tests, may limit the combined company s ability to engage in some transactions, including the following:

incurring additional indebtedness and issuing preferred stock;

paying dividends, redeeming capital stock or making other restricted payments or investments;

selling or buying assets, properties or licenses;

developing assets, properties or licenses which the combined company has or in the future may procure;

creating liens on assets;

participating in future FCC auctions of spectrum or private sales of spectrum;

engaging in mergers, acquisitions, business combinations, or other transactions;

entering into transactions with affiliates; and

placing restrictions on the ability of subsidiaries to pay dividends or make other payments. These restrictions could limit the combined company s ability to obtain debt financing, repurchase stock, refinance or pay principal on its outstanding indebtedness, complete acquisitions for cash or indebtedness or react to changes in its operating environment or the economy. Any future indebtedness that the combined company incurs may contain similar or more restrictive covenants. Any failure to comply with the restrictions of the combined company s debt agreements may result in an event of default under these agreements, which in turn may result in defaults or acceleration of obligations under these agreements and other agreements, giving the combined company s lenders the right to terminate any commitments they had made to provide it with further funds and to require the combined company to repay all amounts then outstanding.

## Deutsche Telekom will be subject to non-compete limitations for a finite period of time, after which it may compete with the combined company.

The stockholder s agreement will restrict Deutsche Telekom from competing with the combined company in the United States, Puerto Rico and the territories and protectorates of the United States, which we refer to as the territory, during the period beginning on the date of the closing and ending two years after the date on which Deutsche Telekom beneficially owns less than 10% of the outstanding shares of the combined company s common stock. Specifically, during such period, neither Deutsche Telekom nor any of its other controlled affiliates will be permitted to engage in providing wireless telecommunications services through a facilities-based network in the territory, hold licenses from the FCC related to or necessary to provide such services, act as a reseller, dealer or distributor of such services in the territory, or act as a mobile virtual network operator in the territory. In addition, for the period commencing at the closing and expiring on the first anniversary of the termination of the trademark license in accordance with its terms, Deutsche Telekom may not manufacture, market or distribute any products or services under, or use in any way, the trademark T-MOBILE in connection with any of the activities described in the previous sentence, other than by the combined company and its affiliates in accordance with the terms of the trademark license. See the sections entitled Summary of Ancillary Agreements Stockholder s Agreement and Summary of Ancillary Agreements Trademark License beginning on pages 154 and 158, respectively.

Following the expiration of the periods described above, Deutsche Telekom and its controlled affiliates may compete directly with the combined company in the territory. Because the duration of the covenant not to compete depends on Deutsche Telekom s equity ownership and not the amount of combined company debt it holds, Deutsche Telekom could be a significant holder of the combined company indebtedness and be able to compete with the combined company. Further, although the combined company can pursue opportunities outside the territory, the covenant not to compete does not prevent Deutsche Telekom from competing with the combined company outside of the territory, even while Deutsche Telekom holds a majority of the combined company s common stock. Because Deutsche Telekom has significant resources and substantial knowledge and expertise in the wireless telecommunications industry, such competition may be harmful to the business of the combined company.

## The combined company will be obligated to pay a royalty for the T-Mobile trademarks equal to a percentage of its net revenue from the products and services sold using the T-Mobile trademarks, and it will not be entitled to use the T-Mobile trademarks indefinitely.

Pursuant to the trademark license, the combined company will receive a limited license to certain T-Mobile trademarks, including the right to use the trademark T-Mobile as a name for the combined company, for use in connection with wireless telecommunications, broadband and information services and products used in connection therewith in the territory. Pursuant to the trademark license, the combined company is obligated to pay Deutsche Telekom a royalty in an amount equal to 0.25%, which we refer to as the royalty rate, of net revenue generated by the products and services sold by the combined company under the licensed trademarks. See the section entitled Summary of Ancillary Agreements Trademark License beginning on page 158. In the event that, during the term of the trademark license, the combined company s relative use of the licensed trademarks (versus other trademarks not licensed from Deutsche Telekom, such as MetroPCS) expands, increasing the amount of net revenue utilized to calculate the royalty payments, or the amount of revenue attributable to products or services sold using the licensed trademarks license, the combined company and Deutsche Telekom from the combined company may increase. On the fifth anniversary of the trademark license, the combined company and Deutsche Telekom have agreed to adjust the royalty rate based on the then average commercial royalty rate found under similar licenses for trademarks in the field of wireless telecommunication, broadband and information products and services in the territory through a binding benchmarking process. We cannot predict what the adjusted royalty rate will be in five years with any certainty. It is possible that it will be higher than the current royalty rate, which is more fully described above and under the section entitled Summary of Ancillary Agreements Trademark License beginning on page 158.

In addition, the term of the trademark license is limited. The initial term of the trademark license is approximately five years. The trademark license automatically renews for an additional five year term unless the combined company provides notice of its intent not to renew the trademark license. Thereafter, the trademark license automatically renews for subsequent five year periods unless the combined company provides 12 months notice prior to the expiration of the then-current term. The trademark agreement assigns all goodwill created as a result of the use of the T-Mobile trademark to Deutsche Telekom. The combined company and Deutsche Telekom are also obligated to negotiate a new trademark license in any of the following events: (i) Deutsche Telekom s ownership of the voting power of the outstanding shares of capital stock of the combined company falls to 50% or less; or (ii) any third party becomes the owner or otherwise comes to control, directly or indirectly, 50% or more of the voting power of the outstanding shares of capital stock of the combined company, or otherwise acquires the power to direct or cause the direction of the management and policies of the combined company. While we cannot predict what the adjusted royalty rate would be under such new trademark license, it is possible that it would be higher than the royalty rate, which is more fully described above and under the section entitled Summary of Ancillary Agreements Trademark License beginning on page 158. If the combined company and Deutsche Telekom fail to agree on a new trademark license after such event, either the combined company or Deutsche Telekom may terminate the trademark license to be effective, in the case of clause (i) above, on the third anniversary after notice of termination and, in the case of clause (ii) above, on the second anniversary after notice of termination. Additionally, the combined company may terminate the trademark license at any time upon notice to Deutsche Telekom, with such termination to be effective on the first anniversary after notice of termination is given. See the section entitled Summary of Ancillary Agreements Trademark License beginning on page 158.

Following the termination of the trademark license, the combined company may not use any T-Mobile trademarks, including the right to use the trademark T-Mobile as a name for the combined company. In such a case, the combined company will need to develop, promote and maintain new trademarks and a new brand. Developing, promoting and maintaining new trademarks and a new brand will likely require substantial capital, marketing and other expenditures. There can be no assurance that such new trademark or brand will be as recognizable or valuable as the T-Mobile trademark or brand. If the combined company fails to develop, promote and maintain new trademarks and a new brand, it could reduce the combined company s ability to attract and retain customers which could lead to greater churn, lower growth, reduced revenues, and lower profitability, all

of which could have a material adverse effect on the combined company s business, assets, liabilities, prospects, outlook, financial condition and results of operations. Further, the combined company will lose all of the investment it has made in the T-Mobile trademark. Any of these results could be harmful to the business, assets, liabilities, prospects, outlook, financial condition or results of operations of the combined company.

# The MetroPCS existing stockholder rights plan will remain in effect with respect to the combined company after completion of the transaction and could prevent a change in control of the combined company in instances in which some stockholders may believe a change in control is in their best interests.

MetroPCS currently has an existing stockholder rights plan, which we refer to as the MetroPCS rights plan, that will remain in effect with respect to the combined company after completion of the transaction. MetroPCS executed an amendment to the MetroPCS rights plan prior to execution of the business combination agreement that exempts Deutsche Telekom, including the acquisition of MetroPCS shares in the transaction, from the effects of the MetroPCS rights plan. See Summary of Ancillary Agreements Rights Agreement Amendment beginning on page 160. Pursuant to the MetroPCS rights plan, MetroPCS has issued to its stockholders one preferred stock purchase right for each outstanding share of MetroPCS common stock as of March 27, 2007. The shares issued in the transaction will include these purchase rights. Each right, when exercisable, will entitle its holder to purchase from the combined company a unit consisting of one one-thousandth of a share of series A junior participating preferred stock at \$66.67 per share, subject to adjustment as a result of the reverse stock split. The MetroPCS rights plan is intended to protect stockholders in the event of an unfair or coercive offer to acquire the combined company and to provide the combined company s board of directors with adequate time to evaluate unsolicited offers. The MetroPCS rights plan will cause substantial dilution to a person or group that attempts to acquire the combined company on terms that the board of directors of the combined company does not believe are in the combined company s and its stockholders best interest and may discourage, delay or prevent a merger or acquisition that stockholders may consider favorable, including transactions in which stockholders might otherwise receive a premium for their shares.

#### Adverse changes in the credit markets could increase the combined company s borrowing costs and the availability of funding.

The combined company will require a significant amount of capital to operate and grow its business. We anticipate that the combined company will need to fund its capital needs in part through borrowings in the credit markets in future. Adverse changes in the credit markets, including increases in interest rates, could increase the combined company s cost of borrowing and make it more difficult for the combined company to obtain future financing for its operations. In addition, the combined company s borrowing costs could be affected by short and long-term debt ratings assigned by independent rating agencies, which would be based, in significant part, on its performance as measured by customer credit metrics. A decrease in these ratings would likely increase the combined company s cost of borrowing and/or make it more difficult for it to obtain financing. In addition, a severe disruption in the global financial markets could impact some of the financial institutions with which the combined company would do business, and such instability could affect the combined company s future access to financing.

## If the combined company is unable to take advantage of technological developments in the wireless telecommunications industry on a timely basis, it may experience a decline in the demand for its services, be unable to implement its business strategy, and experience reduced profits.

The incumbent wireless telecommunications industry is experiencing rapid change and disruptive innovation on many fronts as new technologies are developed that offer consumers a broadening array of choices for their communications needs. For example, Apple, Inc. transformed the wireless landscape with the launch of the iPhone handset, Clearwire Corporation hopes to transform the wireless telecommunications industry with fixed mobile convergence, and Google Inc. introduced its open-source Android operating system in 2008. While smartphone use is expected to continue to grow, tablet sales have also increased dramatically. Rapid penetration

of smartphones and tablets will require carriers to invest in device subsidization and network improvements. For example, T-Mobile is in the process of transforming and upgrading its network to be the first in the United States to deploy LTE Release 10 and the first to use multimode integrated radios that can handle Global System for Mobile Communications, which we refer to as GSM, Evolved High Speed Packet Access, which we refer to as HSPA+, and LTE. As part of the network upgrade, T-Mobile will install new equipment in more than 37,000 cell sites and refarm its Personal Communications Service in the PCS 1900 MHz spectrum band from second generation GSM services to HSPA+. Modernizing the combined company s network will carry significant benefits, but it will also involve some risk, because it involves equipment changes, refarming of spectrum, and migration of customers from existing spectrum bands.

In order to grow and remain competitive, the combined company will need to adapt to future changes in technology, enhance its existing offerings, and introduce new offerings to address its customers changing demands. If the combined company is unable to meet future challenges from competing technologies on a timely basis or at an acceptable cost, it could lose customers to its competitors. In general, the development of new services in the wireless telecommunications industry will require the combined company to anticipate and respond to the varied and continuously changing demands of its customers. The combined company may not be able to accurately predict technological trends or the success of new services in the market. In addition, there could be legal or regulatory restraints on its introduction of new services. If the combined company s services fail to gain acceptance in the marketplace, or if costs associated with these services materially increase, the combined company s ability to retain and attract customers could be adversely affected, which could have a material adverse effect on the combined company s business, assets, liabilities, financial condition and results of operations.

#### **Other Risk Factors of MetroPCS**

#### Wireless existing senior credit facility may be terminated prior to the closing of the transaction.

If Wireless successfully issues the \$2.5 billion notes to third parties to refinance the Wireless existing senior credit facility prior to the closing of the transaction, the Wireless existing senior credit facility (including the existing \$100 million revolving credit sub-facility) may be terminated in advance of the closing of the transaction. Accordingly, in such event, if the closing does not occur, (i) the interest payments incurred by Wireless may increase because the \$2.5 billion notes may have a higher interest rate than the Wireless existing senior credit facility, and (ii) Wireless may not have access to a revolving credit facility to cover any cash flow needs of its business.

#### Wireless issuance of the \$1.0 billion notes will result in an increase in long-term indebtedness.

If Wireless issues the \$1.0 billion notes to third parties to raise new capital (or any portion of such notes) prior to the closing of the transaction, Wireless long-term indebtedness will increase by the principal amount of such notes issued. This debt issuance will result in a higher total leverage, and increased financing costs, for Wireless if the transaction does not close.

MetroPCS will be subject to the risks described above. In addition, MetroPCS is, and will continue to be, subject to the risks described in MetroPCS most recent Annual Report on Form 10-K, as updated by subsequent Quarterly Reports on Form 10-Q, all of which are filed with the SEC and incorporated by reference into this proxy statement. The combined company also will be subject to similar risks after completion of the transaction as those described in MetroPCS periodic reports filed with the SEC. You also should be aware that the risk factors disclosed in this proxy statement in MetroPCS filings with the SEC may not describe every risk facing MetroPCS or the combined company or that may affect MetroPCS or the combined company s business, assets, liabilities, financial condition or results of operations. There may be other known, unknown or unpredictable economic, business, competitive, regulatory or other risk factors that could also have a material adverse effect on our business, financial condition and results of operations in the future. In addition, past performance may not be a reliable indicator of future performance and historical trends should not be used to anticipate results on trends on future periods. See the section entitled Where You Can Find More Information beginning on page 210 for the location of information incorporated by reference in this proxy statement.

#### PARTIES TO THE AGREEMENT

#### **MetroPCS** Communications, Inc.

2250 Lakeside Boulevard

Richardson, Texas 75082

MetroPCS currently provides wireless broadband mobile services primarily in select major metropolitan areas in the United States, including: Las Vegas, Nevada; Los Angeles, San Francisco and Sacramento, California; Detroit, Michigan; Dallas/Fort Worth, Texas; Tampa/Sarasota, Orlando, Miami and Jacksonville, Florida; Atlanta, Georgia; Philadelphia, Pennsylvania; New York, New York; Boston, Massachusetts; and Hartford, Connecticut. As of September 30, 2012, MetroPCS held licenses for wireless spectrum suitable for wireless broadband mobile services covering a total population of 142.5 million people in and around many of the largest metropolitan areas in the United States. In addition, MetroPCS has roaming agreements with other wireless broadband mobile carriers that allow it to offer customers service when they are outside its service area. These roaming agreements, together with the area MetroPCS serves with its networks, allow customers to receive service in an area covering over 280 million in total population under the Metro USA<sup>®</sup> brand. MetroPCS provides services using code division multiple access, which we refer to as CDMA, networks using 1xRTT technology and evolution-data only, which we refer to as EVDO, and, long-term evolution, which we refer to as LTE, technology. MetroPCS introduced the first commercial LTE service in the United States in the Las Vegas and Dallas/Fort Worth metropolitan areas in September 2010 and, at that time, launched the world s first dual mode LTE/CDMA handset. Subsequently, through the remainder of 2010 and early 2011, MetroPCS launched LTE service in the remainder of its major metropolitan areas and launched the world s first LTE handset based on the android operating system. In 2012, MetroPCS launched the world s first commercial voice over LTE, which we refer to as VoLTE, service.

MetroPCS is the fifth largest facilities-based wireless broadband mobile communications provider in the United States based on number of customers served. MetroPCS offers wireless broadband mobile services under the MetroPCS<sup>®</sup> brand in selected major metropolitan areas in the United States. MetroPCS provides a variety of wireless broadband mobile communications services to customers on a no long-term contract, paid-in-advance basis. As of September 30, 2012, MetroPCS had approximately 9.0 million customers.

MetroPCS was incorporated in 2004 in the state of Delaware and maintains its corporate headquarters in Richardson, Texas. All services are provided through wholly-owned subsidiaries of Wireless, an indirect wholly-owned subsidiary of MetroPCS. In April 2007, MetroPCS consummated an initial public offering of its common stock and became listed for trading on the NYSE under the symbol PCS.

#### **Deutsche Telekom AG**

Friedrich-Ebert-Alle 140

#### 53113 Bonn, Germany

Deutsche Telekom is an *Aktiengesellschaft* organized in Germany and is one of the world s leading integrated telecommunications companies with over 131 million mobile customers, 33 million fixed-network lines and more than 17 million broadband lines (as of September 30, 2012). Deutsche Telekom and its affiliates provide fixed-network, mobile communications, Internet and IPTV products and services for consumers and ICT solutions for business and corporate customers. Deutsche Telekom is present in around 50 countries and has over 233,000 employees worldwide. Deutsche Telekom and its affiliates generated revenue of EUR 58.7 billion in the 2011 financial year over half of it outside Germany.

#### T-Mobile Global Zwischenholding GmbH

Friedrich-Ebert-Alle 140

#### 53113 Bonn, Germany

Global is a Gesellschaft mit beschränkter Haftung organized in Germany and a direct wholly-owned subsidiary of Deutsche Telekom.

#### **T-Mobile Global Holding GmbH**

Friedrich-Ebert-Alle 140

#### 53113 Bonn, Germany

Holding is a Gesellschaft mit beschrankter Haftung organized in Germany and a direct wholly-owned subsidiary of Global.

#### T-Mobile USA, Inc.

12920 SE 38th Street

#### Bellevue, Washington 98006

T-Mobile, a privately-held Delaware corporation and direct wholly-owned subsidiary of Holding, is a wireless telecommunications carrier that offers mobile communications services under the T-Mobile brand in the United States, Puerto Rico and the U.S. Virgin Islands. T-Mobile operates its business in one reportable segment. T-Mobile s service offerings include contract and non-contract (prepaid) wireless voice, messaging and data services plans, mobile broadband plans and wholesale wireless services. As of September 30, 2012, T-Mobile provides service to more than 33.3 million customers through its nationwide voice and data networks.

As of September 30, 2012, T-Mobile s nationwide coverage footprint offers wireless voice, messaging, and data services across a service area covering over 280 million Americans in most metropolitan areas. In addition, T-Mobile has roaming agreements with other mobile communication network operators that allow mobile communications services to be provided beyond the direct coverage area of the T-Mobile network footprint. Services from T-Mobile are provided using the Global System for Mobile Communications, which we refer to as GSM, General Packet Radio Service, which we refer to as GPRS, Enhanced Data rates for GSM Evolution, which we refer to as EDGE, Universal Mobile Telecommunications Systems, which we refer to as UMTS, and Evolved High Speed Packet Access, which we refer to as HSPA+, technologies. T-Mobile s network modernization plan will result in the launch of 4G services using LTE technology in areas covering a population of 200 million in 2013.

#### THE SPECIAL MEETING

We are furnishing this proxy statement to our stockholders as part of the solicitation of proxies by the MetroPCS board for use at the special meeting, to be held on [], and at any continuation, adjournment, postponement or continuation thereof. This document is first being mailed to our stockholders on or about [].

#### Time and Place of the Special Meeting

The special meeting will be held [in the Bank of America Theater at the Eisemann Center located at 2351 Performance Drive, Richardson, Texas 75082] on [] at 10:00 a.m., local time.

#### **Purpose of the Special Meeting**

The purpose of the special meeting is to vote upon the following proposals:

The stock issuance proposal: A proposal to approve the MetroPCS stock issuance to be made in connection with, and in order to give effect, to the transaction.

The new certificate of incorporation proposal: A proposal to approve the new certificate of incorporation pursuant to the business combination agreement to, among other things, effect certain aspects of the recapitalization, including the reverse stock split.

The change in control payments proposal: A proposal to approve, on a non-binding, advisory basis, the change in control payments.

**The adjournment proposal**: A proposal to approve the continuation, adjournment or postponement of the special meeting, if necessary or appropriate to solicit additional proxies if there are insufficient votes at the time of the special meeting to approve both the stock issuance proposal and the new certificate of incorporation proposal.

## IF THE METROPCS STOCKHOLDERS FAIL TO APPROVE BOTH THE STOCK ISSUANCE PROPOSAL AND THE NEW CERTIFICATE OF INCORPORATION PROPOSAL, THE TRANSACTION WILL NOT OCCUR.

The only MetroPCS stockholder approvals required by the business combination agreement are the stock issuance proposal and the new certificate of incorporation proposal. The change in control payments proposal is non-binding and advisory and the vote on such proposal will have no impact on whether the transaction is completed. In addition, even if the MetroPCS stockholders approve both the stock issuance proposal and the new certificate of incorporation proposal, the transaction may not be completed if the other conditions to closing the transaction are not satisfied or, if allowed by applicable law, waived. We can give no assurance that the conditions to closing the transaction will be satisfied or so waived.

#### **Other Business**

The MetroPCS bylaws provide that only the MetroPCS board may propose business to be discussed at a special meeting of stockholders. Further, at a special meeting of stockholders, the business discussed must be specified in the notice of meeting, or any supplement thereto. At the special meeting, no matters may come before the stockholders other than the proposals presented herein or in any supplement hereto.

#### **Recommendation of the MetroPCS Board**

The MetroPCS board, acting upon the recommendation of the special committee, recommends that you vote as follows:

**Recommended Vote** Proposal Stock Issuance Proposal (Item 1) FOR the stock issuance to be made in connection with, and in order to give effect to, the transaction New Certificate of Incorporation Proposal (Item 2) FOR the new certificate of incorporation pursuant to the business combination agreement to, among other things, effect certain aspects of the recapitalization, including the reverse stock split Change in Control Payments Proposal FOR (Item 3) the approval, on a non-binding, advisory basis, of the compensation that may be paid or may become payable to MetroPCS named executive officers based on or otherwise relating to the transaction Adjournment Proposal (Item 4) FOR the continuation, adjournment or postponement of the special meeting, if necessary or appropriate to solicit additional proxies if there are insufficient votes at the time of the special meeting to approve both the stock issuance proposal and the new

#### **Record Date and Quorum**

The MetroPCS board has established the close of business on [] as the record date for the special meeting. Only record holders of shares of MetroPCS common stock at the close of business on the record date for the special meeting are entitled to notice of, and to vote at, the special meeting and any continuations, adjournments or postponements of the special meeting. No other shares of MetroPCS capital stock are entitled to notice of and to vote at the special meeting. At the close of business on the record date, MetroPCS had outstanding and entitled to vote [] shares of MetroPCS common stock. Holders of MetroPCS common stock have one vote per share on each matter to be acted upon.

certificate of incorporation proposal

In order to conduct business at the special meeting, holders of a majority of the outstanding shares of MetroPCS common stock entitled to vote on the record date must be present in person or represented by proxy at the special meeting for there to be a quorum. In addition, certain unvested shares of restricted stock issued pursuant to the MetroPCS equity incentive compensation plans will count towards a quorum because such shares will be voted by MetroPCS on the proposals to be considered at the special meeting in the same proportion as the rest of the shares of MetroPCS common stock that are voted. It is important that you provide us with your proxy or attend the special meeting in person so that your shares are counted toward the quorum. If you hold your shares through a bank, broker, custodian or other record holder, please refer to your proxy card, voting instruction form, or the information forwarded by your bank, broker, custodian or other record holder to determine how and

#### Table of Contents

#### Edgar Filing: MILLER PETROLEUM INC - Form 10-K

when to vote your shares. Unless you direct your bank, broker, custodian or other record holder on how to vote by the time and date specified by them, they will be unable to vote your shares. We encourage you to provide us with your proxy card even if you plan on attending the special meeting in person to ensure that your vote will be counted.

All shares of MetroPCS common stock represented at the special meeting, including abstentions, will be treated as shares that are present and entitled to vote for purposes of determining the presence of a quorum. Because all of the proposals are considered non-routine matters, we do not expect to receive any broker non-votes with respect to any of the proposals to be considered at the special meeting.

#### Attendance

Only stockholders with an admission ticket will be admitted to the special meeting. If you are a record holder of MetroPCS common stock, an admission ticket is attached to your proxy card. However, if you hold your shares of MetroPCS common stock through a bank, broker, custodian or other record holder, you should ask the bank, broker, custodian or other record holder that holds your shares to provide you with a legal proxy, a copy of your account statement, or a letter from the record holder confirming that you beneficially own or hold MetroPCS common stock as of the close of business on the record date. You also can obtain an admission ticket to the special meeting by presenting this legal proxy, or confirming documentation of your account from your bank, broker, custodian or other record holder, at the special meeting. All stockholders will be required to show a valid, government-issued, picture identification that matches the name on the admission ticket or legal proxy or confirming documentation from your bank, broker, custodian or other record holder before being admitted to the special meeting.

For safety and security purposes, we do not permit any stockholder to bring cameras, video or audio recording equipment, large bags, briefcases or packages into the meeting room or to otherwise record or photograph the special meeting. We also ask that all MetroPCS stockholders attending the special meeting not bring cell phones into the special meeting or that they turn off all cell phones, pagers, and other electronic devices during the special meeting. We reserve the right to inspect any bags, purses or briefcases brought into the special meeting.

#### **Vote Required**

The required vote to approve each proposal generally is as set forth in the table below. Please see the description immediately following the table for more details on the required vote to approve each proposal.

Proposal	Vote Required
Stock Issuance Proposal (Item 1)	Affirmative vote of the majority of the votes cast, provided that the holders of a majority of the MetroPCS common stock outstanding on the record date vote on this proposal
New Certificate of Incorporation Proposal (Item 2)	Affirmative vote of the majority of the MetroPCS common stock outstanding on the record date
Change in Control Payments Proposal (Item 3)	Affirmative vote of the majority of the votes cast
Adjournment Proposal (Item 4)	If a quorum is not present, the affirmative vote of the majority of the shares of MetroPCS common stock present (in person or by proxy) at the special meeting and entitled to vote; if a quorum is present, the affirmative vote of the majority of the votes cast

The stock issuance proposal: Assuming the presence of a quorum, the affirmative vote of a majority of the votes cast (in person or by proxy) by holders of MetroPCS common stock at the special meeting is required to approve the stock issuance proposal, provided that the holders of a majority of the MetroPCS common stock outstanding on the record date vote on the stock issuance proposal. If you vote to abstain, it will have the same effect as voting against this proposal. If you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have no effect on the voting outcome of this proposal, but it will make it more

difficult to have a quorum or meet the requirement that the holders of a majority of the MetroPCS common stock outstanding on the record date vote on this proposal. Accordingly, it is important that you provide us with your proxy or attend the special meeting in person so that your shares are counted toward the quorum and this requirement.

The new certificate of incorporation proposal: Assuming the presence of a quorum, the affirmative vote of a majority of the MetroPCS common stock outstanding on the record date is required to approve the new certificate of incorporation proposal. If you vote to abstain, or if you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have the same effect as voting against this proposal.

The change in control payments proposal: Assuming the presence of a quorum, the affirmative vote of a majority of the votes cast (in person or by proxy) by holders of MetroPCS common stock at the special meeting is required to approve, on a non-binding, advisory basis, the change in control payments proposal. While the MetroPCS board intends to consider the vote resulting from the change in control payments proposal, the vote is advisory and therefore not binding on MetroPCS, the MetroPCS board or the compensation committee of the MetroPCS board. Accordingly, because MetroPCS is contractually obligated to pay the change in control payments, if the transaction is completed, the change in control payments will be payable, subject only to the conditions applicable to such payments, regardless of the outcome of the advisory, non-binding vote. If you vote to abstain, or if you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have no effect on the voting outcome of this proposal.

#### The adjournment proposal:

If a quorum is not present at the special meeting:

The affirmative vote of a majority of the shares of MetroPCS common stock present at the special meeting (in person or by proxy) and entitled to vote on this proposal is required to approve this proposal.

If you vote to abstain, it will have the same effect as voting against this proposal.

If you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have no effect on the voting outcome of this proposal.

If a quorum is present at the special meeting:

The affirmative vote of a majority of the votes cast (in person or by proxy) by holders of MetroPCS common stock at the special meeting is required to approve this proposal.

If you vote to abstain, it will have no effect on the voting outcome of this proposal.

If you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have no effect on the voting outcome of this proposal.

#### Voting by MetroPCS Directors and Executive Officers

As of the record date for the special meeting, MetroPCS directors and executive officers had the right to vote []% of the MetroPCS common stock outstanding and entitled to vote at the special meeting. MetroPCS currently expects that MetroPCS directors and executive officers will vote their shares of MetroPCS common stock in favor of each of the proposals to be considered at the special meeting, although none of them has entered into any agreements obligating them to do so.

## Table of Contents

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

In addition, any unvested shares of restricted stock issued to MetroPCS directors and executive officers pursuant to the MetroPCS equity incentive compensation plans that do not have voting rights will be voted by MetroPCS for the proposals to be voted on at the special meeting in the same proportion as the rest of the shares of MetroPCS common stock that are voted. As of the record date for the special meeting, MetroPCS directors and executive officers held [] unvested shares of restricted stock that will be voted in this manner.

## Voting

If your proxy card is properly executed and received by MetroPCS in time to be voted at the special meeting, the shares of MetroPCS common stock represented by your proxy (including those given electronically via the Internet or by telephone) will be voted in accordance with the instructions that you mark on your proxy card.

Internet voting will be available until 11:59 p.m. EDT on [] or, if the special meeting is continued, adjourned or postponed, until 11:59 p.m. EDT on the day immediately before such continued, adjourned or postponed meeting. Proxy cards submitted by mail and received by MetroPCS after [] at 10:00 a.m. local time may not be considered unless the special meeting is continued, adjourned or postponed, and then only if received before the date and time the continued, adjourned or postponed special meeting is held. Telephone voting will be considered at the special meeting if completed prior to 11:59 p.m. EDT on [] or, if the special meeting is continued, adjourned or postponed, until 11:59 p.m. EDT on the day immediately before such adjourned or postponed special meeting.

Any executed proxies received by MetroPCS for which no voting instructions are provided on one or more proposals will be voted in accordance with the recommendations of the MetroPCS board with respect to such proposal(s) for which no voting instructions are provided.

#### Revocation

You may change or revoke your proxy at any time prior to the vote on the matters at the special meeting or, if the special meeting is continued, adjourned or postponed meeting. If you are a record holder of MetroPCS common stock, you may revoke your proxy at any time prior to the voting deadlines referred to in the section entitled Questions and Answers about the Special Meeting, the Transaction and the Business Combination Agreement How do I vote? beginning on page xiii by (1) delivering to MetroPCS Corporate Secretary at our principal executive office, located at 2250 Lakeside Boulevard, Richardson, Texas 75082, a written revocation that must be received by MetroPCS prior to the date and time of the special meeting, or, if the special meeting is continued, adjourned or postponed, the date and time of such continued, adjourned or postponed meeting, (2) submitting another valid proxy card with a later date by mail, (3) voting by submitting a proxy by telephone or Internet prior to the date and time of the special meeting, or, if the special meeting is continued, adjourned or postponed, the date and time of such continued, adjourned or postponed meeting, or (4) attending the special meeting in person and giving MetroPCS Inspector of Elections notice of your intent to vote your shares in person. If your shares are held in street name, you must contact the record holder of your MetroPCS Corporate Secretary prior to the date and time of the special meeting, or, if the special meeting, or, if the special meeting is continued, adjourned or postponed, by the date and time of such continued, adjourned to revoke your proxy. If you intend to revoke your proxy, you must ensure that such revocation is received by MetroPCS Corporate Secretary prior to the date and time of the special meeting. Any revocation received as of or after that date and time will not be effective. Attendance at the special meeting will not, by itself, revoke a proxy.

#### Anticipated Date of Completion of the Transaction

We expect that the conditions to the transaction will be satisfied or, if allowed by applicable law, waived, and the transaction will be completed, in the first half of 2013. However, we cannot be certain when, or if, the conditions to the transaction will be satisfied or so waived or that the transaction will be completed.

As more fully described in this proxy statement and in the business combination agreement, the completion of the transaction depends on a number of conditions being satisfied or, where legally permissible, waived. For a complete description of the conditions to closing the transaction, see the section entitled Summary of the Business Combination Agreement Conditions to Closing the Transaction beginning on page 149.

#### **No Appraisal Rights**

Under applicable law, MetroPCS stockholders do not have the right to an appraisal of the value of their shares in connection with the transaction.

#### Solicitation of Proxies; Payment of Solicitation Expenses

MetroPCS bears all of the cost of the solicitation of proxies, including the preparation, assembly, printing and mailing of all proxy materials. MetroPCS also reimburses banks, brokers, custodians and other record holders for their costs in forwarding the proxy materials to the beneficial owners or holders of MetroPCS common stock. MetroPCS and its directors, officers, and regular employees also may solicit proxies by mail, personally, by telephone or by other appropriate means. No additional compensation will be paid to directors, officers or other regular employees for such services. In addition, we have retained MacKenzie Partners, Inc., which we refer to as MacKenzie, to aid in the solicitation of proxies by mail, personally, by telephone, e-mail or other appropriate means. For these services, we will pay MacKenzie \$30,000, plus reasonable out-of-pocket expenses.

#### **Questions and Additional Information**

Additional business and financial information about MetroPCS can be found in documents previously filed with the SEC. This information is available to you without charge at the SEC s website at *http://www.sec.gov*. Some of this additional business and financial information is incorporated by reference into this proxy statement. See Where You Can Find More Information. You can also obtain additional copies of this proxy statement, as well as other relevant materials, by visiting MetroPCS website at *www.metropcs.com* or by requesting them in writing using the following contact information:

MetroPCS Communications, Inc.

2250 Lakeside Boulevard

Richardson, Texas 75082

Attention: Investor Relations

(214) 570-4641

#### Email: investor\_relations@metropcs.com

You may also request additional copies from our proxy solicitor, MacKenzie, using the following contact information:

MacKenzie Partners, Inc.

105 Madison Avenue

New York, New York 10016

#### Call Collect (212) 929-5500

or

Call Toll-Free (800) 322-2885

#### Email: proxy@mackenziepartners.com

#### Assistance

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

If you need assistance in completing your proxy card or have questions regarding the special meeting, please contact MacKenzie toll-free at (800) 322-2885. Banks, brokers, custodians or other record holders may call collect at (212) 929-5500.

## Stockholder List

A list of the record holders of MetroPCS common stock entitled to vote at the special meeting will be available at the meeting and for 10 days prior to the meeting, for any purpose germane to the meeting, between the hours of 9:00 a.m. and 4:30 p.m., local time, at our principal executive offices at 2250 Lakeside Boulevard, Richardson, Texas 75082.

#### PROPOSALS SUBMITTED TO STOCKHOLDERS

#### The Stock Issuance Proposal

#### (Item 1 on proxy card)

The business combination agreement provides that, as part of the transaction, MetroPCS will issue and deliver to Holding or its designee shares of MetroPCS common stock equal to 74% of the fully-diluted shares of MetroPCS common stock outstanding immediately following the cash payment.

Under the NYSE Listed Company Manual, a company listed on the NYSE is required to obtain stockholder approval prior to the issuance of common stock, or securities convertible into or exercisable for common stock, in any transaction or series of related transactions if the number of shares of common stock to be issued is, or will be upon issuance, equal to or in excess of 20% of the number of shares of common stock or securities convertible into or exercisable for common stock. As described above, if the transaction closes, MetroPCS will issue 74% of its outstanding shares of MetroPCS common stock in connection with the transaction.

#### **Required Vote**

Assuming the presence of a quorum, the affirmative vote of a majority of the votes cast (in person or by proxy) by holders of MetroPCS common stock at the special meeting is required to approve the stock issuance proposal, provided that the holders of a majority of the MetroPCS common stock outstanding on the record date vote on the stock issuance proposal. If you vote to abstain, it will have the same effect as voting against this proposal. If you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have no effect on the voting outcome of this proposal, but it will make it more difficult to have a quorum or meet the requirement that the holders of a majority of the MetroPCS common stock outstanding on the record date vote on that proposal. Accordingly, it is important that you provide us with your proxy or attend the special meeting in person so that your shares are counted toward the quorum and this requirement.

The MetroPCS board, acting upon the recommendation of the special committee, recommends a vote <u>FOR</u> the stock issuance proposal (Item 1). For a discussion of interests of MetroPCS directors and executive officers in the transaction that may be different from, or in addition to, MetroPCS stockholders generally, see the section entitled Interests of MetroPCS Directors and Officers in the Transaction beginning on page 118.

# IF THE METROPCS STOCKHOLDERS FAIL TO APPROVE THE STOCK ISSUANCE PROPOSAL, THE TRANSACTION WILL NOT OCCUR.

In addition, even if the MetroPCS stockholders approve the stock issuance proposal, the transaction may not be completed if the other conditions to closing the transaction are not satisfied or, if allowed by applicable law, waived. We can give no assurance that the conditions to closing the transaction will be satisfied or so waived.

#### The New Certificate of Incorporation Proposal

#### (Item 2 on proxy card)

In connection with the transaction and conditioned on the completion of the transaction, MetroPCS proposes to amend and restate its existing certificate of incorporation in its entirety in the form of the new certificate of incorporation attached as Exhibit A to the business combination agreement. The new certificate of incorporation will apply to the combined company following the completion of the transaction.

#### Effects of the New Certificate of Incorporation

The new certificate of incorporation effects a number of changes to MetroPCS existing certificate of incorporation, including the following:

<u>Company Name</u>. The new certificate of incorporation will change MetroPCS name to T-Mobile US, Inc. or another name selected by Deutsche Telekom prior to the completion of the transaction.

<u>Par Value and Reverse Stock Split</u>. The new certificate of incorporation will reduce the par value of the MetroPCS common stock and preferred stock from \$0.0001 to \$0.00001. In addition, the new certificate of incorporation will effect the reverse stock split, pursuant to which each share of MetroPCS common stock, par value \$0.0001 per share, issued and outstanding immediately before the effective time automatically will be, without any action on the part of MetroPCS or any MetroPCS stockholder, reclassified as, and converted into, one-half of a validly issued, fully paid and non-assessable share of MetroPCS common stock, par value \$0.0001 per share.

<u>Declassification</u>. The new certificate of incorporation will provide for the declassification of the MetroPCS board with all members of the MetroPCS board being elected annually. The existing certificate of incorporation provides for a classified board where one-third of the members of the MetroPCS board are elected each year.

<u>Board Representation</u>. The new certificate of incorporation will provide Deutsche Telekom with the same board and committee representation rights as are set forth in the stockholder s agreement. Specifically, Deutsche Telekom generally will have the right to designate a number of individuals, each, a Deutsche Telekom designee, to the combined company s board and any committees thereof equal to the percentage of the combined company s common stock beneficially owned by Deutsche Telekom multiplied by the number of directors on the combined company s board (or the number of members of any committee thereof), in each case, rounded to the nearest whole number. These rights will remain in effect as long as Deutsche Telekom beneficially owns 10% or more of the outstanding shares of the combined company s common stock. If at any time the number of Deutsche Telekom designees then serving as directors on the combined company s board or as members of any committee of the combined company s board exceeds the number of Deutsche Telekom designees that Deutsche Telekom is entitled to designate, Deutsche Telekom will be required to cause the number of Deutsche Telekom designees then serving as directors on the MetroPCS board or as members of such committee of the combined company s board representing such excess to resign immediately as directors or committee members, as applicable. In addition, Deutsche Telekom and the combined company will use their reasonable best efforts to cause at least three members of the combined company s board to be considered independent under the rules of the SEC and under applicable listing standards.

The existing certificate of incorporation does not provide any MetroPCS stockholder with board or committee representation rights of this nature. All of the directors of the MetroPCS board, except the chairman, are independent directors.

<u>Director Removal</u>. The new certificate of incorporation will provide that, subject to certain rights of the holders of preferred stock, any director may be removed from office at any time, with or without cause, by the affirmative vote of the holders of at least a majority of the voting power of all of the outstanding shares of MetroPCS capital stock entitled to elect such director, voting separately as a class, at a duly organized meeting of stockholders or by written consent. However, no director designated by Deutsche Telekom may be removed without the prior written consent of Deutsche Telekom. The existing certificate of incorporation provides that, subject to certain rights of holders of preferred stock, any director may be removed from office at any time, with or without cause, by the affirmative vote of the holders of at least a majority of the voting power underlying the outstanding shares of the MetroPCS capital stock entitled to elect such director, voting separately as a class.

<u>Approval Rights</u>. The new certificate of incorporation will provide Deutsche Telekom with the same approval rights as are set forth in the stockholder s agreement with respect to the combined company s ability to

take certain actions without Deutsche Telekom s prior written consent as long as Deutsche Telekom beneficially owns 30% or more of the outstanding shares of the combined company s common stock, such as actions relating to the incurrence of indebtedness, acquisitions or dispositions for consideration in excess of \$1 billion, a change in the size of the combined company s board, equity issuances constituting 10% or more of the then outstanding shares of the combined company s common stock, redemptions, dividends and hiring or terminating without cause the chief executive officer. These approval rights are more fully described in the section entitled Summary of Ancillary Agreements Stockholder s Agreement beginning on page 154.

The existing certificate of incorporation does not provide any MetroPCS stockholder with approval rights of this nature.

<u>Stockholder Meetings</u>. The new certificate of incorporation will provide that a special meeting of the combined company s stockholders (i) may be called by the chairman of the combined company s board or the combined company s chief executive officer and (ii) must be called by the combined company s secretary at the request of (a) a majority of the combined company s board or (b) as long as Deutsche Telekom beneficially owns 25% or more of the outstanding shares of combined company s common stock, the holders of not less than 33-1/3% of the voting power of all of the outstanding voting stock of the combined company entitled to vote generally for the election of directors. The existing certificate of incorporation provides that a special meeting of MetroPCS stockholders (x) may be called by the chairman of the MetroPCS board or MetroPCS chief executive officer or MetroPCS secretary at the request in writing of a majority of all of the directors on the MetroPCS board then in office.

Stockholder Action by Written Consent. The new certificate of incorporation will provide that, as long as Deutsche Telekom beneficially owns 25% or more of the outstanding shares of the combined company s common stock, any action required or permitted to be taken at any annual or special meeting of the combined company s stockholders may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing setting forth the action so taken is signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted. The existing certificate of incorporation does not permit MetroPCS stockholders to act by written consent.

<u>Power to Amend Bylaws</u>. The new certificate of incorporation will provide that the combined company s bylaws may be amended upon the affirmative vote of the holders of shares having a majority of the combined company s voting power. The existing certificate of incorporation requires the affirmative vote of the holders of 66 2/3% of MetroPCS voting power. The bylaws still may be amended by the combined company s board upon the affirmative vote of a majority of the directors then serving.

<u>Redemption</u>. The existing certificate of incorporation provides that MetroPCS may redeem the outstanding MetroPCS common stock or preferred stock of any MetroPCS stockholder in order to cure any violation of a rule or regulation of the FCC. The new certificate of incorporation will provide that such redemption right does not apply with respect to the combined company s common stock or preferred stock beneficially owned by Deutsche Telekom or any of its subsidiaries.

<u>Governing Law; Forum for Adjudication of Disputes</u>. The new certificate of incorporation will provide that the new certificate of incorporation and the internal affairs of the combined company will be governed by and interpreted under the laws of the State of Delaware. In addition, the Court of Chancery of the State of Delaware will be the sole and exclusive forum for (i) any derivative action brought on behalf of the combined company, (ii) any action asserting a claim of breach of a fiduciary duty owed by any director, officer or other employee of the combined company to the combined company or its stockholders, (iii) any action asserting a claim arising pursuant to any provision of the General Corporation Law of the State of Delaware, which we refer to as the DGCL, the new certificate of incorporation or the new bylaws, or (iv) any other action asserting a claim arising under, in connection with, and governed by the internal affairs doctrine. This provision currently is included in MetroPCS bylaws but not the existing certificate of incorporation.

#### Anti-Takeover Effects of the Provisions in our New Certificate of Incorporation and Bylaws

The following provisions of the new certificate of incorporation and our bylaws that will be adopted upon the completion of the transaction could be deemed to have an anti-takeover effect and could delay, defer or prevent a takeover attempt that a stockholder might consider to be in the stockholders best interests.

Advance notice of director nominations and matters to be acted upon at meetings. The combined company s bylaws will contain advance notice requirements for nominations for directors to our board of directors and for proposing matters that can be acted upon by stockholders at stockholder meetings. The existing bylaws contain similar advance notice requirements.

*Amendment to bylaws.* The new certificate of incorporation will provide that the combined company s bylaws may be amended upon the affirmative vote of the holders of shares having a majority of the combined company s voting power. The existing certificate of incorporation requires the affirmative vote of the holders of shares having 66 2/3% of the voting power of MetroPCS. Our bylaws also provide that the combined company s board will also be authorized to make, alter or repeal our bylaws without further stockholder approval.

*Special meeting of stockholders.* The new certificate of incorporation will provide that a special meeting of the combined company s stockholders (i) may be called by the chairman of the MetroPCS board or the combined company s chief executive officer and (ii) must be called by the combined company s secretary at the request of (a) a majority of the combined company s board or (b) as long as Deutsche Telekom beneficially owns 25% or more of the outstanding shares of combined company s common stock, the holders of not less than 33-1/3% of the voting power of all of the outstanding voting stock of the combined company entitled to vote generally for the election of directors. The existing certificate of incorporation does not permit the MetroPCS stockholders to call a special meeting of the stockholders.

*Board representation.* The new certificate of incorporation will provide that Deutsche Telekom generally will have the right to designate a number of Deutsche Telekom designees to the combined company s board and any committees thereof equal to the percentage of the combined company s common stock beneficially owned by Deutsche Telekom multiplied by the number of directors on the combined company s board (or the number of members of any committee thereof), in each case, rounded to the nearest whole number. These rights will remain in effect as long as Deutsche Telekom beneficially owns 10% or more of the outstanding shares of the combined company s common stock. The new certificate of incorporation will provide that all of the directors of the combined company s board will be of one class and will be elected annually. The existing certificate of incorporation provides for a classified board where one-third of the members of the MetroPCS board are elected each year. However, MetroPCS has indicated that it intends to present at its 2013 annual meeting of stockholders an amendment to its existing certificate of incorporation that will declassify the MetroPCS board.

*Special approval rights.* The new certificate of incorporation will provide Deutsche Telekom with the same approval rights as are set forth in the stockholder s agreement with respect to the combined company s ability to take certain actions (including changing the size of the combined company s board of directors or dispositions in excess of \$1 billion) without Deutsche Telekom s prior written consent as long as Deutsche Telekom beneficially owns 30% or more of the outstanding shares of the combined company s common stock. The existing certificate of incorporation does not provide any MetroPCS stockholder with approval rights of this nature.

Authorized but unissued shares. The authorized but unissued shares of our common stock and preferred stock are available for future issuance without stockholder approval. These additional

shares may be used for a variety of corporate purposes, such as for additional public offerings, acquisitions and employee benefit plans. The existence of authorized but unissued and unreserved common stock and preferred stock could render more difficult or discourage an attempt to obtain control of the combined company by means of a proxy contest, tender offer, merger or otherwise. The existing certificate of incorporation contains similar provisions.

*Cumulative voting*. The new certificate of incorporation will not permit cumulative voting in the election of directors. Instead, any election of directors will be decided by a plurality of the votes cast (in person or by proxy) by holders of the combined company s common stock. The existing certificate of incorporation does not permit cumulative voting in the election of directors.

#### **Delaware law**

MetroPCS is a Delaware corporation and is subject to Delaware law, which generally prohibits a publicly held Delaware corporation from engaging in a business combination with an interested stockholder for a period of three years after the time that the person became an interested stockholder, unless:

before such time the board of directors of the corporation approved either the business combination or the transaction in which the person became an interested stockholder;

upon completion of the transaction that resulted in the stockholder becoming an interested stockholder, the interested person owns at least 85% of the voting stock of the corporation outstanding at the time the transaction commenced, excluding shares owned by persons who are directors and also officers of the corporation and by certain employee stock plans; or

at or after such time the business combination is approved by the board of directors of the corporation and authorized at an annual or special meeting of stockholders, and not by written consent, by the affirmative vote of at least 66 2/3% of the outstanding voting stock of the corporation that is not owned by the interested stockholder.

A business combination generally includes mergers, asset sales and similar transactions between the corporation and the interested stockholder, and other transactions resulting in a financial benefit to the stockholder. An interested stockholder is a person:

who, together with affiliates and associates, owns 15% or more of the corporation s outstanding voting stock; or

who is an affiliate or associate of the corporation and, together with his or her affiliates and associates, has owned 15% or more of the corporation s outstanding voting stock within three years.

#### **Required Vote**

The MetroPCS board, acting upon the recommendation of the special committee, has adopted the new certificate of incorporation subject to the completion of the transaction, declared it advisable and has directed that it be submitted for stockholder approval. Assuming the presence of a quorum, the affirmative vote of a majority of the MetroPCS common stock outstanding on the record date is required to approve the new certificate of incorporation proposal. If you vote to abstain, or if you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have the same effect as voting against this proposal.

The MetroPCS board, acting upon the recommendation of the special committee, recommends a vote <u>FOR</u> the new certificate of incorporation proposal (Item 2). For a discussion of interests of MetroPCS directors and executive officers in the transaction that may be different from, or in addition to, MetroPCS stockholders generally, see the section entitled Interests of MetroPCS Directors and Officers in the Transaction beginning on page 118.

# IF THE METROPCS STOCKHOLDERS FAIL TO APPROVE THE NEW CERTIFICATE OF INCORPORATION PROPOSAL, THE TRANSACTION WILL NOT OCCUR.

In addition, even if the MetroPCS stockholders approve the new certificate of incorporation proposal, the transaction may not be completed if the other conditions to closing the transaction are not satisfied or, if allowed by applicable law, waived. We can give no assurance that the conditions to closing the transaction will be satisfied or so waived.

#### The Change in Control Payments Proposal

#### (Item 3 on proxy card)

The Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010 and Rule 14a-21(c) under the Exchange Act require MetroPCS to provide its stockholders with the opportunity to vote to approve, on a non-binding, advisory basis, the compensation that may be paid or become payable to the named executive officers of MetroPCS based on or otherwise relating to the transaction (also known as golden parachute compensation).

The compensation that MetroPCS named executive officers may be entitled to receive that is based on or otherwise relates to the transaction is summarized and quantified in the section entitled Interests of MetroPCS Directors and Officers in the Transaction Quantification of Change in Control and Termination Payments and Benefits to the MetroPCS Named Executive Officers beginning on page 123. This summary includes payments that may become payable under the change in control agreements with each named executive officer and the value of any awards issued pursuant to the MetroPCS equity incentive compensation plans that will vest in connection with the completion of the transaction.

Accordingly, MetroPCS is requesting that holders of MetroPCS common stock approve the following resolution:

RESOLVED, that the stockholders of MetroPCS Communications, Inc. approve, on a non-binding, advisory basis, the compensation that may be paid, or become payable, to its named executive officers based on or otherwise relating to the transaction, the closing of the transaction or actions following the closing of the transaction, as summarized and quantified, including the compensation table and the related narrative named executive officers of MetroPCS birectors and Executive Officers in the Transaction Quantification of Change in Control and Termination Payments and Benefits to the MetroPCS Named Executive Officers (referred to as the change in control payments).

Approval of this proposal is not a condition to completion of the transaction. Because this is an advisory vote, the result will not be binding on MetroPCS, the MetroPCS board or the compensation committee of the MetroPCS board. While the MetroPCS board intends to consider the vote resulting from the change in control payments proposal, the vote is advisory and therefore not binding on MetroPCS, the MetroPCS board or the compensation committee of the MetroPCS board. Accordingly, such compensation, including amounts that MetroPCS is contractually obligated to pay, could still be payable regardless of the outcome of this advisory vote, subject only to the conditions applicable thereto.

#### **Required Vote**

Assuming the presence of a quorum, the affirmative vote of a majority of the votes cast (in person or by proxy) by holders of MetroPCS common stock at the special meeting is required to approve, on a non-binding, advisory basis, the change in control payments proposal. While the MetroPCS board intends to consider the vote resulting from the change in control payments proposal, the vote is advisory and therefore not binding on MetroPCS, the MetroPCS board or the compensation committee of the MetroPCS board. Accordingly, because MetroPCS is contractually obligated to pay the change in control payments, if the transaction is completed, the change in control

payments will be payable, subject only to the conditions applicable to such payments, regardless of the outcome of the advisory vote. If you vote to abstain, or if you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have no effect on the voting outcome of this proposal.

The MetroPCS board, acting upon the recommendation of the special committee, recommends a vote <u>FOR</u> the change in control payments proposal (Item 3). For a discussion of interests of MetroPCS directors and executive officers in the transaction that may be different from, or in addition to, MetroPCS stockholders generally, see the section entitled Interests of MetroPCS Directors and Officers in the Transaction beginning on page 118.

#### The Adjournment Proposal

#### (Item 4 on proxy card)

The special meeting may be continued, adjourned or postponed to another time or place, if necessary or appropriate to permit further solicitation of proxies if necessary to obtain additional votes in favor of the stock issuance proposal or the new certificate of incorporation proposal. If, at the special meeting, the number of shares of MetroPCS common stock present or represented and voting in favor of the stock issuance proposal or the new certificate of incorporation proposal is insufficient to approve the corresponding proposals, MetroPCS intends to move to continue, adjourn or postpone the special meeting in order to enable the MetroPCS board to solicit additional proxies for approval of such proposals.

In the adjournment proposal, MetroPCS is asking its stockholders to authorize the holder of any proxy solicited by the MetroPCS board to vote in favor of granting discretionary authority to the proxy holders, and to each proxy holder individually, to continue, adjourn or postpone the special meeting to another time and place for the purpose of soliciting additional proxies. If the stockholders approve this proposal, MetroPCS could continue, adjourn or postpone the meeting and any continued, adjourned or postponed session of the meeting and use the additional time to solicit additional proxies, including the solicitation of proxies from stockholders who have previously voted.

#### **Required Vote**

If a quorum is not present at the special meeting:

The affirmative vote of a majority of the shares of MetroPCS common stock present at the special meeting (in person or by proxy) and entitled to vote on this proposal is required to approve this proposal.

If you vote to abstain, it will have the same effect as voting against this proposal.

If you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have no effect on the voting outcome of this proposal.

If a quorum is present at the special meeting:

The affirmative vote of a majority of the votes cast (in person or by proxy) by holders of MetroPCS common stock at the special meeting is required to approve this proposal.

If you vote to abstain, it will have no effect on the voting outcome of this proposal.

## Edgar Filing: MILLER PETROLEUM INC - Form 10-K

If you fail to vote or fail to instruct your bank, broker, custodian or other record holder how to vote, it will have no effect on the voting outcome of this proposal.

The MetroPCS board, acting upon the recommendation of the special committee, recommends a vote <u>FOR</u> the adjournment proposal (Item 4). For a discussion of interests of MetroPCS directors and executive officers in the transaction that may be different from, or in addition to, MetroPCS stockholders generally, see the section entitled Interests of MetroPCS Directors and Officers in the Transaction beginning on page 118.

## THE TRANSACTION

#### **Background of the Transaction**

We are a facilities-based wireless broadband mobile carrier that utilizes spectrum licensed by the FCC to provide our wireless broadband mobile services to our customers. Our FCC licenses and the spectrum we can utilize under them are integral to our ability to provide our services. As a solely facilities-based wireless broadband mobile carrier, if we do not have licensed spectrum in a geographic area, we do not sell our wireless broadband mobile services in that area. Further, as a general matter, the amount of spectrum licensed to us in an area also determines the capacity of our networks, the services we can offer and the number of customers we can serve. On average, we have 22 MHz of spectrum in the major metropolitan areas where we provide wireless broadband mobile services using our networks considerably less than our larger competitors. Because of our smaller spectrum position in the areas we serve using our networks, we are at a competitive disadvantage to our larger competitors.

Since our inception, one of our key strategies has been to expand the areas and customers we serve using our networks, which requires the acquisition of additional spectrum. Over time, we have acquired spectrum and expanded service from our first four major metropolitan areas to the thirteen major metropolitan areas we now serve using our networks. While we historically had been largely a voice and text messaging company with limited data services, starting in 2008, we realized that data services were becoming increasingly important to our existing customers and potential new customers and that such services would require additional spectrum. As a result, in 2008, we started to work with vendors to develop long-term evolution, or LTE, and, in 2009, we signed a contract to construct the first commercial LTE system in the United States, which service we launched in late 2010. Also in 2010, we launched our first relatively low cost smartphone on the Android operating system with unlimited data on our existing code division multiple access, or CDMA, 1xRTT network, and our customers quickly started to use substantial amounts of data services. These data services consumed substantially more spectrum capacity than voice and text services and confirmed that we needed to acquire additional spectrum to remain competitive.

MetroPCS management, together with the MetroPCS board, regularly monitors and assesses developments in the wireless telecommunications industry and reviews our competitive position, growth strategies and available opportunities, including, among other things, possible business combinations with other companies, as part of their ongoing efforts to implement MetroPCS business strategies, strengthen MetroPCS business, acquire additional spectrum and enhance stockholder value. The MetroPCS board undertakes this review through, among other things, presentations from management and investment bankers engaged by MetroPCS. In evaluating our competitive position in recent years, management and the MetroPCS board determined that the acquisition of additional spectrum was critical to meet increasing customer demands for data services, continue customer growth, maintain quality of service, allow the deployment of new technologies and enhance stockholder value. As a result of management s and the MetroPCS board s analysis of our competitive position, starting in 2007, the MetroPCS board agreed that MetroPCS management should explore opportunities to acquire additional spectrum both within its existing licensed areas and in new metropolitan areas as appropriate.

Since that time, MetroPCS has pursued spectrum acquisitions on numerous fronts, but we have faced significant challenges. While we have advocated for the reallocation and auctioning of additional Federal government spectrum, only a very modest amount of spectrum has been made available relative to the demand for spectrum in the market. We also lobbied for revised auction rules that would not favor large incumbent operators over relatively smaller competitors like MetroPCS, but our efforts have been largely unsuccessful. In addition, we have encountered difficulties in completing acquisitions of additional spectrum from third parties, particularly in light of MetroPCS relative size and resources compared to its larger competitors.

For example, in 2007, MetroPCS filed an application to participate in Auction 73 held by the FCC for 700 MHz spectrum. We then participated in Auction 73 in 2008, but we were only able to acquire a 6 x 6 MHz channel in the Boston area, even though we had additional funds available to spend.

In 2009, we became aware of an opportunity to acquire a satellite operator ( Company A ) with 20 MHz of broadband satellite spectrum. In 2010, we considered an acquisition of a different satellite operator ( Company B ), also with 20 MHz of broadband satellite spectrum. We engaged an investment banker, executed nondisclosure agreements with Company A and Company B and conducted due diligence with the counterparties over a number of months. During this time, the MetroPCS board and the finance committee of the MetroPCS board, which was charged with assisting MetroPCS management in evaluating strategic transactions, were involved in reviewing the two strategic alternatives, including by discussing the two strategic alternatives with MetroPCS management and our investment banker at multiple MetroPCS board and finance committee meetings. In early 2011, after MetroPCS management determined that Company A s spectrum was more attractive to MetroPCS than Company B s spectrum, the MetroPCS board authorized MetroPCS management to make a binding offer to acquire Company A. Notwithstanding the competitive bid made by MetroPCS, Company A accepted an offer from another company ( Company C ). Company A s investment banker advised us that Company C s offer was higher than our proposed purchase price. We ultimately decided not to pursue Company B because we believed Company A s and Company B s spectrum was more valuable to a single operator than if each was held separately and because Company B s spectrum was less desirable spectrum on its own. Company C then also acquired Company B. MetroPCS executives also met with Company C executives to discuss possible strategic transactions between MetroPCS and Company C, including a potential sale of spectrum, a joint venture or the contribution of Company C s satellite spectrum in exchange for equity in MetroPCS. The MetroPCS board and finance committee discussed the possible strategic transactions on multiple occasions, and the MetroPCS board authorized management to make several proposals to Company C, which MetroPCS did through a series of telephonic and in-person meetings. However, we were unable to agree with Company C on the terms of any such transaction.

During this period, in light of our inability to make a larger spectrum acquisition at Auction 73 and our failed bid for Company A, we concluded that we would need additional capital if we wanted to acquire sufficient spectrum to meet our growing customer demand. Accordingly, in early 2011, we raised approximately \$1 billion in the public debt markets for general corporate purposes, including opportunistic spectrum acquisitions, which, together with our existing resources, provided us with approximately \$1.5 billion to acquire spectrum.

On March 25, 2011, MetroPCS management met with senior executives of another wireless company (Company D) to discuss possible strategic transactions, including the acquisition of all or part of Company D by MetroPCS or the acquisition of some of Company D s spectrum. From time to time thereafter until the execution of the business combination agreement with Deutsche Telekom, we and Company D held discussions regarding potential transactions that would increase our spectrum holdings. In late 2011, pursuant to the MetroPCS board s authorization, we sent Company D s chief executive officer a sizeable non-binding written indication of interest to acquire certain spectrum assets from Company D. In response, Company D proposed alternative structures for the sale of certain of its spectrum assets on different economic terms. Thereafter, we continued to negotiate the terms of a potential transaction from time to time, including the spectrum and geographic areas to be included in such a transaction and the related economic terms, but had not reached an agreement by the time we entered into the business combination agreement with Deutsche Telekom.

During this same period, a broker contacted a senior executive of MetroPCS regarding a sale by another wireless carrier (Company E) of its excess spectrum in certain geographic areas where we did not provide service at that time and some geographic areas that overlapped existing MetroPCS service areas. In May 2011, pursuant to the authorization of the finance committee, we delivered a non-binding oral indication of interest to Company E for the acquisition of spectrum in a number of metropolitan areas. Over the next month, we negotiated for the purchase of Company E s excess spectrum and submitted additional bids. In June 2011, the broker informed us that Company E had chosen to sell its excess spectrum to one of our larger competitors.

During generally the same period, we were invited to submit a bid to acquire spectrum from another seller of spectrum, but we ultimately determined not to bid on the spectrum because we concluded that the assets were not sufficiently compatible with our existing network.

Beginning in mid-2011, we approached another wireless company (Company F) to inquire about the potential acquisition of selected spectrum and/or other assets of a wireless company that Company F had agreed to acquire. In connection with these discussions, MetroPCS, Company F and the wireless company to be acquired by Company F entered into a nondisclosure agreement which included a standstill provision that terminated in the event that we became a party to, or publicly announced, a change of control transaction with a third party. MetroPCS general practice is to include a standstill provision in our nondisclosure agreements when MetroPCS would be disclosing confidential information to a counterparty that the counterparty could use to attempt an unsolicited acquisition of MetroPCS. We then exchanged information, and principals of MetroPCS and Company F met numerous times over several months regarding a possible transaction. MetroPCS engaged Credit Suisse Securities (USA) LLC, which we refer to as Credit Suisse, which we had retained in connection with prior unrelated transactions, and another financial advisor to advise us on the possible transaction. The MetroPCS board and finance committee reviewed the possible alternative transactions on multiple occasions, and the MetroPCS board authorized MetroPCS management to make a series of non-binding proposals to Company F. In connection with these non-binding proposals, we and Company F exchanged draft term sheets outlining the principal terms of the possible transaction and negotiated the significant deal points during the fall of 2011. Nevertheless, Company F ultimately did not pursue a transaction with MetroPCS.

In light of the numerous unsuccessful attempts to secure a meaningful amount of additional spectrum and with limited prospects for the acquisition of significant additional spectrum from third parties or FCC spectrum auctions, the MetroPCS board and management periodically discussed whether a combination with another wireless telecommunications company, either as the acquirer or the target, would provide us with the scale and resources, including spectrum, to better meet our long-term strategic objectives and enhance stockholder value. The MetroPCS board and management weighed the potential benefits and risks of such a combination against the benefits and risks of remaining a stand-alone company with uncertain opportunities for acquiring additional spectrum.

In late September 2011, while we were still in discussions with Company F, another wireless company ( Company G ) contacted us regarding possible strategic transactions, including a possible acquisition of MetroPCS or commercial resale arrangement. We are a party to a roaming agreement with Company G and had previously discussed a variety of potential commercial transactions with Company G, including a resale arrangement. In September 2011, the finance committee of the MetroPCS board authorized MetroPCS management to proceed with discussions with Company G to explore a possible acquisition of MetroPCS by Company G. We and Company G then entered into a mutual nondisclosure agreement. However, we agreed that until we reached an agreement on economic terms for the acquisition, neither MetroPCS nor Company G would conduct due diligence and Company G would not be required to enter into a standstill agreement with us.

During this period, senior executives of MetroPCS met in person with senior executives of Company G in Dallas, Texas, and communicated that we were willing to discuss a potential transaction with Company G but that MetroPCS was not otherwise for sale and would not commit to engage in a more detailed due diligence process until Company G and MetroPCS agreed upon a value for MetroPCS. Senior executives of MetroPCS conducted several in-person meetings and teleconferences with senior executives of Company G over the next six weeks to discuss various matters relating to MetroPCS. During these discussions, Company G expressed a preference for an all-stock deal, while we expressed a preference for a mix of consideration consisting of stock and cash.

We began to consult with J.P. Morgan Securities LLC, which we refer to as JPMorgan, whom we had retained in connection with prior unrelated transactions, as our financial advisor regarding the potential

transaction with Company G. JPMorgan, at our direction, met with Company G s financial advisor in October 2011 to discuss matters relating to a potential combination.

During this period, MetroPCS management apprised the MetroPCS board of the discussions with Company G, and the directors were supportive of moving forward with selected preliminary due diligence to determine whether Company G would make an attractive offer. At a MetroPCS board meeting on November 10, 2011, management briefed the directors on the status of discussions with Company G, and the directors and management discussed MetroPCS continued need to acquire additional spectrum and that a combination with Company G could satisfy that need. The MetroPCS board also discussed the other strategic discussions taking place with Company F and Company D at that same time.

On November 16, 2011, senior executives of Company G and MetroPCS held an in-person meeting in Dallas, Texas, at which Company G offered to acquire all of the outstanding shares of MetroPCS common stock in exchange for cash and a fixed value of stock of Company G in a ratio of 60% stock and 40% cash equating to aggregate consideration valued at approximately \$12 per share of MetroPCS common stock based on Company G s stock price at that time.

On November 19, 2011, the MetroPCS board met to consider Company G s proposal, and, following discussion and consultation with JPMorgan, including a discussion of various financial aspects of Company G s proposal, the MetroPCS board s initial consensus was that the proposed price implied by Company G s proposal was too low at that time, did not take into account a variety of factors supporting a higher value for MetroPCS and contained a percentage of cash consideration that was insufficient given our projected cash balance at the time of a possible closing. The directors also discussed our strategic alternatives to the potential sale, including potential opportunities to acquire additional spectrum, either through private market transactions or FCC auctions, including the continuing discussions with Company F and Company D, and the opportunities and execution risks related to remaining a stand-alone company. While the directors had not made the decision to sell MetroPCS, the directors nonetheless discussed with JPMorgan appropriate ways for MetroPCS to determine whether other parties were interested in acquiring MetroPCS and might be willing to pay a higher price than Company G and the appropriate time to contact such parties if the discussions with Company G were to continue. The MetroPCS board and JPMorgan discussed several factors related to whether we should contact other parties at that time and, if so, which third parties to contact. These factors included (a) the identity of the parties viewed as most likely to be interested, (b) the perceived ability of such parties to engage in a transaction or pay a higher price for MetroPCS than Company G, (c) the risk of a leak of the discussions by the contacted parties and how the risk of a leak likely increases as a greater number of parties is contacted, (d) the likelihood that a leak might discourage Company G from continuing its discussions or increasing its proposal and (e) the likelihood that contacting other parties could create pressure on Company G to increase its offer. Following discussion with JPMorgan regarding other potential acquirers of MetroPCS and the uncertainty of meaningful other strategic or financial interested buyers surfacing at that time due to the regulatory environment and various other factors, and in light of the MetroPCS board not having made any decision to sell MetroPCS or commence a process to do so at that time, the MetroPCS board determined, after discussions with JPMorgan, to see if a deal with Company G could be firmed-up and then revisit whether to approach other potential buyers. Consequently, the MetroPCS board directed JPMorgan to meet with Company G s financial advisor to explain the MetroPCS board s perspective on the appropriate valuation for MetroPCS and provide analysis to Company G s financial advisor demonstrating that a higher price than Company G s proposed price was appropriate.

On November 21, 2011, JPMorgan met with Company G s financial advisor to discuss the economic terms of Company G s proposal and the MetroPCS board s rationale supporting a higher price.

On November 22, 2011, the MetroPCS board met again to discuss Company G s offer and to receive feedback from JPMorgan regarding its meeting with Company G s financial advisor. Following discussion and consultation with JPMorgan, the MetroPCS board directed management to deliver a counteroffer to Company G consisting of cash and Company G stock equating to aggregate consideration valued at \$15.50 per share of

MetroPCS common stock based on Company G s stock price at that time, which included a fixed exchange ratio of Company G stock and 6.20 per share in cash and would have equated to a 60/40 equity to cash ratio. MetroPCS management delivered the counteroffer to Company G.

On November 28, 2011, a senior executive of Company G contacted Mr. Roger Linquist, chief executive officer and chairman of the MetroPCS board, and proposed a purchase price of 5.00 per share in cash and an amount of Company G stock which equated to an aggregate purchase price of between 12.11 and 13.09 per share based on Company G stock price at that time.

On December 1, 2011, the MetroPCS board met to discuss Company G s new proposal. Attorneys from Akin Gump Strauss Hauer & Feld LLP, which the MetroPCS board had retained as special counsel to the MetroPCS board, were present at the meeting. Following discussion and consultation with JPMorgan, the MetroPCS board continued to believe that Company G had failed to value MetroPCS appropriately for the reasons described in the prior meetings and instructed management and JPMorgan to make another proposal to Company G for \$6.20 per share in cash with a higher exchange ratio range for the stock component of the consideration, which equated to an aggregate value of approximately \$15.13 per share based on Company G s stock price at that time. On December 5, 2011, Mr. Linquist communicated the counteroffer to a senior executive of Company G.

On December 13, 2011, a senior executive of Company G contacted Mr. Linquist and conveyed a revised offer of 5.50 per share in cash and a fixed exchange ratio of Company G stock, for an aggregate value of approximately 13.39 per share based on Company G stock price at that time. Mr. Linquist requested that Company G make its proposal to MetroPCS in writing, which the Company G senior executive agreed to do.

On December 15, 2011, the MetroPCS directors met to consider the most recent proposal from Company G, which the directors, in consultation with JPMorgan, determined was still too low. The directors again discussed with management strategic alternatives available to us, including the status of discussions with Company D and Company F at that time. After consultation with JPMorgan, the directors authorized MetroPCS management to make a counteroffer of \$5.50 per share in cash and a higher fixed exchange ratio, for an aggregate value of approximately \$13.69 per share based on Company G s stock price at that time, after it received the written offer from Company G. The MetroPCS directors also instructed MetroPCS management to commence due diligence activities in the event of acceptance of the counterproposal by the Company G board and the execution by Company G of a standstill agreement with MetroPCS.

In addition, at the December 15, 2011 MetroPCS board meeting, the MetroPCS board considered the formal engagement of JPMorgan as the MetroPCS board s financial advisor. With the assistance of Akin Gump, the MetroPCS board examined JPMorgan s potential conflicts of interest vis-à-vis Company G. The MetroPCS board approved the engagement of JPMorgan, subject to confirmation that JPMorgan did not have any disqualifying conflict of interest. In accordance with the MetroPCS board s direction, following further communications by Akin Gump and MetroPCS management with JPMorgan regarding potential conflicts, we and JPMorgan subsequently executed an engagement letter with respect to JPMorgan s services (including JPMorgan s services previously described). The engagement letter with JPMorgan provided for JPMorgan to receive total compensation payable at closing based on a percentage of the consideration received by the MetroPCS stockholders in any transaction and fixed cash fees payable upon the announcement of a transaction and rendering of an opinion, if any, with respect thereto, in each case credited against the total compensation.

On December 21, 2011, Deutsche Telekom contacted Credit Suisse, which had represented MetroPCS in connection with a previous transaction in which Deutsche Telekom had been involved, and requested that Credit Suisse convey to MetroPCS that Deutsche Telekom was interested in meeting with MetroPCS to discuss potential strategic opportunities involving Deutsche Telekom s subsidiary T-Mobile and MetroPCS. Later that day, Credit Suisse conveyed Deutsche Telekom s interest to a senior executive of MetroPCS. We thereafter

began to consult with Credit Suisse as a MetroPCS financial advisor regarding a potential transaction with Deutsche Telekom and, in January 2012, regarding the potential transaction with Company G.

MetroPCS management discussed the communication from Deutsche Telekom with the members of the finance committee of the MetroPCS board, who encouraged MetroPCS management to meet with the Deutsche Telekom representatives to obtain more information regarding Deutsche Telekom s interest.

On December 22, 2011, in response to Mr. Linquist s December 1<sup>th</sup> request that Company G submit its proposal in writing, a senior executive of Company G provided a written summary of Company G s proposal of \$5.50 per share in cash and the fixed exchange ratio of Company G stock, for an aggregate value of approximately \$13.39 per share based on Company G s stock price at that time. The next day, pursuant to the MetroPCS board s prior authorization, Mr. Linquist contacted the senior executive of Company G by telephone and conveyed the MetroPCS board s counteroffer of \$5.50 per share in cash plus Company G stock at a higher exchange ratio, for an aggregate value of approximately \$13.69 per share based on Company G s stock price at that time.

On January 3, 2012, the senior executive of Company G and Mr. Linquist had a conversation in which Mr. Linquist indicated that any further progress on the transaction would be subject to reaching agreement on the economic terms of the transaction and confirmation that Company G s board was supportive of the transaction.

In anticipation that we and Company G would reach agreement on the economic terms and the Company G board would confirm its support of the transaction, over the next several days, we and Company G negotiated an amendment to our mutual nondisclosure agreement that added a standstill provision that would terminate upon MetroPCS becoming a party to, or publicly announcing, a change of control transaction with another party, with the understanding that such amendment would not be executed unless and until the parties reached agreement on the economic terms.

On January 6, 2012, Mr. Linquist talked by telephone with the chief executive officer of Company G regarding the potential transaction and Company G s internal approval process for the potential transaction.

On January 13, 2012, Mr. Linquist, together with Mr. J. Braxton Carter, chief financial officer of MetroPCS, and Mr. Thomas Keys, president and chief operating officer of MetroPCS, met with senior executives of Deutsche Telekom in Germany to discuss potential strategic opportunities, including a potential combination of MetroPCS and T-Mobile. At that meeting, the MetroPCS senior executives discussed our business, management team, and capital structure, including our cash-on-hand. The MetroPCS senior executives expressed an interest in combining MetroPCS with T-Mobile in a transaction that would permit MetroPCS to control the combined company. MetroPCS noted the strength of its management team, advocating that if MetroPCS controlled a combined MetroPCS/T-Mobile enterprise, we would be able to generate improved returns for Deutsche Telecom s investment. Following the meeting, Mr. Linquist shared the content of the discussions with Deutsche Telekom with the members of the finance committee of the MetroPCS board.

On January 16, 2012, Mr. Linquist met in-person with the chief executive officer of Company G in Dallas, Texas, to discuss the potential transaction between the parties.

On January 19, 2012, we sent a draft mutual nondisclosure agreement between MetroPCS and T-Mobile to Deutsche Telekom which included a standstill provision that would terminate upon our becoming a party to, or publicly announcing, a change of control transaction with another party. The parties executed the agreement on February 3, 2012.

On January 19, 2012, a senior executive of Company G informed Mr. Linquist that Company G had agreed to move forward with due diligence and to pursue the proposed acquisition of MetroPCS on the basis of our last proposal. The next day, Mr. Linquist reported the conversation to the MetroPCS board, and the MetroPCS board

approved moving forward with due diligence and the negotiation of definitive agreements, subject to confirmation that the board of Company G had approved the economic terms of our most recent proposal. On January 20, 2012, Company G sent us a draft of an agreement and plan of merger. On January 21, 2012, Mr. Linquist sought assurances from Company G that its board had approved in principle the transaction on the economic terms of our most recent proposal. On January 23, 2012, a senior executive of Company G confirmed to Mr. Linquist that its board had agreed to move forward with the transaction on our economic terms provided that Company G could conclude that the transaction would be accretive and that integration would not be materially problematic. On that basis, we began to engage in due diligence and the negotiation of definitive agreements with Company G.

On January 25, 2012, Deutsche Telekom communicated to MetroPCS that it would like to begin due diligence immediately. At the special meeting of the MetroPCS board on January 26, 2012, Mr. Linquist described Deutsche Telekom s interest in commencing due diligence and making a proposal to MetroPCS. The MetroPCS board requested that Mr. Linquist communicate to Deutsche Telekom that, if it planned to make a proposal, it should submit it quickly.

On January 26, 2012, Gibson Dunn & Crutcher, LLP, which we refer to as Gibson Dunn, counsel to MetroPCS, provided comments on Company G s draft agreement and plan of merger to Company G s counsel. Over the next several weeks, we and Company G and our respective counsel negotiated the terms of definitive transaction agreements and continued our due diligence.

On January 31, 2012, Mr. Linquist contacted Mr. René Obermann, the chief executive officer of Deutsche Telekom, and informed him that, due to timing constraints relating to other strategic transactions being considered by MetroPCS, if Deutsche Telekom planned to make a proposal, we would need to receive such proposal from Deutsche Telekom by no later than mid-February. The next day, Mr. Thorsten Langheim, senior vice president of M&A of Deutsche Telekom telephoned Mr. Carter and informed him that Deutsche Telekom s internal governance processes would make delivery of a proposal to us by mid-February extremely difficult.

The MetroPCS board met on February 7, 2012. At this meeting, MetroPCS management updated the MetroPCS directors on the final terms of the engagement of JPMorgan and recommended that the MetroPCS board also approve the engagement of Credit Suisse as a second financial advisor. After consulting with Akin Gump, special counsel to the MetroPCS board, the MetroPCS board considered the benefits of engaging a second financial advisor in connection with the proposed transaction with Company G. The MetroPCS board approved the engagement of Credit Suisse in connection with the proposed transaction with Company G, subject to confirmation that Credit Suisse did not have any disqualifying conflict of interest relating to a potential transaction with Company G. In accordance with the MetroPCS board s direction, following further communications by Akin Gump and MetroPCS management with Credit Suisse regarding potential conflicts, we entered into an engagement letter with Credit Suisse provided for Credit Suisse to receive a fixed amount of cash compensation for delivery of a fairness opinion if requested by the MetroPCS board and a separate fixed amount of cash compensation upon closing of a transaction. At this meeting, MetroPCS management also reported to the MetroPCS board regarding the ongoing negotiations with Company G and the discussions with Deutsche Telekom. In addition, during the meeting, Akin Gump reviewed for the MetroPCS directors their fiduciary duties in the context of the potential transaction with Company G.

On February 12, 2012, at the request of Deutsche Telekom, Messrs. Linquist and Carter met with Messrs. Obermann and Langheim in Dallas, Texas, and discussed a potential combination of MetroPCS and T-Mobile. Messrs. Obermann and Langheim conveyed Deutsche Telekom s preliminary view that Deutsche Telekom should own between 72% and 75% of the equity in a combined MetroPCS/T-Mobile company, with MetroPCS stockholders owning the remainder. Deutsche Telekom s proposal also contemplated that the combined company would be capitalized with \$1.4 billion of cash and \$18.4 billion of debt, consisting of

\$8.0 billion of new third party debt and \$10.4 billion of intercompany debt, and that the combined company would pay a \$500 million dividend each year. Deutsche Telekom also explained that under its proposal it would receive a cash payment at closing of approximately \$4.3 billion, which was important to Deutsche Telekom because it provided Deutsche Telekom a partial liquidity event with respect to its investment in T-Mobile, and the MetroPCS stockholders would receive no cash at closing. They also discussed the potential merits of combining Company E with the combined MetroPCS/T-Mobile business. On February 13, 2012, Mr. Linquist communicated a summary of the Deutsche Telekom proposal to the MetroPCS board.

On February 17, 2012, the MetroPCS board held a meeting at which the timing of a possible transaction with Company G and the proposal by Deutsche Telekom were discussed. In addition, the directors were updated on the status of the negotiations and open issues related to the Company G merger agreement and related legal documents. JPMorgan and Credit Suisse also updated the directors with respect to the financial aspects of the Company G transaction.

During this period, MetroPCS, Company G and our respective counsel completed negotiations of the definitive transaction agreements and our mutual due diligence investigations. At this time, we also agreed that the Company G board would vote on the transaction on February 22, 2012, and the MetroPCS board would vote on the transaction on February 23, 2012.

On February 22, 2012, the MetroPCS board and the Company G board each held separate meetings to consider the final terms of the transaction between MetroPCS and Company G. At the MetroPCS board meeting, the MetroPCS directors reviewed with JPMorgan and Credit Suisse the terms of the Company G transaction. The directors also discussed with Credit Suisse the terms of the transaction proposed by Deutsche Telekom, and the MetroPCS board and its advisors discussed various related matters, including the conditionality and uncertainties of Deutsche Telekom s proposal, the dilutive nature of Deutsche Telekom s proposal versus the accretive nature of Company G s proposal, the synergies and execution risks associated with each proposal, the large upfront amount of financing needed under the Deutsche Telekom proposal, the fact that the merger agreement with Company G had been fully negotiated and the potential risk of losing the transaction with Company G in the event of a delay, as well as the provisions in the merger agreement with Company G that allowed MetroPCS board determined that the interests of the MetroPCS stockholders would not be served by delaying execution of the merger agreement with Company G in order to pursue discussions with Deutsche Telekom. Akin Gump also reviewed for the directors their fiduciary duties relative to considering the two alternative transactions. The MetroPCS board then adjourned until February 23, 2012, without taking any action to approve the proposed transaction with Company G, pending receipt of information from Company G that its board had approved the transaction. Later on February 22, 2012, a senior executive of Company G called Mr. Carter to inform him that the board of Company G had determined not to proceed with an acquisition of MetroPCS. Management of MetroPCS then informed the MetroPCS directors of Company G s decision not to proceed with the proposed transaction.

The next day, the MetroPCS directors held a meeting to discuss Company G s decision not to proceed. The directors revisited their analysis of our strategic alternatives, including our viability as a stand-alone company and prospects for obtaining spectrum other than through a business combination transaction. The consensus of the directors remained that acquiring additional spectrum was essential to being a viable competitor in the industry and enhancing stockholder value, but we were limited in our opportunities to acquire additional spectrum. Acknowledging the challenges in acquiring meaningful additional spectrum, the directors authorized MetroPCS management to explore further whether a combination of MetroPCS and T-Mobile could be accomplished on attractive terms. In accordance with this authorization, MetroPCS management worked with JPMorgan and Credit Suisse for several weeks to determine a potential counterproposal to convey to Deutsche Telekom.

Shortly after Company G s decision not to proceed, various news outlets reported that MetroPCS and Company G had been in advanced discussions for a transaction in which Company G was to have acquired MetroPCS, and MetroPCS had been agreeable to such a transaction, but Company G had determined not to proceed. We did not publicly acknowledge the information being reported in the various news outlets.

On March 2, 2012, the MetroPCS directors met to discuss a proposed counterproposal to Deutsche Telekom. The directors discussed a counterproposal that would have MetroPCS acquiring T-Mobile, with Deutsche Telekom receiving a 49% ownership interest in the combined company, \$10 billion in cash and \$7 billion in debt securities of the combined company, while stockholders of MetroPCS would receive no cash but instead would, in the aggregate, own 51% of the equity of the combined company. Retention by MetroPCS stockholders of a majority of the voting interest in MetroPCS was beneficial to MetroPCS stockholders and Deutsche Telekom because it would enable our existing debt to remain outstanding after the transaction because the change of control provisions under the agreements governing that debt would not be triggered and our low cost debt would not have to be refinanced. The proposal also contemplated that Deutsche Telekom might be permitted to sell down a portion of its ownership interest in the combined company at closing or shortly thereafter. MetroPCS management, JPMorgan and Credit Suisse discussed with the directors that this counterproposal was intended to convert the transaction contemplated by Deutsche Telekom s initial proposal, which envisioned Deutsche Telekom owning between 72% and 75% of the combined company and receiving an approximately \$4.3 billion cash payment and \$10.4 billion of combined company debt securities at closing, into a transaction whereby we would acquire T-Mobile and Deutsche Telekom would receive a larger cash payment and hold a reduced amount of intercompany debt from the combined company at closing, which would provide Deutsche Telekom with liquidity with respect to its T-Mobile investment. MetroPCS management and the MetroPCS board s financial advisors discussed with the MetroPCS board that the counterproposal would allow the combined company to take advantage of both (a) our capital structure by retaining our existing debt outstanding after closing because there would be no change of control and (b) additional third party debt capacity that JPMorgan and Credit Suisse believed would be available to the combined company to fund the larger cash payment. In connection with this discussion, the MetroPCS board requested more information from MetroPCS management on expected synergies from the transaction and projections for T-Mobile s business. The directors also emphasized that any future communication with Deutsche Telekom should make clear that MetroPCS was not for sale and was simply exploring whether a potential combination might be feasible on acceptable terms.

The MetroPCS directors met again on March 5, 2012, and MetroPCS management reported on T-Mobile s projections and its initial estimate of the expected synergies from a MetroPCS/T-Mobile combination. MetroPCS management preliminarily estimated that potential synergies could range between \$5.7 billion and \$8.7 billion, including as a result of MetroPCS not spending \$1.5 billion of its cash-on-hand that had been earmarked in the 2012 budget for spectrum acquisitions. The directors discussed various aspects of the synergies, including the time period required to realize the synergies and the likelihood of the synergies being realized. Following discussion and based on the projected synergies and projections, the directors authorized MetroPCS management to provide a counterproposal to Deutsche Telekom as previously discussed in the March 2, 2012 MetroPCS board meeting. On March 5, 2012, Mr. Linquist sent a letter to Mr. Obermann outlining MetroPCS proposal.

On March 9, 2012, Messrs. Linquist and Carter met with Mr. Obermann and other senior executives of Deutsche Telekom in Bonn, Germany, to discuss MetroPCS proposal. At the meeting, Deutsche Telekom discussed with the MetroPCS senior executives that Deutsche Telekom was seeking to sell certain of T-Mobile s tower assets. Deutsche Telekom proposed a counteroffer pursuant to which Deutsche Telekom would receive 63% of the ownership interest in the combined company assuming T-Mobile executed a transaction to sell its tower assets and Deutsche Telekom received the proceeds. MetroPCS proposed that Deutsche Telekom s voting interest in the combined company would be 49%, with the MetroPCS stockholders holding the remaining 51% voting interest, which Deutsche Telekom said it would consider. Further, the parties discussed that Deutsche Telekom could reduce its economic ownership interest in the combined company to 49% at a later stage. In addition, Deutsche Telekom proposed that (a) the combined company would commit to making a \$300 million annual distribution to stockholders, (b) a majority of the combined company s board would be appointed by

MetroPCS stockholders other than Deutsche Telekom, (c) Deutsche Telekom would receive \$10 billion in cash at closing financed by fully committed third party debt of the combined company and (d) the combined company would issue \$7 billion in debt securities to Deutsche Telekom at closing on terms similar to the third party debt, to be refinanced by the combined company within three years after closing. Over the next several days, MetroPCS management discussed the Deutsche Telekom counteroffer with JPMorgan and Credit Suisse.

Also on March 9, 2012, a senior executive of Company F contacted Mr. Mark Stachiw, MetroPCS general counsel, and, referencing recent press reports regarding MetroPCS and Company G, inquired whether MetroPCS was for sale. Mr. Stachiw declined to acknowledge the accuracy of the press reports and stated that MetroPCS was not for sale at that time but that the MetroPCS board was always open to evaluating strategic alternatives. Later in March 2012, Mr. Linquist spoke with the chief executive officer of Company F, who conveyed that Company F was not interested in an acquisition of MetroPCS at that time.

On March 20, 2012, the MetroPCS directors held a special meeting to discuss the Deutsche Telekom counteroffer. After discussion and consultation with JPMorgan and Credit Suisse, the directors requested that MetroPCS management and the MetroPCS board s financial advisors discuss with them at a subsequent meeting additional financial aspects of the counteroffer to assist in their evaluation. The directors also examined the alternative of remaining a stand-alone company with the attendant prospects for obtaining additional spectrum and the relative merits and risks of remaining a stand-alone company versus engaging in a business combination transaction. In addition, Mr. Stachiw and Mr. Linquist reported on their respective discussions with representatives of Company F.

On March 21, 2012, Messrs. Linquist and Stachiw and the chairman of the board and another senior executive from Company C had a dinner meeting at the invitation of Company C. The Company C executives inquired whether the press reports regarding MetroPCS and Company G were true, whether MetroPCS was for sale and whether a potential transaction between Company C and MetroPCS would be possible on mutually acceptable terms. The senior executives of MetroPCS did not deny the press reports and indicated that MetroPCS was not for sale at that time but that we were in the process of evaluating strategic alternatives and that if Company C had a strategic alternative it wanted to propose, the senior executives of MetroPCS would bring it to the MetroPCS board. The chairman of the board of Company C conveyed to the MetroPCS senior executives that Company C would contact MetroPCS if Company C wanted to pursue the discussions further or propose a strategic alternative for the MetroPCS board to consider.

Beginning in March 2012 and extending through the summer of 2012, Mr. Linquist and other senior executives of MetroPCS engaged in conversations with the chief executive officer and other senior executives of another wireless company (Company H) regarding advanced wireless services, or AWS, and 700 MHz spectrum that Company H might be required to divest in connection with Company H s transaction with another company. We indicated to Company H that we were interested in purchasing such spectrum and that we had significant liquidity to do so, although we never extended a formal offer. Company H did not demonstrate any particular interest in selling the AWS spectrum to us. Ultimately Company H entered into an agreement to sell the spectrum to another company.

As a follow-up to the March 21, 2012 meeting, on March 23, 2012, the MetroPCS directors held another special meeting to discuss the Deutsche Telekom counteroffer. The directors discussed various financial aspects of the counteroffer with the MetroPCS board s financial advisors and the proposed economics of the transaction under both a scenario giving full effect to the estimated synergies and scenarios giving a reduced or no effect to the estimated synergies. After discussing various aspects of the valuation of the two companies, the directors authorized MetroPCS management to continue negotiations with Deutsche Telekom for a business combination pursuant to which (a) MetroPCS stockholders would retain between 40% and 45% of the ownership interest in the combined company, with the remainder of the ownership interest being held by Deutsche Telekom, (b) although Deutsche Telekom s equity interest would be over 50%, Deutsche Telekom s voting interest would not exceed 49% of the combined company and (c) Deutsche Telekom would not have the right to appoint a

majority of the board members of the combined company. The directors and their advisors discussed that this structure would permit our existing low cost debt to remain in place after closing because no change of control provisions in the agreements governing the debt would be triggered, which would reduce the amount of debt to be financed at closing. The directors indicated they were generally in agreement with the other deal points previously proposed by Deutsche Telekom on March 9, 2012, but disagreed that the \$7 billion in debt securities to be issued to Deutsche Telekom should be required to be refinanced in three years. In addition, the directors agreed that the proceeds from the tower assets sale would be retained by Deutsche Telekom, subject to our review of and consent to the terms of the leases of the tower assets back to T-Mobile. In connection with the discussion of the potential transaction with Deutsche Telekom, the directors engaged in additional discussion regarding our strategic alternatives and the challenges facing us if we were to remain a stand-alone company, such as our difficulty in acquiring additional spectrum and competing against larger competitors with greater resources. In addition, Mr. Linquist reported on his contacts with Company C and Company H.

During the MetroPCS board meeting, the independent MetroPCS board members, comprising all directors other than Mr. Linquist, met in executive session. These directors discussed whether a special committee of the MetroPCS board should be formed to ensure that the negotiation process with Deutsche Telekom was conducted in the best interests of MetroPCS and the MetroPCS stockholders in light of the possibility that Mr. Linquist, a MetroPCS board member, and some members of MetroPCS management team could remain in senior positions with the combined company if an acquisition by MetroPCS of T-Mobile on the terms proposed by MetroPCS were to occur. Following the executive session, Mr. Linquist and other MetroPCS senior executives rejoined the board meeting, and Mr. James N. Perry, Jr. summarized the deliberations of the independent directors. He reported that, though the directors believed that the negotiations had been and would be conducted in the future in a manner favorable to MetroPCS and the MetroPCS stockholders, the directors determined that, in keeping with best corporate governance practices, they should establish a special committee consisting of all of the independent MetroPCS directors, with Mr. Perry being the chairman of the special committee, and that the establishment of such a special committee was in the best interests of MetroPCS and the MetroPCS stockholders. Consequently, the full MetroPCS board adopted resolutions establishing the special committee and granting the special committee the authority to, among other things, (a) evaluate, consider and respond to the potential transaction with Deutsche Telekom and any alternatives to such transaction, (b) monitor, direct and participate in the negotiations, (c) make or accept, reject or seek to modify the terms and conditions of the potential transaction with Deutsche Telekom or any alternatives to such transaction, (d) recommend to the full MetroPCS board whether the MetroPCS board should approve the potential transaction with Deutsche Telekom or any alternatives to such transaction and (e) engage, at our expense, the committee s own legal counsel and financial advisors.

Over the next several days, management of MetroPCS, JPMorgan and Credit Suisse discussed a potential counterproposal to Deutsche Telekom. On March 29, 2012, Messrs. Linquist and Carter met in person with Mr. Obermann and other senior executives of Deutsche Telekom to discuss terms and conditions for the potential transaction and proposed that Deutsche Telekom receive a 55% ownership interest in the combined company and MetroPCS stockholders retain a 45% ownership interest. In response, Mr. Obermann conveyed that Deutsche Telekom would accept no less than a 60% ownership interest in the combined company, but Deutsche Telekom would consider limiting Deutsche Telekom s voting interest to 49% of the voting interests in the combined company. However, MetroPCS and Deutsche Telekom had different expectations regarding the extent of Deutsche Telekom s consent rights with respect to actions of the combined company. Deutsche Telekom still contemplated that it would receive \$10 billion in cash at closing financed by fully committed third party debt of the combined company, the combined company would issue \$7 billion in debt securities to Deutsche Telekom at closing and Deutsche Telekom would retain all proceeds from the sale of T-Mobile s tower assets. The parties did not agree on the issue of whether the \$7 billion in debt securities issued to Deutsche Telekom would be required to be refinanced within three years after closing.

In addition, during this period, Mr. Perry, on behalf of the special committee, discussed the appointment of legal counsel and financial advisors for the special committee. After consultation with the other members of the special committee, Mr. Perry, on behalf of the special committee, engaged Akin Gump as the special

committee s counsel based in part on that firm s prior role as independent legal counsel to the board during the discussions with Company G. In addition, Mr. Perry, on behalf of the special committee, engaged Evercore as the special committee s independent financial advisor based in part on Evercore s familiarity with the telecommunications industry in general, and with MetroPCS in particular. On behalf of the special committee, Akin Gump negotiated the terms of Evercore s engagement agreement. For a description of Evercore s compensation, see the section entitled The Transaction Opinion of the Financial Advisor to the MetroPCS Special Committee beginning on page 91.

On April 3, 2012, the MetroPCS directors met to discuss Deutsche Telekom s latest position on ownership in the combined company. Representatives of JPMorgan, Credit Suisse and Evercore summarized the terms of Deutsche Telekom s latest proposal, and the directors discussed the proposal with the MetroPCS board s and the special committee s financial advisors, including the proposed implied ownership ratio of the combined company based on various financial metrics. The directors also discussed other aspects of the proposed transaction, including the potential means to limit Deutsche Telekom s voting power to less than 50%, possible approaches to reduce Deutsche Telekom s ownership interest to less than 50% as soon as possible after closing, the terms of the debt securities to be issued to Deutsche Telekom and the combined company s annual dividend policy. In addition, the directors discussed the synergies that were likely to be created through the combination and the value for MetroPCS stockholders attributable to such synergies.

The special committee then met separately with its advisors to discuss the potential transaction. When the full MetroPCS board meeting reconvened, the special committee authorized management to proceed with the due diligence phase of the transaction on the basis of Deutsche Telekom s proposed terms of the 60/40 ownership ratio, \$10 billion of cash consideration to be paid to Deutsche Telekom funded by new third party debt of the combined company, \$7 billion of debt to be issued by the combined company to Deutsche Telekom, the limitation of Deutsche Telekom s voting interest to 49% of the combined company and Deutsche Telekom s inability to appoint a majority of the directors of the combined company. The special committee also instructed MetroPCS management to keep the special committee apprised of all developments in the potential transaction.

On April 5, 2012, Mr. Linquist discussed with Mr. Obermann the terms on which we were prepared to move forward with due diligence. Over the next two weeks, we exchanged due diligence request lists with Deutsche Telekom and worked on regulatory issues relating to the exchange of information.

On April 9, 2012, and through April 18, 2012, we sent drafts of the proposed transaction agreements prepared by Gibson Dunn to the MetroPCS special committee s legal and financial advisors for their input.

On April 18, 2012, Mr. Carter shared our first quarter financial results and initial revised long range forecast based on our first quarter financial results with senior executives of Deutsche Telekom. First quarter financial results included declines in net income of 63%, income from operations of 32% and adjusted EBITDA of 8% relative to the first quarter in the prior year, and the revised long range forecast showed markedly lower projected results. In addition, earnings per share for the first quarter dropped to \$0.06 per share from \$0.25 per share for the fourth quarter of 2011, a 76% decline. The first quarter financial results were significantly below analysts expectations.

After taking into account the input of the special committee s advisors and JPMorgan and Credit Suisse, on April 19, 2012, we sent to Deutsche Telekom a proposed draft of a stock purchase agreement for the transaction and a term sheet setting forth other aspects of the proposed transaction.

We publicly announced our first quarter financial results on April 26, 2012, and the MetroPCS stock price dropped by approximately 16% over the next ten days.

On April 27, 2012, Mr. Langheim informed Mr. Carter that, due to our lower than anticipated first quarter results and the revised long range forecast, Deutsche Telekom was no longer interested in pursuing a transaction

on the economic terms previously discussed. Mr. Langheim also conveyed that Deutsche Telekom expected to retain consent rights with respect to certain actions of the combined company even if its voting interest in the combined company was less than 50% and that, if MetroPCS could not agree to grant these consent rights, Deutsche Telekom would have difficulty moving forward with a transaction. Mr. Carter shared Deutsche Telekom s reaction to our financial results with Mr. Linquist, who later that day contacted Mr. Obermann to request an in-person meeting to discuss the transaction in light of the first quarter results and revised long range forecast. On April 30, 2012, Messrs. Langheim and Carter talked again and agreed that the due diligence process should be put on hold pending further discussion and agreement between the parties on the economic terms of a potential transaction.

Over the next two weeks, MetroPCS management refined its long range forecast, and we exchanged financial information with Deutsche Telekom in an attempt to arrive at revised economic terms for a combination and discussed how the two companies should be valued. During these discussions, the proposed transaction changed substantially. Deutsche Telekom proposed that the ownership ratio in the combined company be adjusted significantly to 76% for Deutsche Telekom and 24% for the MetroPCS stockholders, with the MetroPCS stockholders to receive an aggregate cash payment of \$1.5 billion at the closing of the transaction, but that Deutsche Telekom would no longer be limited to a voting interest of 49%, Deutsche Telekom would be able to appoint a majority of the board of the combined company and Deutsche Telekom would determine the post-closing management composition of the combined company. As a result, according to the Deutsche Telekom proposal, the transaction would no longer be structured so that MetroPCS stockholders would control the combined company, and the transaction would trigger change of control provisions in certain of our existing debt agreements.

On May 16, 2012, Messrs. Linquist and Carter met with Mr. Obermann and other Deutsche Telekom senior executives in Boston, Massachusetts, to discuss the revised nature and terms of a potential transaction. At the meeting, Mr. Obermann referenced the approximately 30% decline in our stock price since Mr. Obermann s March 29, 2012 meeting with Messrs. Linquist and Carter and noted that T-Mobile had outperformed its budget during such period. Mr. Obermann communicated that Deutsche Telekom, in light of these changed circumstances, was revising its proposal to provide that MetroPCS stockholders would retain a 22.5% ownership stake in the combined company and receive \$1.5 billion in cash at the closing, Deutsche Telekom would receive economic and voting shares representing the remaining 77.5% interest in the combined company with no obligation to sell its shares, and Deutsche Telekom would appoint seven of nine directors to the board of the combined company (with the tenth director being the chief executive officer of the combined company) and have certain consent rights with respect to actions to be taken by the combined company. In addition, Mr. Obermann stated that Deutsche Telekom still desired that Mr. Linquist be named chairman and chief executive officer of the combined company and Mr. Carter be named chief financial officer of the combined company, and that the other senior leadership roles for the combined company be filled with the best alternatives from the ranks of the senior management of MetroPCS and T-Mobile. Mr. Obermann also indicated that, in light of our first quarter results and the downward adjustment to our long range forecast, Deutsche Telekom had determined to value MetroPCS based on analyst estimates of our 2012 financial performance rather than our internally generated forecast for 2012. The MetroPCS senior executives reacted negatively to the Deutsche Telekom economic proposal, and after considerable discussion at the meeting, Mr. Obermann revised his proposal and proposed that Deutsche Telekom would be willing to move forward with a deal that included a \$2 billion aggregate cash payment to MetroPCS stockholders at closing and MetroPCS stockholders retaining a 20% ownership stake in the combined company. Mr. Obermann also noted that Deutsche Telekom s proposal was conditioned upon Deutsche Telekom s obtaining third party refinancing of approximately \$15 billion of intercompany debt owed by T-Mobile to Deutsche Telekom to facilitate a liquidity event for Deutsche Telekom and that Deutsche Telekom would undertake such refinancing, including agreeing to the terms of the new debt, rather than MetroPCS. Finally, Mr. Obermann stated that Deutsche Telekom would not make any additional changes to its proposal and was willing to move forward only if we agreed to Deutsche Telekom s proposal in its entirety.

On May 17, 2012, senior executives of MetroPCS met with senior executives of Company E at Company E s request and discussed whether the two companies should explore a strategic transaction, including a combination of their businesses. At the meeting, the senior executives discussed whether any discussions between MetroPCS and Company E regarding a potential combination of their operations would be worthwhile, and it was the consensus of the senior executives that such discussions would not be productive at that time.

On May 22, 2012, the MetroPCS directors discussed the updated proposal from Deutsche Telekom with JPMorgan, Credit Suisse and Evercore and the various benefits and risks of proceeding with a transaction with Deutsche Telekom versus pursuing other strategic alternatives or remaining a stand-alone company and each alternative s impact on stockholder value. The directors also discussed whether the Deutsche Telekom proposal truly represented Deutsche Telekom s final proposal as communicated by Mr. Obermann or whether it would be worthwhile for MetroPCS to make a counteroffer. After considerable discussion, the directors concluded that a counteroffer should be made to Deutsche Telekom on terms consisting of a 22.5% ownership stake for MetroPCS stockholders in the combined company and a \$2 billion cash payment to MetroPCS stockholders at closing. Mr. Linquist reported to the directors regarding the conversations with Company E, and the consensus of the directors was that further discussions with Company E were unlikely to be productive and could potentially complicate any potential transaction with Deutsche Telekom.

On May 25, 2012, Mr. Linquist spoke by telephone with Mr. Obermann and delivered the counteroffer approved by the MetroPCS board of 22.5% ownership of MetroPCS stockholders in the combined company plus the \$2 billion aggregate cash payment to MetroPCS stockholders.

On May 30, 2012, Messrs. Linquist and Obermann spoke by telephone, and Mr. Obermann conveyed that while Deutsche Telekom had some flexibility and could perhaps move to a 20.5% ownership interest for MetroPCS stockholders, Deutsche Telekom was unlikely to accept our counteroffer of 22.5%.

During this time, Messrs. Carter and Langheim had multiple conversations, and the parties considered presentations from several investment banking firms, regarding whether Deutsche Telekom could secure the commitments for third-party refinancing of the intercompany debt contemplated by its proposal and the terms of such financing and related commitments.

The MetroPCS board held its regular quarterly meeting on May 30, 2012, during which, among other things, Mr. Linquist updated the directors on the latest conversations with Deutsche Telekom. Over the next several weeks, senior executives and financial advisors of the MetroPCS board and Deutsche Telekom discussed the financing commitments that would be required to refinance the Deutsche Telekom/T-Mobile intercompany debt and the challenges presented by attempting to secure such financing.

On June 5, 2012, Wachtell, Lipton, Rosen & Katz, which we refer to as Wachtell Lipton, counsel to Deutsche Telekom, distributed to us and Gibson Dunn a draft business combination agreement and a term sheet outlining various aspects of proposed governance for the combined company reflecting Deutsche Telekom s proposal, which we provided to the special committee s advisors. The term sheet contemplated a governance structure for the combined company pursuant to which, among other things, Deutsche Telekom would have consent rights with respect to a number of specified actions by the combined company. On June 12, 2012, Mr. Stachiw and attorneys from Gibson Dunn and Wachtell Lipton discussed by telephone the legal issues presented in the draft business combination agreement. On June 14, 2012, Gibson Dunn distributed a revised draft of the business combination agreement to Deutsche Telekom.

On June 18, 2012, Messrs. Carter and Langheim met in Athens, Greece, and discussed the economic terms of the potential transaction further. Mr. Langheim indicated to Mr. Carter that, in a fundamental change from its prior proposal given Deutsche Telekom s previously stated interest in experiencing a liquidity event with respect to its T-Mobile investment, Deutsche Telekom was exploring the possibility that third party refinancing of T-Mobile s intercompany debt to Deutsche Telekom would not be a condition to closing the transaction due to

unfavorable conditions in the credit markets; instead, Deutsche Telekom would convert all of its existing intercompany debt into new debt securities of the combined company and would receive no cash payment at closing. Messrs. Carter and Langheim discussed possible revised economic terms of the transaction in light of the refinancing issue, including that possibly MetroPCS stockholders could retain a 22.5% ownership interest in the combined company and receive a \$1.5 billion aggregate cash payment.

In June 2012, as part of our ongoing and occasional discussions with Company D regarding potential transactions, we participated in discussions with Company D and a substantial investor in Company D regarding potential strategic alternatives that would allow us to acquire spectrum from Company D or for Company D to be split between its investors and MetroPCS. These discussions continued thereafter from time to time but did not lead to an agreement by the time we executed the business combination agreement with Deutsche Telekom.

On June 19, 2012, Gibson Dunn discussed governance matters pertaining to the combined company with Wachtell Lipton and distributed to Wachtell Lipton a list of governance principles for use in the negotiations.

On June 20, 2012, a representative of Credit Suisse discussed with Mr. Langheim the rollover of the intercompany debt discussed by Messrs. Langheim and Carter in Athens. Mr. Langheim indicated that the financing matters he had discussed with Mr. Carter, which he said significantly changed the economics of the transaction for Deutsche Telekom because it would no longer receive a large cash payment at closing, were still under consideration by Deutsche Telekom.

On June 21, 2012, the MetroPCS directors held a special meeting to discuss the most recent developments in the negotiations with Deutsche Telekom. The MetroPCS directors and senior executives, together with the financial advisors for the MetroPCS board and the special committee, discussed whether the 22.5% / \$1.5 billion structure discussed between Messrs. Carter and Langheim constituted a firm proposal from Deutsche Telekom in light of the refinancing issue and, if so, whether the proposal represented a move backward in economic value for MetroPCS stockholders because, even though the ownership percentage had increased, the amount of the cash payment to MetroPCS stockholders had been reduced. The directors concluded that it was important to determine quickly whether the parties could resolve the open economic issues, particularly given Deutsche Telekom s statements regarding its concerns about the inability to refinance the intercompany debt, and whether the recent discussions between Messrs. Langheim and Carter represented a counterproposal from Deutsche Telekom. Mr. Linquist agreed to contact Mr. Obermann and encourage him to agree to an in-person meeting where the parties could resolve the economic terms.

On June 22, 2012, Mr. Linquist contacted Mr. Obermann and requested an in-person meeting to attempt to reach agreement on the economic terms of the proposed combination. Mr. Obermann responded that he was agreeable to a meeting. Mr. Obermann also indicated that, as a result of unfavorable conditions in the credit markets, the structure for the transaction previously contemplated, which provided that a significant amount of the Deutsche Telekom/T-Mobile intercompany debt was to be refinanced with third parties and Deutsche Telekom was to receive a substantial cash payment at closing, was no longer feasible and that the parties would need to work together to find an alternative structure that would permit Deutsche Telekom to experience a substantial liquidity event in connection with the transaction. The parties agreed to meet in person in Germany on July 9, 2012, to discuss further.

On June 25, 2012, Mr. Stachiw and attorneys from Gibson Dunn met with Mr. Axel Lützner, in-house counsel at Deutsche Telekom, and attorneys from Wachtell Lipton in New York to discuss open issues under the proposed business combination agreement and relating to the governance of the combined company, including (a) the restrictions sought by Deutsche Telekom on the MetroPCS board s ability to engage in discussions with other potential acquirers after execution of the business combination agreement and to change its recommendation of the transaction with Deutsche Telekom, (b) Deutsche Telekom s view that we should not be permitted to terminate the business combination agreement in order to enter into an agreement with another party for a superior transaction prior to a stockholder vote, also known as a force-the-vote

provision, (c) whether Deutsche Telekom would be required to pay a reverse break-up fee in the event the transaction could not close because of a regulatory impediment, (d) whether Deutsche Telekom would have, and the extent of, any Deutsche Telekom consent rights with respect to actions of the combined company, (e) the ability of either MetroPCS or Deutsche Telekom to obtain recourse post-closing for breaches of the business combination agreement and (f) the ability of Deutsche Telekom to sell a significant ownership stake in the combined company in the future without the other stockholders being entitled to participate in any control premium received in such a transfer. The parties discussed the issues at length, but they were unable to reach agreement on them. In addition, the parties acknowledged that the issues surrounding the refinancing of the Deutsche Telekom/T-Mobile intercompany debt and its implications for Deutsche Telekom s desire for a substantial liquidity event were significant and that further work and cooperation were necessary to resolve those issues. Mr. Lützner agreed to discuss the open issues with his colleagues at Deutsche Telekom and get back to Mr. Stachiw.

On July 5, 2012, Messrs. Stachiw and Lützner and attorneys from Gibson Dunn and Wachtell Lipton reconvened by telephone conference to discuss the major open issues in the transaction and Deutsche Telekom s responses to the issues raised by MetroPCS. In the discussion, Deutsche Telekom emphasized that, given its failed transaction to sell T-Mobile to AT&T, it wanted the maximum protections legally permissible in the business combination agreement to ensure that we would not be able to pursue alternative transactions or terminate the business combination agreement in favor of another transaction unless our stockholders failed to approve the Deutsche Telekom transaction. In addition, Deutsche Telekom sought a number of consent rights with respect to actions of the combined company, no reverse break-up fee for a regulatory failure, no post-closing indemnification and no formalized approval process for approval of combined company actions by the independent directors, other than as required by law or advisable in the judgment of the combined company s board of directors on a case-by-case basis. Finally, Deutsche Telekom argued that there should be no restrictions on its ability to sell its shares in the combined company or any requirement for it to afford the other stockholders an opportunity to participate in a significant sale and thereby participate in any resulting control premium.

On July 9, 2012, Messrs. Linquist and Carter met in person in Germany with Messrs. Obermann and Langheim and Mr. Timotheus Höttges, chief financial officer of Deutsche Telekom. Deutsche Telekom expressed that, in light of poor conditions in the financing markets, it was reluctantly willing to continue to be the lender under the Deutsche Telekom/T-Mobile intercompany debt and forego its desired liquidity event with respect to its T-Mobile investment in order to facilitate a transaction. As envisaged by Deutsche Telekom, the combined company would be capitalized with \$20.5 billion in debt, including \$3.5 billion of third-party debt to be raised by MetroPCS and fully committed at the signing of the business combination agreement with no back-stop from Deutsche Telekom, \$2.0 billion of existing MetroPCS senior notes that would remain outstanding with no back-stop from Deutsche Telekom and \$15.0 billion of debt owed by the combined company to Deutsche Telekom, which would contain market terms. Deutsche Telekom also proposed that MetroPCS stockholders would receive an aggregate cash payment of \$1.5 billion at the closing and retain a 22.5% ownership interest in the combined company, with Deutsche Telekom receiving a 77.5% ownership interest. In addition, Deutsche Telekom s representation on the combined company s board would be proportionate to its ownership in the combined company. In addition, Deutsche Telekom again raised the concept of combining Company E with MetroPCS and T-Mobile concurrently with the combination of MetroPCS and T-Mobile, which we indicated we did not want to pursue in connection with the combination with T-Mobile.

On July 17, 2012, a senior executive of Company C contacted Mr. Stachiw, indicating that Company C was considering strategic alternatives and inquiring whether we would be interested in discussing a possible transaction with Company C. Mr. Stachiw conveyed that we were in the process of exploring strategic alternatives and if Company C made a proposal, Mr. Stachiw would convey it to the MetroPCS board. Over the next several days, the senior executive of Company C and Mr. Stachiw talked on several occasions, and the

<sup>79</sup> 

senior executive of Company C indicated that Company C was considering making a proposal to acquire MetroPCS on terms similar to the publicly rumored terms of the failed transaction between MetroPCS and Company G and that it would be helpful if Company C could have our 2012 budget and five year long range forecast. Mr. Stachiw indicated that he thought the MetroPCS board would be interested in considering such a proposal and would be meeting on July 24, 2012, but Company C should recognize that our stock was trading at a low price relative to its historical trading price and that the parties would need to execute a nondisclosure agreement before we could provide the requested information to Company C. The Company C senior executive also indicated that, if the requested information could be delivered promptly to Company C, he expected a proposal to be forthcoming later in the week.

On July 24, 2012, the MetroPCS directors conducted a telephone conference with JPMorgan, Credit Suisse and Evercore during which the financial advisors provided an update on the most recent Deutsche Telekom proposal. In particular, JPMorgan, Credit Suisse and Evercore described the proposed capital structure for the combined company, including potential terms for the debt securities to be issued by the combined company to Deutsche Telekom under its proposal. Mr. Stachiw described his conversations with the senior executive of Company C, and the directors discussed the possible offer from Company C and encouraged MetroPCS management to execute a nondisclosure agreement with Company C.

On that same day, we sent Company C a nondisclosure agreement containing a standstill that would terminate if we entered into, or publicly announced, a change of control transaction with a third party. After considerable negotiation, we entered into the nondisclosure agreement with Company C on July 31, 2012. Later that day, we sent our 2012 business plan and five year long range forecast to Company C.

On July 26, 2012, we announced our second quarter financial results, which included record-high adjusted EBITDA and strong performance in other financial metrics. In the next two days, the MetroPCS stock price climbed almost 44%.

Also on July 26, 2012, we executed an amendment to our engagement agreement with JPMorgan pursuant to which we agreed to pay a fee to JPMorgan equal to a percentage of the total consideration to be received by MetroPCS stockholders in any transaction with Deutsche Telekom.

The MetroPCS directors conducted a telephone conference on July 31, 2012. JPMorgan, Credit Suisse and Evercore discussed with the directors additional financial analysis regarding the latest Deutsche Telekom proposal, including Deutsche Telekom s intention to execute the sale of T-Mobile s tower assets as a prepaid lease transaction and its financial impact on the combined company. In particular, they discussed that Deutsche Telekom s proposed structure for the tower assets sale would result in the recording of an approximately \$2.4 billion financing obligation liability on the financial statements of the combined company, which might in the view of JPMorgan, Credit Suisse and Evercore adversely affect the expected market valuation and credit ratings of the combined company. Taking into account this new information regarding the expected structuring of the tower assets sale and its potential effect on the combined company s valuation and credit ratings, the directors discussed various alternative responses to the Deutsche Telekom proposal. The directors also discussed the potential offer from Company C, which had not been received. Ultimately the MetroPCS directors instructed MetroPCS management to obtain from Deutsche Telekom a 27.5% ownership interest in the combined company and \$1.5 billion in cash for the MetroPCS stockholders to reflect the consequences of the tower assets sale on the value and credit ratings of the combined company.

Later that day, Mr. Carter contacted Mr. Langheim by telephone and proposed a counteroffer of \$1.5 billion in cash and a 27.5% ownership interest in the combined company. Mr. Langheim responded that he did not believe Deutsche Telekom would agree to grant the MetroPCS stockholders more than a 25% ownership interest. Shortly thereafter, Mr. Linquist had a telephone conversation with Mr. Obermann in which Mr. Linquist relayed the counteroffer of \$1.5 billion and a 27.5% ownership interest. Mr. Obermann responded that Deutsche Telekom would agree to a compromise of \$1.5 billion and a 26% ownership interest. Mr. Linquist indicated that he would present that compromise to the MetroPCS board.

The MetroPCS directors convened again on a conference call on August 3, 2012, and discussed Deutsche Telekom s response. JPMorgan, Credit Suisse and Evercore discussed the Deutsche Telekom proposal, and, after consulting with the MetroPCS board s and special committee s financial advisors, the MetroPCS directors authorized MetroPCS management to move forward with the transaction on the basis of the most recent proposal from Deutsche Telekom, conduct due diligence and negotiate the definitive agreements. On August 5, 2012, Mr. Carter contacted Mr. Langheim and communicated that the MetroPCS directors were prepared to move forward with the \$1.5 billion / 26% ownership interest transaction.

Also on August 3, 2012, we executed another engagement agreement with Credit Suisse with respect to the transaction with Deutsche Telekom which provided that Credit Suisse would receive a transaction fee based on a percentage of the enterprise value attributable to MetroPCS in any transaction with Deutsche Telekom.

On August 7, 2012, we received a written indication of interest from Company C stating that Company C was interested in acquiring MetroPCS for \$11 per share, payable 30% in cash and 70% in stock of Company C. The indication of interest was non-binding and subject to customary qualifications, including the need to conduct due diligence and negotiate definitive documentation for the transaction.

On August 10, 2012, the MetroPCS directors conducted a telephone conference to discuss the proposal from Company C. JPMorgan, Credit Suisse and Evercore discussed the Company C proposal and how it compared negatively to the proposed transaction with Deutsche Telekom on the basis of various financial metrics. In addition, the directors noted that the approximately \$1.2 billion of aggregate cash consideration offered by Company C was less than half of our projected year-end 2012 cash balance. Pursuant to the directors instructions, later that day we responded to Company C in writing, stating that the proposal was timely because MetroPCS was currently evaluating other strategic options, but that the Company C proposal was not adequate, both in terms of absolute value as well as the cash consideration per share, particularly in light of MetroPCS projected year-end cash balance, that we were open to Company C submitting an improved proposal, and that we were prepared to move quickly if an improved proposal that was attractive to the MetroPCS board was made.

Beginning in mid-August 2012, we and Deutsche Telekom engaged in customary due diligence, granting each other and our respective representatives access to electronic data rooms containing additional confidential materials regarding various business, financial, legal, regulatory, employee benefits and other matters regarding our respective businesses. We continued to exchange information and hold conference calls for due diligence purposes during the next six weeks.

On August 14, 2012, a senior executive of MetroPCS contacted a senior executive of Company G to discuss the Company G senior executive s publicly announced impending departure from Company G. In their conversation, the Company G senior executive told the MetroPCS senior executive that Company G was evaluating whether to make a proposal to acquire MetroPCS. The senior executive of Company G indicated that in his view the reasons for Company G s decision not to proceed with its proposed acquisition of MetroPCS earlier in the year were no longer applicable and that he was advocating to the board of Company G that Company G make an offer to acquire MetroPCS. The MetroPCS senior executive stated that it was important that any such offer be made very soon because we were considering a variety of alternatives. The Company G senior executive indicated that if Company G determined to make an offer, Company G could deliver its proposal to MetroPCS before the end of September.

On August 14, 2012, Wachtell Lipton distributed a revised draft of the business combination agreement and a draft of a stockholder s agreement between MetroPCS and Deutsche Telekom, which we distributed to the special committee s advisors. Wachtell Lipton and Gibson Dunn proceeded to negotiate aspects of the agreements over the following weeks. On August 22, 2012, Gibson Dunn sent to Wachtell Lipton a list of significant issues arising under the business combination and stockholder s agreements, which we distributed to the special committee s advisors. In general, the same major issues that had previously been discussed by the lawyers at their June 25, 2012 meeting in New York remained outstanding.

On August 22, 2012, Mr. Carter spoke with a senior executive of Company C, who asked questions concerning our response to Company C s written indication of interest. The Company C senior executive indicated that Company C would discuss a potential revised offer for MetroPCS further internally and respond if Company C was interested in pursuing the matter further. In the conversation, the Company C senior executive noted the significant recent increase in our stock price and indicated that the increased stock price would likely make it more difficult for Company C to make an offer that would be acceptable to the MetroPCS board.

On August 23, 2012, Mr. Obermann informed Mr. Linquist that Deutsche Telekom had undertaken a search for a chief executive officer for T-Mobile and that the search had identified an individual for the position, but until the employment agreement was finalized, the individual s identity would remain confidential. At that time, Mr. Obermann expressed the desire to have Mr. Linquist become the non-executive chairman of the board of the combined company. Mr. Linquist indicated he would consider Mr. Obermann s proposal, but he was inclined not to accept it. Mr. Obermann also indicated to Mr. Linquist that Deutsche Telekom remained interested in having Mr. Carter be the chief financial officer of the combined company.

On August 27, 2012, Mr. Linquist spoke to the chairman of the board of Company C to determine Company C s interest in making an improved offer. The chairman of the board of Company C responded that he was focused on other business matters, but Company C was considering whether to submit an improved offer and understood our timing constraints.

On August 30, 2012, the MetroPCS board held its regular quarterly board meeting during which, among other things, it discussed the progress of the Deutsche Telekom transaction. The legal advisors reported on the major outstanding issues under the business combination and stockholder s agreements. Mr. Linquist reported on his conversation with the chairman of the board of Company C. To ensure that we would not forgo any superior opportunity by entering into a business combination transaction with Deutsche Telekom, the MetroPCS directors determined that Company G, which was viewed as the most likely alternative buyer to Deutsche Telekom, particularly in light of the recent conversation with the Company G senior executive, should be contacted to determine whether it was interested in making an offer to acquire MetroPCS and to emphasize further the importance of making any proposal quickly. There was also discussion regarding whether to contact Company F, but the consensus was that Company F was not interested in acquiring MetroPCS. Mr. Linquist also shared with the MetroPCS directors in executive session that Deutsche Telekom had selected a chief executive officer for T-Mobile and that Mr. Obermann had asked Mr. Linquist to be the non-executive chairman of the board of the combined company. Mr. Linquist indicated that he was considering declining any appointment to be the non-executive chairman of the board of the combined company, but the MetroPCS directors asked Mr. Linquist to reconsider, which he agreed to do. Mr. Linquist also advised the MetroPCS directors that Deutsche Telekom had indicated it wanted Mr. Carter to be the chief financial officer of the combined company.

On August 31, 2012, a member of the MetroPCS special committee contacted a Company G board member who was the chairman of Company G s finance committee and relayed that we were examining our strategic alternatives and were likely to make a decision in the near term. The Company G board member echoed what the senior executive of Company G previously had told the senior executive of MetroPCS regarding Company G s previously withdrawn proposal and further indicated that Company G was considering whether to make an offer to acquire MetroPCS.

On September 5, 2012, senior executives of MetroPCS and Deutsche Telekom, attorneys from Gibson Dunn and Wachtell Lipton, Deutsche Telekom s financial advisor, JPMorgan and Credit Suisse met in New York to conduct in-person negotiations regarding material transaction issues. The major issues under the business combination agreement included (a) a potential purchase price adjustment based on any deviation in T-Mobile s working capital, capital expenditures or marketing and subscriber acquisition costs from targeted amounts, (b) matters relating to the timing of filing the proxy statement and holding the MetroPCS stockholders meeting, (c) provisions limiting the ability of the MetroPCS board to pursue other transactions, terminate the business

combination agreement to accept another proposal, and change its recommendation in favor of the Deutsche Telekom transaction, including the force-the-vote provisions, (d) conditions that were required to be satisfied in order for closing to occur, (e) whether there would be a reverse break-up fee payable by Deutsche Telekom to us in the event of a regulatory impediment to closing, (f) the amount of the termination fee payable by us to Deutsche Telekom under certain circumstances and (g) whether the business combination agreement would provide for post-closing indemnification for breaches of the agreement. Under the stockholder s agreement, the primary issues related to the extent of Deutsche Telekom s consent rights and Deutsche Telekom s ability to sell a large number of shares in the combined company without requiring that the other stockholders be given the opportunity to participate in the sale and thereby participate in any resulting control premium. The parties also began to negotiate the specific terms of the combined company s debt to be issued to Deutsche Telekom, a major commercial issue.

On September 10 and 11, 2012, Messrs. Linquist, Carter and Keys met with Messrs. Obermann, Langheim and Höttges and other senior executives of Deutsche Telekom in Bonn, Germany, to discuss potential business plans for the combined company and the governance framework for the combined company. At this meeting, Deutsche Telekom reported that it had hired a chief executive officer for T-Mobile and planned for him to serve as the chief executive officer of the combined company. Deutsche Telekom stated that it wanted Mr. Carter to be the chief financial officer of the combined company and Mr. Linquist to be the non-executive chairman of the board of the combined company and that it could not guarantee Mr. Keys a position with the combined company because it wanted the new chief executive officer to make decisions regarding the other members of the management team. Mr. Linquist indicated that he was not interested in being the non-executive chairman of the board of the combined company, but Mr. Obermann asked that Mr. Linquist reconsider, which he agreed to do. At this meeting, the parties also agreed that they would work to complete negotiations of the business combination agreement so that the transaction could be announced in the first week of October.

On September 13, 2012, the MetroPCS directors met by telephone to discuss developments from the meeting with Deutsche Telekom in Germany. Mr. Linquist reported that the parties made plans to meet again in New York in the following week with the goal of resolving major open issues at that time. During the executive session of the meeting, Mr. Linquist reiterated his position regarding the non-executive chairman s role with the combined company, and he further reported that, in addition to Deutsche Telekom s stated interest in naming Mr. Carter as the chief financial officer of the combined company, it had indicated that Mr. Keys might also have a senior leadership role but had not guaranteed him a position.

Senior executives, including Messrs. Linquist, Carter, Keys and Stachiw from MetroPCS and Messrs. Obermann, Langheim and Höttges from Deutsche Telekom, and legal teams for MetroPCS and Deutsche Telekom met again in New York on September 20, 2012, along with the newly appointed chief executive officer of T-Mobile. At the meeting, we negotiated and ultimately resolved major outstanding issues under the business combination and stockholder s agreements. Among other things, we and Deutsche Telekom agreed to (a) a one-way purchase price adjustment payable by Deutsche Telekom to the combined company in the event there was a shortfall in T-Mobile s working capital, capital expenditures or marketing and subscriber acquisition spending relative to targeted amounts, (b) grant MetroPCS the right under certain circumstances to delay its stockholders meeting, (c) grant MetroPCS more flexibility to consider other transactions and change its recommendation in favor of the Deutsche Telekom transaction but not permit MetroPCS to terminate the business combination agreement to enter into a superior proposal, (d) eliminate closing conditions relating to Deutsche Telekom s obtaining financing, (e) include a reverse break-up fee payable by Deutsche Telekom to MetroPCS in the event of a regulatory impediment to closing, (f) the amount of the termination fee payable by MetroPCS to Deutsche Telekom under certain circumstances, (g) not provide for post-closing indemnification for breaches of the agreement, (h) limit Deutsche Telekom s consent rights with respect to the combined company s operations post-closing, (i) include a provision in the stockholder s agreement providing that, in the event that Deutsche Telekom transferred 30% or more of its shares of the combined company to a transferee, Deutsche Telekom would require the transferee to make an offer to acquire all of the shares of the combined company to a

held by Deutsche Telekom on the same terms, (j) include a standstill provision in the stockholder s agreement whereby Deutsche Telekom could not acquire more than 80.1% of the combined company without making an offer to acquire the remainder of the combined company and (k) the terms of Deutsche Telekom s covenant not to compete with the combined company. In addition, at the meeting, Deutsche Telekom stated that it had agreed to back-stop the issuance of notes to refinance our \$2.5 billion credit facility and, if needed, to satisfy change in control put obligations in connection with our \$2.0 billion of outstanding senior notes. Deutsche Telekom also agreed to back-stop an additional \$1.0 billion of new senior notes issuable by MetroPCS on or before closing and to provide a \$500 million working capital facility to the combined company. Mr. Obermann again raised with Mr. Linquist the question of whether Mr. Linquist would serve as non-executive chairman of the board of the combined company, and Mr. Linquist declined. Mr. Obermann then requested that Mr. Linquist serve as a director of the combined company, which Mr. Linquist said he would consider.

Over the next two weeks, we and Deutsche Telekom and our respective representatives worked to complete mutual due diligence and finalize the terms of the business combination agreement, stockholder s agreement, the other acquisition-related agreements and the financing-related documents. In addition, we and Deutsche Telekom and our respective representatives met to discuss and negotiate terms of the combined company s debt.

On September 24, 2012, we provided drafts of the business combination and stockholder s agreements to the MetroPCS special committee and its financial and legal advisors.

On September 25, 2012, the MetroPCS directors met by telephone to discuss the most recent developments in negotiations with Deutsche Telekom and to discuss with the MetroPCS board s and the special committee s financial advisors the financial aspects of the transaction. In addition, Ernst & Young, which we had engaged to assist with the due diligence investigation of T-Mobile, delivered a detailed presentation of its findings to the directors. Akin Gump reviewed for the directors their fiduciary duties in connection with the proposed transaction. During a separate special committee meeting, the special committee discussed with Evercore in greater detail the terms of the proposed transaction and the proposed management of the combined company.

On September 26, 2012, Mr. Linquist called Mr. Obermann and told him that he had considered the offer to be a director but had decided to decline. However, Mr. Linquist indicated that if Mr. Obermann wanted Mr. Linquist to be available to assist with matters relating to the combined company post-closing, he would agree to a mutually acceptable consulting arrangement, but the terms of such consulting agreement were not discussed.

On September 29, 2012, we provided updated drafts of the business combination and stockholder s agreements, the amended and restated certificate of incorporation, the bylaws and the T-Mobile disclosure letter to the business combination agreement to the MetroPCS special committee and its financial and legal advisors. In addition, the parties met in New York to try to reach agreement on the commercial terms of the combined company s debt and the terms on which Deutsche Telekom would backstop certain financing transactions.

On October 2, 2012, various news outlets began reporting on a possible transaction involving Deutsche Telekom, T-Mobile and MetroPCS. In response to these reports, we and Deutsche Telekom and MetroPCS issued separate press releases confirming that discussions regarding such a transaction were occurring.

Also on October 2, 2012, we entered into an agreement with JPMorgan pursuant to which we agreed to pay JPMorgan a cash fee upon completion of the transaction with T-Mobile as consideration for JPMorgan s services evaluating various financing structures with respect to the transaction with T-Mobile and other transactions considered by MetroPCS.

Also on October 2, 2012, the parties reached agreement on the commercial terms of the combined company s debt and the terms on which Deutsche Telekom would backstop certain financing transactions.

On the evening of October 2, 2012, the MetroPCS board and special committee met in person in New York to consider the final terms of the transaction with Deutsche Telekom. Representatives of MetroPCS management, JPMorgan, Credit Suisse, Evercore, Gibson Dunn and Akin Gump attended the meeting. Following some preliminary business, the MetroPCS board meeting was recessed to allow the special committee meeting to occur. Akin Gump opened the special committee meeting with a review for the directors of their fiduciary duties in the context of the proposed transaction. Management of MetroPCS then delivered a detailed report on T-Mobile s business. The special committee also discussed in detail the potential synergies from the combination with T-Mobile, as estimated by MetroPCS management to be \$6.7 billion of operational and capital synergies, with the opportunity for an additional approximately \$800 million of other synergies, including that the full realization of the synergies would take a number of years due to the time required to decommission our network and move our customers to the combined company s network. MetroPCS management also reviewed our long range forecast and the assumption that we would spend \$1.5 billion on spectrum acquisitions in 2012 if we remained a stand-alone company. JPMorgan and Credit Suisse also described to the special committee the financing arrangements for the transaction, including the material terms of the financing and the material steps to be accomplished to complete the financing. Gibson Dunn reviewed the material terms of the business combination agreement, stockholder s agreement and other material transaction documents. The directors on the special committee also received a summary of matters that arose in our due diligence investigation of T-Mobile. Evercore explained its analysis of the transaction to the special committee and rendered its oral opinion, subsequently confirmed in writing, that as of the date of the opinion and based on and subject to the assumptions, qualifications and limitations on the scope of the review undertaken by Evercore as discussed in its opinion, the transaction was fair, from a financial point of view, to MetroPCS and the MetroPCS stockholders (other than Deutsche Telekom and its affiliates).

During the special committee meeting, a senior executive of Company G contacted a senior executive of MetroPCS, and they engaged in a conversation by telephone. In that conversation, the senior executive of Company G conveyed that Company G was working on a transaction that would result in a large new investment of capital in Company G and that the Company G board was scheduled to hold a meeting at the end of the week, at which it would consider whether to move forward with a proposal to acquire MetroPCS. The senior executive of Company G also explained that, if the Company G board decided to move forward with an acquisition of MetroPCS, it would take several weeks for Company G to prepare a proposal. The Company G senior executive then mentioned that he had seen the recent news reports regarding a potential combination between MetroPCS and T-Mobile and hoped such a transaction would not occur, but, if MetroPCS did enter into an agreement to be acquired, he hoped any break-up fee payable by MetroPCS to pursue an alternative transaction would be reasonable.

The special committee then met in executive session with Akin Gump, Evercore and selected senior executives of MetroPCS. The senior executive of MetroPCS who had talked to the Company G senior executive earlier in the evening then summarized the telephone call for the special committee. The directors asked questions regarding the conversation and whether the MetroPCS board should wait for an offer that might come from Company G in lieu of approving the transaction with Deutsche Telekom. The directors weighed the fully negotiated transaction with Deutsche Telekom against the possibility of an offer from Company G, particularly in light of Company G s previous decision earlier in the year not to move forward with an acquisition of MetroPCS and the directors conclusion that the termination fee in the business contribution agreement was reasonable. The directors, management or their advisors that would be reasonably likely to result in a value to MetroPCS stockholders in excess of the value presented in the Deutsche Telekom transaction (including as a result of the outreach to other potentially interested parties that had previously occurred). At the conclusion of the presentations and related discussions, the special committee unanimously recommended that the MetroPCS board approve the transaction. Following receipt of this recommendation from the special committee. The MetroPCS board meeting was reconvened and the MetroPCS board unanimously approved the transaction with Deutsche Telekom and recommended that the MetroPCS stockholders vote FOR the proposals contained in this proxy statement.

After the conclusion of the MetroPCS board meeting and the meeting of Deutsche Telekom s supervisory board, which also approved the transaction, MetroPCS and Deutsche Telekom executed the business combination agreement on the morning of October 3, 2012. Also on the morning of October 3, 2012, MetroPCS and Deutsche Telekom issued a joint press release announcing the execution and delivery of the business combination agreement.

#### Reasons for the Transaction; Recommendation of the MetroPCS Board and the Special Committee

As described in Background of the Transaction, the MetroPCS board carefully evaluated our competitive position and determined that the acquisition of additional spectrum was critical to meet increasing customer demands for data services, continue customer growth, maintain quality of service, allow the deployment of new technologies and enhance stockholder value. In response to the MetroPCS board s directives, MetroPCS management attempted over the past four years to acquire a meaningful amount of spectrum to both expand MetroPCS licensed area and to increase the amount of spectrum in its existing service areas. However, MetroPCS was unable to acquire a meaningful amount of additional spectrum and has concluded that it faces limited prospects in the future for the acquisition of significant additional spectrum from third parties or FCC spectrum auctions. In light of these circumstances, the MetroPCS board evaluated the combination with T-Mobile, which would substantially increase MetroPCS spectrum assets and improve its competitive position. In evaluating whether to recommend that you vote

for the proposals and whether MetroPCS should enter into the business combination agreement and the transaction, including the MetroPCS stock issuance and the new certificate of incorporation, the MetroPCS board, including the special committee, consulted with MetroPCS management and its legal, financial and other advisors, and also considered numerous factors, including the following factors that each of the MetroPCS board and the special committee viewed as supporting its decision to approve the business combination agreement and the transaction:

the implied value of the consideration to be received by the MetroPCS stockholders, calculated with the assistance of its financial advisors, which represents a compelling valuation for MetroPCS and its stockholders;

the MetroPCS stockholders receiving in the aggregate \$1.5 billion (or approximately \$4.08 per share calculated on a pre-reverse stock split basis), allowing the MetroPCS stockholders an opportunity to realize certain value for a significant portion of their investment immediately upon completion of the transaction;

the MetroPCS equityholders immediately prior to the completion of the transaction collectively continuing to own 26% of the fully-diluted MetroPCS common stock (before taking into account any stock options that are subsequently cashed out in connection with the transaction) immediately upon completion of the transaction, allowing the MetroPCS stockholders to participate in any future earnings or growth of the combined company, synergies and cost savings, and future appreciation in the value of the combined company s common stock;

Deutsche Telekom s agreement to provide the financing in connection with the transaction, as more fully described under the caption Summary of the Financing beginning on page 112;

the creation, by combining MetroPCS and T-Mobile, of a leading value carrier in the U.S. wireless telecommunications industry that will benefit from its increased size and resources, including benefits from a more diverse customer base, increased spectrum assets, network coverage and capacity and improved marketing and purchasing scale;

the expectation that the combined company will roll out an LTE network that includes at least 20x20 MHz in a number of major metropolitan areas;

the complementary nature of MetroPCS and T-Mobile s networks and spectrum assets that will facilitate the migration of MetroPCS subscribers onto T-Mobile s network and, once combined, create a nationwide network footprint and greater customer value and choice;

management s estimate, consistent with Deutsche Telekom s and T-Mobile s estimate, that the transaction could result in between \$6 billion and \$7 billion in cost synergies;

the combined company having a highly experienced management team with extensive experience in the wireless telecommunications industry, and the combined company being able to draw upon the intellectual capital, technical expertise and experience of a deeper and more diverse workforce;

the expectation that the combined company would have an attractive growth profile and increased financial flexibility and continuing access to public capital markets;

MetroPCS right initially to appoint at least two directors to the combined company s board of directors and the expectation that two executive officers of MetroPCS will be executive officers of the combined company, which may facilitate continuity between MetroPCS and the combined company;

the MetroPCS board s and the special committee s belief that the transaction is more favorable to the MetroPCS stockholders than the alternatives to the transaction, including remaining a stand-alone company, which belief was based on the MetroPCS board s and the special committee s review, with the assistance of its financial advisors, of the strategic alternatives available to MetroPCS after the MetroPCS board s, the special committee s and management s discussions with other likely acquirers, as more fully described under the section entitled Background of the Transaction beginning on page 64;

the MetroPCS board s and the special committee s understanding of the business, operations, financial condition, earnings and prospects of MetroPCS, including the challenges faced by MetroPCS in acquiring spectrum assets necessary to be a viable stand-alone company and MetroPCS other challenges and risks as a stand-alone company, as more fully described under the section entitled The Transaction Background of the Transaction beginning on page 64;

the presentation to the special committee on October 2, 2012, and financial analysis reviewed therewith, of Evercore, and the opinion, dated October 2, 2012 of Evercore, addressed to the special committee in its capacity as such, to the effect that, as of that date and based on, and subject to, assumptions made (including the payment by MetroPCS to its stockholders of the cash payment), matters considered and limitations on the scope of review undertaken by Evercore as set forth therein, the MetroPCS stock issuance equal to 74% of the fully-diluted shares of MetroPCS common stock outstanding immediately following the cash payment in consideration for the acquisition by MetroPCS from a subsidiary of Deutsche Telekom of the T-Mobile shares, is fair, from a financial point of view, to MetroPCS and its stockholders (other than Deutsche Telekom and its affiliates) as more fully described in the section entitled The Transaction Opinion of the Financial Advisor to the MetroPCS Special Committee beginning on page 91;

the likelihood that the transaction would be completed, based on, among other things:

the MetroPCS board s and the special committee s belief, after consultation with its internal and outside legal counsel, that the transaction is likely to receive necessary regulatory approvals without material adverse conditions within the outside date for termination of the business combination agreement;

the business combination agreement s provisions that, in the event of a failure of the transaction only for regulatory reasons, Deutsche Telekom will pay MetroPCS a \$250 million termination fee;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

Deutsche Telekom s inability to terminate the business combination agreement due to a decline in MetroPCS business, assets, liabilities, prospects, outlook, financial condition or results of operations unless such decline has a material adverse effect on us;

Deutsche Telekom s inability to pursue alternative acquisition proposals for T-Mobile under the business combination agreement;

Deutsche Telekom s commitment to provide all necessary financing in order to complete the transaction pursuant to the terms of the business combination agreement and the fact that Deutsche Telekom does not have the ability to terminate the business combination agreement for a failure to obtain such financing;

MetroPCS ability, under certain circumstances pursuant to the business combination agreement, to obtain specific performance to prevent breaches of the business combination agreement by Deutsche Telekom, Global, Holding or T-Mobile and to enforce specifically the terms of the business combination agreement, including Deutsche Telekom s obligations to provide financing in connection with the transaction;

the belief that the outside date (as defined in the business combination agreement) will likely allow sufficient time to complete the transaction; and

the strong commitment on the part of the parties to complete the transaction pursuant to their respective obligations under the terms of the business combination agreement;

the provisions of the business combination agreement allowing a recommendation change by the MetroPCS board if MetroPCS receives a superior proposal (as described in the section entitled Summary of the Business Combination Agreement No Solicitation of Alternative Proposals beginning on page 142) from a third party or experiences an intervening event and if, in either case, the MetroPCS board determines that a failure to make a recommendation change would be reasonably likely to be inconsistent with its fiduciary obligations to MetroPCS and/or its stockholders under applicable law, subject to MetroPCS obligation to pay Deutsche Telekom a termination fee of \$150 million upon termination of the business combination agreement under certain circumstances;

the restrictions within the stockholder s agreement on Deutsche Telekom s ability to transfer any shares of the combined company s common stock for six months after the closing of the transaction or Deutsche Telekom s ability to transfer any shares of the combined company s common stock at any time after such period in any transaction that would result in the transferee owning 30% or more of the combined company s common stock at the same price and on the same terms and conditions as the proposed transfer, thereby helping to ensure the public stockholders share the benefits of any sale-of-control transaction;

the provisions of the stockholder s agreement requiring that certain post-closing matters (including determinations with respect to working capital adjustments) will be made by the combined company s directors who are not affiliated with Deutsche Telekom;

the prohibition within the stockholder s agreement on Deutsche Telekom acquiring more than 80.1% of the outstanding shares of the combined company s common stock unless Deutsche Telekom were to make an offer to acquire all of the then remaining outstanding shares of the combined company s common stock at the same price and on the same terms and conditions as the proposed acquisition from all other stockholders of the combined company, thereby helping to ensure the public stockholders obtain a fair price in any squeezeout transaction;

the restrictions within the stockholder s agreement on Deutsche Telekom s ability to engage in certain competing activities with the combined company in the territory during the period beginning on the date of the closing of the transaction and ending on the date that is two years after the date on which Deutsche Telekom beneficially owns less than 10% of the outstanding shares of the combined company s common stock;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

the terms of the trademark license, including the provisions permitting the combined company to continue to use the T-Mobile trademarks, the exclusive nature of the license in the territory, the provision maintaining the royalty payment the same for five years, Deutsche Telekom s obligation

to indemnify the company against trademark infringement claims with respect to certain licensed T-Mobile trademarks and the wind down period after termination or expiration of the trademark license;

the prohibitions in the business combination agreement against Deutsche Telekom asserting claims against the combined company or its subsidiaries regarding any intellectual property owned by Deutsche Telekom and used by T Mobile or any of its subsidiaries immediately prior to the closing date;

the restrictions within the stockholder s agreement on Deutsche Telekom s (and Deutsche Telekom s director designees ) ability to, support, enter into or vote in favor of any transaction between, or involving both (A) the combined company and (B) Deutsche Telekom or an affiliate of Deutsche Telekom, unless such transaction is approved by a majority of the directors on the combined company s board, which majority includes a majority of the directors on the combined company s board that are not affiliates of Deutsche Telekom;

the inclusion in the new certificate of incorporation of a number of provisions that are more favorable to stockholders than the comparable provisions in MetroPCS existing certificate of incorporation; for example, the new certificate of incorporation will provide for a declassified board, stockholder rights to call a special meeting and act by written consent, and a lower stockholder voting threshold to amend the combined company s bylaws;

the special committee s retention of independent legal and financial advisors; and

the contractual benefits that MetroPCS management and its advisors were able to obtain as a result of extensive negotiations with Deutsche Telekom, which the MetroPCS board and the special committee concluded reflected the best terms that Deutsche Telekom would be willing to provide at that time.

The MetroPCS board and the special committee also considered a variety of potentially negative factors in their deliberations concerning the business combination agreement and the transaction, including the MetroPCS stock issuance and the new certificate of incorporation, including the following:

forecasts of future results of operations and synergies are necessarily estimates based on assumptions, and that for these and other reasons there is a risk of not capturing anticipated operational synergies and cost savings between MetroPCS and T-Mobile and the risk that other anticipated benefits might not be realized;

because MetroPCS stockholders are receiving a significant amount of cash in the transaction, they will have a smaller ongoing equity participation in the combined company and, therefore, will receive less of the benefit of any synergies or any increase in the value of the combined company;

the amount of MetroPCS common stock to be issued to Holding or its designee in the transaction is a fixed percentage, meaning Deutsche Telekom s and the existing MetroPCS stockholders relative ownership of the combined company immediately following the closing may not reflect changes in the relative values of T-Mobile and MetroPCS between signing and closing;

the fact that certain of MetroPCS existing management team, including Roger D. Linquist, are not expected to continue as officers of the combined company;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

the substantial costs to be incurred in connection with the transaction, including the substantial cash and other costs of integrating the businesses of MetroPCS and T-Mobile, as well as the transaction expenses arising from the transaction and the delayed nature of the anticipated cost and capital expenditure savings and synergies of the transaction versus when the costs to create such cost and capital expenditure savings and synergies will be incurred by the combined company;

the substantial time and effort of management required to complete the transaction and related disruptions to the operations of MetroPCS business, and the risk of diverting management resources for an extended period of time to accomplish this combination;

the potential adverse effect of the transaction on MetroPCS business and relationships with employees, customers, suppliers, distributors, dealers, regulators and the communities in which it operates;

the risk that uncertainties about the completion of the transaction and contractual restrictions in the business combination agreement involving the operation of MetroPCS business could disrupt MetroPCS operations, limit its ability to take advantage of attractive opportunities, including purchasing spectrum and making changes to its business, or have an adverse effect on its business and results of operations;

the risk that governmental entities may not approve the transaction or may impose conditions on the parties to the business combination agreement in order to gain approval for the transaction that may adversely impact the ability of the combined enterprise to realize the synergies that are projected to occur in connection with the transaction;

the possibility that the transaction may not be completed, or that completion may be unduly delayed, for reasons beyond the control of the parties to the business combination agreement;

the \$150 million termination fee that MetroPCS may be required to pay to Deutsche Telekom under certain circumstances upon the termination of the business combination agreement;

the potential that the termination fee provisions of the business combination agreement and the requirement that the MetroPCS board call and hold the special meeting, regardless of whether or not the MetroPCS board has made a recommendation change, could affect a bona fide alternative proposal for MetroPCS;

the inclusion of the non-solicitation restrictions in the business combination agreement that could limit, in whole or in part, MetroPCS ability to negotiate or enter into an alternative transaction;

the restrictions in the business combination agreement preventing MetroPCS from terminating the business combination agreement to accept a superior proposal until after the special meeting and only then if the MetroPCS stockholders do not approve the stock issuance proposal and the new certificate of incorporation proposal;

that Deutsche Telekom will be a controlling stockholder of the combined company immediately upon completion of the transaction, and will have approval rights with respect to certain actions of the combined company (see the section entitled Summary of Ancillary Agreements Stockholder s Agreement beginning on page 154), each of which could have a negative impact on the combined company s stock price;

that Deutsche Telekom will own a substantial amount of the combined company s indebtedness, certain actions, including amendments and waivers with respect to the combined company s indebtedness and may be accomplished only if Deutsche Telekom consents to such action;

the combined company s lack of ownership of the T-Mobile trademarks and Deutsche Telekom s ability to terminate the trademark license under certain circumstances;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

that Deutsche Telekom s substantial equity and debt ownership in the combined company could cause it to take actions for its own benefit that could be adverse to the interests of the combined company s stockholders;

the absence of a contractual right of the combined company to make indemnification claims against Deutsche Telekom for the breach of any representations and warranties or covenants made by Deutsche Telekom or its subsidiaries in the business combination agreement;

the vesting of all outstanding annual cash performance awards, stock options and restricted stock for all directors, employees and officers of MetroPCS as a result of the transaction;

the risks described in the sections entitled Risk Factors beginning on page 33 and Cautionary Note Regarding Forward-Looking Statements beginning on page 210; and

that MetroPCS directors and executive officers have interests in the transaction that may be different from, or in addition to, those of MetroPCS stockholders (see the section entitled Interests of MetroPCS Directors and Officers in the Transaction beginning on page 118).

This discussion of the information and factors considered by each of the MetroPCS board and the special committee in reaching its conclusions and recommendation includes the material factors considered by the MetroPCS board and the special committee, respectively, but is not intended to be exhaustive. In view of the wide variety of factors considered in connection with its evaluation of the transaction and the complexity of these matters, the MetroPCS board did not find it practicable, and it did not attempt, to quantify, rank or assign any relative or specific weights to the various factors that it considered in reaching its determination to approve the business combination agreement and the transaction and to recommend that MetroPCS stockholders vote in favor of the MetroPCS stock issuance and the new certificate of incorporation. The MetroPCS board and the special committee conducted an overall analysis of the factors described above, including through discussions with, and questioning of, MetroPCS management and outside legal, financial and other advisors regarding certain of the matters described above. In considering the factors described above, individual members of the MetroPCS board or the special committee may have given differing weights to different factors.

The MetroPCS board, acting upon the recommendation of the special committee, unanimously approved the business combination agreement and has determined that the transaction, including the MetroPCS stock issuance and the new certificate of incorporation, is appropriate, advisable and in the best interests of MetroPCS and its stockholders. The MetroPCS board, acting upon the recommendation of the special committee, unanimously recommends that MetroPCS stockholders vot<u>e</u> FOR the stock issuance proposal <u>and</u> FOR the new certificate of incorporation proposal.

#### **Opinion of the Financial Advisor to the MetroPCS Special Committee**

In April 2012, Evercore was retained by MetroPCS on behalf of the special committee of the MetroPCS board to act as financial advisor to the special committee with respect to potential strategic transactions. On October 2, 2012, at a meeting of the special committee, Evercore delivered to the special committee an oral opinion, which opinion was confirmed by delivery of a written opinion dated October 2, 2012, to the effect that, as of that date and based on and subject to assumptions made (including the payment by MetroPCS to its stockholders of the cash payment), matters considered and limitations on the scope of review undertaken by Evercore as set forth therein, the MetroPCS stock issuance equal to 74% of the fully-diluted shares of MetroPCS common stock outstanding immediately following the cash payment (the shares of MetroPCS common stock to be so issued in the MetroPCS stock issuance being referred to as the business combination consideration) in consideration for the acquisition, which we refer to as the business combination, by MetroPCS from a subsidiary of Deutsche Telekom of the T-Mobile shares, is fair, from a financial point of view, to MetroPCS and its stockholders (other than Deutsche Telekom and its affiliates). A copy of Evercore s written opinion was also provided by the special committee to the MetroPCS board, in its capacity as such, in connection with the MetroPCS board s evaluation of the transaction and receipt of the special committee s recommendation to the MetroPCS board.

The full text of Evercore s written opinion, dated October 2, 2012, which sets forth, among other things, the procedures followed, assumptions made, matters considered and limitations on the scope of review undertaken in rendering its opinion, is attached as Annex B to this proxy statement and is incorporated by reference in its entirety into this proxy statement. Evercore s opinion was directed to the special committee, in its capacity as such, and addresses only the fairness to MetroPCS and its stockholders (other than Deutsche Telekom and its affiliates), from a financial point of view, assuming payment of the cash payment, of MetroPCS issuance of the business combination consideration in consideration for the T-Mobile shares. The opinion does not address any other term or aspect of the

business combination agreement or the transaction or any term or aspect of any other agreement or instrument contemplated by the business combination agreement or entered into or amended in connection with the transaction and does not constitute a recommendation to the special committee or to any other persons in respect of the business combination agreement or the transaction, including as to how any holder of shares of MetroPCS common stock should vote or act in respect of the business combination agreement or the transaction.

In connection with rendering its opinion, Evercore, among other things:

reviewed certain publicly available business and financial information relating to MetroPCS and T-Mobile that Evercore deemed to be relevant;

reviewed certain non-public historical financial statements and other non-public historical financial and operating data relating to MetroPCS prepared and furnished to Evercore by the management of MetroPCS;

reviewed certain non-public historical financial statements and other non-public historical financial and operating data relating to T-Mobile prepared by the management of Deutsche Telekom and T-Mobile and furnished to Evercore by MetroPCS;

reviewed certain non-public projected financial data relating to MetroPCS prepared and furnished to Evercore by the management of MetroPCS;

reviewed certain non-public projected financial data relating to T-Mobile prepared by the management of Deutsche Telekom and T-Mobile, which we refer to as the T-Mobile management projected financial data, furnished to Evercore by MetroPCS as well as certain adjustments thereto made at the direction of MetroPCS management, or, as so adjusted, the adjusted T-Mobile projected financial data;

reviewed certain non-public projected operating data relating to MetroPCS prepared and furnished to Evercore by the management of MetroPCS;

reviewed certain non-public projected operating data relating to T-Mobile prepared by the management of Deutsche Telekom and T-Mobile, which we refer to as the T-Mobile management projected operating data, furnished to Evercore by MetroPCS as well as certain adjustments thereto made at the direction of MetroPCS management, or, as so adjusted, the adjusted T-Mobile projected operating data;

discussed the past and current operations, financial projections and current financial condition of MetroPCS and T-Mobile with the management of MetroPCS (including their views on the risks and uncertainties of achieving such projections);

discussed the past and current operations, financial projections and current financial condition of MetroPCS and T-Mobile with the management of Deutsche Telekom and T-Mobile (including their views on the risks and uncertainties of achieving such projections);

reviewed the amount and timing of the cost savings and operating synergies estimated by MetroPCS management to result from the business combination, which we refer to as the estimated synergies;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

reviewed the amount, timing and use of certain tax attributes of the combined company as estimated by management of MetroPCS;

reviewed the reported prices and the historical trading activity of the MetroPCS common stock;

compared the financial performance of MetroPCS and its stock market trading multiples with those of certain other publicly traded companies that Evercore deemed relevant;

reviewed certain valuation multiples relating to the business combination with those of certain other transactions that Evercore deemed relevant;

reviewed a draft of the business combination agreement dated October 1, 2012; and

performed such other analyses and examinations and considered such other factors that Evercore deemed appropriate. For purposes of its analysis and opinion, Evercore assumed and relied upon, without undertaking any independent verification of, the accuracy and completeness of all of the information publicly available, and all of the information supplied or otherwise made available to, discussed with, or reviewed by Evercore, and Evercore assumed no liability therefor. With respect to the projected financial and operating data relating to MetroPCS, Evercore assumed that they had been reasonably prepared on bases reflecting the best available estimates and good faith judgments of management of MetroPCS as to the matters covered thereby. Evercore expressed no view as to any projected financial or operating data relating to MetroPCS or the assumptions on which they were based. With respect to T-Mobile management projected financial data and the T-Mobile management projected operating data, Evercore assumed that they had been reasonably prepared on bases reflecting the best available estimates and good faith judgments of management of Deutsche Telekom and T-Mobile as to the matters covered thereby. With respect to adjusted T-Mobile projected financial data and the adjusted T-Mobile projected operating data, Evercore assumed that the adjustments which MetroPCS management directed Evercore to make to the T-Mobile management projected financial data and the T-Mobile management projected operating data had been reasonably determined by MetroPCS management on bases reflecting the best available estimates and good faith judgments of management of MetroPCS as to the matters covered thereby. Evercore expresses no view as to any projected financial or operating data relating to T-Mobile (whether prepared by management of Deutsche Telekom or T-Mobile, or as adjusted at the direction of management of MetroPCS) or the assumptions on which they are based. Evercore also assumed that the estimated synergies were reasonably obtainable, were estimated on bases reflecting the best currently available estimates and good faith judgments of the future competitiveness, operating and regulatory environments and related financial performance of MetroPCS and T-Mobile and will be realized in the amounts and at the times indicated thereby. Evercore expressed no view as to the estimated synergies or the assumptions on which they were based. With respect to the amount, timing and use of the tax attributes of the combined company estimated by management of MetroPCS, Evercore assumed that such estimates are reasonable and will be realized, although Evercore expressed no view as to such estimates or the assumptions on which they were based. For purposes of Evercore s analysis and opinion, at the request of the special committee, Evercore relied on the projections prepared by the management of MetroPCS with respect to projected financial and operating data of MetroPCS, including the estimated synergies and estimates as to the amount, timing and use of tax attributes of the combined company, and on the adjusted T-Mobile projected financial data and the adjusted T-Mobile projected operating data.

For purposes of rendering its opinion, Evercore assumed that the executed business combination agreement will be the same in all respects material to its analysis as the draft dated October 1, 2012 reviewed by Evercore, that the representations and warranties of each party contained in the business combination agreement are true and correct, that each party will perform all of the covenants and agreements required to be performed by it under the business combination agreement and that all conditions to the completion of the transaction will be satisfied without material waiver or modification thereof. Evercore also assumed with the consent of the special committee that the terms of the Deutsche Telekom notes and the working capital revolving credit facility (including any fees payable to Deutsche Telekom in connection therewith) are on terms no less favorable to MetroPCS than could be obtained from an unaffiliated third party lender. Evercore further assumed that all governmental, regulatory or other consents, approvals or releases necessary for the completion of the transaction will be obtained without any material delay, limitation, restriction or condition that would have an adverse effect on MetroPCS, T-Mobile or the completion of the business combination or materially reduce the benefits to MetroPCS of the transaction.

Evercore did not make or assume any responsibility for making any physical inspection, independent valuation or appraisal of the assets or liabilities of MetroPCS or T-Mobile, nor was Evercore furnished with any such valuation or appraisals, nor did Evercore evaluate the solvency or fair value of MetroPCS or T-Mobile under any state or federal laws relating to bankruptcy, insolvency or similar matters. In addition, Evercore assumed that the outcome of any current and pending litigation affecting MetroPCS or T-Mobile will not be material to its analysis. Evercore s opinion was necessarily based upon information made available to it as of the date of its delivery and financial, economic, market and other conditions as they existed and as could be evaluated as of such date. Evercore noted that subsequent developments may affect Evercore s opinion and that Evercore does not have any obligation to update, revise or reaffirm its opinion.

Evercore was not asked to pass upon, and expressed no opinion with respect to, any matter other than, assuming payment of the cash payment, whether MetroPCS issuance of the business combination consideration in consideration for the T-Mobile shares is fair, from a financial point of view, to MetroPCS and its stockholders (other than Deutsche Telekom and its affiliates). Evercore did not express any view on, and its opinion did not address, any other term or aspect of the business combination agreement or the transaction or any term or aspect of any other agreement or instrument contemplated by the business combination agreement or entered into or amended in connection with the transaction, including, without limitation, (i) the fairness of the reverse stock split or of the cash payment or any other consideration to be received in connection therewith by the holders of MetroPCS common stock or any other securities or creditors or constituencies of MetroPCS, (ii) the terms of the Deutsche Telekom notes and the working capital revolving credit facility (including the amount and nature of the fees payable to Deutsche Telekom in connection therewith) or (iii) the fairness of the amount or nature of any compensation to be paid or payable to any of the officers, directors or employees of MetroPCS, or any class of such persons, whether relative to the reverse stock split, the cash payment, the business combination consideration or otherwise.

Evercore assumed that any modification to the structure of the transaction would not vary in any respect material to its analysis. Evercore s opinion does not address the relative merits of the transaction as compared to other business or financial strategies that might be available to MetroPCS, nor does it address the underlying business decision of MetroPCS to engage in the business combination or any other portion of the transaction, including the reverse stock split and the cash payment. In arriving at its opinion, Evercore was not authorized to solicit, and did not solicit, interest from any third party with respect to the acquisition of any or all of the MetroPCS common stock or other extraordinary transaction involving MetroPCS. Evercore s opinion does not constitute a recommendation to the special committee, the MetroPCS board or to any other persons in respect of the business combination agreement or the transaction, including as to how any holder of shares of MetroPCS common stock should vote or act in respect of the transaction. Evercore expressed no opinion as to the price at which shares of MetroPCS common stock will trade at any time. Evercore is not legal, regulatory, accounting or tax experts and has assumed the accuracy and completeness of assessments by MetroPCS, T-Mobile and their respective advisors with respect to legal, regulatory, accounting and tax matters. The issuance by Evercore of its opinion was approved by an Opinion Committee of Evercore.

## Summary of Evercore s Financial Analysis

Set forth below is a summary of the material financial analyses conducted by Evercore and reviewed with the special committee on October 2, 2012, in connection with Evercore s opinion dated October 2, 2012.

Except as described above under the section entitled Opinion of the Financial Advisor to the Special Committee, the special committee imposed no instructions or limitations on Evercore with respect to the investigations made or the procedures followed by Evercore in rendering its opinion. Evercore s opinion was only one of many factors considered by the special committee in its evaluation of the transaction and should not be viewed as determinative of the views of the special committee, MetroPCS board or MetroPCS management with respect to the transaction or the business combination consideration. See the section entitled The Transaction Reasons for the Transaction; Recommendation of the MetroPCS Board and the Special Committee beginning on page 86.

The business combination consideration was determined through negotiations between MetroPCS, Deutsche Telekom, Holding and T-Mobile and was approved by the special committee and the MetroPCS board. Evercore did not recommend any specific business combination consideration nor did it indicate that any given business combination consideration constituted the only appropriate business combination consideration.

In connection with the review of the transaction by the special committee, Evercore performed a variety of financial and comparative analyses, which are summarized below, for purposes of rendering its opinion. The preparation of a fairness opinion is a complex process and is not necessarily susceptible to partial analysis or summary description. Selecting portions of the analyses or of the summary described below, without considering the analyses as a whole, could create an incomplete view of the processes underlying Evercore s opinion. In arriving at its determination, Evercore considered the results of all the analyses summarized below and did not draw, in isolation, conclusions from or with regard to any one analysis or factor considered by it for purposes of its opinion. Rather, Evercore made its determination as to fairness on the basis of its experience and professional judgment after considering the results of all the analyses. In addition, Evercore may have considered various assumptions more or less probable than other assumptions, so that the range of valuations resulting from any particular analysis described above should therefore not be taken to be Evercore s view of the value of MetroPCS or T-Mobile. No company used in the analyses summarized below as a comparison is identical to MetroPCS or T-Mobile, and no transaction used is identical to the transaction. Accordingly, such analyses may not necessarily utilize all companies or transactions and judgments concerning financial and operating characteristics and other factors that could affect the acquisition, public trading or other values of the companies or transactions used, including judgments and assumptions with regard to industry performance, general business, economic, market and financial conditions and other matters, many of which are beyond the control of MetroPCS, Deutsche Telekom or T–Mobile.

Evercore conducted the analysis summarized below for the purpose of providing an opinion to the special committee as to the fairness to MetroPCS and its stockholders (other than Deutsche Telekom and its affiliates), from a financial point of view, of MetroPCS issuance of the business combination consideration in consideration for the T-Mobile shares, assuming payment of the cash payment. This analysis does not purport to be appraisals or to necessarily reflect the prices at which the business or securities of MetroPCS or T-Mobile actually may trade or be sold. Estimates contained in these analyses are not necessarily indicative of actual future results, which may be significantly more or less favorable than suggested by such estimates. Accordingly, estimates used in, and the results derived from, the analyses summarized below are inherently subject to substantial uncertainty, and Evercore does not assume any responsibility if future results are materially different from those forecasted in such estimates.

Except as otherwise noted, the following quantitative information, to the extent that it is based on market data, is based on market data as it existed on or before October 2, 2012, and is not necessarily indicative of current or future market conditions.

The following summary of financial analyses includes information presented in tabular format. These tables alone do not constitute a complete description of the financial analyses and must be read together with the text of each summary in order to understand fully the financial analyses. Considering the tables below without considering the full narrative description of the financial analyses, including the methodologies and assumptions underlying the analyses, could create a misleading or incomplete view of such financial analyses.

In conducting its analysis, Evercore used various methodologies to review the valuation of each of MetroPCS and T-Mobile on a stand-alone basis and MetroPCS and T-Mobile on a relative basis, to assess the fairness of the business combination consideration to be issued by MetroPCS in consideration for the T-Mobile shares, assuming payment of the cash payment. Specifically, Evercore conducted analyses of historical share price, research analyst price targets, selected publicly traded companies, selected precedent transactions,

discounted cash flow, implied percentage ownership based upon selected publicly traded companies and discounted cash flow analyses, relative contribution to the combined company and has / gets analysis. However, Evercore only relied upon the analyses of implied percentage ownership based upon selected publicly traded companies and discounted cash flow analyses, relative contribution to the combined company and has / gets analysis for purposes of its opinion. For purposes of each valuation methodology applied to MetroPCS, Evercore assumed that MetroPCS will spend \$1.5 billion on spectrum in 2012. For purposes of each valuation methodology (other than the discounted cash flow analysis) applied to T-Mobile, Evercore included \$2.4 billion related to tower financing obligations in its calculation of net debt.

## Stand-Alone Valuation Analyses

Analysis of Selected Publicly Traded Companies Generally. Evercore compared certain financial and operating information and commonly used valuation measurements for MetroPCS and T-Mobile to corresponding information and measurements for a group of five publicly traded comparable companies that participate predominantly in the wireless communications industry, referred to as the selected comparable companies, in order to derive implied per share equity value reference ranges for MetroPCS and T-Mobile. The selected comparable companies were:

Sprint

U.S. Cellular

Leap Wireless

Atlantic Tele-Network

nTelos

Evercore selected the companies listed above because, based on Evercore professional judgment and experience in the wireless communications industry, their businesses and operating profiles are relevant to that of MetroPCS. However, because of the inherent differences between the businesses, operations and prospects of MetroPCS and the businesses, operations and prospects of the selected comparable companies, no comparable company is exactly the same as MetroPCS. Therefore, Evercore believed that it was inappropriate to, and therefore did not, rely solely on the quantitative results of the comparable company analysis. Accordingly, Evercore also made qualitative judgments (as more fully explained in the subsequent sentence) based on its experience in the wireless communications industry concerning differences between the financial and operating characteristics and prospects of MetroPCS and the selected comparable companies that would affect the public trading values of each in order to provide a context in which to consider the results of the quantitative analysis. These qualitative judgments related primarily to the differing sizes, capital structures, growth prospects, profitability levels and degrees of operational risk of MetroPCS and the selected comparable companies.

Evercore reviewed, among other things, enterprise values, calculated as equity value based on closing stock prices on September 26, 2012, plus debt, preferred stock and minority interests, less cash and cash equivalents and investments, as a multiple of the estimated earnings before interest, taxes, depreciation and amortization, which we refer to as EBITDA, of the selected comparable companies for calendar years 2012 and 2013. Financial forecasts for MetroPCS and T-Mobile were based on (i) certain non-public projected financial data relating to MetroPCS prepared and furnished to Evercore by the management of MetroPCS and (ii) certain non-public projected financial data relating to T-Mobile prepared by the management of Deutsche Telekom and T-Mobile furnished to Evercore by MetroPCS as adjusted at the direction of MetroPCS management. Evercore also used publicly available information concerning historical and projected financial performance for the selected comparable companies, including published historical financial information and publicly available third-party research.

*Analysis of Selected Publicly Traded Companies As Applied to MetroPCS.* Evercore then applied ranges of selected enterprise value to EBITDA (post stock-based compensation expense) multiples derived from the selected comparable companies (the mean and median of which was 5.3 and 5.5 respectively for 2012 Estimated EBITDA and 5.1 and 5.2 respectively for 2013 Estimated EBITDA) to corresponding financial data of MetroPCS based on certain non-public projected financial data relating to MetroPCS prepared and furnished to Evercore by the management of MetroPCS. The high and low of the relevant multiples derived for each of the selected comparable companies is reflected in the column of the chart titled All selected comparable companies. Based on the multiple ranges for the selected comparable companies, Evercore selected the range of multiples reflected in the column of the chart below titled Selected Valuation Multiple Ranges for MetroPCS, which Evercore judged, based on their financial advisory experience, to be most appropriate in order to perform their analysis of MetroPCS. This analysis indicated the following implied equity value per share valuation reference ranges for MetroPCS as presented below:

Valuation Methodology	All Selected Comparable Companies	Selected Valuation Multiple Ranges for MetroPCS	Implied Equity Valu per share Valuation Reference Range for MetroPCS
Enterprise Value as a Multiple of:			
2012 Estimated EBITDA	4.2 6.3x	5.0 6.5x	\$ 7.55 \$12.9
2013 Estimated EBITDA	4.2 5.6x	4.75 5.75x	\$ 7.01 \$10.7

*Analysis of Selected Publicly Traded Companies As Applied to T-Mobile.* In order to provide a basis for certain of the other analyses performed by Evercore and described below under Relative Valuation Analyses, Evercore then applied ranges of selected enterprise value to EBITDA multiples derived from the selected comparable companies (the mean and median of which was 5.3 and 5.5 respectively for 2012 Estimated EBITDA) to corresponding financial data of T-Mobile based on information and projections provided by T-Mobile and Deutsche Telekom management as adjusted by MetroPCS management. Based on the multiple ranges for the selected comparable companies (the high and low of such multiples being reflected in the column of the chart below titled All selected comparable companies ), Evercore selected the range of multiples reflected in the column of the chart below titled Selected Valuation Multiple Ranges for T-Mobile that Evercore judged, based on their financial advisory experience, to be most appropriate in order to perform their analysis of T-Mobile. This analysis indicated the following implied equity value reference ranges for T-Mobile:

Valuation Methodology	All Selected Comparable Companies	Selected Valuation Multiple Ranges for T-Mobile	Implied Equity Value Valuation Reference Ranges for T-Mobile (\$ in millions)
Enterprise Value as a Multiple of:			
2012 Estimated EBITDA	4.2 6.3x	5.0 6.5x	\$ 7,843 \$15,416
2013 Estimated EBITDA	4.2 5.6x	4.75 5.75x	\$ 6,979 \$12,111

None of the selected comparable companies utilized as a comparison is identical to MetroPCS or T-Mobile. Accordingly, Evercore believes the analysis of publicly traded comparable companies is not simply mathematical. Rather, it involves complex considerations and qualitative judgments, reflected in Evercore s opinion, concerning differences in financial and operating characteristics and other factors that could affect the public trading value of the selected comparable companies to which MetroPCS and T-Mobile are compared.

*Discounted Cash Flow Analysis of MetroPCS.* As part of its analysis, and in order to estimate the implied present value of the equity value per share for MetroPCS, Evercore prepared a discounted cash flow analysis for MetroPCS.

A discounted cash flow analysis is a valuation methodology used to derive a valuation of an asset by calculating the present value of estimated future cash flows to be generated by the asset. Present value refers to the current value of future cash flows or amounts and is obtained by discounting those future cash flows or amounts by a discount rate that takes into account macro-economic assumptions and estimates of risk, the opportunity cost of capital, expected returns and other appropriate factors. Evercore performed a discounted cash flow analysis for MetroPCS by adding (1) the present value of MetroPCS projected after-tax unlevered free cash flows for fiscal years 2013 through 2016 to (2) the present value of certain net operating losses of MetroPCS to (3) the present value of the terminal value of MetroPCS as of the end of fiscal year 2016. For each year, unlevered free cash flow was derived as follows: EBITDA plus certain non-cash adjustments less taxes less capital expenditures less changes in working capital, where changes in working capital can either be positive or negative. Terminal value refers to the value at a particular point in time of all future cash flows to be generated by an asset.

Evercore estimated a range of terminal values as of the end of fiscal year 2016 calculated based on multiples of 4.5x to 5.5x, which Evercore selected based on its professional judgment and experience in the wireless telecommunications industry. Evercore performed a discounted cash flow analysis using a range of discount rates from 8% to 9%. The discount rates were based on calculations of the weighted average cost of capital of MetroPCS, which Evercore derived based on MetroPCS estimated cost of equity (derived using the Capital Asset Pricing Model) and MetroPCS estimated cost of debt at an assumed target capital structure. Evercore calculated per share equity values by first determining a range of enterprise values of MetroPCS by adding the present values of the after-tax unlevered free cash flows, certain net operating losses and terminal values for each terminal value multiple and discount rate scenario, and then subtracting from the enterprise values the net debt, calculated as total debt minus cash and investments, of MetroPCS management, this analysis indicated the following implied per share equity value reference ranges for MetroPCS:

	Implied per S Referen	hare Equit ce Ranges f	•	
Valuation Methodology	Me	etroPCS		
Discounted Cash Flow Analysis	\$	13.07	\$17.59	
Elaw Anglusia of T Mahila In order to provide a basis for as	stain of the other englying	a manfammaa	d hy Erran	ana and

*Discounted Cash Flow Analysis of T-Mobile.* In order to provide a basis for certain of the other analyses performed by Evercore and described below under Relative Valuation Analyses, Evercore prepared a discounted cash flow analysis for T-Mobile of after-tax unlevered free cash flows for fiscal years 2013 through 2016, using projections provided by T-Mobile and Deutsche Telekom management as adjusted by MetroPCS management. Evercore performed a discounted cash flow analysis for T-Mobile by adding (1) the present value of T-Mobile s projected after-tax unlevered free cash flows for fiscal years 2013 through 2016 to (2) the present value of certain net operating losses of T-Mobile to (3) the present value of the terminal value of T-Mobile as of the end of fiscal year 2016. For each year, unlevered free cash flow was derived as follows: EBITDA less certain tower lease related adjustments less taxes less capital expenditures less changes in working capital, whereas changes in working capital can either be positive or negative.

Evercore estimated a range of terminal values as of the end of fiscal year 2016 calculated based on multiples of 4.5x 5.5x, which Evercore selected based on its professional judgment and experience in the wireless telecommunications industry. Evercore performed a discounted cash flow analysis using a range of discount rates from 8% to 9%. Evercore calculated equity values of T-Mobile by adding the present values of the after-tax unlevered free cash flows, certain net operating losses and terminal values for each terminal value multiple and discount rate scenario. Based on financial estimates provided by T-Mobile and Deutsche Telekom management as adjusted by MetroPCS management, this analysis indicated the following equity value reference ranges for T-Mobile:

	Equity Value Reference Ranges				
Valuation Methodology	for T-Mobile	e ( <b>\$ in m</b>	illions)		
Discounted Cash Flow Analysis	\$	12,635	\$17,891		

## **Relative Valuation Analyses**

*Implied Percentage Ownership Analysis.* Based on the implied valuations for each of MetroPCS and T-Mobile derived above under Analysis of Selected Publicly Traded Companies As Applied to MetroPCS, Analysis of Selected Publicly Traded Companies As Applied to T-Mobile, Discounted Cash Flow Analysis of MetroPCS and Discounted Cash Flow Analysis of T-Mobile, Evercore calculated an implied equity ownership range for MetroPCS in the combined company, which we refer to as the MetroPCS implied equity ownership range, and compared it to the proposed equity ownership for MetroPCS in the combined company to result from the transaction.

For each of the analyses referred to above, Evercore calculated the low end of each MetroPCS implied equity ownership range assuming the lowest implied per share equity value for MetroPCS and the highest implied equity value for T-Mobile derived from each of the foregoing valuation analyses. Evercore then calculated the high end of each MetroPCS implied equity ownership range assuming the highest implied per share equity value for MetroPCS and the lowest implied equity value for T-Mobile derived from each of the foregoing valuation analyses. Evercore also calculated the midpoint implied equity ownership assuming the midpoint implied per share equity value for both MetroPCS and T-Mobile derived from each of the foregoing analyses. In each of these calculations, Evercore assumed that MetroPCS had made the cash payment to its stockholders.

The proposed equity ownership percentage of holders of shares of MetroPCS common stock in the combined company after giving effect to the transaction was compared to MetroPCS implied equity ownership utilizing the non-public projected financial data relating to MetroPCS prepared and furnished to Evercore by the management of MetroPCS, as set forth below:

Method	MetroPCS Implied Equity Ownership Range	Equity Ownership	
Selected Publicly Traded Companies			
Analysis			
2012 EBITDA	7.6%	29.4%	16.3%
2013 EBITDA	8.1%	25.8%	15.5%
Discounted Cash Flow Analysis	15.8%	28.8%	21.7%

Evercore observed that the proposed aggregate equity ownership percentage of holders of shares of MetroPCS Common Stock in the combined company of 26% fell above or within each of the implied equity ownership ranges set forth above.

*Contribution Analysis.* Evercore analyzed the respective contributions of MetroPCS and T-Mobile to the revenue and EBITDA of the combined company for estimated fiscal year 2012 through estimated fiscal year 2014. This analysis did not include any estimated synergies and was based on (i) certain non-public projected financial data relating to MetroPCS prepared and furnished to Evercore by the management of MetroPCS and (ii) certain non-public projected financial data relating to T-Mobile prepared by the management of Deutsche Telekom and T-Mobile furnished to Evercore by MetroPCS as adjusted at the direction of MetroPCS management. For purposes of this analysis, Evercore assumed that the pro forma company is valued at a range of 2013 combined estimated EBITDA of 5.0x to 5.5x, which Evercore selected based on its professional judgment and experience in the wireless telecommunications industry. Evercore then calculated each of MetroPCS and T-Mobile s enterprise value based on their respective relative contribution of estimated revenue and EBITDA. Evercore subtracted net debt, calculated as total debt minus cash and cash equivalents, from enterprise value to derive the equity value for each of MetroPCS and T-Mobile. For MetroPCS, Evercore deducted the cash payment from MetroPCS calculated equity value. These equity values were used to calculate the implied equity ownership of MetroPCS in the combined company. This analysis was compared to the proposed aggregate 26% equity ownership in the combined company by current holders of shares of MetroPCS common stock assuming payment by MetroPCS to its stockholders of the cash payment and after giving effect to the business combination. The following table presents the results of this analysis:

# Implied MetroPCS Ownership

	2013E EBITDA Multiple				
Metric	5.0x	5.25x	5.50x		
Revenue					
2012	14.1%	15.1%	15.8%		
2013	20.0%	20.4%	20.6%		
2014	22.8%	22.9%	22.9%		
EBITDA					
2012	14.3%	15.3%	16.0%		
2013	14.6%	15.5%	16.2%		
2014	25.0%	24.9%	24.8%		

Evercore observed that the proposed aggregate 26% equity ownership of holders of shares of MetroPCS common stock in the combined company, assuming payment by MetroPCS to its stockholders of the cash payment and after giving effect to the business combination, was greater than any implied MetroPCS equity ownership derived from Evercore s contribution analysis set forth above.

## Has / Gets Analysis

Evercore also reviewed the following metrics for MetroPCS both on a stand-alone and pro forma basis:

Implied per share equity value based on discounted cash flow analysis

Implied per share equity value based on multiples analysis

With respect to the implied per share equity value based on discounted cash flow analysis, Evercore assumed an 8.5% weighted average cost of capital and 5.0x terminal multiple for both MetroPCS and T-Mobile, which were based on Evercore s professional judgment and experience in the wireless telecommunications industry. The pro-forma implied equity value per share was equal to the quotient obtained by dividing (A) the sum of (1) 26% multiplied by an amount equal to the sum of (i) MetroPCS stand-alone discounted cash flow implied equity value, plus (ii) T-Mobile s stand-alone discounted cash flow implied equity value, plus (iii) the present value of the estimated synergies, plus (iv) the after-tax value of the avoided spectrum cost assumed by MetroPCS in its stand-alone projections, less (v) the net present value of the loss of certain net operating losses as a result of the business combination, less (vi) estimated transaction expenses, less (vii) the cash payment, and

(2) the cash payment by (B) the fully diluted shares outstanding of MetroPCS common stock. For illustrative purposes, Evercore calculated pro forma equity value assuming three different scenarios: (1) no realization of the estimated synergies; (2) realization of 50% of the estimated synergies or net present value of \$3.3 billion; and (3) realization of 100% of the estimated synergies or net present value of \$6.7 billion.

With respect to the implied per share equity value based on multiples analysis, the stand-alone value is based on the share price as of September 25, 2012. The pro-forma implied equity value per share was equal to the quotient obtained by dividing (A) the sum of (1) 26% multiplied by an amount equal to the sum of (i) the pro forma equity value calculated by applying 5.25x multiple, which Evercore selected based on its professional judgment and experience in the wireless telecommunications industry, to the estimated pro forma 2013 EBITDA and deducting estimated net debt of both MetroPCS and T-Mobile, plus (ii) the present value of the estimated synergies, less (iii) the estimated transaction expenses, less (iv) the cash payment, and (2) the cash payment by (B) the fully diluted shares outstanding of MetroPCS common stock. For illustrative purposes, Evercore calculated pro forma equity value assuming three different scenarios: (1) no realization of estimated synergies; (2) realization of 50% of the estimated synergies or net present value of \$3.3 billion; and (3) realization of 100% of the estimated synergies or net present value of \$3.3 billion; and (3) realization of 100% of the estimated synergies or net present value of (a) weighted average cost of capital from 8.0% to 9.0% and (b) terminal value multiples from 4.5x to 5.5x, in the case of its discounted cash flow has / gets analysis, and (ii) a range of multiples from 4.75x to 5.75x, in the case of its discounted cash flow has / gets analysis, and (iii) a range of multiples from 4.75x to 5.75x, in the case of its discounted cash flow has / gets analysis, and (iii) a range of multiples from 4.75x to 5.75x, in the case of its discounted cash flow has / gets analysis, and (iii) a range of multiples from 4.75x to 5.75x, in the case of its discounted cash flow has / gets analysis, and (iii) a range of multiples from 4.75x to 5.75x, in the case of its discounted cash flow has / gets analysis, and (iii) a range of multiples from 4.75x to 5.75x, in the case of its d

The comparison of stand-alone implied per share equity values to the pro forma implied equity values using both discounted cash flow analysis and multiples analysis indicated that, based on the 26% ownership split provided in the business combination agreement, pro forma implied per share values were higher than stand-alone implied per share values.

Discounted Cash Flow Has / Gets Analysis

Valuation Matheolala au	Status	No Estin		S	n Estimated ynergies	Est	6.7bn imated
Valuation Methodology	Quo	Synerg	gies		(50%)	Synerg	ies (100%)
MetroPCS Share of Combined Company DCF							
plus Cash Payment	\$ 15.31	\$ 1	7.90	\$	20.11	\$	22.32
	Multiples Has /	Gets Analysi.	s				

Valuation Methodology	Status Ouo	Not Including Estimated Synergies	\$3.3bn Estimated Synergies (50%)		Estimated ies (100%)
Total Value per a Share of MetroPCS Common	Quo	5 yner gres	(50 %)	Synerg	100 / 0
Stock	\$ 11.25	\$ 12.83	\$ 15.10	\$	17.33
Other Information					

*Historical Share Price*. Evercore noted that the trailing low and high 52-week closing trading prices for shares of MetroPCS common stock, as of October 2, 2012, were \$5.59 per share and \$12.01 per share, respectively.

*Research Analyst Price Targets*. Evercore reviewed publicly available research analyst reports for MetroPCS published on or after July 25, 2012 that were available to it as of October 2, 2012. The table below sets forth (i) the name of each firm that published such a report, (ii) the date on which it issued such report, (iii) the price target for MetroPCS set forth in such report, if any, and (iv) if set forth in such report, the period of time set forth in such report for achieving such price target:

Firm Name	<b>Report Date</b>	Target Price (\$)	Target Period (Months)
JPMorgan	10/01/12	11.00	12
Credit Suisse	09/28/12	9.00	
Sanford C. Bernstein & Co	09/28/12	9.00	12
Guggenheim Securities LLC	09/28/12	8.25	12
Wells Fargo Securities, LLC	09/20/12		
Piper Jaffray	09/17/12	8.00	12
Macquarie	09/07/12	9.50	12
Oppenheimer & Co	09/07/12		
Canaccord Genuity Corp	08/29/12	12.00	12
BMO Capital Markets	08/22/12	10.00	
Robert W. Baird & Co	08/22/12	8.00	12
Nomura	08/22/12	6.00	12
EVA Dimensions	08/20/12		
Argus Research Corp	08/01/12		
FBR Capital Markets	07/30/12	11.00	12
Deutsche Bank	07/27/12	12.00	12
RBC Capital Markets	07/27/12	7.00	12
Jefferies	07/27/12	10.00	12
Raymond James	07/27/12		
Pivotal Research Group LLC	07/27/12	7.50	12
Goldman Sachs	07/26/12	10.00	12
Pacific Crest Securities	07/26/12		
William Blair & Co	07/26/12		
National Securities	07/26/12		
Barclays	07/25/12	8.00	

Note: Neither Evercore nor MetroPCS is adopting the foregoing price targets for any purpose by including such price targets in this proxy statement, and neither Evercore nor MetroPCS has expressed, and does not express, any view as to any such target prices or as to the manner or basis upon which they were derived by any research analyst. Evercore provided such price target information to the special committee for informational purposes only. Such information was not utilized or relied upon by Evercore for valuation purposes.

*Review of Selected Precedent Transactions.* Evercore reviewed the financial terms, to the extent publicly available, of nine merger and acquisition transactions announced between June 2008 and August 2012 of companies that Evercore, based on its professional experience with merger and acquisition transactions in the wireless telecommunications industry, deemed relevant. Evercore chose the transactions, referred to as the selected precedent transactions, based on the similarity of the target companies in the transactions to MetroPCS in terms of the size, mix, margins and other characteristics of their businesses. However, given the unique structure of the business combination, Evercore concluded that no single selected precedent transaction was truly representative of the transaction and, as a result, Evercore did not utilize its analysis of selected precedent transactions in arriving at its opinion. The selected precedent transactions were:

Acquirer AT&T, Inc.	<b>Target</b> NextWave Wireless Inc.	Date Transaction Announced August 2012
AT&T, Inc.	T-Mobile USA, Inc.	March 2011
Sprint Nextel Corporation	iPCS, Inc.	October 2009
Sprint Nextel Corporation	Virgin Mobile USA, Inc.	July 2009
Atlantic Tele-Network, Inc.	Verizon Wireless (Divestiture of Alltel Corporation Assets)	June 2009
AT&T, Inc.	Verizon Wireless (Divestiture of Alltel Corporation Assets)	May 2009
Verizon Communications Inc.	AT&T, Inc. (Divestiture of Centennial Communications Corp. Assets)	May 2009
AT&T, Inc.	Centennial Communications Corp.	November 2008
Verizon Communications Inc.	Alltel Corporation	June 2008

For each of the selected precedent transactions, Evercore calculated certain financial multiples for the target company derived from certain publicly available information for the target company. Specifically, in performing this analysis, Evercore determined the multiples of enterprise value to the forward twelve months of EBITDA (the mean of which was 5.8x and the median of which was 6.7x) for the selected precedent transactions. Evercore then used these multiples to estimate the implied per share equity value reference ranges for MetroPCS. Based on financial estimates provided by MetroPCS management, this analysis indicated the following implied per share equity value reference ranges for MetroPCS:

	Selected Valuation Multip	le Range	Implied Equity Value per share Valuation Reference Range for
Valuation Methodology	for MetroPCS	0	MetroPCS
Enterprise Value as a Multiple of:			
2013 Estimated EBITDA	5.00	7.00x	\$ 7.93 \$15.34
Because the reasons for, and the circumstances	surrounding, each of the selected preced	ent transactions	ns analyzed were so diverse, and because
of the inherent differences between the operati	ons and the financial condition of MetroF	CS and the con	mpanies involved in the selected precedent
transactions, Evercore believes that a comparat	ble transaction analysis is not simply mat	hematical. Rath	her, it involves complex considerations

transactions, Evercore believes that a comparable transaction analysis is not simply mathematical. Rather, it involves complex considerations and qualitative judgments, reflected in the opinion of Evercore, concerning differences between the characteristics of these transactions and the transaction that could affect the value of the subject companies and MetroPCS.

## General

Evercore is an internationally recognized investment banking firm and, as part of its investment banking activities, is regularly engaged in the valuation of businesses and their securities in connection with mergers and

acquisitions, negotiated underwritings, competitive bids, secondary distributions of listed and unlisted securities, private placements and valuations for corporate and other purposes. The special committee selected Evercore because of its expertise, reputation and experience in the wireless telecommunications industry generally and because its investment banking professionals have had substantial experience in transactions comparable to the business combination.

As provided under the terms of Evercore s engagement, MetroPCS has paid Evercore an aggregate fee of \$4.0 million, of which \$1.5 million was paid upon execution of Evercore s engagement letter, \$2.0 million was paid upon delivery by Evercore of its opinion without regard to the conclusions reached therein, and \$500,000 was paid as an additional discretionary amount that the special committee elected to pay to Evercore in its sole discretion. MetroPCS also agreed to reimburse Evercore s expenses and to indemnify Evercore against certain liabilities arising out of its engagement. During the two year period prior to the date of its opinion, no material relationship existed between Evercore and its affiliates and Deutsche Telekom, T-Mobile or MetroPCS, pursuant to which compensation was received by Evercore or its affiliates as a result of such a relationship. Evercore may provide financial or other services to Deutsche Telekom, T-Mobile or MetroPCS in the future and in connection with any such services Evercore may receive compensation.

In the ordinary course of business, Evercore or its affiliates may actively trade the securities, or related derivative securities, or financial instruments of MetroPCS, Deutsche Telekom or T–Mobile and their respective affiliates, for its own account and for the accounts of its customers and, accordingly, may at any time hold a long or short position in such securities or instruments.

## **Certain Unaudited MetroPCS Forecasts**

Although MetroPCS periodically may issue limited public guidance concerning its expected financial performance, MetroPCS does not as a matter of course publicly disclose detailed financial forecasts. However, in the course of due diligence in connection with the negotiation of the transaction, MetroPCS provided certain non-public financial forecasts for the years ending December 31, 2012, 2013, 2014 and 2015, which we refer to as the MetroPCS forecasts, to its financial advisors and to Deutsche Telekom and its financial advisors. The MetroPCS forecasts were prepared by MetroPCS management based upon certain internal financial forecasts that are based on reasonable expectations, beliefs, opinions and assumptions of MetroPCS management at the time they were made and are subjective. While the MetroPCS forecasts are being included in this proxy statement, the MetroPCS forecasts were not prepared with a view toward complying with published guidelines established by the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information. In the view of MetroPCS management, the MetroPCS forecasts were prepared on a reasonable basis and reflected the best then-currently available estimates and judgments of MetroPCS or any other recipient of this information considered, or now considers, this information to be necessarily predictive of actual future results, and does not constitute an admission or representation by any person that such information is material, or that the expectations, beliefs, opinions and assumptions that underlie such MetroPCS forecasts remain the same as of the date of this proxy statement, and readers are cautioned not to place undue reliance on the prospective financial information.

This prospective financial information has been prepared by, and is the responsibility of, MetroPCS management. Neither MetroPCS independent auditor nor any other independent accountant have examined, compiled, or performed any procedures with respect to the prospective financial information and, accordingly, none have expressed an opinion or any other form of assurance with respect thereto. MetroPCS independent auditor s report included in MetroPCS 2011 Annual Report on Form 10-K incorporated by reference herein relates to MetroPCS historical financial information and does not extend to the prospective financial information and should not be read to do so. The MetroPCS forecasts:

were based upon numerous expectations, beliefs, opinions and assumptions, as further described below, many of which are beyond the control of MetroPCS and may not prove to be accurate and that MetroPCS management at the time believed to be reasonable;

were originally prepared in the second quarter of 2012;

do not necessarily reflect current estimates or expectations, beliefs, opinions or assumptions management of MetroPCS may have about prospects for MetroPCS business, changes in general business or economic conditions, or any other transaction or event that has occurred or that may occur and that was not anticipated at the time the forecasts were prepared;

may not reflect estimates, beliefs, opinions or analysis about any other financial period and may not reflect current results, which may be significantly more favorable or less favorable than as set forth below; and

are not, and should not be regarded as, a representation that any of the expectations contained in, or forming a part of, the MetroPCS forecasts will be achieved.

Key assumptions underlying the MetroPCS forecasts include:

Strong revenue growth from 2012 to 2016, largely driven by return to net subscriber growth in 2013 and anticipated increase in ARPU driven by 4G LTE service plans. ARPU represents (a) service revenues less pass through charges for the measurement period, divided by (b) the sum of the average monthly number of customers during such period.

Return to EBITDA margin expansion from 2012 to 2016, largely driven by revenue growth, moderate reduction in churn and declines to cost per gross customer addition and cost per user per month due primarily to scale.

Acquisition of additional spectrum for \$1.5 billion in 2012.

The expected decline in capital expenditures in 2013 onwards due the expected completion of MetroPCS 4G/LTE network build-out.

All of the financial information contained in this section entitled Certain Unaudited MetroPCS Projections is forward-looking in nature. This information relates to multiple future years and such information by its nature becomes less predictive with each succeeding year. Actual future financial results may materially vary from the forward-looking information presented above. The financial information is based on assumptions, beliefs and opinions at the time made, which MetroPCS believed to be reasonable at that time.

The information above does not take into account any circumstances or events occurring after the respective dates on which it was prepared. MetroPCS does not intend to update or revise any of the information presented above. The information presented above constitutes forward-looking statements. For additional information on factors which may cause actual future financial results to materially vary from the information presented above, see the sections entitled Cautionary Note Regarding Forward-Looking Statements and Risk Factors beginning on pages 29 and 33, respectively.

## MetroPCS Forecasts Provided by MetroPCS

The following information was included in the MetroPCS forecasts:

	2012	2013	2014	2015
(\$ in millions)				
Service Revenue	\$ 4,549	\$ 4,746	\$ 5,047	\$ 5,366
EBITDA (1)	\$ 1,331	\$ 1,359	\$ 1,556	\$ 1,736
Capital Expenditures	\$ 990	\$ 930	\$ 892	\$ 855

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

(1) EBITDA is defined as consolidated net income plus depreciation and amortization; gain (loss) on disposal of assets; stock-based compensation expense; gain (loss) on extinguishment of debt; provision for income taxes; interest expense; minus interest and other income and non-cash items increasing consolidated net income.

## Possible Benefits of the Transaction

In connection with MetroPCS and Deutsche Telekom s respective evaluations of the transaction, representatives of MetroPCS and Deutsche Telekom discussed various potential benefits to MetroPCS of the transaction, including, among other things, potential capital expenditure and operating cost savings (including from reduced operating expenses relating to towers, backhaul and roaming, savings in capacity and expansion capital expenditures, the cost advantage of HSPA+ over CDMA, and savings from procurement and back office efficiencies and common platform efficiencies), potential synergies from expanding the business (including from leveraging technologies from both companies, leveraging market-development and other complementary business capabilities), potential tax synergies, potential operating efficiencies and other benefits.

MetroPCS has disclosed that it expects that the transaction could result in total cost synergies having a net present value of up to \$6 billion to \$7 billion. In particular, MetroPCS expects that the combined company will be able to realize cost synergies with an annual run rate of approximately \$1.2 billion to \$1.5 billion, including (i) approximately \$600 million to \$700 million in annual run rate savings resulting from reduced operating expenses relating to towers, backhaul and roaming, (ii) approximately \$400 million to \$500 million in annual run rate savings resulting from savings in capacity and expansion capital expenditures, (iii) approximately \$200 million to \$300 million in annual run rate savings resulting from the cost advantage of HSPA+ over CDMA and savings from procurement and back office efficiencies, and (iv) up to \$50 million in annual run rate savings resulting from common platform efficiencies. Both MetroPCS and Deutsche Telekom were aware that the amounts of any benefits to the combined company as a result of the transaction were estimates, that they may change, and that achieving any of the benefits would be subject to a number of uncertainties. For additional information related to limitations on these benefits, see the section entitled Risk Factors beginning on page 33.

#### **Certain Unaudited T-Mobile Forecasts**

Although Deutsche Telekom periodically may issue limited public guidance concerning T-Mobile s expected financial performance, Deutsche Telekom does not as a matter of course publicly disclose detailed financial forecasts for T-Mobile. However, in the course of due diligence in connection with the negotiation of the transaction, Deutsche Telekom provided certain non-public financial forecasts with respect to T-Mobile for the years ending December 31, 2012, 2013, 2014 and 2015, which we refer to as the T-Mobile forecasts, to its financial advisors and to MetroPCS and its financial advisors. The T-Mobile forecasts were prepared by Deutsche Telekom s and T-Mobile s management based upon certain internal financial forecasts that are based on reasonable expectations, beliefs, opinions and assumptions of Deutsche Telekom and T-Mobile forecasts were not prepared with a view toward complying with published guidelines established by the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information. In the view of Deutsche Telekom s and T-Mobile s management, the T-Mobile forecasts were prepared on a reasonable basis and reflected the best then-currently available estimates and judgments of Deutsche Telekom, T-Mobile, MetroPCS or any other recipient of this information considered, or now considers, this information to be necessarily predictive of actual future results, and does not constitute an admission or representation by any person that such information is material, or that the expectations, beliefs, opinions and assumptions that underlie such T-Mobile forecasts remain the same as of the date of this proxy statement, and readers are cautioned not to place undue reliance on the prospective financial information.

This prospective financial information has been prepared by, and is the responsibility of, Deutsche Telekom s and T-Mobile s management. Neither T-Mobile s independent auditor nor any other independent accountant have examined, compiled, or performed any procedures with respect to the prospective financial information and, accordingly, none have expressed an opinion or any other form of assurance with respect thereto. T-Mobile s independent auditors report included in this proxy statement relates to T-Mobile s historical

information. It does not extend to the prospective financial information and should not be read to do so. The T-Mobile forecasts:

were based upon numerous expectations, beliefs, opinions and assumptions, as further described below, many of which are beyond the control of Deutsche Telekom and T-Mobile and may not prove to be accurate;

were finalized on August 30, 2012;

do not necessarily reflect current estimates or expectations, beliefs, opinions or assumptions management of Deutsche Telekom and T-Mobile may have about prospects for T-Mobile s business, changes in general business or economic conditions, or any other transaction or event that has occurred or that may occur and that was not anticipated at the time the forecasts were prepared;

are not necessarily indicative of current values or future performance, which may be significantly more favorable or less favorable than as set forth below; and

are not, and should not be regarded as, a representation that any of the expectations contained in, or forming a part of, the T-Mobile forecasts will be achieved.

Key assumptions underlying the T-Mobile forecasts include:

Service revenues return to growth starting in 2014 driven by positive branded contract net customer additions by the end of 2013. Branded contract gross customer additions increases are driven by investment in the growth of the Business-to-Business segment, increases in customer volumes related to bring your own device programs enabled by spectrum refarming from T-Mobile s network modernization program and the launch of unlimited data offerings. Churn is expected to decline across all segments over the planning period as a result of network coverage improvements as part of the network modernization program, and continuing churn reduction program quality improvements, as well as a stronger phone line-up. Deeper investment in the Company s prepaid product line-up and ability to host Mobile Virtual Network Operators also contributes to service revenue growth.

EBITDA improvement trends begin in 2014, driven by growing contract customer revenue, the expansion of the T-Mobile USA Value plan across all channels which will reduce cost per gross addition, a decrease in churn from network modernization and other programs, and continued benefits from cost savings programs.

Capital expenditures will be elevated in 2012 and 2013 due to T-Mobile s network modernization program which has a target LTE deployment of 35,000 network sites by the end of the second quarter of 2013. Capital expenditures are expected to decrease in 2014 and 2015 as the network modernization program ends.

### T-Mobile Forecasts Provided by Deutsche Telekom

The following information was included in the T-Mobile forecasts:

2012	2013	2014	2015					
(in millions)								

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

Service Revenue	\$ 17,326	\$ 16,639	\$ 17,041	\$ 17,902
EBITDA Base Case (1)	\$ 4,919	\$ 4,559	\$ 5,084	\$ 5,500
EBITDA Adjusted (1),(2)	\$ 4,919	\$ 5,009	\$ 5,084	\$ 5,500
Capital Expenditures	\$ 3,297	\$ 3,085	\$ 2,485	\$ 2,185

- (1) EBITDA is a non-GAAP financial measure, which T-Mobile defines as earnings before interest, income taxes, depreciation, amortization and impairment charges. T-Mobile adjusts EBITDA to exclude transactions that are not reflective of T-Mobile s ongoing operating performance.
- (2) 2013 includes additional adjustments for an aggregate amount of \$450 million related to non-recurring costs and the effects of customer churn, marketing and cost management programs.

#### Adjustments to T-Mobile Forecasts by MetroPCS Management

MetroPCS management made certain adjustments to the T-Mobile forecasts provided by Deutsche Telekom based on its judgment and experience in the wireless communications industry. The following table sets forth such adjustments:

	2012	2013 (in mil	2014 lions)	2015
Service Revenue:				
From T-Mobile Forecast	\$ 17,326	\$ 16,639	\$17,041	\$17,902
Adjustment(1)	0	(338)	(495)	(318)
Service Revenue	\$ 17,326	\$ 16,301	\$ 16,546	\$ 17,584
EBITDA:				
From T-Mobile Forecast	\$ 4,919	\$ 4,559	\$ 5,084	\$ 5,500
Adjustment(2)	130	573	(165)	(134)
EBITDA Adjusted	\$ 5,049	\$ 5,132	\$ 4,919	\$ 5,366
Capital Expenditures:				
From T-Mobile Forecast	\$ 3,297	\$ 3,085	\$ 2,485	\$ 2,185
Adjustment(1)	0	(63)	(76)	(42)
Capital Expenditures	\$ 3,297	\$ 3,022	\$ 2,409	\$ 2,143

Note: All T-Mobile forecasts were provided by Deutsche Telekom, and all adjustments were made by MetroPCS management.

- (1) Service Revenue and Capital Expenditures were adjusted by MetroPCS management to reflect possible upside adjustments to T-Mobile s forecast of customer churn, customer acquisition costs and customer additions based on MetroPCS management s experience and discussions with T-Mobile and Deutsche Telekom.
- (2) EBITDA was adjusted by MetroPCS management to reflect (a) possible upside adjustments to customer churn, customer acquisition costs and customer additions based on MetroPCS management s experience and discussions with T-Mobile and Deutsche Telekom, (b) potential upside to the 2013 EBITDA forecast based on MetroPCS management s view of the performance of T-Mobile based on discussions with T-Mobile management, (c) the effects of GAAP adjustments related to the T-Mobile tower transaction and (d) certain non-recurring costs. MetroPCS believes that the information provided in this section and in the sections entitled Certain Unaudited MetroPCS Forecasts and Certain Unaudited T-Mobile Forecasts on pages 104 and 106, respectively, of this proxy statement is a materially accurate summation of the financial and operating data furnished to, and reviewed by, Evercore.

## Listing of MetroPCS Common Stock Issued in the Transaction

It is a condition to the completion of the transaction that the shares of MetroPCS common stock that will be issued to Holding or its designee in the transaction will have been approved for listing on the NYSE, subject to official notice of issuance.

## **Regulatory Approvals**

The parties have agreed to use their reasonable best efforts to obtain all authorizations, consents, orders or other approvals required from governmental entities to complete the transaction.

#### Federal Communications Commission

Under the Communications Act of 1934, as amended, which we refer to as the Communications Act, as a condition to, and before the completion of, the transaction, the FCC must approve the transfer of control of MetroPCS licenses and authorizations in connection with the transaction, which will result from 74% of the fully-diluted shares of MetroPCS common stock being owned by Deutsche Telekom following the completion of the transaction. In connection with such approval, the FCC must determine whether Deutsche Telekom is qualified to control MetroPCS licenses and authorizations and whether the transfer of control of such licenses is consistent with the public interest, convenience and necessity. Since Deutsche Telekom and Holding are not incorporated in the United States, the FCC must also issue a declaratory ruling pursuant to Section 310 of the Communications Act that the foreign ownership of MetroPCS is not inconsistent with the public interest. MetroPCS and T-Mobile filed transfer of control applications with the FCC on October 18, 2012.

#### United States Antitrust Laws

Under the HSR Act and the rules promulgated under that act by the FTC, the transaction may not be completed until notifications have been given and information furnished to the FTC and to the Antitrust Division, and the specified waiting period has been terminated or has expired without the commencement of a lawsuit. MetroPCS and T-Mobile each filed notification and report forms under the HSR Act with the FTC and the Antitrust Division on October 19, 2012. On November 19, 2012, MetroPCS and Deutsche Telekom received a request for additional information, which we refer to as the second request, from the Antitrust Division. The effect of the second request is to extend the waiting period imposed by the HSR Act until 30 days after MetroPCS and Deutsche Telekom have substantially complied with the second request, unless that period is extended voluntarily by the parties or terminated sooner by the Antitrust Division. At any time before or after completion of the transaction, the FTC or the Antitrust Division could act under the antitrust laws to prevent a substantial lessening of competition or the creation of a monopoly, including by seeking to enjoin completion of the transaction or seeking divestiture of substantial assets, businesses or product lines of MetroPCS or T-Mobile.

The transaction could also be the subject of challenges by private parties or state attorneys general under the state or federal antitrust laws.

## Committee on Foreign Investment in the United States

Under the Exon-Florio Amendment to the Defense Production Act of 1950, the President of the United States has the authority to investigate and, where necessary, suspend or prohibit any foreign acquisition, merger or takeover of companies engaged in U.S. interstate commerce or determined to threaten U.S. national security. By executive order, the President has delegated his investigatory powers under the Exon Florio Amendment to the CFIUS, an interagency committee chaired by the U.S. Treasury Department. Deutsche Telekom anticipates filing a voluntary notification of the transaction with the CFIUS in early 2013, seeking confirmation that the transaction contemplated by the business combination agreement does not threaten national security. The CFIUS has 30 days from the date of that filing to determine whether to pursue further investigation of the transaction.

## **Other Telecommunications Approvals**

MetroPCS and T-Mobile may make filings with state public utility commissions, domestic regulatory authorities, and, if required, foreign regulatory authorities in order to complete the transaction.

## Litigation Relating to the Transaction

Since the announcement on October 3, 2012 of the execution of the business combination agreement, MetroPCS, Deutsche Telekom, Global, Holding, T-Mobile (we refer to Deutsche Telekom, Global, Holding and

T-Mobile, collectively, herein as the T-Mobile defendants) and the members of the MetroPCS board, which we refer to as the MetroPCS board members, including an officer of MetroPCS, have been named as defendants in multiple putative stockholder derivative and class action complaints challenging the transaction.

As of the date of this proxy statement, the lawsuits include:

a putative class action lawsuit filed by Paul Benn, an alleged MetroPCS stockholder, on October 11, 2012 in the Delaware Court of Chancery, *Paul Benn v. MetroPCS Communications, Inc. et al.*, Case No. C.A. 7938-CS, which we refer to as the Benn action;

a putative class action lawsuit filed by Joseph Marino, an alleged MetroPCS stockholder, on October 11, 2012 in the Delaware Court of Chancery, *Joseph Marino v. MetroPCS Communications, Inc. et al.*, Case No. C.A. 7940-CS, which we refer to as the Marino action;

a putative class action lawsuit filed by Robert Picheny, an alleged MetroPCS stockholder, on October 22, 2012 in the Delaware Court of Chancery, *Robert Picheny v. MetroPCS Communications, Inc. et al.*, Case No. C.A. 7971-CS, which we refer to as the Picheny action;

a putative class action filed by James S. McLearie, an alleged MetroPCS stockholder, on November 5, 2012 in the Delaware Court of Chancery, *James McLearie v. MetroPCS Communications, Inc. et al.*, Case No. C.A. 8009-CS, which we refer to as the McLearie action, and together with the Benn action, the Marino action and the Picheny action, the Delaware actions;

a putative class action and shareholder derivative action filed by Adam Golovoy, an alleged MetroPCS stockholder, on October 10, 2012 in the Dallas, Texas County Court at Law, *Adam Golovoy et al. v. Deutsche Telekom et al.*, Cause No. CC-12-06144-A, which we refer to as the Golovoy action; and

a putative class action and shareholder derivative action filed by Nagendra Polu and Fred Lorquet, who are alleged MetroPCS stockholders, on October 10, 2012 in the Dallas, Texas County Court at Law, *Nagendra Polu et al. v. Deutsche Telekom et al.*,

Cause No. CC-12-06170-E, which we refer to as the Polu action, and together with the Golovoy action, the Texas actions. The various plaintiffs in the lawsuits allege that the individual defendants breached their fiduciary duties by, among other things, failing to (i) obtain sufficient value for the MetroPCS stockholders in the transaction, (ii) establish a process that adequately protected the interests of the MetroPCS stockholders, and (iii) adequately ensure that no conflicts of interest occurred. The plaintiffs also allege that the individual defendants breached their fiduciary duties by agreeing to certain terms in the business combination agreement that allegedly restricted the defendants ability to obtain a more favorable offer, including certain of the provisions described in the sections entitled Summary of the Business Combination Agreement No Solicitation of Alternative Proposals, Change in the MetroPCS Board s Recommendation and Effect of Termination beginning pages 142, 144 and 152, respectively, and that those provisions, together with the support agreement and rights agreement amendment described in the sections entitled Summary of Ancillary Agreements Voting and Support Agreement and Rights Agreement Amendment beginning on pages 158 and 160, respectively, constitute breaches of the individual defendants fiduciary duties. The plaintiffs seek injunctive relief, unspecified damages, an order rescinding the business combination agreement, unspecified punitive damages, attorney s fees, other expenses, and costs. All of the plaintiffs seek a determination that their alleged claims may be asserted on a class-wide basis. In addition, the plaintiffs in the Texas actions assert putative derivative claims, as stockholders on behalf of MetroPCS, against the individually named defendants for breach of fiduciary duty, abuse of control, gross mismanagement, unjust enrichment and corporate waste in connection with the transaction.

On November 5, 2012, the plaintiff in the Golovoy action filed a motion seeking to restrain and enjoin the MetroPCS and the MetroPCS board members, which we refer to collectively as the MetroPCS defendants, from complying with the force-the-vote provision in the business combination agreement and from declaring a distribution date under, or issuing rights certificates in conjunction with, MetroPCS rights agreement, which we

refer to as the Texas TRO motion. On November 12, 2012, the MetroPCS defendants filed a motion to dismiss or stay the Texas actions based on a mandatory forum selection provision in the MetroPCS bylaws, which requires that all derivative claims and all claims for breach of fiduciary duty against the MetroPCS board members must be filed and litigated only in the Delaware Court of Chancery, and sought dismissal for failure to plead standing to pursue derivative claims on behalf of MetroPCS.

On November 16, 2012, the trial court in the Golovoy action, which we refer to as the Texas trial court, issued a temporary restraining order, which we refer to as the TRO order, restraining the MetroPCS defendants from complying with the force the vote provision in the business combination agreement and from declaring a distribution date under, or issuing rights certificates in conjunction with, MetroPCS rights agreement, and set a temporary injunction hearing for November 29, 2012. On November 19, 2012, the MetroPCS defendants and the T-Mobile defendants filed a petition for writ of mandamus and a motion to stay, which we refer to as the Texas mandamus petition, with the Court of Appeals for the Fifth District at Dallas, which we refer to as the Texas appellate court, to stay and overturn the TRO order based on the mandatory forum selection provision in the MetroPCS bylaws, which requires that the claims in the Texas actions must be dismissed and pursued only in the Delaware Court of Chancery, and on a lack of evidence supporting the findings in the TRO order or establishing a basis for such TRO order, and to stay the temporary injunction hearing. On November 20, 2012, the Texas appellate court stayed the Texas trial court strial court stayed the scheduled temporary injunction hearing, and ordered briefing on the issues raised in the petition for writ of mandamus.

On November 28, 2012, the plaintiff in the Marino action filed an amended class action complaint alleging breach of fiduciary duty by the MetroPCS board members in connection with the terms of the business combination agreement, as well as alleging that MetroPCS has failed to make full and fair disclosure of all information and analyses presented to and considered by the MetroPCS board members, and alleging that the T-Mobile defendants aided and abetted such claimed breaches of fiduciary duty, and motions seeking expedited proceeding and discovery and to enjoin the defendants from taking any action to consummate the business combination between MetroPCS and the T-Mobile defendants. No hearing has been set on these motions. On November 30, 2012, all of Delaware actions were consolidated into a single action, now captioned *MetroPCS Communications, Inc. Shareholder Litigation*, Consolidated C.A. No. 7938-CS. The MetroPCS defendants have agreed to provide expedited discovery to the plaintiffs in the Marino action.

On January 8, 2013, the Texas appellate court conditionally granted the Texas mandamus petition and ordered the Texas trial court to vacate the TRO order, render an order denying the Texas TRO motion, and render an order granting the MetroPCS defendants and T-Mobile defendants motion to stay the action until MetroPCS defendants and T-Mobile defendants motion to dismiss or stay the action is decided by the Texas trial court. A hearing is currently set on such motion for January 25, 2013.

The MetroPCS defendants plan to defend vigorously against the claims made in the Delaware actions and the Texas actions.

#### SUMMARY OF THE FINANCING

The transaction is expected to be financed by the issuance by Wireless or T-Mobile of senior unsecured notes in an aggregate principal amount of up to \$18.5 billion as follows:

\$15.0 billion of senior unsecured notes, which we refer to as the \$15.0 billion notes, to be issued by T-Mobile and purchased by Deutsche Telekom to refinance certain intercompany indebtedness owed by T-Mobile and its subsidiaries to Deutsche Telekom and its subsidiaries (excluding T-Mobile and its subsidiaries);

\$2.5 billion of senior unsecured notes, which we refer to as the \$2.5 billion notes, which may be offered by Wireless to third-party investors and, to the extent not sold to third-party investors prior to the completion of the transaction, will be purchased by Deutsche Telekom upon the closing of the transaction, the proceeds of which we intend to use to refinance the Wireless existing senior credit facility; and

\$1.0 billion of senior unsecured notes, which we refer to as the \$1.0 billion notes, which may be offered by Wireless to third-party investors and, to the extent not sold to third-party investors prior to the completion of the transaction, will be purchased by Deutsche Telekom upon the closing of the transaction, the proceeds of which we intend to use for general corporate purposes.

In addition to the notes issued to finance the transaction, Deutsche Telekom (or one of its subsidiaries if the obligations of such subsidiary thereunder are unconditionally guaranteed by Deutsche Telekom) will make available for the benefit of T-Mobile and its subsidiaries, on the closing date of the transaction, an unsecured revolving credit facility with a maximum principal amount of no less than \$500 million to be used for working capital and other general corporate purposes, on terms substantially as set forth on Exhibit H to the business combination agreement, which we refer to as the working capital revolving credit facility.

Deutsche Telekom has committed, pursuant to the terms of the business combination agreement, to purchase (or to cause one or more of its subsidiaries to purchase) (i) the \$15.0 billion notes and (ii) any portion of the \$2.5 billion notes and the \$1.0 billion notes that are not sold to third-party investors. The economic terms, including the interest rate, the tenor, the no-call period and the redemption premium, of each series of notes to be purchased by Deutsche Telekom, which we refer to as the Deutsche Telekom notes, will be determined as set forth in the Deutsche Telekom notes pricing schedule, attached as Exhibit F to the business combination agreement. The proceeds of any issuances of debt securities by MetroPCS, or any of its subsidiaries, including Wireless, to third parties following the date of the business combination agreement will reduce the commitments of Deutsche Telekom in respect of the \$2.5 billion notes and the \$1.0 billion notes in the manner and to the extent described in the financing backstop terms and conditions schedule, attached as Exhibit I to the business combination agreement.

Each series of senior unsecured notes that may be offered by Wireless to third-party investors on or prior to the closing of the transaction, which we refer to as the permitted Wireless notes, will (a) have a maturity date of not less than seven and not more than twelve years from the date of issuance thereof, (b) have a call protection pricing schedule that is customary for high yield debt securities, (c) have a non-call period for permitted Wireless notes (i) with maturities of seven years, of not more than three years from the date of issuance, (ii) with maturities of greater than seven years and not greater than nine years, of not more than five years from the date of issuance, (iii) with maturities of greater than eleven years, of not more than five years from the date of issuance, and (iv) with maturities of greater than eleven years, of not more than six years from the date of issuance, (d) have an effective yield to maturity, at time of issuance thereof (taking into account any issuance fees, including underwriting fees, or original issue discount thereon), that is not greater than the initial yield that would be applicable to non-reset Deutsche Telekom notes of the same tenor, if such Deutsche Telekom notes were to be issued on the same date as such permitted Wireless notes, as calculated in accordance with the Deutsche Telekom notes pricing schedule, attached as Exhibit F to the business combination

agreement, (e) expressly permit the transaction and the related transactions (without the need to obtain any waiver, pay any fee, or make any offer to purchase), and (f) otherwise be on the terms set forth in the description of notes attached as Exhibit G to the business combination agreement; provided, further, that the proceeds of any permitted Wireless notes will be used solely as permitted in the business combination agreement.

On December 5, 2012, in connection with the consent solicitation and the entry by Wireless into the revised supplemental indentures, MetroPCS and Deutsche Telekom entered into a letter agreement, which we refer to as the Deutsche Telekom notes letter agreement, that, among other things, amended Exhibit G to the business combination agreement to reflect certain of the amendments to the covenants, events of default and other non-economic terms that were made applicable to the Wireless existing notes as a result of the consent solicitation. As a result, to the extent set forth in the Deutsche Telekom notes letter agreement, any notes issued by MetroPCS or its subsidiaries after the date of the business combination agreement that will be assumed by T-Mobile at the closing are required to have covenants, events of default and other non-economic terms that match those that were made applicable to the Wireless existing notes as a result of the consent solicitation (to the extent such revisions will continue to be effective as to the Wireless existing notes following the completion of the transaction).

# \$15.0 Billion Notes to Refinance T-Mobile-Deutsche Telekom Intercompany Indebtedness, and up to \$3.5 Billion Notes as Backstop for \$2.5 Billion Wireless Credit Agreement Refinancing and \$1.0 Billion Wireless New Notes

The \$15.0 billion notes will be issued by T-Mobile to Deutsche Telekom or a subsidiary of Deutsche Telekom pursuant to an indenture, which we refer to as the Deutsche Telekom notes indenture, containing the terms set forth in the description of notes attached as Exhibit G to the business combination agreement, as amended by the Deutsche Telekom notes letter agreement. The Deutsche Telekom notes will be unsecured. The Deutsche Telekom notes will be guaranteed by the combined company (the direct parent of T-Mobile following the transaction) and by all of T-Mobile s wholly-owned domestic restricted subsidiaries (other than immaterial subsidiaries), all of T-Mobile s restricted subsidiaries that guarantee certain of T-Mobile s indebtedness, and any future subsidiary of the combined company that directly or indirectly owns any of T-Mobile s equity interests.

The \$15.0 billion notes will have maturities ranging from six to eleven years. In addition, the \$15.0 billion notes will be divided into (i) six series of senior unsecured notes having interest rates that remain constant through maturity, which we refer to as the non-reset notes, and (ii) six series of senior unsecured notes, one-third of which will be re-priced every six months, beginning two years after the date of issuance and ending three years after the date of issuance, which we refer to as the reset notes. The no-call period with respect to each series of non-reset notes will range from four to six years after the issuance thereof. The no-call period with respect to each series of the \$15.0 billion notes will be in a principal amount of \$1.25 billion. In addition, to the extent Deutsche Telekom purchases the \$2.5 billion notes or the \$1.0 billion notes, which we refer to, collectively, as the additional notes (described further below), such notes will be divided equally into non-reset notes and reset notes, and will have maturities varying between three and eight years.

Both the reset notes and non-reset notes will be priced at the closing of the transaction, and the reset notes will be re-priced at the applicable time, according to a formula, the first component of which is a reference yield which is based upon (i) three indices of high-yield bonds issued by telecommunications companies (50% weight (or 2/3s weight, if qualifying securities of the type described in either (but not both) of the following clauses (ii) and (iii) are not available at the time of calculation), (ii) the prices of comparable bonds issued by Sprint Nextel Corporation or any successor or assign thereof (25% weight (or 1/3 weight, if qualifying securities of the type described in the following clause (iii) are not available at the time of calculation), (ii) the prices of the type described in the following clause the time of calculation), (ii) the prices of the type described in the following clause (iii) are not available at the time of calculation), (ii) the prices of the type described in the following clause the time of calculation or 2/3 weight, if qualifying securities of the type described in the following clause (iii) are not available at the time of calculation or 2/3 weight (or 1/3 weight, if qualifying securities of the type described in the following clause (iii) are not available at the time of calculation or zero weight if qualifying securities of the type described in this clause (ii) are not available at the time of calculation)) and (iii) the prices of Wireless securities (25% weight (or 1/3 weight, if qualifying securities))

of the type described in the previous clause (ii) are not available at the time of calculation or zero weight if qualifying securities of the type described in this clause (iii) are not available at the time of calculation)), all as of the applicable time (and provided that the yield of each index, bond or other qualifying security shall be increased (or decreased) for purposes of this calculation by 12.5 basis points per year, calculated to the day, by which the effective tenor of such index, bond or security (calculated as the tenor resulting in the yield to worst) is less than (or greater than) eight years. The reference yield will then be adjusted as follows: (1) *plus* 100 basis points for reset notes or 187.5 basis points for non-reset notes, (2) *plus* or *minus* 12.5 basis points per year, calculated to the day, by which the remaining tenor of the series of notes being repriced is longer or shorter than eight years; (3) *plus* a distribution fee of 200 basis points (spread in the coupon based upon the tenor of the applicable note).

By way of example, in the case of the \$15.0 billion notes, if the transaction closed on [] (based on applicable prices for the indices and securities described in clauses (i) through (iii) above assuming such a closing date):

The series of non-reset notes with the shortest tenor of six years would have a coupon of [ ]% per annum and a no-call period of two years, would be redeemable at a premium equal to half of the coupon in year three, at a premium of a quarter of the coupon in year four and would be redeemable at par in years five and six.

The series of non-reset notes with the longest tenor of eleven years would have a coupon of [ ]% per annum and a no-call period of five years, would be redeemable at a premium equal to half of the coupon in year six, at a premium of a third of the coupon in year seven, at a premium of a sixth of the coupon in year eight and would be redeemable at par in years nine through eleven.

The series of reset notes with the shortest tenor of six years would initially have a coupon of []% per annum, which coupon would reset on the second anniversary of the issuance thereof. This series of notes would be no-call through the reset date and for a period of two years thereafter, would be redeemable at a premium equal to half of the coupon in year five and would be redeemable at par in year six.

The series of reset notes with the longest tenor of eleven years would initially have a coupon of [ ]% per annum, which coupon would reset on the third anniversary of the issuance thereof. This series of notes would be no-call through the reset date and for a period of three years thereafter, would be redeemable at a premium equal to half of the coupon in year seven, at a premium of a quarter of the coupon in year eight and would be redeemable at par in years nine through eleven.

In the case of the additional notes, if the transaction closed on [ ] (based on applicable prices for the indices and securities described in clauses (i) through (iii) above assuming such a closing date):

The series of non-reset notes with the shortest tenor of three years would have a coupon of [ ]% per annum and would be no-call through maturity.

The series of non-reset notes with the longest tenor of eight years would have a coupon of []% per annum and a no-call period of four years, would be redeemable at a premium equal to half of the coupon in year five, at a premium of a quarter of the coupon in year six, and would be redeemable at par in years seven and eight.

The series of reset notes with the shortest tenor of three years would initially have a coupon of [ ]% per annum, which coupon would reset on the second anniversary of the issuance thereof (or, under some circumstances, depending on the amounts of additional notes issued, six months after the second anniversary of the issuance thereof, as described in Exhibit F to the business combination agreement). This series of notes would be no-call through maturity.

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

The series of reset notes with the longest tenor of eight years would initially have a coupon of [ ]% per annum, which coupon would reset on the second anniversary of the issuance thereof (or, under

some circumstances, depending on the amounts of additional notes issued, six months after the second anniversary of the issuance thereof, or on the third anniversary of the issuance thereof, as described in Exhibit F to the business combination agreement). This series of notes would be no-call through the reset date and for a period of three years thereafter (or, in the case of reset notes with an eight year tenor that reset on the third anniversary of the issuance thereof, for a period of two years thereafter), would be redeemable at a premium equal to half of the coupon in the first year after the end of the no-call, at a premium of a quarter of the coupon in the succeeding year, and would be redeemable at par thereafter.

The Deutsche Telekom notes indenture will contain customary events of default, covenants and other terms, including, among other things, covenants that restrict the ability of the issuer and its subsidiaries to, *inter alia*, pay dividends and make certain other restricted payments, incur indebtedness and issue preferred stock, create liens on assets, sell or otherwise dispose of assets, enter into transactions with affiliates and enter new lines of business, all as described in the description of notes attached as Exhibit G to the business combination agreement. These covenants include certain customary baskets, exceptions and incurrence-based ratio tests. The Deutsche Telekom notes indenture will not contain any financial maintenance covenants.

Pursuant to an agreement to be entered into by T-Mobile and Deutsche Telekom on the closing date and described on Exhibit J to the business combination agreement, Deutsche Telekom, as holder of the Deutsche Telekom notes, will have certain special rights, and will be subject to certain special restrictions, that do not apply to other holders of those notes, including among other things (i) a more broadly defined change in control put right, (ii) restrictions on its ability to tender shares into a change in control offer following a change in control resulting from a transfer of common stock of T-Mobile by Deutsche Telekom unless all holders of common stock are required or entitled to participate on the same terms, (iii) a right to consent to equity issuances the proceeds of which would be used to redeem notes held by Deutsche Telekom, and (iv) a right to consent to any redemption of the Deutsche Telekom notes held by Deutsche Telekom with the proceeds of any equity issuance by T-Mobile or the combined company.

#### \$2.5 Billion to Refinance Wireless Credit Agreement

It is currently contemplated that the \$2.5 billion notes will be issued and the proceeds will be used to pay off the approximately \$2.5 billion in principal amount of indebtedness, which we refer to as the credit agreement refinancing, currently owed by Wireless under the Third Amended and Restated Credit Agreement, dated as of March 17, 2011, among Wireless, as Borrower, the Guarantors party thereto, the Lenders from time to time parties thereto, and JPMorgan Chase Bank, N.A., as Administrative Agent, as modified by the Incremental Commitment Agreement, dated as of May 10, 2011 and as further amended and restated, supplemented or modified from time to time, which we refer to as the Wireless existing senior credit facility. Wireless, in consultation with Deutsche Telekom, is permitted to offer and sell permitted Wireless notes in an amount sufficient for the credit agreement refinancing at any time prior to the closing of the transaction. In the event that the credit agreement refinancing has not been consummated for the full principal amount of the Wireless existing senior credit facility indebtedness on or prior to the closing of the transaction, Deutsche Telekom will purchase additional Deutsche Telekom notes at the closing of the transaction in an amount necessary to repay the Wireless existing senior credit facility in full, up to a maximum amount of \$2.5 billion. Any additional Deutsche Telekom notes would be issued under the Deutsche Telekom notes indenture described above with the maturities and pricing described above.

Deutsche Telekom will be entitled to a commitment fee, payable by T-Mobile within one business day after the closing of the transaction, equal to 150 basis points of the \$2.5 billion Deutsche Telekom commitment amount; provided that T-Mobile, as the wholly-owned subsidiary of the combined company following the transaction, will be entitled to a fee reduction equal to (1) 100 basis points of the amount of Deutsche Telekom s commitment that is reduced with proceeds from the issuance of permitted Wireless notes, which we refer to as the take-out proceeds, within four and one-half months after pro forma financial statements giving effect to the

transaction are available (which pro forma financials shall be deemed to be available after this proxy statement is filed in definitive form with the SEC) or (2) 50 basis points of the amount of Deutsche Telekom s commitment that is reduced with take-out proceeds between four and one-half months and seven and one-half months following the availability of the pro forma financials.

#### \$1.0 Billion Wireless New Notes

In addition, the business combination agreement permits Wireless or its direct parent company, in consultation with Deutsche Telekom, to issue up to \$1.0 billion of additional permitted Wireless notes prior to the closing of the transaction. In the event that the \$1.0 billion notes are not sold to third party investors by the closing of the transaction or are sold to third party investors in an aggregate principal amount of less than \$1.0 billion, Deutsche Telekom will purchase additional Deutsche Telekom notes at the closing of the transaction in an amount equal to such shortfall. Any additional Deutsche Telekom notes would be issued under the Deutsche Telekom notes indenture described above with the maturities and pricing described above. It is currently anticipated that the proceeds of the \$1.0 billion notes will be used for general corporate purposes.

Deutsche Telekom will be entitled to a commitment fee, payable by T-Mobile within one business day after the closing of the transaction, equal to 150 basis points of the \$1.0 billion Deutsche Telekom commitment amount; provided that T-Mobile, as the wholly-owned subsidiary of the combined company following the transaction, will be entitled to a fee reduction equal to (1) 100 basis points of the amount of Deutsche Telekom s commitment that is reduced with take-out proceeds within nine months after the signing of the business combination agreement or (2) 50 basis points of the amount of Deutsche Telekom s commitment that is reduced with take-out proceeds between nine and twelve months after the signing of the business combination agreement.

#### \$500 Million Working Capital Revolving Credit Facility

The business combination agreement further provides that Deutsche Telekom or one of its subsidiaries will make the working capital revolving credit facility available to T-Mobile and its subsidiaries on the closing date of the transaction. The working capital revolving credit facility will be unsecured but guaranteed by the combined company and by all of T-Mobile s wholly-owned domestic restricted subsidiaries (other than immaterial subsidiaries), all of T-Mobile s restricted subsidiaries that guarantee certain of T-Mobile s indebtedness, and any future subsidiary of the combined company that directly or indirectly owns any of T-Mobile s equity interests.

The working capital revolving credit facility will have an availability period of up to five years. Borrowings under the working capital revolving credit facility will bear interest at a variable rate based on the London Interbank Offered Rate *plus* a spread of between 250 and 300 basis points, to be determined by reference to the borrower s debt-to-cash flow ratio. Also in connection with the working capital revolving credit facility, T-Mobile will pay Deutsche Telekom (1) an upfront fee equal to 50 basis points of the maximum principal amount of the working capital revolving credit facility, payable within one business day after the closing of the transaction, and (2) an unused commitment fee, payable quarterly, ranging from 25 to 50 basis points of any undrawn portion of the working capital revolving credit facility, to be determined by reference to the borrower s debt-to-cash flow ratio.

The working capital revolving credit facility will contain events of default, representations, warranties, covenants and other terms that are customary and substantially consistent with the Wireless existing senior credit facility, with changes to reflect certain baskets and exceptions and other terms provided in the Deutsche Telekom notes indenture. The working capital revolving credit facility will also include a financial covenant requiring that the borrower s debt-to-cash flow ratio shall not exceed 4.0 to 1.0, which applies as a condition to borrowing (tested at the time of the borrowing giving pro forma effect to the borrowing) and at any time there are borrowings outstanding (tested on a quarterly basis).

#### Consent Solicitation for Amendments to Wireless Existing Notes

On December 5, 2012, Wireless commenced the consent solicitation, seeking to amend the indentures governing each series of the Wireless existing notes. On December 14, 2012, following the receipt of the requisite consents in the consent solicitation, Wireless, the guarantors named therein and the trustee entered into revised supplemental indentures that now govern the Wireless existing notes. Among other things, the revised supplemental indentures modified the definition of Change in Control so that the consummation of the transaction will not be considered a change in control under the indentures governing the Wireless existing notes. The revised supplemental indentures also conformed the covenants, events of default and other non-economic terms previously applicable to the Wireless existing notes to certain covenants, events of default and other non-economic terms of the function of the transaction, but that will apply only until such time, if any, as such notes are assumed by T-Mobile upon the consummation of the transaction, but that will be permanent if the transaction is not consummated. As required under the terms of the business combination agreement, Deutsche Telekom consented to the proposed amendments to the indentures governing the Wireless existing notes.

As a result of the consummation of the consent solicitation and the entry into the revised supplemental indentures relating to the Wireless existing notes, Deutsche Telekom s commitment, pursuant to the business combination agreement, to purchase additional notes in an amount sufficient to satisfy such change of control obligations, has been terminated, and the commitment fee of 150 basis points of the \$2.0 billion Deutsche Telekom commitment with respect to the Wireless existing notes has been reduced by 100 basis points and the remaining 50 basis points will be paid by T-Mobile at the closing of the transaction.

## INTERESTS OF METROPCS DIRECTORS AND OFFICERS IN THE TRANSACTION

In considering the recommendation of the MetroPCS board that you vote to approve the stock issuance proposal and the new certificate of incorporation proposal, you should be aware that some of MetroPCS directors and officers have interests in the transaction that are different from, or in addition to, those of MetroPCS stockholders generally. The MetroPCS board and the special committee were aware of and considered these potential interests, among other matters, in evaluating the business combination agreement and the transaction, and in recommending to you that you approve the MetroPCS stock issuance and the amendment and restatement of MetroPCS certificate of incorporation in connection with the transaction.

#### **Certain Director Relationships**

James N. Perry, Jr., a member of the MetroPCS board, is a managing director of Madison Dearborn Partners, LLC, a private equity firm, which is an affiliate of Madison Dearborn, one of MetroPCS greater than 5% stockholders, and a general partner of various investment funds affiliated with Madison Dearborn. Madison Dearborn entered into the support agreement with Deutsche Telekom relating to the transaction, which is described in Summary of the Ancillary Agreements Voting and Support Agreement beginning on page 158.

#### **Positions after Completion of the Transaction**

Following the completion of the transaction, two existing members of the MetroPCS board, who are expected to be James N. Perry, Jr. and John (Jack) F. Callahan, Jr., will continue to be directors of MetroPCS, and it is anticipated that certain of the executive officers of MetroPCS will continue to be executive officers of the combined company, as described under the section entitled Board of Directors and Management after the Transaction beginning on page 126. Any executive officer, including the named executive officers, who is not identified prior to the closing of the transaction to be an executive officer of the combined company following the closing will not be deemed to be an executive officer of the combined company following the closing.

For at least one year after the completion of the transaction, the combined company will, and Deutsche Telekom will cause the combined company to, provide any of the combined company s continuing employees with compensation (other than equity compensation), severance pay and employee benefits that are substantially comparable, in the aggregate, to the existing compensation (other than equity compensation), severance pay and benefits currently provided by MetroPCS. In addition, each continuing employee of the combined company will receive service credit under any new combined company benefit plans to the extent credited under MetroPCS existing benefit plans for all purposes of determining eligibility to participate, vesting and level of benefits, including for purposes of vacation, severance and paid time off benefits (but not for benefit accrual purposes, for the purpose of qualifying a subsidized early retirement benefit, or if such credit would result in a duplication of benefits). After the closing, the combined company agreed to use its reasonable best efforts to make a decision whether to continue the employment of any combined company employee and communicate such decision within one year of closing to such combined company employee. The business combination agreement further provides that employees are not third-party beneficiaries of such agreement.

#### Severance Pay Plan

MetroPCS board previously adopted a severance pay plan, which we refer to as the severance plan, that provides for severance benefits to all officers of MetroPCS, including its executive officers. Eligible officers will not receive any compensation under the severance plan solely due to the completion of the transaction.

The severance plan provides that an eligible officer listed in the table below will be entitled to a severance payment in accordance with the guidelines indicated below if his or her employment is terminated by MetroPCS without cause or by the eligible officer for good reason.

Tier	Position	Severance Payment	Severance Period
Tier 1	Chief Executive Officer	2.0 times Annual Compensation (as defined below)	24 months
		+ Pro-Rata Additional Payment	
Tier 2	President and Chief Operating Officer, Vice Chairmen, Executive Vice Presidents, Senior Vice Presidents,	1.5 times Annual Compensation	18 months
	and Vice Presidents who report directly to the Chief Executive Officer	+ Pro-Rata Additional Payment	
Tier 3	All other Vice Presidents who are officers of MetroPCS, with the exception of those Vice Presidents who report directly to the Chief	.75 times Annual Compensation	9 months
	Executive Officer (and not including staff vice presidents or regional vice presidents)	+ Pro-Rata Additional Payment	

For purposes of the calculation of the severance payments, Annual Compensation is an amount equal to the annualized base salary for the eligible officer, plus the eligible officer s Pro-Rata Additional Payment, which is the product, pro-rated for the number of days the eligible officer was employed by MetroPCS or an affiliate of MetroPCS during the calendar year in which his or her employment was severed, of the percentage set forth below for the eligible officer multiplied by the eligible officer s annualized base salary. Additionally, the eligible officer would receive a one-time payment equal to the eligible officer s Pro-Rata Additional Payment attributable to the year in which the termination of employment occurs.

Tier	Position	Additional Payment (percentage of annualized base salary)
Tier 1	Chief Executive Officer	140%
Tier 2	President and Chief Operating Officer Vice Chairman, Chief Financial Officer	90% 80%
	Vice Chairman, General Counsel & Secretary	75%
	Senior Vice Presidents	65%
	Vice Presidents who report directly to the Chief Executive Officer	40%
	Vice President, Regional General Manager	50%
Tier 3	Vice Presidents other than Regional General Managers and Vice Presidents who report directly to the Chief Executive Officer	40%

For a Tier 1, Tier 2 or Tier 3 officer to be deemed an eligible officer and receive the benefits described above in full, the officer must have been continuously employed by MetroPCS or an affiliate of MetroPCS for a period of two or more years following the officer s hire date. In certain circumstances, new officers and other officers who have not been in continuous service with MetroPCS for a period of two years will be entitled to a pro-rata portion of the severance payments described above based on the officer s length of service with MetroPCS.

In addition to the severance payments described above, an eligible officer will also be entitled to an amount equal to the sum of all accrued and unpaid salary as of the date of termination, any reimbursement of business

expenses incurred prior to termination, any accrued vacation pay not paid, any vested and unpaid annual cash performance awards and any vested or accrued other benefits, plus a reimbursement in an amount equal to the aggregate applicable monthly premiums required for the eligible officer and his or her dependents to continue health coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985, which we refer to as COBRA, during the eligible officer s applicable severance period as set forth in the chart above.

Under the terms of the severance plan, the severance payments and the COBRA reimbursements will be paid monthly in substantially equal increments in accordance with MetroPCS normal payroll practices during the eligible officer s applicable severance period as set forth in the chart above. The first such monthly payment will be made on the 60th day following the eligible officer s termination of employment. The payment of the severance payments and the COBRA reimbursements is conditioned upon the eligible officer s execution and delivery of a customary release agreement in favor of MetroPCS and certain affiliated parties. In addition to customary release language, the release agreement includes a non-compete provision applicable for a period equal to the eligible officer s applicable severance period. If the eligible officer elects to accept employment with a listed MetroPCS competitor or otherwise breaches the non-compete provisions of the release agreement, MetroPCS may stop making severance payments and reimbursing for COBRA coverage.

Upon a termination by MetroPCS without cause or by the eligible officer for good reason, the eligible officer s outstanding awards under MetroPCS equity incentive compensation plans will receive the following treatment:

All unvested stock option awards granted pursuant to the Second Amended and Restated MetroPCS, Inc. 1995 Stock Option Plan, as amended, which we refer to as the 1995 Plan, the Amended and Restated MetroPCS Communications, Inc. 2004 Equity Incentive Compensation Plan, which we refer to as the 2004 Plan, and the MetroPCS Communications, Inc. 2010 Equity Incentive Compensation Plan, which we refer to as the 2010 Plan and which we refer to, collectively with the 1995 Plan and the 2004 Plan, as the MetroPCS equity incentive compensation plans, will be immediately forfeited without any further payment;

All vested but unexercised stock option awards granted pursuant to the 1995 Plan shall remain exercisable by the eligible officer for a period of three months following the eligible officer s termination of employment;

All vested but unexercised stock option awards granted pursuant to the 2004 Plan or the 2010 Plan shall remain exercisable by the eligible officer for a period of six months following termination of employment;

All unvested restricted stock granted pursuant to the MetroPCS equity incentive compensation plans shall be immediately forfeited without further payment;

Any unvested annual performance awards granted pursuant to the MetroPCS equity incentive compensation plans shall be immediately forfeited without further payment; and

All other awards under the MetroPCS equity incentive compensation plans shall be immediately forfeited without further payment.

#### **Change in Control Agreements**

In 2010, MetroPCS entered into change in control agreements that provide for payments to be made to certain key officers, including all of the MetroPCS named executive officers, whose employment is terminated by MetroPCS without cause or as a condition to the completion of a change in control transaction, or by the employee for good reason, during the 18 month period following a change in control (which includes the completion of the transaction), which we refer to as the protection period. The benefits payable to any executive officer under the change in control agreements are in lieu of any other payments or benefits payable under any other severance plan, policy or arrangement maintained by MetroPCS, including the severance pay plan described above.

Executive officers will generally be entitled to the following severance benefits if their employment is terminated during the protection period by MetroPCS or the combined company without cause or by the executive officers for good reason:

Position	Lump Sum Severance Payment	Benefits Coverage Period
Chief Executive Officer	2.5 times annual base salary + Bonus (as defined in the Change in Control Agreement)	30 months
Vice Chairman, Executive Vice Presidents, Senior Vice Presidents, and Vice Presidents who report directly to the Chief Executive Officer	2 times annual base salary + Bonus	24 months
All other Vice Presidents	1 times annual base salary + Bonus	12 months

In addition to the lump-sum severance payment described above, each officer and his or her dependents will be entitled to health and dental benefits coverage for the stated severance benefit period following termination and any amounts owed to such officer as of the date of termination, including among other things, accrued and unpaid salary, reimbursement of expenses, accrued vacation pay and any pro-rata portion of his or her annual cash incentive award for the year in which he or she was terminated. The change in control agreements also provide that any payments made pursuant to such agreements will be reduced to the minimum extent necessary (but not below zero) to avoid the characterization of any such payments (together with any other payments or benefits payable under plans, programs, or agreements of MetroPCS) as excess parachute payments under Sections 280G of the Code. The severance benefits are payable on the 60 day following the officer s termination and are conditioned upon the officer s execution and delivery of a customary release agreement in favor of MetroPCS and certain affiliated parties. In addition to customary release language, the release agreement includes a non-compete provision applicable for a certain period following termination. If the officer elects to accept employment with a listed competitor of MetroPCS or otherwise breaches the non-compete provisions, the officer will be required to re-pay certain of the severance benefits payable under the change in control agreement and MetroPCS shall be entitled to cease providing any health or dental benefit coverage on a prospective basis.

In addition to the cash payment and benefits describe above, any outstanding equity awards and incentive compensation awards held by the officer under any MetroPCS equity incentive compensation plans will immediately vest and become exercisable upon the closing of the transaction (without regard to any termination of employment). In addition, any annual cash performance awards attributable to each officer will immediately vest and be deemed earned in full at the target level as of the date of the completion of the transaction without regard to any applicable performance cycle, restriction or condition being completed or satisfied or without regard to any termination of employment. Such vesting and payment is not conditioned on a termination of employment after the completion of the transaction.

#### **Retention Agreements**

The business combination agreement provides that, during the period between the date of the business combination agreement and the closing of the transaction, MetroPCS may, subject to certain conditions, enter into retention agreements with officer and non-officer employees in an aggregate amount not to exceed \$25 million. Neither the overall size of these retention payments nor the allocations to particular participants has been determined. Under the business combination agreement, any retention payment to any officer, including a named executive officer, is subject to Deutsche Telekom s consent.

#### **Equity Incentive Compensation Plans**

The business combination agreement provides that during the period between the date of the business combination agreement and the closing of the transaction, MetroPCS may, subject to certain conditions, make (a) grants of stock options or restricted stock under the MetroPCS equity incentive compensation plans to officers and employees of MetroPCS and its subsidiaries hired or promoted in each case after the date of the business combination agreement consistent with past practice and in the ordinary course of business and (b) annual grants of stock options and restricted stock under the MetroPCS equity incentive compensation plans to officers, employees and directors in such amounts as are consistent with past practice and in the ordinary course of business provided that the total amount of such awards made pursuant to clause (a) and (b) does not exceed 5,750,000 shares of MetroPCS common stock.

Effective as of the completion of the MetroPCS stock issuance, all outstanding annual cash performance and equity awards under MetroPCS equity incentive compensation plans, including each outstanding stock option and each share of restricted stock, will automatically vest and (in the case of stock options) become exercisable and will remain outstanding. The annual cash performance awards will be paid upon the completion of the transaction at the target payment amount for such award.

In addition, except as provided below, at the time the amendment of MetroPCS certificate of incorporation becomes effective, each outstanding option to acquire MetroPCS common stock will be adjusted to take into account the reverse stock split and the cash payment. The number of shares of MetroPCS common stock to be acquired pursuant to outstanding options will be reduced to reflect the reverse stock split and the exercise price of the options will be increased to reflect the reverse stock split, with the per share amount of the cash payment deducted from the options adjusted per-share exercise price. Restricted stock will be adjusted to account for the reverse stock split in the same manner as all other MetroPCS common stock, and holders of restricted stock will share in the cash payment.

Notwithstanding and in lieu of the above, holders of stock options with an exercise price that is less than the average closing price of MetroPCS common stock based on a five-day trading average before the closing ignoring any market effect of the reverse stock split and cash payment, which we refer to as in-the-money stock options, may elect to receive cash in lieu of their in-the-money stock options during the five days following the closing of the transaction at a price per share equal to the average closing price of MetroPCS common stock based on a five-day trading average before the closing ignoring any market effect of the reverse stock split and cash payment minus the exercise price of the option, less required tax withholding. Any in-the-money stock options issued under the 1995 Plan that have an exercise price equal to or less than the per share amount of the cash payment (or approximately \$4.08 per share calculated on a pre-reverse stock split basis), which we refer to as low exercise price stock options, will be automatically cashed-out in the same manner. Any stock options that are not cashed out at the closing of the transaction, including any in-the-money stock options (other than low exercise price stock options) for which the holder thereof does not elect to receive cash at closing, will be adjusted for the reverse stock split and the per share amount of the cash payment as described above and will remain outstanding, 100% vested and exercisable in accordance with its terms. Holders of stock options will not receive any portion of the cash payment.

#### Director and Officer Indemnification and Insurance

From and after the completion of the transaction, the combined company will assume and honor the obligations of MetroPCS with respect to all rights to indemnification and exculpation from liabilities, including advancement of expenses, for acts or omissions occurring at or prior to the completion of the transaction now existing in favor of the current or former directors or officers of MetroPCS. In addition, MetroPCS will purchase directors and officers liability insurance policies for all current and former MetroPCS directors and officers for a period of six years following the closing of the transaction for events arising at or prior to the closing of transaction, subject to certain limitations on the amount of premiums payable under such policies.

#### Quantification of Change in Control and Termination Payments and Benefits to the MetroPCS Named Executive Officers

The following table sets forth the amount of payments and benefits that each MetroPCS named executive officer would receive in connection with or otherwise related to the transaction, assuming the completion of the transaction occurred on October 31, 2012 and (except as provided otherwise in the footnotes to the table) the employment of the named executive officer was terminated other than for cause or the named executive officer resigned for good reason, in each case on such date, and the named executive officer received no additional annual cash performance awards or additional stock options or shares of restricted stock on or after October 31, 2012. The actual payment may vary significantly depending upon the date the transaction closes. The payments and benefits are subject to a non-binding, advisory vote of MetroPCS stockholders. For additional details regarding the terms of the payments quantified below, see the section entitled Interests of MetroPCS Directors and Officers in the Transaction beginning on page 118.

1	2	3
	-	~

## CHANGE IN CONTROL AND TERMINATION COMPENSATION

				Perquisites /	Tax		
Named Executive			Pension/	Benefits	<b>Reimburse-</b>		
Officer	Cash (1)	Equity (2)	NQDC	(3)	ment	Other (4)	Total (5)
Roger D. Linquist	\$ 8,180,561	\$18,727,307	\$	\$ 30,764	\$	\$	\$ 26,938,632
Thomas C. Keys	3,204,360	5,069,332		36,570			8,310,262
J. Braxton Carter	2,740,870	3,812,593		36,122			6,589,585
Mark A. Stachiw	2,149,725	4,243,735		36,570			6,430,030
Dennis T. Currier	1,392,325	921,377		36,570			2,350,272

(1) As described above, this amount represents the double trigger lump-sum cash severance payment that the named executive officer would be entitled to receive under the change in control agreements following a qualifying termination of employment after the completion of the transaction that includes the pro-rata portion of the named executive officer s annual cash incentive award for the year assuming both the change in control occurred and the officer was terminated on October 31, 2012. In addition, this amount also includes the single trigger annual cash performance award attributable to the named executive officer that will immediately vest and be deemed earned in full at the applicable target level as of the date the change in control occurred. Each named executive officer would also be paid any amounts earned or vested and unpaid to him as of the date of termination, including among other things, accrued and unpaid salary, reimbursement of business expenses, and accrued vacation pay, which amounts are not reflected on the table above. This amount also assumes that all payments and benefits under the change in control agreements and other plans, programs or arrangements of MetroPCS would not be deemed to be excess parachute payments under Sections 280G of the Code, in which case the payments under the change in control agreements are necessary to avoid such characterization.
Single Trigger Annual Cash Performance Award

Named Executive Officer	Cash
Roger D. Linquist	\$ 1,337,000
Thomas C. Keys	529,200
J. Braxton Carter	432,800
Mark A. Stachiw	330,750
Dennis T. Currier	201,500

(2) As described above, this amount equals the value of all stock options to purchase MetroPCS common stock and unvested shares of restricted stock held by the executive as of October 31, 2012 that will automatically vest on a single trigger basis upon completion of the transaction (without any adjustment to reflect the reverse stock split or cash payment). Stock options are valued at the excess of the price per share of MetroPCS common stock over the applicable exercise price (ignoring any market effect of the reverse stock split and cash payment). For purposes of these calculations because the price per share that MetroPCS stockholders will receive is not a fixed dollar amount, MetroPCS has (as required under Item 402(t) of Regulation S-K) used the average closing price per share of MetroPCS common stock over the five business days following the public announcement of the transaction on October 3, 2012, which was \$12.49. In addition to the stock options and unvested restricted stock described above and below, the business combination agreement authorizes MetroPCS to make additional awards of stock options and restricted stock, including annual awards, prior to the closing of the transaction in such amounts as are consistent with past practice and the ordinary course of business not to exceed 5,750,000 shares of MetroPCS common stock in the aggregate for all officers and employees of MetroPCS. Any such future awards will also automatically vest on a single trigger basis upon the completion of the transaction. No such awards have yet been granted to any named executive officer, and thus are not reflected in the above or below tables.

#### Single Trigger Vesting

		Restricted MetroPCS		
Named Executive Officers	Stock Options	Common Stock	Total	
Roger D. Linquist	\$ 12,795,856	\$ 5,931,451	\$ 18,727,307	
Thomas C. Keys	1,846,650	3,222,682	5,069,332	
J. Braxton Carter	1,310,684	2,501,909	3,812,593	
Mark A. Stachiw	2,734,518	1,509,217	4,243,735	
Dennis T. Currier	205,800	715,577	921,377	

- (3) As described above, this amount represents the value of the double trigger health and welfare continuation benefits provided to each executive under the terms of the change in control agreements upon a qualifying termination of employment. For purposes of calculating these amounts 2013 COBRA rates were used.
- (4) As described above, MetroPCS may, subject to the satisfaction of certain conditions, make retention payments up to an aggregate \$25 million to employees, including the named executive officers, during the period between the date of the business combination agreement and the closing of the transaction. Neither the overall size, nor the allocations to particular participants, of these retention payments has been determined. Any retention payment to a named executive officer requires the prior consent of Deutsche Telekom. For information on the share ownership of the named executive officers, see the table entitled Beneficial Ownership of MetroPCS Directors, Executive Officers and Persons Owning More than 5% of the Outstanding Shares of MetroPCS Common Stock beginning on page 161.
- (5) This amount includes the aggregate dollar value of the sum of all amounts reported in the preceding columns.

#### BOARD OF DIRECTORS AND MANAGEMENT AFTER THE TRANSACTION

The business combination agreement provides that MetroPCS will increase the size of the MetroPCS board, which will become the combined company s board, to 11 directors upon completion of the transaction. The board will consist of two existing directors of MetroPCS, who are expected to be (i) James N. Perry, Jr. and John (Jack) F. Callahan, Jr., (ii) John J. Legere, the new chief executive officer of the combined company and (iii) eight directors designated by Deutsche Telekom. Pursuant to the stockholder s agreement, at least three of the directors on the board must be considered independent under the rules of the SEC and under applicable listing standards, which could include James N. Perry, Jr. and John (Jack) F. Callahan, Jr. (both of whom may be considered independent under SEC and applicable listing standards). After completion of the transaction, Deutsche Telekom generally will have the right to designate a number of individuals to the board and any committees thereof equal to the percentage of the combined company s common stock beneficially owned by Deutsche Telekom multiplied by the number of directors on the board (or the number of members of any committee thereof), in each case, rounded to the nearest whole number. These rights will remain in effect as long as Deutsche Telekom beneficially owns 10% or more of the outstanding shares of the combined company s common stock. The business combination agreement provides that after the completion of the transaction, John J. Legere, currently President and Chief Executive Officer of T-Mobile, will serve as Chief Executive Officer of the combined company; J. Braxton Carter, currently Chief Financial Officer and Vice Chairman of MetroPCS, will serve as Chief Financial Officer of the combined company; James (Jim) C. Alling, currently Chief Operations Officer of T-Mobile, will serve as Chief Operating Officer, T-Mobile USA; and Thomas C. Keys, currently President and Chief Operating Officer of MetroPCS, will serve as Chief Operating Officer, MetroPCS. See Summary of the Business Combination Agreement and Summary of the Ancillary Agreements Stockholder s Agreement.

None of the combined company s named executive officers currently employed by MetroPCS have entered into compensation agreements in connection with their employment by the combined company.

#### Officers of the Combined Company

*John J. Legere* joined T-Mobile in September 2012 as President and Chief Executive Officer with over 32 years experience in the U.S. and global telecommunications and technology industries. Prior to joining T-Mobile, Mr. Legere served as CEO of Global Crossing Limited, or Global Crossing, where he successfully transformed the company to become a leading provider of IP services worldwide. Before this, he was CEO of Asia Global Crossing, originally a Microsoft Corporation, Softbank Corp. and Global Crossing joint venture. Previously, he served as president of Dell Computer Corporation s operations in Europe, the Middle East, Africa and the Asia-Pacific region. Mr. Legere also worked at AT&T for 18 years in a number of senior positions, including president of AT&T Asia Pacific, president of AT&T Solutions Outsourcing Unit, and head of global strategy and development. He began his career at New England Telephone in 1980. Mr. Legere received a Bachelor s degree in Business Administration from the University of Massachusetts; a Master of Science degree as an Alfred P. Sloan Fellow at the Massachusetts Institute of Technology; a Master of Business Administration degree from Fairleigh Dickinson University; and completed Harvard Business School s Management Development Program. Mr. Legere serves on the CTIA Board of Directors and on the Board of the New York Road Runners and NYC Marathon, Achilles International and Shoe4Africa. Mr. Legere is also a Board Trustee for the Dana Farber Cancer Institute.

*J. Braxton Carter* became the Vice Chairman of MetroPCS in May 2011 and currently serves as MetroPCS Chief Financial Officer. Mr. Carter served as Executive Vice President and Chief Financial Officer from February 2008 until May 2011. From March 2005 to February 2008, Mr. Carter served as Senior Vice President and Chief Financial Officer. Mr. Carter served as Vice President, Corporate Operations from February 2001 to March 2005. Mr. Carter also serves as a director of all of MetroPCS corporate subsidiaries and as a member of the management committee of each of the limited liability companies. Previously, Mr. Carter served as a director of MetroPCS Wireless, Inc., and its wholly-owned subsidiaries from July 2001 to December 2004. Prior to joining MetroPCS Communications, Mr. Carter was Chief Financial Officer and Chief Operating Officer of

PrimeCo PCS, the successor entity of PrimeCo Personal Communications formed in March 2000. He held various senior management positions with PrimeCo Personal Communications, including Chief Financial Officer and Controller, from 1996 until March 2000. Mr. Carter also has extensive senior management experience in the retail industry, spent ten years in public accounting and is also a certified public accountant. Mr. Carter presently serves on the board of directors of e-Rewards, Inc., and serves as Chairman of the Audit Committee and also serves on the Board of Advisors of Amdocs Limited.

*James (Jim) C. Alling* joined T-Mobile in 2009 and currently serves as T-Mobile s Chief Operating Officer. In this role, Mr. Alling is in charge of all customer facing activities for a subscription base of over 33 million users. Before joining T-Mobile, Mr. Alling worked as an executive at Starbucks Corporation, which we refer to as Starbucks, for 11 years in a variety of leadership roles. He joined Starbucks in 1997 and was promoted into a series of roles with ever-increasing scope and responsibility eventually becoming the President of Starbucks Coffee USA and later President of Starbucks Coffee International. Mr. Alling began his career in 1985 at Nestle S.A., which we refer to as Nestle, where he held various senior management positions in the packaged goods marketing sector before eventually becoming a VP/General Manager for Nestle USA. Mr. Alling received a BA from DePauw University in Greencastle, Indiana with a double major in Economics and Spanish; he then obtained a Master of International Management from the Thunderbird School of International Management.

*Thomas C. Keys* became President of MetroPCS in May 2011, in addition to his current position as Chief Operating Officer of MetroPCS in which he has served since June 2007. Mr. Keys also served as MetroPCS President from June 2007 to December 2007. Previously, Mr. Keys served as MetroPCS Senior Vice President, Market Operations, West, from January 2007 until June 2007, and as MetroPCS Vice President and General Manager, Dallas, from April 2005 until January 2007. Mr. Keys also serves as a director of all of MetroPCS corporate subsidiaries and as a member of the management committee of each of the limited liability companies. Prior to joining MetroPCS, Mr. Keys served as the President and Chief Operating Officer for VCP International Inc., a Dallas-based wholesale distributor of wireless products, from July 2002 to April 2005. Prior to joining VCP International Inc., Mr. Keys served as the Senior Vice President, Business Sales for WebLink Wireless, Inc., (formerly PageMart Wireless, Inc., the surviving entity upon merger with PageMart, Inc., that is now known as USA Mobility) from March 1999 to June 2002, which included leading and managing the national sales and distribution efforts, and in other senior management positions with WebLink Wireless, Inc., from January 1993 to March 1999.

#### **Directors of the Combined Company**

John (Jack) F. Callahan, Jr. has served as a director of MetroPCS since November 2008 and has served as a member of MetroPCS Audit Committee. Mr. Callahan became the Executive Vice President and Chief Financial Officer of the McGraw-Hill Companies, a New York Stock Exchange listed company and leading global financial information and education company, in December 2010. Previously, Mr. Callahan was the Executive Vice President and Chief Financial Officer of Dean Foods Company, a New York Stock Exchange food and beverage company. Mr. Callahan joined Dean Foods in May 2006. Before joining Dean Foods, he held a number of positions with PepsiCo and Frito Lay, including Senior Vice President of Corporate Strategy and Development for PepsiCo Inc. Chief Financial Officer for Frito Lay International, and Senior Vice President of Strategy and Planning at Frito Lay North America. Before joining PepsiCo Inc., he held various positions at The General Electric Company and McKinsey & Company.

*James N. Perry*, *Jr.*, a director of MetroPCS since November 2005 and the Chairman of MetroPCS Nominating and Corporate Governance Committee and a member of MetroPCS Audit and Finance Committees, is a Managing Director of Madison Dearborn Partners, LLC, a Chicago-based private equity investing firm, where he specializes in investing in companies in the communications industry. Prior to co-founding Madison Dearborn Partners, LLC, in 1992, Mr. Perry was with First Chicago Venture Capital for eight years. An affiliate of Madison Dearborn Partners, LLC, is an investor in MetroPCS. Mr. Perry also presently serves on the boards of the following private companies and non-profit organizations: New Asurion Corporation, Sorenson Communications, Inc., The

Topps Company, Inc., Univision Communications, Inc., NextG Networks, Inc., the Chicago Public Media board and the School Board of the Archdiocese of Chicago. Mr. Perry previously served on the board of directors of Nextel Partners from July 2003 to June 2006.

The remaining directors of the combined company have not yet been determined but will be identified by Deutsche Telekom prior to the completion of the transaction.

#### ACCOUNTING TREATMENT

MetroPCS prepares its financial statements in accordance with GAAP. The transaction will be accounted for as a reverse acquisition under the acquisition method of accounting, which requires determination of the accounting acquirer. The accounting guidance for business combinations, referred to as Accounting Standards Codification 805, provides that in identifying the acquiring entity in a combination effected through an exchange of equity interests, all pertinent facts and circumstances must be considered, including: the relative voting rights of the stockholders of the constituent companies in the combined company, the existence of a large minority voting interest in the combined entity if no other owner or organized group of owners has a significant voting interest, the composition of the board of directors and senior management of the combined company, the relative size of each company and the terms of the exchange of equity securities in the business combination, including payment of any premium.

Because T-Mobile s indirect stockholder, Deutsche Telekom, will be entitled to designate the majority of the board of directors of the combined company, MetroPCS stockholders will receive the cash payment and Deutsche Telekom will receive a majority of the equity securities of the combined company, T-Mobile is considered to be the acquirer of MetroPCS for accounting purposes. This means that T-Mobile will allocate the purchase price to the fair value of MetroPCS assets and liabilities at the acquisition date, with any excess purchase price being recorded as goodwill.

## NO APPRAISAL RIGHTS

Under applicable law, MetroPCS stockholders do not have the right to an appraisal of the value of their shares in connection with the transaction.

#### MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES

The following is a summary of certain material U.S. federal income tax consequences of the reverse stock split and cash payment to U.S. holders and non-U.S. holders (as such terms are defined below) of shares of MetroPCS common stock.

This summary does not purport to consider all aspects of U.S. federal income taxation that might be relevant to U.S. holders or non-U.S. holders of shares of MetroPCS common stock. The summary is based on the Internal Revenue Code of 1986 as amended, which we refer to as the Code, final, temporary or proposed U.S. Treasury regulations, administrative rulings and court decisions in effect as of the date of this proxy statement, all of which are subject to change at any time, possibly with retroactive effect. Any such change could alter the U.S. federal income tax consequences described herein. No ruling has been or will be sought from the Internal Revenue Service, which we refer to as the IRS, as to the U.S. federal income tax consequences of the reverse stock split and cash payment. Accordingly, there can be no assurance that the IRS will not challenge any of the U.S. federal income tax consequences described herein.

For purposes of this summary, we use the term U.S. holder to mean:

a citizen or individual resident of the United States;

a corporation, or other entity treated as a corporation for U.S. federal income tax purposes, created or organized under the laws of the United States or any state thereof (or the District of Columbia);

a trust if it (1) is subject to the primary supervision of a court within the United States and one or more U.S. persons have the authority to control all substantial decisions of the trust, or (2) has a valid election in effect under applicable U.S. Treasury regulations to be treated as a U.S. person; or

an estate that is subject to U.S. federal income tax on its income regardless of its source. For purposes of this summary, we use the term non-U.S. holder to mean a holder of shares of MetroPCS common stock that is neither a U.S. holder nor a partnership (or other entity or arrangement treated as a partnership for U.S. federal income tax purposes).

This summary only addresses U.S. federal income tax consequences of the reverse stock split and cash payment to holders of shares of MetroPCS common stock that hold their shares as a capital asset within the meaning of Section 1221 of the Code (generally, property held for investment). Further, this summary does not address all aspects of U.S. federal income taxation that may be relevant to a holder of shares of MetroPCS common stock in light of such holder s particular circumstances or that may be applicable to holders subject to special treatment under U.S. federal income tax laws (including, for example, banks or other financial institutions, insurance companies, real estate investment trusts or regulated investment companies, broker-dealers, dealers in securities or currencies, traders in securities that have elected to use a mark-to-market method of accounting, tax-exempt entities including governmental authorities (both U.S. and non-U.S.), U.S. holders whose functional currency is not the U.S. dollar, holders who acquired shares of MetroPCS common stock that vests as of the closing of the transaction, holders who hold shares of MetroPCS common stock in an individual retirement or other tax-deferred account, holders subject to the alternative minimum tax provisions of the Code, U.S. expatriates, holders who hold shares of MetroPCS common stock as part of a hedge, straddle, integration, constructive sale, conversion or other risk reduction transaction, and S corporations, partnerships or other pass-through entities (). In addition, no information is provided herein with respect to the tax consequences of the MetroPCS recapitalization that includes a reverse stock split under applicable state, local or non-U.S. tax laws or U.S. federal income tax laws.

If a partnership (including an entity or arrangement classified as a partnership for U.S. federal income tax purposes) holds shares of MetroPCS common stock, the tax treatment of a partner in the partnership generally

will depend on the status of the partners and the activities of the partnership. If a holder is a partner in a partnership holding shares of MetroPCS common stock, such holder should consult its tax advisor.

The U.S. federal income tax consequences described below are not intended to constitute a complete description of all tax consequences relating to the reverse stock split and cash payment. Holders of shares of MetroPCS common stock should consult with their tax advisors regarding the tax consequences of the reverse stock split and cash payment to them, including the effects of U.S. federal, state and local, non-U.S. income and other tax laws.

#### The Reverse Stock Split and Cash Payment

#### U.S. Holders of Shares of MetroPCS Common Stock

For U.S. federal income tax purposes, the reverse stock split and cash payment should be integrated and treated as a recapitalization within the meaning of Section 368(a)(1)(E) of the Code in which each MetroPCS stockholder exchanges each share MetroPCS common stock for one-half of a share of the combined company s common stock plus the per share amount of the cash payment, which we refer to as the MetroPCS recapitalization. In the event of such recapitalization treatment, the U.S. federal income tax consequences of the MetroPCS recapitalization to U.S. holders of MetroPCS common stock will be, in general, as follows:

A U.S. holder generally would recognize taxable gain, but not loss, on the MetroPCS recapitalization in an amount equal to the lesser of (i) the cash received (excluding any cash received in lieu of a fractional share, as discussed below) as part of the per share amount of the cash payment or (ii) the excess, if any, of (A) the sum of the per share amount of the cash payment received pursuant to the MetroPCS recapitalization and the fair market value of the shares of the common stock of the combined company received by such holder over (B) such holder s tax basis in the holder s shares of MetroPCS common stock. Any such gain would be capital gain provided that one of the Section 302 tests described below was satisfied and would be long-term capital gain if the holder s holding period in shares of MetroPCS common stock exceeds one year on the date of the MetroPCS recapitalization. Long-term capital gains of a non-corporate taxpayer are subject to reduced rates of taxation. If none of the Section 302 tests were satisfied, the U.S. holder s gain generally would be treated as a dividend distribution under Section 301 of the Code to the extent of such holder s ratable share of MetroPCS current and accumulated earnings and profits (as determined under U.S. federal income tax principles) and then as capital gain.

A U.S. holder s aggregate tax basis in the combined company s common stock received pursuant to the MetroPCS recapitalization would equal the aggregate tax basis in the MetroPCS common stock surrendered (excluding the portion of the tax basis that is allocable to any fractional share), decreased by the cash payment received and increased by the income and gain recognized in the exchange (excluding any gain attributable to cash received in lieu of a fractional share).

A U.S. holder sholding period for the combined company s common stock received would include the holding period for the MetroPCS common stock surrendered.

A U.S. holder who receives cash in lieu of a fractional share of the combined company s common stock generally would recognize capital gain or loss in the amount equal to the difference, if any, between the amount of cash received in lieu of the fractional share and the tax basis allocated to such fractional share of the combined company s common stock.

The Section 302 tests referenced above are based on Section 302 of the Code, and generally provide that a U.S. holder will recognize capital gain as long as (i) the receipt of cash results in a substantially disproportionate redemption with respect to the U.S. holder, (ii) the receipt of cash is not essentially equivalent to a dividend with respect to the U.S. holder or (iii) the exchange results in a complete termination of the U.S. holder s interest in the combined company. Each of these tests, referred to as the Section 302 tests, is explained in more detail below.

In general, because the MetroPCS stock issuance will reduce the percentage ownership of each U.S. holder of shares of MetroPCS common stock in the common stock of the combined company compared to its percentage ownership of MetroPCS, the MetroPCS recapitalization should result in a substantially disproportionate redemption with respect to each such U.S. holder and accordingly a Section 302 test should be satisfied. Holders should consult their tax advisors to determine the application of the Section 302 tests to their particular circumstances.

The Section 302 tests are:

<u>Substantially Disproportionate Test</u>. The MetroPCS recapitalization generally will result in a substantially disproportionate redemption with respect to a U.S. holder of shares of MetroPCS common stock if, among other things, the percentage of the outstanding shares of the common stock of the combined company actually and constructively owned by the U.S. holder immediately after the MetroPCS recapitalization and the MetroPCS stock issuance is less than 80% of the percentage of the shares of MetroPCS common stock actually and constructively owned by the U.S. holder before the MetroPCS recapitalization and the MetroPCS stock issuance.

Not Essentially Equivalent to a Dividend Test. The receipt of cash in the MetroPCS recapitalization will be treated as not essentially equivalent to a dividend if the reduction in a U.S. holder s proportionate interest in the combined company as a result the MetroPCS recapitalization and the MetroPCS stock issuance (when compared to the U.S. holder s proportionate interest in MetroPCS immediately prior to the MetroPCS recapitalization and the MetroPCS stock issuance) constitutes a meaningful reduction of the U.S. holder s proportionate interest given the U.S. holder s particular facts and circumstances. The IRS has indicated in a published revenue ruling that even a small reduction in the percentage interest of a stockholder whose relative stock interest in a publicly held corporation is minimal and who exercises no control over corporate affairs should constitute a meaningful reduction.

<u>Complete Termination Test</u>. A U.S. holder of shares of MetroPCS common stock may be able to satisfy the complete termination test if such holder sells or otherwise disposes of all of such holder s shares of the common stock of MetroPCS or the combined company contemporaneously with the completion of the MetroPCS recapitalization and the MetroPCS stock issuance and as part of a single integrated plan which includes participation by such holder in the MetroPCS recapitalization. However, there is some uncertainty as to whether the complete termination test applies in such circumstances. U.S. holders should consult their own tax advisors as to this matter in light of their particular circumstances and the applicable law.

In applying the Section 302 tests, U.S. holders must take into account not only shares of the common stock of the combined that they actually own but also shares they are treated as owning under the constructive ownership rules of Section 318 of the Code. Under the constructive ownership rules, a U.S. holder is treated as owning any shares that are owned (actually and in some cases constructively) by certain related individuals and entities as well as shares that the U.S. holder has the right to acquire by exercise of an option or warrant or by conversion or exchange of a security. Also, contemporaneous acquisitions or dispositions of the common stock of the combined company may be deemed to be part of a single integrated transaction and, if so, may be taken into account in determining whether any of the Section 302 tests, described above, are satisfied. **Due to the factual nature of the Section 302 tests, U.S. holders should consult their tax advisors to determine whether the receipt of cash in the MetroPCS recapitalization qualifies for sale or exchange treatment in their particular circumstances.** 

#### Additional Tax on Net Investment Income

For taxable years beginning after December 31, 2012, U.S. holders that are not corporations will generally be subject to a 3.8% tax (the Medicare tax) on the lesser of (1) the U.S. holder s net investment income for the taxable year and (2) the excess of the U.S. holder s modified adjusted gross income for the taxable year over

certain thresholds. A U.S. holder s net investment income will generally include any income or gain recognized by such holder with respect to the MetroPCS recapitalization, unless such income or gain is derived in the ordinary course of the conduct of such U.S. holder s trade or business (other than a trade or business that consists of certain passive or trading activities).

## Non-U.S. Holders of Shares of MetroPCS Common Stock

Payments of cash pursuant to the MetroPCS recapitalization to a non-U.S. holder (or such holder s agent) will be subject to withholding of U.S. federal income tax at a rate of 30%, unless a reduced rate of withholding is applicable pursuant to an income tax treaty or an exemption from withholding is applicable because such payments are effectively connected with the non-U.S. holder s conduct of a trade or business within the United States (and, if an income tax treaty applies, the payments are generally attributable to a United States permanent establishment maintained by such non-U.S. holder). In order to claim a reduction of or an exemption from withholding tax, a non-U.S. holder must provide a validly completed and executed IRS Form W-8BEN (with respect to income tax treaty benefits) or IRS Form W8-ECI (with respect to amounts effectively connected with the conduct of a trade or business within the United States) claiming such reduction or exemption before the payment is made. A non-U.S. holder that qualifies for an exemption from withholding by delivering IRS Form W-8ECI generally will be subject to U.S. federal income tax on income derived from the MetroPCS recapitalization at the same rates applicable to U.S. holders. Additionally, in the case of a corporate non-U.S. holder, such income may be subject to branch profits tax at a rate of 30% (or a lower rate specified in an applicable income tax treaty). Exchanging non-U.S. holders can obtain the applicable IRS forms from the IRS website at *www.irs.gov*.

A non-U.S. holder may be eligible to obtain a refund of all or a portion of any tax withheld (i) if such holder meets one of the Section 302 tests (the substantially disproportionate, the not essentially equivalent to a dividend, or the complete termination test) described in the section entitled Material U.S. Federal Income Tax Consequences The Reverse Stock Split and Cash Payment U.S. Holders of Shares of MetroPCS Common Stock or (ii) if such holder is otherwise able to establish that no tax or a reduced amount of tax is due. Non-U.S. holders should consult their own tax advisors regarding the particular tax consequences to them of the MetroPCS recapitalization, including the application of U.S. federal income tax withholding, their potential eligibility for a withholding tax reduction or exemption, and the refund procedure.

#### Information Reporting and Backup Withholding

Payments of cash made in connection with the MetroPCS recapitalization may, under certain circumstances, be subject to information reporting and backup withholding (currently at a rate of 28%). The current backup withholding rate of 28% is scheduled to expire on December 31, 2012, and the backup withholding rate will be 31% for amounts paid after December 31, 2012, unless further extended by the U.S. Congress. To avoid backup withholding, each holder of shares of MetroPCS common stock that does not otherwise establish an exemption should furnish its taxpayer identification number and comply with the applicable certification procedures. Backup withholding is not an additional tax and amounts withheld will be allowed as a credit against the holder s U.S. federal income tax liability and may entitle such holder to a refund, provided the required information is timely furnished to the IRS.

#### SUMMARY OF THE BUSINESS COMBINATION AGREEMENT

The following is a summary of the material provisions of the business combination agreement, which is attached as Annex A to this proxy statement and which is incorporated by reference herein in its entirety. The summary in this section, and elsewhere in this proxy statement, is qualified in its entirety by reference to the business combination agreement. This summary does not purport to be complete and may not contain all of the information about the business combination agreement that is important to you. We encourage you to read carefully the business combination agreement in its entirety before making any decisions regarding the transaction because it is the principal document governing the transaction.

This summary of the business combination agreement has been included to provide you with information regarding the terms of the business combination agreement. It is not intended to provide any other factual information about MetroPCS, Deutsche Telekom or T-Mobile or their respective subsidiaries or affiliates. The representations, warranties, and covenants contained in the business combination agreement were made only for purposes of that agreement and as of specific dates, were solely for the benefit of the parties to the business combination agreement, were for the purposes of allocating risks between the parties to the business combination agreement, and may be subject to limitations agreed upon by the parties, including being qualified by confidential disclosures made by each contracting party to the other. Moreover, information concerning the subject matter of the representations, warranties, and covenants may change after the date of the business combination agreement, and subsequent information may or may not be fully reflected in public disclosures by MetroPCS, Deutsche Telekom or T-Mobile. Accordingly, you should read the representations and warranties in the business combination agreement not in isolation but only in conjunction with the other information about MetroPCS, Deutsche Telekom and T-Mobile and their respective subsidiaries that the respective companies include in reports, statements, documents and other filings they make with the SEC or other governmental agencies.

#### Structure of the Transaction

On October 3, 2012, MetroPCS entered into the business combination agreement with Deutsche Telekom, Global, Holding and T-Mobile. Pursuant to the terms and subject to the conditions set forth in the business combination agreement, the transaction is structured as follows. MetroPCS will (i) effect a recapitalization that includes a reverse stock split, pursuant to which each share of MetroPCS common stock outstanding as of the effective time will represent thereafter one-half of a share of MetroPCS common stock, and (ii) as part of the recapitalization, make a cash payment in an amount equal to \$1.5 billion (or approximately \$4.08 per share pre-reverse stock split), without interest, in the aggregate to the record holders of MetroPCS common stock immediately following the effective time. Immediately following the cash payment, MetroPCS will issue and deliver to Holding or its designee shares of MetroPCS common stock equal to 74% of the fully-diluted shares of MetroPCS common stock outstanding immediately following the cash payment, and Holding will deliver to MetroPCS all of the shares of capital stock of T-Mobile. The closing price of MetroPCS common stock used in the above calculation will be the average, rounded to the nearest one ten-thousandth, of the closing price of a share of MetroPCS common stock on the NYSE for the five full NYSE trading days immediately preceding the closing of the transaction (without giving effect to any adjustment for the MetroPCS reverse stock split or the cash payment) whether through the operation of the NYSE s ex-dividend procedures or otherwise. In addition, unless otherwise agreed to by the parties, on the business day immediately following the closing of the transaction, MetroPCS, Inc., will merge with and into Wireless, with Wireless continuing as the surviving entity and, immediately thereafter, Wireless will merge with and into T-Mobile, with T-Mobile continuing as the surviving entity.

#### **Effective Time; Closing**

The effective time of the reverse stock split will be the date and time specified by MetroPCS in the new certificate of incorporation filed with the Secretary of State of the State of Delaware. Unless the parties otherwise agree, the closing of the transaction will take place on the third business day after all conditions to the completion

of the transaction have been satisfied or, to the extent such conditions can be waived under applicable law, waived (other than those conditions that by their terms are to be satisfied at the closing). These conditions are more fully described in the section entitled Summary of the Business Combination Agreement Conditions to Closing the Transaction beginning on page 149. The closing will be deemed to have occurred and will be effective as of 12:01 a.m., prevailing Eastern Time, on the date of the closing.

The parties expect that the conditions to the transaction will be satisfied or, if allowed by applicable law, waived, and the transaction will be completed, in the first half of 2013. However, the parties cannot be certain when, or if, the conditions to the transaction will be satisfied or so waived, or that the transaction will be completed.

#### Adjustment to the Consideration for the Transaction

The business combination agreement provides that Deutsche Telekom will make a cash payment to T-Mobile at the closing equal to Deutsche Telekom s estimate of the sum of (i) the excess, if any, of \$1.3 billion over T-Mobile s working capital as of the closing (as calculated pursuant to the business combination agreement), plus (ii) any deficiency between the amount actually spent by T-Mobile and its subsidiaries during the period between the signing of the business combination agreement and the closing on capital expenditures, marketing and subscriber acquisition and retention activities when compared to specified amounts for such items during such period. Following the closing, the parties will determine the difference between such estimated amount and the amount that should have been paid at the closing and the appropriate party, between Deutsche Telekom and T-Mobile, will make an adjustment payment to the other party.

#### **Cash Payment**

At or prior to the completion of the transaction, MetroPCS will appoint a payment agent to handle the cash payment to the record holders of MetroPCS common stock immediately prior to the effective time. Promptly after the completion of the transaction, the payment agent automatically will deliver to each MetroPCS stockholder that held MetroPCS common stock in book-entry form immediately prior to the effective time the portion of the cash payment to which such MetroPCS stockholder is entitled. For each MetroPCS stockholder that held a certificate representing shares of MetroPCS common stock immediately prior to the reverse stock split, which we refer to as a MetroPCS stockholder is entitled promptly following such MetroPCS stockholder is delivery to the payment to which such MetroPCS stockholder is entitled promptly following such MetroPCS stockholder is delivery to the payment agent of its MetroPCS stock certificate and properly completed letter of transmittal, as described in more detail in the following section.

#### **Exchange of MetroPCS Common Stock**

Upon the effective time, the reverse stock split will occur and each share of MetroPCS common stock issued and outstanding immediately prior to the effective time automatically will be converted into one-half of a validly issued, fully paid and non-assessable share of MetroPCS common stock. This conversion of shares of MetroPCS common stock pursuant to the reverse stock split does not require any action on the part of MetroPCS or any of its stockholders.

As a result, for MetroPCS stockholders that hold MetroPCS common stock in book-entry form, the number of shares of MetroPCS common stock in your account automatically will be adjusted as a result of the reverse stock split. For MetroPCS stockholders that hold a MetroPCS stock certificate, your certificate will automatically represent the number of shares of MetroPCS common stock into which the shares of MetroPCS common stock represented by such MetroPCS stock certificate have been combined pursuant to the reverse stock split.

Notwithstanding the foregoing, for MetroPCS stockholders that hold a MetroPCS stock certificate, the payment agent will send you a letter of transmittal as promptly as practicable after the completion of the transaction. The letter of transmittal will specify that delivery will be effected, and risk of loss and title to any

certificates shall pass, only upon proper delivery of your MetroPCS stock certificate to the payment agent. The letter of transmittal will be accompanied by instructions for exchanging your MetroPCS stock certificate for a certificate that represents the number of shares of MetroPCS common stock into which the shares of MetroPCS common stock represented by such MetroPCS certificate have been combined pursuant to the reverse stock split. Please note that you will be required to complete and return this letter of transmittal, together with your MetroPCS stock certificate, in order to receive your portion of the cash payment. As a result, please do not return your MetroPCS stock certificate with the enclosed proxy card.

## **Fractional Shares**

MetroPCS will not issue any fractional shares of MetroPCS common stock resulting from the reverse stock split. Instead of any fractional shares, the payment agent will pay MetroPCS stockholders an amount in cash for any fraction of a share calculated by multiplying (i) the fractional share interest to which such stockholder would otherwise be entitled by (ii) the average closing price, rounded to the nearest one ten-thousandth, of a share of MetroPCS common stock on the NYSE for the five full NYSE trading days immediately preceding the closing date, without giving effect to any adjustment for the reverse stock split or the cash payment, whether through the operation of the NYSE s ex-dividend procedures or otherwise.

# Effect on Annual Cash Performance Awards, Stock Options and Restricted Stock Issued Pursuant to MetroPCS Equity Incentive Compensation Plans

Effective as of the completion of the MetroPCS stock issuance, all outstanding equity and incentive awards under MetroPCS equity incentive compensation plans, including all outstanding annual cash performance awards, stock options and each share of restricted stock, will automatically vest (at the target payment amount for annual cash performance awards) and (in the case of stock options) become exercisable and will remain outstanding.

In addition, except as provided below, at the time the amendment of MetroPCS certificate of incorporation becomes effective, each outstanding option to acquire MetroPCS common stock will be adjusted to take into account the reverse stock split and the cash payment. The number of shares of MetroPCS common stock to be acquired pursuant to outstanding options will be reduced to reflect the reverse stock split and the exercise price of the options will be increased to reflect the reverse stock split, with the per share amount of the cash payment made to holders of MetroPCS common stock deducted from the options adjusted per share exercise price. Restricted stock will be adjusted to account for the reverse stock split in the same manner as all other stock, and holders of restricted stock will share in the cash payment. Holders of stock options will not receive any portion of the cash payment.

Notwithstanding and in lieu of the above, holders of stock options with an exercise price that is less than the average closing price of MetroPCS common stock based on a five-day trading average before the closing ignoring any market effect of the reverse stock split and cash payment, which we refer to as in-the-money stock options, may elect to receive cash in lieu of their in-the-money stock options during the five days following the closing of the transaction at a price per share equal to the average closing price of MetroPCS common stock based on a five-day trading average before the closing ignoring any market effect of the reverse stock split and cash payment minus the exercise price of the option, less required tax withholding. Any in-the-money stock options issued under the Second Amended and Restated MetroPCS, Inc. 1995 Stock Plan, as amended, that have an exercise price equal to or less than the per share amount of the cash payment (or approximately \$4.08 per share calculated on a pre-reverse stock split basis), which we refer to as low exercise price stock options, will be automatically cashed-out in the same manner. Any stock options that are not cashed-out at the closing of the transaction, including any in-the-money stock options (other than low exercise price stock options) for which the holder thereof does not elect to receive cash at closing, will be adjusted for the reverse stock split and the per share amount of the cash payment as described above and will remain outstanding, 100% vested and exercisable in accordance with their terms. In the money stock options that are cashed out will reduce the number of shares outstanding after the stock issuance and will cause Deutsche Telekom s interest in the combined company to increase proportionately.

#### **Representations and Warranties**

The business combination agreement contains representations and warranties of each of MetroPCS, Deutsche Telekom, Global, Holding and T-Mobile. MetroPCS and T-Mobile generally make reciprocal representations and warranties, and Deutsche Telekom, Global, Holding make representations and warranties as direct or indirect owners of T-Mobile.

#### Representations and Warranties of Deutsche Telekom, Global and Holding

Deutsche Telekom, Global and Holding make representations and warranties related to, among other topics, the following:

organization and good standing;

authority relative to the execution and delivery of the business combination agreement, and the execution, delivery and enforceability of the business combination agreement (also on behalf of T-Mobile);

absence of conflicts with, or violations of, organizational documents and other agreements or obligations and required consents;

absence of requirements for governmental filings other than the ones listed in the business combination agreement to consummate the transaction;

ownership of subsidiaries, including good and valid title to all equity interests of T-Mobile;

broker s fees payable in connection with the transaction;

licenses from the FCC and public utility commissions held by Deutsche Telekom, Global and Holding;

no ownership of MetroPCS common stock as of the signing of the business combination agreement; and

with respect to Deutsche Telekom, that it has sufficient funds to consummate the transaction, including to fund any additional Deutsche Telekom notes pursuant to its commitments described in the section entitled Summary of the Financing beginning on page 112.

#### **Representations and Warranties of T-Mobile**

T-Mobile makes representations and warranties related to, among other topics, the following:

organization, good standing and qualification;

capitalization;

## Edgar Filing: MILLER PETROLEUM INC - Form 10-K

ownership of subsidiaries;

absence of conflicts with, or violations of, organizational documents and other agreements or obligations and required consents;

absence of requirements for governmental filings other than the ones listed in the business combination agreement to consummate the transaction;

financial statements;

absence of undisclosed liabilities;

internal controls;

absence of certain litigation;

employee benefits matters;

compliance with applicable laws and licenses;

absence of certain changes and events between the end of the most recently completed fiscal year and the date the parties executed and delivered the business combination agreement;

insurance matters;

environmental matters;

tax matters;

labor matters;

intellectual property;

material contracts;

owned and leased property;

sufficiency and ownership of its assets and business;

related-party agreements; and

prohibited payments. Representations and Warranties of MetroPCS

MetroPCS makes representations and warranties related to, among other topics, the following:

organization, good standing and qualification;

capitalization;

ownership of subsidiaries;

authority relative to the execution and delivery of the business combination agreement, and the execution, delivery and enforceability of the business combination agreement;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

absence of conflicts with, or violations of, organizational documents and other agreements or obligations and required consents;

absence of requirements for governmental filings other than the ones listed in the business combination agreement to consummate the transaction;

financial statements and SEC reports;

absence of undisclosed liabilities;

internal controls;

absence of certain litigation;

employee benefits matters;

compliance with applicable laws and licenses;

absence of certain changes and events between the end of the most recently completed fiscal year and the date the parties executed and delivered the business combination agreement;

insurance matters;

environmental matters;

tax matters;

labor matters;

intellectual property;

material contracts;

owned and leased property;

related-party agreements;

prohibited payments;

broker s fees payable in connection with the transaction;

the MetroPCS common stock issued in the transaction;

the MetroPCS rights agreement amendment;

the required vote of the MetroPCS stockholders; and

the post-closing restructuring tax treatment.

Many of these representations and warranties are qualified by the knowledge of certain specified officers, materiality or material adverse effect limitations, disclosures set forth in schedules delivered at the time the business combination agreement was signed and/or, in the case of MetroPCS, references to its filings with the SEC.

Material adverse effect, with respect to MetroPCS, is defined in the business combination agreement to mean (i) an effect that would prevent or materially delay the ability of MetroPCS to consummate the transaction, or (ii) a material adverse effect on the financial condition, properties, assets, liabilities, business or results of operations of MetroPCS and its subsidiaries, taken as a whole, except that the definition of material adverse effect excludes from clause (ii):

any circumstance generally affecting (x) the territory, or global economy or territory or global financial, debt, credit, capital or securities markets or (y) the wireless telecommunications and wireless information products and services industry in the territory, unless such circumstance has a disproportionate impact on MetroPCS and its subsidiaries, taken as a whole, vis-à-vis other companies in the wireless telecommunications and wireless information services industry in the territory;

any circumstance resulting from any declared or undeclared acts of war, terrorism, outbreaks or escalations of hostilities, sabotage or civil strife or threats thereof, unless such circumstance has a disproportionate impact on MetroPCS and its subsidiaries, taken as a whole, vis-à-vis other companies in the wireless telecommunications and wireless information services industry in the territory;

any act of God or weather-related circumstance, unless such circumstance has a disproportionate impact on MetroPCS and its subsidiaries, taken as a whole, vis-à-vis other companies in the wireless telecommunications and wireless information services industry in the territory;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

any circumstance resulting from any change in applicable laws or regulatory or enforcement developments, unless such circumstance has a disproportionate impact on MetroPCS and its subsidiaries, taken as a whole, vis-à-vis other companies in the wireless telecommunications and wireless information services industry in the territory;

any circumstance resulting from any change in GAAP, unless such circumstance has a disproportionate impact on MetroPCS and its subsidiaries, taken as a whole, vis-à-vis the prepaid operations of other companies in the wireless telecommunications and wireless information services industry in the territory;

any circumstance resulting from any failure by MetroPCS or its subsidiaries to meet any estimates, projections, budgets or forecasts of revenues or earnings for any period ending on or after the date hereof, or any rumors, predictions or reports of such failure (but not the underlying cause of such failure);

any circumstance resulting from the announcement, pendency or public disclosure of the business combination agreement and the transaction;

any circumstance resulting from any action required to be taken or omitted to be taken pursuant to the business combination agreement; and

any circumstance resulting from any decline in the price or trading volume of the MetroPCS common stock on the NYSE (but not the underlying cause of such decline).

The definition of material adverse effect, with respect to T-Mobile, is generally and, to the extent applicable, reciprocal. Any determination of material adverse effect also excludes the effects of matters disclosed in the parties schedules delivered at the time the business combination agreement was signed or the matters specifically identified in the notes to the parties financial statements.

#### **Conduct of Business**

Each of MetroPCS and T-Mobile has undertaken certain covenants in the business combination agreement restricting the conduct of its business between the date of the business combination agreement and the closing. In general, each of MetroPCS and T-Mobile has agreed to conduct its business in the ordinary course and, to the extent consistent therewith, use its commercially reasonable efforts to preserve its business organizations intact, maintain existing relations and goodwill with governmental entities, customers, suppliers, distributors, dealers, retailers, creditors, lessors, employees and business associates, and keep available the services of its present employees, officers and agents, in each case except as may be required by law or the business combination agreement or previously disclosed in writing to the other party, or with the other party s consent, which may not be unreasonably withheld and must be provided or denied within 10 days.

In addition, each of MetroPCS and T-Mobile has agreed to various specific restrictions relating to the conduct of its business between the date of the business combination agreement and the closing, including the following (in each case except as may be required by law or the business combination agreement or previously disclosed in writing to the other party, or with the other party s consent, which may not be unreasonably withheld and must be provided or denied within 10 days):

amending its organizational documents, unless such amendment would not be reasonably expected to prevent, materially delay or materially impair the completion of the transaction;

merging or consolidating with any other person, or authorizing, recommending, proposing or announcing a plan of liquidation, dissolution, consolidation, restructuring, recapitalization or any other reorganization other than transactions solely among T-Mobile or MetroPCS and their respective subsidiaries;

acquiring assets from any other person with a value or purchase price in excess of, in the case of T-Mobile, \$100 million, and, in the case of MetroPCS, \$50 million (or \$35 million for acquiring spectrum licenses), in each case in the aggregate, unless such acquisition is (i) in the ordinary course of business, (ii) in material compliance with the provisions for expenses and capital expenditures of the T-Mobile or MetroPCS, as applicable, business plan, or (iii) pursuant to an agreement in effect on the date of the business combination agreement for consideration not exceeding, in the case of T-Mobile, \$25 million, and, in the case of MetroPCS, \$10 million;

issuing, selling or encumbering its equity interests (except for customary exceptions and, for MetroPCS, in connection with the exercise of outstanding options or existing financing arrangements, including a provision by which MetroPCS may issue up to 5,750,000 additional shares pursuant to stock options and/or restricted stock awards, all of which will vest upon the closing of the transaction);

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

entering into any agreement with respect to the voting of its equity interests;

creating or incurring any encumbrance (other than customary permitted encumbrances or in the ordinary course of business consistent with past practice) on its assets that, individually or in the

aggregate, is material to it and its subsidiaries, taken as a whole, or would reasonably be expected to prevent, materially delay or materially impair the completion of the transaction;

making loans, guarantees, capital contributions to or investments in excess of, in the case of T-Mobile, \$50 million, and, in the case of MetroPCS, \$20 million;

declaring or paying dividends or otherwise making distributions, except that T-Mobile will be permitted to make or pay cash distributions or dividends, including to Holding;

incurring additional debt or modifying the terms of existing debt, except (i) as permitted under the business combination agreement, (ii) in the ordinary course of business consistent with past practice, (iii) as contemplated by its business plan, and (iv) for MetroPCS as required by its existing financing arrangements;

except as contemplated by its approved capital expenditure budget in the business plan, making capital expenditures in excess of, in the case of T-Mobile, \$100 million (or \$200 million in the event of a significant increase in the data demand of the business), and, in the case of MetroPCS, \$70 million (or \$135 million in the event of a significant increase in the data demand of the business);

entering into or amending certain restricted contracts and, with respect to certain other material contracts, failing to provide the other party with written notice of the entry into such material contract within 10 business days after such entry;

in the case of T-Mobile, entering into or amending certain intercompany contracts;

changing its financial accounting policies or procedures (except as required by changes in GAAP, the International Financial Reporting Standards, or the SEC, by governmental entities or by law) or writing up, writing down or writing off the book value of its assets (except in the ordinary course of business consistent with past practice or as may be consistent with its financial accounting policies and GAAP);

entering into new lines of business or in any conduct that may require the receipt, the transfer or the application for a license that would reasonably be expected to prevent, materially delay or impair the consummation of the transaction;

entering into new geographic areas outside of the territory other than in support of business or operations within the territory;

applying for, requesting or filing for any license the receipt of which would reasonably be likely to prevent, materially impair or materially delay the completion of the transaction;

settling material litigation, except to the extent of its reserves on its most recent balance sheet or for an amount less than, in the case of T-Mobile, \$50 million, and, in the case of MetroPCS, \$12.5 million, for any individual or group of related settlements without the imposition of material restrictions;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

making a change in any tax election or method of tax accounting, or settling any tax controversy for an amount materially in excess of the applicable amount reserved on its most recent balance sheet or filing an amended tax return, in each case to the extent that such action would have a material adverse effect on T-Mobile or MetroPCS, as applicable;

transferring licenses or wireless spectrum, other than (i) point-to-point microwave licenses, business radio licenses and experimental licenses, (ii) exchanges of spectrum licenses within the same FMA that improve contiguity within such FMA but do not worsen the expected contiguity between T-Mobile and MetroPCS and their respective subsidiaries, or (iii) license swaps transferring, in the case of T-Mobile, 300 million licensed MHz POPs (MHz of license multiplied by population covered) or fewer and involving cash consideration of \$10 million or less, and, in the case of MetroPCS, 150 million licensed MHz POPs or fewer in exchange for spectrum licenses covering at least 75% of the number of licensed MHz POPs transferred and involving cash consideration or less;

transferring any other material assets, including licenses, except (i) in the ordinary course of business consistent with past practice, (ii) with a value or purchase price not exceeding, in the case of T-Mobile, \$100 million, and, in the case of MetroPCS, \$50 million, or (iii) pursuant to an existing agreement for consideration not exceeding, in the case of T-Mobile, \$25 million, and, in the case of MetroPCS, \$10 million;

except for up to \$25 million in MetroPCS retention payments to non-officer employees or to officers with Deutsche Telekom s consent, amending or increasing certain employee compensation and benefits;

transferring transmission towers other than (i) dispositions solely among it and its subsidiaries, (ii) decommissioning or transferring towers in the ordinary course of business or (iii) in one or more transactions scheduled to close prior to the closing date;

acquiring wireless spectrum that would reasonably be expected to prevent, materially delay or materially impair the completion of the transaction or result in required divestitures of assets, except pursuant to definitive agreements entered into prior to the signing of the business combination agreement;

making a fundamental change to its network technologies or principal billing systems (except for system upgrades, improvements and modernization, equipment replacement and similar matters consistent with the business plan);

in the case of MetroPCS, taking any action that would prevent or impede the merger of (i) MetroPCS, Inc., with and into Wireless or (ii) Wireless with and into T-Mobile, in each case, from qualifying as a reorganization within the meaning Section 368(a) of the Code; or

authorizing, or committing, resolving, announcing, offering, agreeing or entering into an agreement to do or take, any of the foregoing actions or any actions inconsistent with the foregoing. **Proxy Statement and Stockholder Meeting** 

# MetroPCS and Deutsche Telekom have agreed to cooperate to, and MetroPCS must within 45 calendar days following the date of the business combination agreement, prepare and file with the SEC this preliminary proxy statement relating to the required MetroPCS stockholder approvals. We also must use our reasonable best efforts, and Deutsche Telekom must cooperate with us, to resolve all SEC comments with respect to this proxy statement as promptly as practicable after receipt of such comments, and we must use our reasonable best efforts to cause this proxy statement in definitive form to be mailed to our stockholders as promptly as practicable following the clearance of this proxy statement by the SEC.

In addition, we have agreed to use our reasonable best efforts to solicit from our stockholders proxies in favor of the required MetroPCS stockholder approvals and take all other actions necessary or advisable to secure the required MetroPCS stockholder approvals. We also have agreed to call, give notice of, convene and hold the special meeting no later than 45 business days after the date this proxy statement is cleared by the SEC, except that (i) we may (and, at the request of Deutsche Telekom, we must) postpone or adjourn the special meeting to a date no more than 15 days after its originally noticed date only to the extent reasonably required in order to solicit additional proxies so as to establish a quorum and obtain the required MetroPCS stockholder approvals, and (ii) we may postpone or adjourn the special meeting to allow time for the filing and dissemination of any supplemental or amended disclosure document that the MetroPCS board has determined in good faith (after consultation with its outside legal counsel) is necessary or required to be filed and disseminated under applicable laws.

#### No Solicitation of Alternative Proposals

Each of MetroPCS and Deutsche Telekom has agreed that, from the time of the execution and delivery of the business combination agreement until the completion of the transaction, neither it nor any of its subsidiaries will, that it will not authorize or permit any of its and their respective directors and officers to, and that it will not

authorize, and will use its reasonable best efforts not to permit, any of its and their other respective representatives to, directly or indirectly, (i) initiate, solicit or knowingly encourage or knowingly take or continue any other action to facilitate the submission of any inquiry, indication of interest, proposal or offer that constitutes, or would reasonably be expected to lead to, an acquisition proposal, (ii) participate in any discussions or negotiations regarding, or that would reasonably be expected to lead to, any acquisition proposal, (iii) furnish any non-public information or data regarding it or any of its subsidiaries to, or afford access to its properties, personnel, books and records to, any person in connection with or in response to or in circumstances that would reasonably be expected to lead to, any acquisition proposal, (iv) take any action to make the provisions of any fair price, moratorium, control share acquisition, business combination or other similar anti-takeover statute or regulation (including any transaction under, or a third party becoming an interested stockholder under, Section 203 of the DGCL), any restrictive provision of any applicable anti-takeover provision in its organizational documents or, for MetroPCS, in its stockholder rights plan, as amended, inapplicable to any person other than the other parties to the business combination agreement or to any transactions constituting or contemplated by an acquisition proposal, or (v) resolve or agree to do any of the foregoing. Additionally, MetroPCS and its subsidiaries, from the time of the execution and delivery of the business combination agreement, must, and must cause their respective directors and officers and shall use their reasonable best efforts to cause their other representatives to, cease and terminate any and all existing activities, discussions or negotiations with any person with respect to an acquisition proposal. An acquisition proposal with respect to a party means, other than the transaction, any inquiry, proposal, offer, or other expression or indication of interest with respect to any direct or indirect acquisition or purchase, in one transaction or a series of transactions, and whether through any merger, reorganization, consolidation, tender offer, self-tender, exchange offer, stock acquisition or issuance, asset acquisition, binding share exchange, business combination, recapitalization, liquidation, dissolution, joint venture or otherwise, of (A) assets or businesses of such party (or, in the case of Deutsche Telekom, of T-Mobile) and its subsidiaries that generate or represent 20% or more of the (x) net revenues or net income of such party (or, in the case of Deutsche Telekom, of T-Mobile) and its subsidiaries, taken as a whole, immediately prior to such transaction, (y) aggregate licensed MHz POPs of such party s (or, in the case of Deutsche Telekom, of T-Mobile s) FCC licenses immediately prior to such transaction, or (z) total assets (based on fair market value) of such party (or, in the case of Deutsche Telekom, of T-Mobile) and its subsidiaries, taken as a whole, immediately prior to such transaction, or (B) 20% or more of any class of capital stock, equity interests, other equity securities, economic interests or voting power of such party (or, in the case of Deutsche Telekom, any capital stock, other equity securities or voting power of T-Mobile), any of its subsidiaries or any resulting parent company.

Notwithstanding the restrictions above, the MetroPCS board will be permitted, prior to the receipt of the required MetroPCS stockholder approvals, to furnish, or cause to be furnished, information to a third party bidder making a bona fide, unsolicited written acquisition proposal and its representatives, and engage in discussions or negotiations with such bidder and its representatives with respect to such acquisition proposal if (i) such proposal did not result from a breach of the MetroPCS non-solicitation obligations described above, (ii) the MetroPCS board determines in good faith, after consultation with its outside legal counsel and financial advisors, that such acquisition proposal constitutes, or is reasonably likely to result in, a superior proposal, (iii) the MetroPCS board concludes in good faith, after consultation with its outside legal counsel, that the failure to take such action with respect to such acquisition proposal would be reasonably likely to be inconsistent with the MetroPCS board s fiduciary obligations to MetroPCS and/or its stockholders under applicable law, (iv) MetroPCS promptly notified Deutsche Telekom (and in any event within 24 hours) of the receipt and terms and conditions of such acquisition proposal (including delivering to Deutsche Telekom a copy of such acquisition proposal and any related draft agreements and other written materials) and (v) prior to furnishing any information to such bidder, MetroPCS than those contained in the confidentiality agreement between MetroPCS and T-Mobile). A superior proposal may material respect to MetroPCS than those contained in the confidentiality agreement between MetroPCS and T-Mobile). A

superior proposal means any *bona fide* written acquisition proposal with respect to MetroPCS (with all references to 20% in the definition of acquisition proposal being treated as references to 50% for these purposes) made by a third party that the MetroPCS board determines in good faith, after consultation with its outside legal counsel and financial advisors, would be more favorable to MetroPCS stockholders from a

financial point of view than the transactions contemplated in the business combination agreement, taking into account any proposal by Deutsche Telekom in writing to amend or modify the terms of the transaction, the identity of the person making such acquisition proposal, and the terms, conditions, timing, likelihood of completion and legal, financial, and regulatory aspects of such acquisition proposal.

#### Change in the MetroPCS Board s Recommendation

The MetroPCS board has agreed that neither it nor any committee of the MetroPCS board may (i) withdraw or withhold, amend, modify or qualify in any manner adverse to Deutsche Telekom its recommendation in favor of the required MetroPCS stockholder approvals or make any public announcement inconsistent with its recommendation, or publicly propose to do any of the foregoing, (ii) approve, adopt, endorse, recommend or take a neutral position (other than any factually accurate public statement by MetroPCS that solely describes MetroPCS receipt of an acquisition proposal and the operation of the business combination agreement with respect thereto or any stop, look and listen communication or similar communication of the type contemplated by Rule 14d-9(f) under the Exchange Act) on any acquisition proposal or any inquiry or proposal that would reasonably be expected to lead to an acquisition proposal, (iii) following the date any acquisition proposal or any material modification thereto is first made public, sent or given to the stockholders of MetroPCS, fail to issue a press release that expressly reaffirms its recommendation within 10 business days following Deutsche Telekom s written request to do so (which request may only be made once with respect to any acquisition proposal and each material modification thereto), (iv) fail to include its recommendation in the proxy statement (we refer to any action described in clause (i), (ii), (iii) or (iv), whether taken by MetroPCS to abandon, terminate, delay or fail to consummate the transaction.

Notwithstanding the foregoing, at any time prior to the receipt of the required MetroPCS stockholder approvals, the MetroPCS board may effect an adverse recommendation change in response to an intervening event or in response to an acquisition proposal (as described in the section entitled Summary of the Business Combination Agreement No Solicitation of Alternative Proposals beginning on page 142) if the MetroPCS board determines in good faith, after consultation with MetroPCS outside legal counsel and financial advisors, that such acquisition proposal constitutes a superior proposal and such superior proposal is not withdrawn and that, after consultation with MetroPCS outside legal counsel, the failure to effect such adverse recommendation change would reasonably be likely to be inconsistent with the MetroPCS board s fiduciary obligations to MetroPCS and/or its stockholders under applicable law. MetroPCS can only effect an adverse recommendation change if it is in compliance with its non-solicitation obligations described above (except for immaterial non-compliance that is not reasonably related to the adverse recommendation change) and promptly notified Deutsche Telekom (and in any event within 24 hours) of the receipt of the acquisition proposal, including the identity of the person making such superior proposal and, if the acquisition proposal is in writing, provided a copy of such acquisition proposal and any related draft agreements and other written materials, or, if the acquisition proposal is oral, a detailed summary thereof. Before making an adverse recommendation change, MetroPCS must provide written notice to Deutsche Telekom that the MetroPCS board is prepared to make an adverse recommendation change and the reasons therefor and give Deutsche Telekom five business days to amend or revise its offer. If the bidder that made the superior proposal makes another proposal or materially amends its proposal, then MetroPCS must allow Deutsche Telekom three business days to respond to such updated proposal (and any subsequent updated proposal). Intervening event means a circumstance material to MetroPCS and its subsidiaries, taken as a whole, that did not occur, arise or become known to the MetroPCS board or was not reasonably foreseeable by the MetroPCS board, in each case prior to the date of the business combination agreement (or if known or reasonably foreseeable prior to the date of the business combination agreement, the material consequences of which were not known or reasonably foreseeable prior to the date of the business combination agreement), which circumstance, or any material consequence thereof, becomes known to the MetroPCS board prior to the receipt of the required MetroPCS stockholder approvals, except that the receipt, existence or terms of an acquisition proposal do not constitute an intervening event.

If the MetroPCS board, taking into account all amendments or revisions committed to by Deutsche Telekom, determines, after consultation with its outside legal counsel and financial advisors, that failure to effect such adverse recommendation change would reasonably be likely to be inconsistent with the MetroPCS board s fiduciary obligations to MetroPCS and/or its stockholders under applicable law and that the acquisition proposal remains a superior proposal, then it can make an adverse recommendation change. However, unless Deutsche Telekom terminates the business combination agreement as a result thereof, MetroPCS will nonetheless continue to be obligated to hold the special meeting and submit the proposals described in this proxy statement to its stockholders for approval and to comply with its other obligations under the business combination agreement. MetroPCS may not terminate the business combination agreement to accept a superior acquisition proposal until after the special meeting and only then if the MetroPCS stockholders fail to approve the stock issuance proposal and the new certificate of incorporation proposal.

#### **Regulatory Filings and Other Actions**

MetroPCS and Deutsche Telekom have agreed to cooperate with each other and use their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, necessary, proper or advisable under the business combination agreement and applicable laws to consummate the transaction as promptly as reasonably practicable, including:

preparing and filing as promptly as reasonably practicable all documentation to effect necessary notices, reports and other filings (including by filing promptly on a date agreed to by the parties the notification and required form under the HSR Act and all applications and necessary and appropriate filings and any necessary and appropriate amendments required to be filed with the FCC, any PUCs or similar state or foreign regulatory bodies);

obtaining as promptly as reasonably practicable all governmental consents;

furnishing all information required or reasonably requested for any application or other filing to be made pursuant to any applicable laws in connection with the transaction;

keeping the other parties informed in all material respects of any material communication received by such party from, or given by such party to, any governmental entity and of any material communication received or given in connection with any proceeding by a private party, in each case relating to the transaction;

negotiating, proposing and/or agreeing to the sale, divestiture, license, disposition or hold separate of any asset and other actions, restrictions, limitations or conditions required to obtain any consents, registrations, approvals, permits or authorizations in connection with the transaction;

conducting or agreeing to conduct its business post-closing in any manner as necessary to obtain any consents, registrations, approvals, permits or authorizations in connection with the transaction;

agreeing to any order, action or regulatory condition of any regulatory body, whether in an approval proceeding or another regulatory proceeding; and

defending against the entry of any decree, order, or judgment that would restrain, prevent or delay the closing, including defending any lawsuits or other legal proceedings, whether judicial or administrative, challenging the business combination agreement or the completion of the transaction.

Notwithstanding the foregoing, MetroPCS and Deutsche Telekom are not required to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets (including FCC licenses), businesses or product lines of

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

MetroPCS, Deutsche Telekom, T-Mobile, any of their respective subsidiaries, or any combination thereof, that would have a material adverse effect on the business, assets, liabilities, financial condition or results of operations of MetroPCS, T-Mobile and their respective subsidiaries, taken as a whole.

In addition, MetroPCS is not permitted to agree to any actions, restrictions or conditions with respect to obtaining any consents, registrations, approvals, permits or authorizations in connection with the transaction without the prior written consent of Deutsche Telekom.

#### Financing

MetroPCS and Deutsche Telekom have agreed to cooperate in good faith to effect the financing transactions described under the section entitled Summary of the Financing beginning on page 112.

#### Expenses

Generally, the parties have agreed to bear their respective expenses, costs and fees. Notwithstanding the foregoing, Deutsche Telekom has agreed to be responsible for the fees of T-Mobile and its subsidiaries incurred in connection with the execution and delivery of the business combination agreement and the completion of the transaction. In addition, Deutsche Telekom has agreed initially to be responsible, and if the transaction closes the combined company will reimburse Deutsche Telekom at closing, for (i) fees, costs, charges and expenses incurred in connection with obtaining a credit rating pre-determination for MetroPCS and the Wireless existing notes, (ii) commitment fees, costs, charges and expenses incurred in connection with obtaining the debt financing in connection with the transaction (except as described in the immediately following sentence), (iii) fees for services provided in connection with the financial statements (including the pro forma financial statements) included in this proxy statement and (iv) HSR filing fees. MetroPCS has agreed to be responsible for (i) commitment fees, costs, charges and expenses incurred in connection with the issuance of any permitted Wireless notes in connection with the transaction and any consent solicitations or similar transactions to secure the waiver of the holders of the Wireless existing notes to any change in control resulting from the transaction, and (ii) costs and expenses of MetroPCS counsel incurred by MetroPCS or its subsidiaries in connection with the Deutsche Telekom notes issued in the transaction.

#### Governance

In connection with the transaction and conditioned on the completion of the transaction, MetroPCS will amend and restate its existing certificate of incorporation in its entirety in the form of the new certificate of incorporation attached as Exhibit A to the business combination agreement, and the new certificate of incorporation of the combined company immediately following the completion of the transaction. As part of the new certificate of incorporation, MetroPCS will change its name to T-Mobile US, Inc. or another name selected by Deutsche Telekom prior to the closing. The new certificate of incorporation will also provide for the declassification of the MetroPCS board with members generally serving one year terms and provide Deutsche Telekom will the same board and committee representation rights as are set forth in the stockholder s agreement. The new certificate of incorporation will also contain the same approval rights as are set forth in the stockholder s agreement, which restrict the combined company s ability to take certain actions without Deutsche Telekom s prior written consent as long as Deutsche Telekom beneficially owns 30% or more of the outstanding shares of the combined company s common stock. The new certificate of incorporation entitled Proposals Submitted to the Stockholders The New Certificate of Incorporation Proposal beginning on page 57.

MetroPCS also will amend and restate its existing bylaws in the form of the new bylaws attached as Exhibit B to the business combination agreement, and the new bylaws will be the bylaws of the combined company immediately following the completion of the transaction. The new bylaws will include similar amendments to reflect changes in the new certificate of incorporation, including those related to the declassification of the MetroPCS board, MetroPCS board and committee composition, stockholder action by written consent and the ability to call a special meeting of stockholders. In addition, the new bylaws will provide that special meetings of the MetroPCS board may be called by, among others, a majority of the directors on the MetroPCS board, whereas the existing bylaws of MetroPCS permit two directors to call a special meeting of the MetroPCS board.

Effective as of the closing of the transaction, the combined company s board will be comprised of eleven directors. The board will consist of (i) two existing directors of MetroPCS, who are expected to be James N. Perry, Jr. and John (Jack) F. Callahan, Jr., (ii) John J. Legere, the new chief executive officer of the combined company and (iii) eight directors designated by Deutsche Telekom. Pursuant to the stockholder s agreement, at least three of the directors on the board must be considered independent under the rules of the SEC and under applicable listing standards. MetroPCS and Deutsche Telekom have agreed to cooperate in good faith to identify each of these directors prior to the closing.

Effective as of the closing of the transaction, John J. Legere, currently President and Chief Executive Officer of T-Mobile, will serve as Chief Executive Officer of the combined company, J. Braxton Carter, currently Chief Financial Officer and Vice Chairman of MetroPCS, will serve as Chief Financial Officer of the combined company, James (Jim) C. Alling, currently Chief Operations Officer of T-Mobile, will serve as Chief Operating Officer, T-Mobile USA, and Thomas C. Keys, currently President and Chief Operating Officer of MetroPCS, will serve as Chief Operating Officer, MetroPCS. MetroPCS and Deutsche Telekom have agreed to cooperate in good faith to identify additional executive officers prior to the closing. Any MetroPCS or T-Mobile executive officer not identified prior to the closing as an executive officer of MetroPCS post-closing will be deemed not to be an executive officer of MetroPCS post-closing.

#### **Tower Transaction**

Prior to the date of the business combination agreement, T-Mobile entered into an agreement with respect to the transfer of certain of its tower assets. Pursuant to the business combination agreement, the parties agreed that T-Mobile will be permitted to consummate such agreement or any other transfer on terms that, taken as a whole, have an equivalent or more favorable economic cost to T-Mobile and its subsidiaries as compared to such agreement. On Friday, November 30, 2012, an initial closing was consummated pursuant to which nearly all of such tower assets were transferred, and the associated proceeds were received by T-Mobile. On December 7, 2012, in accordance with the business combination agreement, such proceeds (net of fees and expenses) were distributed to Deutsche Telekom. From time to time, additional subsequent closings will occur with respect to the small percentage of tower assets that were not transferred on November 30, 2012. All proceeds (net of fees and expenses) resulting from such subsequent closings that occur prior to the closing date of the business combination agreement, and resulting proceeds (net of fees and expenses) that are received, from and after the closing date of the business combination agreement, the combined company will use its reasonable best efforts to transfer such tower assets on terms acceptable to Deutsche Telekom, and to transfer the proceeds (net of fees and expenses) of such transfer to Deutsche Telekom.

#### **Intellectual Property**

Deutsche Telekom has agreed not to assert claims against the combined company or its subsidiaries regarding any intellectual property owned by Deutsche Telekom and used by T-Mobile or any of its subsidiaries immediately prior to the closing date as long as the combined company does not use such intellectual property for any activities, products and services or for any other purposes other than those for which T-Mobile used such intellectual property immediately prior to the closing date. This agreement by Deutsche Telekom is limited to the territory and does not extend to any legal successors and assignees of MetroPCS. Additionally, the agreement does not apply to any intellectual property with respect to which Deutsche Telekom and T-Mobile, or as the case may be, MetroPCS and Deutsche Telekom have made a written agreement, including the trademark license described below.

#### **Employee Matters**

For at least one year after the completion of the transaction, MetroPCS will, and Deutsche Telekom will cause MetroPCS to, provide any of MetroPCS continuing employees with compensation (other than equity compensation), severance pay and employee benefits that are substantially comparable, in the aggregate, to the

existing compensation (other than equity compensation), severance pay and benefits currently provided by MetroPCS. In addition, each continuing employee of MetroPCS will receive service credit under any new combined company benefit plans to the extent credited under MetroPCS existing benefit plans for all purposes of determining eligibility to participate, vesting and level of benefits, including for purposes of vacation, severance and paid time off benefits (but not for benefit accrual purposes, for the purpose of qualifying a subsidized early retirement benefit, or if such credit would result in a duplication of benefits).

In addition, MetroPCS will, and Deutsche Telekom will cause MetroPCS to, use its reasonable best efforts to (i) make a decision regarding whether to continue the employment of each of MetroPCS continuing employees in a position and on terms of employment comparable to such employee s position and terms of employment as of the closing and (ii) communicate such decision to each such employee and carry out such decision during the one- year period immediately following the closing.

#### **Director and Officer Indemnification**

The parties have agreed that all rights to exculpation, indemnification and advancement of expenses of MetroPCS and T-Mobile s current and former directors, officers and employees immediately prior to the closing will continue after the closing. In addition, the combined company will, and Deutsche Telekom has agreed to cause the combined company to, purchase a tail directors and officers liability insurance policy for all current and former MetroPCS directors and officers covering a period of six years following the closing for events arising at or prior to the closing.

#### **Other Covenants**

The business combination agreement contains certain other covenants and agreements, including covenants and agreements relating to:

confidentiality and access by each party to certain information about the other parties during the period prior to the closing;

cooperating in connection with public announcements;

causing the resignation of each director and officer of the parties that will not continue in such capacity following the closing;

establishing a joint working group to prepare for the integration of T-Mobile and its subsidiaries after the closing into MetroPCS internal control structure and procedures for financial reporting compliance with the requirements of Rule 404 of the Sarbanes-Oxley Act and MetroPCS financial reporting structure;

addressing during the period prior to the closing T-Mobile s contracts to which Deutsche Telekom or its affiliates is a counterparty;

making customer communications;

developing transition arrangements;

Deutsche Telekom not acquiring any MetroPCS common stock;

cooperating in the defense or settlement of any stockholder litigation relating to the transaction; and

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

MetroPCS using its reasonable best efforts to cause the MetroPCS common stock to be issued in the MetroPCS stock issuance to be approved for listing on the NYSE, subject to official notice of issuance, prior to the closing.

#### **Conditions to Closing the Transaction**

The obligations of Deutsche Telekom, Global, Holding, T-Mobile and MetroPCS to complete the transaction are subject to the satisfaction or, if permitted under applicable law, waiver of the following conditions:

the required MetroPCS stockholder approvals having been received;

the waiting period (and any extensions thereof) applicable to the completion of the transaction under the HSR Act having expired or been earlier terminated without requiring Deutsche Telekom or MetroPCS to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets, businesses or product lines of MetroPCS, Deutsche Telekom, T-Mobile, or any of their respective subsidiaries, or any combination thereof, that would cause a material adverse effect on the business, assets, liabilities, financial condition or results of operations of MetroPCS, T-Mobile and their respective subsidiaries, taken as a whole, which we refer to as a regulatory material adverse condition;

all governmental consents and waivers required to be obtained from the FCC in connection with the completion of the transaction having been granted by the FCC by final order without requiring Deutsche Telekom or MetroPCS to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets, business or product lines of MetroPCS, Deutsche Telekom, T-Mobile, or any of their respective subsidiaries, or any combination thereof, that would cause a regulatory material adverse condition, provided that such governmental consents to be granted by the FCC shall not be required to have been granted by final order in the event that (i) waiting to receive the final order would require the parties to extend the outside date described in the section entitled Summary of the Business Combination Agreement Termination beginning on page 151, (ii) all other conditions to closing are met and (iii) neither party has appealed or sought reconsideration of the authorizations granted by the FCC in connection with the completion of the transaction;

certain other governmental consents agreed upon by the parties having been obtained by final order, without requiring Deutsche Telekom or MetroPCS to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets, businesses or product lines of MetroPCS, Deutsche Telekom, T-Mobile, or any of their respective subsidiaries, or any combination thereof, that would cause a regulatory material adverse condition;

the CFIUS having terminated its review under 31 C.F.R. Part 800 and, where applicable, its investigation, without unresolved national security concerns with respect to the transaction, except as would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the business, assets, liabilities, financial condition or results of operations of MetroPCS and its subsidiaries, taken as a whole (after giving effect to the transaction), or the ability of either party to consummate the transaction, and without requiring Deutsche Telekom or MetroPCS to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets, businesses or product lines of MetroPCS, Deutsche Telekom, T-Mobile, or any of their respective subsidiaries, or any combination thereof, that would cause a regulatory material adverse condition;

all other governmental consents required to be obtained in connection with the consummation of the transaction, including from any PUCs or similar regulatory bodies and from any foreign regulatory bodies under any foreign antitrust, competition or similar laws or any foreign public service or utility commissions or similar foreign regulatory bodies, having been obtained by final order, except as would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the business, assets, liabilities, financial condition or results of operations of MetroPCS and its subsidiaries, taken as a whole (after giving effect to the transaction) or the ability of either party to consummate the transaction, and without requiring Deutsche Telekom or

MetroPCS to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets, businesses or product lines of MetroPCS, Deutsche Telekom, T-Mobile, or any of their respective subsidiaries, or any combination thereof, that would cause a regulatory material adverse condition; and

no governmental entity having enacted, issued, promulgated, enforced or entered any law, statute, ordinance, rule, regulation, judgment, injunction, decree or other order (whether temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits the completion of the transaction.

The obligations of MetroPCS to complete the transaction are also subject to the satisfaction of the following conditions:

all representations and warranties of Deutsche Telekom, Global, Holding and T-Mobile being true and correct, as of the date of the business combination agreement and as of the closing, subject in certain cases to de minimis, materiality and, in most of the cases, material adverse effect qualifiers;

each of Deutsche Telekom, Global, Holding and T-Mobile having performed in all respects its obligations to transfer at the closing to MetroPCS all of the equity interests of T-Mobile and satisfy any consideration adjustment to be paid in cash at the closing;

each of Deutsche Telekom, Global, Holding and T-Mobile having made available any notes that T-Mobile is required to issue pursuant to the business combination agreement and the working capital revolving credit facility;

each of Deutsche Telekom, Global, Holding and T-Mobile having performed in all material respects its other covenants and agreements set forth in the business combination agreement;

no circumstance having occurred that has had or would reasonably be expected to have a material adverse effect with respect to T-Mobile;

MetroPCS having received a fully executed counterpart of the stockholder s agreement and trademark license; and

T-Mobile having delivered a Certificate of Non-Foreign Status to MetroPCS; provided, that if T-Mobile fails to deliver such certificate, MetroPCS shall be permitted to withhold from the consideration otherwise payable to Deutsche Telekom pursuant to the business combination agreement any amounts required to be withheld pursuant to Section 1445 of the Code.

The obligations of Deutsche Telekom, Global, Holding and T-Mobile to complete the transaction are subject to the satisfaction of the following conditions:

all representations and warranties of MetroPCS being true and correct, as of the date of the business combination agreement and as of the closing, subject in certain cases to de minimis, materiality and, in most of the cases, material adverse effect qualifiers;

MetroPCS having performed in all respects its obligations with respect to the reverse stock split, cash payment and MetroPCS stock issuance;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

MetroPCS having performed in all material respects its other covenants and agreements set forth in the business combination agreement;

no circumstance having occurred that has had or would reasonably be expected to have a material adverse effect with respect to MetroPCS;

Deutsche Telekom having received a fully executed counterpart of the stockholder s agreement and trademark license;

Deutsche Telekom having received a payoff letter reasonably acceptable to it with respect to the termination of the Wireless existing senior credit facility and all commitments or other extensions of credit thereunder and the satisfaction and discharge of all principal, premium, if any, interest, fees and other amounts then due or outstanding thereunder and the satisfaction, release and discharge of all security interests, mortgages, liens and other encumbrances on MetroPCS and its subsidiaries properties and assets over MetroPCS securing such obligations;

except to the extent refinanced pursuant to a change in control offer on or prior to the closing, the Wireless existing notes remaining outstanding, there being no event of default in respect of any of the Wireless existing notes, and the completion of the transaction alone not giving rise to any fact, event, circumstance or effect that with notice or lapse of time would constitute an event of default in respect of any of the Wireless existing notes; and

the MetroPCS common stock to be issued to Holding or its designee in the transaction being approved for listing on the NYSE, subject to official notice of issuance.

# Termination

The business combination agreement may be terminated at any time prior to the closing, even after the receipt of the required MetroPCS stockholder approvals, under the following circumstances:

by mutual written consent of MetroPCS and Deutsche Telekom;

by either MetroPCS or Deutsche Telekom, if any governmental entity of competent jurisdiction shall have issued a final and non-appealable order or taken any other final and non-appealable action permanently enjoining, restraining, denying or otherwise prohibiting the consummation of the transaction; provided that the party seeking to terminate the business combination agreement shall have used its reasonable best efforts to have such order lifted if and to the extent required by the business combination agreement;

by either MetroPCS or Deutsche Telekom, if the transaction is not completed on or before the outside date; except that if the conditions set forth in the second, third, fourth, fifth and sixth bullet points under Summary of the Business Combination Agreement Conditions to Closing the Transaction beginning on page 149 have not been satisfied by October 3, 2013, either party may extend the outside date from time to time, by written notice to the other party given prior to the outside date in effect prior to such notice, to a date not later than January 3, 2014;

by Deutsche Telekom (provided that none of Deutsche Telekom, Global, Holding and T-Mobile is then in material breach of any representation, warranty, covenant or other agreement in the business combination agreement), in the event that (i) a breach by MetroPCS of any representation, warranty, covenant or other agreement contained in the business combination agreement would result in a failure of a condition to the closing, and (ii) such breach (A) if curable, has not been cured within 30 calendar days following MetroPCS receipt of written notice from Deutsche Telekom, or if the outside date is less than 30 calendar days from such notice, has not been or cannot reasonably be expected to be cured by the outside date, or (B) is not curable;

by MetroPCS (provided that MetroPCS is not then in material breach of any representation, warranty, covenant or other agreement in the business combination agreement), in the event that (i) a breach by Deutsche Telekom, Global, Holding or T-Mobile of any representation, warranty, covenant or other agreement contained in the business combination agreement would result in a failure of a condition to the closing, and (ii) such breach (A) if curable, has not been cured within 30 calendar days following Deutsche Telekom s receipt of written notice from MetroPCS, or if the outside date is less than 30 calendar days from such notice, has not been or cannot reasonably be expected to be cured by the outside date, or (B) is not curable;

by either Deutsche Telekom or MetroPCS, if the required MetroPCS stockholder approvals are not obtained at the special meeting, or at any adjournment or postponement thereof, at which a vote seeking such required MetroPCS stockholder approvals was taken, except that no party may terminate the business combination agreement pursuant to this bullet point if such party has breached in any material respect any of its obligations under the business combination agreement in any manner that would reasonably be expected to cause the failure to obtain the required MetroPCS stockholder approvals at the special meeting or at any adjournment or postponement thereof;

by Deutsche Telekom, prior to the receipt of the MetroPCS stockholder approvals, if there is an adverse recommendation change, whether or not such adverse recommendation change is in compliance with the requirements described in the section entitled Summary of the Business Combination Agreement Change in the MetroPCS Board's Recommendation beginning on page 144;

by Deutsche Telekom, if after the date of the business combination agreement there is a material adverse effect with respect to MetroPCS and such material adverse effect is not curable or, if curable, (i) is not cured within 30 calendar days after written notice is given by Deutsche Telekom to MetroPCS or (ii) if the outside date is less than 30 calendar days from such notice, has not been or cannot reasonably be expected to be cured by the outside date; or

by MetroPCS, if after the date of the business combination agreement there is a material adverse effect with respect to T-Mobile and such material adverse effect is not curable or, if curable, (i) is not cured within 30 calendar days after written notice is given by MetroPCS to Deutsche Telekom or (ii) if the outside date is less than 30 calendar days from such notice, has not been or cannot reasonably be expected to be cured by the outside date.

#### **Effect of Termination**

If the business combination agreement is validly terminated, the business combination agreement will become void and of no effect with no liability on the part of any party (or of any of its directors, officers, employees, agents, legal and financial advisors or other representatives), except that no termination will relieve any party of any liability or damages resulting from any willful breach of any material provision of the business combination agreement, the confidentiality agreement between MetroPCS and T-Mobile or any other agreement delivered in connection with the business combination agreement prior to its termination.

MetroPCS will be obligated to pay a termination fee of \$150 million to Deutsche Telekom if:

Deutsche Telekom terminates the business combination agreement because there has been an adverse recommendation change;

MetroPCS or Deutsche Telekom terminates the business combination agreement because the required MetroPCS stockholder approvals are not obtained following (i) a material breach by MetroPCS of the covenants requiring MetroPCS to file the proxy statement, call and hold the special meeting, not solicit alternative transaction proposals or continue to recommend that its stockholders approve the stock issuance proposal and the new certificate of incorporation proposal, and such material breach is reasonably related to the failure to obtain the required MetroPCS stockholder approvals, or (ii) an adverse recommendation change; or

MetroPCS or Deutsche Telekom terminates the business combination agreement because the required MetroPCS stockholder approvals are not obtained (other than under the circumstances described in the immediately preceding bullet point) or because the outside date has passed, and (i) an acquisition proposal has been made and is pending at the time of termination and, within twelve months after such termination, MetroPCS enters into, publicly approves or submits to its stockholders for approval, an agreement with respect to an acquisition proposal, or it consummates an acquisition proposal (which in each case need not be the same proposal or with the same party

that made the earlier proposal), or (ii) an acquisition proposal has been made but was withdrawn prior to the stockholder meeting at which MetroPCS stockholders voted not to grant the required MetroPCS stockholder approvals and, within twelve months after such termination, MetroPCS enters into, publicly approves or submits to its stockholders for approval, an agreement with respect to an acquisition proposal with the same party that made the earlier proposal that had been withdrawn.

Deutsche Telekom will be obligated to pay a termination fee of \$250 million to MetroPCS if the business combination agreement is terminated by MetroPCS or Deutsche Telekom (i) because a regulatory agency issues a final order prohibiting the completion of the transaction for regulatory law reasons or (ii) due to a failure to obtain the necessary regulatory approvals by the outside date.

#### **Specific Performance**

The parties have agreed in the business combination agreement that irreparable damage would occur if any provision of the business combination agreement were not performed in accordance with its terms and that, except in a circumstance where a termination fee is payable by one of the parties, the parties will be entitled to an injunction or injunctions to prevent breaches of the business combination agreement or to enforce specifically the performance of its terms and provisions without any requirement to post bond, in addition to any other remedy to which they may be entitled at law or in equity.

#### SUMMARY OF ANCILLARY AGREEMENTS

#### Stockholder s Agreement

In connection with the closing of the transaction, MetroPCS and Deutsche Telekom will enter into the stockholder s agreement, a form of which is attached as Exhibit C to the business combination agreement and which is incorporated by reference herein in its entirety. The following is a summary of the material provisions of the stockholder s agreement. The summary in this section and elsewhere in this proxy statement is qualified in its entirety by reference to the stockholder s agreement. This summary does not purport to be complete and may not contain all of the information about the stockholder s agreement that is important to you. We encourage you to read carefully the stockholder s agreement in its entirety before making any decisions regarding the transaction.

Pursuant to the business combination agreement, MetroPCS and Deutsche Telekom have agreed to enter into the stockholder s agreement at the completion of the transaction. The stockholder s agreement will govern Deutsche Telekom s ownership of debt and equity securities of the combined company following the completion of the transaction.

#### **Board Representation**

Pursuant to the stockholder s agreement, Deutsche Telekom generally will have the right to designate a number of individuals, each of which we refer to as a Deutsche Telekom designee, to the combined company s board and any committees thereof equal to the percentage of the combined company s common stock beneficially owned by Deutsche Telekom multiplied by the number of directors on the combined company s board (or the number of members of any committee thereof), in each case, rounded to the nearest whole number. The combined company and Deutsche Telekom will use their reasonable best efforts to cause Deutsche Telekom designees to be elected to the combined company s board. These rights will remain in effect as long as Deutsche Telekom designees then serving as directors on the combined company s board or as members of any committee of the combined company s board exceeds the number of Deutsche Telekom designees that Deutsche Telekom is entitled to designate, Deutsche Telekom will be required to cause the number of Deutsche Telekom designees then serving as directors on the MetroPCS board or as members of such committee of the combined company s board representing such excess to resign immediately as directors or committee members, as applicable.

The parties have agreed to use their reasonable best efforts to cause at least three members of the combined company s board to be considered independent under the rules of the SEC and under applicable listing standards. In addition, each Deutsche Telekom designee must not be prohibited or disqualified from serving as a director on the combined company s board pursuant to any rule or regulation of the SEC, the NYSE or any other or additional exchange on which securities of the combined company are listed or by applicable law. Deutsche Telekom will, and will cause any Deutsche Telekom designee to, provide such information as the combined company reasonably requests.

#### **Specified Actions**

Pursuant to the stockholder s agreement, as long as Deutsche Telekom beneficially owns 30% or more of the outstanding shares of the combined company s common stock, the combined company will not take the following actions without Deutsche Telekom s prior written consent, which consent Deutsche Telekom may withhold in its sole discretion:

create, incur, issue, assume or otherwise become liable for (including through a merger, acquisition or otherwise) or refinance or guarantee any indebtedness (excluding any permitted debt, as defined in the stockholder s agreement) that would result in the combined company and its subsidiaries, on a consolidated basis, having or being liable for indebtedness in an aggregate principal amount that

would result in the debt to cash flow ratio, as defined in the Deutsche Telekom notes indenture, for the combined company s most recently ended four full fiscal quarters for which financial statements are available to be greater than 5.25 to 1.0 on a *pro forma* basis as if the additional indebtedness had been incurred at the beginning of such four-quarter period;

take any action or enter into any transaction that would reasonably be expected to result in a breach of or default under any credit agreement, indenture, note, or similar instrument or security to which Deutsche Telekom or any of its affiliates is a party or is bound;

acquire (including by way of merger, recapitalization, reorganization, liquidation or dissolution) any business, debt or equity interests, operations or assets of any person, or make any investment in or loan to any person, in any single transaction or series of related transactions (excluding the acquisition of products and equipment in the ordinary course of business), for consideration in excess of \$1 billion;

sell, lease, transfer, encumber (other than permitted liens, as defined in the stockholder s agreement) or otherwise dispose of (including by way of merger, recapitalization, reorganization, liquidation or dissolution) any division, business, or operations of the combined company or any of its subsidiaries, or any equity interests of the combined company or any of its subsidiaries, in any single transaction or series of related transactions, for consideration in excess of \$1 billion;

change the size of the combined company s board;

issue any equity or equity-linked securities or other voting securities of the combined company or any of its subsidiaries, in any single transaction or series of related transactions, (i) constituting 10% or more of the then outstanding shares of the combined company s common stock (other than grants of incentive awards to officers or employees of the combined company or its subsidiaries that are approved by the combined company s board or the applicable committee thereof or issuances of securities to the combined company or any of its wholly-owned subsidiaries) or (ii) for the purpose of redeeming or purchasing any indebtedness of the combined company held by Deutsche Telekom or its affiliates;

(i) except as required in the combined company s governing documents, repurchase or redeem any equity (or equity-based) securities of the combined company or any of its non-wholly-owned subsidiaries, or (ii) make any extraordinary or in-kind dividend with respect to any of the equity (or equity-based) securities of the combined company or any of its subsidiaries, other than a dividend on a *pro rata* basis with respect to all stockholders of the combined company, or a dividend to the combined company or any of its wholly-owned subsidiaries; or

hire, or terminate without cause, the combined company s chief executive officer, or agree to do so. In addition, the combined company will not be permitted to amend its governing documents in any manner that could adversely affect Deutsche Telekom or its rights under the stockholder s agreement as long Deutsche Telekom beneficially owns 5% or more of the outstanding shares of the combined company s common stock.

#### Debt Defaults

Pursuant to the stockholder s agreement, the combined company will be required to notify Deutsche Telekom any time it is reasonably likely that the combined company will default on any indebtedness with a principal amount greater than \$75 million, which we refer to as a potential default. Thereupon, Deutsche Telekom will have the right, but not the obligation, to provide new debt financing to the combined company up to the amount of the indebtedness that is the subject of the potential default plus any applicable prepayment or other penalties, on the same terms and conditions as such indebtedness (together with any waiver of the potential default). If Deutsche Telekom elects to provide the combined company must take any actions reasonably requested by Deutsche Telekom (i) to prepare documentation

reflecting the terms and conditions of the new debt financing; (ii) to repay the indebtedness that is the subject of the potential default; and (iii) to take any other action necessary or desirable to avert the potential default.

#### Information

As long as Deutsche Telekom beneficially owns 10% or more of the outstanding shares of the combined company s common stock, the combined company will provide Deutsche Telekom with the following information and consultation rights: (i) Deutsche Telekom will be entitled to consult with the officers of the combined company with respect to the combined company s business and financial matters, including management s proposed annual operating plans, and, upon request, members of management will meet with representatives of Deutsche Telekom at mutually agreeable times and places for such consultation, including to review progress in achieving said plans; (ii) the combined company will furnish Deutsche Telekom with such available financial and operating data and other information with respect to the business and properties of the combined company and its subsidiaries as Deutsche Telekom may reasonably request; and (iii) Deutsche Telekom will be entitled to inspect all books and records and facilities and properties of the combined company at reasonable times and intervals.

Subject to the requirements of applicable law, regulations and rules, Deutsche Telekom generally has agreed to, and to cause its representatives and its Deutsche Telekom designees to, keep confidential all information of the combined company and its affiliates obtained by Deutsche Telekom and its Deutsche Telekom designees. Deutsche Telekom also will, and will cause its controlled affiliates, representatives, and directors on the MetroPCS board that are its affiliates to, comply with applicable law regarding insider trading in the combined company s securities to the extent any of them is in possession of information of the combined company and its affiliates.

Deutsche Telekom also has agreed to use its commercially reasonable efforts to, until the next subsequent filing of the combined company s Annual Report on Form 10-K, assist the combined company with the integration of T-Mobile and its subsidiaries into the combined company s internal control structure and procedures for financial reporting compliance with the requirements of Rule 404 of the Sarbanes-Oxley Act.

#### **Director Consent Rights**

During the term of the stockholder s agreement, Deutsche Telekom will not be permitted to, and will be required to cause the Deutsche Telekom designees then serving as directors on the combined company s board not to, support, enter into or vote in favor of any transaction between, or involving both (A) the combined company and (B) Deutsche Telekom or an affiliate of Deutsche Telekom, unless such transaction is approved by a majority of the directors on the combined company s board, which majority includes a majority of the directors on the combined company s board that are not affiliates of Deutsche Telekom. In addition, Deutsche Telekom has agreed that the directors on the combined company s board that are not affiliates of Deutsche Telekom will direct and make any determinations with respect to the combined company s post-closing actions relating to the adjustment of consideration under the business combination agreement.

#### Acquisitions of the Combined Company s Common Stock

Pursuant to the stockholder s agreement, Deutsche Telekom and its affiliates will generally be prohibited from acquiring more than 80.1% of the outstanding shares of the combined company s common stock unless it makes an offer to acquire all of the then remaining outstanding shares of the combined company s common stock at the same price and on the same terms and conditions as the proposed acquisition from all other stockholders of the combined company, which is either (i) accepted or approved by the majority of the directors, which majority includes a majority of the directors that are not affiliates of Deutsche Telekom, or (ii) accepted or approved by disinterested stockholders.

# Lock-Up Period

Deutsche Telekom will be prohibited from transferring any shares of the combined company s common stock during the six-month period after the closing.

# Transfers of the Combined Company s Common Stock

Following the lock-up period, Deutsche Telekom and its affiliates may freely transfer any shares of the combined company s common stock, subject to applicable law, provided that Deutsche Telekom will be prohibited from transferring any shares of the combined company s common stock in any other transaction that would result in the transferee owning more than 30% of the outstanding shares of the combined company s common stock unless such transferee offers to acquire all of the then outstanding shares of the combined company s common stock at the same price and on the same terms and conditions as the proposed transfer.

# **Registration Rights**

The stockholder s agreement will include certain registration rights for equity securities (including the combined company s common stock) and debt securities, which we refer to, collectively, as registrable securities, of the combined company beneficially owned by Deutsche Telekom and acquired in connection with the transaction or in the future. The combined company must file a shelf registration statement covering all registrable securities within 30 days after the closing, and Deutsche Telekom generally will have the right to request that the combined company file, from time to time, a registration statement or prospectus supplement to a registration statement (i) with respect to equity securities so long as it owns 5% or more of the combined company s common stock and (ii) with respect to debt securities so long as it holds any debt securities issued by the combined company.

Notwithstanding the foregoing, Deutsche Telekom will be subject to the following limitations with respect to its registration rights:

the expected proceeds from the sale of registrable securities to be included in any requested registration statement or prospectus supplement must be \$100 million or greater;

with respect to equity securities, Deutsche Telekom must wait 90 days between requests; and

the combined company will be entitled to postpone and delay, for reasonable periods of time not in excess of 60 days, and in no event more than twice in any 12-month period, the filing or effectiveness of any such requested registration statement or prospectus supplement, if one or more executive officers of the combined company determines in good faith that any such filing or the offering or sale of any equity securities thereunder would (i) impede, delay or otherwise interfere with any pending or contemplated material acquisition, disposition, corporate reorganization or other similar material transaction involving the combined company, (ii) based upon advice from the combined company s investment banker or financial advisor, materially and adversely impede, delay or otherwise interfere with any pending or contemplated financing, offering or sale of any class of securities by the combined company, (iii) require disclosure of material non-public information which, if disclosed at such time, would not be in the best interests of the combined company and its stockholders, or (iv) have a material adverse effect on the combined company.

In addition, Deutsche Telekom has piggyback registration rights with respect to any offering initiated by the combined company or any of its other stockholders. These piggyback registration rights will be subject to cutback procedures in the event the piggyback offering is oversubscribed.

Any transferee of Deutsche Telekom who acquires at least 5% of either the registrable equity securities or the registrable debt securities pursuant to a transaction that is not registered under the Securities Act will be entitled to enjoy the same registration rights as Deutsche Telekom as long as the registrable securities held by such transferee may not be sold or disposed of pursuant to Rule 144 without volume limitations at the time when such transferee seeks to exercise its registration rights.

#### Non-Competition

The stockholder s agreement will restrict Deutsche Telekom s ability to compete with the combined company in the United States, Puerto Rico and the territories and protectorates of the United States during the period beginning on the date of the closing and ending on the date that is two years after the date on which Deutsche Telekom beneficially owns less than 10% of the outstanding shares of the combined company s common stock. Specifically, during such period, neither Deutsche Telekom nor any of its controlled affiliates will be permitted to engage in providing wireless telecommunications services through a facilities-based network in the United States, Puerto Rico and the territories and protectorates of the United States, hold licenses from the FCC related to or necessary to provide such services, act as a reseller, dealer or distributor of such services in the United States, Puerto Rico and the territories and protectorates of the United States, or act as a mobile virtual network operator in the United States, Puerto Rico and the territories and protectorates of the United States. In addition, for the period commencing at the closing and expiring on the first anniversary of the termination of the trademark license in accordance with its terms, Deutsche Telekom may not manufacture, market or distribute any products or services under, or use in any way, the trademark T-MOBILE in connection with any of the activities described in the previous sentence, other than by the combined company and its affiliates in accordance with the terms of the trademark license is more fully described under the section entitled Statemary of Ancillary Agreements Trademark License beginning on page 158.

#### Voting and Support Agreement

Concurrently, and in connection, with entering into the business combination agreement, Deutsche Telekom and Madison Dearborn entered into a voting and support agreement, which we refer to as the support agreement, a form of which is attached as Exhibit E to the business combination agreement and which is incorporated by reference herein in its entirety. The following is a summary of the material provisions of the support agreement. The summary in this section and elsewhere in this proxy statement is qualified in its entirety by reference to the support agreement. This summary does not purport to be complete and may not contain all of the information about the support agreement that is important to you. We encourage you to read carefully the support agreement in its entirety before making any decisions regarding the transaction.

On October 3, 2012, Deutsche Telekom and Madison Dearborn entered into the support agreement in the form attached as Exhibit E to the business combination agreement, pursuant to which, subject to the conditions set forth therein, Madison Dearborn agreed to, among other things, vote all shares of MetroPCS common stock beneficially owned by it, as well as any additional securities which it may acquire or own, (i) in favor of the required MetroPCS stockholder approvals and (ii) against certain third party proposals to acquire MetroPCS and against any other actions that could reasonably be expected to materially impede, interfere with, delay, postpone, discourage or adversely affect the transaction or any other transactions contemplated by the business combination agreement. In addition, Madison Dearborn has agreed to substantially similar non-solicitation restrictions as those imposed upon by MetroPCS pursuant to the business combination agreement.

As of [], the record date, Madison Dearborn beneficially owned approximately []% of the outstanding MetroPCS common stock.

#### **Trademark License**

In connection with the completion of the transaction, MetroPCS and Deutsche Telekom will enter into a trademark license, which we refer to as the trademark license, a form of which is attached as Exhibit D to the business combination agreement and which is incorporated by reference herein in its entirety. The following is a summary of the material provisions of the trademark license. The summary in this section and elsewhere in this proxy statement is qualified in its entirety by reference to the trademark license. This summary does not purport to be complete and may not contain all of the information about the trademark license that is important to you. We encourage you to read carefully the trademark license in its entirety before making any decisions regarding the transaction.

Pursuant to the business combination agreement, MetroPCS and Deutsche Telekom have also agreed to enter into the trademark license, pursuant to which the combined company will receive (i) a limited, exclusive, non-revocable and royalty-bearing license to certain T-Mobile trademarks (including Internet domains) for use in connection with telecommunications and broadband products and services in the United States, (ii) a limited, non-exclusive, non-revocable and royalty-bearing license to use certain other trademarks for use in connection with telecommunications and broadband products and services in the United States and (iii) free of charge, the right to use the trademark T-Mobile as a name for the combined company. The parties have agreed to finalize the list of licensed trademarks prior to the closing.

#### Term

The initial term of the trademark license is five years. The trademark license automatically renews for an additional five year term unless the combined company provides notice of its intent not to renew the trademark license. Thereafter, the trademark license automatically renews for subsequent five year periods unless either the combined company or Deutsche Telekom provides 12 months notice prior to the expiration of the then-current term. Additionally, the combined company may terminate the trademark license at any time upon prior notice, and Deutsche Telekom can terminate the trademark license if the combined company abandons the trademarks licensed thereunder.

#### Royalty

The combined company is obligated to pay Deutsche Telekom a royalty in an amount equal to 0.25%, which we refer to as the royalty rate, of the net revenue generated by products and services sold by the combined company under the licensed trademarks. Net revenues includes all revenues generated by the combined company in connection with the sale of products and services using the licensed trademarks, including inbound roaming revenue earned by the combined company, but products and services sold by the combined company under the MetroPCS brand or trademarks owned by the combined company are excluded from the net revenue so long as licensed trademarks are not used in conjunction therewith, other than to non-prominently refer to the name of the company. On the fifth anniversary of the trademark license, the combined company and Deutsche Telekom have agreed to adjust the royalty rate to the royalty rate found under similar licenses for trademarks in the field of wireless telecommunication, broadband and information products and services in the territory through a binding benchmarking process.

#### Quality Control Requirements

The trademark license contains certain quality control requirements that the combined company is obligated to maintain. For instance, the combined company is obligated to use the licensed trademarks in accordance with the Deutsche Telekom trademark standards and guidelines and Deutsche Telekom has the right to review representative samples of products that use the licensed trademarks. Further, Deutsche Telekom must approve each advertising campaign that uses the licensed trademarks, whether in print, online or on television. Additionally, the trademark license establishes a brand advisory committee comprising two representatives from the combined company and two representatives from Deutsche Telekom. The brand advisory committee s purpose is to implement the trademarks standards and guidelines and establish procedures for approving advertising campaigns.

#### Renegotiation

The combined company and Deutsche Telekom are obligated to negotiate a new trademark license in any of the following events: (i) Deutsche Telekom has 50% or less of the voting power of the outstanding shares of capital stock of the combined company or (ii) any third party owns or controls, directly or indirectly, 50% or more of the voting power of the outstanding shares of capital stock of the combined company, or otherwise has the power to direct or cause the direction of the management and policies of the combined company. If the

combined company and Deutsche Telekom fail to agree on a new trademark license, either the combined company or Deutsche Telekom may terminate the trademark license and such termination shall be effective, in the case of clause (i) above, on the third anniversary after notice of termination and, in the case of clause (ii) above, on the second anniversary after notice of termination.

#### Wind Down

The combined company has the right to continue to sell products under the licensed trademarks for a period of one year after termination or expiration of the trademark license. Additionally, MetroPCS has the right to continue to use advertising materials bearing the licensed trademarks for a period of up to six months after termination or expiration of the trademark license.

#### Indemnification and Liability Limits

Deutsche Telekom is obligated to indemnify the combined company against trademark infringement claims with respect to certain licensed T-Mobile trademarks and has the right (but not the obligation) to indemnify the combined company against trademark infringement claims with respect to certain other licensed trademarks. If Deutsche Telekom chooses not to defend the combined company against trademark infringement claims with respect to certain other licensed trademarks, the combined company has the right to defend itself against such claim. The combined company is obligated to indemnify Deutsche Telekom against third party claims due to the combined company s advertising or anti-competitive use by the combined company of the licensed trademarks. Except for indemnification obligations and intentional misconduct, the liability of the combined company and Deutsche Telekom is limited to EUR 1 million per calendar year.

#### **Rights Agreement Amendment**

Concurrently, and in connection, with entering into the business combination agreement, MetroPCS entered into the rights agreement amendment, which is attached as Exhibit 4.1 to MetroPCS Current Report on Form 8-K filed October 3, 2012 and which is incorporated by reference herein in its entirety. The following is a summary of the material provisions of the rights agreement amendment. The summary in this section and elsewhere in this proxy statement is qualified in its entirety by reference to the rights agreement amendment. This summary does not purport to be complete and may not contain all of the information about the rights agreement amendment that is important to you. We encourage you to read carefully the rights agreement amendment in its entirety before making any decisions regarding the transaction.

On October 3, 2012, MetroPCS entered into Amendment No. 1, which we refer to as the rights agreement amendment, to the Rights Agreement, dated as of March 29, 2007, which we refer to as the rights agreement, by and between MetroPCS and American Stock Transfer & Trust Company, LLC, as Rights Agent. The rights agreement amendment provides that none of the adoption, approval, execution, delivery, announcement or performance of the business combination agreement or related transactions, nor the completion of the transaction, will (i) cause the rights (as defined in the rights agreement) to become exercisable, (ii) cause Deutsche Telekom or any of its affiliates or associates (as such terms are defined in the rights agreement) to become an acquiring person (as defined in the rights agreement) or (iii) give rise to a flip-in-event, a flip-over-event, a triggering event, a distribution date, or a stock acquisition date (as such terms are defined in the rights agreement).

# BENEFICIAL OWNERSHIP OF METROPCS DIRECTORS, EXECUTIVE OFFICERS AND PERSONS OWNING MORE THAN 5% OF THE OUTSTANDING SHARES OF METROPCS COMMON STOCK

The following table shows the amount of MetroPCS common stock beneficially owned as of October 31, 2012 (unless otherwise indicated) by each person known by MetroPCS to own beneficially more than 5% of the outstanding shares of MetroPCS common stock, by each of MetroPCS directors and named executive officers and by all directors and executive officers of MetroPCS as a group. Unless otherwise indicated, each person has sole investment and voting power over the securities listed in the table. The beneficial ownership information has been presented in accordance with SEC rules and is not necessarily indicative of beneficial ownership for any other purpose. Unless otherwise indicated below and except to the extent authority is shared by spouses under applicable law, to our knowledge, each of the persons set forth below has sole voting and investment power with respect to all shares of MetroPCS common stock shown as beneficially owned by them. The number of shares of MetroPCS common stock used to calculate each listed person s percentage ownership of each such class includes the shares of MetroPCS common stock underlying options or other convertible securities held by such person that are exercisable or vest within 60 days after October 31, 2012.

	Beneficial	Common Stock Beneficially Owned	
	Number	Percentage	
Directors and Named Executive Officers(1)(18):			
Roger D. Linquist (2)	13,391,621	3.58%	
Thomas C. Keys (3)	1,803,602	*	
J. Braxton Carter (4)	1,062,287	*	
Mark A. Stachiw (5)	865,567	*	
Dennis T. Currier (6)	118,158	*	
W. Michael Barnes (7)	341,463	*	
John (Jack) F. Callahan, Jr. (8)	91,232	*	
C. Kevin Landry (9)	1,492,400	*	
Arthur C. Patterson (10)	5,395,536	1.47%	
James N. Perry, Jr. (11)(12)	30,850,744	8.39%	
All directors and Named Executive Officers as a group (10 persons)	55,412,610	14.63%	
Beneficial Owners of More Than 5%:			
BlackRock Inc. (13)	23,979,956	6.52%	
40 East 52 <sup>nd</sup> Street			
New York, New York 10022			
Capital Research Global Investors (14)	19,363,700	5.27%	
333 South Hope Street			
Los Angeles, CA 90071			
Madison Dearborn Capital Partners IV, L.P. (11)(12)	30,581,312	8.32%	
Three First National Plaza, Suite 4600			
Chicago, IL 60602			
Paulson & Co. Inc. (15)	23,800,000	6.47%	
1251 Avenue of the Americas			
New York, New York 10020			
Prudential Financial, Inc. (16)	20,125,020	5.47%	
751 Broad Street			
Newark, New Jersey 07102-3777			
Vanguard Group, Inc.(17)	20,464,455	5.57%	
100 Vanguard Boulevard			
Malvern, PA 19355-2331			

\* Represents less than 1%.

- Unless otherwise indicated, the address of each person is c/o MetroPCS Communications, Inc., 2250 Lakeside Blvd., Richardson, Texas 75082.
- (2) Includes 6,773,148 shares of MetroPCS common Stock issuable upon exercise of options and 54,583 shares of restricted stock as granted under the MetroPCS equity incentive compensation plans, 4,863,890 shares of MetroPCS common stock held directly by Mr. Linquist, and 1,700,000 shares of MetroPCS common stock held by THCT Partners, LTD, a partnership with which Mr. Linquist is affiliated, may be deemed to be a member of a group under Section 13d-3 of the Exchange Act, and may be deemed to share voting and/or investment power with respect to the shares owned by such entities. Mr. Linquist disclaims beneficial ownership of such shares, except to the extent of his interest in such shares arising from his interests in THCT Partners, LTD. Mr. Linquist has dispositive power with respect to the MetroPCS common stock held by THCT Partners, LTD.
- (3) Includes 1,636,410 shares of MetroPCS common stock issuable upon exercise of options and 33,021 shares of restricted stock as granted under the MetroPCS equity incentive compensation plans.
- (4) Includes 951,412 shares of MetroPCS common stock issuable upon exercise of options and 20,313 shares of restricted stock as granted under the MetroPCS equity incentive compensation plans.
- (5) Includes 788,278 shares of MetroPCS common stock issuable upon exercise of options and 13,021 shares of restricted stock as granted under the MetroPCS equity incentive compensation plans.
- (6) Includes 84,790 shares of MetroPCS common stock issuable upon exercise of options and 5,729 shares of restricted stock as granted under the MetroPCS equity incentive compensation plans.
- (7) Includes 306,419 shares of MetroPCS common stock issuable upon exercise of options and 1,500 shares of restricted stock as granted under the MetroPCS equity incentive compensation plans.
- (8) Includes 80,732 shares of MetroPCS common stock issuable upon exercise of options and 1,500 shares of restricted stock as granted under the MetroPCS equity incentive compensation plans.
- (9) Includes 28,000 shares of MetroPCS common stock issuable upon exercise of stock options and 1,500 shares of restricted stock as granted to Mr. Landry under the MetroPCS equity incentive compensation plans and 7,000 shares of MetroPCS common stock held directly by Mr. Landry. All other shares attributed to Mr. Landry are owned directly by TA Atlantic and Pacific V L.P., TA Investors II L.P., TA IX L.P., TA Strategic Partners Fund A L.P. and TA Strategic Partners Fund B L.P., with which Mr. Landry is affiliated, may be deemed to be a member of a group (hereinafter referred to as TA Associates, et al) under Section 13d-3 of the Exchange Act and may be deemed to share voting and/or investment power with respect to the shares owned by such entities. Mr. Landry disclaims beneficial ownership of such shares, except to the extent of his interest in such shares arising from his interests in TA Associates, et al.
- (10) Includes 205,556 shares of MetroPCS common stock issuable upon exercise of options and 1,500 shares of restricted stock as granted to Mr. Patterson under the MetroPCS equity incentive compensation plans and 396,162 shares of MetroPCS common stock held directly by Mr. Patterson. All other shares attributed to Mr. Patterson are owned directly by ACP Family Partnership L.P., Ellmore C. Patterson Partners, ACP 2007 Accel-7 GRAT U/A/D 4/2/07 and ACP 2007 Accel-10 GRAT U/A/D 4/2/07, with which Mr. Patterson may be deemed to share voting and/or investment power with respect to the shares owned by such entities. Mr. Patterson disclaims beneficial ownership of such shares, except to the extent of his pecuniary interest therein.

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

(11) Includes 258,932 shares of MetroPCS common stock issuable upon exercise of options and 1,500 shares of restricted stock subject as granted to Mr. Perry under the MetroPCS equity incentive compensation plans and 9,000 shares of MetroPCS common stock held directly by Mr. Perry. All other shares attributed to Mr. Perry are owned directly by Madison Dearborn Capital Partners IV, L.P., which we refer to as MDCP IV, and Madison Dearborn Partners IV, L.P., which we refer to as MDCP IV, and Madison Dearborn Partners IV, L.P., which we refer to as MDCP IV and a limited partner of MDP IV, and therefore may be deemed to share voting and investment power over such shares and therefore to beneficially own such shares. Mr. Perry disclaims any beneficial ownership of such shares, except to the extent of his pecuniary interest in such shares arising from his interests in MDP IV.

- (12) MDCP IV and MDP IV may be deemed to be a group under Section 13d-3 of the Exchange Act and the shares held by MDCP IV may be deemed to be beneficially owned by MDP IV, the sole general partner of MDCP IV. As the sole members of a limited partner committee of MDP IV that has the power, acting by majority vote, to vote or dispose of the shares held directly by MDCP IV, Paul J. Finnegan and Samuel M. Mencoff have shared voting and investment power over such shares. Messrs. Finnegan and Mencoff, and MDP IV each disclaims any beneficial ownership of any shares held by MDCP IV, except to the extent of their respective pecuniary interests therein.
- (13) Based on a Schedule 13F reporting beneficial ownership as of September 30, 2012, BlackRock, Inc. as the parent holding company of the subsidiaries listed in such Schedule 13F, has sole voting power and sole dispositive power over 23,979,956 shares of MetroPCS common stock.
- (14) Based on a Schedule 13G/A reporting beneficial ownership as of October 31, 2012, Capital Research Global Investors, a division of Capital Research and Management Company, which we refer to as CRMC, is deemed to be the beneficial owner of 19,363,700 shares as a result of CRMC acting as investment adviser to various investment companies registered under Section 8 of the Investment Company Act of 1940.
- (15) Paulson & Co. Inc. and/or certain of its affiliated entities, which we refer to collectively as Paulson, provide investment management services to certain pooled investment vehicles and managed accounts, which we refer to collectively as the funds. Based on the Schedule 13F for the quarter end dated September 30, 2012 filed by Paulson, the funds may be deemed to be the beneficial owners of, in the aggregate, 23,800,000 shares of MetroPCS Communications, Inc., which we refer to as the fund shares. Paulson disclaims beneficial ownership of the fund shares for purposes of the Exchange Act.
- (16) Based on a Schedule 13F reporting beneficial ownership as of September 30, 2012, Prudential Financial, Inc. may have direct or indirect voting and/or investment discretion over 843,632 shares which are held for its own benefit or for the benefit of its clients by its separate accounts, externally managed accounts, registered investment companies, subsidiaries and/or other affiliates. Prudential Financial, Inc. is also a Parent Holding Company and the indirect parent of Jennison Associates LLC, which we refer to as Jennison, which separately filed a Schedule 13F reporting beneficial ownership as of September 30, 2012, of 19,281,388 shares held by entities to which it furnishes investment advice. Because Jennison does not file Schedule 13F jointly with Prudential shares of MetroPCS common stock reported by Jennison are not included in the shares reported by Prudential Financial, Inc.
- (17) Based on a Schedule 13F reporting beneficial ownership as of September 30, 2012, Vanguard Group, Inc. is deemed to be the beneficial owner of 20,464,455 shares as a result of Vanguard Group, Inc. (or its affiliate) acting as investment adviser to various investment portfolios.
- (18) Does not include the following number of shares of unvested restricted stock that are not beneficially owned and have no voting and/or investment power, but are held by the following individuals:

	Number
Name	of Shares
Roger D. Linquist	420,313
Thomas C. Keys	225,000
J. Braxton Carter	180,000
Mark A. Stachiw	107,813
Dennis T. Currier	51,563
W. Michael Barnes	7,500
John (Jack) F. Callahan, Jr.	7,500
C. Kevin Landry	7,500
Arthur C. Patterson	7,500
James N. Perry, Jr.	7,500

# MANAGEMENT S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS OF T-MOBILE

#### **Company Overview**

T-Mobile USA, Inc. ( T-Mobile ) is a wireless telecommunications carrier that offers mobile communications services under the T-Mobile brand in the United States, Puerto Rico and the U.S. Virgin Islands. T-Mobile operates its business in one reportable segment. T-Mobile s service offerings include contract and non-contract ( prepaid ) wireless voice, messaging and data services plans, mobile broadband plans and wholesale wireless services. As of September 30, 2012, T-Mobile provides service to more than 33.3 million customers through its nationwide voice and data networks.

As of September 30, 2012, T-Mobile s nationwide coverage footprint offers wireless voice, messaging, and data services across a service area covering over 280 million Americans in most metropolitan areas. In addition, T-Mobile has roaming agreements with other mobile communication network operators that allow mobile communications services to be provided beyond the direct coverage area of the T-Mobile network footprint. Services from T-Mobile are provided using the Global System for Mobile Communications (GSM), General Packet Radio Service (GPRS), Enhanced Data rates for GSM Evolution (EDGE), Universal Mobile Telecommunications Systems (UMTS) and Evolved High Speed Packet Access (HSPA+) technologies. T-Mobile s network modernization plan will result in the launch of 4G services using Long Term Evolution (LTE) technology in areas covering a population of 200 million in 2013.

#### Customers

T-Mobile generates revenue from three primary categories of customers: branded contract, branded prepaid and wholesale. Branded contract customers generally include customers that activated service on a contract, and branded prepaid customers include customers who generally pay in advance or did not activate service on an annual contract. Wholesale customers include Machine-to-Machine (M2M) customers and Mobile Virtual Network Operators (MVNO) customers that operate on the T-Mobile network, but are managed by wholesale partners. T-Mobile generates the majority of its revenues by providing wireless communication services to branded contract customers. Therefore, T-Mobile s ability to acquire and retain branded contract customers is significant to its business, including the generated by providing wireless communication services revenues were generated by providing wireless communication services to branded customers, and 5% for wholesale customers, roaming and other services.

#### Services and Products

T-Mobile offers affordable 4G nationwide wireless communications services through a variety of pricing plans, including contact and no-contract service plans.

In the third quarter of 2011, T-Mobile introduced Value plans which bring more choice and value to branded contract customers. Value plans provide wireless services on a subscription contract basis without customers needing to purchase a bundled handset. Customers on T-Mobile Value plans benefit from reduced monthly service charges and can choose whether to use their own compatible handset on T-Mobile s network or they can purchase a handset from T-Mobile. Customers who choose to purchase their handset from T-Mobile pay close to the full suggested retail price for the handset. Depending on their credit profile, qualifying customers have the choice of either paying for a handset at the point-of-sale or financing a portion of the purchase price over a 20-month installment period. For each handset sold, T-Mobile Value plans result in increased equipment revenue, compared to traditional bundled price plans that typically offer a discounted handset combined with higher service charges. As a result, T-Mobile Value plans resulted in increased net income during the period of activation while service revenues are lower over the service contract service period as further described in Results of Operations under Equipment Sales.

Data plan growth and smartphone adoption have significantly increased in the last three years, positively contributing to T-Mobile s branded revenues and equipment sales. Data revenues increased 55% from the first quarter of 2009 to the third quarter of 2012. Over the same period, the number of customers using smartphones increased rapidly from approximately 1 million to 12 million as of September 30, 2012. Given the current high demand for data usage by wireless customers, T-Mobile expects data usage and smartphone adoption to continue to increase.

In 2011, T-Mobile also introduced Monthly 4G plans, a branded prepaid service offering that provides customers with a variety of plan choices including unlimited talk, text and web service on the T-Mobile nationwide 4G network at flat monthly rates without the requirement of an annual contract.

T-Mobile sells services, devices and accessories through T-Mobile owned and operated retail stores and independent third party retail outlets and over the Internet through the T-Mobile website and a variety of third party web locations. T-Mobile sells a wide variety of wireless devices and accessories, including smartphones, wirelessly enabled computers (i.e., notebooks and tablets), and data cards which are manufactured by various suppliers. T-Mobile sells devices directly to consumers, as well as to agents and other third party distributors for resale. Similar to other wireless service providers, some of T-Mobile s service plans provide customers substantial equipment subsidies to initiate or upgrade service. In December 2012, T-Mobile announced that it has entered into an agreement with Apple, Inc. to sell its products in 2013.

#### **Operating Trends**

During the three years ended December 31, 2011, T-Mobile experienced declines in revenues primarily as a result of sustained net losses of branded customers and more significantly, branded contract customers. The losses of branded contract customers were driven by competitive pressures, including the exclusive launch by other wireless carriers of certain smartphones and other devices not offered by T-Mobile, including the iPhone, the promotion of deeply discounted handsets and the targeting of existing T-Mobile customers by T-Mobile s competitors. Over the three years ended December 31, 2011, branded contract customer losses were offset by net additions of wholesale customers and, in 2011, by branded prepaid customers. As a result, there has been a significant shift in T-Mobile s customer portfolio as branded contract customers declined from 80% of T-Mobile s 32.8 million total customers at the beginning of 2009 to 67% of its 33.2 million total customers as of December 31, 2011.

These declines have been partially offset by growth in average revenue per user ( ARPU ) for branded contract customers, due primarily to strong growth in revenue from data usage as a result of broad and increasing adoption of smartphones by T-Mobile customers. Over the same period, T-Mobile has also experienced declines in operating income and adjusted operating income before depreciation, amortization, impairment and other adjustments ( Adjusted OIBDA , see Performance Measures ). These declines were primarily attributable to the declines in revenue and increases in the costs of handsets sold. The increases in costs of handsets sold relates primarily to the increasing number of more expensive smartphones sold.

In response to these trends, T-Mobile launched the Challenger Strategy described below and entered into strategic initiatives to transform T-Mobile and improve its operational and financial performance. While T-Mobile continued to face intense competition and declines in branded contract customers and revenues during the nine months ended September 30, 2012, it is starting to see improvements in branded net customer losses and branded contract customer churn following increasing adoption of T-Mobile s Value plans. In addition, during the nine months ended September 30, 2012, branded prepaid revenues and customers have increased compared to the respective period in 2011, due primarily to strong prepaid customer acquisitions and improved retention following the May 2011 launch of T-Mobile s Monthly 4G plans. T-Mobile anticipates that its Challenger Strategy and initiatives described below will have a positive impact on future results of operations and cash flows.

#### Competition

The wireless telecommunications industry is highly competitive and T-Mobile faces substantial and increasing competition in all aspects of its business. T-Mobile s competitors include a wide range of facilities based carriers such as AT&T Inc. (AT&T), Verizon Wireless (Verizon) and Sprint Nextel Corporation (Sprint) and a larger number of regional providers of wireless communications services such as MetroPCS Communications, Inc. (MetroPCS) and Leap Wireless International, Inc., as well as resellers of those services. Attractive promotions offering consumers discounted handsets, heavy spending on advertising and increasing wireless market penetration all contribute to a highly competitive industry environment. T-Mobile is the fourth largest wireless carrier in the U.S. and as such lacks the size and scale advantages enjoyed by its three larger competitors, AT&T, Verizon and Sprint. AT&T and Verizon, in particular, have been successful in recent years in increasing their share of the market for contract customers. T-Mobile competes for customers based principally on differentiation of its mobile communications network services, such as network quality and coverage area, wireless device offerings, price, and customer service.

As discussed in Operating Trends, T-Mobile has experienced significant losses of branded contract customers since 2009. In response to these trends, T-Mobile launched the Challenger Strategy that focuses on key strategic initiatives to improve T-Mobile s business.

# Challenger Strategy

In January 2011, T-Mobile announced the launch of the Challenger Strategy which focused on deploying the HSPA+ network, improving the handset portfolio, making smartphones affordable, reducing customer churn, restructuring operations and driving operating efficiencies through programs such as the Company s Reinvent cost reduction program, as described below. In February 2012, subsequent to the termination of the potential acquisition of T-Mobile by AT&T, T-Mobile announced a reinvigoration of the Challenger Strategy, which included the launch of strategic initiatives for modernization of the T-Mobile network, and plans for additional investment to drive growth in the Business to Business (B2B) and MVNO market segments in addition to continuing efforts to reduce churn and drive operating efficiencies.

*Network Investment* Core to the T-Mobile Challenger Strategy is the planned investment of \$4 billion in network modernization and 4G evolution, which will improve existing voice and data coverage and pave the way for LTE service in 2013. T-Mobile expects to broadly deploy LTE in 2013 to reach over 200 million Americans by the end of that year. The timing for launching LTE allows T-Mobile to take advantage of the latest and most advanced LTE technology infrastructure, improving the overall capacity and performance of T-Mobile s 4G network, while optimizing spectrum resources. In addition, the timing allows T-Mobile to benefit from a more mature LTE device ecosystem while continuing to meet the growing demand for data with a powerful 4G experience using its HSPA+ network. Nearly 95 percent of T-Mobile device sales in the third quarter of 2012 were 3G and 4G smartphones. As data usage and smartphone adoption accelerate, fewer T-Mobile customers are utilizing 2G services. As a result, T-Mobile is able to re-farm existing spectrum holdings by reducing the amount of 1900 MHz PCS spectrum being used for GSM to deploy HSPA+ 4G services in the PCS band and make room in the Advanced Wireless Services ( AWS ) band for LTE. In addition to creating capacity for LTE deployment on AWS spectrum, transitioning HSPA+ to the PCS band will harmonize the T-Mobile spectrum bands with other U.S. and international carriers which will allow more international data roamers to use the T-Mobile network when visiting the United States, while also increasing device compatibility.

As part of T-Mobile s Challenger Strategy, T-Mobile also launched the Reinvent program with the goal of driving operational efficiencies and an improved cost structure for T-Mobile and to reinvest a portion of those savings into customer acquisition programs. The Reinvent program is designed to deliver long-term cost-savings of \$1.7 billion per year by 2015.

**B2B Market Segment Investment** The T-Mobile Challenger Strategy also includes plans for additional investment in the B2B market segment. T-Mobile estimates that the B2B market represents a growing area within the wireless telecommunications industry, and includes enterprise and small business customers whose revenues are estimated to grow more rapidly than in the consumer space. The B2B market in the United States is estimated to generate approximately \$70 billion in revenues annually. As of December 31, 2011, T-Mobile had more than five million government and business lines of service and estimates T-Mobile s market share of the B2B subscriber base to approximate 5%. T-Mobile aims to expand its B2B market share over the next four years through strategic investments that include increasing the B2B sales force, introducing compelling new B2B rate plans (such as new mobile broadband and international data roaming plans), and delivering B2B-enabling systems and solutions.

*Churn Reduction* Reducing branded contract customer churn remains a key strategic focus for T-Mobile. The monthly churn rate for T-Mobile branded contract customers has decreased from 2.6% in the third quarter of 2010 to 2.3% in the third quarter of 2012. Reducing churn is a multi-year effort that involves improving many aspects of customer interactions. The following are the key initiatives of the T-Mobile churn reduction efforts:

Improving network quality with improved in-home coverage and overall modernization

Improving customers sales and onboarding experience

Providing customers with incentives to renew contracts

Ensuring customers are on the right rate plan to most effectively meet their needs T-Mobile anticipates higher churn rates in the fourth quarter of 2012 compared to the third quarter of 2012 as a result of multiple factors. The factors include a large number of T-Mobile customers contracts expiring in the fourth quarter of 2012, competitive pressures from other wireless carriers offering the iPhone 5, and competitors holiday promotions offering discounted handset pricing on certain sought-after devices.

#### Acquisitions and Dispositions

In December 2011, Deutsche Telekom AG (Deutsche Telekom), T-Mobile s parent company, acquired the right to the transfer of spectrum licenses from AT&T as part of the break-up consideration in connection with the termination of the proposed acquisition of T-Mobile by AT&T. In May 2012, T-Mobile obtained the spectrum licenses from Deutsche Telekom in a non-cash transaction in return for satisfaction of \$1.2 billion in accounts receivable from affiliates. This transfer provided T-Mobile with AWS mobile spectrum in 128 service areas, including 12 of the top 20 metropolitan areas.

In September 2012, T-Mobile completed a spectrum license transaction with Verizon for the purchase and exchange of certain AWS spectrum licenses in 218 service areas across the U.S. The transaction improves T-Mobile s spectrum position in 15 of the top 25 metropolitan areas in the U.S. and realigns its existing spectrum holdings. Both of these transactions provide T-Mobile with critical AWS spectrum, enhancing network capacity as well as network performance, and allowing T-Mobile to meet the growing consumer demand for 4G mobile broadband.

In September 2012, T-Mobile entered into definitive agreements with Crown Castle International (CCI) regarding the sale of rights to operate approximately 7,200 wireless communication towers owned by T-Mobile. Prior to closing on November 30, 2012, the tower portfolio included in the transaction was reduced to 7,105 sites. During the fourth quarter of 2012, T-Mobile received net proceeds of \$2,469 million of which T-Mobile distributed \$2,403 million to its parent, Deutsche Telekom. Under the terms of the transaction, CCI acquired the sole rights to lease and operate the wireless communication towers through either purchase of the towers, or under long-term prepaid lease and management agreements. Furthermore, CCI was granted an option to acquire any sites that were not sold to CCI at the end of their respective least terms. T-Mobile will lease back and continue to have rights to space on the towers to operate its wireless communication equipment. The master lease agreement governing the

lease-back of tower space provides T-Mobile with an initial term and several optional renewals at lease rentals that are consistent with those it has for similar sites where T-Mobile leases space on third-party owned towers.

#### **Results of Operations**

In this section, T-Mobile discusses the overall results of its consolidated operations.

#### Year Ended December 31, 2011 Compared to Year Ended December 31, 2010

Set forth below is a summary of certain financial information for the periods indicated:

	Year Ended December 31,			
	2011	2010	Change	
	(in mi	lions)		
Revenues	¢ 1 < 220	¢ 1 < 500	(2) (7	
Branded contract revenues	\$ 16,230	\$ 16,538	(2)%	
Branded prepaid revenues	1,307	1,384	(6)%	
Wholesale revenues	443	199	**	
Roaming and other service revenues	501	612	(18)%	
Total service revenues	18,481	18,733	(1)%	
Equipment sales	1,901	2,404	(21)%	
Other revenues	236	210	12%	
Total revenues	20,618	21,347	(3)%	
Operating expenses				
Network costs, excluding depreciation and amortization	4,952	4,895	1%	
Cost of equipment sales	3,646	4,237	(14)%	
Customer acquisition, excluding depreciation and amortization	3,185	3,205	(1)%	
General and administrative, excluding depreciation and amortization	3,543	3,535	0%	
Depreciation and amortization	2,982	2,773	8%	
Impairment charges	6,420		**	
Other, net	169	(3)	**	
Total operating expenses	24,897	18,642	34%	
Operating (loss) income	(4,279)	2,705	**	
Other (expense) income				
Interest expense to affiliates	(670)	(556)	21%	
Income tax benefit (expense)	216	(822)	**	
Other, net	15	30	(50)%	
Net (loss) income	\$ (4,718)	\$ 1,357	**	

#### \*\* Not meaningful

*Branded Contract Revenues.* Branded contract revenues decreased by \$308 million, or 2%, to \$16.2 billion for the year ended December 31, 2011 from \$16.5 billion for the year ended December 31, 2010. The decrease was primarily attributable to declines in the number of branded contract customers, which decreased by 9% during 2011. This decrease in revenues was partially offset by increased data revenues from

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

customer adoption of data plans and a more robust offering of 3G/4G wireless devices. Smartphone customers accounted for 40% of total branded customers at the end of 2011, up from 28% at the end of 2010. In the fourth quarter of 2010, T-Mobile began directly providing handset insurance services, resulting in branded contract revenues of \$627

million for the year ended December 31, 2011 and \$114 million for the portion of the year ended December 31, 2010. Previously, handset insurance was provided through a third party insurance provider and respective revenues from sales of replacement handsets to the insurance provider were recorded in equipment sales, and commissions earned related to sales of handset insurance were reported within roaming and other service revenues. In addition, the introduction of reconnection fees in the third quarter of 2011 positively impacted branded contract revenues by approximately \$171 million in 2011.

*Branded Prepaid Revenues.* Branded prepaid revenues decreased by \$77 million, or 6%, to \$1.3 billion for the year ended December 31, 2011 from \$1.4 billion for the year ended December 31, 2010. The decrease was primarily attributable to the decline in the average number of branded prepaid customers due to the strategic phase-out of certain prepaid products, primarily the FlexPay No Contract product, due to the higher churn rates from these products. In 2011, T-Mobile introduced flat rate Monthly 4G plans and grew data revenues through rising sales of prepaid data plans, which helped partially offset the decline in branded prepaid revenue.

*Wholesale Revenues.* Wholesale revenues increased by \$244 million to \$443 million for the year ended December 31, 2011 from \$199 million for the year ended December 31, 2010. The increase was primarily attributable to growth in MVNO customers in 2011 compared to 2010. M2M revenues remained consistent in 2011 and 2010.

*Roaming and Other Service Revenues.* Roaming and other service revenues decreased by \$111 million, or 18%, to \$501 million for the year ended December 31, 2011 from \$612 million for the year ended December 31, 2010. The decrease was primarily the result of changes to the handset insurance program, resulting in T-Mobile selling and providing handset insurance services to customers directly rather than having the services provided through a third party. As a result of this change, commencing in the fourth quarter of 2010, all revenues for the handset insurance program have been recorded in branded contract revenues as described above. In 2010, T-Mobile recorded \$73 million in roaming and other service revenues related to handset insurance program services billing and other service fees paid to T-Mobile by the third party handset insurance provider.

*Equipment Sales.* Equipment sales decreased by \$503 million, or 21%, to \$1.9 billion for the year ended December 31, 2011 from \$2.4 billion for the year ended December 31, 2010. The decline was primarily driven by a 23% decrease in handset unit sales volumes year over year related to a 17% decrease in branded contract gross customer additions as described in Performance Measures, partially offset by higher average revenues per handset sold driven by increased smartphone adoption. In addition, in the fourth quarter of 2010, T-Mobile began directly providing handset insurance services which had previously been provided through a third party. As a result of this change, commencing in the fourth quarter of 2010, all revenues related to the handset insurance program were recorded in branded contract revenues, as described above. In 2010, T-Mobile recorded \$149 million in equipment sales for the handset insurance program for handsets sold to the third party provider to fulfill handset insurance claims from T-Mobile s customers.

*Other Revenues.* Other revenues increased by \$26 million to \$236 million for the year ended December 31, 2011 from \$210 million for the year ended December 31, 2010. The increase is primarily due to higher cell site co-location rental income.

*Network Costs.* Network costs, excluding depreciation and amortization, increased by \$57 million, or 1%, to \$5.0 billion for the year ended December 31, 2011 from \$4.9 billion for the year ended December 31, 2010. The increase was primarily attributable to higher fixed line costs and lease expense related to upgrading the mobile network to HSPA+.

*Cost of Equipment Sales.* Cost of equipment sales decreased by \$591 million, or 14%, to \$3.6 billion for the year ended December 31, 2011 from \$4.2 billion for the year ended December 31, 2010. The decrease was primarily attributable to lower handset unit sales volumes. Also contributing to the decrease in cost of equipment sales was the impact of directly providing handset insurance services which had previously been provided by a third party as described above. Prior to the fourth quarter of 2010, T-Mobile s handset insurance services were

provided by a third party, therefore, handsets used to fulfill T-Mobile s customers insurance claims were sold to the third party. The decrease in sales volumes was partially offset by a higher average cost per handset driven by increased smartphone adoption.

*Customer Acquisition.* Customer acquisition, excluding depreciation and amortization, remained relatively consistent at approximately \$3.2 billion for each of the years ended December 31, 2011 and 2010. Lower commission costs, driven by fewer branded gross customer additions, were largely offset by higher advertising costs associated with marketing T-Mobile s 4G HSPA+ network.

*General and Administrative*. General and administrative, excluding depreciation and amortization expense, remained relatively consistent year over year and was approximately \$3.5 billion for the years ended December 31, 2011 and 2010. The slight increase in 2011 compared to 2010 was primarily due to higher bad debt expense associated with new products, such as deposit products, and changes in customer mix toward subprime customers that have higher delinquency rates. This increase was partially offset by lower employee-related costs due in part to fewer customer support employees driven by lower customer care call volumes. In addition, the effects of ongoing cost management programs in 2011 helped control expense growth.

*Depreciation and Amortization.* Depreciation and amortization expense increased by \$209 million, or 8%, to \$3.0 billion for the year ended December 31, 2011 from \$2.8 billion for the year ended December 31, 2010. The increase in depreciation expense in 2011 compared to 2010 was associated with the build out of the T-Mobile 4G HSPA+ network. In addition, as a result of a study performed over the useful lives of property and equipment, T-Mobile increased the useful lives of certain asset classes effective January 1, 2010, resulting in a reduction of depreciation of approximately \$268 million for the year ended December 31, 2010 (see Management s Discussion and Analysis of Financial Condition and Results of Operations of T-Mobile Critical Accounting Policies and Estimates ).

*Impairment Charges.* Impairment charges of \$6.4 billion were recorded for the year ended December 31, 2011 as a result of T-Mobile s annual assessment for impairment of goodwill and other indefinite-lived assets. The impairment charges were recorded on goodwill and spectrum licenses in the amounts of \$3.9 billion and \$2.5 billion, respectively. In step one of the 2011 goodwill impairment test, the carrying value of the reporting unit was determined to exceed its fair value due to T-Mobile experiencing legal and regulatory challenges against a planned sale of the reporting unit, customer declines and lower service revenues, which resulted in a reduction in the estimated fair value of the reporting unit as compared to the 2010 estimate of fair value. As the estimated fair value of the reporting unit was lower than its carrying value, T-Mobile performed the second step of the impairment test, as described in Note 2 to the T-Mobile consolidated financial statements for the three years ended December 31, 2011, and recognized the non-cash impairment charge of \$3.9 billion as described above. In its annual impairment test of spectrum licenses for 2011 T-Mobile estimated the fair value using the Greenfield approach, which is an income approach. Due to adverse changes in the competitive landscape and regulatory environment in 2011, T-Mobile s management changed its assumptions on which market participants would be able to transact for the asset leading to declines in the estimated cash flows used to value the spectrum licenses. At December 31, 2011, the estimated fair value of the spectrum licenses was lower than its carrying value therefore T-Mobile recognized the impairment charge of \$2.5 billion as described above. There was no impairment recorded for the year ended December 31, 2010. For further discussion on the impairment charges and the annual assessment for impairment of goodwill and other indefinite-lived assets, see Note 6 to the T-Mobile consolidated financial statements for the year ended December 31, 2011.

*Other, net.* Other, net operating expenses of \$169 million were recorded for the year ended December 31, 2011. These costs were primarily related to employee-related costs associated with the proposed acquisition of T-Mobile by AT&T, which was terminated in December 2011.

*Interest Expense to Affiliates.* Interest expense increased by \$114 million, or 21%, to \$670 million for the year ended December 31, 2011 from \$556 million for the year ended December 31, 2010. The increase in interest

expense was a result of higher average debt balances with Deutsche Telekom, which are included in payables due to affiliates. In the fourth quarter of 2010, T-Mobile acquired additional debt from Deutsche Telekom of approximately \$5.2 billion, which resulted in higher interest expense in 2011.

*Income Tax Benefit (Expense).* Income tax benefit was \$216 million for the year ended December 31, 2011 compared to income tax expense of \$822 million for the year ended December 31, 2010. The decrease in income tax expense of \$1.0 billion in 2011 was primarily due to the impairment charges on spectrum licenses recorded in 2011, as described above. The effective tax rate in 2011 was 4.4% compared to 37.8% in 2010. The decrease in the effective tax rate in 2011 compared to 2010 was primarily due to the impairment charges recorded in 2011.

*Net (Loss) Income.* Net loss was \$4.7 billion for the year ended December 31, 2011 compared to net income of \$1.4 billion for the year ended December 31, 2010. The change from net income to net loss from 2010 to 2011 was a result of the \$6.4 billion impairment charges recorded on goodwill and spectrum licenses, which was partially offset by the related income tax benefit of \$1.0 billion for the year ended December 31, 2011.

#### Year Ended December 31, 2010 Compared to Year Ended December 31, 2009

Set forth below is a summary of certain financial information for the periods indicated:

	Year Ended December 31,			
	2010	2009	Change	
	(in mi	llions)		
Revenues	¢ 1 < 500	A 1 4 5 5 0	(1) (1	
Branded contract revenues	\$ 16,538	\$ 16,750	(1)%	
Branded prepaid revenues	1,384	1,488	(7)% **	
Wholesale revenues	199	59		
Roaming and other service revenues	612	663	(8)%	
Total service revenues	18,733	18,960	(1)%	
Equipment sales	2,404	2,403		
Other revenues	210	168	25%	
Total revenues	21,347	21,531	(1)%	
Operating expenses				
Network costs, excluding depreciation and amortization	4,895	4,936	(1)%	
Cost of equipment sales	4,237	3,856	10%	
Customer acquisition, excluding depreciation and amortization	3,205	3,382	(5)%	
General and administrative, excluding depreciation and amortization	3,535	3,442	3%	
Depreciation and amortization	2,773	2,859	(3)%	
Other, net	(3)		**	
Total operating expenses	18,642	18,475	1%	
Operating income (loss)	2,705	3,056	(11)%	
Other income (expense)				
Interest expense to affiliates	(556)	(740)	(25)%	
Income tax expense	(822)	(860)	(4)%	
Other, net	30	20	50%	
Net income	\$ 1,357	\$ 1,476	(8)%	

\*\* Not meaningful

*Branded Contract Revenues.* Branded contract revenues decreased by \$212 million, or 1%, to \$16.5 billion for the year ended December 31, 2010 from \$16.8 billion for the year ended December 31, 2009. The decrease was primarily attributable to declines in the number of branded contract customers, which decreased by 4% during 2010. The decline in branded contract revenues was partially offset by a 22% increase in branded contract data revenues from greater customer adoption of data plans.

*Branded Prepaid Revenues.* Branded prepaid revenues decreased by \$104 million, or 7%, to \$1.4 billion for the year ended December 31, 2010 from \$1.5 billion for the year ended December 31, 2009. The decrease was primarily attributable to the decline in the average number of branded prepaid customers, which decreased nearly 5% in 2010 compared to 2009.

*Wholesale Revenues.* Wholesale revenues increased by \$140 million, to \$199 million for the year ended December 31, 2010 from \$59 million for the year ended December 31, 2009. The increase was primarily attributable to growth of MVNO customers.

*Roaming and Other Service Revenues.* Roaming and other service revenues decreased by \$51 million, or 8%, to \$612 million for the year ended December 31, 2010 from \$663 million for the year ended December 31, 2009. The decrease was primarily due to roaming revenues from customers of international roaming partners using T-Mobile s network.

*Equipment Sales.* Equipment sales remained relatively consistent year over year and were \$2.4 billion for the years ended December 31, 2010 and 2009. The quantity of handset units sold and the average price per handset sold remained comparable year over year.

*Other Revenues.* Other revenues increased by \$42 million to \$210 million for the year ended December 31, 2010 from \$168 million for the year ended December 31, 2009. The increase is primarily due to higher cell site co-location rental income.

*Network Costs.* Network costs, excluding depreciation and amortization, decreased slightly by \$41 million, or 1%, and totaled approximately \$4.9 billion for each of the years ended December 31, 2010 and 2009. The decrease in network costs was primarily attributable to lower roaming expenses offset by higher fixed line costs and lease expense related to operating the T-Mobile HSPA+ network.

*Cost of Equipment Sales.* Cost of equipment sales increased by \$381 million, or 10%, to \$4.2 billion for the year ended December 31, 2010 from \$3.9 billion for the year ended December 31, 2009. The increase was primarily attributable to higher average costs per handset sold compared with the prior year driven by increasing adoption of higher cost smartphone devices.

*Customer Acquisition.* Customer acquisition, excluding depreciation and amortization, decreased by \$177 million, or 5%, to \$3.2 billion for the year ended December 31, 2009. This decrease was primarily attributable to reductions in advertising expenses as a result of cost savings initiatives. In addition, lower commission costs resulting from fewer branded customer gross additions contributed to the decrease in customer acquisition expense.

*General and Administrative*. General and administrative, excluding depreciation and amortization, increased by \$93 million, or 3%, to \$3.5 billion for the year ended December 31, 2010 from \$3.4 billion for the year ended December 31, 2009. The increase was primarily attributable to higher bad debt expense related to certain products introduced in 2010 and increased commissions related to customer retention initiatives, partially offset by lower customer service outsourcing expenses.

*Depreciation and Amortization.* Depreciation and amortization decreased by \$86 million, or 3%, to \$2.8 billion for the year ended December 31, 2010 from \$2.9 billion for the year ended December 31, 2009. The decrease was the result of changes in useful life assumptions for certain asset classes effective January 1, 2010 based upon results of an asset life study, which resulted in increases to the useful lives for certain assets. This decrease was partially offset by additional depreciation on assets acquired and put into service in connection with the build out of the HSPA+ network in 2010.

*Interest Expense to Affiliates.* Interest expense decreased by \$184 million, or 25%, to \$556 million for the year ended December 31, 2010 from \$740 million for the year ended December 31, 2009. The decrease was primarily attributable to lower average debt balances with Deutsche Telekom, which are included in payables due to affiliates in the consolidated financial statements.

*Income Tax Expense.* Income tax expense was \$822 million and \$860 million for the years ended December 31, 2010 and December 31, 2009, respectively. The decrease in income tax expense of \$38 million was primarily driven by lower net income before income taxes for 2010 compared to 2009. The effective tax rate was 37.8% and 36.9% for the years ended December 31, 2010 and December 31, 2009, respectively.

*Net Income*. Net income decreased \$119 million, or 8%, to approximately \$1.4 billion for the year ended December 31, 2010 from approximately \$1.5 billion for the year ended December 31, 2009. The decrease in net income was driven by the factors described above.

### Nine Months Ended September 30, 2012 Compared to Nine Months Ended September 30, 2011

Set forth below is a summary of certain financial information for the periods indicated:

	Nine Mont Septem 2012	ber 30, 2011	Change
Revenues	(in mil	lions)	
Branded contract revenues	\$ 11,105	\$ 12,264	(9)%
Branded prepaid revenues	1,241	957	30%
Wholesale revenues	407	315	29%
Roaming and other service revenues	333	380	(12)%
Total service revenues	13,086	13,916	(6)%
Equipment sales	1,524	1,352	13%
Other revenues	200	171	17%
Total revenues	14,810	15,439	(4)%
Operating expenses			
Network costs, excluding depreciation and amortization	3,515	3,750	(6)%
Cost of equipment sales	2,456	2,772	(11)%
Customer acquisition, excluding depreciation and amortization	2,323	2,364	(2)%
General and administrative, excluding depreciation and amortization	2,681	2,643	1%
Depreciation and amortization	2,391	2,221	8%
Impairment charges	8,134		**
Restructuring costs	90		**
Other, net	(136)	64	**
Total operating expenses	21,454	13,814	55%
Operating (loss) income	(6,644)	1,625	**
Other income (expense)			
Interest expense to affiliates	(487)	(481)	1%
Income tax expense	(272)	(469)	(42)%
Other, net	75	4	**
Net (loss) income	\$ (7,328)	\$ 679	**

#### \*\* Not meaningful

*Branded Contract Revenues.* Branded contract revenues decreased by \$1.2 billion, or 9%, to \$11.1 billion for the nine months ended September 30, 2012 from \$12.3 billion for the nine months ended September 30, 2011. The decrease was primarily attributable to the 10% decline in the number of branded contract customers from September 30, 2011 to September 30, 2012. Branded contract revenues were also negatively impacted by the growth in Value plans. Compared to traditional bundled price plans, Value plans result in recording lower service revenues over the service contract period, while recognizing higher equipment revenues at the time of the sale as Value plans do not include subsidized (discounted) handsets. These decreases were partially offset by a \$137 million, or 4%, increase in data revenues from customer adoption of data plans and a more robust offering of 3G/4G devices. Smartphone customers accounted for 57% of total branded contract customers at the end of September 2012, up from 44% at the end of September 2011. In addition, T-Mobile introduced reconnection fees in the third quarter of 2011, which totaled \$182 million and \$87 million for the nine months ended September 30, 2012 and 2011, respectively.

*Branded Prepaid Revenues.* Branded prepaid revenues increased by \$284 million, or 30%, to \$1.2 billion for the nine months ended September 30, 2012 from \$957 million for the nine months ended September 30, 2011. The increase was primarily attributable to growth in branded prepaid customers driven by the success of Monthly 4G plans, which were introduced in the second quarter of 2011.

*Wholesale Revenues.* Wholesale revenues increased by \$92 million, or 29%, to \$407 million for the nine months ended September 30, 2012 from \$315 million for the nine months ended September 30, 2011. The increase was primarily attributable to growth in MVNO customers during 2012.

*Roaming and Other Service Revenues.* Roaming and other service revenues decreased by \$47 million, or 12%, to \$333 million for the nine months ended September 30, 2012 from \$380 million for the nine months ended September 30, 2011. The decrease was primarily attributable to lower roaming data revenues due to rate reductions with certain roaming partners.

*Equipment Sales.* Equipment sales increased by \$172 million, or 13%, to \$1.5 billion for the nine months ended September 30, 2012 from \$1.4 billion for the nine months ended September 30, 2011. The increase was primarily attributable to sales program changes in connection with T-Mobile Value plans. Sales of handsets in connection with T-Mobile Value plans result in higher equipment revenues compared to traditional bundled price plans, because handsets are sold separately and are discounted less at point of sale. Additionally, smartphone sales growth contributed to the year over year increase in equipment sales as smartphones have higher revenue per unit sold compared to other phones.

*Network Costs.* Network costs excluding depreciation and amortization decreased by \$235 million, or 6%, to \$3.5 billion for the nine months ended September 30, 2011. The decrease was primarily attributable to lower roaming expenses related to a decline in customer base and associated usage compared to the nine months ended September 30, 2011.

*Cost of Equipment Sales.* Cost of equipment sales decreased by \$316 million, or 11%, to \$2.5 billion for the nine months ended September 30, 2012 from \$2.8 billion for the nine months ended September 30, 2011. The decrease was primarily attributable to lower handset unit sales volumes, which decreased 12% year over year.

*Customer Acquisition.* Customer acquisition excluding depreciation and amortization decreased by \$41 million, or 2%, to \$2.3 billion for the nine months ended September 30, 2012 from \$2.4 billion for the nine months ended September 30, 2011. The decrease was primarily attributable to lower commissions expense on fewer customer gross additions partially offset by an increase in advertising expense.

*General and Administrative*. General and administrative, excluding depreciation and amortization increased slightly to \$2.7 billion for the nine months ended September 30, 2012 from \$2.6 billion for the nine months ended September 30, 2011. The slight increase was primarily attributable to higher bad debt expense partially offset by lower employee-related expenses driven by a reduction in workforce as a result of restructuring initiatives, which were implemented in the first half of 2012.

*Depreciation and Amortization.* Depreciation and amortization increased by \$170 million, or 8%, to \$2.4 billion for the nine months ended September 30, 2012 from \$2.2 billion for the nine months ended September 30, 2011. The increase was due to accelerated depreciation recorded on equipment to be replaced in connection with the network modernization efforts.

*Impairment Charges.* An impairment charge of \$8.1 billion was recorded on goodwill for the nine months ended September 30, 2012 (see Note 5 to the T-Mobile condensed consolidated financial statements for the nine months ended September 30, 2012). T-Mobile assesses the carrying value of its goodwill and other indefinite-lived intangible assets (spectrum licenses) for potential impairment annually as of December 31 or more frequently if events or changes in circumstances indicate that such assets might be impaired. On October 3, 2012,

Deutsche Telekom and MetroPCS announced the business combination agreement, which provides for the combination of T-Mobile and MetroPCS. Under the terms of the business combination agreement, Deutsche Telekom will receive 74% of the fully-diluted shares of common stock of the combined company (calculated pursuant to the business combination agreement) in exchange for its contribution of all of T-Mobile s common stock. T-Mobile determined that the announced transaction was a triggering event for a goodwill impairment assessment. The fair value of T-Mobile implied by using the market value of MetroPCS and the exchange terms contemplated in the business combination agreement was less than the carrying amount, including goodwill, of the T-Mobile s single reporting unit as of September 30, 2012. T-Mobile used the fair value implied by the transaction to estimate the fair value of the reporting unit in step one of its goodwill impairment test. Because the first step in the impairment process indicated that the carrying value exceeded the fair value of the reporting unit as of September 30, 2012, T-Mobile performed the second step in the goodwill impairment test. As a result of the second step impairment analysis, T-Mobile concluded that the implied goodwill was \$0, and recognized a non-cash impairment charge of \$8.1 billion for the nine months ended September 30, 2012. T-Mobile attributes this impairment to the business impacts from the highly competitive environment in the U.S. wireless telecommunications industry and the ongoing challenges in attracting and retaining branded contract customers. No impairment charges were incurred in the nine months ended September 30, 2011.

*Restructuring Costs.* Restructuring costs of \$90 million were recorded in the nine months ended September 30, 2012. During the nine months ended September 30, 2012, T-Mobile consolidated call center operations from 24 to 17 facilities and also restructured and optimized operations in other parts of the business in order to strengthen T-Mobile s competiveness in the U.S. marketplace. No restructuring costs were incurred during the nine months ended September 30, 2011.

*Other, net.* Other, net in the operating expenses section reflects the net effect of transaction related gains and costs. The nine months ended September 30, 2012 reflected a net gain of \$136 million compared to net cost of \$64 million for the nine months ended September 30, 2011. The net gain for the nine months ended September 30, 2012 primarily relates to a gain recorded on an AWS spectrum license exchange, partially offset by expenses associated with the terminated AT&T acquisition of T-Mobile. The net costs recorded during the nine months ended September 30, 2011 primarily related to employee costs associated with the terminated AT&T acquisition of T-Mobile.

*Interest Expense to Affiliates.* Interest expense was consistent at \$487 million for the nine months ended September 30, 2012 compared to \$481 million for the nine months ended September 30, 2011.

*Income Tax Expense*. Income tax expense was \$272 million and \$469 million for the nine months ended September 30, 2012 and 2011, respectively. The decrease in income tax expense for the nine months ended September 30, 2012 compared to the same period in 2011 was primarily due to tax benefit recorded on goodwill impairment in 2012. The effective tax rate was (3.9)% and 40.8% for the nine months ended September 30, 2012 and 2011, respectively. The decrease in the effective tax rate for the nine months ended September 30, 2012 compared to the same period in 2011 was primarily due to the goodwill impairment recorded in 2012.

*Net (Loss) Income*. Net loss was \$7.3 billion for the nine months ended September 30, 2012 compared to net income of \$679 million for the nine months ended September 30, 2011. The change from net income to net loss was a result of the \$8.1 billion impairment charge recorded on goodwill for the nine months ended September 30, 2012.

### **Critical Accounting Policies and Estimates**

The following discussion and analysis of T-Mobile s financial condition and results of operations are based upon the T-Mobile consolidated financial statements, which have been prepared in accordance with GAAP. You should read this discussion and analysis in conjunction with the T-Mobile consolidated financial statements and the related notes thereto contained elsewhere in this proxy statement. The preparation of financial statements in

conformity with GAAP requires the use of estimates and assumptions that affect the reported amounts of certain assets, liabilities, revenues and expenses, and related disclosure of contingent assets and liabilities at the date of the financial statements. T-Mobile bases its estimates on historical experience and on various other assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions.

T-Mobile believes the following critical accounting policies affect the more significant judgments and estimates used in the preparation of the T-Mobile consolidated financial statements.

#### Allowances for Uncollectible Accounts

T-Mobile maintains an allowance for estimated losses resulting from the failure of customers to make required payments. When determining the allowance, T-Mobile considers the probability of recovery based on past experience taking into account current collection trends and general economic factors. Collection risks are assessed for each type of receivable based upon historical and expected write-offs, net of recoveries, and an analysis of the aged accounts receivable balances with reserves generally increasing as the receivable ages. To the extent that actual loss experience differs significantly from historical trends or assumptions, the required allowance amounts could differ from the estimate. Total allowance for uncollectible accounts receivable as of September 30, 2012 and December 31, 2011 was approximately 13.2% and 11.3%, respectively, of the total amount of gross accounts receivable, including long-term accounts receivable. A 10% change in the amount estimated to be uncollectible as of December 31, 2011, would result in a corresponding change in bad debt expense of approximately \$25 million.

#### Depreciation

The depreciation of assets, including underlying management estimates of useful lives, is described in Note 5 to the T-Mobile consolidated financial statements for the three years ended December 31, 2011. Depreciable life studies are performed periodically to confirm the appropriateness of depreciable lives for certain categories of property and equipment. These studies take into account actual usage, physical wear and tear, replacement history and assumptions about technology evolution. When these factors indicate that an asset s useful life is different from the previous assessment, the remaining book values are depreciated prospectively over the adjusted remaining estimated useful life.

As a result of a study performed by T-Mobile over the useful lives of its property and equipment, T-Mobile increased the useful lives of certain asset classes effective January 1, 2010, resulting in a reduction of depreciation of approximately \$268.0 million for the year ended December 31, 2010. There were no changes in useful lives as a result of depreciable life studies for the nine month period ended September 30, 2012 or the years ended December 31, 2011 or 2009. In connection with an assessment of the reasonably assured lease term of cell site leases T-Mobile shortened useful life assumptions for certain long-lived assets tied to cell sites effective July 1, 2010 resulting in approximately \$76.0 million additional depreciation expense in the year ended December 31, 2010.

#### Evaluation of Goodwill and Indefinite-Lived Intangible Assets for Impairment

T-Mobile assesses the carrying value of goodwill and other indefinite-lived intangible assets (including spectrum licenses) for potential impairment annually as of December 31 or more frequently if events or changes in circumstances indicate that such assets might be impaired. Indicators of impairment include the impacts of significant adverse changes in legal factors, market and economic conditions, T-Mobile s operational performance and ability to achieve strategic plans, actions by regulators, changes in competition and market share, the potential for the sale or disposal of all or a significant portion of the T-Mobile business, and possible significant adverse changes to the cost or continued availability of capital for the expansion and enhancement of the T-Mobile wireless network.

T-Mobile uses a two-step process to determine a possible impairment of goodwill. In the first step, T-Mobile compares the fair value of the reporting unit, calculated using a market approach and/or a discounted cash flow method, to its carrying value. When using a discounted cash flow method, the future cash flow assumptions are based upon T-Mobile s estimates of revenues, OIBDA margin and a long-term growth rate taking into consideration expected industry and market conditions. The resulting cash flows are then discounted using a weighted average cost of capital reflecting the risks associated with the business and the projected cash flows. T-Mobile evaluates each significant assumption, both individually and in the aggregate, to determine the reasonableness of the fair value of the reporting unit. If the carrying amount of the reporting unit exceeds the fair value, the second step of the test is performed to measure the amount of impairment loss, if any.

In the second step, T-Mobile determines the fair values of all of the assets and liabilities of the reporting unit, including those that currently may not be recorded. The excess of the fair value of the reporting unit over the sum of the fair value of all of those assets and liabilities represents the implied goodwill amount, which is then compared to the recorded goodwill. If the implied fair value of goodwill is lower than the carrying amount of goodwill, then an impairment loss is recognized.

T-Mobile tests its spectrum licenses for impairment on an aggregate basis, consistent with the management of the overall business at a national level. T-Mobile estimates the fair value of the licenses using the Greenfield approach, which is an income approach that estimates the price at which an orderly transaction to sell the asset would take place between market participants at the measurement date under current market conditions. The Greenfield approach values the licenses by calculating the cash flow generating potential of a hypothetical start-up company that goes into business with no assets except the asset to be valued (in this case, licenses). The value of the licenses can be considered as equal to the present value of the cash flows of this hypothetical start-up company. T-Mobile bases the assumptions underlying the Greenfield approach on a combination of market participant data and T-Mobile s historical results, trends and business plans. Future cash flows in the Greenfield approach are based on estimates and assumptions of market participant revenues, OIBDA margin, network build-out period, and a long-term growth rate for a market participant taking into consideration expected industry and market conditions. The cash flows are discounted using a weighted average cost of capital reflecting the risks associated with the business and the projected cash flows. If the carrying amount of spectrum licenses exceeds the fair value, an impairment loss is recognized.

The valuation approaches utilized to estimate fair value for the purposes of the T-Mobile impairment tests of goodwill and spectrum licenses require the use of assumptions and estimates, which involve a degree of uncertainty, including primarily revenues, OIBDA margins and long-term growth rates. If actual results or future expectations are not consistent with the assumptions, this may result in the recording of significant impairment charges on goodwill or spectrum licenses. The most significant assumptions within the valuation models are the discount rate, revenues, OIBDA margins and the long-term growth rate. As a result of an impairment test performed as of September 30, 2012, T-Mobile recorded an impairment charge on goodwill. For further discussion of the impairment test performed as of September 30, 2012 and the impairment charge recorded, see Note 5 to the T-Mobile condensed consolidated financial statements for the nine month period ended September 30, 2012. As the result of the annual impairment test in 2011, T-Mobile recorded impairment charges on goodwill and spectrum licenses. For further discussion of the annual impairment test and impairment charges recorded for the year ended December 31, 2011, see Notes 2 and 6 to the T-Mobile consolidated financial statements for the valued December 31, 2011, see Notes

#### Fair Value of Financial Instruments

T-Mobile accounts for certain assets and liabilities at fair value. Fair value is a market-based measurement that should be determined based on assumptions that market participants would use in pricing an asset or liability. As a basis for considering such assumptions, the Company uses a three-tier value hierarchy, which prioritizes the inputs used in measuring fair value as follows:

Level 1 Observable inputs that reflect quoted prices in active markets for identical assets or liabilities;

Level 2 Inputs other than the quoted prices in active markets that are observable either directly or indirectly, and;

Level 3 Unobservable inputs for which there is little or no market data, which require the Company to develop its own assumptions.

As of December 31, 2011 and September 30, 2012, T-Mobile did not have any significant assets or liabilities balances categorized as Level 3 in the three-tier hierarchy. As of December 31, 2010, the only asset for which the fair value measurement is classified as Level 3 in the three-tier hierarchy was T-Mobile s investments in auction rate securities, which is disclosed in Note 10 of the T-Mobile consolidated financial statements for the three years ended December 31, 2011. T-Mobile used a discounted cash flow valuation model and an internal analysis by management of other-than-temporary impairment factors to determine the fair value. The estimates and assumptions include terms, credit rating, management s assumptions regarding expected liquidation rate, other market transactions and information received by third party brokers.

Derivative instruments are recorded on the balance sheet at their fair values and are classified as Level 2 in the three-tier value hierarchy. T-Mobile s derivative instruments are comprised of interest rate and cross currency interest rate swaps entered into in connection with long-term borrowings from related parties. T-Mobile uses discounted cash flow techniques to determine the fair value, which incorporates market-based observable inputs such as interest rates, credit spreads, forward currency exchange rates and estimates and assumptions such as instrument s term, notional amounts, discount rate and credit risk.

#### **Rent** Expense

Most of the leases for T-Mobile s tower sites have fixed rent escalations which provide for periodic increases in the amount of rent payable over time. T-Mobile calculates straight-line rent expense for each of these leases based on the fixed non-cancellable term of the lease plus all periods, if any, for which failure to renew the lease imposes a penalty on T-Mobile in such amount that a renewal appears, at lease inception, to be reasonably assured. T-Mobile is therefore required at lease inception to make significant assumptions in determining and assessing the factors that constitute a penalty . In doing so, T-Mobile primarily considers costs incurred in acquiring and developing new sites, the useful life of site improvements and equipment costs, future economic conditions and the extent to which improvements in wireless technologies can be incorporated into a current assessment of whether an economic compulsion will exist in future to renew a lease.

#### Income Taxes

T-Mobile s estimate of income taxes and the significant items giving rise to deferred tax assets and liabilities includes recent cumulative earnings experience by taxing jurisdiction, expectations of future income, the carry forward periods available for tax reporting purposes, and other relevant factors. A valuation allowance is maintained against deferred tax assets when it is more likely than not that some portion or all of the deferred tax assets will not be realized. Actual income taxes could vary from these estimates due to future changes in income tax law or the final review of T-Mobile tax returns by federal, state or foreign tax authorities.

T-Mobile accounts for uncertainty in income taxes recognized in the financial statements in accordance with Accounting Standards Codification 740. T-Mobile uses judgment to determine whether it is more likely than not

that a tax position will be sustained upon examination based on the technical merits of the position and adjusts the unrecognized tax benefits in light of changes in facts and circumstances, such as changes in tax law, interactions with taxing authorities and developments in case law.

#### **Performance Measures**

In managing the T-Mobile business and assessing financial performance, T-Mobile supplements the information provided by financial statement measures (GAAP measures) such as operating income (loss) with non-GAAP measures, including Adjusted OIBDA, which measures the financial performance of operations, and several customer focused performance metrics that are widely used in the wireless communications industry. In addition to metrics involving the numbers of customers, these metrics also include ARPU, which measures service revenue per customer. For a reconciliation of non-GAAP performance measures and a further discussion of these measures, see the Management s Discussion and Analysis of Financial Condition and Results of Operations of T-Mobile Reconciliation of non-GAAP Financial Measures .

The following table sets forth the number of ending customers and net customer additions (losses) for the years ended December 31, 2011, 2010 and 2009.

	Year Ended December 31,		
	2011	2010 (in thousands)	2009
Customers, end of period			
Branded contract customers	22,367	24,574	25,642
Branded prepaid customers	4,819	4,497	5,011
Total branded customers	27,186	29,071	30,654
M2M customers	2,429	1,873	1,122
MVNO customers	3,569	2,790	2,015
Total wholesale customers	5,999	4,663	3,137
Total T-Mobile USA customers, end of period	33,185	33,734	33,790
Net customer additions (losses)			
Branded contract customers	(2,206)	(1,069)	(619)
Branded prepaid customers	321	(513)	(53)
Total branded customers	(1,885)	(1,582)	(672)
M2M customers	556	751	577
MVNO customers	780	775	1,127
Total wholesale customers	1,336	1,526	1,705
Total T-Mobile USA net customer additions (losses)	(549)	(56)	1,033

Note : Certain customer numbers may not add due to rounding.

#### **Total Customers**

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

A customer is defined as a SIM card with a unique T-Mobile mobile identity number which generates revenue. Branded contract and branded prepaid customers are categorized depending on the type of rate plan selected. Branded contract customers generally include customers that are on a contract, and branded prepaid customers include customers who generally pay in advance or did not activate service with an annual contract. Wholesale customers include M2M and MVNO customers that operate on the T-Mobile network, but are managed by wholesale partners.

T-Mobile net customer losses were 549,000 for the year ended December 31, 2011, compared to 56,000 net customer losses for the year ended December 31, 2010. At December 31, 2011, T-Mobile had 33.2 million customers, a 1.6% decrease from the customer total as of December 31, 2010. The decrease in total customers was primarily driven by fewer branded contract customer additions and higher wholesale customer churn, specifically MVNO customers.

Net customer losses were 56,000 for the year ended December 31, 2010, compared to 1,033,000 net customer additions for the year ended December 31, 2009. At December 31, 2010, T-Mobile had 33.7 million customers, a 0.2% decrease from the customer total as of December 31, 2009. The decrease in total customers was primarily driven by a fewer branded customer gross additions partially offset by higher churn from MVNO customers.

#### **Branded** Customers

Branded contract net customer losses were 2,206,000 for the year ended December 31, 2011, compared to 1,069,000 branded contract net customer losses for the year ended December 31, 2010. The increase in branded contract customer losses was primarily attributable to fewer branded contract customer gross additions. Branded contract customer gross additions decreased in part due to the discontinuation of certain products such as T-Mobile s FlexPay Contract product, as a focus on improving overall quality of the branded contract customer base. In addition, increased competitive pressures, including the introduction of two additional national competitors (Verizon and Sprint) offering the iPhone in 2011 and the launch of the iPhone 4S by competitors in the fourth quarter of 2011, contributed to branded customer losses.

Branded contract net customer losses were 1,069,000 for the year ended December 31, 2010, compared to 619,000 branded contract net customer losses for the year ended December 31, 2009. The increase in branded contract net customer losses was primarily attributable to fewer branded contract customer gross additions as a result of competitive intensity. T-Mobile s customer performance has been impacted by our competitors ability to negotiate exclusive rights to innovative devices, such as the iPhone, that have proven to be attractive to wireless consumers. During 2010, several touchscreen smartphones were introduced to the market, supported by aggressive handset pricing and nationwide advertising drawing consumer demand away from T-Mobile.

Branded prepaid net customer additions were 321,000 for the year ended December 31, 2011, compared to 513,000 net customer losses for the year ended December 31, 2010. The significant improvement in net prepaid customer additions in 2011 was due primarily to growth of Monthly 4G plans. In addition, the improvement in branded prepaid net customer additions was due to lower churn, including the impacts from discontinuing certain products with higher churn, such as FlexPay No-Contract, in 2011.

Branded prepaid net customer losses were 513,000 for the year ended December 31, 2010, compared to 53,000 net customer losses for the year ended December 31, 2009. The increase in branded prepaid customer losses was due to fewer branded prepaid customer additions and higher branded prepaid customer deactivations driven by competitive intensity. During 2010, T-Mobile s prepaid competitors began offering monthly billed prepaid service products, including data, which were popular with consumers. This trend negatively impacted T-Mobile s customer deactivations until the launch of the T-Mobile Monthly 4G plans in 2011.

#### Wholesale

Wholesale net customer additions were 1,336,000 for the year ended December 31, 2011, compared to net customer additions of 1,526,000 for the year ended December 31, 2010. The decrease in wholesale customer net additions was driven primarily by higher MVNO customer deactivations in 2011 compared to 2010. The increase in customer deactivations in 2011 resulted from a relatively consistent rate of churn year-over year being applied against a growing MVNO customer base.

Wholesale net customer additions were 1,526,000 for the year ended December 31, 2010, compared to net customer additions of 1,705,000 for the year ended December 31, 2009. The decrease in wholesale customer net additions was driven primarily by higher MVNO customer deactivations compared to 2009. The increase in customer deactivations is a reflection of a consistent churn rate year-over-year applied against a larger MVNO customer base.

### Churn

	Year	Year Ended December 31,		
	2011	2010	2009	
Branded churn	3.30%	3.20%	3.10%	
Branded contract churn	2.70%	2.40%	2.40%	
Branded prepaid churn	6.70%	7.60%	7.10%	

Churn is defined as the number of customers whose service was discontinued, expressed as a rounded monthly percentage of the average number of customers during the specified period. T-Mobile believes that churn, which is a measure of customer retention and loyalty, provides relevant and useful information and is used by management to evaluate the operating performance of T-Mobile s business.

Branded churn was 3.30% for the year ended December 31, 2011, compared to 3.20% for the year ended December 31, 2010 and 3.10% for the year ended December 31, 2009. This year over year increase in 2011 was primarily due to intense competition in the industry. The success of Monthly 4G plans introduced in 2011 partially offset the churn increase.

Branded contract churn was 2.70% for the year ended December 31, 2011, compared to 2.40% for the year ended December 31, 2010 and 2.40% for the year ended December 31, 2009. This year over year increase of branded contract churn in 2011 over 2010 was primarily due to higher churn of branded contract customers as a result of competitive market conditions including promotions by our competitors offering handset discounts and the impact of large national competitors such as Verizon, AT&T and Sprint, but not T-Mobile, offering the iPhone. Additionally, in 2011 T-Mobile experienced customer losses which it believes may relate to uncertainty caused by the pending, but ultimately unsuccessful, acquisition of T-Mobile by AT&T.

Branded prepaid churn was 6.70% for the year ended December 31, 2011, compared to 7.60% for the year ended December 31, 2010 and 7.10% for the year ended December 31, 2009. The year over year decrease in branded prepaid churn in 2011 was driven primarily by the discontinuation of certain products that historically had higher churn. The year over year increase in branded prepaid churn in 2010 primarily related to a high churn profile associated with T-Mobile s Flex-Pay No Contract product.

#### Average Revenue Per User

	Year I	Year Ended December 31,		
	2011	2010	2009	
ARPU (blended)	\$ 45.86	\$46.33	\$47.43	
ARPU (branded contract)	\$ 57.56	\$ 54.78	\$ 53.59	
ARPU (branded prepaid)	\$ 24.27	\$ 24.18	\$ 24.82	
Data ARPU (branded contract)	\$ 17.07	\$ 13.92	\$ 11.03	

ARPU represents the average monthly service revenue earned from customers. Blended ARPU is calculated by dividing service revenues for the specified period by the average customers during the period, and further dividing by the number of months in the period. Branded contract ARPU is calculated by dividing branded

contract service revenues for the specified period by the average branded contract customers during the period, and further dividing by the number of months in the period. Branded prepaid ARPU is calculated by dividing branded prepaid service revenues for the specified period by the average branded prepaid customers during the period, and further dividing by the number of months in the period. Branded contract data ARPU is calculated by dividing branded contract data revenues for the specified period by the average branded contract customers during the period, and further dividing by the number of months in the period. T-Mobile believes ARPU provides management with useful information to evaluate the service revenues generated from T-Mobile s customer base.

Blended ARPU, which represents ARPU attributable to T-Mobile s entire business, was \$45.86 and \$46.33 for the years ended December 31, 2011 and 2010, respectively, a decrease of \$0.47. The slight decrease in blended ARPU was primarily attributable to the change in customer portfolio mix towards branded prepaid and wholesale customers, which traditionally have lower ARPU. Blended ARPU was \$47.43 for the year ended December 31, 2009. The \$1.10 decrease in blended ARPU from the year ended December 31, 2010 was primarily attributable to losses of higher generating ARPU customers.

Branded contract ARPU was \$57.56 for the year ended December 31, 2011, an increase from \$54.78 for the year ended December 31, 2010 and \$53.59 for the year ended December 31, 2009. The year over year increase in branded contract ARPU in 2011 and 2010 was a result of data revenue growth, partially offset by lower voice revenues.

Branded prepaid ARPU was \$24.27 for the year ended December 31, 2011, a slight increase from \$24.18 for the year ended December 31, 2010, but down from \$24.82 for the year ended December 31, 2009. The slight year over year increase in branded prepaid ARPU in 2011 was driven by growth of Monthly 4G products that were introduced in the third quarter of 2011. In 2010, branded prepaid ARPU decreased compared to 2009 as a result of a change in the mix of customers.

Branded contract data ARPU for the year ended December 31, 2011 was \$17.07 compared to \$13.92 for the year ended December 31, 2010 and \$11.03 for the year ended December 31, 2009. Significant increases in branded contract data ARPU of \$3.15, or 23%, for the year 2011 compared to 2010, and \$2.89, or 26%, for the year 2010 compared to 2009 were driven by the continued growth in the number of customers using smartphones with data plans. Smartphone customers accounted for 49% of total branded contract customers at the end of 2011, up from 33% at the end of 2010. Additionally, the continued upgrade of the T-Mobile 3G and 4G networks helped drive Internet access revenue growth through the increased customer adoption of mobile broadband data plans in 2011.

#### Adjusted OIBDA

	Year I	Year Ended December 31,		
	2011	2010	2009	
	(do	(dollars in millions)		
Adjusted OIBDA	\$ 5,310	\$ 5,478	\$ 5,915	
Adjusted OIBDA margin	29%	29%	31%	
Operating (loss) income	\$ (4,279)	\$ 2,705	\$ 3,056	
Operating (loss) income margin	(21)%	13%	14%	

T-Mobile defines OIBDA, a non-GAAP financial measure, as operating income before, depreciation, amortization, impairment charges and other adjustments. In a capital-intensive industry such as wireless telecommunications, T-Mobile believes Adjusted OIBDA to be a meaningful measure of T-Mobile s operating performance. Adjusted OIBDA should not be construed as an alternative to operating income or net income as determined in accordance with GAAP, as an alternative to cash flows from operating activities as determined in accordance with GAAP or as a measure of liquidity. T-Mobile uses Adjusted OIBDA and Adjusted OIBDA

margin as an integral part of its planning and internal financial reporting processes, to evaluate the performance of its business by senior management and to compare its performance with that of many of its competitors. T-Mobile believes that operating income is the financial measure calculated and presented in accordance with GAAP that is the most directly comparable to Adjusted OIBDA. Adjusted OIBDA margin is calculated as Adjusted OIBDA divided by total service revenues expressed as a percentage. The GAAP financial measure that is most directly comparable to Adjusted OIBDA margin is operating margin, which represents operating income divided by total revenues. OIBDA is adjusted to exclude transactions that are not reflective of T-Mobile s ongoing operating performance and is detailed in the section entitled Reconciliation of non-GAAP Financial Measures . OIBDA was adjusted in the fourth quarter of 2011 to exclude costs related to the AT&T transaction that are not reflective of T-Mobile s ongoing operating performance.

Adjusted OIBDA for the year ended December 31, 2011 decreased to \$5.3 billion from \$5.5 billion for the year ended December 31, 2010 and \$5.9 billion for the year ended December 31, 2009. The Adjusted OIBDA decreases in 2011 and 2010 were primarily the result of lower service revenues driven by the decrease in branded contract customers. Additionally, in 2010 compared to 2009, there were higher operating expenses primarily driven by increased customer adoption of smartphones, which have higher cost of equipment sales.

#### Nine Months Ended September 30, 2012 Compared to Nine Months Ended September 30, 2011

The following table sets forth the number of ending customers and net customer additions (losses) for the nine months ended September 30, 2012 and 2011.

	Nine Months Ended September 30, 2012 2011	
	(in thous	sands)
Customers, end of period		
Branded contract customers	20,809	23,074
Branded prepaid customers	5,659	4,599
Total branded customers	26,468	27,673
M2M customers	2,954	2,525
MVNO customers	3,905	3,514
Total wholesale customers	6,859	6,038
Total T-Mobile USA customers, end of period	33,327	33,711
Net customer additions (losses)		
Branded contract customers	(1,559)	(1,499)
Branded prepaid customers	841	101
Total branded customers	(719)	(1,399)
M2M customers	525	652
MVNO customers	336	724
Total wholesale customers	861	1,376
Total T-Mobile USA net customer additions (losses)	142	(23)

Note: Certain customer numbers may not add due to rounding.

### **Total Customers**

Net customer additions were 142,000 for the nine months ended September 30, 2012, compared to 23,000 net customer losses for the nine months ended September 30, 2011. Total customers were approximately 33.3 million as of September 30, 2012, a decrease of 1.1% over the customer total as of September 30, 2011. During the nine months ended September 30, 2012, branded contract customer losses were partially offset by branded prepaid customer growth.

### **Branded** Customers

Branded contract net customer losses were 1,559,000 for the nine months ended September 30, 2012, compared to 1,499,000 net customer losses for the nine months ended September 30, 2011. The increase in branded contract customer losses is primarily attributable to fewer branded contract customer gross additions partially offset by improved branded contract customer churn. The decrease in customer gross additions was primarily a result of customers responding positively to competitor advertising campaigns and promotional offerings such as handset discounts as well as lingering concerns by certain customers about the uncertainty caused by the pending, but ultimately unsuccessful acquisition of T-Mobile by AT&T.

Branded prepaid net customer additions were 841,000 for the nine months ended September 30, 2012, compared to 101,000 net customer additions for the nine months ended September 30, 2011. The significant improvement in branded prepaid net customer additions in the first nine months of 2012 was due to the continued success of Monthly 4G plans and customer migration to prepaid plans due to the discontinuation of the FlexPay No Contract product that historically had higher churn.

### Wholesale

Wholesale net customer additions were 861,000 for the nine months ended September 30, 2012, compared to net customer additions of 1,376,000 for the nine months ended September 30, 2011. The decrease in wholesale net customer additions was primarily due to fewer MVNO gross customer additions and higher M2M customer churn. Slower MVNO partner growth was driven by reduced demand for T-Mobile s products and services by MVNO partners, while M2M customer churn resulted primarily from M2M partners deactivating lines to align with their customer demand. MVNO partners often have relationships with multiple carriers and through promotions can impact specific carriers results.

### Churn

		Nine Months Ended September 30,	
	2012	2011	
Branded churn	3.10%	3.20%	
Branded contract churn	2.30%	2.60%	
Branded prepaid churn	6.20%	6.70%	

Branded churn was 3.10% for the nine months ended September 30, 2012, compared to 3.20% for the nine months ended September 30, 2011. The decrease in branded churn was primarily driven by the strategic phase-out of certain products that had historically higher churn, the introduction of new products with higher customer satisfaction and the continued strategic focus on churn reduction, including credit optimization initiatives.

Branded contract churn was 2.30% for the nine months ended September 30, 2012, down 30 basis points from the nine months ended September 30, 2011. The decrease in branded contract churn was the result of the strategic phase-out of the FlexPay Contract product that historically had higher churn and the continued focus on churn reduction initiatives.

Branded prepaid churn was 6.20% for the nine months ended September 30, 2012, down 50 basis points from the nine months ended September 30, 2011. The improvement in branded prepaid churn was primarily attributed to the continued success of the T-Mobile Monthly 4G plans and the strategic phase-out of the FlexPay No Contract product, which historically had higher churn.

#### Average Revenue Per User

		Nine Months Ended September 30,		
	2012	2011		
ARPU (blended)	\$ 43.73	\$ 45.97		
ARPU (branded contract)	\$ 57.21	\$ 57.35		
ARPU (branded prepaid)	\$ 26.55	\$ 24.05		
Data ARPU (branded contract)	\$ 19.14	\$ 16.74		

Blended ARPU was \$43.73 and \$45.97 for the nine months ended September 30, 2012 and 2011, respectively, a decrease of \$2.24. The decrease in blended ARPU was primarily attributable to the change in customer portfolio mix towards Value plans, branded prepaid and wholesale customers, which traditionally have lower ARPU.

Branded contract ARPU was \$57.21 for the nine months ended September 30, 2012, a slight decrease from \$57.35 for the nine months ended September 30, 2011. The slight decrease in branded contract ARPU was primarily due to the branded contract customer base migration to Value plans introduced in the third quarter of 2011, which have lower ARPUs, partially offset by higher data revenues from the continued adoption of data plans and smartphone penetration.

Branded prepaid ARPU was \$26.55 for the nine months ended September 30, 2012, an increase from \$24.05 for the nine months ended September 30, 2011. The increase in branded prepaid ARPU was primarily due to the growth of the Monthly 4G plans, which include data services.

Branded contract data ARPU for the nine months ended September 30, 2012 was \$19.14 an increase from \$16.74 for the nine months ended September 30, 2011. The increase in branded contract data ARPU was primarily attributable to the increase in smartphone sales with data plans.

#### Adjusted OIBDA

		Nine Months Ended September 30,		
	2012	2011		
	(dollars in )	millions)		
Adjusted OIBDA	\$ 3,838	\$ 3,910		
Adjusted OIBDA margin	29%	28%		
Operating (loss) income	\$ (6,644)	\$ 1,625		
Operating (loss) income margin	(45)%	11%		

Adjusted OIBDA for the nine months ended September 30, 2012 decreased to \$3.8 billion from \$3.9 billion for the nine months ended September 30, 2011. The decrease was primarily due to a decline in service revenues, offset partially by equipment revenue growth and operating expense reductions. Equipment revenue improved in the first nine months of 2012 related to the launch of T-Mobile s Value plans for which T-Mobile does not offer subsidies (discounts) on devices. Additionally, operating expenses decreased in the first nine months of 2012 primarily due to lower cost of equipment sales driven by lower handset unit sales volumes, lower employee-related expenses and the effects of ongoing cost management programs. These combined operating expense reductions were partially offset by higher bad debt expense.

### **Reconciliation of non-GAAP Financial Measures**

A non-GAAP financial measure is defined as a numerical measure of a company s financial performance that (i) excludes amounts, or is subject to adjustments that have the effect of excluding amounts, that are included in the comparable measure calculated and presented in accordance with GAAP in the statement of income or statement of cash flows, or (ii) includes amounts, or is subject to adjustments that have the effect of including amounts, that are excluded from the comparable measure so calculated and presented.

Adjusted OIBDA is a non-GAAP financial measure utilized by T-Mobile s management to monitor the financial performance of its operations. This measurement, together with GAAP measures such as revenue and operating income, assists management in its decision-making process related to the operation of the business. T-Mobile uses Adjusted OIBDA internally as a metric to evaluate and compensate T-Mobile s personnel and management for their performance, and as a benchmark to evaluate its operating performance in comparison to its competitors. Management also uses Adjusted OIBDA to measure, from period-to-period, T-Mobile s ability to provide cash flows to meet future debt services, capital expenditures and working capital requirements and fund future growth. T-Mobile believes that analysts and investors use Adjusted OIBDA as a supplemental measure to evaluate our overall operating performance and that this metric facilitates comparisons with other wireless communications companies. Adjusted OIBDA has limitations as an analytical tool and should not be considered in isolation or as a substitute for income from operations, net income, or any other measure of financial performance reported in accordance with GAAP. In addition, other wireless carriers may calculate this measure differently. OIBDA is adjusted to exclude transactions that are not reflective of our ongoing operating performance and is detailed in the tables below.

The following tables illustrate the calculation of Adjusted OIBDA and reconcile Adjusted OIBDA to operating income (loss) which T-Mobile considers to be the most directly comparable GAAP financial measure to Adjusted OIBDA.

	Year Ended December 31,		er 31,
	2011	2010 (in millions)	2009
Adjusted OIBDA	\$ 5,310	\$ 5,478	\$ 5,915
Depreciation and amortization	(2,982)	(2,773)	(2,859)
Adjusted operating income (excl. impairment and transaction-related costs)	2,328	2,705	3,056
Impairment charges	(6,420)		
Transaction-related costs(1)	(187)		
Operating (loss) income	\$ (4,279)	\$ 2,705	\$ 3,056

	Nine Mont Septeml	
	2012	2011
	(in mil	lions)
Adjusted OIBDA	\$ 3,838	\$ 3,910
Depreciation and amortization	(2,391)	(2,221)
Adjusted operating income (excl. impairment, restructuring and transaction-related costs)	1,447	1,689
Impairment charges	(8,134)	
Restructuring charges	(90)	
Other transactions(1)	133	(64)
Operating (loss) income	\$ (6,644)	\$ 1,625

(1) Represents AT&T transaction-related costs incurred from the now terminated AT&T acquisition of T-Mobile from Q2 2011 to Q3 2012, and a net gain from a spectrum swap transacted in Q3 2012. Other transactions may not agree in total to the other, net classification in the Condensed Consolidated Statements of Operations and Comprehensive Income (Loss) due to certain routine operating activities that are not excluded from Adjusted OIBDA.

#### Liquidity and Capital Resources

T-Mobile s principal sources of liquidity are existing cash and cash equivalents, short-term money market investments with Deutsche Telekom (accounts receivable from affiliates), and cash generated from operations. At September 30, 2012, T-Mobile s liquid assets were \$755 million, consisting of \$430 million cash and cash equivalents and \$325 million short-term investments with Deutsche Telekom. At December 31, 2011, T-Mobile s liquid assets were \$2.2 billion, consisting of \$390 million cash and cash equivalents and \$1.8 billion short-term investments with Deutsche Telekom. T-Mobile evaluates its cash position on a regular basis and invests amounts that exceed short term cash requirements in U.S. dollars with Deutsche Telekom.

T-Mobile determines future liquidity requirements, for both operations and for capital expenditures, based in large part upon projected financial and operating performance. T-Mobile regularly reviews and updates these projections for changes in current and projected financial and operating results, general economic conditions, the competitive landscape and other factors. T-Mobile believes that its existing liquid assets, anticipated cash flows from operations, and ability to access additional financing through Deutsche Telekom will be sufficient to meet its anticipated liquidity requirements in the near term.

There are a number of risks and uncertainties that could cause T-Mobile s financial and operating results and capital requirements to differ materially from its projections and that could cause its liquidity to differ materially from the assessment set forth above. See the section entitled Risk Factors beginning on page 33.

Compared to December 31, 2011, T-Mobile s cash and cash equivalents and short-term investments with Deutsche Telekom decreased by \$1.5 billion to \$755 million as of September 30, 2012. The decrease was primarily due to the transfer of spectrum licenses of \$1.2 billion from Deutsche Telekom in exchange for short-term investments (accounts receivable from affiliates) held with Deutsche Telekom. As part of the break-up consideration of the terminated transaction for the sale of T-Mobile to AT&T, Deutsche Telekom received \$1.2 billion of spectrum licenses from AT&T. The spectrum licenses were transferred to T-Mobile in May 2012 in exchange for short-term investments held with Deutsche Telekom.

Compared to December 31, 2010, T-Mobile s cash and cash equivalents and short-term investments with Deutsche Telekom increased by \$1.8 billion to \$2.2 billion at December 31, 2011. The increase was primarily a result of fewer intercompany notes payables with Deutsche Telecom maturing during 2011 compared to the prior

year. T-Mobile settled maturing notes payables to affiliates balances of \$500 million in 2011 compared to \$2.6 billion in 2010, utilizing short-term investments (accounts receivables from affiliates) held with Deutsche Telekom. The settlement of T-Mobile s notes payable to affiliates with short-term investments held with Deutsche Telekom is a non-cash transaction and does not appear on the consolidated statements of cash flows.

### Capital Expenditures

The construction of T-Mobile s network and the marketing and distribution of its wireless communications products and services have required, and will continue to require, substantial amounts of liquidity (see also Management s Discussion and Analysis of Financial Condition and Results of Operations of T-Mobile Contractual Obligations ). Historically, T-Mobile s liquidity requirements have been driven primarily by capital expenditures for spectrum licenses and the construction, expansion and upgrades of its network infrastructure. T-Mobile s capital expenditures for property and equipment for the nine months ended September 30, 2012 were \$2.0 billion. The expenditures related to spectrum licenses for the nine months ended September 30, 2012 were \$1.5 billion, including the transfer of spectrum licenses of \$1.2 billion described above. T-Mobile s capital expenditures for property and equipment for the years ended December 31, 2011, 2010 and 2009 were \$2.8 billion and \$3.7 billion, respectively. The expenditures related to spectrum licenses for the years ended December 31, 2011, 2010 and 2009 were \$23 million, \$18 million and \$34 million, respectively.

The property and equipment capital expenditures for the nine months ended September 30, 2012 primarily relate to T-Mobile s network modernization plans to deploy 4G LTE in certain markets during 2012 and more broadly in 2013. The capital expenditures for the year ended December 31, 2011 were primarily associated with continued expansion of T-Mobile s network coverage and the upgrade of T-Mobile s UMTS network to HSPA+ 42 technology. The capital expenditures for the years ended December 31, 2010 were primarily associated with the expansion of T-Mobile s network coverage and the upgrade to HSPA+. The capital expenditures for the year ended December 31, 2009 were primarily associated with the roll-out of the 3G (UMTS/HSPA) network as well as the continued focus on the improvement of T-Mobile s network quality and expansion of network coverage.

T-Mobile expects to incur capital expenditures for property and equipment, and spectrum licenses of approximately \$3.3 billion and \$1.8 billion (including the transfer of spectrum licenses of \$1.2 billion described above), respectively, during 2012.

#### **Capital Resources**

T-Mobile s sources of funds, primarily from operations and, to the extent necessary, from financing arrangements with Deutsche Telekom or its affiliates, are sufficient to meet ongoing operating and investing requirements. T-Mobile expects that its capital spending requirements will continue to be financed primarily through internally generated funds. Should additional debt or equity financing be needed to fund additional development activities or to maintain an appropriate capital structure to ensure T-Mobile s financial flexibility, T-Mobile historically has negotiated and expects that it would continue to negotiate the necessary financing with Deutsche Telekom.

T-Mobile s total capital consists of notes payable to affiliates (current and long-term) and stockholder s equity. At September 30, 2012, T-Mobile s notes payable to affiliates totaled \$14.9 billion and stockholder s equity was \$8.5 billion. Stockholder s equity decreased from December 31, 2011 due to the net loss recorded in the nine months ended September 30, 2012, as described above in the Results of Operations. As of December 31, 2011 and 2010, T-Mobile s notes payables to affiliates totaled \$15.6 billion and \$16.3 billion, respectively, and stockholder s equity was \$15.8 billion and \$20.5 billion, respectively. The decrease in stockholder s equity in 2011was primarily due to the net loss recorded in 2011, largely attributable to a non-cash impairment charge against goodwill and spectrum licenses as described above in Management s Discussion and Analysis of Financial Condition and Results of Operations of T-Mobile Results of Operations .

As described elsewhere in this proxy statement, on October 3, 2012, Deutsche Telekom and MetroPCS announced the signing of a business combination agreement (the business combination agreement ) to combine T-Mobile and MetroPCS. Pursuant to the terms and subject to the conditions set forth in the business combination agreement, T-Mobile will refinance its net intercompany indebtedness (net of notes payable to affiliates and accounts receivable from affiliates) by issuing to Deutsche Telekom \$15.0 billion of senior unsecured notes. Additional financing terms contemplated in the business combination agreement for the new combined entity is covered in the Financing section of this proxy statement.

T-Mobile had a \$150 million unsecured credit facility with U.S. Bank National Association available for working capital and other corporate purposes that was not extended and expired on December 31, 2011.

In September 2012, T-Mobile entered into definitive agreements with CCI regarding the sale of rights to operate approximately 7,200 wireless communication towers owned by T-Mobile. Prior to closing on November 30, 2012, the tower portfolio included in the transaction was reduced to 7,105 sites. During the fourth quarter of 2012, T-Mobile received net proceeds of \$2,469 million of which T-Mobile distributed \$2,403 million to its parent, Deutsche Telekom. Under the terms of the transaction, CCI acquired the sole rights to lease and operate the wireless communication towers through either purchase of the towers, or under long-term prepaid lease and management agreements. Furthermore, CCI was granted an option to acquire any sites that were not sold to CCI at the end of their respective least terms. T-Mobile will lease back and continue to have rights to space on the towers to operate its wireless communication equipment. The master lease agreement governing the lease-back of tower space provides T-Mobile with an initial term and several optional renewals at terms that are consistent with those it has for similar sites where T-Mobile leases space on third-party owned towers.

### Cash Flows

The following table shows cash flow information for the three years ended December 31, 2011, 2010 and 2009:

	Year	Year Ended December 31,			
	2011	2011 2010			
		(in millions)			
Net cash provided by operating activities	\$ 4,980	\$ 4,905	\$ 5,437		
Net cash used in investing activities	\$ (4,699)	\$ (5,126)	\$ (5,603)		
Net cash provided by financing activities	\$	\$ 123	\$ 67		

#### **Operating** Activities

Cash provided by operating activities increased by \$75 million to \$5.0 billion for the year ended December 31, 2011 from \$4.9 billion for the year ended December 31, 2010. The increase was primarily attributable to cash inflows from changes in working capital accounts, mainly accounts receivable. The improvement in cash flows from the above was largely offset by the decrease in operating income, exclusive of non cash items such as depreciation and amortization, impairment and other non-cash charges in 2011 compared to the prior year.

Cash provided by operating activities decreased by \$532 million to \$4.9 billion during the year ended December 31, 2010 from \$5.4 billion for the year ended December 31, 2009. The decrease was primarily attributable to lower cash flows from changes in working capital accounts, mainly accounts receivable related to the decline in T-Mobile s branded customer base in 2010, as well as changes in inventory balances resulting from improvements in inventory procurement and management processes in 2009.

### **Investing** Activities

Cash used in investing activities was \$4.7 billion during the year ended December 31, 2011 compared to \$5.1 billion during the year ended December 31, 2010. The decrease was primarily due to \$310 million fewer funds invested with Deutsche Telekom as short-term investments on a net basis (short term affiliate loan receivable, net) during 2011 compared to 2010.

Cash used in investing activities was \$5.1 billion during the year ended December 31, 2010 compared to \$5.6 billion during the year ended December 31, 2009. The decrease was primarily due to \$868 million lower capital expenditures for property and equipment during 2010. During 2009, T-Mobile launched its 3G UMTS/HSPA network resulting in higher capital spending during 2009. In 2010, the focus of capital expenditures for property and equipment was on upgrading the HSPA network to HSPA+ which is less capital intensive. The decrease in network capital expenditures was partially offset by a \$427 million increase in additional funds invested with Deutsche Telecom as short-term investments (short term affiliate loan receivable, net) on a net basis during 2010 compared to 2009.

### **Financing** Activities

There was no cash provided by financing activities during the year ended December 31, 2011 compared to cash provided by financing activities of \$123 million during the year ended December 31, 2010. The financing activities in 2010 related primarily to \$116 million in additional long-term debt borrowings from Deutsche Telekom. During the years ended December 31, 2011 and 2010, T-Mobile settled notes payables to affiliates with Deutsche Telekom through a non-cash transaction by offsetting borrowings from Deutsche Telekom with short-term investments held by Deutsche Telekom of \$0.5 billion and \$2.6 billion, respectively.

Cash provided by financing activities was \$123 million during the year ended December 31, 2010 compared to cash provided by financing activities of \$67 million during the year ended December 31, 2009. The change in cash provided by financing activities was a result of no debt repayments to Deutsche Telekom in 2010, compared to debt repayments of \$133 million in 2009. New long-term borrowings from Deutsche Telekom were \$116 million in 2010 compared to \$200 million in 2009.

#### Cash Flows

The following table shows cash flow information for the nine months ended September 30, 2012 and 2011:

		Nine Months Ended September 30,		
	2012	2011		
	(in mi	llions)		
Net cash provided by operating activities	\$ 2,707	\$ 3,570		
Net cash used in investing activities	\$ (2,667)	\$ (3,258)		
Net cash provided by financing activities	\$	\$		

#### **Operating** Activities

Cash provided by operating activities decreased by \$863 million to \$2.7 billion during the nine months ended September 30, 2012 from \$3.6 billion for the nine months ended September 30, 2011. The decrease was primarily attributable to changes in working capital accounts, including payments for AT&T transaction related costs and restructuring costs during the nine months ended September 30, 2012.

#### **Investing** Activities

Cash used in investing activities was \$2.7 billion during the nine months ended September 30, 2012 compared to \$3.3 billion during the nine months ended September 30, 2011. The decrease was primarily due to

\$803 million fewer funds invested with Deutsche Telekom on a net basis (short term affiliate loan receivable, net) during the nine months ended September 30, 2012 compared to the same period in the prior year. The decrease in funds invested with Deutsche Telekom in 2012 was the result of utilizing accounts receivable from affiliates in exchange for spectrum licenses from Deutsche Telekom. The spectrum licenses were received from AT&T as part of the break-up consideration of the terminated AT&T transaction for the sale of T-Mobile.

### **Financing** Activities

Cash provided by financing activities was consistent during the nine months ended September 30, 2012 compared to the nine months ended September 30, 2011, with no cash provided by financing activities in either period. During the nine month period ended September 30, 2012 and 2011, T-Mobile settled inter-company debt with Deutsche Telekom through non-cash transactions utilizing short-term investments held by Deutsche Telekom of \$644 million and \$500 million, respectively.

#### **Off-Balance Sheet Arrangements**

T-Mobile does not participate in, or secure, financing for any unconsolidated entities.

#### **Contractual Obligations**

Current accounting standards require disclosure of material obligations and commitments to making future payments under contracts, such as debt, lease agreements, and purchase obligations. T-Mobile discloses its payables to affiliates in Note 11 and commitments and contingencies in Note 18 to the T-Mobile consolidated audited financial statements for the three years ended December 31, 2011.

The following table provides aggregate information about T-Mobile s contractual obligations as of December 31, 2011. Other commitments and obligations are included in the table based on the year of required payment or an estimate of the year of payment.

	Payments Due by Period as of December 31, 2011				
	Less Than 1			More Than 5	
	Year	1 - 3 Years	3 - 5 Years (in millions)	Years	Total
Payables to affiliates (1)	\$ 644	\$ 3,621	\$ 2,905	\$ 7,914	\$ 15,084
Interest on payables to affiliates	616	1,523	1,052	5,298	8,489
Non-dedicated transportation lines	489	997	890	457	2,833
Operating leases, including dedicated					
transportation lines	1,791	3,167	2,692	5,681	13,331
Purchase obligations (2)	398	21	2		421
-					
Total contractual obligations	\$ 3,938	\$ 9,329	\$ 7,541	\$ 19,350	\$ 40,158

- (1) Represents principal amounts of payables to affiliates at maturity.
- (2) T-Mobile calculated the minimum obligation for certain agreements to purchase goods or services based on termination fees that can be paid to exit the contract. Termination penalties are included in the above table as payments due in less than one year, as this is the earliest T-Mobile could exit these contracts. If T-Mobile elects to exit these agreements, termination fees for all such contracts could be approximately \$231 million.

The following table provides aggregate information about T-Mobile s contractual obligations as of September 30, 2012. Other commitments and obligations are included in the table based on the year of required payment or an estimate of the year of payment.

	Payments Due by Period as of September 30, 2012					
	Less Than 1			More Than 5		
	Year	1 - 3 Years	3 - 5 Years (in millions)	Years	Total	
Payables to affiliates (1)	\$ 1,273	\$ 4,253	\$ 1,000	\$ 7,910	\$ 14,436	
Interest on payables to affiliates	624	1,321	1,035	4,842	7,822	
Non-dedicated transportation lines	518	1,029	828	255	2,630	
Operating leases, including dedicated transportation						
lines	1,770	3,323	3,065	6,514	14,672	
Purchase obligations (2)	2,012	491	2,825		5,328	
Total contractual obligations	\$ 6,197	\$ 10,417	\$ 8,753	\$ 19,521	\$ 44,888	

(1) Represents principal amounts of payables to affiliates at maturity.

(2) T -Mobile calculated the minimum obligation for certain agreements to purchase goods or services based on termination fees that can be paid to exit the contract. Termination penalties are included in the above table as payments due in less than one year, as this is the earliest T-Mobile could exit these contracts. If T -Mobile elects to exit these agreements, termination fees for all such contracts could be approximately \$750 million.

The tables do not include the payments to be made on T-Mobile s interest rate swaps and cross currency interest rate swaps. Certain of T-Mobile s other noncurrent liabilities have been excluded from the tables due to the uncertainty of the timing of payments, combined with the absence of historical trending to be used as a predictor of such payments.

The purchase obligations reflected in the table above are primarily commitments to purchase handsets and accessories, equipment, software, programming and network services, and marketing activities, which will be used or sold in the ordinary course of business. These amounts do not represent T-Mobile s entire anticipated purchases in the future, but represent only those items for which T-Mobile is contractually committed. Where T-Mobile is committed to make a minimum payment to the supplier regardless of whether it takes delivery, T-Mobile has included only that minimum payment as a purchase obligation.

#### **Related Party Transactions**

See Note 16 to the T-Mobile consolidated financial statements for the three years ended December 31, 2011 and Note 12 to the T-Mobile condensed consolidated financial statements for the nine months ended September 30, 2012.

#### **Restructuring Costs**

T-Mobile consolidated its call center operations from 24 to 17 facilities during the period ended September 30, 2012. In addition, T-Mobile restructured and optimized operations in other parts of the business to strengthen T-Mobile s competiveness in the U.S. marketplace. For further discussion, see Note 13 to the T-Mobile condensed consolidated financial statements for the nine months ended September 30, 2012.

#### Effect of New Accounting Standards

In May 2011, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) 2011-04, Amendments to Achieve Common Fair Value Measurement and Disclosure Requirements in GAAP and IFRS. The new guidance limits the highest-and-best-use measure to nonfinancial assets, permits certain financial assets and liabilities with offsetting positions in market or counterparty credit risks to be measured at a net basis, and provides guidance on the applicability of premiums and discounts and expands the disclosures on Level 3 inputs by requiring quantitative disclosure of the unobservable inputs and assumptions, as well as description of the valuation processes and the sensitivity of the fair value to changes in unobservable inputs. The new guidance is effective for interim and annual periods beginning after December 15, 2011. The Company adopted this accounting pronouncement on January 1, 2012 with no material impact on results of operations, financial conditions or its financial disclosures.

In September 2011, FASB issued ASU 2011-08, Testing Goodwill for Impairment. It provides an option to first assess qualitative factors to determine whether it is necessary to perform the two-step quantitative goodwill impairment test. Given this option, an entity no longer would be required to calculate the fair value of a reporting unit unless the entity determines, based on that qualitative assessment, that it is more likely than not that its fair value is less than its carrying amount. The new guidance is effective for interim and annual periods beginning after December 15, 2011. T-Mobile adopted this accounting pronouncement on January 1, 2012 with no material impact on results of operations, financial conditions or its financial disclosures.

In July 2012, FASB issued ASU 2012-02, Intangibles Goodwill and Other (Topic 350): Testing Indefinite-Lived Intangible Assets for Impairment . This update simplifies the guidance for testing the decline in the realizable value (impairment) of indefinite-lived intangible assets other than goodwill. The amendments in this Update are effective for annual and interim impairment tests performed for fiscal years beginning after September 15, 2012. T-Mobile has adopted these amendments for early impairment tests performed during the year ending December 31, 2012 with no material impact on results of operations, financial conditions or its financial disclosures.

#### Quantitative and Qualitative Disclosures about Financial, Credit, Market and Liquidity Risk

T-Mobile is exposed to various types of market risks in the normal course of business, primarily from changes in interest rates and foreign currency exchange rates. These risks, along with other business risks, impact T-Mobile s cost of capital. T-Mobile s policy is to manage exposure related to fluctuations in interest rates and foreign exchange in order to manage capital costs, control financial risks and maintain financial flexibility over the long term. In managing market risks, T-Mobile employs derivatives according to documented policies and procedures, including interest rate swap agreements for fluctuation in interest rates and cross currency interest rate swap agreements to manage foreign currency risk along with interest rate risk. The cross currency interest rate swap agreements manage foreign currency risk and interest rate risk related to T-Mobile s notes payables to affiliates denominated in Euros. T-Mobile does not use derivatives for trading or speculative purposes. T-Mobile does not foresee significant changes in the strategies used to manage market risk in the near future.

#### Interest Rate Risk

T-Mobile is exposed to changes in interest rates, primarily on its notes payable to affiliates. As of December 31, 2011, T-Mobile had \$15.6 billion in notes payable with Deutsche Telekom. Changes in interest rates can lead to significant fluctuations in the fair value of T-Mobile s debt instruments. To manage interest expense, T-Mobile utilizes interest rate swaps. T-Mobile has established interest rate risk limits that are closely monitored by measuring interest rate sensitivities of its debt and interest rate derivatives portfolios.

To perform the sensitivity analysis on its notes payable to affiliates balances, T-Mobile assessed the risk of a change in the fair value from the effect of a hypothetical interest rate change of 100 basis points on T-Mobile s variable and fixed rate debt. As of December 31, 2011, the change in the fair value of T-Mobile s notes payable to affiliates is shown in the table below:

December 31, 2011	Fair Value	Fair Value assuming +100 basis point shift (in millions)		lue assuming basis point shift
Current and long-term notes payables to				
affiliates	\$ 16,870	\$	15,879	\$ 17,980

To perform the sensitivity analysis on its interest rate swaps not designated as hedges, T-Mobile assessed the risk of loss in fair values from the effect of a hypothetical interest rate change of 100 basis points on T-Mobile s portfolio of interest rate swaps. As of December 31, 2011, the change in the fair value of T-Mobile s interest rate swaps is shown in the table below:

	December 31, 2011	Fair Value assuming Fair Value +100 basis point shift (in millions)		-100 ba	e assuming asis point hift	
	Interest rate swaps	\$ 96	\$	52	\$	141
-	· · · ·					

# Foreign Exchange Risk

While the vast majority of T-Mobile s financial transactions are conducted in U.S. dollars, T-Mobile is exposed to foreign currency exchange risk through its Euro denominated notes payable to affiliates. T-Mobile hedges exchange risk involved with its Euro denominated notes payables to affiliates through cross currency interest rate swaps. As of December 31, 2011, T-Mobile s notes payable to affiliates denominated in Euros have been economically hedged at issuance through cross currency interest rate swaps. Risks related to both interest rate risk and foreign currency exchange fluctuations associated with the underlying interest and principal payments were eliminated through swaps from fixed-rate foreign currencies to fixed-rate U.S. dollars at terms exactly mirroring those of the underlying Euro borrowing arrangements. At the time of maturity or payment, T-Mobile expect the gains or losses from its cross currency interest rate swaps to offset the losses and gains on its Euro denominated notes payable to affiliates.

#### T-MOBILE AND METROPCS UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

The following unaudited pro forma condensed combined financial information present the unaudited pro forma condensed combined balance sheet and unaudited pro forma condensed combined statements of operations based upon the combined historical financial statements of T-Mobile and MetroPCS, after giving effect to the transaction between T-Mobile and MetroPCS and adjustments described in the accompanying notes. The transaction will be accounted for as a reverse acquisition under the acquisition method of accounting, which requires determination of the accounting acquirer. The accounting guidance for business combinations, Accounting Standards Codification 805, provides that in identifying the acquiring entity in a combination effected through an exchange of equity interests, all pertinent facts and circumstances must be considered, including: the relative voting rights of the stockholders of the constituent companies in the combined company, the existence of a large minority voting interest in the combined entity if no other owner or organized group of owners has a significant voting interest, the composition of the board of directors and senior management of the combined company, the relative size of each company and the terms of the exchange of equity securities in the business combination, including payment of any premium.

Because T-Mobile s indirect stockholder, Deutsche Telekom, will be entitled to designate the majority of the board of directors of the combined company, MetroPCS stockholders will receive the cash payment and Deutsche Telekom will receive a majority of the equity securities and voting rights of the combined company, T-Mobile is considered to be the acquirer of MetroPCS for accounting purposes. This means that T-Mobile will allocate the purchase price to the fair value of MetroPCS assets and liabilities at the acquisition date, with any excess purchase price being recorded as goodwill.

The unaudited pro forma condensed combined balance sheet as of September 30, 2012 reflects the transaction as if it occurred on September 30, 2012. The unaudited pro forma condensed combined statements of operations for the year ended December 31, 2011 and the nine months ended September 30, 2012 reflect the transaction as if it occurred on January 1, 2011, the beginning of the earliest period presented.

The unaudited pro forma condensed combined financial information should be read in conjunction with the audited and unaudited historical financial statements of each of T-Mobile and MetroPCS and the notes thereto, as well as the disclosures contained in each company s Management s Discussion and Analysis of Financial Condition and Results of Operations. Additional information about the basis of presentation of this information is provided in Note 1 hereto.

The unaudited pro forma condensed combined financial information was prepared in accordance with Article 11 of Regulation S-X. The unaudited pro forma adjustments reflecting the transaction have been prepared in accordance with business combination accounting guidance as provided in Accounting Standards Codification 805, and reflect the allocation of the preliminary purchase price to the acquired assets and liabilities based upon their estimated fair values, using the assumptions set forth in the notes to the unaudited pro forma condensed combined financial information.

The unaudited pro forma condensed combined financial information is provided for informational purposes only and is not necessarily indicative of the operating results or financial position that would have occurred if the transaction had been completed as of the dates set forth above, nor is it indicative of the future results or financial position of the combined company. The unaudited pro forma condensed combined financial information also does not give effect to the potential impact of current financial conditions, any anticipated synergies, operating efficiencies or cost savings that may result from the transaction or any integration costs. Furthermore, the unaudited pro forma condensed combined statements of operations do not include certain nonrecurring charges and the related tax effects which result directly from the transaction as described in the notes to the unaudited pro forma condensed combined financial information.

# Unaudited Pro Forma Condensed Combined Balance Sheet

### As of September 30, 2012

### (in millions)

		orical	Pro Forma		Pro Forma
	T-Mobile	MetroPCS	Adjustments	Notes	Combined
Assets					
Current assets					
Cash and cash equivalents	\$ 430	\$ 2,057	\$ (1,500)	(3a)	\$ 2,053
			1,000	(5)	
	4.012	1 1 2 0	66	(6)	5 (04
Other current assets	4,913	1,128	(437)	(5)	5,604
Total current assets	5,343	3,185	(871)		7,657
Property and equipment, net	12,535	4,197	(2,457)	(3b)	14,275
Goodwill			791	(3b)	791
Spectrum licenses	14,360	2,562	1,238	(3b)	18,160
Other intangible assets, net	43		650	(3b)	693
Other assets	332	128			460
Total assets	\$ 32,613	\$ 10,072	\$ (649)		\$ 42,036
Liabilities and Stockholders Equity					
Current liabilities					
Accounts payable and accrued liabilities	\$ 2,848	\$ 450	\$		\$ 3,298
Other current liabilities	2,014	335	(1,479)	(5)	¢ 3,290 870
	_,		(-,,)	(-)	
Total current liabilities	4,862	785	(1,479)		4,168
Long-term payables to affiliates	13,620		1,380	(5)	15,000
Long-term debt, net		4,731	1,000	(5)	8,200
			2,469	(6)	
Other long-term liabilities	5,655	1,234	90	(3b)	6,678
			(305)	(3c)	
			4	(5)	
Total long-term liabilities	19,275	5,965	4,638		29,878
Total liabilities	24,137	6,750	3,159		34,046
Stockholders equity					
Common stock and paid-in capital	31,600	1,815	(1,500)	(3a)	31,260
Common stock and paid-in capital	51,000	1,015	1,944	(3a) (3b)	51,200
			(196)	(5)	
			(190)	(6)	
Accumulated other comprehensive loss	(9)	(5)	(2,+03)	(0) (3b)	1
		(3)	10	(5)	1
Treasury stock		(10)	10	(3b)	
Accumulated (deficit) earnings	(23,115)	1,522	(1,522)	(3b)	(23,271)
( <u>-</u>	(20,110)	1,0 ==	(1,522)	(5)	(,_/1)

Total stockholders equity	8,476	3,322	(3,808)	7,990
Total liabilities and stockholders equity	\$ 32,613	\$ 10,072	\$ (649)	\$ 42,036

Edgar Filing: MILLER PETROLEUM INC - Form 10-K

The accompanying notes are an integral part of, and should be read together with, this unaudited

pro forma condensed combined financial information.

Unaudited Pro Forma Condensed Combined Statement of Operations

For the Twelve Months Ended December 31, 2011

(in millions, except share and per share amounts)

		Historical			Pro Forma		Pro Forma		
	T	Mobile	Me	troPCS		stments	Notes		mbined
Revenues					0				
Total revenues	\$	20,618	\$	4,847	\$	135	(6)	\$	25,600
Operating expenses									
Network costs, excluding depreciation									
and amortization		4,952		1,474					6,426
Cost of equipment sales		3,646		1,440					5,086
Selling, general and administrative		6,728		644			(7)		7,372
Depreciation and amortization		2,982		539		(230)	(4a)		3,415
						124	(4b)		
Impairment charges		6,420							6,420
Other, net		169		3					172
Total operating expenses		24,897		4,100		(106)			28,891
Operating (loss) income		(4,279)		747		241			(3,291)
Other (expense) income		(1,27)		, , ,		211			(3,271)
Other expense, net		(655)		(268)		(539)	(5)		(1,582)
		(000)		(200)		(120)	(6)		(1,502)
Total other expense, net		(655)		(268)		(659)			(1,582)
(Loss) income before income taxes		(4,934)		479		(418)			(4,873)
Income tax benefit (expense)		216		(178)		156	(4c)		194
Net (loss) income	\$	(4,718)	\$	301	\$	(262)		\$	(4,679)
Not (loss) in some nor sommer share									
Net (loss) income per common share Basic	¢	(1(10))	¢	0.92			(4.1)	¢	(6.50)
Basic	\$	(16.12)	\$	0.83			(4d)	\$	(6.50)
Diluted	\$	(16.12)	\$	0.82			(4d)	\$	(6.50)
Weighted average shares									
Basic	29	2,669,971	360	),410,168			(4d)	71	9,539,741
Diluted	29	2,669,971	363	3,837,940			(4d)	71	9,539,741

The accompanying notes are an integral part of, and should be read together with, this unaudited

pro forma condensed combined financial information.

Unaudited Pro Forma Condensed Combined Statement of Operations

For the Nine Months Ended September 30, 2012

(in millions, except share and per share amounts)

		Historical			Pro Forma		Pro Forma		
	T	-Mobile	Me	troPCS		istments	Notes		mbined
Revenues					0				
Total revenues	\$	14,810	\$	3,817	\$	101	(6)	\$	18,728
Operating expenses									
Network costs, excluding depreciation									
and amortization		3,515		1,130					4,645
Cost of equipment sales		2,456		1,003					3,459
Selling, general and administrative		5,004		507			(7)		5,511
Depreciation and amortization		2,391		469		(220)	(4a)		2,734
						94	(4b)		
Impairment charges		8,134							8,134
Other, net		(46)		5					(41)
Total operating expenses		21,454		3,114		(126)			24,442
Operating (loss) income		(6,644)		703		227			(5,714)
Other (expense) income		(-) )							(-). )
Other expense, net		(412)		(152)		(417)	(5)		(1,071)
•				, ,		(90)	(6)		
Total other expense, net		(412)		(152)		(507)			(1,071)
(Loss) income before income taxes		(7,056)		551		(280)			(6,785)
Income tax (expense) benefit		(272)		(188)		104	(4c)		(356)
Net (loss) income	\$	(7,328)	\$	363	\$	(176)		\$	(7,141)
Net (loss) income per common share									
Basic	\$	(25.04)	\$	0.99			(4d)	\$	(9.91)
Diluted	\$	(25.04)	\$	0.99			(4d)	\$	(9.91)
Weighted average shares									
Basic	29	2,669,971	363	3,190,434			(4d)	72	0,929,874
Diluted	29	2,669,971	364	,440,115			(4d)	72	0,929,874

The accompanying notes are an integral part of, and should be read together with this unaudited

pro forma condensed combined financial information.

### 1. Basis of Presentation

The historical financial information has been adjusted to give pro forma effect to events that are (i) directly attributable to the transaction, (ii) factually supportable, and (iii) with respect to the unaudited pro forma condensed combined statements of operations, expected to have a continuing impact on the combined results. The pro forma adjustments are preliminary and based on estimates of the fair value and useful lives of the assets acquired and liabilities assumed and have been prepared to illustrate the estimated effect of the transaction and certain other adjustments. The final determination of the purchase price allocation will be based on the fair values of assets acquired and liabilities assumed as of the date the transaction closes, and could result in a significant change to the unaudited pro forma condensed combined financial information, including goodwill.

T-Mobile s historical results are derived from T-Mobile s audited consolidated statement of operations and comprehensive (loss) income for the year ended December 31, 2011, unaudited condensed consolidated balance sheet as of September 30, 2012 and unaudited condensed consolidated statement of operations and comprehensive income (loss) for the nine months ended September 30, 2012 under GAAP. MetroPCS historical results include the audited consolidated statement of income and comprehensive income for the year ended December 31, 2011, unaudited condensed consolidated statement of statement of income and comprehensive income for the year ended December 31, 2011, unaudited condensed consolidated balance sheet as of September 30, 2012 and unaudited condensed consolidated statement of income and comprehensive income for the nine months ended September 30, 2012 under GAAP.

## **Description of Transaction**

On October 3, 2012, MetroPCS entered into the business combination agreement, by and among Deutsche Telekom, Global, Holding, T-Mobile and MetroPCS. Pursuant to the terms and subject to the conditions set forth in the business combination agreement, including receipt of the required MetroPCS stockholder approval of certain of the proposals described in this proxy statement, MetroPCS will: (i) effect a recapitalization that includes the reverse stock split of the MetroPCS common stock, pursuant to which each share of MetroPCS common stock outstanding as of the effective time of the reverse stock split will represent thereafter one-half of a share of MetroPCS common stock; (ii) as part of the recapitalization, make the cash payment in an amount equal to \$1.5 billion (or approximately \$4.08 per share pre-reverse stock split), without interest, in the aggregate to the record holders of MetroPCS common stock immediately following the effective time of the reverse stock split; and (iii) immediately following the cash payment, issue and deliver to Holding or its designee shares of MetroPCS common stock equal to 74% of the fully-diluted shares of MetroPCS common stock outstanding immediately following the cash payment, and Holding will deliver to MetroPCS all of the T-Mobile shares.

In addition, unless otherwise agreed to by the parties, on the business day immediately following the closing of the transaction, MetroPCS, Inc., a direct wholly-owned subsidiary of MetroPCS, will merge with and into its direct wholly-owned subsidiary Wireless, with Wireless continuing as the surviving entity and, immediately thereafter, Wireless will merge with and into T-Mobile, with T-Mobile continuing as the surviving entity. In this proxy statement, we refer to the reverse stock split, the cash payment, the MetroPCS stock issuance, the mergers and the other transactions contemplated by the business combination agreement, collectively, as the transaction.

The transaction is expected to be financed by the issuance by Wireless or T-Mobile of senior unsecured notes, which are assumed to be issued at market. Certain expected embedded features of these notes may result in future fair value adjustments and associated impacts to our statements of operations. However, as future changes in fair value are not determinable, an adjustment would not be factually supportable, and therefore no mark-to-market adjustment is made in the unaudited pro forma condensed combined statement of operations. The senior unsecured notes in an aggregate principal amount of up to \$18.5 billion are to be issued or offered as follows:

The \$15.0 billion notes will be issued by T-Mobile and purchased by Deutsche Telekom to refinance certain intercompany indebtedness owed by T-Mobile (aggregate principal amount of \$14.4 billion) and its subsidiaries to Deutsche Telekom and its subsidiaries (excluding T-Mobile and its subsidiaries).

The \$2.5 billion notes, which may be offered by Wireless to third-party investors and, to the extent not sold to third-party investors prior to the completion of the transaction, will be purchased by Deutsche Telekom upon the closing of the transaction, the proceeds of which MetroPCS intends to use to refinance the Wireless existing senior credit facility. The expected impact related to such refinancing has not been reflected in this unaudited pro forma condensed combined financial information as the terms of the \$2.5 billion notes are not known and therefore an adjustment would not be factually supportable.

The \$1.0 billion notes, which may be offered by Wireless to third-party investors and, to the extent not sold to third-party investors prior to the completion of the transaction, will be purchased by Deutsche Telekom upon the closing of the transaction, the proceeds of which MetroPCS intends to use for general corporate purposes.

In addition to the notes issued to finance the transaction, Deutsche Telekom (or one of its subsidiaries if the obligations of such subsidiary thereunder are unconditionally guaranteed by Deutsche Telekom) will make available for the benefit of T-Mobile and its subsidiaries, on the closing date of the transaction, the working capital revolving credit facility, which is a revolving unsecured credit facility with a maximum principal amount of no less than \$500.0 million to be used for working capital and other general corporate purposes, on terms substantially as set forth on Exhibit H to the business combination agreement.

Additionally, an adjustment to the consideration for the transaction may be made under the terms of the business combination agreement; however, because the conditions for the adjustment as defined in the business combination agreement are not factually supportable as of the date of the unaudited pro forma condensed combined financial information, no adjustment to the consideration for the transaction is reflected in the unaudited pro forma condensed combined financial information.

MetroPCS and T-Mobile expect to incur approximately \$55.0 million of transaction costs, including commitment fees. This estimate excludes financing fees related to the \$15.0 billion notes which are expected to be incurred and not estimable as of the date of this proxy. No adjustments have been reflected in the unaudited pro forma condensed combined financial information as such adjustments are considered not material.

Certain nonrecurring charges have been excluded from the unaudited pro forma condensed combined statements of operations in accordance with Article 11. The excluded nonrecurring charges are the acceleration of MetroPCS share based payments and the related severance and change in control payments. The estimated charge relating to the acceleration of MetroPCS share based payments is approximately \$72.0 million to be incurred by MetroPCS prior to the combination of the two companies. Integration efforts have not yet occurred, therefore the actual amount of severance and change in control payments that will ultimately be paid cannot be predicted with any degree of certainty.

For more details regarding the transaction, including the financing, see the sections entitled The Transaction and Summary of the Financing beginning on pages 64 and 112, respectively.

### 2. Calculation of Estimated Purchase Consideration

The purchase consideration in a reverse acquisition is determined with reference to the value of equity that the accounting acquirer (in this case T-Mobile, the legal subsidiary) would have had to issue to the owners of the accounting acquiree (MetroPCS, the legal parent) to give them the same percentage interest in the combined entity. A preliminary estimate of the purchase price, assuming the transaction closed on November 12, 2012, is as follows (in millions, except shares and stock price):

Number of MetroPCS shares outstanding (i)	367,	,651,349
Adjusted MetroPCS common stock price (ii)	\$	5.97
Impact of reverse stock split:		
Estimated number of shares	183,	,825,675
Estimated adjusted common stock price	\$	11.94
Fair value of MetroPCS shares Fair value of stock options (iii) Cash consideration paid to stock option holders (iii)	\$	2,195 62 2
Estimated purchase price	\$	2,259

- (i) Number of MetroPCS common stock issued and outstanding as of November 12, 2012, including 3,503,050 shares of MetroPCS unvested restricted stock, which will immediately vest upon closing.
- (ii) Closing price of MetroPCS common stock on the New York Stock Exchange on November 12, 2012 of \$10.05 per share is adjusted by the estimated \$4.08 per share impact of the \$1.5 billion cash payment (see Note 3a below). The cash payment is a return of capital, made as part of the recapitalization to the MetroPCS stockholders prior to the stock issuance to Holding or its designee. MetroPCS will make the cash payment in an amount equal to \$1.5 billion (or approximately \$4.08 per share pre-reverse stock split), without interest, in the aggregate to the record holders of MetroPCS common stock immediately following the effective time of the reverse stock split.
- (iii) Pursuant to the business combination agreement, upon closing, the unvested equity awards will immediately vest and stock option holders will receive stock options of the combined entity upon closing, subject to stockholder s right for options where exercise price is less than the average closing price for the five days preceding the closing to elect to cash out such options. Stock options with low exercise prices, as defined in the business combination agreement, are assumed to be cancelled in exchange for cash consideration. Therefore, the fair value of stock options contemplates all stock options outstanding, adjusted for those options meeting the definition of low exercise price subject to cash payment.

For pro forma purposes, the fair value of the T-Mobile common stock used in determining the purchase price was \$10.05 per share based on the closing price of MetroPCS common stock on November 12, 2012. The final purchase consideration could significantly differ from the amounts presented in the unaudited pro forma condensed combined financial information due to movements in MetroPCS common stock price as of the closing date of the transaction. A sensitivity analysis related to the fluctuation in the MetroPCS common stock price was performed to assess the impact a hypothetical change of 10% on the closing price of MetroPCS common stock on November 12, 2012 would have on the estimated purchase price and goodwill as of the closing date.

The following table shows the change in stock price, estimated purchase price and goodwill (dollars in millions, except stock price):

		Estimated	
Change in stock price	Stock price	Purchase Price	Goodwill
Increase of 10%	\$ 11.06	\$ 2,628	\$ 1,160
Decrease of 10%	9.05	1,889	421

### **Preliminary Purchase Price Allocation**

Under the acquisition method of accounting, the identifiable assets acquired and liabilities assumed of MetroPCS will be recorded at the acquisition date fair values and added to those of T-Mobile. The pro forma adjustments are preliminary and based on estimates of the fair value and useful lives of the assets acquired and liabilities assumed as of September 30, 2012 and have been prepared to illustrate the estimated effect of the transaction. The allocation is dependent upon certain valuation and other studies that will not be completed until after the transaction has closed. Accordingly, the pro forma purchase price allocation is subject to further adjustments as additional information becomes available and as additional analyses and final valuations are conducted following the completion of the transaction. There can be no assurances that these additional analyses and final valuations will not result in significant changes to the estimates of fair value set forth below.

The following table sets forth a preliminary allocation of the estimated purchase consideration to the identifiable tangible and intangible assets acquired and liabilities assumed of MetroPCS, with the excess recorded as goodwill (dollars in millions):

Current and other assets (1)	\$ 736
Property and equipment	1,740
Goodwill	791
Spectrum licenses	3,800
Other intangible assets	650
Total assets	\$ 7,717
Current liabilities less current portion of debt	\$ (749)
Deferred tax liabilities, net (2)	(696)
Other long-term liabilities	(315)
Debt (3)	(2,198)
Total liabilities	\$ (3,958)
Net assets prior to the cash payment	\$ 3,759
Less: cash payment to MetroPCS stockholders	(1,500)
Net assets acquired	\$ 2,259
Tet ussets acquired	<i>\(\ \L,\(\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \</i>

- (1) Current and other assets are presented excluding cash and cash equivalents, short-term investments, and deferred tax assets totaling \$2,573 million.
- (2) MetroPCS unadjusted net deferred tax liabilities are \$1,001 million as of September 30, 2012.

(3) Debt includes long-term debt, including current portion, net of cash and cash equivalents and short-term investments. Acquisition date fair values for net property and equipment were calculated utilizing a cost approach that estimates the fair value of property and equipment needed to replace the functionality provided by the existing property and equipment. The estimated acquisition date fair values of property and equipment reflect a significant decrease in the carrying value of MetroPCS property and equipment due to advances in telecommunications equipment technology allowing a market participant to utilize a smaller quantity of property and equipment in a wireless network to achieve the same functionality. Additionally, MetroPCS cell sites are concentrated in dense urban areas, where a market participant would have significant overlapping coverage, thus the acquisition date fair value is impacted by the accelerated decommissioning of a large number of cell sites.

The final determination of the purchase price allocation upon the closing of the transaction will be based on MetroPCS net assets acquired as of that date and will depend on a number of factors, which cannot be predicted with any certainty at this time. The purchase price allocation may change materially based on the receipt of more detailed information. Therefore, the actual allocations will differ from the pro forma adjustments

# Table of Contents

presented.

#### 3. Notes to Unaudited Pro Forma Condensed Combined Balance Sheet

- (a) Represents the \$1.5 billion aggregate cash payment made as a return of capital by MetroPCS to its stockholders as part of the recapitalization prior to the MetroPCS stock issuance to Holding or its designee (see Note 2 above).
- (b) Reflects the acquisition method of accounting based on the estimated fair value of the assets and liabilities of MetroPCS as discussed in Note 2 above. Includes the elimination of MetroPCS historical stockholders equity accounts because MetroPCS is not considered to be the accounting acquirer.
- (c) Represents adjustments to reflect the reduction of net deferred tax liabilities as a result of recording the acquired assets and assumed liabilities of MetroPCS at their fair value as well as recognition of MetroPCS net operating losses related to equity compensation which were previously unrecognized in the financial statements. The valuation allowances, which are included in net deferred tax liabilities, are unchanged from MetroPCS and T-Mobile s respective historical valuation allowances as neither MetroPCS nor T-Mobile anticipates a material change to its valuation allowances due to the transaction.

#### 4. Notes to Unaudited Pro Forma Condensed Combined Statements of Operations

(a) Represents adjustments to record depreciation and amortization expense related to the reduced basis of property and equipment by \$2.5 billion (see Note 3b), which have been recorded at estimated fair value on a pro forma basis and will be depreciated and amortized over the estimated remaining useful lives on a straight-line basis utilizing T-Mobile s useful life assumptions as provided for each class of property and equipment in the table below. The useful life assumptions differ from MetroPCS property and equipment useful life assumptions because each company s assumptions are based on its own historical experience with similar assets and its own intended use for the assets, while also taking into account anticipated technological or other changes.

(dollars in millions)	Useful Lives	Adjus	sted Basis
Wireless communications systems	3-20 years	\$	1,624
Furniture and equipment	3-5 years		40
Construction-in-progress			76

\$ 1,740

Historical depreciation and amortization expense was adjusted for the fair value adjustment decreasing the basis of property and equipment, as well as the useful life assumption changes. The fair value adjustment to basis most significantly impacts certain network-related equipment and construction costs in wireless communication systems. Historically reported depreciation was reduced in part due to the asset basis adjustment, in particular to network equipment representing 3G or third generation technologies as well as redundant macro cell sites. Offsetting the impact of the adjustment to basis is an increase in depreciation due to generally shorter T-Mobile useful life assumptions as introduced by differences in policy as well as the intended use of the property and equipment. The approximate impact of each is illustrated below:

(dollars in millions)	 Ended er 31, 2011	Septen	ths Ended aber 30, 012
Historical depreciation and amortization expense	\$ 539	\$	469
Adjustments related to change in:			
Asset basis of property and equipment, net	(312)		(270)
Useful life assumptions	82		50
Pro forma adjustment	(230)		(220)

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

Adjusted MetroPCS depreciation and amortization expense

\$ 309 **\$** 249

(b) Represents adjustments to record amortization expense related to other identifiable intangible assets calculated on a straight-line basis. These identifiable intangible assets include:

*Subscriber relationships (\$550 million)* Represents relationships with U.S. domestic subscribers that are expected to have an estimated useful life of approximately five years.

*Trade names (\$100 million)* Represents the MetroPCS brand that has been estimated to have a useful life of approximately seven years.

Indefinite lived intangible assets, including spectrum licenses and goodwill, are not subject to amortization but will be tested for impairment annually or more frequently if events or changes in circumstances indicate that the assets might be impaired.

(c) Represents adjustments to income tax expense for each period as a result of the tax impact on the pro forma adjustments. MetroPCS and T-Mobile utilized their respective statutory tax rates to compute the income tax expense related to each entity s pro forma condensed combined statements of operations adjustments as follows:

		D	Year E ecember		
(dollars in millions)	Me	troPCS	T-	Mobile	Total
Pro Forma Adjustments	\$	55	\$	(473)	\$ (418)
Statutory Rate		38.6%		37.5%	
Tax Impact	\$	(21)	\$	177	\$ 156

				hs Ended 30, 2012	
(dollars in millions)	Me	troPCS	T-	Mobile	Total
Pro Forma Adjustments	\$	91	\$	(371)	\$ (280)
Statutory Rate		38.6%		37.5%	
Tax Impact	\$	(35)	\$	139	\$ 104

(d) Represents the loss per share, taking into consideration the pro forma weighted average shares outstanding calculated assuming the acceleration of the vesting of the restricted stock, applying the reverse stock split and the issuance of common stock equal to 74% of the fully-diluted shares of common stock outstanding.

	Year Ended December 31, 2011	Nine Months Ended September 30, 2012
Weighted average shares outstanding	360,410,168	363,190,434
Unvested restricted shares	3,503,050	3,503,050
	363,913,218	366,693,484
Reverse 1:2 stock split	0.50	0.50
	181,956,609	183,346,742
Issuance of shares to Deutsche Telekom	537,583,132	537,583,132
Pro forma basic and diluted weighted average shares	719,539,741	720,929,874

### 5. Financing Agreements

As described in Note 1, the transaction is expected to be financed by the issuance of senior unsecured notes to T-Mobile and MetroPCS. The adjustments represent the reduction of \$15.1 billion to current and long term payables to affiliates of \$1.5 billion (including accrued interest) and \$13.6 billion, respectively, the removal of \$437 million of other current assets, which represents net receivable from affiliate and interest rate swaps, and a loss on extinguishment of swaps of \$156 million. The adjustments represent the assumption of T-Mobile s intercompany indebtedness by Deutsche Telekom, including the write-off of T-Mobile s cross currency interest rate swaps and interest rate swaps related to such indebtedness. The addition of the \$15.0 billion notes reflects the notes T-Mobile will issue to Deutsche Telekom prior to or upon the closing of the transaction. Pursuant to the business combination agreement, the net receivable of \$196 million resulting from the exchange of indebtedness will be distributed to Deutsche Telekom.

The estimated committed interest rates for the \$15.0 billion senior notes are 8.16% and 7.28% for the permanent notes (\$7.5 billion) and reset notes (\$7.5 billion), respectively. The estimated committed interest rates are calculated based on current market information and a formula as outlined in the business combination agreement. See \$15.0 Billion Notes to Refinance T-Mobile-Deutsche Telekom Intercompany Indebtedness, and up to \$3.5 Billion Notes as Backstop for \$2.5 Billion Wireless Credit Agreement Refinancing and \$1.0 Billion Wireless New Notes on page 113 of the Revised Preliminary Proxy Statement. The first component of the formula is a reference yield which is based upon (i) three indices of high-yield bonds issued by telecommunications companies (50% weight), (ii) the prices of comparable bonds issued by Sprint Nextel Corporation (25% weight) and (iii) the prices of Wireless securities (25% weight). In calculating the estimated committed interest rates, the reference yield was adjusted as follows: (1) plus 100 basis points for reset notes or 187.5 basis points for non-reset notes (assuming an eight year term). As a result of T-Mobile s exchange of indebtedness and removal of the related interest rate swaps, the pro forma adjustments reflect the adjustment to historical interest expense to record the estimated pro forma interest expense under the senior notes of \$1.15 billion for the year ended December 31, 2011 and \$870 million for the nine months ended September 30, 2012.

The pro forma adjustments also reflect the incurrence by MetroPCS of \$1.0 billion of additional fixed-rate senior unsecured notes to third-party investors. The \$1.0 billion is incremental to (i) the pre-existing \$2.0 billion in senior unsecured notes, which will remain outstanding with terms amended pursuant to revised supplemental indentures relating thereto entered into on December 14, 2012, and (ii) the pre-existing \$2.5 billion senior secured credit facility, which is to be refinanced in accordance with the terms of the business combination agreement. With the limited information available with regards to the terms of refinancing, the weighted average interest rate for MetroPCS senior secured credit facility of 5.065% and 4.614% for the year ended December 31, 2011 and the nine months ended September 30, 2012, respectively, was used for the purpose of calculating the interest expense presented in the unaudited pro forma condensed combined statements of operations on the incremental \$1.0 billion. The pro forma adjustments reflect the estimated incremental pro forma interest expense of \$51 million for the year ended December 31, 2011 and \$35 million for the nine months ended September 30, 2012.

A sensitivity analysis on interest expense for the year ended December 31, 2011 and for the nine months ended September 30, 2012, has been performed to assess the effect of a change of 12.5 basis points of the hypothetical interest rate on the \$15.0 billion notes and the \$1.0 billion of fixed-rate senior unsecured notes offered by Wireless to third-party investors and Deutsche Telekom.

The following table shows the change in interest expense for the \$15.0 billion notes (in millions):

Interest expense assuming	Year Ended December 31, 2011	Nine Months Ended September 30, 2012
Increase of 0.125%	\$ 1,177	\$ 1,139
Decrease of 0.125%	883	854

The following table shows the change in interest expense for the \$1.0 billion of fixed-rate senior unsecured notes offered by Wireless (in millions):

	Year Ended	Nine Months Ended		
Interest expense assuming	December 31, 2011	September 30, 2012		
Increase of 0.125%	\$ 52	\$ 36		
Decrease of 0.125%	50	34		
naantian				

## 6. Tower Transaction

On September 28, 2012, T-Mobile and Crown Castle International Corp. (CCI) entered into definitive agreements under which CCI was to acquire rights to approximately 7,200 wireless communication tower sites owned by T-Mobile (also referred to in this document as the tower transaction). Prior to closing on November 30, 2012, the tower portfolio included in the transaction was reduced to 7,105 sites. During the fourth quarter of 2012, T-Mobile received net proceeds of \$2,469 million of which it distributed \$2,403 million to its parent, Deutsche Telekom. Under the agreements, CCI acquired the sole right to lease and operate the wireless communication tower sites through the purchase of 896 wireless communications tower sites, and through long-term prepaid lease and management agreements for the remaining 6,209 wireless communications tower sites. Furthermore, CCI was granted an option to acquire any wireless communications tower sites that were not sold to CCI at the end of their respective lease terms. T-Mobile will lease back and continue to have rights to space on the wireless communications tower sites required to operate its wireless communications tower sites which were sold to CCI, will be accounted for as a financing transaction. T-Mobile will record rental income for the lease of the portion of the towers not leased back by T-Mobile. Pursuant to the business combination agreement with MetroPCS, the parties agreed that T-Mobile will be permitted to consummate the tower transaction or any other transfer of the T-Mobile wireless communications tower sites on terms that, taken as a whole, have an equivalent or more favorable economic impact to T-Mobile and its subsidiaries as compared to such tower transaction.

The adjustments in the unaudited pro forma condensed combined balance sheet assume the tower transaction occurred on September 30, 2012. All proceeds from the tower transaction are included as a financial liability on the balance sheet, and the net proceeds are considered to be distributed to Deutsche Telekom. The adjustments in the unaudited pro forma condensed combined statements of operations assume the tower transaction occurred on January 1, 2011, resulting in approximately \$135 million and \$101 million in co-location revenue from leasing the portion of wireless communications tower sites not leased back by T-Mobile to CCI for the year ended December 31, 2011 and the nine months ended September 30, 2012, respectively, and approximately \$120 million and \$90 million in interest expense related to the financial liability for the year ended December 31, 2011 and the nine months ended September 30, 2012, respectively. The effective interest rate used was 4.9% for purposes of the unaudited pro forma condensed combined financial statements based upon the imputed interest rate applied in the transaction.

#### 7. Trademark License

MetroPCS and Deutsche Telekom have agreed to enter into a trademark license in connection with the completion of the transaction, under which the combined company is obligated to pay Deutsche Telekom a royalty in an amount equal to 0.25% of the net revenue generated by products and services sold by the combined company under the licensed trademarks. Under the trademark license, products and services sold by the combined company under the MetroPCS brand or trademarks owned by the combined company are excluded from the royalty calculation so long as they are not used in conjunction with the trademarks subject to the trademark license.

On the fifth anniversary of the trademark license, the combined company and Deutsche Telekom have agreed to adjust the royalty rate based on the then average commercial royalty rate found under similar licenses for trademarks in the field of wireless telecommunication, broadband and information products and services in the territory through a binding benchmarking process. The adjustment of the royalty rate will depend on the then average commercial royalty rates charged in the field of wireless telecommunication, broadband and information products and services in the territory covered by the trademark license, which can be affected by a number of factors, including the royalty rate charged by others in the relevant field and the relative value of the T-Mobile trademark. Royalty rates vary considerably, are dependent on a number of factors which cannot be known at this time, and can change year-by-year, making it difficult at this time to determine any definitive estimate of what the current average commercial royalty rate will be in five years.

Within the unaudited pro forma condensed combined statements of operations, no pro forma adjustments are presented for the trademark license because it is expected that the expenses associated with the trademark license will be substantially the same as in the T-Mobile consolidated audited financial statements for the three years ended December 31, 2011 beginning on page F-3 of this proxy statement. The financial statements of T-Mobile reflect royalty expenses from an existing licensing agreement between Deutsche Telekom and T-Mobile which management believes approximate the expenses under the new trademark license. Additionally, the combined company expects to continue to sell MetroPCS services and products under the MetroPCS brand or trademarks and not use such MetroPCS brand or trademarks in conjunction with the trademarks subject to the licensing agreements; and, therefore, the combined company is not expected to incur royalty expenses on net revenues generated from the sale of MetroPCS branded services and products.

# DATES FOR SUBMISSION OF STOCKHOLDER PROPOSALS FOR 2013 ANNUAL MEETING

MetroPCS will hold its 2013 annual meeting of stockholders regardless of whether the transaction has been completed.

*Proposals Pursuant to Rule 14a-8.* Pursuant to Rule 14a-8 under the Exchange Act, MetroPCS stockholders may present proper proposals for inclusion in our proxy statement and for consideration at our 2013 annual meeting of stockholders. To be eligible for inclusion in our proxy statement with respect to our 2013 annual meeting of stockholders under Rule 14a-8, your proposal must be received by us no later than the close of business on December 17, 2012, and must otherwise comply with Rule 14a-8. While the MetroPCS board will consider stockholder proposals, MetroPCS reserves the right to omit from our proxy statement stockholder proposals that we are not required to include under the Exchange Act, including Rule 14a-8.

*Business Proposals and Nominations Pursuant to our Bylaws.* Under MetroPCS bylaws, in order to nominate a director or bring any other business before the stockholders at the 2013 annual meeting of stockholders that will not be included in our proxy statement with respect to our 2013 annual meeting of stockholders pursuant to Rule 14a-8, you must comply with the procedures and timing specifically described in our bylaws. In addition, assuming the date of the 2013 annual meeting of stockholders is not more than 30 days before and not more than 60 days after the anniversary date of the 2012 annual meeting of stockholders, you must notify MetroPCS in writing and such written notice must be delivered to MetroPCS Secretary no earlier than January 24, 2013, and no later than February 23, 2013.

A copy of MetroPCS bylaws setting forth the requirements for the nomination of director candidates by stockholders and the requirements for proposals by stockholders may be obtained free of charge from MetroPCS Secretary at the address indicated on the first page of this proxy statement or on MetroPCS website at *www.metropcs.com* under the Investor Relations tab, Corporate Governance. A nomination or proposal that does not comply with the above procedures will be disregarded. Compliance with the above procedures does not require MetroPCS to include the proposed nominee or proposal in our proxy solicitation material.

#### WHERE YOU CAN FIND MORE INFORMATION

MetroPCS files annual, quarterly and current reports, proxy statements and other information with the SEC. You may read and copy these reports, proxy statements and other information at the SEC s Public Reference Section at 100 F Street, N.E., Washington, D.C. 20549. Our SEC filings are available to the public on the SEC s website at *http://www.sec.gov*.

The SEC allows us to incorporate by reference information into this proxy statement. This means that we are disclosing important business and financial information to you by referring you to another document filed separately with the SEC. These documents contain important information about us and our financial condition. The information incorporated by reference is considered to be part of this proxy statement. Information that we file later with the SEC will automatically update and supersede the information included or incorporated by reference into this proxy statement.

We incorporate by reference into this proxy statement the documents listed below and any filings we make with the SEC under Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act from the date of this proxy statement until the date that the MetroPCS special meeting is held. The information incorporated by reference is an important part of this proxy statement. Nothing in this proxy statement shall be deemed to incorporate information furnished to, but not filed with, the SEC. Any statement in a document incorporated by reference into this proxy statement will be deemed to be modified or superseded to the extent a statement contained in (1) this proxy statement, or (2) any other subsequently filed document that is incorporated by reference into this proxy statement modifies or supersedes such statement.

MetroPCS Annual Report on Form 10-K for the fiscal year ended December 31, 2011

MetroPCS Quarterly Reports on Form 10-Q for the three months ended March 31, 2012, June 30, 2012 and September 30, 2012

MetroPCS Current Reports on Form 8-K filed on January 5, 2012, February 14, 2012, May 31, 2012, August 3, 2012, October 2, 2012, October 3, 2012, November 19, 2012, December 5, 2012, December 7, 2012 and December 17, 2012 (other than documents or portions of those documents not deemed to be filed)

MetroPCS Definitive Proxy Statement on Schedule 14A filed on April 16, 2012 You may request a copy of these filings, at no cost, using the following contact information:

MetroPCS Communications, Inc.

2250 Lakeside Boulevard

Richardson, Texas 75082

Attention: Investor Relations

(214) 570-4641

#### Email: investor\_relations@metropcs.com

You may also request a copy of these filings from our proxy solicitor, MacKenzie, using the following contact information:

MacKenzie Partners, Inc.

105 Madison Avenue

New York, New York 10016

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

# Call Collect (212) 929-5500

or

Call Toll-Free (800) 322-2885

Email: proxy@mackenziepartners.com

THIS PROXY STATEMENT DOES NOT CONSTITUTE THE SOLICITATION OF A PROXY IN ANY JURISDICTION TO OR FROM ANY PERSON TO WHOM OR FROM WHOM IT IS UNLAWFUL TO MAKE SUCH PROXY SOLICITATION IN THAT JURISDICTION. YOU SHOULD RELY ONLY ON THE INFORMATION CONTAINED OR INCORPORATED BY REFERENCE IN THIS PROXY STATEMENT TO VOTE YOUR SHARES OF METROPCS COMMON STOCK AT THE SPECIAL MEETING. WE HAVE NOT AUTHORIZED ANYONE TO PROVIDE YOU WITH INFORMATION THAT IS DIFFERENT FROM WHAT IS CONTAINED IN THIS PROXY STATEMENT. THIS PROXY STATEMENT IS DATED []. YOU SHOULD NOT ASSUME THAT THE INFORMATION CONTAINED IN THIS PROXY STATEMENT IS ACCURATE AS OF ANY DATE OTHER THAN THAT DATE, AND THE MAILING OF THIS PROXY STATEMENT TO STOCKHOLDERS DOES NOT CREATE ANY IMPLICATION TO THE CONTRARY.

## INDEX TO FINANCIAL STATEMENTS OF T-MOBILE

<u>T-Mobile USA, Inc. Consolidated Financial Statements</u> <u>T-Mobile USA, Inc. Condensed Consolidated Interim Financial Statements</u> F-3 F-34

T-Mobile USA, Inc.

Index

	Page(s)
Consolidated Financial Statements	
December 31, 2011, 2010 and 2009	
Report of Independent Auditors	F-3
Consolidated Balance Sheets	F-4
Consolidated Statements of Operations and Comprehensive Income (Loss)	F-5
Consolidated Statements of Stockholder s Equity	F-6
Consolidated Statements of Cash Flows	F-7
Notes to Consolidated Financial Statements Condensed Consolidated Interim Financial Statements	F-8

# September 30, 2012 and December 31, 2011

Condensed Consolidated Balance Sheets	F-34
Condensed Consolidated Statements of Operations and Comprehensive Income (Loss)	F-35
Condensed Consolidated Statements of Cash Flows	F-36
Notes to Condensed Consolidated Interim Financial Statements	F-37

#### **Report of Independent Auditors**

To the Board of Directors and

Stockholder of T-Mobile USA, Inc.:

In our opinion, the accompanying consolidated balance sheets and the related consolidated statements of operations and comprehensive income (loss), of stockholder s equity, and of cash flows present fairly, in all material respects, the financial position of T-Mobile USA, Inc. and its subsidiaries at December 31, 2011 and 2010, and the results of their operations and their cash flows for each of the three years in the period ended December 31, 2011 in conformity with accounting principles generally accepted in the United States of America. These financial statements are the responsibility of the Company s management. Our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

/s/ PricewaterhouseCoopers LLP

Seattle, Washington

April 20, 2012, except for earnings per share disclosed in Note 15 to the consolidated financial statements, as to which the date is November 16, 2012.

T-Mobile USA, Inc.

**Consolidated Balance Sheets** 

December 31, 2011 and 2010

(dollars in millions, except per share amounts)	2011	2010
Assets		
Current assets		
Cash and cash equivalents	\$ 390	\$ 109
Accounts receivable, net of allowances for uncollectible accounts of \$347 and \$302, respectively	2,697	2,857
Accounts receivable from affiliates	1,820	310
Inventory	455	621
Current portion of deferred tax assets, net	668	914
Other current assets	572	500
Total current assets	6,602	5,311
Property and equipment, net of accumulated depreciation of \$15,599 and \$13,801, respectively	12,703	13,213
Goodwill	8,134	12,044
Spectrum licenses	12,814	15,282
Other intangible assets, net of accumulated amortization of \$216 and \$163, respectively	61	113
Investments in and advances to unconsolidated affiliates	35	33
Long-term investments	22	92
Other assets	238	203
	250	205
Total assets	\$ 40,609	\$ 46,291
Liabilities and Stockholder s Equity		
Current liabilities		
Accounts payable	\$ 1,826	\$ 2,170
Accrued liabilities	1,232	1,078
Current payables to affiliates	1,046	805
Deferred revenue	257	299
Other current liabilities	143	103
Total current liabilities	4,504	4,455
Long-term payables to affiliates	15,049	15,854
Deferred tax liabilities	3,282	3,756
Deferred rents	1,672	1,446
Other long-term liabilities	317	288
Other long-term indomnes	517	200
Total long-term liabilities	20,320	21,344
Commitments and contingencies (Note 18)		
Stockholder s equity		
Common stock, par value \$0.000001 per share, and paid-in capital; 500,000,000 shares authorized,		
292,669,971 shares issued and outstanding	31,600	31,600
Accumulated other comprehensive loss	(28)	(39)
Accumulated deficit	(15,787)	(11,069)
Total stockholder s equity	15,785	20,492

Total liabilities and stockholder s equity

The accompanying notes are an integral part of these consolidated financial statements.

\$ 40,609

\$ 46,291

T-Mobile USA, Inc.

Consolidated Statements of Operations and Comprehensive Income (Loss)

Years Ended December 31, 2011, 2010 and 2009

dollars in millions, except per share amounts)		2011		2010		2009
Revenues Branded contract revenues	\$	16,230	\$	16,538	\$	16,750
Branded prepaid revenues	φ	1,307	φ	1,384	φ	1,488
Wholesale revenues		443		1,584		59
Roaming and other service revenues		501		612		663
Comming and other service revenues		501		012		005
Fotal service revenues		18,481		18,733		18,960
Equipment sales		1,901		2,404		2,403
Other revenues		236		210		168
Fotal revenues		20,618		21,347		21,531
Operating expenses						
Network costs, excluding depreciation and amortization		4,952		4,895		4,936
Cost of equipment sales		3,646		4,237		3,856
Customer acquisition, excluding depreciation and amortization		3,185		3,205		3,382
General and administrative, excluding depreciation and amortization		3,543		3,535		3,442
Depreciation and amortization		2,982		2,773		2,859
mpairment charges		6,420				
Other, net		169		(3)		
Fotal operating expenses		24,897		18,642		18,475
Operating (loss) income		(4,279)		2,705		3,056
Other (expense) income						
Interest expense to affiliates		(670)		(556)		(740
Interest income		25		14		12
Other (expense) income, net		(10)		16		8
Fotal other expense, net		(655)		(526)		(720
Loss) income before income taxes		(4,934)		2,179		2,336
income tax benefit (expense)		216		(822)		(860
Net (loss) income		(4,718)		1,357		1,476
Net income attributable to non-controlling interest				(3)		(6
Net (loss) income attributable to T-Mobile USA, Inc.	\$	(4,718)	\$	1,354	\$	1,470
Other comprehensive (loss) income, net of tax						
Unrealized loss on derivatives held as cash flow hedges, net of tax of						
\$28, \$42 and \$0, respectively		(47)		(71)		
Unrealized gain on foreign currency translation, net of tax of \$29, \$23						
and \$0, respectively		49		40		
						(8

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

Unrealized gain (loss) on available-for-sale securities, net of tax of \$5, \$0 and \$5, respectively						
Total comprehensive (loss) income attributable to T-Mobile USA, Inc.	\$	(4,707)	\$	1,323	\$	1,462
Net (loss) income per common share (Note 15):						
Basic and diluted	\$	(16.12)	\$	4.63	\$	5.02
Weighted average shares:						
Basic and diluted	292	2,669,971	292	2,669,971	292	2,669,971

The accompanying notes are an integral part of these consolidated financial statements.

T-Mobile USA, Inc.

Consolidated Statements of Stockholder s Equity

# Years Ended December 31, 2011, 2010 and 2009

(dollars in millions)	Number of Common Shares	Par Value and Additional Paid-in Capital	Accume Oth Comprel Los	er 1ensive	cumulated Deficit	1	al T-Mobile USA, Inc. ockholder s Equity		ontrolling terest	Total ckholder s Equity
Balances at December 31,	Shares	Capital	1.0.	55	Denen		Equity	111	ierest	Equity
2008	292,669,971	\$ 36,594	\$		\$ (13,906)	\$	22,688	\$	95	\$ 22,783
Income tax expense on share-based compensation awards and share-based compensation expense		(1)	·			·	(1)			(1)
Net income attributable to										
noncontrolling interest									6	6
Net income attributable to T-Mobile USA, Inc.					1,470		1,470			1,470
Other comprehensive loss, net										
of tax				(8)			(8)			(8)
Balances at December 31,				, í						, ,
2009	292,669,971	36,593		(8)	(12,436)		24,149		101	24,250
Equity distribution of paid-in	2,00,,,,,	50,575		(0)	(12,150)		21,119		101	21,230
capital in exchange for										
payables to affiliates		(5,000)					(5,000)			(5,000)
Income tax expense on share-based compensation awards and loss from conversion to cash-settled										
awards		7			(2)		5			5
Net income attributable to noncontrolling interest									3	3
Derecognition of										
noncontrolling interest					15		15		(104)	(89)
Net income attributable to T-Mobile USA, Inc.					1,354		1,354			1,354
Other comprehensive loss, net of tax				(31)			(31)			(31)
Balances at December 31,										
2010	292,669,971	31,600		(39)	(11,069)		20,492			20,492
Net loss					(4,718)		(4,718)			(4,718)
Other comprehensive income, net of tax				11			11			11
Balances at December 31, 2011	292,669,971	\$ 31,600	\$	(28)	\$ (15,787)	\$	15,785	\$		\$ 15,785

The accompanying notes are an integral part of these consolidated financial statements.

T-Mobile USA, Inc.

**Consolidated Statements of Cash Flows** 

Years Ended December 31, 2011, 2010 and 2009

(dollars in millions)	2011	2010	2009
Operating activities			
Net (loss) income	\$ (4,718)	\$ 1,357	\$ 1,476
Adjustments to reconcile net (loss) income to net cash provided by operating activities			
Impairment charges	6,420		
Depreciation and amortization	2,982	2,773	2,859
Income tax (benefit) expense	(216)	822	860
Amortization of debt discount and premium, net	(84)	(59)	(39)
Bad debt expense	713	619	528
Deferred rent expense	218	219	243
Other non-cash income and expense, net	(43)	(52)	(1)
Changes in operating assets and liabilities			
Accounts receivable	(558)	(862)	(468)
Inventory	166	19	291
Other current and long-term assets	(182)	62	(31)
Accounts payable	(51)	(14)	(181)
Other current and accrued liabilities	157	21	(100)
Accrued liability for transaction-related costs	176		
Net cash provided by operating activities	4,980	4,905	5,437
Investing activities			
Purchases of property and equipment	(2,729)	(2,819)	(3,687)
Expenditures related to spectrum licenses	(23)	(18)	(3,007)
Short term affiliate loan receivable, net	(2,005)	(2,315)	(1,888)
Proceeds from disposals of property and equipment	2	(2,313)	(1,000)
Proceeds from disposals of financial assets	73	20	3
Investments in and advances to unconsolidated affiliates, net	(17)	(11)	2
investments in and advances to unconsolidated armates, net	(17)	(11)	2
Net cash used in investing activities	(4,699)	(5,126)	(5,603)
Financing activities			
Long-term debt repayments to affiliates			(133)
Long-term borrowings from affiliates		116	200
Other, net		7	
Net cash provided by financing activities		123	67
Change in cash and cash equivalents	281	(98)	(99)
	201	(96)	(99)
Cash and cash equivalents	100	207	207
Beginning of year	109	207	306
End of year	\$ 390	\$ 109	\$ 207

The accompanying notes are an integral part of these consolidated financial statements.

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

#### 1. Organization

T-Mobile USA, Inc. (T-Mobile or the Company) is Deutsche Telekom AG s (Deutsche Telekom) wholly-owned U.S. mobile communications segment. T-Mobile provides mobile communications services under the T-Mobile brand in the United States, Puerto Rico and the U.S. Virgin Islands. T-Mobile provides services using the Global System for Mobile Communications (GSM), General Packet Radio Service (GPRS), Enhanced Data rates for GSM Evolution (EDGE), Universal Mobile Telecommunications Systems (UMTS) and Evolved High Speed Packet Access (HSPA+) technologies. T-Mobile s voice and data networks cover over 280 million people. At the end of 2011, T-Mobile s most advanced 4G HSPA+ network covered 184 million people. During 2010, T-Mobile established a wholly-owned single-parent captive insurance company to provide reinsurance for handset insurance policies and extended warranty contracts offered to T-Mobile s wireless customers.

#### 2. Summary of Significant Accounting Policies Consolidation and Basis of Presentation

The consolidated financial statements (financial statements) of T-Mobile include the accounts of all wholly-owned and all majority-owned subsidiaries over which T-Mobile exercises control, as well as variable interest entities where T-Mobile is deemed to be the primary beneficiary (Note 8). Entities over which T-Mobile exercises significant influence, but does not control and is not the primary beneficiary are accounted for using the equity method. Entities over which T-Mobile is not able to exercise significant influence are accounted for under the cost method. Certain prior period amounts have been reclassified to conform to the current period presentation. Intercompany transactions and balances have been eliminated in consolidation.

#### **Operating Segments**

The Company operates in a single operating segment and a single reporting segment as a wireless communications carrier. As of December 31, 2011 and 2010, and for the years ended, December 31, 2011, 2010 and 2009, all of the Company s revenues and long-lived assets related to operations in the United States, Puerto Rico and the U.S. Virgin Islands.

## **Cash and Cash Equivalents**

T-Mobile includes highly liquid interest-earning investments with remaining maturities of three months or less at the date of purchase as cash equivalents. Cash equivalents are stated at cost, which approximates fair value. As of December 31, 2011 and 2010, outstanding checks of \$222.1 million and \$198.7 million, respectively, are included in accounts payable.

#### Accounts Receivable and Allowance for Uncollectible Accounts

Accounts receivable consist of amounts billed and currently due from customers, other carriers and third-party retail channels ( dealers ), as well as revenues earned but not yet billed at the end of each period. T-Mobile maintains an allowance for estimated losses resulting from uncollectible balances based on a number of factors, including collection experience, aging of the accounts receivable portfolio, credit quality of the customer base and other qualitative factors such as macro-economic conditions. The Company writes off account balances if collection efforts are unsuccessful and future collection is unlikely based on customer credit ratings and the length of time from the original billing date.

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

The Company offers certain retail customers the option to pay for their handset and other purchases in installments over a period of up to 20 months. At the time of sale, the Company imputes interest on the transaction and records the deferred interest as a reduction to equipment revenues and the related accounts receivable. Interest income is recognized over the financed term of the customer contract. T-Mobile maintains an allowance for estimated losses resulting from uncollectible balances based on the aging of the Company s equipment installment plan balances and expected write-off experience (Note 4).

The following table summarizes the changes in the Company s allowance for uncollectible accounts related to its short-term accounts receivable balances:

(dollars in millions)	2011	2010	2009
Allowance, Beginning of Year	\$ 302	\$ 278	\$ 235
Bad debt expense	713	619	528
Write-offs	(698)	(598)	(485)
Change in deferred interest on short-term installment receivables (Note 4)	30	3	
Allowance, End of Year	\$ 347	\$ 302	\$ 278

#### Inventory

The Company s inventory consists primarily of wireless handsets and accessories, which are valued at the lower of cost or market (determined using current replacement cost). T-Mobile sells handsets and other mobile communication devices separately and in connection with service contracts. As part of the strategy to acquire and retain new customers the Company sometimes sells handsets and other mobile communication devices, in connection with a service contract, at a stated price below its standard cost. Because the handset discount is part of the Company s strategy for acquiring and retaining customers, the loss on the sale of handset ( handset subsidy ) is recognized at the time of the sale. The handset subsidy is expected to be recovered through future service revenue on the customer contract. Shipping and handling costs paid to wireless handset, device and accessory vendors are included in the standard cost of inventory. T-Mobile records inventory write-downs for obsolete and slow-moving items based on inventory turnover trends and historical experience.

#### Long-Lived Assets

Long-lived assets include assets such as property and equipment and intangible assets that do not have indefinite lives. The Company assesses potential impairments to its long-lived assets when there is evidence that events or changes in circumstances indicate that the carrying value may not be recoverable and exceeds its fair value. The carrying value of a long-lived asset or asset group is not recoverable if it exceeds the sum of the undiscounted cash flows expected to result from the use and eventual disposition of the asset or asset group. An impairment loss is measured as the amount by which the carrying amount of a long-lived asset or asset group exceeds its fair value.

Property and equipment are recorded at cost less accumulated depreciation and impairments, if any. Costs of major replacements and improvements are capitalized while expenditures that do not enhance or extend the asset s useful life are charged to operating expenses as incurred. Construction costs, labor and overhead incurred in the expansion or enhancement of T-Mobile s wireless network are capitalized. Capitalization commences with pre-construction period administrative and technical activities, which includes obtaining leases, zoning approvals and building permits, and ceases at the point at which the asset is ready for its intended use. T-Mobile capitalizes interest associated with the acquisition or construction of property and equipment. Capitalized interest is reported as a reduction in interest expense and depreciated as part of the

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

cost of the related asset. The Company recognized capitalized interest of \$23.8 million, \$35.0 million and \$56.2 million for the years ended December 31, 2011, 2010 and 2009, respectively. Depreciation commences once assets have been placed in service and is computed using the straight-line method over the estimated useful life of each asset. Depreciable life studies are performed periodically to confirm the appropriateness of depreciable lives for certain categories of property and equipment. These studies take into account actual usage, physical wear and tear, replacement history and assumptions about technology evolution. When these factors indicate that an asset s useful life is different from the previous assessment, the remaining book values are depreciated prospectively over the adjusted remaining estimated useful life. Leasehold improvements are depreciated over the shorter of their estimated useful lives or the related lease term.

T-Mobile records a liability for the fair value of legal obligations associated with the retirement of tangible long-lived assets and a corresponding increase in the carrying amount of the related asset in the period in which the obligation is incurred. Over time, the liability is accreted to its present value, and the capitalized cost is depreciated over the estimated useful life of the asset. The Company s obligations relate primarily to certain legal obligations to remediate leased property on which the Company s network infrastructure and administrative assets are located. The Company s liability associated with the retirement of long-lived assets was \$114.9 million and \$104.8 million as of December 31, 2011 and 2010, respectively. The corresponding asset related to the liability for the retirement of long-lived assets was \$59.3 million and \$63.5 million as of December 31, 2011 and 2010, respectively. The Company recognized accretion expense in network costs of \$6.9 million, \$5.5 million and \$3.0 million for the years ended December 31, 2011, 2010 and 2009, respectively. During the years ended December 31, 2011 and 2010, no significant liabilities related to the asset retirement obligation have been incurred or settled.

#### **Software Capitalization**

The Company capitalizes certain costs incurred in connection with developing or acquiring internal use software. Capitalization on software commences once the final selection of software has been made and management authorizes and commits to funding the software project, and ceases at the point at which the software is ready for its intended use. Capitalized costs include direct development costs associated with internal use software, including internal direct labor costs and external costs of materials and services. Capitalized software costs are included in property and equipment, net and are being amortized on a straight-line basis over a period of up to seven years. Costs incurred during the preliminary project stage, as well as maintenance and training costs are expensed as incurred.

#### Goodwill

As of December 31, 2011, the carrying value of T-Mobile s goodwill is \$8.1 billion. Goodwill consists of the excess of the purchase price over the fair value of net identifiable assets acquired in purchase business combinations. At December 31, 2011 and 2010, the majority of goodwill is related to Deutsche Telekom s acquisition of VoiceStream Wireless Corporation and Powertel, Inc. (now jointly T-Mobile) on May 31, 2001.

#### Spectrum Licenses

As of December 31, 2011, the carrying value of T-Mobile s spectrum licenses is \$12.8 billion. The Company has determined that its spectrum licenses should be treated as indefinite-lived intangible assets. Spectrum licenses are carried at costs incurred to acquire spectrum licenses and the costs, if any, to prepare the licenses for use. The Federal Communications Commission (FCC) issues spectrum licenses that

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

authorize wireless carriers to provide service in specific geographic service areas for up to fifteen years. The spectrum licenses provide the Company with the exclusive right to utilize designated radio frequency spectrum to provide mobile communication services. The FCC has specific standards applying to wireless spectrum license renewals which provide that it will award a spectrum license renewal to a licensee that has met certain standards based on past performance. Historically, the FCC has granted license renewals routinely and at a nominal cost. The spectrum licenses held by the Company expire at various dates; however, the Company believes it will be able to meet all requirements necessary to secure renewal of its spectrum licenses. Moreover, the Company has determined that there are currently no legal, regulatory, contractual, competitive, economic or other factors that limit the useful lives of its spectrum licenses and therefore treats the spectrum licenses as indefinite-lived intangible assets.

Spectrum licenses acquired as part of an exchange of nonmonetary assets are valued at fair value at the time the exchange occurs. If the transaction lacks commercial substance or the fair value is not measurable the acquired spectrum licenses are recorded at the book value of the assets tendered.

#### Impairment Tests of Goodwill and Indefinite-lived Intangible Assets

The Company assesses the carrying value of its goodwill and other indefinite-lived intangible assets (spectrum licenses) for potential impairment annually as of December 31 or more frequently if events or changes in circumstances indicate that such assets might be impaired.

The Company uses a two-step process to determine impairments of goodwill. The first step involves determining the fair value of the reporting unit (calculated using a market approach and/or a discounted cash flow method) and comparing that measurement to the carrying value. If the fair value is less than the carrying value, there is a potential impairment and step two is performed. In the second step, a determination of the fair values of all of the assets and liabilities of the reporting unit is made, including those that may not be currently recorded, such as unfavorable/favorable leases and customer lists. The excess of the fair value of the reporting unit over the sum of the fair value of all of those assets and liabilities represents a new implied goodwill amount, which is then compared to the recorded goodwill. If the implied fair value of goodwill is lower than the carrying amount of goodwill then an impairment loss is recognized.

The Company tests its spectrum licenses for impairment on an aggregate basis, consistent with the Company s management of the overall business at a national level. The Company estimates fair value using the Greenfield approach, which is an income approach to estimate the price at which an orderly transaction to sell the asset would take place between market participants at the measurement date under current market conditions. If the estimated fair value of the spectrum licenses is lower than the carrying amount, an impairment loss is recognized.

#### Fair Value of Financial Instruments

As of December 31, 2011 and 2010, the carrying values of cash and cash equivalents, accounts receivable, accounts receivable from affiliates and accounts payable approximate fair value due to the short-term maturities of these instruments. There were no financial instruments with a carrying value materially different from their fair value, based on quoted market prices or rates for the same or similar instruments, or internal valuation models.

#### Investments

Investment securities consist of auction rate securities ( ARS ) and securities held in a grantor trust related to the Company s nonqualified deferred compensation plan. T-Mobile classifies its investment securities as

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

available for sale. As of December 31, 2011, the Company s ARS investments are included in other current assets as they are expected to be sold within one year. As of December 31, 2010, the Company s ARS investments were included in long-term investments. The securities related to the Company s nonqualified deferred compensation plan are included in long-term investments (Note 13).

#### **Derivative Instruments and Hedging Activities**

Derivative instruments are recorded on the balance sheet at their fair values (Note 10). Changes in the fair value of derivative instruments are recorded each period in interest expense or other comprehensive (loss) income, depending on the type of derivative and whether the derivative is designated as part of an effective hedge transaction. T-Mobile does not enter into derivatives for trading or speculative purposes. The Company has entered into derivative agreements to manage exposure to fluctuations in interest rates, through interest rate swap agreements. The Company also entered into derivative agreements to manage foreign currency risk along with interest rate risk, through cross currency interest rate swap agreements. The interest rate swap agreements are not designated as hedging instruments and, accordingly, changes in fair value related to such agreements are reflected within interest expense. The cross currency interest rate swap agreements manage foreign currency risk and interest rate risk related to the Company s payables to affiliates denominated in Euros. The cross currency interest rate swaps are designated as cash flow hedges and meet the criteria to qualify for hedge accounting. At the inception of the cash flow hedges and quarterly thereafter, the Company performs an assessment to determine whether changes in the fair values or cash flows of the hedged transaction. If at any time subsequent to the inception of the cash flow hedges, the assessment indicates that the derivative is no longer highly effective as a hedge, the Company will discontinue hedge accounting and recognize all subsequent derivative gains and losses in results of operations. The effective portion of the change in fair value of the cross currency interest rate swaps is recorded in other comprehensive (loss) income and reclassified to interest expense in the period in which the hedged transaction affects earnings.

#### **Revenue Recognition**

Service revenues are earned from providing access to and usage of the Company s mobile communications network and are recognized when the service is rendered. Branded contract service revenues may be billed in advance or in arrears depending on the plan or contract entered into by the customer. Branded prepaid service revenues include revenues earned from pay-in-advance customers generally not originated under contract; recognition of prepaid revenue is deferred until services are rendered or the prepaid balance expires. Access revenue from customers paying a recurring charge for specified services is recognized ratably over the service period. Usage revenue, including roaming revenue and long-distance revenue is recognized when the service is rendered. Wholesale revenues are earned for providing services to mobile virtual network operators and machine-to-machine customers and are recognized when the service is provided. Roaming and other service revenues primarily include revenues from other wireless carriers for roaming by their customers on the Company s network. Equipment sales, including those on installment plans (Note 4), are composed of revenues from the sale of mobile communication devices and accessories and are recognized when the products are delivered to and accepted by the customer or dealer.

The Company sells both wireless services and equipment to customers through its company-owned sales channels. For contracts that involve multiple components, such as wireless services and equipment, revenue is allocated between the separate units of accounting, based on such components relative selling prices on a standalone basis, subject to the requirement that revenue recognized is limited to the amounts already received from the customer that are not contingent upon the delivery of additional products or services to

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

the customer in the future. For sales made by third parties, the Company defers nonrefundable, up-front activation fees and associated costs to the extent of the related revenues and amortizes these over the estimated term of the customer relationship.

Federal Universal Service Fund (USF) and other fees which are assessed by various governmental authorities in connection with the services that the Company provides to its customers are reported on a gross basis in service revenues and network costs on the accompanying Consolidated Statements of Operations and Comprehensive Income (Loss). For the years ended December 31, 2011, 2010 and 2009, the Company recorded approximately \$462.8 million, \$460.1 million and \$396.0 million, respectively, of USF and other fees on a gross basis. Sales, use and excise taxes for all service plans are reported on a net basis in general and administrative expenses on the accompanying Consolidated Statements of Operations and Comprehensive Income (Loss).

#### Lease Accounting

The Company has operating leases for its cell sites, retail locations, corporate offices and dedicated transportation lines, some of which have escalating rentals during the initial lease term and during subsequent optional renewal periods. The Company recognizes rent expense on a straight-line basis, over the initial lease term and renewal periods that are considered reasonably assured at the inception of the lease (Note 18).

#### Advertising Expense

T-Mobile expenses the cost of advertising and other promotional expenditures to market the Company s brand as incurred. Advertising expense was \$711.0 million, \$582.4 million and \$693.8 million for the years ended December 31, 2011, 2010 and 2009, respectively.

#### **Income Taxes**

Deferred tax assets and liabilities are recognized based on temporary differences between the financial statement and tax bases of assets and liabilities using enacted tax rates expected to be in effect when these differences are realized. A valuation allowance is maintained against deferred tax assets when it is more likely than not that some portion or all of the deferred tax assets will not be realized.

The Company accounts for uncertainty in income taxes recognized in the financial statements in accordance with Accounting Standards Codification (ASC) ASC 740, Income Taxes, which provides guidance on the financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return. The Company uses judgment to determine whether it is more likely than not that a tax position will be sustained upon examination based on the technical merits of the position and adjusts unrecognized tax benefits in light of changes in facts and circumstances, such as changes in tax law, interactions with taxing authorities and developments in case law.

#### Other Comprehensive (Loss) Income

Other comprehensive (loss) income in 2011, 2010 and 2009 consisted of adjustments, net of tax, related to unrealized gains (losses) on available-for-sale securities, unrealized losses on cash flow hedging derivatives and unrealized gains on foreign currency translation. These are reported in accumulated other comprehensive loss as a separate component of stockholder s equity until realized in earnings.

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

The following table summarizes the components included in accumulated other comprehensive loss as of December 31, 2011 and 2010:

(dollars in millions)	2011	2010
Unrealized (loss) on derivatives held as cash flow hedges, net of tax	\$ (118)	\$(71)
Unrealized gain on foreign currency translation, net of tax	89	40
Unrealized gain (loss) on available-for-sale securities, net of tax	1	(8)
Accumulated other comprehensive loss	\$ (28)	\$ (39)

#### **Stock-based Compensation**

At December 31, 2011, T-Mobile did not have any authorized stock option plans except for suspended legacy plans, of which no options remained authorized or available for future grants. Stock options outstanding represent options to purchase shares of Deutsche Telekom American Depository Receipts. In 2010, Deutsche Telekom ceased listing its American Depository Receipts on the New York Stock Exchange, and as a result, all outstanding options were converted into cash settled stock appreciation rights with equivalent terms. Outstanding cash-settled stock appreciation rights are included in long-term payables to affiliates at fair value and were not material as of December 31, 2011 or 2010.

#### **Concentrations of Risk**

The Company sells handsets, wireless data devices and prepaid cards to T-Mobile s dealer network. T-Mobile s policy is to limit the amount of credit granted to individual dealers and to regularly monitor the dealer accounts receivable balances and activity. No dealer or customer accounted for greater than 10% of revenues or accounts receivable at December 31, 2011 or 2010.

T-Mobile purchases a substantial portion of its wireless infrastructure equipment and handsets from only a few major suppliers. Loss of any of these suppliers could adversely affect operations until a comparable substitute could be found.

Local and long-distance telephone and other companies provide certain communication services to T-Mobile. Disruption of these services could adversely affect operations until an alternative telecommunication provider could be found.

#### Use of Estimates

The preparation of consolidated financial statements in conformity with U.S. generally accepted accounting principles (GAAP) requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, revenues and expenses, and the disclosure of contingencies at the date of the consolidated financial statements. Significant estimates include service revenues earned but not yet billed, allowances for uncollectible accounts and sales returns, tax liabilities, deferred income taxes, the estimated useful lives of the Company s wireless communications equipment and other long-lived assets, reasonably assured renewal terms for operating leases, and fair value measurements related to goodwill, spectrum licenses, intangible assets, investments and derivatives. T-Mobile bases estimates on historical experience, where applicable, and other assumptions that management believes are reasonable under the circumstances. These estimates are inherently subject to judgment and actual results may differ from such estimates if management s assumptions prove invalid or conditions change.

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

#### **Recently-Issued Accounting Standards**

In June 2011, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) 2011-05, Presentation of Comprehensive Income. The new guidance eliminates the current option to report other comprehensive income and its components in the statement of changes in equity. Instead, an entity will be required to present either a continuous statement of net income and other comprehensive income or in two separate but consecutive statements. The Company has adopted this recently-issued accounting pronouncement for the year ended December 31, 2011.

In May 2011, the FASB issued ASU 2011-04, Amendments to Achieve Common Fair Value Measurement and Disclosure Requirements in U.S. GAAP and IFRS. The new guidance limits the highest-and-best-use measure to nonfinancial assets, permits certain financial assets and liabilities with offsetting positions in market or counterparty credit risks to be measured at a net basis, and provides guidance on the applicability of premiums and discounts. Additionally, the new guidance expands the disclosures on Level 3 inputs by requiring quantitative disclosure of the unobservable inputs and assumptions, as well as description of the valuation processes and the sensitivity of the fair value to changes in unobservable inputs. The new guidance will be effective for T-Mobile for the year ending December 31, 2012. The Company does not expect this recently-issued accounting pronouncement to have a material impact on results of operations, financial conditions or its financial disclosures.

In September 2011, FASB issued ASU 2011-08, Testing Goodwill for Impairment. It provides an option to first assess qualitative factors to determine whether it is necessary to perform the two-step quantitative goodwill impairment test. Given this option, an entity no longer would be required to calculate the fair value of a reporting unit unless the entity determines, based on that qualitative assessment, that it is more likely than not that its fair value is less than its carrying amount. The new guidance will be effective for T-Mobile for the year ending December 31, 2012. The Company does not expect this recently-issued accounting pronouncement to have a material impact on results of operations, financial conditions or its financial disclosures.

#### 3. Termination of Agreement for Sale of T-Mobile to AT&T

On March 20, 2011, Deutsche Telekom and AT&T Inc. ( AT&T ) entered into a definitive agreement (the Stock Purchase Agreement ) under which AT&T would have acquired T-Mobile from Deutsche Telekom in a cash and stock transaction valued at approximately \$39 billion, subject to adjustment in accordance with the agreement. On August 31, 2011, the U.S. Department of Justice ( DOJ ) filed a complaint in the Federal District Court of Washington, D.C. to block the acquisition. On November 24, 2011, AT&T and Deutsche Telekom withdrew their pending applications at the FCC for the transfer of T-Mobile spectrum licenses to AT&T as part of AT&T s acquisition of T-Mobile. On December 20, 2011, AT&T and Deutsche Telekom jointly announced the termination of the Stock Purchase Agreement. AT&T agreed to provide Deutsche Telekom with additional consideration ( break-up consideration ) under the terminated Stock Purchase Agreement. As a result, AT&T made a \$3.0 billion cash payment to Deutsche Telekom in December 2011. The break-up consideration also included Advanced Wireless Services ( AWS ) spectrum and a long-term agreement on UMTS roaming within the U.S. that was subsequently received in 2012 (Note 19). None of the components of the break-up consideration are reflected in T-Mobile s consolidated financial statements as of December 31, 2011.

In connection with this terminated transaction, T-Mobile incurred AT&T transaction-related costs of \$186.9 million for the year ended December 31, 2011, primarily consisting of employee-related expenses. In addition, T-Mobile is expected to be reimbursed for certain costs associated with the terminated transaction in accordance with a reimbursement agreement with AT&T. As of December 31, 2011, a receivable of \$9.5 million is included in other current assets for the specific reimbursable expenses.

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

#### 4. Equipment Installment Plan Receivables

The Company offers certain retail customers the option to pay for their handsets and other purchases in installments over a period of up to 20 months. At the time of sale, the Company imputes interest on the installment receivables and records the deferred interest as a reduction to equipment revenues and the related accounts receivable. Interest income is recognized over the financed term of the customer contract. The current portion of the Company s equipment installment plan receivables is included in accounts receivable, net and was \$159.7 million and \$157.2 million as of December 31, 2011 and 2010, respectively. The long-term portion of the equipment installment plan receivables is included in other assets and was \$95.9 million and \$29.0 million as of December 31, 2011 and 2010, respectively.

#### **Credit Quality**

The Company assesses the collectability and credit quality of the equipment installment plan receivables based upon a variety of factors, including aging of the accounts receivable portfolio, credit quality of the customer base, historical write-off experience, payment trends and other qualitative factors such as macro-economic conditions.

Based upon customer credit profiles, the Company classifies customers into the categories of Prime and Subprime . Prime customers are those with lower delinquency risk and Subprime customers are those with higher delinquency risk. Some customers within the Subprime category are required to pay an advance deposit for wireless service and equipment financed under the equipment installment plan.

The balance and aging of the equipment installment plan receivables on a gross basis by credit category for the year ended December 31, 2011 are as follows:

		Credit	Category	
(dollars in millions)	Prime	Sub	prime	Total
Unbilled	\$ 124	\$	164	\$ 288
Current	5		7	12
Past due	3		3	6
Total equipment installment plan receivables	\$ 132	\$	174	\$ 306

The Company records bad debt expense based on an estimate of the percentage of equipment revenue that will not be collected. This estimate is based on a number of factors including historical write-off experience, credit quality of the customer base, and other factors such as macro-economic conditions. The Company monitors the aging of its equipment installment plan receivables and writes-off account balances if collection efforts are unsuccessful and future collection is unlikely based on customer credit ratings and the length of time from the original billing date.

Activity in the allowance for credit losses for the equipment installment plan receivables was as follows:

(dollars in millions)	2011	2010	2009
Allowance, Beginning of Year	\$ 34	\$ 16	\$ 1
Change in deferred interest on short-term and long-term installment receivables	36	5	
Bad debt expense	35	65	24
Write-offs	(54)	(52)	(9)

\$51 \$34 \$16

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

Included in the allowance for credit losses is deferred interest of \$41.1 million and \$5.2 million for the years ended December 31, 2011 and 2010, respectively. There was no deferred interest included in the allowance for credit losses for the year ended December 31, 2009.

#### 5. Property and Equipment

(dollars in millions)	Useful Lives	2011	2010
Buildings and improvements	Up to 40 years	\$ 998	\$ 963
Wireless communications systems	3 20 years	20,068	18,919
Capitalized software	3 7 years	4,222	3,765
Furniture and equipment	3 5 years	2,007	2,015
Construction in progress		1,007	1,352
Accumulated depreciation and amortization		(15,599)	(13,801)
Property and equipment, net		\$ 12,703	\$ 13,213

Depreciation expense relating to property and equipment was \$2.9 billion, \$2.7 billion and \$2.8 billion for the years ended December 31, 2011, 2010 and 2009, respectively.

As a result of changes in useful life assumptions for property and equipment based on performance of a useful life study, the Company increased the useful lives of certain asset classes effective January 1, 2010, resulting in a reduction of depreciation of approximately \$268.0 million for the year ended December 31, 2010.

As a result of a reassessment of cell site lease renewal options reasonably assured of exercise (Note 18), the Company shortened useful life assumptions for certain long-lived assets related to cell sites effective July 1, 2010, resulting in approximately \$76.0 million additional depreciation expense in 2010.

#### 6. Goodwill, Spectrum Licenses and Other Intangible Assets Goodwill and Spectrum Licenses

The change in carrying values of goodwill and spectrum licenses for the years ended December 31, 2011 and 2010 is as follows:

(dollars in millions)	lances at ember 31, 2009	a	litions nd posals	 lances at ember 31, 2010	Additions and Disposals	Im	pairment	lances at ember 31, 2011
Goodwill (at cost)	\$ 18,446	\$	19	\$ 18,465	\$	\$		\$ 18,465
Accumulated impairment	(6,421)			(6,421)			(3,910)	(10,331)
Goodwill	\$ 12,025	\$	19	\$ 12,044	\$	\$	(3,910)	\$ 8,134

Spectrum licenses (at cost)	\$ 23,187	\$ 26	\$ 23,213	\$ 38	\$	\$ 23,251
Accumulated impairment	(7,931)		(7,931)	4	(2,510)	(10,437)
Spectrum licenses	\$ 15,256	\$ 26	\$ 15,282	\$ 42	\$ (2,510)	\$ 12,814

In connection with its annual assessment for impairment of goodwill and indefinite-lived intangible assets as of December 31, 2011, the Company recorded a non cash impairment charge of \$3.9 billion against the carrying value of goodwill and \$2.5 billion against the carrying value of its spectrum licenses. Both amounts are included in impairment charges in the Consolidated Statements of Operations and Comprehensive

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

Income (Loss). The Company also recorded a related deferred tax benefit of \$1.0 billion to reflect the impact on the respective deferred tax liability due to the reduced book to tax basis difference of goodwill and spectrum licenses. No impairments were indicated as of December 31, 2010.

#### Goodwill

In step one of the 2011 goodwill impairment test, the carrying value of the reporting unit was determined to exceed its fair value due to the Company experiencing legal and regulatory challenges against a planned sale of the reporting unit, subscriber declines and lower service revenues, which resulted in a reduction in the estimated fair value of the reporting unit as compared to the 2010 estimate of fair value.

The Company estimated the fair value of the reporting unit using an income approach, specifically based on the present value of estimated future cash flows. Future cash flows were based on the Company s estimates of revenues, operating income before interest, depreciation and amortization (OIBDA) margin and a long-term growth rate taking into consideration expected industry and market conditions. The cash flows were discounted using a weighted average cost of capital reflecting the risks associated with the business and the projected cash flows.

The estimated fair value of the reporting unit was lower than its carrying value, so the Company performed the second step of the impairment test, as described in Note 2 to the Consolidated Financial Statements, to determine the amount of the goodwill impairment.

#### Licenses

In its annual impairment test of spectrum licenses the Company estimated the fair value using the Greenfield approach, which is an income approach. Future cash flows were based on the Company s estimates and assumptions of revenues, OIBDA margin, network build-out period, and a long-term growth rate for a market participant taking into consideration expected industry and market conditions. The cash flows were discounted using a weighted average cost of capital reflecting the risks associated with the business and the projected cash flows. Due to adverse changes in the competitive landscape and regulatory environment in 2011, management changed its assumptions on which market participants would be able to transact for the asset leading to declines in the estimated cash flows used to value the spectrum licenses. At December 31, 2011, the estimated fair value of the spectrum licenses was lower than its carrying value so the Company recognized the impairment charge as described above.

#### **Other Intangible Assets**

The components of intangible assets subject to amortization are as follows:

			2011		2010					
(dollars in millions)	Useful Lives	Gross Carrying Amount		mulated rtization	Car	Net Trying Iount	Gross Carrying Amount	mulated rtization	Car	Net rrying nount
Customer list	1 6 years	\$ 209	\$	(186)	\$	23	\$ 209	\$ (138)	\$	71
Other	Up to 25 years	68		(30)		38	67	(25)		42
		\$ 277	\$	(216)	\$	61	\$ 276	\$ (163)	\$	113

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

#### **Amortization Expense**

Amortization expense for intangible assets subject to amortization was \$51.3 million, \$51.7 million and \$52.5 million for the years ended December 31, 2011, 2010 and 2009, respectively. Estimated aggregate future amortization expense for intangible assets subject to amortization is \$25.2 million in 2012, \$6.8 million in 2013, \$2.7 million in 2014, \$2.5 million in 2015, \$2.4 million in 2016 and \$20.9 million thereafter.

### 7. Investments in and Advances to Unconsolidated Affiliates

The Company s investments in and advances to unconsolidated affiliates relate primarily to an investment in Iowa Wireless Services LLC (Iowa Wireless). Iowa Wireless provides mobile wireless cellular phone communication services to rural areas of Iowa, Illinois and Missouri. T-Mobile has a 53.6% ownership interest in Iowa Wireless. However, the Company does not have a controlling interest in the entity as it does not have a majority of the voting interest, and therefore accounts for its investment in Iowa Wireless on an equity basis. The Company s share of the income and losses of Iowa Wireless is not significant to the financial statements.

Additionally, T-Mobile makes contributions to JVL Ventures LLC, a joint venture created in 2010 with AT&T Mobility and Verizon Wireless chartered with implementing a national mobile commerce payment network. The Company accounts for its 20.0% investment in this joint venture on an equity basis and records its share of the joint venture s operating losses. The Company s share of the losses is not material to the financial statements. The Company contributed \$17.5 million and \$9.8 million to the joint venture during the years ended December 31, 2011 and 2010, respectively. In accordance with the terms of the joint venture agreement, certain performance measures may be triggered in 2012 and the Company may opt to contribute additional capital ratably based on its equity ownership percentage over the operating life of the joint venture.

#### 8. Variable Interest Entity

Cook Inlet/VoiceStream GSM VII PCS Holdings LLC, (CIVS VII) is a joint venture funded by contributions from the Company and Cook Inlet Voice and Data Services, Inc. (Cook Inlet) with the majority (50.1%) of the initial capital contributions provided by Cook Inlet. CIVS VII is managed by Cook Inlet and owns spectrum licenses. The Company utilizes these spectrum licenses under certain operating agreements and compensates CIVS VII based on minutes of use. Each partner provides funding for the venture based on the initial ownership proportions until certain maximums are reached. The maximum aggregate contribution is \$80 million for Cook Inlet, which was reached in 2005. The maximum aggregate contribution for T-Mobile is \$453 million. Prior to 2010, T-Mobile s capital contribution was \$155.2 million, which represented 66% of the total capital in the joint venture. In 2010, CIVS VII made a distribution to Cook Inlet, increasing T-Mobile s ownership from 66% to 80%. The results of CIVS VII have been consolidated in the Company s financial statements, which include \$267.9 million and \$252.3 million in assets held by the joint venture as of December 31, 2011 and 2010, respectively.

In conjunction with the joint venture agreement for CIVS VII, the Company entered into an Exchange Rights Agreement with Cook Inlet in November 2004. The exchange rights granted Cook Inlet the right to exchange its ownership interest in the joint venture for Deutsche Telekom shares or cash. This original Exchange Rights Agreement expired in 2010, and an amended Exchange Rights Agreement between T-Mobile and Cook Inlet was entered into in December 2010. The amended agreement allows Cook Inlet, with advance notice, to exchange its ownership interest in the joint venture for cash equal to the sum of Cook Inlet s original contribution to the joint venture plus accrued interest. The agreement also provides

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

T-Mobile, with advance notice, the right to terminate Cook Inlet s exchange right during the first five years, or to cause Cook Inlet to exchange its joint venture interest for cash beginning on the sixth anniversary and continuing for 10 years. The updated exchange right does not meet the definition of a derivative instrument. The terms of the amended Exchange Rights Agreement entered into in 2010 are accounted for as a financing of the Company s purchase of Cook Inlet s interest in the joint venture, resulting in the derecognition of a noncontrolling interest.

As of December 31, 2011, the exchange right, which represented a liability of \$88.3 million, had not been exercised. On January 25, 2012, T-Mobile gave notice of its intent to terminate Cook Inlet s exchange rights pursuant to the amended Exchange Rights Agreement dated December 20, 2010. On February 2, 2012, Cook Inlet gave notice that it will exercise its exchange rights. As of October 15, 2012, both T-Mobile s notice of its intent to terminate the exchange rights, as well as Cook Inlet s notice that it will exercise its exchange rights, were mutually terminated pursuant to a Rescission Agreement entered into by the parties. As such, both of the parties rights were restored and are currently in existence (Note 19).

#### 9. Accrued Liabilities

Accrued liabilities are summarized as follows:

(dollars in millions)	2011	2010
Property and other taxes	\$ 482	\$ 528
Payroll and benefits	293	302
Dealer commissions	143	105
Toll and interconnect	39	19
Accrued liability for transaction-related costs	176	
Other	99	124
	\$ 1,232	\$ 1,078

#### 10. Fair Value Measurements and Derivative Instruments

The Company accounts for certain assets and liabilities at fair value. Fair value is a market-based measurement that should be determined based on assumptions that market participants would use in pricing an asset or liability. As a basis for considering such assumptions, the Company uses a three-tier value hierarchy, which prioritizes the inputs used in measuring fair value as follows:

Level 1 Observable inputs that reflect quoted prices in active markets for identical assets or liabilities;

Level 2 Inputs other than the quoted prices in active markets that are observable either directly or indirectly; and

Level 3 Unobservable inputs for which there is little or no market data, which require the Company to develop its own assumptions. T-Mobile uses observable market data, when available. Assets and liabilities of the Company measured at fair value include interest rate swaps, cross currency interest rate swaps designated as cash flow hedges, auction rate securities, and investments and obligations related to the Company s nonqualified deferred compensation plan. During the years ended December 31, 2011 and 2010, the Company did not have any transfers between Levels 1, 2 or 3 in the three-tier value hierarchy.

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

#### **Interest Rate Swaps**

The Company manages interest rate risk by entering into interest rate swaps, which are included in other current assets and other current liabilities as of December 31, 2011. The total notional amount of such interest rate swaps held by the Company at December 31, 2011 was \$2.8 billion. Interest rate swaps are valued using discounted cash flow techniques. These techniques incorporate market-based observable inputs such as interest rates and credit spreads, considering each instrument s term, notional amount, discount rate and credit risk.

The Company s interest rate swaps are classified as Level 2 in the three-tier value hierarchy. The Company recorded gains on the change in the fair value of the interest rate swaps of \$73.4 million, \$117.1 million and \$34.9 million during the years ended December 31, 2011, 2010 and 2009, respectively, which were included in interest expense.

#### **Cross Currency Interest Rate Swaps**

In 2010, the Company entered into three cross currency interest rate swaps designated as cash flow hedges to exchange the proceeds from approximately \$2.3 billion of intercompany Euro denominated notes payable to affiliates assumed on October 1, 2010 (Note 16) into U.S. dollars and to fix the future interest and principal payments in U.S. dollars, as well as mitigate the impact of foreign currency transaction gains or losses over the terms of the payables to affiliates extending to 2025. The fair value of the cross currency interest rate swaps as of December 31, 2011 was \$8.3 million included in other assets, and \$16.6 million included in other long-term liabilities. The fair value of the cross currency interest rate swaps as of December 31, 2010 was \$53.2 million included in other assets, and \$1.5 million included in other long-term liabilities. Cross currency interest rate swaps are valued using discounted cash flow techniques. These techniques incorporate market-based observable inputs such as interest rates and credit spreads, considering each instrument s term, notional amount, discount rate and credit risk.

The Company evaluates hedge effectiveness at the inception of the hedge prospectively as well as retrospectively and at the end of each reporting period, records any ineffective portion of the hedging instruments in interest expense. In the years ended December 31, 2011 and 2010, the hedges were evaluated as 100% effective, thus no gain (loss) was recognized in interest expense due to hedge ineffectiveness.

The Company s cross currency interest rate swaps are classified as Level 2 in the three-tier value hierarchy. For the year ended December 31, 2011, a \$47.0 million, net of tax, loss on the cross currency interest rate swaps was recognized in other comprehensive (loss) income. For the year ended December 31, 2010, a \$70.8 million, net of tax, loss on the cross currency interest rate swaps was recognized in other comprehensive (loss) income.

#### **Auction Rate Securities**

The Company s ARS are included in other current assets. These securities are primarily AAA-rated with remaining maturities through 2043; the underlying assets are government-backed student loans. Due to the lack of availability of observable market quotes for the Company s auction rate securities portfolio, the fair value was estimated based on a discounted cash flow valuation model and an internal analysis by management of other-than-temporary impairment factors. The discounted cash flow model considered terms, credit rating and management s assumptions regarding expected liquidation date. Management also considered other market transactions and information received from third party brokers in order to conclude that the fair value based on the discounted cash flow valuation model was reasonable. Significant inputs to the valuation are unobservable in active markets and are therefore classified as Level 3 in the three-tier

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

#### December 31, 2011, 2010 and 2009

value hierarchy. Prior to June 30, 2011, T-Mobile s investment in ARS was reported at fair value with unrealized gains and losses included as a component of other comprehensive (loss) income. At June 30, 2011, the Company recognized \$5.0 million in interest expense as an other-than-temporary loss related to a change in its ARS investment strategy to sell below the par value of the securities. Based on T-Mobile s expectations to sell its ARS in the next year, the Company reclassified these securities from long-term investments to other current assets in 2011.

#### Nonqualified Deferred Compensation Plan

Included in long-term investments and other long-term liabilities are available for sale securities and obligations, respectively, relating to the Company s nonqualified deferred compensation plan, which are valued using quoted market prices in active markets or broker-dealer quotations. The nonqualified deferred compensation plan assets and liabilities are classified as Level 1 in the three-tier value hierarchy.

The following tables provide a summary by level of the fair value of financial instruments as of December 31, 2011 and 2010 that are measured on a recurring basis:

		11		
(dollars in millions)	Level 1	Level 2	Level 3	Total
Assets				
Interest rate swaps	\$	\$ 96	\$	\$ 96
Cross currency interest rate swaps		8		8
Auction rate securities			7	7
Nonqualified deferred compensation plan	22			22
	\$ 22	\$ 104	\$7	\$ 133
Liabilities				
Cross currency interest rate swaps	\$	\$ 17	\$	\$ 17
Nonqualified deferred compensation plan	22			22
	\$ 22	\$ 17	\$	\$ 39

		20	)10	
(dollars in millions)	Level 1	Level 2	Level 3	Total
Assets				
Interest rate swaps	\$	\$ 78	\$	\$ 78
Cross currency interest rate swaps		53		53
Auction rate securities			74	74
Nonqualified deferred compensation plan	17			17
	\$17	\$ 131	\$ 74	\$ 222
Liabilities				
Interest rate swaps	\$	\$9	\$	\$9

Cross currency interest rate swaps Nonqualified deferred compensation plan	17	2	2 17
	\$ 17	\$ 11	\$ \$ 28

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

The following table summarizes the changes in fair value of the Company s Level 3 assets for the years ended December 31, 2011, 2010 and 2009:

	Aucti	on Rate Secu	rities
(dollars in millions)	2011	2010	2009
Balance, Beginning of Year	\$ 74	\$97	\$ 121
Unrealized losses included in other comprehensive (loss) income			(13)
Sales	(76)	(23)	(11)
Reversal of unrealized loss	14		
Realized loss on auction rate securities	(5)		
Balance, End of Year	\$ 7	\$ 74	\$ 97

The following table summarizes the fair values of derivatives not designated as hedging instruments in the Consolidated Balance Sheets at December 31, 2011 and 2010:

	Designated	Derivatives Not Designated as Hedging Instruments	
(dollars in millions)	2011	20	010
Assets			
Interest rate swaps	\$ 96	\$	78
	\$ 96	\$	78
Liabilities			
Interest rate swaps	\$	\$	9
	\$	\$	9

The following table summarizes the activity related to derivatives not designated as hedging instruments in the Consolidated Statements of Operations and Comprehensive Income (Loss) for the years ended December 31, 2011, 2010 and 2009:

	D	Derivatives Not Designated as Hedging Instruments		
(dollars in millions)	2011	2010	2009	
Amount of gain recognized in income				
Interest rate swaps	\$ 73	\$ 117	\$ 35	
	\$ 73	\$ 117	\$ 35	

The following table summarizes the fair values of derivatives designated as hedging instruments in the Consolidated Balance Sheets at December 31, 2011 and 2010:

	Designated a	Derivatives Designated as Hedging Instruments	
(dollars in millions)	2011	2010	
Assets			
Cross currency interest rate swaps	\$ 8	\$ 53	
	\$ 8	\$ 53	
Liabilities			
Cross currency interest rate swaps	\$ 17	\$ 2	
	\$ 17	\$ 2	

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

The following tables summarizes the activity related to derivatives designated as hedging instruments in the Consolidated Statements of Operations and Comprehensive Income (Loss) for the years ended December 31, 2011, 2010 and 2009:

	Derivatives Designated as Hedging Instruments		
(dollars in millions)	2011	2010	2009
Amount of loss recognized in other comprehensive income (loss), net of tax			
Cross currency interest rate swaps	\$ (47)	\$ (71)	\$
	\$ (47)	\$ (71)	\$

#### 11. Payables to Affiliates

Payables to affiliates consist of notes payable, net of discounts and premiums, and other payables to affiliates. T-Mobile has obtained funding from Deutsche Telekom or its affiliates to meet working capital, capital expenditure, and other obligations. Notes payable arrangements with affiliates accrue interest from the date of issuance at stated interest rates or LIBOR plus an applicable margin, with accrued interest payable semi-annually, quarterly or monthly. The applicable interest rate on certain notes payable is subject to periodic change based on changes in the credit rating of Deutsche Telekom.

On October 1, 2010, T-Mobile distributed additional paid-in capital of \$5.0 billion to Deutsche Telekom funded through the assumption of \$5.2 billion of notes payable to affiliates, net of \$0.2 billion due for cross currency interest rate swaps entered into as economic hedges in connection with the debt assumption (Notes 12 and 16). As part of the transaction, the Company assumed three notes payable to affiliates denominated in Euros that have maturities extending to 2025, which are included in the Notes payable to affiliates, thereafter line.

As of December 31, 2011, payables to affiliates were as follows:

(dollars in millions)	2011
Notes payable to affiliates, due 2012 (2.815%)	\$ 644
Notes payable to affiliates, due 2013 (1.770% 7.099%)	1,273
Notes payable to affiliates, due 2014 (2.550% 3.531%)	2,348
Notes payable to affiliates, due 2015 (2.689%)	1,905
Notes payable to affiliates, due 2016 (2.738%)	1,000
Notes payable to affiliates, thereafter (3.992% 8.195%)	7,914
	15,084
Unamortized discount and premium, net	543
Other payables to affiliates	468
Total payable to affiliates	16,095
Less: Current maturities of long-term payables to affiliates	644
Less: Other current payables to affiliates	402

Long-term payables to affiliates

\$ 15,049

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

As of December 31, 2010, payables to affiliates were as follows:

(dollars in millions)	2010
Notes payable to affiliates, due 2011 (3.552%)	\$ 500
Notes payable to affiliates, due 2012 (2.849%)	644
Notes payable to affiliates, due 2013 (1.747% 7.099%)	1,273
Notes payable to affiliates, due 2014 (2.575% 3.501%)	2,348
Notes payable to affiliates, due 2015 (2.726%)	1,905
Notes payable to affiliates, thereafter (2.698% 8.195%)	8,993
	15,663
Unamortized discount and premium, net	630
Other payables to affiliates	366
Total payable to affiliates	16,659
Less: Current maturities of long-term payables to affiliates	500
Less: Other current payables to affiliates	305
Long-term payables to affiliates	\$ 15,854

The fair value of the Company s notes payables to affiliates is estimated based on comparable market yields for similar issuances. The following table presents the carrying amounts and fair values of the Company s current and long-term notes payables to affiliates:

	20	)11	2	010
	Carrying		Carrying	
(dollars in millions)	Amount	Fair Value	Amount	Fair Value
Notes payables to affiliates	\$ 15,627	\$ 16,870	\$ 16,293	\$ 17,472

Although the Company has determined the estimated fair value amounts using available market information and commonly accepted valuation methodologies, considerable judgment is required in interpreting market data to develop fair value estimates. The fair value estimates are based on information available at December 31, 2011 and 2010. As such, the Company s estimates are not necessarily indicative of the amount that the Company could realize in a current market exchange and current estimates of fair value could differ significantly.

#### 12. Stockholder s Equity

In October 2010, T-Mobile distributed additional paid-in capital of \$5.0 billion to Deutsche Telekom in a transaction that also included the assumption of payables to affiliates (Notes 11 and 16).

13. Employee Savings and Compensation Plans Employee Retirement Savings Plan

The Company sponsors a retirement savings plan for the majority of its employees under section 401(k) of the Internal Revenue Code and similar plans. The plans allow employees to contribute a portion of their pretax income in accordance with specified guidelines. The plans match a percentage of employee contributions up to certain limits. Employer matching contributions were \$51.5 million, \$57.6 million and \$54.9 million for the years ended December 31, 2011, 2010 and 2009, respectively.

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

#### Nonqualified Deferred Compensation Plan

The Company maintains a nonqualified deferred compensation plan for certain employees. Assets related to the plan are held in a grantor trust and are subject to the claims of creditors in the event of bankruptcy. As of December 31, 2011, assets of \$22.4 million were recorded at market value and classified as long-term investments and the related \$22.4 million deferred liability was included in other long-term liabilities. As of December 31, 2010, assets of \$17.3 million were recorded at market value and classified as long-term investments and the related \$17.4 million deferred liability was included in other long-term liabilities. Realized investment gains and losses are recognized in earnings and changes in unrealized gains and losses, not considered to be other than temporary, are recorded in other comprehensive (loss) income.

#### **Executive Compensation Plan**

For the years up to 2010, T-Mobile granted performance-based cash bonus awards under its Phantom Share Plan (PSP). Under the PSP, awards could be earned, in whole or in part, based upon customer growth on a sliding scale from 60% to 150% of the original number of phantom shares granted.

The value of a phantom share appreciated or depreciated from its \$10 per share face value in proportion to the change in the Company s appraised enterprise value over the performance period. The value of an award was determined by multiplying the number of phantom share awards earned by the appraised value of a phantom share. Awards were earned and paid out ratably over a performance period of up to three years. Achievement of performance targets was determined annually and required approval by the Supervisory Board of Deutsche Telekom.

Effective January 1, 2011, T-Mobile replaced the PSP with a performance-based Long Term Incentive Plan (LTIP) tied to the Company s EBITDA and customer satisfaction targets. LTIP awards are earned over a performance period of up to three years with 50% of the target value earned on a ratable schedule and 50% of the target value earned at the end of the three year performance period.

Compensation expense reported within operating expenses related to the Company s PSP and executive compensation plan was \$51.6 million, \$39.0 million and \$16.3 million for the years ended December 31, 2011, 2010 and 2009, respectively. Payments of \$32.5 million, \$41.4 million, and \$83.9 million were made to participants related to T-Mobile s executive bonus plans during the years ended December 31, 2011, 2010 and 2009, respectively.

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

# 14. Income Taxes

The total income tax (benefit) expense is summarized as follows:

(dollars in millions)	2011	2010	2009
Current tax (benefit) expense			
Federal	\$ (15)	\$ 25	\$ 30
State and foreign	32	38	33
Total current tax expense	17	63	63
Deferred tax (benefit) expense			
Federal	(383)	729	767
State and foreign	(65)	(14)	11
Deferred tax (benefit) expense exclusive of the effect of the change in valuation allowance	(448)	715	778
Change in valuation allowance	215	45	20
Total deferred tax (benefit) expense	(233)	760	798
Other noncurrent tax contingencies		(1)	(1)
Total income tax (benefit) expense	\$ (216)	\$ 822	\$ 860

Significant components of deferred income tax assets and liabilities, tax effected, are as follows:

(dollars in millions)	2011	2010
Deferred tax assets		
Loss carryforwards	\$ 2,426	\$ 2,135
Leases	672	586
Reserves and accruals	296	270
Federal alternative minimum tax credits	171	186
Other	164	176
Total deferred tax assets	3,729	3,353
Deferred tax liabilities		
Intangible assets	(3,542)	(4,300)
Property and equipment	(2,271)	(1,589)
Other	(39)	(31)
Total deferred tax liabilities	(5,852)	(5,920)
Less: Valuation allowance	(491)	(275)
Net deferred tax liabilities	\$ (2,614)	\$ (2,842)

(dollars in millions)	2011	2010
Current deferred tax asset, net	\$ 668	\$ 914
Non-current deferred tax liability, net	(3,282)	(3,756)
Net deferred tax liability	\$ (2,614)	\$ (2,842)

As of December 31, 2011, the Company has approximately \$5.6 billion in federal net operating loss (NOL) carryforwards and \$4.1 million in capital loss carryforwards. These federal NOL carryforwards will expire between 2012 and 2030. The Company s ability to utilize NOL carryforwards in any given year may be limited by certain events, including a significant change in ownership interest. The Company s

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

capital loss carryforwards will expire in 2016. The Company s ability to utilize capital loss carryforwards is limited by generating capital gains to offset the capital loss within the carryforward period. The Company has available federal Alternative Minimum Tax credit carryforwards for tax purposes of approximately \$171.1 million which may be used indefinitely to reduce regular federal income taxes.

As of December 31, 2011, the Company has \$18.7 million in unrecognized tax benefits that, if recognized, would affect the Company s annual effective tax rate. The Company reports interest and penalties separately from income tax expense. During 2011, accrued interest and penalties associated with unrecognized tax benefits were insignificant. The Company does not anticipate any significant increases or decreases to unrecognized tax benefits during the next twelve months.

A reconciliation of the beginning and ending amount of unrecognized tax benefits is as follows:

(dollars in millions)	2011	2010	2009
Gross tax contingencies, beginning of year	\$ 19	\$ 121	\$123
Gross decreases to tax positions in prior periods		(102)	(2)
Gross increases to current period tax positions	78		
Gross tax contingencies, end of year	\$ 97	\$ 19	\$ 121

A valuation allowance is recorded when it is more likely than not that some portion or all of the deferred tax assets will not be realized. The ultimate realization of deferred tax assets depends on the ability to generate sufficient income of the appropriate character within the carryforward period available under tax law and in the appropriate taxing jurisdictions. The Company considers many factors when assessing the likelihood of future realization of deferred tax assets, including recent cumulative earnings experience by taxing jurisdiction, expectations of future income, the carryforward periods available for tax reporting purposes, and other relevant factors. As of December 31, 2011, the Company s valuation allowance relates to certain federal and state NOL carryforwards of \$11.1 million and \$442.4 million, respectively, that have been determined not to be more likely than not realizable. Furthermore, an additional \$37.0 million of valuation allowance relates to stock option deductions included in the federal NOL carryforwards which are not realizable until they offset current taxes payable.

Due to T-Mobile s NOL carryforwards, all tax years remain subject to examination by the Internal Revenue Service, as well as a significant number of state jurisdictions, although many are closed for assessment.

The reconciliation between the United States federal statutory income tax rate and T-Mobile s effective income tax rate is as follows:

	2011	2010	2009
Federal statutory income tax rate	35.00%	35.00%	35.00%
State taxes, net of federal benefit	2.56	2.21	2.77
Change in valuation allowance	(4.36)	2.07	0.86
Impairment charges	(28.27)		
Other, net	(0.54)	(1.52)	(1.71)
Effective income tax rate	4.39%	37.76%	36.92%

# 15. Earnings per Share

Basic and diluted (loss) earnings per share is computed as net (loss) income divided by the weighted-average number of common shares outstanding for the period. Previously granted stock options for Deutsche Telekom stock were granted as part of the Company s legacy stock option incentive plan and

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

therefore resulted in stock-based compensation expense to the Company. However, these awards do not affect the Company s equity structure and, therefore, do not represent potentially dilutive securities of the Company. As the Company has not issued any potentially dilutive securities, basic and diluted net (loss) income calculations are identical.

The computation of basic and diluted (loss) earnings per share ( EPS ) was as follows:

(dollars in millions, except per share amounts)	2	2011		2010		2009
Basic and Diluted EPS:						
Net (loss) income	\$	(4,718)	\$	1,357	\$	1,476
Net income attributable to noncontrolling interest				(3)		(6)
Net (loss) income attributable to T-Mobile USA	\$	(4,718)	\$	1,354	\$	1,470
Weighted average shares outstanding basic and diluted	292	,669,971	29	2,669,971	2	92,669,971
Net (loss) income per common share basic and diluted	\$	(16.12)	\$	4.63	\$	5.02

#### 16. Related Party Transactions

T-Mobile has obtained funding from Deutsche Telekom or its affiliates to meet working capital, capital expenditure, and other obligations. As of December 31, 2011, 2010 and 2009, the Company had significant payables due to Deutsche Telekom or its affiliates (Note 11). On October 1, 2010, T-Mobile distributed additional paid-in capital of \$5.0 billion to Deutsche Telekom, funded through the assumption of \$5.2 billion of notes payable to affiliates, net of \$0.2 billion due for cross currency interest rate swaps entered into as economic hedges in connection with the debt assumption (Notes 11 and 12). The Company had information technology service commitments with Deutsche Telekom or its affiliates of \$26.0 million as of December 31, 2011 (Note 18).

Transactions associated with Deutsche Telekom or its affiliates are included in various items in the accompanying Consolidated Balance Sheets, Consolidated Statements of Operations and Comprehensive Income (Loss), and Consolidated Statements of Cash Flows. The following tables summarize the transactions with Deutsche Telekom or its affiliates:

(dollars in millions)	2011	2010	2009
Related party revenues related to roaming agreements	\$ 20	\$ 22	\$ 21
Related party expenses related to roaming agreements	11	14	17
Fees incurred for use of the T-Mobile brand	52	54	13
Interest expense related to payables to affiliates	680	569	778
Net gains related to changes in fair value of interest rate swaps	73	117	35
Net loss related to changes in the fair value of cash flow hedges	47	71	

(dollars in millions)	2011	2010
Amount owed to the Company for affiliate receivables and cash management agreement	\$ 1,820	\$ 310

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

#### December 31, 2011, 2010 and 2009

The Company has intercompany interest rate swap agreements (swaps) and cross currency interest rate swaps with Deutsche Telekom or its affiliates. The cross currency interest rate swaps are designated as cash flow hedges:

	=	011 NotionaNu	-	2010 oNotionaNu		2009 ofNotional
(dollars in millions for notional amounts)	Contracts	AmountsC	ontract	sAmountsCo	ontrac	tsAmounts
Interest rate swaps entered into during the year	S	5	5	\$ 1,750	8	\$ 3,100
Interest rate swaps terminated during the year					13	3,300
Interest rate swaps matured during the year	5	2,000	2	900	3	550
Interest rate swaps outstanding at year-end	8	2,800	13	4,800	10	3,950
Cash flow hedges entered into during the year			3	2,300		
Cash flow hedges outstanding at year-end	3	2,300	3	2,300		

As of December 31, 2011, the intercompany interest rate swaps outstanding had an aggregate net fair value of \$96.4 million and the cash flow hedges have an aggregate negative net fair value of \$8.3 million. As of December 31, 2010, the intercompany interest rate swaps outstanding had an aggregate net fair value of \$68.8 million and cash flow hedges had an aggregate net fair value of \$51.7 million.

For the purposes of securing T-Mobile s obligations to provide handset insurance services, Deutsche Telekom issued a \$58.0 million letter of credit on the Company s behalf in 2010. As of December 31, 2011 and 2010, there was no balance outstanding on the letter of credit.

T-Mobile is party to network access and reciprocal wholesale agreements with certain affiliates that are controlled by Cook Inlet (Note 8). The agreements entitle each party to utilize airtime on the other party s system, and/or utilize wireless system infrastructure in certain agreed-upon markets. The agreements are structured such that each party resells airtime purchased from the other, and related fees are charged and paid between the parties.

#### 17. Supplemental Cash Flow Information

(dollars in millions)	2011	2010	2009
Interest and income tax payments			
Interest payments affiliates	\$ 831	\$ 678	\$ 833
Interest payments other	4	5	3
Income tax payments, net of refunds received	16	70	62
Noncash investing and financing activities			
Decrease in accounts payable for purchases of property and equipment	294	17	245
Relinquishment of accounts receivable from affiliates in satisfaction of notes payable to			
affiliates	500	2,615	1,492
Equity distribution in exchange for payables to affiliates (Notes 11, 12 and 16)		5,000	
Close out of interest rate swap contracts (Note 16)			104
Assumption of payables to affiliates in exchange for derivatives		166	
Rollover of current payables to affiliates		3,150	700
Acquisition of the remaining interest in a rural wireless carrier in exchange for spectrum			
licenses		32	
Noncash portion of spectrum license swap transaction	50		

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

#### 18. Commitments and Contingencies Operating Leases

T-Mobile operates cell sites, switch sites, retail stores and office facilities, which are leased with contractual terms expiring between 2012 and 2027. The majority of cell site leases have an initial term of five years, with renewal options for varying additional five-year periods. Cell site leases were originally determined to have lease terms of 20 to 25 years due to renewal options that were considered at lease inception to be reasonably assured of exercise. As of June 30, 2010, the Company determined that certain renewal options pertaining to cell site leases were no longer reasonably assured due to developments in market factors and technological changes. As a result, the Company considers that renewal options on such leases are reasonably assured of exercise only up to a total term ending closest to 15 years from inception or acquisition of the lease.

The Company continues to recognize rent expense, including the effect of fixed increases in rent, on a straight-line basis over the term estimated at inception or acquisition of the lease. Beginning in 2010, the Company assessed reasonably assured renewals for all new leases and began recording rent expense on a straight-line basis over a term ending closest to 15 years. The disclosure of the future minimum lease payments under operating leases has been modified to exclude rentals during periods covered by renewal options that are no longer considered to be reasonably assured of exercise.

T-Mobile has operating leases with local exchange carriers for dedicated transportation lines. The original terms of these commitments vary from five years up to ten years.

Future minimum payments required under dedicated transportation lines and other operating leases over their remaining terms, including reasonably assured renewals are summarized below:

(dollars in millions) Years Ending December 31,	Tran	edicated sportation Lines	OI	Other perating Leases
2012	\$	232	\$	1,559
2013		119		1,510
2014		65		1,473
2015		25		1,415
2016		2		1,250
Thereafter				5,681
	\$	443	\$	12,888

Aggregate rental expense for cell sites, switch sites, retail stores and office facilities, including accounting for lease expense on a straight line basis was \$1.7 billion for the years ended December 31, 2011 and 2010, and \$1.6 billion for the year ended December 31, 2009. Aggregate rental expense for transportation lines under operating leases was \$670.0 million, \$779.3 million and \$741.8 million for the years ended December 31, 2011, 2010 and 2009, respectively.

#### **Other Commitments**

T-Mobile has commitments with local exchange carriers for non-dedicated transportation lines. The original terms of these commitments vary from five years up to ten years. Additionally included within the table below, the Company has entered into marketing sponsorship agreements and various other commitments with a variety of suppliers primarily to purchase handsets, network services, equipment, software and other

# Table of Contents

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

items in the ordinary course of business, with various terms, through 2016. These amounts are not reflective of the Company s entire anticipated purchases under the related agreements, but are generally determined based on the non-cancelable quantities or termination amounts to which the Company is contractually obligated.

Future minimum payments required under non-dedicated transportation lines and other purchase commitments over their remaining terms are summarized below:

(dollars in millions)	Non-Dedicated Transportation Lines	Other Purchase Commitmen	
Years Ending December 31,			
2012	\$ 489	\$	398
2013	502		17
2014	495		4
2015	479		1
2016	411		1
Thereafter	457		
	\$ 2,833	\$	421

#### **Bank Line of Credit**

T-Mobile had a \$150 million unsecured credit facility with U.S. Bank National Association available for working capital and other corporate purposes. This facility was not extended and expired December 31, 2011. The credit facility had variable interest rates based on LIBOR, plus a specified margin. The credit facility contained provisions to maintain compliance with certain covenants. The Company had no borrowings outstanding as of December 31, 2011 and 2010.

#### **Capital Contribution Commitments**

T-Mobile, AT&T Mobility and Verizon Wireless have partnered to create a national mobile commerce payment network. In accordance with the terms of the joint venture agreement, certain performance measures may be triggered in 2012 and the Company may opt to contribute additional capital ratably based on its equity ownership percentage over the operating life of the joint venture.

#### **Contingencies and Litigations**

T-Mobile is party to various claims, legal actions and complaints from time to time in the ordinary course of business. While the Company cannot predict the outcome of these matters with absolute certainty, the Company believes that all such matters are adequately indemnified by third parties, are without merit or are of such kind, or involve such amounts, that unfavorable disposition would not likely have a material adverse effect on the Consolidated Balance Sheets, Consolidated Statements of Operations and Comprehensive Income (Loss) or Consolidated Statements of Cash Flows. These statements are based on the Company s current understanding of relevant facts and circumstances. As such, the Company s view of these matters is subject to inherent uncertainties and may change in the future.

19. Subsequent Events

On January 25, 2012, T-Mobile gave notice of its intent to terminate Cook Inlet s exchange rights pursuant to the amended Exchange Rights Agreement dated December 20, 2010. On February 2, 2012, Cook Inlet

T-Mobile USA, Inc.

Notes to Consolidated Financial Statements

December 31, 2011, 2010 and 2009

gave notice of its intent to exercise its exchange rights. In accordance with the Exchange Rights Agreement, on December 21, 2012, Cook Inlet will exchange its interest in the joint venture for cash.

On March 22, 2012, T-Mobile announced its intent to consolidate its call center operations from 24 to 17 facilities by the end of the second quarter of 2012.

In December 2011, pursuant to the termination of the Stock Purchase Agreement providing for the sale of T-Mobile by Deutsche Telekom to AT&T, Deutsche Telekom received break-up consideration consisting of \$3.0 billion in cash and the right to the transfer of spectrum licenses and a long-term roaming agreement within the U.S.

Subsequent events were evaluated for disclosure through April 20, 2012, the date on which the financial statements were issued.

T-Mobile USA, Inc.

Condensed Consolidated Balance Sheets as of September 30, 2012 and December 31, 2011

## (Unaudited)

(dollars in millions, except per share amounts)

(dollars in millions, except per share amounts)	Sep	tember 30, 2012	Dec	ember 31, 2011
Assets				
Current assets				
Cash and cash equivalents	\$	430	\$	390
Accounts receivable, net of allowances for uncollectible accounts of \$410 and \$347, respectively		2,636		2,697
Accounts receivable from affiliates		325		1,820
Inventory		497		455
Current portion of deferred tax assets, net		669		668
Other current assets		786		572
Total current assets		5,343		6,602
Property and equipment, net of accumulated depreciation of \$17,410 and \$15,599, respectively		12,535		12,703
Goodwill		,		8,134
Spectrum licenses		14,360		12,814
Other intangible assets, net of accumulated amortization of \$237 and \$216, respectively		43		61
Investments in and advances to unconsolidated affiliates		33		35
Long-term investments		31		22
Other assets		268		238
Total assets	\$	32,613	\$	40,609
Liabilities and Stockholder s Equity				
Current liabilities				
Accounts payable	\$	1,710	\$	1,826
Accrued liabilities		1,138		1,232
Current payables to affiliates		1,578		1,046
Deferred revenue		273		257
Other current liabilities		163		143
Total current liabilities		4,862		4,504
Long-term payables to affiliates		13,620		15,049
Deferred tax liabilities		3,543		3,282
Deferred rents		1,816		1,672
Other long-term liabilities		296		317
Total long-term liabilities		19,275		20,320
Commitments and contingencies (Note 15)				
Stockholder s equity				
Common stock, par value \$0.000001 per share, and paid-in capital; 500,000,000 shares				
authorized, 292,669,971 shares issued and outstanding		31,600		31,600
Accumulated other comprehensive loss		(9)		(28)
Accumulated deficit		(23,115)		(15,787)
Total stockholder s equity		8,476		15,785

Total liabilities and stockholder s equity	\$ 32,613	\$ 40,609

The accompanying notes are an integral part of these condensed consolidated financial statements.

T-Mobile USA, Inc.

Condensed Consolidated Statements of Operations and Comprehensive Income (Loss)

## Nine months ended September 30, 2012 and 2011

## (Unaudited)

(dollars in millions, except per share amounts)	Nine Mor Septen	nths Ende nber 30,	ed
	2012	,	2011
Revenues			
Branded contract revenues	\$ 11,105	\$	12,264
Branded prepaid revenues	1,241		957
Wholesale revenues	407		315
Roaming and other service revenues	333		380
Total service revenues	13,086		13,916
Equipment sales	1,524		1,352
Other revenues	200		171
Total revenues	14,810		15,439
Operating expenses			
Network costs, excluding depreciation and amortization	3,515		3,750
Cost of equipment sales	2,456		2,772
Customer acquisition, excluding depreciation and amortization	2,323		2,364
General and administrative, excluding depreciation and amortization	2,681		2,643
Depreciation and amortization	2,391		2,221
Impairment charges	8,134		2,221
Restructuring costs	90		
Other, net	(136)		64
Total operating expenses	21,454		13,814
Operating (loss) income	(6,644)		1,625
Other (expense) income			
Interest expense to affiliates	(487)		(481)
Interest income	53		15
Other income (expense), net	22		(11)
Total other expense, net	(412)		(477)
(Loss) income before income taxes	(7,056)		1,148
Income tax expense	(7,030) (272)		(469)
Net (loss) income	\$ (7,328)	\$	679
Other comprehensive (loss) income, net of tax			
Unrealized gain (loss) on derivatives held as cash flow hedges, net of tax of \$10 and \$48,			
respectively	17		(80)
	2		
Unrealized gain (loss) on foreign currency translation, net of tax of \$2 and \$7, respectively	Z		(12)

## Edgar Filing: MILLER PETROLEUM INC - Form 10-K

Unrealized gain on available-for-sale securities, net of tax of \$0 and \$5, respectively				9
Total comprehensive (loss) income	\$	(7,309)	\$	596
Net (loss) income per common share: Basic and diluted	\$	(25.04)	\$	2.32
Weighted average shares: Basic and diluted	292	2,669,971	292	,669,971

The accompanying notes are an integral part of these condensed consolidated financial statements.

T-Mobile USA, Inc.

**Condensed Consolidated Statements of Cash Flows** 

## Nine months ended September 30, 2012 and 2011

## (Unaudited)

(dollars in millions)		ths Ended iber 30, 2011
Operating activities		
Net (loss) income	\$ (7,328)	\$ 679
Adjustments to reconcile net (loss) income to net cash provided by operating activities		
Impairment charges	8,134	
Depreciation and amortization	2,391	2,221
Income tax expense	272	469
Amortization of debt discount and premium, net	(61)	(64)
Bad debt expense	676	483
Deferred rent expense	141	168
Other non-cash income	(225)	(40)
Changes in operating assets and liabilities		
Accounts receivable	(630)	(422)
Inventory	(42)	101
Other current and long-term assets	(127)	(111)
Accounts payable	(336)	(113)
Other current and accrued liabilities	(30)	143
Accrued liabilities related to restructuring and transaction-related costs	(128)	56
Net cash provided by operating activities	2,707	3,570
Investing activities		
Purchases of property and equipment	(2,003)	(2,178)
Expenditures related to spectrum licenses	(379)	(15)
Short term affiliate loan receivable, net	(297)	(1,100)
Proceeds from disposals of property and equipment	3	1
Proceeds (to acquire)/dispose of financial assets	(7)	45
Investments in and advances to unconsolidated affiliates, net	16	(11)
Net cash used in investing activities	(2,667)	(3,258)
Financing activities		
Net cash provided by financing activities		
Change in cash and cash equivalents	40	312
Cash and cash equivalents		
Beginning of period	390	109
End of period	\$ 430	\$ 421

The accompanying notes are an integral part of these condensed consolidated financial statements.

T-Mobile USA, Inc.

Notes to Condensed Consolidated Interim Financial Statements

(Unaudited)

#### 1. Consolidation and Basis of Presentation

The accompanying unaudited condensed consolidated interim financial statements include the balances and results of operations of T-Mobile USA, Inc. (T-Mobile , or the Company ). The financial statements of T-Mobile include the accounts of all majority-owned subsidiaries over which T-Mobile exercises control, as well as variable interest entities where T-Mobile is deemed to be the primary beneficiary. Entities over which T-Mobile exercises significant influence, but does not control and is not the primary beneficiary are accounted for using the equity method. Entities over which T-Mobile is not able to exercise significant influence are accounted for using the cost method. Certain prior period amounts have been reclassified to conform to the current period presentation. Intercompany transactions and balances have been eliminated in consolidation.

The Condensed Consolidated Balance Sheets as of September 30, 2012 and December 31, 2011, the Condensed Consolidated Statements of Operations and Comprehensive Income (Loss) and the Condensed Consolidated Statements of Cash Flows for the periods ended September 30, 2012 and 2011 should be read in conjunction with the consolidated audited financial statements and notes thereto for the three years ended December 31, 2011. In the opinion of management, the unaudited condensed consolidated interim financial statements reflect all adjustments (consisting of normal, recurring adjustments) which are necessary for a fair presentation of the Company s results for the periods presented. The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of certain assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from management s estimates and operating results for the interim periods are not necessarily indicative of the operating results for the entire fiscal year.

The Company operates in a single operating segment and a single reporting unit as a wireless communications carrier. As of September 30, 2012 and December 31, 2011, and for the nine months ended September 30, 2012 and 2011, all of the Company s revenues and long-lived assets related to operations in the United States, Puerto Rico and the U.S. Virgin Islands.

Federal Universal Service Fund (USF) and other fees, which are assessed by various governmental authorities in connection with the services that the Company provides to its customers, are reported on a gross basis in service revenues and network costs on the accompanying Condensed Consolidated Statements of Operations and Comprehensive Income (Loss). The Company recorded USF and other fees on a gross basis of \$348.9 million and \$358.6 million, respectively, for the nine months ended September 30, 2012 and 2011. Sales, use and excise taxes for all service plans are reported on a net basis in general and administrative expenses in the accompanying Condensed Consolidated Statements of Operations and Comprehensive Income (Loss).

#### 2. Recently Adopted Accounting Standards

In May 2011, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) 2011-04, Amendments to Achieve Common Fair Value Measurement and Disclosure Requirements in U.S. GAAP and IFRS. The new guidance specifies that the concepts of highest and best use and valuation premise are relevant only when measuring the fair value of nonfinancial assets, permits certain financial assets and liabilities with offsetting positions in market or counterparty credit risks to be measured at a net basis, and provides guidance on the application of premiums and discounts in a fair value measurement. Additionally, for measurements categorized within Levels 2 and 3 of the fair value hierarchy, the new guidance requires a description of the valuation technique(s) and the inputs used in the fair value measurement. If there has been a change in valuation technique, the new guidance requires disclosure of that

T-Mobile USA, Inc.

#### Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

change and the reason(s) for making it. The guidance also requires quantitative information about the significant unobservable inputs used in Level 3 fair value measurements. However, the quantitative disclosures are not required for fair value measurements related to the reporting for goodwill after its initial recognition in a business combination. The new guidance is effective for T-Mobile for the year ending December 31, 2012. The Company has adopted this recently-issued accounting pronouncement for the year ending December 31, 2012, with no material impact on results of operations, financial condition or its financial disclosures.

In September 2011, the FASB issued ASU 2011-08, Testing Goodwill for Impairment . It provides an option to first assess qualitative factors to determine whether it is necessary to perform the two-step quantitative goodwill impairment test. Given this option, an entity no longer would be required to calculate the fair value of a reporting unit unless the entity determines, based on that qualitative assessment, that it is more likely than not that its fair value is less than its carrying amount. The new guidance is effective for T-Mobile for the year ending December 31, 2012. The Company has adopted this recently-issued accounting pronouncement for the year ending December 31, 2012, with no material impact on results of operations, financial condition or its financial disclosures.

In July 2012, the FASB issued ASU 2012-02, Intangibles Goodwill and Other (Topic 350): Testing Indefinite-Lived Intangible Assets for Impairment . This ASU simplifies the guidance for testing the decline in the realizable value (impairment) of indefinite-lived intangible assets other than goodwill. It allows an entity the option of first performing a qualitative assessment to determine whether it is more likely than not that an indefinite-lived intangible asset is impaired. The amendments in this Update are effective for annual and interim impairment tests performed for fiscal years beginning after September 15, 2012. The Company has adopted these amendments for the year ending December 31, 2012, with no material impact on results of operations, financial condition or its financial disclosures.

#### 3. Proposed Transaction with MetroPCS

On October 3, 2012, Deutsche Telekom AG ( Deutsche Telekom ) and MetroPCS Communications, Inc. ( MetroPCS ) announced the signing of a business combination agreement ( business combination agreement ) to combine T-Mobile and MetroPCS. The transaction will be accounted for as a reverse acquisition under the acquisition method of accounting with T-Mobile considered to be the accounting acquirer based upon the terms of the transaction which include the ability of T-Mobile s stockholder, Deutsche Telekom, to nominate the majority of the board of directors of the combined company, and Deutsche Telekom receiving a majority of the voting rights in the combined company. Based on the determination that T-Mobile is the accounting acquirer in the transaction, T-Mobile will allocate the purchase price to the fair value of MetroPCS assets and liabilities at the acquisition date, with any excess purchase price being recorded as goodwill. The transaction is expected to close in the first half of 2013 subject to MetroPCS shareholder approval, regulatory approvals and other customary closing conditions. Pursuant to the terms and subject to the conditions set forth in the agreement, the key elements of the agreement are:

MetroPCS will effect a recapitalization that includes a reverse stock split of the MetroPCS common stock, pursuant to which each share of MetroPCS common stock outstanding as of the effective time of the reverse stock split will represent therefore one-half of a share of MetroPCS common stock and an aggregate cash payment of \$1.5 billion to the MetroPCS stockholders.

MetroPCS will acquire all of T-Mobile s common stock in exchange for issuing to Deutsche Telekom 74% of the fully diluted shares of MetroPCS common stock (calculated pursuant to the business combination agreement).

T-Mobile USA, Inc.

#### Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

T-Mobile will refinance its existing intercompany debt with Deutsche Telekom by issuing Deutsche Telekom \$15.0 billion senior unsecured notes. Deutsche Telekom will provide the combined company with a \$0.5 billion unsecured revolving credit facility and purchase up to \$5.5 billion in additional senior unsecured notes under certain circumstances.

Since the announcement on October 3, 2012 of the execution of the business combination agreement, MetroPCS, Deutsche Telekom (and certain of its subsidiaries), T-Mobile and the members of the MetroPCS board, including an officer of MetroPCS, have been named as defendants in multiple stockholder derivative and class action complaints challenging the transaction. The lawsuits generally allege, among other things, that the transaction fails to properly value MetroPCS and that the individual defendants breached their fiduciary duties in approving the business combination agreement and, in some of the lawsuits, that those breaches were aided and abetted by Deutsche Telekom (and certain of its subsidiaries), and T-Mobile. The lawsuits seek, among other things, injunctive relief enjoining the defendants from completing the transaction on the agreed-upon terms, monetary relief and attorneys fees and costs. While citing a number of specific issues, the Company believes these cases have no merit.

#### 4. Equipment Installment Plan Receivables

The Company offers certain retail customers the option to pay for their handsets and other purchases in installments over a period of up to 20 months. At the time of sale, the Company imputes interest on the installment receivables and records the deferred interest as a reduction to equipment revenues and the related accounts receivable. Interest income is recognized over the financed term of the customer contract. The current portion of the Company s equipment installment plan receivables is included in accounts receivable, net and was \$343.7 million and \$159.7 million as of September 30, 2012 and December 31, 2011, respectively. The long-term portion of the equipment installment plan receivables is included in other assets and was \$141.6 million and \$95.9 million as of September 30, 2012 and December 31, 2011, respectively.

#### **Credit Quality**

The Company assesses the collectability and credit quality of the equipment installment plan receivables based upon a variety of factors, including aging of the accounts receivable portfolio, credit quality of the customer base, historical write off experience, payment trends, and other qualitative factors such as macro-economic conditions.

Based upon customer credit profiles, the Company classifies customers into the categories of Prime and Subprime . Prime customers are those with lower delinquency risk and Subprime customers are those with higher delinquency risk. Some customers within the Subprime category are required to pay an advance deposit for wireless service and equipment financed under the equipment installment plan.

There were no material movements between the credit categories as a result of new information that has arisen during the nine month period ended September 30, 2012.

#### T-Mobile USA, Inc.

## Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

The balance and aging of the equipment installment plan receivables on a gross basis by credit category are as follows:

		Credit Categories as of September 30, 2012					
(dollars in millions)	Prime	Subprime	Total				
Unbilled	\$ 229	\$ 319	\$ 548				
Current	8	14	22				
Past Due	3	7	10				
Total equipment installment plan receivables	\$ 240	\$ 340	\$ 580				

Total equipment installment plan receivables

		edit Categories as December 31, 2011	
(dollars in millions)	Prime	Subprime	Total
Unbilled	\$ 124	\$ 164	\$ 288
Current	5	7	12
Past Due	3	3	6
Total equipment installment plan receivables	\$ 132	\$ 174	\$ 306

The Company records bad debt expense based on an estimate of the percentage of equipment revenue that will not be collected. This estimate is based on a number of factors including historical write-off experience, credit quality of the customer base, and other factors such as macro-economic conditions. The Company monitors the aging of its equipment installment plan receivables and writes off account balances if collection efforts are unsuccessful and future collection is unlikely based on customer credit profile and the length of time from the original billing date.

Activity in the allowance for credit losses for the equipment installment plan receivables was as follows:

(dollars in millions)	nber 30, 012	nber 31, )11
Allowance, Beginning of Period	\$ 51	\$ 34
Change in deferred interest on short-term and long-term installment		
receivables	30	36
Bad debt expense	65	35
Write-offs	(52)	(54)
Allowance, End of Period	\$ 94	\$ 51

Included in the allowance for credit losses is deferred interest of \$71.4 million and \$41.1 million for the nine month period ended September 30, 2012 and the year ended December 31, 2011, respectively.

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

#### T-Mobile USA, Inc.

#### Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

#### 5. Goodwill and Spectrum Licenses

The change in carrying values of goodwill and spectrum licenses during the nine months ended September 30, 2012 was as follows:

(dollars in millions)	Dece	ances at mber 31, 2011	tions and sposals	Imj	pairment	llances at tember 30, 2012
Goodwill (at cost)	\$	18,465	\$	\$		\$ 18,465
Accumulated impairment		(10,331)			(8,134)	(18,465)
	\$	8,134	\$	\$	(8,134)	\$
Spectrum licenses (at cost)	\$	23,251	\$ 1,511	\$		\$ 24,762
Accumulated impairment		(10,437)	35			(10,402)
	\$	12,814	\$ 1,546	\$		\$ 14,360

The Company assesses the carrying value of its goodwill and other indefinite-lived intangible assets (spectrum licenses) for potential impairment annually as of December 31 or more frequently if events or changes in circumstances indicate that such assets might be impaired. On October 3, 2012, Deutsche Telekom and MetroPCS announced the business combination agreement, which provides for the combination of T-Mobile and MetroPCS. Under the terms of the business combination agreement, Deutsche Telekom will receive 74% of the fully-diluted shares of common stock of the combined company (calculated pursuant to the business combination agreement) in exchange for its contribution of all of T-Mobile s common stock (Note 3).

The Company determined that the announced transaction was a triggering event for a goodwill impairment assessment. The fair value of T-Mobile implied by using the market value of MetroPCS and the exchange terms contemplated in the business combination agreement was less than the carrying amount, including goodwill, of the Company s single reporting unit as of September 30, 2012. The Company used the fair value implied by the transaction to estimate the fair value of the reporting unit in step one of its goodwill impairment test as it incorporates observable inputs that are considered as Level 2 in the fair value hierarchy. In determining the fair value of the reporting unit, the Company also considered alternative valuation methodologies, including valuation based upon earnings-multiples and discounted cash flow. Historically, the Company estimated the fair value of the reporting unit using a discounted cash flow approach when there was a lack of comparable observable market data. Because the first step in the impairment process indicated that the carrying value exceeded the fair value of the reporting unit as of September 30, 2012, the Company performed the second step in the goodwill impairment test.

As a result of the second step impairment analysis, the Company concluded that the implied goodwill was \$0, and recognized a non-cash impairment charge of \$8.1 billion for the nine months ended September 30, 2012. The Company attributes this impairment to the business impacts from the highly competitive environment in the U.S. wireless telecommunications industry and the ongoing challenges in attracting and retaining branded contract customers. No impairment charge was recognized for the nine months ended September 30, 2011.

The Company has determined that its spectrum licenses should be treated as indefinite-lived intangible assets. Spectrum licenses are carried at costs incurred to acquire spectrum licenses and the costs, if any, to prepare the licenses for use. The Federal Communications Commission (FCC) issues spectrum licenses that authorize wireless carriers to provide service in specific geographic service areas for up to fifteen years. The spectrum licenses provide the Company with the exclusive right to utilize designated radio

T-Mobile USA, Inc.

## Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

frequency spectrum to provide cellular communication services. The FCC has specific standards applying to wireless spectrum license renewals which provide that it will award a spectrum license renewal to a licensee that has met certain standards based on past performance. Historically, the FCC has granted license renewals routinely and at a nominal cost. The spectrum licenses held by the Company expire at various dates; however, the Company believes it will be able to meet all requirements necessary to secure renewal of its spectrum licenses. Moreover, the Company has determined that there are currently no legal, regulatory, contractual, competitive, economic or other factors that limit the useful lives of its spectrum licenses and therefore treats the spectrum licenses as indefinite-lived intangible assets.

During the nine months ended September 30, 2012, the carrying value of T-Mobile s spectrum licenses increased as a result of spectrum acquisition activities. T-Mobile acquired spectrum licenses valued at \$1.2 billion from Deutsche Telekom, which Deutsche Telekom had received from AT&T Inc. ( AT&T ) as a component of the break-up consideration related to the failed acquisition of T-Mobile by AT&T in 2011. In addition, T-Mobile completed a transaction to purchase from and exchange with Verizon Communications Inc. ( Verizon ) certain Advanced Wireless Services ( AWS ) spectrum licenses. As a result of the transaction, the Company received AWS spectrum licenses and transferred certain of its AWS spectrum licenses along with a cash payment to Verizon. A gain of \$155.9 million was recognized as a result of this transaction and is recorded in other, net. There were no indicators of impairment of the Company spectrum licenses during the nine months ended September 30, 2012 or 2011.

## 6. Investments in and Advances to Unconsolidated Affiliates

The Company s investments in and advances to unconsolidated affiliates relate primarily to an investment in Iowa Wireless Services LLC (Iowa Wireless). T-Mobile plans to sell network assets and spectrum licenses to Iowa Wireless in a transaction expected to be completed in the fourth quarter of 2012 (Note 16). As a result of this expected transaction, spectrum licenses and property equipment valued at \$15.5 million were transferred to held-for-sale during the nine months ended September 30, 2012.

During the nine months ended September 30, 2012 and 2011, the Company contributed \$21.6 million and \$10.9 million, respectively, to JVL Ventures LLC, a joint venture between T-Mobile, AT&T and Verizon. The Company s share of the losses is not material to the financial statements. In accordance with the terms of the joint venture agreement, certain performance measures may be triggered in 2012 and the Company may opt to contribute additional capital ratably based on its equity ownership percentage over the operating life of the joint venture.

## 7. Variable Interest Entity

Cook Inlet/VoiceStream GSM VII PCS Holdings LLC, ( CIVS VII ) is a joint venture funded by contributions from the Company and Cook Inlet Voice and Data Services, Inc. ( Cook Inlet ) with the majority (50.1%) of the initial capital contributions provided by Cook Inlet. CIVS VII is managed by Cook Inlet and owns spectrum licenses. The Company utilizes these spectrum licenses under certain operating agreements and compensates CIVS VII based on minutes of use. T-Mobile is required to consolidate the results of CIVS VII because T-Mobile is considered to be the primary beneficiary and substantially all of the joint venture s activities are carried out on behalf of the Company. The results of CIVS VII have been consolidated in the Company s financial statements, which include \$267.8 million and \$267.9 million in assets held by the joint venture as of September 30, 2012 and December 31, 2011, respectively.

In conjunction with the joint venture agreement for CIVS VII, the Company has an exchange rights agreement with Cook Inlet which as amended in 2010 allows Cook Inlet, with advance notice, to exchange its ownership interest in the joint venture for cash equal to the sum of Cook Inlet s original contribution to

T-Mobile USA, Inc.

## Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

the joint venture plus accrued interest. The amended agreement provides T-Mobile, with advance notice, the right to cause Cook Inlet to exchange its joint venture interest for cash. The exchange right does not meet the definition of a derivative instrument.

During the first quarter of 2012, T-Mobile gave notice of its intent to terminate Cook Inlet s exchange rights pursuant to the amended exchange rights agreement dated December 20, 2010 and Cook Inlet gave notice that they would exercise their interest in the joint venture for cash. However, subsequent to period-end, both T-Mobile and Cook Inlet entered into a rescission agreement to mutually terminate their respective notices (Note 16).

## 8. Fair Value Measurements and Derivative Instruments

The Company accounts for certain assets and liabilities at fair value. Fair value is a market-based measurement that should be determined based on assumptions that market participants would use in pricing an asset or liability. As a basis for considering such assumptions, the Company uses a three-tier value hierarchy, which prioritizes the inputs used in measuring fair value as follows:

Level 1 Observable inputs that reflect quoted prices in active markets for identical assets or liabilities;

Level 2 Inputs other than the quoted prices in active markets that are observable either directly or indirectly; and

Level 3 Unobservable inputs for which there is little or no market data, which require the Company to develop its own assumptions. T-Mobile uses observable market data, when available. Assets and liabilities of the Company measured at fair value include interest rate swaps, cross currency interest rate swaps designated as cash flow hedges, auction rate securities, and investments and obligations related to the Company s nonqualified deferred compensation plan. During the periods ended September 30, 2012 and 2011, the Company did not have any transfers between Levels 1, 2 or 3 in the three-tier value hierarchy.

T-Mobile does not enter into derivatives for trading or speculative purposes. The Company has entered into derivative agreements to manage exposure to fluctuations in interest rates, through interest rate swap agreements, and to manage foreign currency risk along with interest rate risk, through cross currency interest rate swap agreements. The interest rates swap agreements are not designated as hedging instruments and, accordingly, changes in fair value related to such agreements are reflected within interest expense. The cross currency interest rate swap agreements manage foreign currency risk and interest rate risk related to the Company s payables to affiliates denominated in Euros. The cross currency interest rate swaps are designated as cash flow hedges and meet the criteria to qualify for hedge accounting. At the inception of the cash flow hedges and quarterly thereafter, the Company performs an assessment to determine whether changes in the fair values or cash flows of the derivatives are deemed highly effective in offsetting changes in the fair values or cash flows of the hedged transaction. If at any time subsequent to the inception of the cash flow hedges, the assessment indicates that the derivative is no longer highly effective as a hedge, the Company will discontinue hedge accounting and recognize all subsequent derivative gains and losses in results of operations. The effective portion of the change in fair value of the cross currency interest rate swaps is recorded in other comprehensive (loss) income and reclassified to interest expense in the period in which the hedged transaction affects earnings. The ineffective portion of the change in fair value of the cross currency interest rate swaps is recorded in other comprehensive (loss) income and reclassified to interest expense in the period in which the hedged transaction affects earnings. The ineffective portion of the change in fair value of the cross currency interest rate swaps is recorded in other comprehensive (loss) income and reclassifi

T-Mobile USA, Inc.

#### Notes to Condensed Consolidated Interim Financial Statements

(Unaudited)

#### **Interest Rate Swaps**

The Company manages interest rate risk by entering into interest rate swaps, which are included in other current assets and other current liabilities as of September 30, 2012 and December 31, 2011. The total notional amount of such interest rate swaps held by the Company as of September 30, 2012 and December 31, 2011 was \$4.1 billion and \$2.8 billion, respectively. Interest rate swaps are valued using discounted cash flow techniques. These techniques incorporate market-based observable inputs such as interest rates and credit spreads, considering each instrument s term, notional amount, discount rate and credit risk.

The Company s interest rate swaps are classified as Level 2 in the three-tier value hierarchy. During the nine months ended September 30, 2012 and 2011, gains on the change in the fair value of these interest rate swaps of \$74.1 million and \$74.8 million, respectively, were reported as a reduction to and included in interest expense.

#### **Cross Currency Interest Rate Swaps**

The fair value of the Company s cross currency interest rate swaps as of September 30, 2012 was \$5.0 million included in other assets and \$1.9 million included in other long-term liabilities. The fair value of the cross currency interest rate swaps as of December 31, 2011 was \$8.3 million included in other assets and \$16.6 million included in other long-term liabilities. Cross currency interest rate swaps are valued using discounted cash flow techniques. These techniques incorporate market-based observable inputs such as interest rates and credit spreads, considering each instrument s term, notional amount, discount rate and credit risk. The Company s cross currency interest rate swaps are classified as Level 2 in the three-tier value hierarchy.

The Company evaluates hedge effectiveness at the inception of the hedge prospectively as well as retrospectively and, at the end of each reporting period, records any ineffective portion of the hedging instruments in expense. During the nine months ended September 30, 2012 and 2011, the hedges were evaluated as 100% effective, thus no gain (loss) was recognized in interest expense due to hedge ineffectiveness.

#### Nonqualified Deferred Compensation Plan

Included in long-term investments and other long-term liabilities are available for sale securities and obligations, respectively, relating to the Company s nonqualified deferred compensation plan, which are valued using quoted market prices in active markets or broker-dealer quotations. The nonqualified deferred compensation plan assets and liabilities are classified as Level 1 in the three-tier value hierarchy.

#### T-Mobile USA, Inc.

## Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

The following tables provide a summary by level of the fair value of financial instruments as of September 30, 2012 and December 31, 2011 that are measured on a recurring basis:

(dollars in millions)	Level 1	At Septeml Level 2	ber 30, 2012 Level 3	Total
Assets				
Interest rate swaps	\$	\$ 137	\$	\$137
Cross currency interest rate swaps		5		5
Auction rate securities			7	7
Nonqualified deferred compensation plan	31			31
	\$ 31	\$ 142	\$7	\$ 180
Liabilities				
Cross currency interest rate swaps	\$	\$ 2	\$	\$ 2
Nonqualified deferred compensation plan	31			31
	\$ 31	\$ 2	\$	\$ 33

(dollars in millions)	At December 31, 2011 Level Level			
	1	Level 2	3	Total
Assets				
Interest rate swaps	\$	\$ 96	\$	\$ 96
Cross currency interest rate swaps		8		8
Auction rate securities			7	7
Nonqualified deferred compensation plan	22			22
	\$ 22	\$ 104	\$7	\$ 133
Liabilities				
Cross currency interest rate swaps	\$	\$ 17	\$	\$ 17
Nonqualified deferred compensation plan	22			22
	\$ 22	\$ 17	\$	\$ 39

The following table summarizes the fair values of derivatives not designated as hedging instruments in as of September 30, 2012 and December 31, 2011:

(dollars in millions)

Derivatives Not Designated as Hedging Instruments

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

	September 30, 2012	1ber 31, )11
Assets		
Interest rate swaps	\$ 137	\$ 96
	\$ 137	\$ 96
Liabilities		
Interest rate swaps	\$	\$
	\$	\$

#### T-Mobile USA, Inc.

## Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

The following table summarizes the activity related to derivatives not designated as hedging instruments for nine month periods ended September 30, 2012 and 2011:

(dollars in millions)	Derivatives Not Hedging In Nine Mont Septeml	struments hs Ended per 30,
	2012	2011
Amount of gain recognized in income		
Interest rate swaps	\$ 74	\$ 75
	\$ 74	\$ 75

The following table summarizes the fair values of derivatives designated as hedging instruments as of September 30, 2012 and December 31, 2011:

(dollars in millions)	Derivatives Designated a Hedging Instruments		
	September 30, 2012		1ber 31, 011
Assets			
Cross currency interest rate swaps	\$ 5	\$	8
	\$ 5	\$	8
Liabilities			
Cross currency interest rate swaps	\$ 2	\$	17
	\$ 2	\$	17

The following table summarizes the activity related to derivatives designated as hedging instruments for the nine month periods ended September 30, 2012 and 2011:

(dollars in millions)	Derivatives Designated Hedging Instrument Nine Months Ended September 30,	
	2012	2011
Amount of income (loss) recognized in other comprehensive (loss) income, net of tax		
Cross currency interest rate swaps	\$ 17	\$ (80)

\$ 17	\$ (80)

T-Mobile USA, Inc.

#### Notes to Condensed Consolidated Interim Financial Statements

(Unaudited)

#### 9. Payables to Affiliates

Payables to affiliates consist of notes payable, net of discounts and premiums, and other payables to affiliates due as of September 30, 2012 for each of the maturity periods, as follows:

(dollars in millions)	Carrying Amount
Notes payable to affiliates, due 2013 (1.772% 7.099%)	\$ 1,273
Notes payable to affiliates, due 2014 (2.721% 3.526%)	2,348
Notes payable to affiliates, due 2015 (2.877%)	1,905
Notes payable to affiliates, due 2016 (2.744%)	1,000
Notes payable to affiliates, due 2017	
Notes payable to affiliates, thereafter (3.701% 8.195%)	7,910
	14,436
Unamortized discount and premium, net	481
Other payables to affiliates	281
Total payable to affiliates	15,198
Less: Current maturities of long-term payables to affiliates	1,273
Less: Other current payables to affiliates	305
Long-term payables to affiliates	\$ 13,620

Payables to affiliates consist of notes payable, net of discounts and premiums, and other payables to affiliates due as of December 31, 2011 for each of the maturity periods, as follows:

(dollars in millions)	Carrying Amount
Notes payable to affiliates, due 2012 (2.815%)	\$ 644
Notes payable to affiliates, due 2013 (1.770% 7.099%)	1,273
Notes payable to affiliates, due 2014 (2.550% 3.531%)	2,348
Notes payable to affiliates, due 2015 (2.689%)	1,905
Notes payable to affiliates, due 2016 (2.738%)	1,000
Notes payable to affiliates, thereafter (3.992% 8.195%)	7,914
	15,084
Unamortized discount and premium, net	543
Other payables to affiliates	468
Total payable to affiliates	16,095
Less: Current maturities of long-term payables to affiliates	644
Less: Other current payables to affiliates	402

## Long-term payables to affiliates

\$ 15,049

The fair value of the Company s notes payable to affiliates is estimated based on comparable market yields for similar debt. The following table presents carrying amounts and fair values of the current and long-term notes payables to affiliates:

(dollars in millions)	September 30, 2012 December 31		er 31, 2011	
	Carrying		Carrying	
	Amount	Fair Value	Amount	Fair Value
Notes payable to affiliates	\$ 14,917	\$ 14,889	\$ 15,627	\$ 16,870

T-Mobile USA, Inc.

#### Notes to Condensed Consolidated Interim Financial Statements

(Unaudited)

#### 10. Income Taxes

The effective tax rate was (3.85)% and 40.82% for the nine months ended September 30, 2012 and 2011, respectively. For the nine months ended September 30, 2012, the Company s effective tax rate differs from the statutory federal rate of 35.0% primarily due to permanent book and tax differences resulting from the goodwill impairment (Note 5). For the nine months ended September 30, 2011, our effective tax rate differs from the statutory federal rate of 35.0% primarily due to net state and foreign taxes, and changes in valuation allowance.

Income tax expense was \$272 million and \$469 million for the nine months ended September 30, 2012 and 2011, respectively. The decrease in income tax expense for the nine months ended September 30, 2012 compared to the same period in 2011 was primarily due to lower book income after permanent items.

#### 11. Earnings per share

Basic and diluted (loss) earnings per share is computed as net (loss) income divided by the weighted-average number of common shares outstanding for the period. As the Company has not issued any potentially dilutive securities, basic and diluted net (loss) income calculations are identical.

The computation of basic and diluted (loss) earnings per share ( EPS ) was as follows:

(dollars in millions, except per share amounts)		Nine Months Ended September 30,			
	2012	2011			
Basic and Diluted EPS:					
Net (loss) income	\$ (7,328)	\$ 679			
Weighted average shares outstanding basic and diluted	292,669,971	292,669,971			
Net (loss) income per common share basic and diluted	\$ (25.04)	\$ 2.32			

#### 12. Related Party Transactions

T-Mobile has obtained funding from Deutsche Telekom or its affiliates to meet working capital, capital expenditure, and other obligations. As of each of September 30, 2012 and December 31, 2011, the Company had significant payables due to Deutsche Telekom or its affiliates (Note 9).

Transactions associated with Deutsche Telekom or its affiliates are included in various items in the accompanying Condensed Consolidated Balance Sheets, Condensed Consolidated Statements of Operations and Comprehensive Income (Loss), and Condensed Consolidated Statements of Cash Flows. The following tables summarize the transactions with Deutsche Telekom or its affiliates:

(dollars in millions)

Nine Months Ended September 30, 2012 2011

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

Related party revenues related to roaming agreements	\$ 12	\$ 17
Related party expenses related to roaming agreements	5	9
Fees incurred for use of T-Mobile brand	37	39
Interest expense related to payables to affiliates	484	490
Net gains related to changes in the fair value of the interest rate swaps	74	75
Net gain (loss) related to changes in the fair value of the cash flow hedges	17	(80)

T-Mobile USA, Inc.

#### Notes to Condensed Consolidated Interim Financial Statements

#### (Unaudited)

(dollars in millions)		Sept	ember 30, 2012		mber 31, 2011
Amount owed to the Company for affiliate receivables and cash n	nanagement				
agreement		\$	325	\$	1,820

The Company has intercompany interest rate swap agreements (swaps) and cross currency interest rate swaps, which are designated as cash flow hedges, with Deutsche Telekom or its affiliates (Note 8). There were no cash flow hedges entered into, terminated, or matured during the nine month periods ended September 30, 2012 and 2011. The following table summarizes the activity related to the swaps for the nine month period ended September 30, 2012 and year ended December 31, 2011:

	2	2012		2011
	Number of	Notional	Number of	Notional
(dollars in millions for notional amount)	contracts	Amounts	contracts	Amounts
Interest rate swaps entered into during the period	3	\$ 2,300		\$
Interest rate swaps matured into during the period	3	\$ 1,050	5	\$ 2,000
Interest rate swaps outstanding at period-end	8	\$ 4,050	8	\$ 2,800

T-Mobile is party to network access and reciprocal wholesale agreements with certain affiliates that are controlled by Cook Inlet (Note 7). The agreements entitle each party to utilize airtime on the other party s system, and/or utilize wireless system infrastructure in certain agreed-upon markets. The agreements are structured such that each party resells airtime purchased from the other, and related fees are charged and paid between the parties.

#### 13. Restructuring Costs

During the nine months ended September 30, 2012, T-Mobile consolidated its call center operations from 24 to 17 facilities. In addition, T-Mobile restructured and optimized operations in other parts of the business to strengthen T-Mobile s competiveness in the U.S. marketplace. Major costs incurred primarily related to lease buyout costs, severance payments and other personnel-related restructuring costs. In regards to these restructuring activities, the Company recognized total costs of approximately \$90 million during the nine months ended September 30, 2012 as restructuring costs in the Consolidated Statements of Operations and Comprehensive (Loss) Income. Substantially all costs related to this consolidation effort have been incurred through this date. As of September 30, 2012, \$42 million of costs incurred were included in accrued liabilities.

Activity associated with restructuring costs in accrued liabilities was as follows:

(dollars in millions)	2012
Balance as of December 31, 2011	\$
Personnel related restructuring costs	55
Non-personnel related restructuring costs	35
Cash payments	(48)
Balance as of September 30, 2012	\$ 42

T-Mobile USA, Inc.

## Notes to Condensed Consolidated Interim Financial Statements

(Unaudited)

## 14. Supplemental Cash Flow Information

(dollars in millions)		Nine Months Ended September 30,	
	2012	2011	
Interest and income tax payments			
Interest payments affiliates	\$ 630	\$ 631	
Interest payments other	14	3	
Income tax payments, net of refunds received	24	15	
Noncash investing and financing activities			
(Increase)/decrease in accounts payable for purchases of property and equipment	(233)	306	
Relinquishment of accounts receivable from affiliates in satisfaction of notes payable to affiliates	644	500	
Noncash portion of spectrum license swap transactions	289		
Spectrum licenses received in exchange for accounts receivable from affiliates	1,163		

#### 15. Commitments and Contingencies Commitments

During the third quarter of 2012, the Company and Crown Castle International Corp. (CCI) entered into definitive agreements pursuant to which CCI acquired rights to approximately 7,200 wireless communication towers owned by T-Mobile for approximately \$2.4 billion in cash at closing, which occurred in the fourth quarter of 2012 (see Subsequent Events Note 16). Under the agreements, CCI acquired the sole right to lease and operate these T-Mobile towers through either purchase of the towers, or under long-term prepaid lease and management agreements. Furthermore, CCI was granted an option to acquire any sites that were not sold to CCI at the end of their respective least terms. T-Mobile will lease back and continue to have rights to space on the towers to operate the Company s wireless communications equipment. The transaction will be treated as a financing transaction with the proceeds reported as a financial liability at the time of close.

During the nine months ended September 30, 2012, T-Mobile signed various agreements with suppliers under which it has incurred purchase commitments totaling \$5.1 billion relating to LTE-capable network equipment and handsets.

## Contingencies

T-Mobile is party to various claims, legal actions and complaints from time to time in the ordinary course of business. While the Company cannot predict the outcome of these matters with absolute certainty, the Company believes that all such matters are adequately indemnified by third parties, are without merit or are of such kind, or involve such amounts, that unfavorable disposition would not likely have a material adverse effect on the Condensed Consolidated Balance Sheets, Condensed Consolidated Statements of Operations and Comprehensive (Loss) Income or Condensed Consolidated Statements of Cash Flows. These statements are based on the Company s current understanding of relevant facts and circumstances. As such, the Company s view of these matters is subject to inherent uncertainties and may change in the future.

T-Mobile has been named as a defendant in multiple shareholder lawsuits related to its proposed business combination agreement with MetroPCS (Note 3).

T-Mobile USA, Inc.

Notes to Condensed Consolidated Interim Financial Statements

(Unaudited)

## 16. Subsequent Events

On October 3, 2012, Deutsche Telekom and MetroPCS announced the signing of the business combination agreement, which provides for the combination of T-Mobile and MetroPCS (Note 3).

As of October 15, 2012, both T-Mobile s notice of its intent to terminate the exchange rights and Cook Inlet s notice to exercise its exchange rights were mutually terminated pursuant to a rescission agreement entered into by the parties. As such, both of the parties rights were restored and are currently in existence (Note 7).

T-Mobile plans to sell network assets and spectrum licenses to Iowa Wireless in a transaction expected to be completed in the fourth quarter of 2012 (Note 6). No loss is expected on the sale.

During the third quarter of 2012, the Company and CCI entered into definitive agreements pursuant to which CCI will acquire rights to approximately 7,200 owned by T-Mobile (see Commitments and Contingencies (Note 15)). Prior to closing on November 30, 2012, the tower portfolio included in the transaction was reduced to 7,105 sites. During the fourth quarter of 2012, T-Mobile received net proceeds of \$2,469 million of which the Company distributed \$2,403 million to its parent, Deutsche Telekom.

Subsequent events were evaluated for disclosure through December 20, 2012.

Annex A

EXECUTION COPY

## **BUSINESS COMBINATION AGREEMENT**

by and among

## DEUTSCHE TELEKOM AG

## T-MOBILE GLOBAL ZWISCHENHOLDING GMBH

## T-MOBILE GLOBAL HOLDING GMBH

## **T-MOBILE USA, INC.**

and

## METROPCS COMMUNICATIONS, INC.

Dated as of October 3, 2012

## TABLE OF CONTENTS

ARTICLE I DEFINITIONS AND TERMS		Page A-2
1.1	Definitions	A-2
1.2	Other Definitional Provisions	A-15
ARTICI	LE II RECAPITALIZATION; STOCK PURCHASE; CLOSING	A-16
2.1	The Recapitalization	A-16
2.2	Stock Purchase	A-21
2.3	Mergers	A-21
2.4	Adjustment	A-22
2.5	Closing	A-24
<u>ARTICI</u>	LE III REPRESENTATIONS AND WARRANTIES	A-24
3.1	Representations and Warranties Regarding DT, Global and Holding	A-24
3.2	Representations and Warranties Regarding TMUS and its Subsidiaries	A-26
3.3	Representations and Warranties of MetroPCS	A-40
<u>ARTICI</u>	LE IV COVENANTS	A-56
4.1	Interim Operations of TMUS	A-56
4.2	Interim Operations of MetroPCS	A-60
4.3	Proxy Statement	A-64
4.4	MetroPCS Stockholders Meeting	A-65
4.5	No Solicitation MetroPCS	A-65
4.6	No Solicitation DT	A-68
4.7	Access: Post-Signing Deliverables	A-69
4.8	Publicity	A-70
4.9	Expenses	A-71
4.10	Resignations	A-71
4.11	Filings; Other Actions; Notification.	A-71
4.12	Financial Working Group	A-72
4.13	Financing	A-72
4.14	Regulatory Compliance	A-74
4.15	Further Action	A-74
4.16	Intercompany Arrangements	A-74
4.17	Customer Communications	A-75
4.18	Employee Matters	A-75
4.19	MetroPCS Directors and Officers; Name	A-76
4.20	Transition Arrangements	A-76
4.21	Intellectual Property	A-77
4.22	Confidentiality	A-77
4.23	Indemnification: Release	A-77
4.24	MetroPCS Common Stock	A-78
4.25	Sale of Towers	A-78
4.26	Notification of Certain Matters	A-78
4.27	Litigation	A-79
4.28	Anti-Takeover Statutes	A-79
4.29	Control of Operations	A-79
4.30	Listing of TMUS Stock Consideration	A-79

## TABLE OF CONTENTS

#### (continued)

		Page
ARTICLE V CONDITIONS		A-79
5.1	Conditions to Each Party s Obligation to Effect the Transaction	A-79
5.2	Conditions to Obligations of MetroPCS	A-80
5.3	Conditions to Obligations of DT, Holding and TMUS	A-81
<u>ARTICL</u>	LE VI TERMINATION	A-82
6.1	Termination	A-82
6.2	Effect of Termination and Abandonment	A-83
ARTICLE VII MISCELLANEOUS AND GENERAL		A-85
7.1	<u>Survival</u>	A-85
7.2	Amendment; Waivers, Etc	A-85
7.3	Counterparts	A-86
7.4	Governing Law; Jurisdiction; Forum; Waiver of Trial by Jury	A-86
7.5	Notices	A-86
7.6	Entire Agreement	A-87
7.7	Specific Performance	A-87
7.8	No Third-Party Beneficiaries	A-87
7.9	Severability	A-87
7.10	Interpretation	A-88
7.11	Assignment	A-88
7.12	Limitation of Liability	A-88
7.13	Securities Matters	A-88
7.14	<u>Transfer Taxes</u>	A-89
7.15	Reliance of Other Parties	A-90
7.16	Effect of Breaches	A-90
Exhibits		

- Exhibit A New MetroPCS Certificate
- Exhibit B New MetroPCS Bylaws
- Exhibit C Stockholder s Agreement
- Exhibit D Trademark License
- Exhibit E Voting and Support Agreement
- Exhibit F DT Note Pricing Schedule
- Exhibit G Description of DT Notes
- Exhibit H TMUS Working Capital Facility
- Exhibit I DT Financing Backstop Commitment
- Exhibit J Terms of Noteholder Agreement Exhibit K Directors and Executive Officers
  - mon K Directors and Executive Officers

## **BUSINESS COMBINATION AGREEMENT**

BUSINESS COMBINATION AGREEMENT (this <u>Agreement</u>), dated as of October 3, 2012, by and among DEUTSCHE TELEKOM AG, an *Aktiengesellschaft* organized and existing under the laws of the Federal Republic of Germany (<u>DT</u>), T-MOBILE GLOBAL ZWISCHENHOLDING GMBH, a *Gesellschaft mit beschränkter Haftung* organized and existing under the laws of the Federal Republic of Germany (<u>Global</u>), T-MOBILE GLOBAL HOLDING GMBH, a *Gesellschaft mit beschränkter Haftung* organized and existing under the laws of the Federal Republic of Germany (<u>Holding</u>), T-MOBILE USA, INC., a Delaware corporation (<u>MetroPCS</u>).

#### RECITALS

WHEREAS, DT owns all of the issued and outstanding shares of capital stock of Global, which owns all of the issued and outstanding shares of capital stock of Holding, which owns all of the issued and outstanding shares of capital stock of TMUS;

WHEREAS, MetroPCS desires to effect a recapitalization and reverse split of its common stock, par value \$0.0001 per share (the <u>MetroPCS</u> <u>Common Stock</u>), as described herein;

WHEREAS, in order to effect such recapitalization and the MetroPCS Reverse Stock Split, and to give effect to the other provisions herein and therein, MetroPCS desires to amend and restate its certificate of incorporation in the form attached as Exhibit A (the <u>New MetroPCS Certificate</u>) and its bylaws in the form attached as Exhibit B (the <u>New MetroPCS Bylaws</u>);

WHEREAS, as part of, and effective upon, such recapitalization and the MetroPCS Reverse Stock Split, MetroPCS desires to pay to its stockholders the MetroPCS Cash Amount, upon the terms and subject to the conditions set forth herein;

WHEREAS, upon, and subject to, such recapitalization, the MetroPCS Reverse Stock Split and the Cash Payment, Holding desires to sell to MetroPCS, and MetroPCS desires to purchase from Holding, the TMUS Shares in exchange for the TMUS Stock Consideration, upon the terms and subject to the conditions set forth herein;

WHEREAS, the MetroPCS Board has (a) approved the execution, delivery and performance of this Agreement, (b) determined that the New MetroPCS Certificate is advisable in connection with the Transaction in accordance with the provisions of the DGCL, and (c) resolved to recommend the approval of the New MetroPCS Certificate and the MetroPCS Share Issuance by the MetroPCS Stockholders, upon the terms and subject to the conditions set forth in this Agreement and in accordance with the provisions of the DGCL and the rules and regulations of the NYSE, as applicable;

WHEREAS, the execution, delivery and performance of this Agreement has been authorized by all requisite action of the board of directors or similar governing body of DT, Global, Holding and TMUS;

WHEREAS, concurrently with the Closing, and as a condition and inducement to the parties willingness to enter into this Agreement, (a) DT and MetroPCS shall enter into a Stockholder s Agreement, dated as of the Closing Date, substantially in the form attached as Exhibit C (the <u>Stockholder s Agreement</u>), and (b) DT and MetroPCS shall enter into a Trademark License Agreement, dated as of the Closing Date, substantially in the form attached as Exhibit D (the <u>Trademark License</u> and, together with the Stockholder s Agreement<u>, the An</u>cillary <u>Agreements</u>); and

WHEREAS, concurrently herewith, as a condition and inducement to the parties willingness to enter into this Agreement, Madison Dearborn Capital Partners IV, L.P. is entering into a voting and support agreement in the form attached as Exhibit E.

NOW, THEREFORE, in consideration of the premises, and of the representations, warranties, covenants and agreements contained herein, and other good and valuable consideration the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

#### ARTICLE I

#### **DEFINITIONS AND TERMS**

1.1 Definitions. As used in this Agreement the following terms shall have the following respective meanings:

Actual Adjustment Amount shall have the meaning set forth in Section 2.4(g).

Additional DT Notes shall have the meaning set forth in Section 4.13(c).

Adjusted MetroPCS Stock Option shall have the meaning set forth in Section 2.1(d)(i).

Adjusted Per-Share Option Exercise Price shall have the meaning set forth in Section 2.1(d)(i).

<u>Adjustment Amount</u> shall mean the sum of (i) the excess, if any, of \$1,300,000,000 over the TMUS Working Capital as of 12:01 a.m., prevailing Eastern Time, on the Closing Date, *plus* (ii) the Spending Deficiency Amount.

<u>Affiliate</u> shall mean, with respect to any Person, a Person that directly or indirectly through one or more intermediaries, Controls, is Controlled by, or is under common Control with such Person.

Agreement shall have the meaning set forth in the Preamble.

Ancillary Agreements shall have the meaning set forth in the Recitals.

<u>Beneficially Own</u> shall mean, with respect to any securities, (i) having beneficial ownership of such securities for purposes of Rule 13d-3 or 13d-5 under the Exchange Act (or any successor statute or regulation), (ii) having the right to become the Beneficial Owner of such securities (whether such right is exercisable immediately or only after the passage of time or the occurrence of conditions) pursuant to any agreement, arrangement or understanding, or upon the exercise of conversion rights, exchange rights, rights, warrants or options, or otherwise, or (iii) having an exercise or conversion privilege or a settlement payment or mechanism with respect to any option, warrant, convertible security, stock appreciation, swap agreement or other security, contract right or derivative position, whether or not currently exercisable, at a price related to the value of the securities for which Beneficial Ownership is being determined or a value determined that increases in value as the value of the securities for which Beneficial Ownership is being determined increases or that provides to the holder an opportunity, directly or indirectly, to profit or share in any profit derived from any increase in the value of the securities for which Beneficial Ownership is being determined (excluding any interests, rights, options or other securities set forth in Rule 16a-1(c)(1)-(5) or (7) promulgated pursuant to the Exchange Act).

<u>Business</u> shall mean the business of marketing, selling, offering, promoting or providing wireless telecommunications and wireless information products and services, and all products and services ancillary or related thereto, including products and services offered as of the date hereof by MetroPCS and its Subsidiaries, or TMUS and its Subsidiaries, as applicable, in the Territory.

<u>Business Day</u> shall mean any day other than a Saturday, a Sunday, a federal holiday or a day on which banks in the City of New York or in Bonn, Germany are authorized or obligated by Law to close.

<u>Cash Payment</u> shall have the meaning set forth in Section 2.1(c).

<u>CDM</u>A shall have the meaning set forth in Section 3.3(p)(ii)(I).

<u>Chosen Courts</u> shall have the meaning set forth in Section 7.4.

<u>Circumstance</u> shall mean any event, occurrence, fact, condition, effect, change or development.

<u>Closing</u> shall have the meaning set forth in Section 2.5.

<u>Closing Date</u> shall have the meaning set forth in Section 2.5.

<u>Code</u> shall have the meaning set forth in Section 3.2(g)(ii).

<u>Communications Act</u> shall have the meaning set forth in Section 3.1(c)(i).

<u>Confidentiality Agreement</u> shall mean, collectively, (i) the confidentiality agreement, dated January 31, 2012, between MetroPCS and TMUS, (ii) the Clean Team Confidentiality Agreement, dated April 19, 2012, between MetroPCS and TMUS, and (iii) the Common Interest Agreement, dated April 19, 2012, between MetroPCS and TMUS, in each case as amended, amended and restated, supplemented or modified from time to time.

<u>Control</u> shall mean the possession, direct or indirect, of the power to direct, or cause the direction of, the management and policies of a Person, whether through the ownership of voting securities, voting equity, limited liability company interests, general partner interests, or voting interests, by contract or otherwise.

<u>D&O Tail Policy</u> shall have the meaning set forth in Section 4.23(b).

<u>Damages</u> shall mean any and all demands, claims, actions or causes of action, assessments, losses, damages, liabilities, diminution in value, costs and expenses, including interest, penalties and reasonable attorneys fees and expenses, in each case on a basis net of any actual benefit received.

<u>D</u>T shall have the meaning set forth in the Preamble.

<u>DT Notes</u> shall have the meaning set forth in Section 4.13(b).

DT Termination Amount shall have the meaning set forth in Section 6.2(d).

<u>DGC</u>L shall mean the General Corporation Law of the State of Delaware.

<u>Disputed Item</u> shall have the meaning set forth in Section 2.4(d).

<u>Effective Time</u> shall have the meaning set forth in Section 2.1(a).

<u>Encumbrance</u> (including, with correlative meaning, the ter<u>m</u> Encumber ) shall mean any lien, pledge, charge, claim, encumbrance, hypothecation, security interest, option, lease, license, mortgage, easement or other restriction or third-party right of any kind, including any right of first refusal, tag-along or drag-along rights or restriction on voting, transferring, lending, disposing or assigning, in each case other than pursuant to the Stockholder s Agreement.

<u>Environmental La</u>w shall mean any applicable Law relating to (i) the protection of the environment (including air, water, soil and natural resources) or (ii) the use, storage, handling, release or disposal of any Hazardous Substance or waste, in each case as presently in effect.

<u>Equity Interests</u> shall mean (i) any capital stock of a corporation, any partnership interest, any limited liability company interest or any other equity interest; (ii) any security or right convertible into, exchangeable for, or evidencing the right to subscribe for, any such stock, equity interest or security referred to in clause (i); (iii) any stock appreciation right, contingent value right or similar security or right that is derivative of any such stock, equity interest or security referred to in clause (i) or (ii); and (iv) any contract to grant, issue, award, convey or sell any of the foregoing.

<u>ERISA</u> shall have the meaning set forth in Section 3.2(g)(i).

<u>ERISA Affiliate</u> means any entity that would be considered a single employer with TMUS under Section 4001(b) of ERISA or a member of a group of entities which includes TMUS for purposes of Section 414(b), (c), (m) or (o) of the Code.

Estimated Adjustment Amount shall have the meaning set forth in Section 2.4(a).

Estimated TMUS Closing Statement shall have the meaning set forth in Section 2.4(a).

Exchange Act shall mean the Securities Exchange Act of 1934, as amended.

<u>Excluded Liabilities</u> shall mean all liabilities of DT and its Affiliates, other than liabilities of TMUS and its Subsidiaries to the extent related to the business operated by TMUS and its Subsidiaries on or prior to the Effective Time.

<u>Executory Period</u> shall have the meaning set forth in the definition of Spending Deficiency Amount.

<u>FAA</u> shall have the meaning set forth in Section 3.2(h)(ii).

FAA Rules shall have the meaning set forth in Section 3.2(h)(v).

<u>FC</u>C shall have the meaning set forth in Section 3.1(c)(i).

FCC Licenses shall mean the TMUS FCC Licenses and the MetroPCS FCC Licenses, as applicable.

FCC Rules shall have the meaning set forth in Section 4.14(a).

<u>Final Order</u> shall mean any action or decision of a Governmental Entity (i) that has not been vacated, reversed, set aside, annulled or suspended, (ii) as to which no request for a stay or similar request is pending, no stay is in effect, and any deadline for filing such request that may be designated by statute or regulation has passed without the filing of any such request, (iii) as to which no timely petition for rehearing or reconsideration, application for review, or other protest is pending before such Governmental Entity and the time for the filing of any such petition, application or protest designated by statute, regulation or otherwise has passed, (iv) that is not under reconsideration or review on such Governmental Entity s own motion and the time within which it may effect such reconsideration or review designated by statute, regulation or otherwise has passed, and (v) that is not then under administrative or judicial review and as to which there is no notice of appeal or other application for administrative or judicial review pending or in effect, and any deadline for filing any such appeal or other application for administrative or judicial review pending or nucleas passed, unless, in the cases of clauses (ii) through (v), the parties mutually agree in writing that such request, stay, petition, application, protest, reconsideration, review, and/or appeal is not reasonably likely to result in vacating, reversing, setting aside, annulling or suspending such action or decision, or in modifying such action or decision in a manner that would reasonably be expected to have or result in a Regulatory Material Adverse Condition.

<u>Financing Sources</u> means the entities that commit to provide or otherwise enter into agreements, commitments, undertakings, contracts or arrangements in connection with or relating to the MetroPCS Finance Transactions or other financings in connection with or relating to the Transaction, including any lenders, noteholders, agents, collateral agents, arrangers, trustees or similar parties.

<u>FMA</u> shall have the meaning set forth in Section 4.1(r).

<u>GAAP</u> shall mean United States generally accepted accounting principles.

<u>Glob</u>al shall have the meaning set forth in the Preamble.

<u>Governmental Consents</u> shall mean all notices, reports and other filings required to be made prior to the Closing by DT or MetroPCS or any of their respective Subsidiaries with, and all consents, registrations, approvals, permits, clearances, licenses, certificates, waivers and authorizations required to be obtained prior to the Closing by DT or MetroPCS or any of their respective Subsidiaries from, any Governmental Entity in connection with the execution and delivery of this Agreement and the consummation of the Transaction.

<u>Governmental Entity</u> shall have the meaning set forth in Section 3.1(c)(i).

<u>Hazardous Substance</u> shall mean any substance that is (i) listed, classified or regulated pursuant to any Environmental Law; (ii) any petroleum product or by-product, asbestos-containing material, lead-containing paint or plumbing, polychlorinated biphenyls, mold, radioactive material or radon; and (iii) any other substance which may be the subject of regulatory action by any Governmental Entity in connection with any Environmental Law.

<u>Hedge Agreement</u> shall mean any agreement or arrangement with respect to any swap, cap, collar, forward, future or derivative transaction or option or similar agreement, whether exchange traded, over-the-counter or otherwise, involving, or settled by reference to, one or more rates, currencies, commodities, equity or debt instruments or securities, or economic, financial or pricing indices or measures of economic, financial or pricing risk or value or any similar transaction or any combination of these transactions.

Holding shall have the meaning set forth in the Preamble.

<u>HSR Act</u> shall have the meaning set forth in Section 3.1(c)(i).

In-the-Money MetroPCS Stock Option shall have the meaning set forth in Section 2.1(d)(iii).

<u>Indebtedness</u> shall mean (i) all liabilities for borrowed money, whether current or funded, secured or unsecured, all obligations evidenced by bonds, debentures, notes or similar instruments, and all liabilities in respect of mandatorily redeemable or purchasable capital stock or securities convertible into capital stock; (ii) all liabilities for the principal amount of the deferred and unpaid purchase price of real property and equipment that have been delivered; (iii) all liabilities in respect of any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which liabilities are required to be classified and accounted for under GAAP as capital leases; (iv) all liabilities in respect of Hedge Agreements; (v) all liabilities for the reimbursement of any obligor on any letter of credit, banker s acceptance or similar credit transaction securing obligations of any other Person of a type described in clauses (i), (ii), (iii) or (iv) above to the extent of the obligation secured; and (vi) all liabilities as guarantor of obligations of any other Person of a type described in clauses (i), (ii), (iii), (iii), (iv) or (v) above, to the extent of the obligation guaranteed.

<u>Independent Accountant</u> shall mean a certified public accountant satisfactory to MetroPCS and D<u>T</u>; provided, that if MetroPCS and DT do not appoint an Independent Accountant within 10 days after either MetroPCS or DT gives notice to the other of a request therefor, either of them may request the American

Arbitration Association to appoint as the Independent Accountant a partner in the New York office of a nationally recognized independent registered public accounting firm based on its determination that the partner has had no material relationships with the parties or their respective Affiliates within the preceding two years and taking into account such firm s material relationships during the preceding two years with the parties and their respective Affiliates, and such appointment shall be final, binding and conclusive on MetroPCS and DT.

Insurance Policies shall have the meaning set forth in Section 3.2(j).

Insured Parties shall have the meaning set forth in Section 4.23(b).

<u>Intellectual Property</u> shall mean all rights in intellectual property of any type throughout the world, including the following: (i) all trademarks, service marks, brand names, product names and slogans, certification marks, collective marks, d/b/a s, assumed names, Internet domain names, logos, symbols, trade dress, trade names and any and every other form of trade identity and other indicia of origin, all applications and registrations therefor and renewals thereof and all goodwill associated therewith and symbolized thereby (the items listed in this clause (i) collectively, <u>Trademarks</u>); (ii) all inventions and discoveries, whether or not reduced to practice, patents, including utility patents and design patents, industrial designs and utility models, invention disclosures, all applications and registrations for the foregoing, including reissues, divisionals, continuations, continuations-in-part, supplementary protection certificates, extensions, reexaminations, renewals thereof, and any counterparts (foreign or otherwise) claiming priority therefrom which priority may be claimed, and all inventions disclosed therein and improvements thereto; (iii) proprietary and confidential information, trade secrets and know-how, including processes, schematics, business methods, formulae, drawings, research and development, prototypes, models, designs, customer lists and supplier lists, all other confidential or proprietary technical, business and other information and all rights in any jurisdiction to limit the use or disclosure thereof (the items listed in this clause (iii) collectively, <u>Trade Secrets</u>); (iv) published and unpublished works of authorship (including databases and other compilations of information, mask works and Software), works for hire, the copyrights therein and thereto and all registrations and applications therefor and renewals, extensions, restorations and reversions thereof; and (v) all other intellectual property, industrial or similar proprietary rights recognized under any jurisdiction wor

<u>Intercompany Contracts</u> shall mean all TMUS Contracts between TMUS or one or more of its Subsidiaries, on the one hand, and DT or one or more of its Subsidiaries (other than TMUS and its Subsidiaries), on the other hand, other than the DT Notes.

<u>Intercompany Indebtedness</u> shall mean any Indebtedness that would be owed by TMUS or one of its Subsidiaries to DT or one of its Subsidiaries (other than TMUS and its Subsidiaries) or by DT or one of its Subsidiaries (other than TMUS and its Subsidiaries) to TMUS or one of its Subsidiaries, other than the DT Notes, the Additional DT Notes or any indebtedness issued pursuant to Section 4.25.

<u>Intervening Event</u> shall mean a Circumstance material to MetroPCS and its Subsidiaries, taken as a whole, that did not occur, arise or become known to the MetroPCS Board or was not reasonably foreseeable by the MetroPCS Board, in each case prior to the date hereof (or if known or reasonably foreseeable prior to the date hereof, the material consequences of which were not known or reasonably foreseeable prior to the date hereof), which Circumstance, or any material consequence thereof, becomes known to the MetroPCS Board prior to the receipt of the MetroPCS Stockholder Approval; <u>provided</u>, <u>however</u>, that the receipt, existence or terms of a MetroPCS Acquisition Proposal shall not constitute an Intervening Event.

<u>IRS</u> shall have the meaning set forth in Section 3.2(g)(ii).

<u>IT Assets</u> shall mean computers, Software, firmware, middleware, servers, workstations, routers, hubs, switches, data communications lines and all other information technology equipment, and all documentation associated therewith.

Knowledge of MetroPCS shall mean the actual knowledge of the Persons listed on Schedule 1.1(a) of the MetroPCS Disclosure Letter.

Knowledge of TMUS shall mean the actual knowledge of the Persons listed on Schedule 1.1(a) of the TMUS Disclosure Letter.

Laws shall have the meaning set forth in Section 3.2(h)(i).

<u>Leased Real Property</u> shall mean all real property leased or subleased by TMUS and its Subsidiaries or by MetroPCS and its Subsidiaries, as applicable.

<u>Liabilities</u> means any and all debts, Indebtedness, liabilities, commitments and obligations of any kind, whether fixed, contingent or absolute, matured or unmatured, liquidated or unliquidated, accrued or not accrued, asserted or not asserted, known or unknown, determined, determinable or otherwise, whenever or however arising (including, whether arising out of any contract or tort based on negligence or strict liability) and whether or not the same would be required by GAAP to be reflected in financial statements or disclosed in the notes thereto.

<u>Licensed MHz POPs</u> shall mean, with respect to any FCC License, (i) the population of each geographic area covered by such FCC License based on the 2010 United States census, <u>multiplied</u> by (ii) the aggregate MHz of spectrum authorized by such FCC License in such area.

Licenses shall have the meaning set forth in Section 3.2(h)(i).

Low Exercise Price MetroPCS Stock Option shall have the meaning set forth in Section 2.1(d)(iii).

<u>Material MetroPCS Contracts</u> shall have the meaning set forth in Section 3.3(p)(i).

<u>Material TMUS Contracts</u> shall have the meaning set forth in Section 3.2(o)(i).

<u>MetroPCS</u> shall have the meaning set forth in the Preamble.

<u>MetroPCS Acquisition Proposal</u> shall have the meaning set forth in Section 4.5(j).

MetroPCS Adverse Recommendation Change shall have the meaning set forth in Section 4.5(d).

<u>MetroPCS Benefit Plans</u> shall have the meaning set forth in Section 3.3(h)(i).

MetroPCS Board shall mean the board of directors of MetroPCS.

MetroPCS Book-Entry Shares shall have the meaning set forth in Section 2.1(f)(iii).

<u>MetroPCS Business Plan</u> shall mean MetroPCS s 2012 business plan approved by the MetroPCS Board prior to the date hereof and MetroPCS s 2013 long range planning model, a copy of each of which is attached as Schedule 1.1(b) of the MetroPCS Disclosure Letter.

<u>MetroPCS Cash Amount</u> shall have the meaning set forth in Section 2.1(c).

<u>MetroPCS Cash Deposit</u> shall have the meaning set forth in Section 2.1(e).

<u>MetroPCS Certificate</u> shall have the meaning set forth in Section 2.1(f).

<u>MetroPCS Closing Price</u> means the average, rounded to the nearest one ten thousandth, of the closing price of a share of MetroPCS Common Stock on the NYSE for the five full NYSE trading days immediately preceding the Closing Date, without giving effect to any adjustment for the MetroPCS Reverse Stock Split or the Cash Payment, whether through the operation of the NYSE s ex-dividend procedures or otherwise.

<u>MetroPCS Common Stock</u> shall have the meaning set forth in the Recitals.

MetroPCS Communications Licenses shall have the meaning set forth in Section 3.3(i)(ii).

<u>MetroPCS Consent Offers</u> shall mean any consent solicitations or similar transactions to secure the waiver of the holders of a majority in principal amount of each series of the MetroPCS Existing Notes to any Change of Control resulting from the Transaction or the transactions related thereto.

<u>MetroPCS Contract</u> shall mean any agreement, lease, license, contract, note, mortgage, credit agreement, security agreement, indenture, arrangement, commitment, undertaking or other obligation, whether written or oral, binding upon MetroPCS or any of its Subsidiaries.

MetroPCS Disclosure Letter shall have the meaning set forth in Section 3.3.

<u>MetroPCS Employees</u> shall have the meaning set forth in Section 4.18(a).

<u>MetroPCS Exchange Ratio</u> shall have the meaning set forth in Section 2.1(a).

<u>MetroPCS Existing Credit Agreement</u> means the Third Amended and Restated Credit Agreement, dated as of March 17, 2011, among MetroPCS OpCo, as Borrower, the Lenders from time to time parties thereto, and JPMorgan Chase Bank, N.A., as Administrative Agent, as modified by the Incremental Commitment Agreement, dated as of May 10, 2011, among MetroPCS OpCo, as Borrower, the Guarantors (as defined therein), the financial institutions parties thereto and JPMorgan Chase Bank, N.A., as Administrative Agent, as further amended, amended and restated, supplemented or modified from time to time.

<u>MetroPCS Existing Finance Documents</u> means, collectively, (i) the Indenture, dated September 21, 2010, among MetroPCS OpCo, the Guarantors (as defined therein) and Wells Fargo Bank, N.A., as trustee, the First Supplemental Indenture, dated September 21, 2010, among MetroPCS OpCo, the Guarantors and Wells Fargo Bank, N.A., as trustee, and the Third Supplemental Indenture, dated December 23, 2010, among MetroPCS OpCo, the Guarantors and Wells Fargo Bank, N.A., as trustee, and the Notes (as defined therein), (ii) the Indenture, dated September 21, 2010, among MetroPCS OpCo, the Guarantors (as defined therein) and Wells Fargo Bank, N.A., as trustee, and the Notes (as defined therein), (ii) the Indenture, dated September 21, 2010, among MetroPCS OpCo, the Guarantors (as defined therein) and Wells Fargo Bank, N.A., as trustee, and the Fourth Supplemental Indenture, dated December 23, 2010, among MetroPCS OpCo, the Guarantors and Wells Fargo Bank, N.A., as trustee, and the Fourth Supplemental Indenture, dated December 23, 2010, among MetroPCS OpCo, the Guarantors and Wells Fargo Bank, N.A., as trustee, and the Fourth Supplemental Indenture, dated December 23, 2010, among MetroPCS OpCo, the Guarantors and Wells Fargo Bank, N.A., as trustee, and the Notes (as defined therein), and (iii) the MetroPCS Existing Credit Agreement, together with the Loan Documents (as defined therein), in each case as amended, amended and restated, supplemented or modified from time to time.

<u>MetroPCS Existing Notes</u> means, collectively, MetroPCS OpCo s (i) 7 7/8% Senior Notes due 2018 and (ii) 6 5/8% Senior Notes due 2020.

<u>MetroPCS FCC Licenses</u> shall have the meaning set forth in Section 3.3(i)(ii).

<u>MetroPCS Finance Transactions</u> shall mean, collectively, (i) the issuance of the Permitted MetroPCS Notes, if any, (ii) the MetroPCS Consent Offers, if any, and (iii) other financing transactions, including hedging transactions, reasonably related to the foregoing as DT and MetroPCS may agree.

MetroPCS Financial Statements shall have the meaning set forth in Section 3.3(f)(ii).

MetroPCS HoldCo shall mean MetroPCS, Inc.

MetroPCS Material Adverse Effect shall mean (i) an effect that would prevent or materially delay the ability of MetroPCS to consummate the Transaction, or (ii) a material adverse effect on the financial condition, properties, assets, liabilities, business or results of operations of MetroPCS and its Subsidiaries, taken as a whole; provided, however, with respect to this clause (ii), none of the following shall be deemed to be or constitute a MetroPCS Material Adverse Effect, or be taken into account when determining whether a MetroPCS Material Adverse Effect has occurred or would occur: (A) any Circumstance generally affecting (x) the Territory or global economy or Territory or global financial, debt, credit, capital or securities markets or (y) the wireless telecommunications and wireless information products and services industry in the Territory; (B) any Circumstance resulting from any declared or undeclared acts of war, terrorism, outbreaks or escalations of hostilities, sabotage or civil strife or threats thereof: (C) any act of God or weather-related Circumstance; (D) any Circumstance resulting from any change in (x) GAAP or (y) applicable Laws or regulatory or enforcement developments (in the cases of clauses (A), (B), (C) and (D)(y), except to the extent such Circumstance disproportionately affects MetroPCS and its Subsidiaries, taken as a whole, relative to other companies in the wireless telecommunications and wireless information services industry in the Territory, and, in the case of clause (D)(x), except to the extent such Circumstance disproportionately affects MetroPCS and its Subsidiaries, taken as a whole, relative to the prepaid operations of other companies in the wireless telecommunications and wireless information services industry in the Territory); (E) any Circumstance resulting from any failure by MetroPCS or its Subsidiaries to meet any estimates, projections, budgets or forecasts of revenues or earnings for any period ending on or after the date hereof, or any rumors, predictions or reports of such failure; provided, that the exception in this clause (E) shall not prevent or otherwise affect a determination that any Circumstance underlying such failure has resulted in or contributed to a MetroPCS Material Adverse Effect; (F) any Circumstance resulting from the announcement, pendency or public disclosure of this Agreement and the Transaction; (G) any Circumstance resulting from any action required to be taken or omitted to be taken pursuant to this Agreement; or (H) any Circumstance resulting from any decline in the price or trading volume of, MetroPCS Common Stock on the NYSE; provided, that the exception in this clause (H) shall not prevent or otherwise affect a determination that any Circumstance underlying such decline has resulted in or contributed to a MetroPCS Material Adverse Effect. Any determination of MetroPCS Material Adverse Effect shall exclude the effects of the matters disclosed in the MetroPCS Disclosure Letter or the matters specifically identified in the notes to the MetroPCS Financial Statements.

<u>MetroPCS Material Licenses</u> shall have the meaning set forth in Section 3.3(i)(i).

<u>MetroPCS Merger</u> shall have the meaning set forth in Section 2.3(a).

<u>MetroPCS OpC</u>o shall mean MetroPCS Wireless, Inc..

<u>MetroPCS Owned Intellectual Property</u> shall have the meaning set forth in Section 3.3(o)(i).

MetroPCS Per-Share Cash Amount shall have the meaning set forth in Section 2.1(c).

<u>MetroPCS Permitted Encumbrances</u> shall mean (i) Encumbrances specifically reflected or specifically reserved against or otherwise disclosed in the MetroPCS Financial Statements or the MetroPCS Disclosure Letter; (ii) mechanics, materialmen s, warehousemen s, carriers, workers or repairmen s liens or other common law or statutory Encumbrances arising or incurred in the ordinary course of MetroPCS s business consistent with past practice and that are not material in amount or effect on the business of MetroPCS and its Subsidiaries, taken as a whole; (iii) liens for Taxes, assessments and other governmental charges not yet due and payable or due but not delinquent or being contested in good faith by appropriate proceedings and for which adequate reserves have been established, if and to the extent required by GAAP, in the most recent MetroPCS

Financial Statements; (iv) with respect to real property, (A) easements, quasi-easements, licenses, covenants, rights-of-way, rights of re-entry or other similar restrictions, including any other agreements, conditions or restrictions that would be shown by a current title report or other similar report or listing, in each case that do not or would not materially impair the conduct of business of MetroPCS and its Subsidiaries, taken as a whole, or the use or value of the relevant asset, (B) any conditions that may be shown by a current survey or physical inspection, in each case that do not or would not materially impair the conduct of business of MetroPCS and its Subsidiaries, taken as a whole, or the use or value of the relevant asset, (B) any conditions that may be shown by a current survey or physical inspection, in each case that do not or would not materially impair the conduct of business of MetroPCS and its Subsidiaries, taken as a whole, or the use or value of the relevant asset, and (C) zoning, building, subdivision or other similar requirements or restrictions, in each case that do not or would not materially impair the conduct of business of MetroPCS and its Subsidiaries, taken as a whole, or the use or value of the relevant asset; and (v) Encumbrances granted by or required under the MetroPCS Existing Finance Documents or any Hedge Agreements to which MetroPCS or any of its Subsidiaries is a party and which have been provided to DT prior to the date hereof.

<u>MetroPCS Preferred Stock</u> shall have the meaning set forth in Section 3.3(b)(i).

MetroPCS Qualified Bidder shall have the meaning set forth in Section 4.5(c).

<u>MetroPCS Recommendation</u> shall have the meaning set forth in Section 4.4(a).

MetroPCS Restricted Stock shall mean MetroPCS Common Stock issued, but not vested, under the MetroPCS Benefit Plans.

MetroPCS Reverse Stock Split shall have the meaning set forth in Section 2.1(a).

<u>MetroPCS Rights Agreement</u> shall mean the Rights Agreement, dated as of March 29, 2007, between MetroPCS and American Stock Transfer & Trust Company, as Rights Agent.

<u>MetroPCS SEC Reports</u> shall mean such reports, schedules, forms, statements and other documents required to be filed by MetroPCS under the Exchange Act or any successor statute, and the rules and regulations promulgated thereunder, including pursuant to Section 13(a) or 15(d) thereof, since December 31, 2009 (including the exhibits thereto and documents incorporated by reference therein).

<u>MetroPCS Share Issuance</u> shall have the meaning set forth in Section 2.2(b).

<u>MetroPCS State Licenses</u> shall have the meaning set forth in Section 3.3(i)(ii).

<u>MetroPCS Stock Option</u> shall have the meaning set forth in Section 2.1(d)(i).

<u>MetroPCS Stock Plans</u> shall have the meaning set forth in Section 2.1(d)(iv).

MetroPCS Stockholder Approval shall have the meaning set forth in Section 3.3(d).

<u>MetroPCS Stockholders</u> shall mean the holders of MetroPCS Common Stock.

<u>MetroPCS Stockholders Meeting</u> shall have the meaning set forth in Section 3.3(d).

MetroPCS Subsequent Determination Notice shall have the meaning set forth in Section 4.5(e).

<u>MetroPCS Superior Proposal</u> shall have the meaning set forth in Section 4.5(i).

MetroPCS Superior Proposal Adverse Recommendation Change shall have the meaning set forth in Section 4.5(e).

<u>MetroPCS Termination Amount</u> shall have the meaning set forth in Section 6.2(b).

New MetroPCS Bylaws shall have the meaning set forth in the Recitals.

<u>New MetroPCS Certificate</u> shall have the meaning set forth in the Recitals.

<u>NYS</u>E shall mean the New York Stock Exchange.

<u>Order</u> shall have the meaning set forth in Section 5.1(d).

<u>Organizational Documents</u> shall mean, with respect to any Person, such Person s articles or certificate of association, incorporation, formation or organization, by-laws, limited liability company agreement, partnership agreement or other constituent document or documents, each in its currently effective form as amended from time to time.

<u>Owned Real Property</u> shall mean all real property owned in fee by TMUS and its Subsidiaries or by MetroPCS and its Subsidiaries, as applicable.

Payment Agent shall have the meaning set forth in Section 2.1(e).

<u>Permitted MetroPCS Notes</u> shall mean up to \$3,500,000,000 (and up to an additional \$2,000,000,000 to the extent necessary to satisfy the refinancing of any MetroPCS Existing Notes resulting from any change of control put obligations with respect thereto in connection with the Transaction) of fixed rate senior unsecured notes issued by MetroPCS HoldCo or MetroPCS OpCo in consultation with DT after the date hereof and on or prior to the Closing Date that (a) have a maturity date of not less than 7 and not more than 12 years from the date of issuance thereof, (b) have a call protection pricing schedule that is customary for high yield debt securities, (c) have a non-call period for Permitted MetroPCS Notes (i) with maturities of 7 years, of not more than 3 years from the date of issuance, (ii) with maturities of greater than 7 years and not greater than 9 years, of not more than 4 years from the date of issuance, (iii) with maturities of greater than 9 years from the date of issuance, and (iv) with maturities of greater than 11 years, of not more than 5 years from the date of issuance thereof (taking into account any issuance fees (including underwriting fees) or original issue discount thereon), that is not greater than the initial yield that would be applicable to DT Notes of the same tenor, if such DT Notes were to be issued on the same date as such Permitted MetroPCS Notes, as calculated in accordance with Exhibit F, (e) expressly permit the Transaction (without the need to obtain any waiver, pay any fee, or make any offer to purchase), and (f) are otherwise on the terms set forth in Exhibit G; provided, further, that the proceeds of any Permitted MetroPCS Notes shall be used solely as permitted under this Agreement.

<u>Person</u> shall mean any individual, corporation (including not-for-profit), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, Governmental Entity or other entity of any kind or nature.

Proxy Statement shall have the meaning set forth in Section 3.3(e)(i).

<u>PUCs</u> shall have the meaning set forth in Section 3.1(c)(i).

<u>Regulatory Law</u> shall mean (i) the Sherman Anti-Trust Act of 1890, as amended, (ii) the Clayton Act, as amended, (iii) the HSR Act, (iv) the Federal Trade Commission Act, as amended, (v) any Law analogous to the HSR Act or otherwise regulating antitrust or merger control matters and in each case existing in foreign jurisdictions, (vi) all other Federal, state and foreign, if any, statutes, rules, regulations, orders, decrees, administrative and judicial doctrines and other Laws that are designed or intended to prohibit, restrict or regulate (A) foreign investment or (B) actions having the purpose or effect of monopolization or restraint of trade or lessening of competition, (vii) the Communications Act, (viii) the FCC Rules, and (ix) the rules, regulations and orders of state public utility service or public utility commissions or similar state regulatory bodies.

Regulatory Material Adverse Condition shall have the meaning set forth in Section 4.11(c).

Replacement Welfare Plan shall have the meaning set forth in Section 4.18(b).

<u>Representatives</u> shall mean the directors, officers, employees, Affiliates, agents, investment bankers, financial advisors, attorneys, accountants, brokers, finders, consultants or representatives prior to the Closing of DT and its Subsidiaries or MetroPCS and its Subsidiaries, as applicable.

<u>Resolution Perio</u>d shall have the meaning set forth in Section 2.4(e).

Restricted MetroPCS Contracts shall have the meaning set forth in Section 3.3(p)(ii).

Restricted TMUS Contracts shall have the meaning set forth in Section 3.2(o)(ii).

<u>Sample TMUS Statement</u> shall mean the calculation set forth on Schedule 1.1(b) of the TMUS Disclosure Letter of (i) the TMUS Working Capital as of June 30, 2012, including the asset and liability line items used in such calculation, (ii) the capital expenditures of TMUS and its Subsidiaries for the quarter ended June 30, 2012, including the line items used in such calculation, and (iii) the marketing, subscriber acquisition and subscriber retention expenditures of TMUS and its Subsidiaries for the first eight months of 2012, including the line items used in such calculation.

Sarbanes-Oxley Act shall mean the Sarbanes-Oxley Act of 2002.

<u>SEC</u> shall mean the United States Securities and Exchange Commission.

Securities Act shall mean the Securities Act of 1933, as amended.

<u>Software</u> shall mean computer software, programs and databases in any form, including Internet web sites, web site content, member or user lists and information associated therewith, links, source code, object code, binary code, operating systems, boot loaders, kernels, and specifications, data, databases, database management code, libraries, scripts, utilities, graphical user interfaces, menus, images, icons, forms, methods of processing, software engines, platforms, whether tangible, intangible, separate or embedded, and data formats, all versions, updates, corrections, enhancements, and modifications thereto, and all related documentation, developer notes, comments and annotations.

<u>Spending Deficiency Amount</u> shall mean an amount equal to the sum of (i) an amount (if positive) equal to (A) the number of full days elapsed from and including the date hereof through 12:01 a.m., prevailing Eastern Time, on the Closing Date (the <u>Executory Period</u>), multiplied by \$9,442,000, minus (B) the amount of capital expenditures actually made or expensed by TMUS and its Subsidiaries in the Executory Period (as calculated based on the line items shown on the Sample TMUS Statement for TMUS and its Subsidiaries, in each case, determined in accordance with the accounting principles, practices and methodologies used in the TMUS Applicable Accounting Principles), plus (ii) an amount (if positive) equal to (A) the number of full days elapsed in the Executory Period, multiplied by \$7,869,000, minus (B) the amount of expenditures actually made or expensed by TMUS and its Subsidiaries in the Executory for acquisition and subscriber retention activities in the Executory Period (as calculated based on the line items shown on the Sample TMUS Statement for TMUS Statement for TMUS and its Subsidiaries, in each case, determined in accordance with the accounting principles, practices and methodologies used in the TMUS Applicable Accounting Retention activities in the Executory Period (as calculated based on the line items shown on the Sample TMUS Statement for TMUS and its Subsidiaries, in each case, determined in accordance with the accounting principles, practices and methodologies used in the TMUS Applicable Accounting Principles).

Stock Purchase shall have the meaning set forth in Section 2.2(a).

Stockholder s Agreement shall have the meaning set forth in the Recitals.

<u>Subsidiary</u> shall mean, with respect to any Person, any entity, whether incorporated or unincorporated, of which (i) voting power to elect a majority of the board of directors or others performing similar functions with respect to such other Person is held by the first mentioned Person and/or by any one or more of its Subsidiaries, (ii) a general partnership interest is held by such first mentioned Person and/or by any one or more of its Subsidiaries, where such first mentioned Person (A) does not Beneficially Own a majority of the general partnership interests or voting interests and (B) does not otherwise Control such entity, directly or indirectly, by contract, arrangement or otherwise), or (iii) in excess of 50% of the Equity Interests of such other Person is, directly or indirectly, owned or Controlled by such first mentioned Person and/or by any one or more of its Subsidiaries; <u>provided</u>, that for purposes hereof, Cook Inlet/VS GSM VII PCS Holdings, LLC shall be deemed to be a Subsidiary of TMUS, and <u>provided</u>, <u>further</u>, that for purposes hereof, Iowa Wireless Services LLC shall not be deemed to be a Subsidiary of TMUS.

<u>Tax</u> (including, with correlative meaning, the terms <u>Taxes</u> and <u>Taxable</u>) shall mean all United States federal, state and local and non-United States income, profits, franchise, gross receipts, environmental, customs duty, capital stock, severances, stamp, payroll, sales, employment, unemployment, disability, use, property, withholding, excise, production, value added, occupancy and other taxes, duties or assessments of any nature whatsoever, together with all interest, penalties and additions imposed with respect to such amounts and any interest in respect of such penalties and additions.

<u>Tax Return</u> shall mean all returns and reports (including elections, declarations, disclosures, schedules, estimates, and information returns) required to be supplied to a Taxing Authority relating to Taxes.

<u>Taxing Authority</u> means a Governmental Entity or any subdivision, agency, commission or authority thereof or any quasi-governmental or private body having jurisdiction over the assessment, determination, collection or imposition of any Tax (including the IRS).

<u>Termination Date</u> shall have the meaning set forth in Section 6.1(c).

<u>Territory</u> shall mean the United States, Puerto Rico, and the territories and protectorates of the United States.

<u>TMUS</u> shall have the meaning set forth in the Preamble.

<u>TMUS Applicable Accounting Principles</u> shall mean the accounting principles, practices and methodologies set forth in the Sample TMUS Statement or, to the extent not set forth or reflected therein, as used in TMUS Financial Statements for the fiscal year ended December 31, 2011.

TMUS Acquisition Proposal shall have the meaning set forth in Section 4.6(d).

TMUS Benefit Plans shall have the meaning set forth in Section 3.2(g)(i).

TMUS Board shall mean the board of directors of TMUS.

<u>TMUS Business Plan</u> shall mean TMUS s 2012 and 2013 business plan, a copy of which is attached as Schedule 1.1(c) of the TMUS Disclosure Letter.

<u>TMUS Closing Statement</u> shall have the meaning set forth in Section 2.4(b).

TMUS Closing Statement Dispute Notice shall have the meaning set forth in Section 2.4(d).

TMUS Common Stock shall have the meaning set forth in Section 3.2(b)(i).

TMUS Communications Licenses shall have the meaning set forth in Section 3.2(h)(ii).

<u>TMUS Contract</u> shall mean any agreement, lease, license, contract, note, mortgage, credit agreement, security agreement, indenture, arrangement, commitment, undertaking or other obligation, whether written or oral, binding upon TMUS or any of its Subsidiaries.

TMUS Disclosure Letter shall have the meaning set forth in Section 3.2.

TMUS FCC Licenses shall have the meaning set forth in Section 3.2(h)(ii).

TMUS Financial Statements shall have the meaning set forth in Section 3.2(e)(i).

TMUS Material Adverse Effect shall mean (i) an effect that would prevent or materially delay the ability of DT, Holding or TMUS to consummate the Transaction, or (ii) a material adverse effect on the financial condition, properties, assets, liabilities, business or results of operations of TMUS and its Subsidiaries, taken as a whole; provided, however, with respect to this clause (ii), none of the following shall be deemed to be or constitute a TMUS Material Adverse Effect, or be taken into account when determining whether a TMUS Material Adverse Effect has occurred or would occur: (A) any Circumstance generally affecting (x) the Territory or global economy or Territory or global financial, debt, credit, capital or securities markets or (y) the wireless telecommunications and wireless information products and services industry in the Territory; (B) any Circumstance resulting from any declared or undeclared acts of war, terrorism, outbreaks or escalations of hostilities, sabotage or civil strife or threats thereof; (C) any act of God or weather-related Circumstance; (D) any Circumstance resulting from any change in GAAP or applicable Laws or regulatory or enforcement developments (in the cases of clauses (A), (B), (C) and (D), except to the extent such Circumstance disproportionately affects TMUS and its Subsidiaries, taken as a whole, relative to other companies in the wireless telecommunications and wireless information services industry in the Territory); (E) any Circumstance resulting from any failure by TMUS or its Subsidiaries to meet any estimates, projections, budgets or forecasts of revenues or earnings for any period ending on or after the date hereof, or any rumors, predictions or reports of such failure; provided, that the exception in this clause (E) shall not prevent or otherwise affect a determination that any Circumstance underlying such failure has resulted in or contributed to a TMUS Material Adverse Effect; (F) any Circumstance resulting from any action required to be taken or omitted to be taken pursuant to this Agreement; or (G) any Circumstance resulting from the announcement, pendency or public disclosure of this Agreement and the Transaction. Any determination of TMUS Material Adverse Effect shall exclude the effects of the matters disclosed in the TMUS Disclosure Letter or the matters specifically identified in the notes to the TMUS Financial Statements.

TMUS Material Licenses shall have the meaning set forth in Section 3.2(h)(i).

TMUS Merger shall have the meaning set forth in Section 2.3(b).

TMUS Owned Intellectual Property shall have the meaning set forth in Section 3.2(n)(i).

<u>TMUS Permitted Encumbrances</u> shall mean (i) Encumbrances reflected or reserved against or otherwise disclosed in the TMUS Financial Statements or the TMUS Disclosure Letter; (ii) mechanics , materialmen s, warehousemen s, carriers , workers or repairmen s liens or other common law or statutory Encumbrances arising or incurred in the ordinary course of TMUS s business consistent with past practice and that are not material in amount or effect on the business of TMUS and its Subsidiaries, taken as a whole; (iii) liens for Taxes, assessments and other governmental charges not yet due and payable or due but not delinquent or being contested in good faith by appropriate proceedings and for which adequate reserves have been established if and to the extent required by GAAP, in the most recent TMUS Financial Statements; and (iv) with respect to real property, (A) easements, quasi-easements, licenses, covenants, rights-of-way, rights of re-entry or other similar restrictions, including any other agreements, conditions or restrictions that would be shown by a current title report or other similar report or listing, in each case that do not or would not materially impair the conduct of business of TMUS and its Subsidiaries, taken as a whole, or the use or value of the relevant asset, (B) any

conditions that may be shown by a current survey or physical inspection, in each case that do not or would not materially impair the conduct of business of TMUS and its Subsidiaries, taken as a whole, or the use or value of the relevant asset, and (C) zoning, building, subdivision or other similar requirements or restrictions, in each case that do not or would not materially impair the conduct of business of TMUS and its Subsidiaries, taken as a whole, or the use or value of the relevant asset, and (C) zoning, building, subdivision or other similar requirements or restrictions, in each case that do not or would not materially impair the conduct of business of TMUS and its Subsidiaries, taken as a whole, or the use or value of the relevant asset.

TMUS Shares shall mean all of the Equity Interests of TMUS.

TMUS State Licenses shall have the meaning set forth in Section 3.2(h)(ii).

TMUS Stock Consideration shall have the meaning set forth in Section 2.2(b).

<u>TMUS Working Capital</u> means, for the applicable date and time, (a) the sum of the amounts for the asset line items shown on the Sample TMUS Statement for TMUS and its Subsidiaries, <u>minus</u> (b) the sum of the amounts for the liability line items shown on the Sample TMUS Statement for TMUS and its Subsidiaries, in each case, determined in accordance with the accounting principles, practices and methodologies used in the TMUS Applicable Accounting Principles.

<u>TMUS Working Capital Facility</u> shall mean a revolving credit facility made available by DT (or one of its Subsidiaries if the obligations of such Subsidiary thereunder are unconditionally guaranteed by DT) for the benefit of TMUS and its Subsidiaries, for working capital and other general corporate purposes, with a maximum principal amount of no less than \$500,000,000, which facility shall be on terms and conditions substantially as set forth on Exhibit H and otherwise reasonably acceptable to DT and MetroPCS.

<u>Tower Assets</u> means (i) the Owned Real Property and Leased Real Property owned or leased by TMUS or any of its Subsidiaries that have, as fixtures or appurtenances thereto, cellular transmission towers or building pads therefor owned or leased by TMUS or its Subsidiaries, but excluding any retail stores, business offices or any location where TMUS and its Subsidiaries do not have any facilities operating on TMUS FCC Licenses, and (ii) such cellular transmission towers and building pads therefor, all as set forth in Schedule 1.1(d) of the TMUS Disclosure Letter.

Tower Holdco shall have the meaning set forth in Section 4.25.

Trade Secrets shall have the meaning set forth in the definition of Intellectual Property.

Trademarks shall have the meaning set forth in the definition of Intellectual Property.

Trademark License shall have the meaning set forth in the Recitals.

<u>Transaction</u> shall mean the transactions contemplated by this Agreement.

<u>Transfer Taxes</u> shall mean any and all transfer Taxes (excluding Taxes measured in whole or in part by net income or gain), including sales, use, excise, stock, stamp, documentary, filing, real estate transfer, recording, permit, license, authorization and similar Taxes.

<u>Unresolved Items</u> shall have the meaning set forth in Section 2.4(f).

Voting Debt shall have the meaning set forth in Section 3.2(b)(ii).

<u>WARN Act</u> shall have the meaning set forth in Section 3.2(m)(v).

1.2 Other Definitional Provisions. Unless the express context otherwise requires:

(a) the words hereof, herein, and hereunder and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement;

(b) the words date hereof, when used in this Agreement, shall refer to the date set forth in the Preamble;

(c) the terms defined in the singular have a comparable meaning when used in the plural, and vice versa;

(d) the terms defined in the present tense have a comparable meaning when used in the past tense, and vice versa;

(e) any references herein to Dollars and \$ are to United States Dollars;

(f) any references herein to a specific Section, Schedule, Annex or Exhibit shall refer, respectively, to Sections, Schedules, Annexes or Exhibits of this Agreement;

(g) wherever the word include, includes, or including is used in this Agreement, it shall be deemed to be followed by the words without limitation;

(h) references herein to any gender includes each other gender;

(i) the word or shall not be exclusive; and

(j) references to documents or other materials provided, delivered or made available to DT, TMUS or MetroPCS, as applicable, shall mean that such documents or other materials were present prior to the execution and delivery of this Agreement in the on-line data room maintained for purposes of the Transaction by DT, TMUS or MetroPCS, as applicable, and accessible by the other parties or their Representatives.

#### ARTICLE II

#### **RECAPITALIZATION; STOCK PURCHASE; CLOSING**

2.1 The Recapitalization. On the Closing Date, MetroPCS shall undertake a recapitalization as follows:

(a) <u>Amendment of Certificate of Incorporation: Reverse Stock Split</u>. Upon the terms and subject to the conditions set forth in this Agreement, MetroPCS shall effect a reverse stock split, pursuant to which each share of MetroPCS Common Stock outstanding as of the Effective Time shall thereafter represent 0.5 of a share of MetroPCS Common Stock (the <u>MetroPCS Exchange Ratio</u>) (the <u>MetroPCS Reverse Stock Split</u>), by filing a Certificate of Amendment with the Secretary of State of the State of Delaware, providing for the amendment and restatement of the Certificate of Incorporation of MetroPCS in the form of the New MetroPCS Certificate. The New MetroPCS Certificate shall be the Certificate of Incorporation of MetroPCS from and after the Effective Time, until thereafter changed or amended as provided therein and/or in accordance with its provisions and applicable Law (the date and time the New MetroPCS Certificate becomes effective, the <u>Effective Time</u>).

(b) <u>Amendment of Bylaws</u>. MetroPCS shall take all actions necessary so that, effective as of the Effective Time, the amended Bylaws of MetroPCS shall be amended and restated in the form of the New MetroPCS Bylaws, which shall be the Bylaws of MetroPCS from and after the Effective Time, until thereafter changed or amended as provided therein, in the New MetroPCS Certificate and/or in accordance with applicable Law.

(c) <u>Cash Payment</u>. As part of the recapitalization of MetroPCS, subject to the terms and conditions set forth in this Agreement, and conditioned upon the effectiveness of the MetroPCS Reverse Stock Split, effective immediately following the Effective Time, MetroPCS shall make a payment (the <u>Cash Payment</u>) in cash in an amount equal to \$1,500,000,000, without interest, in the aggregate (the <u>MetroPCS Cash Am</u>ount), to the MetroPCS Stockholders of record immediately following the Effective Time, by paying to each such MetroPCS Stockholder an amount per share of MetroPCS Common Stock held of record by such holder immediately following the Effective Time (with the number of shares of MetroPCS Common Stock held by such holder, for the avoidance of doubt, taking into account the MetroPCS Reverse Stock Split) equal to the MetroPCS Cash

Amount divided by the aggregate number of shares of MetroPCS Common Stock (with the number of shares of MetroPCS Common Stock, for the avoidance of doubt, taking into account the MetroPCS Reverse Stock Split) outstanding immediately following the Effective Time (the <u>MetroPCS Per-Share Cash Amount</u>). The amount to be paid to each holder of record of MetroPCS Common Stock immediately following the Effective Time shall be rounded up to the nearest whole cent.

#### (d) MetroPCS Equity and Equity-Based Awards.

(i) Effective as of the Effective Time, and except as otherwise provided in Section 2.1(d)(iii), each then outstanding option to purchase shares of MetroPCS Common Stock (each, a <u>MetroPCS Stock Option</u>) granted to or held by any current or former employee, officer or director of, or consultant or other service provider to, MetroPCS or any of its Affiliates shall be adjusted such that (A) it shall become an option to purchase a number (rounded down to the nearest whole number) of shares of MetroPCS Common Stock (an <u>Adjusted MetroPCS Stock Option</u>) equal to the product of (1) the number of shares of MetroPCS Common Stock subject to such MetroPCS Stock Option immediately prior to the MetroPCS Reverse Stock Split, multiplied by (2) the MetroPCS Exchange Ratio, and (B) the per share exercise price for MetroPCS Common Stock issuable upon the exercise of such Adjusted MetroPCS Stock Option shall be adjusted to be equal to (rounded up to the nearest cent) (x) the quotient of (1) the exercise price per share of MetroPCS Common Stock for which such MetroPCS Stock Option was exercisable immediately prior to the MetroPCS Reverse Stock Split, divided by (2) the MetroPCS Exchange Ratio, *less* (y) the MetroPCS Per-Share Cash Amount (such adjusted per-share exercise price, the <u>Adjusted Per-Share Option Exercise Price</u>). Except as otherwise provided in this Section 2.1(d)(i) and in Section 2.1(d)(ii), each Adjusted MetroPCS Stock Option shall be subject to the same terms and conditions (including expiration dates and exercise provisions, taking into account, to the extent applicable, Section 2.1(d)(ii) and Section 2.1(d)(iv)), as were applicable to the corresponding MetroPCS Stock Option immediately prior to the Effective Time. This Section 2.1(d)(i) is intended to comply with Section 409A (and, to the extent applicable by reason of Section 409A, Section 424) of the Code and the Treasury Regulations issued thereunder and will be interpreted accordingly.

(ii) Effective as of the consummation of the Stock Purchase, each then outstanding MetroPCS Stock Option (whether or not converted into an Adjusted MetroPCS Stock Option) shall automatically and without any action on behalf of the holder thereof, immediately vest and become exercisable in accordance with its terms.

(iii) Notwithstanding Section 2.1(d)(i), (x) each MetroPCS Stock Option (other than any Low Exercise Price MetroPCS Stock Option) with an exercise price per share immediately prior to the Effective Time (ignoring any adjustment for the MetroPCS Reverse Stock Split or the Cash Payment, including pursuant to Section 2.1(d)(i)) that is less than the MetroPCS Closing Price (an <u>In-the-Money MetroPCS Stock Option</u>) shall, at the written election, as to all or any portion of such In-the-Money MetroPCS Stock Option, of the holder of such In-the-Money MetroPCS Stock Option granted under the Second Amended & Restated 1995 Stock Option Plan of MetroPCS that, if it were adjusted pursuant to Section 2.1(d)(i), would have an Adjusted Per-Share Option Exercise Price that is less than or equal to zero (a <u>Low Exercise Price MetroPCS Stock Option</u>), shall, automatically and without any action on behalf of the holder thereof, in each case of (x) and (y) in lieu of becoming an Adjusted MetroPCS Stock Option, be cancelled and converted, effective as of the consummation of the Stock Purchase, into the right to receive from MetroPCS, as promptly as reasonably practicable and in any event no later than 10 Business Days after the Closing Date, in full settlement of such holder s right thereunder, an amount in cash, without interest, equal to (A) the product of (1) the aggregate number of shares of MetroPCS Common Stock subject to such In-the-Money MetroPCS Stock Option (or portion thereof) for which the holder has elected to receive cash pursuant to this Section 2.1(d)(iii) or Low Exercise Price MetroPCS Stock Option, as applicable, immediately prior to the Effective Time (and in each case ignoring any adjustment for the MetroPCS

Reverse Stock Split and the Cash Payment, including pursuant to Section 2.1(d)(i)), multiplied by (2) the amount by which the MetroPCS Closing Price exceeds the exercise price per share of such In-the-Money MetroPCS Stock Option (or portion thereof) for which the holder has elected to receive cash pursuant to this Section 2.1(d)(iii) or Low Exercise Price MetroPCS Stock Option, as applicable, immediately prior to the Effective Time (and in each case ignoring any adjustment for the MetroPCS Reverse Stock Split and the Cash Payment, including pursuant to Section 2.1(d)(i)), less (B) any Taxes required to be withheld from such payment.

(iv) For the avoidance of doubt, the MetroPCS Reverse Stock Split and Cash Payment contemplated by this Section 2.1 shall be deemed for all purposes of the MetroPCS Benefit Plans (including each of MetroPCS s equity-based compensation plans, as amended, identified on Schedule 3.3(h) of the MetroPCS Disclosure Letter (the <u>MetroPCS Stock Plans</u>)) to have occurred contingent upon the consummation of the Stock Purchase such that, (A) there shall be a Change in Control, Change of Control and Corporate Transaction, as applicable, at the time of the consummation of the Stock Purchase under and pursuant to the terms of the MetroPCS Benefit Plans (including the MetroPCS Stock Plans) and (B) all severance, accelerated vesting, lapsing of restrictions and other rights and benefits that accrue and become effective under the MetroPCS Benefit Plans (including the MetroPCS Stock Plans) upon a Change in Control, Change of Control and Corporation Transaction, as applicable, as a result of the Transaction shall accrue and become effective as of the consummation of the Stock Purchase pursuant to such MetroPCS Benefit Plans (including the MetroPCS Stock Plans) and the provisions hereof. For the further avoidance of doubt, the MetroPCS Reverse Stock Split and Cash Payment shall together be deemed to constitute a recapitalization under each of the MetroPCS Stock Plans, and the applicable provisions of each such MetroPCS Stock Plan shall be construed accordingly. At or prior to the consummation of the Stock Purchase, the MetroPCS Board (or the appropriate committee thereof) and the boards of directors or management committees of its Subsidiaries shall pass such resolutions as may be necessary to effectuate the provisions of this Section 2.1(d), including to (x) ensure that all MetroPCS Stock Options that are not converted into Adjusted MetroPCS Stock Options are cashed out and cancelled effective as of the consummation of the Stock Purchase pursuant to Section 2.1(d)(iii) (and the MetroPCS Stock Plans are hereby deemed amended to the extent necessary to effectuate the provisions of this Section 2.1(d)) and that no holder of any such cashed-out and cancelled MetroPCS Stock Option shall have any right with respect thereto, except as provided in this Section 2.1(d) and (y) amend the award agreements with respect to any MetroPCS Restricted Stock to permit the holder of such MetroPCS Restricted Stock to receive and keep the Cash Payment contemplated by this Section 2.1.

(v) Prior to the Effective Time, MetroPCS shall deliver to each holder of the MetroPCS Stock Options appropriate notices and, if applicable, election forms, setting out the terms applicable to such MetroPCS Stock Options with respect to the Transaction.

(e) <u>Payment Agent: Deposit of MetroPCS Cash Amount</u>. At or prior to the Effective Time, MetroPCS shall deposit, or shall cause to be deposited, with a commercial bank or trust company designated by MetroPCS and reasonably satisfactory to DT (the <u>Payment Agent</u>) for the benefit of the MetroPCS Stockholders of record immediately following the Effective Time, (i) cash in lieu of any fractional shares, to be paid pursuant to Section 2.1(f)(v), and (ii) cash in an amount equal to the MetroPCS Cash Amount (such deposited amount described in clauses (i) and (ii), the <u>MetroPCS Cash Deposit</u>): provided, that, if the Closing does not occur, the Payment Agent shall return, or cause to be returned, the MetroPCS Cash Deposit, and any interest or other income thereon, to MetroPCS on the next Business Day following the termination of this Agreement pursuant to its terms. Except as otherwise agreed to by the parties, the investment of the MetroPCS Cash Deposit shall in all events be limited to direct short-term obligations of, or short-term obligations fully guaranteed as to principal and interest by, the United States government, in commercial paper rated A-1 or P-1 or better by Moody s Investors Service, Inc. or Standard & Poor s Corporation, respectively, or in certificates of deposit, bank repurchase agreements, other bank instruments or direct deposits or banker s acceptances of commercial banks with capital exceeding \$10,000,000 (based on the most recent financial statements of such bank that are then publicly

available); <u>provided</u>, that no investment or loss thereon shall affect the amounts payable to holders of MetroPCS Common Stock pursuant to this Section 2.1. The MetroPCS Cash Deposit shall be used solely for purposes of making the Cash Payment and paying any cash in lieu of fractional shares to be paid pursuant to Section 2.1(f)(v), and shall not be used to satisfy any other obligation of MetroPCS, DT, TMUS or any of their respective Subsidiaries.

### (f) Exchange of Shares.

(i) Notwithstanding anything in this Section 2.1(f) to the contrary, each certificate that immediately prior to the Effective Time represented shares of MetroPCS Common Stock (each, a <u>MetroPCS Certificate</u>) shall thereafter (and without the necessity of presenting the same for exchange) represent that number of shares of MetroPCS Common Stock into which the shares of MetroPCS Common Stock represented by such MetroPCS Certificate shall have been combined pursuant to the MetroPCS Reverse Stock Split, subject to the limitation on fractional shares pursuant to Section 2.1(f)(v).

(ii) As soon as practicable after the Closing, MetroPCS shall instruct the Payment Agent to send to each holder of record of a MetroPCS Certificate immediately prior to the Effective Time, (A) a letter of transmittal (which shall be in customary form and shall specify, among other things, that the delivery shall be effected, and risk of loss and title to such MetroPCS Certificate shall pass, only upon proper delivery of such MetroPCS Certificate to the Payment Agent) and (B) instructions for use in effecting the surrender of such MetroPCS Certificate in exchange for a new certificate representing that number of shares of MetroPCS Common Stock into which the shares of MetroPCS Common Stock represented by such MetroPCS Certificate shall have been combined pursuant to the MetroPCS Reverse Stock Split, the portion of the MetroPCS Cash Amount that such holder has the right to receive pursuant to the Cash Payment and cash in lieu of fractional shares pursuant to Section 2.1(f)(v).

(iii) Upon surrender of a MetroPCS Certificate for cancellation to the Payment Agent, together with a properly completed letter of transmittal, the Payment Agent (A) shall register in the name of the holder of such MetroPCS Certificate the number of whole shares of MetroPCS Common Stock (in the form of book-entry shares, unless the holder of such MetroPCS Certificate expressly requests in writing that such shares be delivered in certificated form) representing, in the aggregate, the whole number of shares of MetroPCS Common Stock, if any, into which the shares of MetroPCS Common Stock represented by such MetroPCS Certificate shall have been combined pursuant to the MetroPCS Reverse Stock Split and (B) shall deliver to the holder of such MetroPCS Certificate a check or wire transfer in same day funds for the amount equal to the portion of the MetroPCS Cash Amount that such holder has the right to receive pursuant to the Cash Payment and cash payable in lieu of fractional shares pursuant to Section 2.1(f)(v). The MetroPCS Certificate so surrendered shall forthwith be cancelled. Promptly (and in any event no more than two Business Days) after the Closing, with respect to each holder of book-entry shares which immediately prior to the Effective Time represented shares of MetroPCS Common Stock (<u>MetroPCS Book-Entry Shares</u>), the Payment Agent (x) shall register in the name of such holder the number of whole shares of MetroPCS Common Stock (in the form of book-entry shares) representing, in the aggregate, the whole number of shares of MetroPCS Common Stock, if any, into which such holder s MetroPCS Book-Entry Shares shall have been combined pursuant to the MetroPCS Reverse Stock Split, and (y) deliver to such holder a check or wire transfer in same day funds for the amount equal to the portion of the MetroPCS Cash Amount that such holder has the right to receive pursuant to the Cash Payment and cash payable in lieu of fractional shares pursuant to Section 2.1(f)(v), without such holder being required to deliver a MetroPCS Certificate or an executed letter of transmittal to the Payment Agent. No interest will be paid or accrued on any MetroPCS Payment, any cash in lieu of fractional shares or any other cash payments payable in respect of any such securities pursuant to this Agreement.

(iv) If any shares of MetroPCS Common Stock that are combined pursuant to the MetroPCS Reverse Stock Split are to be registered in the name of, or if any cash in respect thereof is to be paid to, a Person other than that in whose name the MetroPCS Certificate(s) surrendered pursuant to this Section 2.1(f) is or

are registered (whether as the result of a transfer of such shares or otherwise), it shall be a condition to the registration of such shares and the cash payments that (A) such certificate or certificates so surrendered shall be properly endorsed (or accompanied by an appropriate instrument or instruments of transfer as MetroPCS or the Payment Agent may reasonably request) and otherwise in proper form for transfer, (B) the Person requesting such registration and payment shall pay to the Payment Agent in advance any transfer or other Taxes required by reason of the payment or registration in any name other than that of the registered holder of the shares represented by the MetroPCS Certificate surrendered or required for any other reason, or shall establish to the satisfaction of the Payment Agent that such Tax has been paid or is not payable, and (C) the Person receiving such payment and in whose name such shares are being registered (I) represents and warrants to MetroPCS that such Person is entitled to such payment and shares being registered and (II) agrees to indemnify MetroPCS from and against any and all Damages resulting from, arising out of, or incurred in connection with any claim that any other Person is entitled to such payment and shares being registered.

(v) No certificates or scrip representing fractional shares of MetroPCS Common Stock or book-entry credit of the same will be issued in connection with the MetroPCS Reverse Stock Split, including upon the surrender for exchange of shares of MetroPCS Common Stock, but in lieu thereof each MetroPCS Stockholder who would otherwise be entitled to a fraction of a share of MetroPCS Common Stock in connection with the MetroPCS Reverse Stock Split (after aggregating all fractional shares of MetroPCS Common Stock to be received by such MetroPCS Stockholder) shall receive an amount of cash (rounded up to the nearest whole cent), without interest, equal to the product of such fraction multiplied by the MetroPCS Closing Price.

(vi) Any portion of the MetroPCS Cash Deposit that remains unclaimed by MetroPCS Stockholders or former MetroPCS Stockholders as of the first anniversary of the Closing shall be paid to MetroPCS. Any MetroPCS Stockholders or former MetroPCS Stockholders shall thereafter look only to MetroPCS for the issuance of any certificates of MetroPCS Common Stock or payment of any portion of the MetroPCS Cash Amount or cash in lieu of any fractional shares of MetroPCS Common Stock such MetroPCS Stockholders or former MetroPCS Stockholders are entitled to receive pursuant to this Agreement, without any interest thereon. Notwithstanding the foregoing, none of MetroPCS, the Payment Agent, any of their respective Representatives or any other Person shall be liable to any MetroPCS Stockholder or former MetroPCS Stockholder for any amount delivered in good faith to a public official pursuant to applicable abandoned property, escheat or similar laws.

(vii) In the event any MetroPCS Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact in a form reasonably acceptable to MetroPCS by the Person claiming such certificate to be lost, stolen or destroyed and, if reasonably required by MetroPCS, the posting by such Person of a bond in such amount as MetroPCS may determine is reasonably necessary as indemnity against any claim that may be made against it with respect to such certificate, the Payment Agent shall (A) register in the name of the holder of such MetroPCS Certificate the number of whole shares of MetroPCS Common Stock (in the form of book-entry shares, unless the holder of such MetroPCS Certificate expressly requests in writing that such shares be delivered in certificated form) representing, in the aggregate, the whole number of shares of MetroPCS Common Stock, if any, into which the shares of MetroPCS Common Stock represented by such MetroPCS Certificate shall have been combined pursuant to the MetroPCS Reverse Stock Split and (B) deliver to the holder of such MetroPCS Certificate a check or wire transfer in same day funds for the amount equal to the portion of the MetroPCS Cash Amount that such holder has the right to receive pursuant to the Cash Payment and cash payable in lieu of fractional shares pursuant to Section 2.1(f)(v).

(g) <u>Withholding Rights</u>. The Payment Agent and MetroPCS shall be entitled to deduct and withhold from the MetroPCS Cash Amount and any other amounts otherwise payable pursuant to this Agreement such amounts as the Payment Agent or MetroPCS, as the case may be, is required to deduct and withhold under the Code, or any provision of state, local or foreign Tax Law, with respect to the making of such payment (which deduction,

in the case of the MetroPCS Restricted Stock, shall be made from the shares into which such MetroPCS Restricted Stock is combined pursuant to the MetroPCS Reverse Stock Split). To the extent the amounts are so withheld by the Payment Agent or MetroPCS, as the case may be, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of whom such deduction and withholding was made.

2.2 <u>Stock Purchase</u>. Upon the terms and subject to the conditions set forth in this Agreement, at the Closing and effective immediately following the Cash Payment:

(a) Holding will sell, convey, assign, transfer and deliver to MetroPCS, free and clear of all Encumbrances, and MetroPCS will purchase, acquire and accept from Holding, all of Holding s right, title and interest in and to TMUS Shares, free and clear of all Encumbrances, and shall deliver or cause to be delivered to MetroPCS certificates representing the TMUS Shares, duly endorsed in blank or accompanied by stock powers duly endorsed in blank in proper form for transfer, with appropriate transfer stamps, if any, affixed (the <u>Stock Purchase</u>); provided that notwithstanding the foregoing or any other provision hereof, neither MetroPCS, TMUS nor any of their respective Subsidiaries shall assume or be obligated to pay, perform or otherwise discharge any (and DT shall retain, pay, perform or otherwise discharge without recourse to MetroPCS, TMUS or any of their Subsidiaries all, and as applicable shall reimburse MetroPCS, TMUS and any of their Subsidiaries for their satisfaction of any) Excluded Liabilities;

(b) MetroPCS shall issue and deliver (the <u>MetroPCS Share Issuance</u>) to Holding or its designee, a number of shares of MetroPCS Common Stock equal to 74% of the fully-diluted (as calculated under the treasury method based on the MetroPCS Closing Price after taking into account the MetroPCS Reverse Stock Split and the Cash Payment, but not taking into account any cancellation of MetroPCS Stock Options pursuant to Section 2.1(d)(iii)) shares of MetroPCS Common Stock outstanding immediately following the Cash Payment (on a grossed-up basis to take into account the number of shares of MetroPCS Common Stock so issued to Holding or its designee), which, at DT s election, may be represented by one or more certificates or may be uncertificated (the <u>TMUS Stock Consideration</u>); provided, that in the event of any dividend or distribution (other than the Cash Payment), stock split, reverse stock split (other than the MetroPCS Reverse Stock Split), stock dividend, reorganization, reclassification, merger, combination, recapitalization, or other like change with respect to or affecting shares of MetroPCS Common Stock (or in respect of which a record date or effective date, as applicable, has been declared and passed prior to the Effective Time), and including any stock repurchase or redemption effected on a substantially pro rata basis or in which a majority of the MetroPCS Stockholders participate, which occurs prior to the Effective Time and which affects the number of shares of MetroPCS Common Stock that DT should equitably receive, such number of shares of MetroPCS Common Stock shall be equitably adjusted to the extent necessary to provide the parties the same economic effect as contemplated by this Agreement prior to such transaction or declaration; and <u>provided</u>, <u>further</u>, that the MetroPCS Share Issuance shall be made only in whole shares, and any fractional shares shall be rounded up to the nearest whole share; and

(c) DT shall pay to TMUS the Estimated Adjustment Amount, if any.

2.3 <u>Mergers</u>. Upon the terms and subject to the conditions set forth in this Agreement, on the Business Day immediately following the Closing Date, unless otherwise agreed by the Parties in writing in good faith:

(a) In accordance with Section 253 of the DGCL, MetroPCS shall cause MetroPCS HoldCo to merge with and into MetroPCS OpCo (the <u>MetroPCS Merger</u>), whereupon the separate existence of MetroPCS HoldCo shall cease and MetroPCS OpCo shall continue its existence as the surviving corporation under the laws of the State of Delaware. MetroPCS shall cause MetroPCS OpCo to file with the Secretary of State of Delaware a certificate of merger, upon the filing of which the MetroPCS Merger shall be effective. By virtue of the MetroPCS Merger, (i) each share of common stock, par value \$0.0001 per share, of MetroPCS HoldCo issued and outstanding immediately prior to the effectiveness of the MetroPCS Merger shall automatically be converted into one validly issued, fully paid and nonassessable share of common stock of MetroPCS OpCo, and (ii) each

share of common stock, par value \$0.0001 per share, of MetroPCS OpCo shall be automatically canceled. Prior to the Closing Date, MetroPCS shall cause MetroPCS HoldCo and MetroPCS OpCo to enter into an agreement and plan of merger reflecting the provisions of this Section 2.3(a).

(b) Immediately following the MetroPCS Merger, in accordance with Section 251 of the DGCL, MetroPCS shall cause MetroPCS OpCo to merge with and into TMUS (the <u>TMUS Merger</u>), whereupon the separate existence of MetroPCS OpCo shall cease and TMUS shall continue its existence as the surviving corporation under the laws of the State of Delaware. MetroPCS shall cause TMUS to file with the Secretary of State of Delaware a certificate of merger, upon the filing of which the TMUS Merger shall be effective. By virtue of the TMUS Merger, (i) each share of common stock, par value \$0.0001 per share, of MetroPCS OpCo issued and outstanding immediately prior to the effectiveness of the TMUS Merger shall automatically be canceled, and (ii) each share of common stock, par value \$0.000001 per share, of TMUS shall remain outstanding. Prior to the Closing Date, MetroPCS shall cause MetroPCS OpCo to, and DT shall cause TMUS to, and TMUS shall, enter into an agreement and plan of merger reflecting the provisions of this Section 2.3(b), and MetroPCS and Holding shall provide such consents as may be necessary or advisable to authorize the TMUS Merger.

2.4 <u>Adjustment</u>. (a) At least five Business Days prior to the expected Closing Date (and in any event not more than 10 Business Days prior to the actual Closing Date), DT shall prepare and deliver to MetroPCS a statement (the <u>Estimated TMUS Closing Statement</u>) consisting of a calculation in reasonable detail (including calculations of the TMUS Working Capital and the Spending Deficiency Amount) of the estimated Adjustment Amount, if any (the <u>Estimated Adjustment Amount</u>). The Estimated TMUS Closing Statement shall be signed by a duly authorized officer of DT and prepared in good faith and in accordance with the TMUS Applicable Accounting Principles and using the line items set forth on the Sample TMUS Statement. DT shall provide to MetroPCS and its Representatives such access to the books and records of TMUS and its Subsidiaries and to any other information, including such access to TMUS s and its Subsidiaries employees and work papers of their accountants (subject to MetroPCS entering into, and such accountants agreeing to, a customary agreement relating to such access to work papers in form and substance reasonably acceptable to such accountants), as MetroPCS shall reasonably request, in connection with MetroPCS s review of the Estimated TMUS Closing Statement and preparation of the TMUS Closing Statement. MetroPCS shall have the right to object to the amounts contained in the Estimated TMUS Closing Statement within two Business Days after the delivery of the Estimated TMUS Closing Statement to MetroPCS. DT shall in good faith consider the objections, if any, of MetroPCS to the Estimated TMUS Closing Statement and, if MetroPCS has made any objections, will re-issue an Estimated TMUS Closing Statement containing the Estimated Adjustment Amount no later than two Business Days prior to the Closing Date with any such revisions that DT has determined in good faith are appropriate.

(b) <u>Closing Statement</u>. As promptly as practicable following the Closing Date (but in any event within 90 days thereafter), MetroPCS shall prepare, or cause to be prepared, and deliver to DT a statement (the <u>TMUS Closing Statement</u>) consisting of a calculation in reasonable detail (including calculations of the TMUS Working Capital and the Spending Deficiency Amount) of (i) the Adjustment Amount, if any, and (ii) the amount, if any, payable pursuant to Section 2.4(g). The TMUS Closing Statement shall be signed by a duly authorized officer of MetroPCS and prepared in good faith and in accordance with the TMUS Applicable Accounting Principles and using the line items set forth on the Sample TMUS Statement.

(c) <u>Access to Information</u>. MetroPCS shall provide to DT and its Representatives such access to the books and records of TMUS and its Subsidiaries and to any other information, including such access to TMUS s and its Subsidiaries employees and work papers of the accountants who compiled or reviewed the TMUS Closing Statement or the underlying accounting data (subject to DT entering into, and such accountants agreeing to, a customary agreement relating to such access to work papers in form and substance reasonably acceptable to such accountants), as DT shall reasonably request, in connection with DT s review of the TMUS Closing Statement.

(d) <u>Closing Statement Dispute Notice</u>. The TMUS Closing Statement shall become final, binding and conclusive upon DT and MetroPCS on the 45th day following DT s receipt of the TMUS Closing Statement unless, if DT in good faith disputes one or more items contained in the TMUS Closing Statement on the basis that it was not prepared and calculated in accordance with this Agreement or that there was a mathematical error (a <u>Disputed Item</u>), on or prior to such 45th day, DT delivers to MetroPCS a written notice (a <u>TMUS Closing Statement Dispute Notice</u>) stating that DT disputes one or more Disputed Items and specifying in reasonable detail each Disputed Item. If DT timely delivers a TMUS Closing Statement Dispute Notice, all items in the TMUS Closing Statement that are not Disputed Items shall be final, binding and conclusive as to MetroPCS and DT for all purposes hereunder except for any such items that must be changed or adjusted as a direct consequence of a change or adjustment to a Disputed Item.

(e) <u>Resolution Period</u>. If DT timely delivers a TMUS Closing Statement Dispute Notice, then MetroPCS and DT shall seek in good faith to resolve the Disputed Items during the 30-day period beginning on the date MetroPCS receives the TMUS Closing Statement Dispute Notice (the <u>Resolution Period</u>). If MetroPCS and DT reach agreement with respect to any Disputed Items, MetroPCS shall revise the TMUS Closing Statement to reflect such agreement.

(f) Independent Accountant. If MetroPCS and DT are unable to resolve all Disputed Items during the Resolution Period, then, at the request of either party, MetroPCS and DT shall jointly engage and submit the unresolved Disputed Items (the <u>Unresolved Items</u>) to the Independent Accountant. MetroPCS and DT shall enter into reasonable and customary arrangements for the services to be rendered by the Independent Accountant, including a customary non-disclosure agreement. MetroPCS and DT shall use their reasonable best efforts to cause the Independent Accountant to issue its written determination regarding the Unresolved Items within 30 days after such items are submitted to it for resolution. The Independent Accountant shall make a determination with respect to the Unresolved Items only and in a manner consistent with this Section 2.4 and the TMUS Applicable Accounting Principles. The Independent Accountant shall limit its review only to the Unresolved Items. In reviewing any Unresolved Items, the Independent Accountant may not assign a value to any Unresolved Item that is greater than the greatest value or less than the smallest value for such Unresolved Item claimed by either party. Each party shall use its reasonable best efforts to furnish to the Independent Accountant such work papers and other documents and information pertaining to the Unresolved Items as the Independent Accountant may reasonably request. The determination of the Independent Accountant shall be final, binding and conclusive on MetroPCS and DT absent manifest error. The fees, expenses and costs of the American Arbitration Association, if applicable, and the Independent Accountant shall be borne by DT and MetroPCS in the same proportion as the aggregate amount of the Unresolved Items submitted to the Independent Accountant shall be borne by the Independent Accountant) bears to the total amount of the Unresolved Items submitted to the Independent Accountant shall be accountant) bears to the total amount of the Unresolved Items submitted to the

(g) <u>Final Adjustment</u>. If (i) the Adjustment Amount as set forth on the TMUS Closing Statement delivered by MetroPCS to DT if DT does not timely deliver a TMUS Closing Statement Dispute Notice pursuant to Section 2.4(d), or as agreed by MetroPCS and DT pursuant to Section 2.4(e), or as determined by the Independent Accountant pursuant to Section 2.4(f), as applicable) (the <u>Actual Adjustment Amount</u>) exceeds the Estimated Adjustment Amount, then DT shall pay to TMUS an amount equal to such excess, or (ii) the Estimated Adjustment Amount, then TMUS shall pay to DT an amount equal to such excess.

(h) <u>Method of Payment, Interest, etc</u>. Any amount paid pursuant to Section 2.4(g) shall be (i) increased by interest on such amount at an annual interest rate equal to 4%, from the Closing Date to and including the date of payment based on a 365 day year, and (ii) made within five Business Days after the TMUS Closing Statement becomes final pursuant to Section 2.4(d), (e) or (f) by wire transfer of immediately available cash funds to an account designated by the receiving party.

2.5 <u>Closing</u>. The closing of the Transaction (the <u>Closing</u>) shall take place at the offices of Wachtell, Lipton, Rosen & Katz, 51 West 52nd Street, New York, NY 10019, at 9:00 a.m., local time, on the third Business Day following the satisfaction or waiver (if permissible under applicable Laws) of the conditions set forth in Article V (other than those conditions that by their terms are to be satisfied at the Closing, but subject to the satisfaction or waiver (if permissible under applicable Laws) of such conditions), or at such other time and place as the parties may agree in writing. The Closing Date shall be the date upon which the Closing occurs. The Closing shall be deemed to have occurred and shall be effective as of 12:01 a.m., prevailing Eastern Time, on the Closing Date. In addition to the deliverables described in Sections 2.1 and 2.2, at the Closing, each of MetroPCS and DT shall deliver or cause to be delivered (a) duly executed counterpart signatures to each Ancillary Agreement to which it or any of its Subsidiaries is a party and (b) each of the certificates and other documents contemplated to be delivered by such party or its Subsidiaries pursuant to Article V.

#### **ARTICLE III**

#### **REPRESENTATIONS AND WARRANTIES**

3.1 <u>Representations and Warranties Regarding DT, Global and Holding</u>. DT, Global and Holding hereby represent and warrant to MetroPCS as follows:

(a) <u>Organization and Good Standing</u>. Each of DT, Global and Holding has been duly organized, is validly existing and is in good standing under the Laws of Germany. Prior to the date hereof, MetroPCS has been provided with complete and correct copies of DT s, Global s and Holding s Organizational Documents.

(b) <u>Authorization</u>. Each of DT, Global, Holding and TMUS has all requisite power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the Transaction. Holding has all requisite power and authority to sell the TMUS Shares, subject to the filings and actions referred to in Section 3.2(d). The execution and delivery by DT, Global, Holding and TMUS of this Agreement, the performance of their respective obligations hereunder and the consummation by DT, Global, Holding and TMUS of the Transaction have been duly authorized by all necessary action of DT, Global, Holding and TMUS. This Agreement has been duly executed and delivered by DT, Global, Holding and TMUS and, assuming the due authorization, execution and delivery of this Agreement by MetroPCS, constitutes the legal, valid and binding obligation of DT, Global, Holding and TMUS, enforceable against DT, Global, Holding and TMUS in accordance with its terms, except as limited by applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equitable principles. No authorization by the stockholders of DT, Global, Holding or TMUS is required to consummate the Transaction.

#### (c) Governmental Filings; No Conflicts.

(i) Other than the reports, filings, registrations, consents, approvals, permits, waivers, petitions for declaratory ruling, authorizations and/or notices (A) under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the <u>HSR Act</u>), (B) with, by, from or to the Federal Communications Commission in the United States (such agency or any federal successor agency having similar regulatory powers, the <u>FCC</u>) pursuant to the Communications Act of 1934, as amended and further amended from time to time (the <u>Communications Act</u>), (C) pursuant to any applicable state or territorial public utility Laws and rules, regulations and orders of any state or territorial public utility commissions (<u>PUCs</u>) or similar foreign public utility Laws and rules, regulations and orders of any regulatory bodies regulating telecommunications businesses, in respect of the jurisdictions set forth on Schedule 3.1(c) of the TMUS Disclosure Letter, or (D) required with respect to the Committee on Foreign Investment in the United States under 31 C.F.R Part 800, no material notices, reports or other filings are required to be made or effected by DT, Global, or Holding with, nor are any material consents, regulatory authority, agency, commission, body or other governmental entity

(<u>Governmental Entity</u>) in connection with the execution and delivery of this Agreement by DT, Global, Holding and TMUS, the performance of their respective obligations hereunder or the consummation of the Transaction.

(ii) The execution and delivery of this Agreement by each of DT, Global and Holding, the performance of its respective obligations hereunder and the consummation of the Transaction will not constitute or result in (i) a breach or violation of, or a default under, the Organizational Documents of DT, Global or Holding or any of their respective Subsidiaries (other than TMUS and its Subsidiaries) or (ii) a breach or violation of, a termination (or right of termination) or default under, the creation or acceleration of any obligations under, the change of any rights of DT, Global or Holding or any of their respective Subsidiaries (other than TMUS and its Subsidiaries) under, or the creation of an Encumbrance on, any of the assets of DT, Global or Holding or any of their respective Subsidiaries (other than TMUS and its Subsidiaries) (with or without notice, lapse of time or both) pursuant to any agreement, lease, license, contract, note, mortgage, credit agreement, security agreement, indenture, arrangement or other obligation of, or binding on, DT, Global or Holding or any of their respective Subsidiaries (other than TMUS and its Subsidiaries) (other than TMUS and its Subsidiaries), or (iii) conflict with, breach or violate any Law applicable to DT, Global or Holding or any of their respective Subsidiaries (other than TMUS and its Subsidiaries) or by which its or by which any of their properties are bound or affected, except, in the case of clauses (ii) or (iii) above, for any such breach, violation, termination, default, creation, acceleration or change that would not, individually or in the aggregate, reasonably be likely to prevent, materially delay or materially impair the ability of DT, Global or Holding to consummate the Transaction.

#### (d) Ownership of Global, Holding and TMUS Shares.

(i) DT is the record and beneficial owner of, and has good and valid title to, all of the issued and outstanding shares of capital stock of Global, and Global is the record and beneficial owner of, and has good and valid title to, all of the issued and outstanding shares of capital stock of Holding. Holding is the record and beneficial owner of, and has good and valid title to, the TMUS Shares, free and clear of any Encumbrances, and consummation of the Transaction will vest good and valid title to the TMUS Shares in MetroPCS, free and clear of any Encumbrances.

(ii) There are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements or commitments of any character under which DT, Global or Holding is or may become obligated to sell, or giving any Person a right to acquire, or in any way dispose of, any of the TMUS Shares or any securities or obligations exercisable or exchangeable for, or convertible into, any of the TMUS Shares, and no securities or obligations evidencing such rights are authorized, issued or outstanding. The TMUS Shares are not subject to any voting trust agreement or other contract, agreement or arrangement restricting or otherwise relating to the voting, dividend rights or disposition of TMUS Shares.

(e) <u>Brokers and Finders</u>. Neither DT, Global, Holding or TMUS, nor any of DT s, Global s, Holding s or TMUS s respective officers, directors or employees, has employed any broker or finder for which DT (or a Subsidiary of DT other than TMUS and its Subsidiaries) is not solely responsible for such broker s or finder s fees or incurred any Liability for any brokerage fees, commissions or finder s fees in connection with the Transaction for which DT (or a Subsidiary of DT other than TMUS and its Subsidiaries) is not solely.

(f) <u>Licenses</u>. Schedule 3.1(f) of the TMUS Disclosure Letter sets forth a true and complete list, as of the date hereof, of all Licenses from the FCC or any PUC held by DT and its Subsidiaries (other than TMUS and its Subsidiaries).

(g) <u>Ownership of MetroPCS Common Stock</u>. (i) Neither DT nor any of its Affiliates has been, at any time during the three years preceding the date hereof, an interested stockholder of MetroPCS as defined in Section 203 of the DGCL, and (ii) as of the date hereof, DT and its Affiliates do not Beneficially Own any shares of MetroPCS Common Stock.

(h) <u>Financing</u>. The provisions of Sections 4.13(b), 4.13(c) and 4.13(e) constitute the legal, valid and binding obligation of DT (except as limited by applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equitable principles). There are no conditions precedent related to the funding of the full amount of the DT Notes, the Additional DT Notes and the TMUS Working Capital Facility under Sections 4.13(b), 4.13(c) and 4.13(e). Subject to the occurrence of the Closing Date, the aggregate proceeds of the DT Notes and any such Additional DT Notes issued pursuant to Section 4.13(b) and 4.13(c), together with other financial resources of MetroPCS and TMUS including cash, cash equivalents and marketable securities of MetroPCS, TMUS and their respective Subsidiaries on the Closing Date, are expected to be sufficient to consummate the Transaction upon the terms contemplated by this Agreement and pay all related fees and expenses.

(i) <u>No Other Representations or Warranties</u>. Except for the representations and warranties contained in Sections 3.3 and 7.13 and any representation contained in any certificate delivered pursuant to Section 5.3, DT, Global and Holding acknowledge that neither MetroPCS, nor any Subsidiary of MetroPCS or any other Person on behalf of MetroPCS (including any Representative of MetroPCS), makes any express or implied representation or warranty with respect or relating to MetroPCS, any of its Subsidiaries, or any information provided to DT or any other Person, or DT s use of any such information, including any information, documents, projections, forecasts or other material made available to DT in certain data rooms or management presentations in expectation of the Transaction, and neither DT, Global nor Holding has relied on such information or any other representation or warranty not set forth in this Agreement.

3.2 <u>Representations and Warranties Regarding TMUS and its Subsidiaries</u>. Except as set forth in the corresponding sections of the disclosure letter delivered to MetroPCS by TMUS on or prior to entering into this Agreement (the <u>TMUS Disclosure Letter</u>) (it being agreed that disclosure of any item in any part of the TMUS Disclosure Letter shall be deemed disclosure with respect to any other part to which the relevance of such item is reasonably apparent on its face notwithstanding the omission of a reference or cross-reference thereto) or as specifically identified in the notes to the TMUS Financial Statements, TMUS hereby represents and warrants to MetroPCS as follows:

(a) <u>Organization. Good Standing and Qualification</u>. Each of TMUS and its Subsidiaries is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite power and authority to own and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing in each jurisdiction where the ownership or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so qualified or in good standing or to have such power and authority would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. Prior to the date hereof, MetroPCS has been provided with complete and correct copies of each of TMUS s and its Subsidiaries Organizational Documents, and each as so delivered is in full force and effect.

#### (b) Capitalization.

(i) The authorized capital stock of TMUS consists solely of 500,000,000 shares of common stock, \$0.00001 par value per share (<u>TMUS</u> <u>Common Stock</u>), of which 292,669,971 shares are issued and outstanding, and 10,000,000 shares of preferred stock, \$0.001 par value per share, of which no shares are issued and outstanding, and as of the close of business on October 1, 2012, TMUS had no other shares of TMUS Common Stock or TMUS Preferred Stock reserved or otherwise subject to issuance. The TMUS Shares represent all shares of TMUS Common Stock issued and outstanding. All of the outstanding shares of TMUS Common Stock (A) have been duly authorized and validly issued, (B) are fully paid and nonassessable, and (C) were issued in compliance with all applicable Laws concerning the issuance of securities and not in violation of any preemptive rights, purchase option, call, right of first refusal or any similar right. There are no other Equity Interests of TMUS issued, authorized or outstanding.

(ii) There are no bonds, debentures, notes or other indebtedness having general voting rights, or convertible into securities having such rights (<u>Voting De</u>bt), of TMUS issued or outstanding. There are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements or commitments of any character under which TMUS is or may become obligated to issue or sell, or giving any Person a right to subscribe for or acquire, or in any way dispose of, any Equity Interests or Voting Debt of TMUS or any securities or obligations exercisable or exchangeable for, or convertible into, any Equity Interests or Voting Debt of TMUS, and no securities or obligations evidencing such rights are authorized, issued or outstanding.

(iii) The outstanding shares of TMUS Common Stock are not subject to any voting trust agreement or other contract, agreement or arrangement to which TMUS or any of its Affiliates is a party restricting or otherwise relating to the voting, dividend rights or disposition of such outstanding shares (other than TMUS Permitted Encumbrances). There are no phantom stock or similar rights providing economic benefits based, directly or indirectly, on the value or price of the Equity Interests of TMUS.

(iv) Schedule 3.2(b)(iv) of the TMUS Disclosure Letter lists all Indebtedness described in clauses (i) and (iii) of the definition of Indebtedness. There are no purchase money security interests for which liens have been filed, in each case created, issued, assumed, guaranteed or permitted to exist by each of TMUS and its Subsidiaries as of the date hereof, that are material to TMUS and its Subsidiaries, taken as a whole.

#### (c) Subsidiaries.

(i) A true and complete list of the Subsidiaries of TMUS as of the date hereof is set forth on Schedule 3.2(c)(i) of the TMUS Disclosure Letter, and such list sets forth, with respect to each such Subsidiary, as of the date hereof (A) its jurisdiction of organization or formation and (B) the direct or indirect ownership interest of TMUS in such Subsidiary, as well as the ownership interest of any other Person in such Subsidiary if it is not wholly-owned, directly or indirectly, by TMUS. TMUS does not own, directly or indirectly, any Equity Interests in any Person that is not a Subsidiary of TMUS.

(ii) All of the Equity Interests of each Subsidiary of TMUS owned beneficially or of record, directly or indirectly, by TMUS are owned free and clear of any Encumbrances (other than TMUS Permitted Encumbrances). No Subsidiary of TMUS has any Voting Debt issued or outstanding. There are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements or commitments of any character under which any Subsidiary of TMUS is or may become obligated to issue or sell, or giving any Person a right to subscribe for or acquire, or in any way dispose of, any Equity Interests or Voting Debt of any Subsidiary of TMUS or any securities or obligations exercisable or exchangeable for, or convertible into, any Equity Interests or Voting Debt of any Subsidiary of TMUS, and no securities or obligations evidencing such rights are authorized, issued or outstanding.

(iii) All of the outstanding Equity Interests of the Subsidiaries of TMUS have been duly authorized and are validly issued, fully paid and nonassessable. The outstanding Equity Interests of each Subsidiary of TMUS are not subject to any voting trust agreement or other contract, agreement or arrangement restricting or otherwise relating to the voting, dividend rights or disposition of such Equity Interests. There are no phantom stock or similar rights providing economic benefits based, directly or indirectly, on the value or price of the Equity Interests of any Subsidiary of TMUS.

## (d) Governmental Filings; No Conflicts.

(i) Other than the reports, filings, registrations, consents, approvals, permits, waivers, petitions for declaratory ruling, authorizations and/or notices (A) under the HSR Act, (B) with, by, from or to the FCC pursuant to the Communications Act, (C) pursuant to any applicable state or territorial public utility Laws and rules, regulations and orders of any PUCs or similar foreign public utility Laws and rules, regulations and orders of any regulatory bodies regulating telecommunications businesses, in respect of the jurisdictions

set forth on Schedule 3.2(d) of the TMUS Disclosure Letter, or (D) required with respect to the Committee on Foreign Investment in the United States under 31 C.F.R Part 800, no material notices, reports or other filings are required to be made or effected by TMUS or its Subsidiaries with, nor are any material consents, registrations, approvals, permits or authorizations required to be obtained by TMUS or its Subsidiaries from, any Governmental Entity in connection with the execution and delivery of this Agreement by TMUS, the performance of its obligations hereunder or the consummation of the Transaction.

(ii) The execution and delivery of this Agreement by TMUS, the performance of its obligations hereunder and the consummation of the Transaction will not constitute or result in (A) a breach or violation of, or a default under, the Organizational Documents of TMUS or of any of its Subsidiaries, (B) a breach or violation of, a termination (or right of termination) or default under, the creation or acceleration of any obligations under, the change of any rights of TMUS or any of its Subsidiaries under, or the creation of an Encumbrance (other than a TMUS Permitted Encumbrance) on, any of the assets of TMUS or any of its Subsidiaries (with or without notice, lapse of time or both) pursuant to any agreement, lease, license, contract, note, mortgage, credit agreement, security agreement, indenture, arrangement or other obligation binding upon TMUS or any of its properties are bound or affected, except, in the case of clause (B) or (C) above, for any such breach, violation, termination, default, creation, acceleration or change that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect.

#### (e) Financial Statements; Undisclosed Liabilities.

(i) Prior to the date hereof, MetroPCS has been provided with complete and correct copies of the audited consolidated statements of operations and comprehensive income, changes in stockholders equity and cash flows of TMUS and its Subsidiaries for the fiscal years ended December 31, 2009, 2010 and 2011 and consolidated balance sheets of TMUS and its Subsidiaries as of such dates, and the unaudited consolidated statements of operations and comprehensive income, changes in stockholders equity and cash flows of TMUS and its Subsidiaries for the six-month period ended June 30, 2012 and a consolidated balance sheet of TMUS and its Subsidiaries as of such date (the <u>TMUS</u> <u>Financial Statements</u>). The TMUS Financial Statements (A) have been prepared in accordance with GAAP applied on a consistent basis during the periods involved, except as may be noted therein or in the notes thereto, (B) present fairly, in all material respects, the consolidated financial position of TMUS and its Subsidiaries for the periods then ended and (C) accurately reflect in all material respects the books of account and other financial records of TMUS and its Subsidiaries.

(ii) Neither TMUS nor any of its Subsidiaries has any Liabilities except for (A) Liabilities reflected or reserved against on the balance sheet as of December 31, 2011 included in, or otherwise disclosed in, the TMUS Financial Statements and not heretofore paid or discharged, (B) Liabilities incurred since December 31, 2011 in the ordinary course of business consistent with past practice or (C) Liabilities that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. All of the Liabilities of TMUS and its Subsidiaries arose, directly or indirectly, out of or in connection with the assets used in, or the operations of, the business of TMUS and its Subsidiaries (including products, services, assets and operations ancillary thereto).

(iii) TMUS has designed and maintains a system of internal controls over financial reporting that provides reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP and provides reasonable assurance that fraud is detected and prevented.

(iv) Since December 31, 2009, to the Knowledge of TMUS, no attorney representing TMUS or any of its Subsidiaries, whether or not employed by TMUS or any of its Subsidiaries, has reported to TMUS s chief

legal officer, chief executive officer, audit committee or the TMUS Board evidence of a material violation of securities laws, material breach of fiduciary duty or similar material violation of law, relating to periods after December 31, 2009.

(f) <u>Litigation</u>. As of the date hereof, there is no civil, criminal or administrative action, suit, demand, claim, hearing, proceeding or investigation pending or, to the Knowledge of TMUS, threatened, against TMUS or any of its Subsidiaries, except those that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. Neither TMUS nor any of its Subsidiaries is a party to, or subject to the provisions of, any judgment, order, writ, injunction, decree or award of any Governmental Entity that would, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. No representation or warranty is made in this Section 3.2(f) with respect to Tax matters, which shall be governed exclusively by Section 3.2(g) (Employee Benefits) and 3.2(l) (Taxes), or environmental matters, which shall be governed exclusively by Section 3.2(k) (Environmental Matters).

#### (g) Employee Benefits.

(i) All benefit and compensation plans, contracts, agreements, policies or arrangements sponsored or contributed to by TMUS or any of its Subsidiaries Subsidiaries, including those covering any of its past or present employees, officers or directors (or for which TMUS or any of its Subsidiaries could have any liability), including employee benefit plans within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended (<u>ERISA</u>), and employment agreements, collective bargaining agreements, deferred compensation, change of control, retention, stock option, stock purchase, restricted stock, stock appreciation, phantom share, stock based, incentive, severance and bonus plans (other than any immaterial benefit plans) (the <u>TMUS Benefit Plans</u>) in effect as of the date hereof are listed on Schedule 3.2(g) of the TMUS Disclosure Letter. True and complete copies of all TMUS Benefit Plans listed on Schedule 3.2(g) of the TMUS Disclosure Letter, and of all related material funding documents, actuarial and financial reports, government correspondence, summary plan descriptions, annual reports, IRS determination letters, and audit reports have been provided or made available to MetroPCS prior to the date hereof.

(ii) Each TMUS Benefit Plan was established and, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect, has been documented, maintained and administered in compliance in all respects with the terms thereof and the applicable requirements of ERISA, the Internal Revenue Code of 1986, as amended (the <u>Code</u>), and any other applicable Law. Each TMUS Benefit Plan that is intended to be qualified under Section 401(a) of the Code has received a favorable determination or opinion letter from the Internal Revenue Service (the <u>IRS</u>) with respect to its qualified status under Section 401(a) of the Code or has pending or has time remaining in which to file an application for such determination from the IRS (or TMUS and its Subsidiaries are entitled to rely on a favorable opinion or advisory letter issued by the IRS in accordance with Revenue Procedure 2005-16 with respect to the qualified status of the plan document), and, to the Knowledge of TMUS, there is no fact or circumstance that exists that would, individually or in the aggregate, reasonably be likely to adversely affect or give rise to the revocation of such qualified status. All contributions required to be made under the terms of any TMUS Benefit Plan or applicable Law (including all employer contributions and employee salary reduction contributions) have been timely made or are reflected in the TMUS Financial Statements as of the dates thereof. All unfunded liabilities, if any, under the TMUS Benefit Plans are fully reflected in the TMUS Financial Statements. No event has occurred and no condition exists that would, individually or in the aggregate, reasonably be IKRS or the Code or other applicable Law in respect of any TMUS Benefit Plan.

(iii) Neither TMUS nor any ERISA Affiliate maintains or contributes to or has within the past six complete calendar years maintained or contributed to, or been required to contribute to, or otherwise has any direct or indirect liability with respect to, an employee pension benefit plan (within the meaning of

Section 3(2) of ERISA) that is subject to Title IV of ERISA, is a multiemployer plan (within the meaning of Section 3(37) of ERISA) or is a multiple employer plan (within the meaning of Section 4063 of ERISA or Section 413(c) of the Code). Except as required by the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, or similar state or local law or as is reflected on the TMUS Financial Statements, neither TMUS nor any ERISA Affiliate is obligated to provide any retiree health or life insurance benefits to any employee or former employees of TMUS or any ERISA Affiliate. Each TMUS Benefit Plan can be amended or terminated at any time without liability, other than liability for benefits accrued as of the date of any such amendment or termination.

(iv) Excluding routine, uncontested claims for benefits under any TMUS Benefit Plan and except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect, (A) there is no action, suit, demand, audit, claim, hearing, proceeding or investigation pending against or involving or, to the Knowledge of TMUS, threatened, against or involving any TMUS Benefit Plan before any court or arbitrator or any Governmental Entity, or federal, state or local official that would, individually or in the aggregate, reasonably be likely to subject TMUS or any of its Subsidiaries to a liability, except those first arising after the date hereof in the ordinary course of business, and (B) to the Knowledge of TMUS, there are no facts or circumstances existing that would, individually or in the aggregate, reasonably be likely to give rise to such actions, suits, demands, audits, claims, hearings or proceedings.

(v) No agreement, binding commitment or obligation exists to materially increase benefits under, or adopt any new, TMUS Benefit Plan. There is no provision of any TMUS Benefit Plan or other related contract or agreement, and there has been no amendment to or announcement by, TMUS or any of its Subsidiaries relating to, or change in employee participation or coverage under, any TMUS Benefit Plan that would increase materially the expense of maintaining such plan above the level of the expense incurred therefor for the most recent fiscal year.

(vi) Neither the execution of this Agreement nor the consummation of the Transaction will (whether alone or in connection with any other related event(s)): (A) entitle any employee of TMUS or any of its Subsidiaries to severance pay or any increase in severance pay (or other compensation or benefits) upon any termination of employment; (B) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of any compensation or benefits under, or increase the amount payable pursuant to, or result in the deemed satisfaction of vesting conditions, goals or any other requirements or conditions under, any of the TMUS Benefit Plans; (C) limit or restrict the right of TMUS or any of its Subsidiaries or, after the consummation of the Transaction, MetroPCS or any of its Subsidiaries, to merge, amend or terminate any of the TMUS Benefit Plans (other than solely pursuant to applicable Law); or (D) result in the creation, increase, forgiveness, extension or modification of any loan to any employee, officer or director of TMUS or any of its Subsidiaries.

(vii) No TMUS Benefit Plan or other TMUS practice provides any Person with any amount of additional compensation if such individual is provided amounts subject to excise or additional taxes imposed under Section 409A or 4999 of the Code.

(viii) Each TMUS Benefit Plan that is a nonqualified deferred compensation plan (as defined under Section 409A(d)(1) of the Code) is, except as would not result in a material liability, in documentary compliance with Section 409A of the Code and the guidance provided thereunder and has been operated and administered in compliance in all material respects with Section 409A of the Code and the guidance provided thereunder.

#### (h) Compliance with Laws; Licenses.

(i) The business of TMUS and its Subsidiaries has been, and is being, conducted in compliance with all federal, state, local and foreign laws, statutes and ordinances, common law and all rules, regulations, guidelines, standards, judgments, orders, writs, injunctions, decrees, arbitration awards, agency requirements, licenses and permits of any Governmental Entity (collectively, <u>Laws</u>), except for violations

that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. No investigation or review by any Governmental Entity with respect to TMUS or any of its Subsidiaries is pending or, to the Knowledge of TMUS, threatened, nor has any Governmental Entity indicated an intention to conduct the same, except for such investigations or reviews that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. Each of TMUS and its Subsidiaries has obtained and is in compliance with all permits, licenses, certifications, approvals, registrations, consents, authorizations, franchises, variances, exemptions and orders issued or granted by a Governmental Entity (<u>Licenses</u>) necessary to conduct its business as presently conducted, except those the absence of which would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect (the <u>TMUS Material Licenses</u>).

(ii) Schedule 3.2(h)(ii) of the TMUS Disclosure Letter sets forth a true and complete list, as of the date hereof, of (A) all TMUS Material Licenses and, to the extent not otherwise constituting TMUS Material Licenses, all Licenses issued or granted to TMUS or any of its Subsidiaries by the FCC and all leases for the use of wireless spectrum licensed by other FCC licensees to TMUS or any of its Subsidiaries (such licenses and leases, the <u>TMUS FCC Licenses</u> ), all Licenses issued or granted to TMUS or any of its Subsidiaries by PUCs regulating telecommunications businesses (<u>TMUS State Licenses</u>), and all Licenses issued or granted to TMUS or any of its Subsidiaries by foreign Governmental Entities regulating telecommunications businesses (collectively with the TMUS Material Licenses, TMUS FCC Licenses and TMUS State Licenses, the <u>TMUS Communications Licenses</u>); (B) all pending applications for issuance, grant, assignment or transfer of Licenses, or applications for leases of wireless spectrum, to TMUS or any of its Subsidiaries, that would be TMUS Communications Licenses if issued or granted; (C) all pending applications for assignment or transfer of Licenses, or applications for leases of wireless spectrum, by TMUS or its Subsidiaries to any Person (other than TMUS or its Subsidiaries); and (D) all pending applications by TMUS or any of its Subsidiaries for modification, extension or renewal of any TMUS Communications License; provided that Schedule 3.2(h)(ii) may exclude point-to-point microwave licenses, business radio licenses, experimental licenses and Section 214 certificates and pending applications regarding the same. Each of TMUS and its Subsidiaries is in compliance with its obligations under each of the TMUS FCC Licenses and the rules and regulations of the FCC, and with its obligations under each of the TMUS State Licenses, in each case, except for such failures to be in compliance with Licenses that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. TMUS and its Subsidiaries are not the subject of, and there are no pending nor, to the Knowledge of TMUS, threatened, proceedings, notices of violation, orders of forfeiture or complaints or investigations relating to TMUS Communications Licenses before the FCC, the Federal Aviation Administration (the <u>FAA</u>), or any other Governmental Entity, in each case, except that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. The FCC actions granting all TMUS Communications Licenses, together with all underlying construction permits, have not been reversed, stayed, enjoined, annulled or suspended, and there is not pending or, to the Knowledge of TMUS, threatened any application, petition, objection or other pleading with the FCC, the FAA or any other Governmental Entity that challenges or questions the validity of or any rights of the holder under any such TMUS FCC License, in each case, except that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect.

(iii) TMUS holds the TMUS Communications Licenses, and the TMUS Communications Licenses are issued in the name of TMUS or one of its Subsidiaries. The TMUS Communications Licenses are in full force and effect, are granted without conditions, except for those conditions on the face of such TMUS Communications Licenses or conditions generally applicable to all similarly situated licenses of comparable spectrum, and are free and clear of all Encumbrances (other than TMUS Permitted Encumbrances) or any restrictions which might, individually or in the aggregate, limit the full operation of the TMUS Communications Licenses in any material respect.

(iv) All of the currently operating cell sites and microwave paths of TMUS and its Subsidiaries in respect of which a filing with the FCC was required have been constructed and are currently operated as represented to the FCC in currently effective filings, and modifications to such cell sites and microwave paths have been preceded by the submission to the FCC of all required filings, in each case, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect.

(v) All transmission towers owned or leased by TMUS and its Subsidiaries are (to the Knowledge of TMUS with respect to leased towers) obstruction-marked and lighted by TMUS or its Subsidiaries to the extent required by, and in accordance with, the rules and regulations of the FAA (the <u>FAA Rules</u>), except that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. Appropriate notification to the FAA has been made for each transmission tower owned or leased by TMUS and its Subsidiaries to the extent required to be made by TMUS or any of its Subsidiaries by, and in accordance with, the FAA Rules, in each case, except that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect.

(vi) Neither TMUS nor any of its Subsidiaries holds any TMUS Communications Licenses through a partnership, joint venture or other Person that is not a Subsidiary of TMUS.

(vii) TMUS does not hold any License to offer, and does not offer, any services or products other than wireless telecommunications and wireless information services and products, and any ancillary services or products thereto. TMUS and its Subsidiaries do not conduct any business other than the Business.

(viii) TMUS and its Subsidiaries are fully qualified under the Communications Act and the rules and regulations of the FCC to hold the TMUS FCC Licenses generally. To the Knowledge of TMUS, there are no facts or circumstances relating to the qualifications of TMUS and its Subsidiaries that would prevent or materially delay the grant of any FCC Form 603 application (or other appropriate form) under the FCC Rules and the Communications Act with respect to the Transaction.

(ix) No representation or warranty is made in this Section 3.2(h) with respect to Tax matters, which shall be governed exclusively by Sections 3.2(g) (Employee Benefits) and 3.2(l) (Taxes), or environmental matters, which shall be governed exclusively by Section 3.2(k) (Environmental Matters).

(i) <u>Absence of Certain Changes</u>. Since December 31, 2011 and prior to the date hereof, other than expenses and capital expenditures incurred or made in accordance with the TMUS Business Plan, TMUS and its Subsidiaries have conducted their respective businesses only in, and have not engaged in any material transaction other than in accordance with, the ordinary course of such businesses, and there has not been any:

(i) Circumstance (including any adverse change with respect to any Circumstance existing on or prior to December 31, 2011) that, individually or in the aggregate, has had or would reasonably be likely to have a TMUS Material Adverse Effect;

(ii) merger or consolidation between TMUS or any of its Subsidiaries with any other Person, or any restructuring, reorganization or complete or partial liquidation or similar transaction, or the entry into any agreements or arrangements imposing material changes or restrictions on its assets, operations or businesses, except for any such transactions among wholly-owned Subsidiaries of TMUS;

(iii) acquisition of assets outside of the ordinary course of business consistent with past practice for consideration in excess of \$50,000,000 individually, other than in accordance with the TMUS Business Plan;

(iv) creation or incurrence of any Encumbrance (other than TMUS Permitted Encumbrances) on (x) any TMUS FCC Licenses or (y) on the other assets of TMUS or its Subsidiaries that are, individually or in the aggregate, material to TMUS or any of its Subsidiaries;

(v) loan, advance, guarantee or capital contribution to, or investment in, any Person (other than any of the foregoing to or on behalf of TMUS or any direct or indirect wholly-owned Subsidiary of TMUS and

other than loans or advances to employees and contractors in the ordinary course of business consistent with past practices in an amount not to exceed \$250,000 individually);

(vi) material damage, destruction or other casualty loss with respect to any material asset, or TMUS Owned Real Property, TMUS Leased Real Property or property otherwise used by TMUS or any of its Subsidiaries, whether or not covered by insurance;

(vii) declaration, setting aside or payment of any non-cash distribution with respect to any Equity Interests of TMUS or any of its Subsidiaries (except for distributions by any direct or indirect wholly-owned Subsidiary of TMUS to TMUS or any other such Subsidiary of TMUS);

(viii) incurrence of any Indebtedness for borrowed money other than from DT or any of its wholly-owned Subsidiaries;

(ix) material change in any method of financial accounting or accounting practice by TMUS or any of its Subsidiaries, except for any such change required by changes in GAAP or applicable Law;

(x) increase in the compensation payable or to become payable to its officers or employees (except for increases in the ordinary course of business consistent with past practice);

(xi) fundamental change to any of the important elements of the network technologies or principal billing systems of TMUS and its Subsidiaries (excluding, for the avoidance of doubt, system upgrades, improvements and modernization, equipment replacement and similar matters, in each case within the same fundamental framework of such network technologies and billing systems); or

(xii) agreement to do any of the foregoing.

(j) <u>Insurance</u>. All material fire and casualty, general liability, business interruption, product liability and sprinkler and water damage insurance policies (<u>Insurance Policies</u>) maintained by TMUS or any of its Subsidiaries, together with adequately capitalized self-insurance arrangements, provide adequate coverage for all normal risks incident to the business of TMUS and its Subsidiaries and their respective properties and assets, except for any such failures to maintain such Insurance Policies that would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. As of the date hereof, neither TMUS nor any of its Subsidiaries has received any written notice of cancellation of any material Insurance Policy.

(k) Environmental Matters. Except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect:

(i) since the date that is four years prior to the date hereof, TMUS and its Subsidiaries have been in compliance with all applicable Environmental Laws and have not incurred any Liabilities concerning any Environmental Laws with respect to the business of TMUS and its Subsidiaries;

(ii) there are no writs, injunctions, decrees, awards, orders or judgments outstanding, or any actions, suits, demands, claims, hearings, proceedings or investigations pending or, to the Knowledge of TMUS, threatened, relating to compliance with, or Liability under, any Environmental Law affecting the business of TMUS and its Subsidiaries, other than those first arising after the date hereof in the ordinary course of business;

(iii) to the Knowledge of TMUS, there has been no release, threatened release, contamination or disposal of Hazardous Substances at any property currently or formerly owned or operated in connection with the business of TMUS and its Subsidiaries (including in soils, groundwater, surface water, buildings or other structures) or at any third-party property, or from any waste generated by TMUS or any of its Subsidiaries or any legally responsible predecessor corporation thereof, that has given or would, individually or in the aggregate, reasonably be likely to give rise to any Liability under any Environmental Law for which TMUS or any of its Subsidiaries would incur or share Liability; and

(iv) there are no consent decrees, orders or similar agreements with any Governmental Entity imposing restrictions on the ownership, use or transfer of any real property relating to, or derived from, any Environmental Law, and there are no indemnification or other agreements with any third party (other than ordinary course provisions in leases of real property or in agreements for the acquisition or disposition of assets or businesses) relating to any Liability or potential Liability under any Environmental Law.

(1) <u>Taxes</u>. Except as would not have and would not reasonably be expected to have, individually or in the aggregate, a TMUS Material Adverse Effect:

(i)(A) All Tax Returns required to be filed by, or on behalf of, or with respect to, TMUS and each of its Subsidiaries have been timely filed (taking into account extensions) with the appropriate Taxing Authority and all such Tax Returns are true and complete, and (B) TMUS has, or has caused each of its Subsidiaries to, duly and timely pay all Taxes due and payable, including Taxes required to be withheld from amounts owing to any Person, except in each case of clauses (A) and (B), with respect to matters contested in good faith or for which adequate reserves have been established, in accordance with GAAP, in the most recent TMUS Financial Statements, as adjusted to reflect operations in the ordinary course of business since the date thereof.

(ii) All deficiencies or assessments made in writing as a result of any audit, examination or investigation by any Taxing Authority of Tax Returns of TMUS and its Subsidiaries that are due and payable have been fully paid, and no other audits, examination or investigations by any Taxing Authority relating to any Tax Returns of TMUS and its Subsidiaries are in progress. Neither TMUS nor any of its Subsidiaries has received written notice from any Taxing Authority of the commencement of any audit, examination or investigation not yet in progress. There is no action, suit, demand, claim, hearing or, to the Knowledge of TMUS, proceeding, relating to Taxes pending or, to the Knowledge of TMUS, threatened, against TMUS or any of its Subsidiaries.

(iii) Neither TMUS nor any of its Subsidiaries is a party to any Tax indemnification, Tax allocation or Tax sharing agreement pursuant to which TMUS or any of its Subsidiaries, as applicable, will have any obligation to make any payments after the Closing Date, other than (A) any agreements solely among TMUS and/or its Subsidiaries and (B) Tax provisions in loan agreements, leases, license agreement and other commercial agreements the principal purpose of which does not relate to Taxes. Neither TMUS nor any of Subsidiaries is or could be liable for Taxes of any Person (other than of a member of the affiliated group for United States federal income tax purposes of which TMUS or any of its Subsidiaries is or was the common parent) (x) under Section 1.1502-6 of the Treasury Regulations (or any similar provision of state, local or foreign Law), (y) as a transferee or successor, or (z) otherwise, for any taxable period (or portion thereof) ending on or before the Closing Date for which the applicable statute of limitations (including extensions) is not closed.

(iv) In the past five years, TMUS has not received any IRS private letter ruling or entered into any closing agreements within the meaning of Section 7121 of the Code relating to or with respect to the income and/or assets of TMUS or any of its Subsidiaries. There are no pending requests by TMUS or any of its Subsidiaries for an IRS private letter ruling.

(v) Neither TMUS nor any of its Subsidiaries will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any (A) change in method of accounting for a taxable period ending on or prior to the Closing Date, (B) closing agreement (within the meaning of Section 7121 of the Code or any corresponding or similar provision of state, local, or non-United States income Tax law) entered into on or prior to the Closing Date, (C) installment sale or open transaction disposition made on or prior to the Closing Date, or (D) prepaid amount received on or prior to the Closing Date.

(vi) There are no Encumbrances for Taxes upon any assets of TMUS or any of its Subsidiaries other than TMUS Permitted Encumbrances.

(vii) Within the preceding three years, no written claim has been received by TMUS or any of its Subsidiaries from a Taxing Authority in any jurisdiction where TMUS or any of its Subsidiaries does not file Tax Returns asserting that TMUS or any of its Subsidiaries is or may be subject to taxation by such jurisdiction.

(viii) Neither TMUS nor any of its Subsidiaries has granted any currently effective waiver, extension or comparable consent regarding the application of the statute of limitations with respect to any Taxes or Tax Return, nor has any request for any such waiver, extension or consent been made.

(ix) Within the preceding three years, neither TMUS nor any of its Subsidiaries has been a distributing or controlled corporation in a transaction intended or purported to be governed by Section 355 of the Code.

(x) Neither TMUS nor any of its Subsidiaries has participated in a listed transaction within the meaning of Section 1.6011-4(b)(2) of the Treasury Regulations.

(xi) TMUS is the common parent of an affiliated group, as defined in Section 1504 of the Code, that files a consolidated return for United States federal income tax purposes.

(xii) Neither TMUS nor any of its Subsidiaries is a party to any agreement, contract, undertaking, commitment, arrangement or plan that would result, and neither the execution of this Agreement nor the consummation of the Transaction (whether alone or in connection with any other related event(s)) will result, separately or in the aggregate, in the payment of any excess parachute payment within the meaning of Code Section 280G.

#### (m) Labor Matters.

(i) None of TMUS or its Subsidiaries is party to or otherwise bound by, and, as of the date hereof, none of DT or its Affiliates, including TMUS and its Subsidiaries, are proposing, offering or negotiating to enter into or adopt, any labor and collective bargaining agreements, contracts or other agreements or understandings with a labor union or labor organization relating to, affecting, or in any way binding on TMUS and its Subsidiaries.

(ii) As of the date hereof, neither TMUS nor any of its Subsidiaries is the subject of any proceeding, nor is any proceeding pending, or to the Knowledge of TMUS, threatened asserting that it has committed any material unfair labor practice or seeking to compel it to bargain with any labor union or labor organization.

(iii) There has not been since December 31, 2009, and as of the date hereof, there is not pending or, to the Knowledge of TMUS, threatened any material labor strike, dispute, walk-out, work stoppage, slow-down, union activity, picketing, lockout or other similar occurrence by employees of TMUS or its Subsidiaries.

(iv) TMUS and its Subsidiaries have complied in all material respects with all Laws relating to labor and employment, including those relating to wages, hours, collective bargaining, meals and rest times, unemployment compensation, worker s compensation, equal employment opportunity, age and disability discrimination, immigration control, employee classification, information privacy and security, payment and withholding of taxes, and continuation coverage with respect to group health plans.

(v) Since January 1, 2009, neither TMUS nor any of its Subsidiaries has engaged in any plant closing or mass layoff, as defined in the Worker Adjustment Retraining and Notification Act or any comparable state or local law (collectively, the <u>WARN Act</u>), without complying in all material respects with the notice requirements of the WARN Act.

#### (n) Intellectual Property.

(i) All Intellectual Property owned or held exclusively by TMUS and its Subsidiaries (<u>TMUS Owned Intellectual Property</u>) is exclusively owned or held (beneficially and of record, where applicable) by TMUS

or one of its Subsidiaries, free and clear of all Encumbrances (other than TMUS Permitted Encumbrances), and is not subject to any open source or similar license agreement or distribution model, or to any commitments to any standards-setting or similar organization, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. The TMUS Owned Intellectual Property is valid, subsisting and enforceable, and is not subject to any outstanding order, judgment, decree or agreement adversely affecting TMUS s or its Subsidiaries use of, or their rights to, such Intellectual Property, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect.

(ii) To the Knowledge of TMUS, TMUS and its Subsidiaries have sufficient rights to use all material Intellectual Property used in, or necessary for the conduct of the Business as presently conducted, all of which rights shall survive the consummation of the Transaction unchanged in all respects material to TMUS and its Subsidiaries, taken as a whole. TMUS and its Subsidiaries have taken commercially reasonable measures to protect the TMUS Owned Intellectual Property, and to protect the confidentiality of all Trade Secrets that are owned, used or held for use by TMUS and its Subsidiaries. TMUS and each of its Subsidiaries maintains a policy requiring that upon their hire, employees of TMUS and its Subsidiaries execute confidentiality and intellectual property assignment agreements which prohibit such employees from disclosing TMUS s and its Subsidiaries. Trade Secrets and confidential information without the written approval of an officer of TMUS and which assign to TMUS all Intellectual Property rights developed by such employees during the course of their employment with TMUS or its Subsidiaries.

(iii) Neither TMUS nor any of its Subsidiaries has infringed, misappropriated or otherwise violated the Intellectual Property rights of any third party in the past six years, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. There is no litigation, opposition, cancellation, proceeding, objection or claim pending, asserted or, to the Knowledge of TMUS, threatened, against TMUS or any of its Subsidiaries concerning the ownership, validity, registrability, enforceability, infringement or use of, or licensed right to use, any Intellectual Property, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. To the Knowledge of TMUS, no Person is infringing, misappropriating or otherwise violating any TMUS Owned Intellectual Property right of TMUS or its Subsidiaries in any respect material to TMUS and its Subsidiaries, taken as a whole.

(iv) The material IT Assets used by TMUS or any of its Subsidiaries operate and perform as needed by TMUS and its Subsidiaries to adequately conduct their respective businesses as presently conducted and, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect, the data therein have not been subject to unauthorized access by any Person.

(o) <u>Contracts</u>. (i) Schedule 3.2(o)(i) of the TMUS Disclosure Letter lists all Material TMUS Contracts in effect as of the date hereof. The term <u>Material TMUS Contracts</u> means all of the following types of TMUS Contracts (other than TMUS Contracts solely among or between TMUS and/or its direct or indirect wholly-owned Subsidiaries, Organizational Documents of TMUS and its Subsidiaries, TMUS Benefit Plans or other agreements related to employee benefits and agreements related to labor matters to the extent that such items are provided for in Sections 3.2(g) (Employee Benefits) and 3.2(m) (Labor Matters), respectively):

(A) TMUS Contracts evidencing Indebtedness for borrowed money with a principal amount greater than \$100,000,000;

(B) each TMUS Contract for distribution, supply, inventory, purchase, license or advertising or similar agreement that is reasonably likely to involve consideration of more than \$300,000,000 in the aggregate in any 12-month period, other than any such contract that can be cancelled without penalty or further payment on 90 days or less notice;

(C) TMUS Contracts relating to the acquisition, lease or disposition by TMUS or any of its Subsidiaries of assets and properties for consideration in excess of \$100,000,000, or under which

TMUS or any of its Subsidiaries has any indemnification obligations or any other ongoing obligations that would reasonably be likely to result in payments in excess of \$50,000,000;

(D) TMUS Contracts that are reasonably likely to involve consideration of more than \$300,000,000 in any 12-month period or involved consideration of more than \$300,000,000 in the aggregate during calendar year 2011 or \$600,000,000 in the aggregate over the term of such TMUS Contract;

(E) any TMUS Contract that would reasonably be likely to involve consideration of more than \$50,000,000 in any 12-month period that is an interconnection, bundling or similar agreement in connection with which the equipment, networks and services of TMUS or any of its Subsidiaries are connected to those of another service provider in order to allow their respective customers access to each other s services and networks;

(F) any TMUS Contract that would reasonably be likely to involve consideration of more than \$100,000,000 in any 12-month period that is an agency, dealer, reseller, franchise or other similar contract (except for those that are terminable, without penalty, on 90 days or less notice);

(G) roaming TMUS Contracts that would reasonably be likely to involve payment by or expense to TMUS or any of its Subsidiaries of more than \$100,000,000 in any 12-month period; and

(H) TMUS Contracts that would reasonably be likely to involve consideration of more than \$50,000,000 in any 12-month period pursuant to which TMUS or any of its Subsidiaries licenses Intellectual Property to or from any Person.

(ii) Schedule 3.2(o)(ii) of the TMUS Disclosure Letter lists all Restricted TMUS Contracts in effect as of the date hereof. The term <u>Restricted TMUS Contracts</u> means all of the following types of TMUS Contracts (other than TMUS Contracts solely among or between TMUS and/or its direct or indirect wholly-owned Subsidiaries, Organizational Documents of TMUS and its Subsidiaries, TMUS Benefit Plans or other agreements related to employee benefits and agreements related to labor matters to the extent that such items are provided for in Sections 3.2(g) (Employee Benefits) and 3.2(m) (Labor Matters), respectively):

(A) joint venture, partnership, limited partnership or limited liability company agreements relating to the formation, creation, operation, existence, management or control of any joint venture, partnership, limited partnership or limited liability company that is not wholly-owned, directly or indirectly, by TMUS;

(B) TMUS Contracts that purport to limit in any material respect either the type of business in which TMUS or any of its Subsidiaries, or MetroPCS or its Affiliates (other than TMUS and/or its Subsidiaries) after the Closing (other than any TMUS Contracts between MetroPCS or its Affiliates, on the one hand, and TMUS and its Subsidiaries, on the other hand), may engage or the manner or locations in which any of them may so engage in any business or purport to create any material exclusive relationship;

(C) TMUS Contracts that could require the disposition of any material operations or line of business of TMUS or any of its Subsidiaries;

(D) TMUS Contracts that grant most favored nation status to any third party that paid or received consideration of more than \$50,000,000 in any 12-month period, other than business-to-business service contracts under which no third party paid or received consideration of more than \$100,000,000 in any 12-month period;

(E) any TMUS Contract that grants any right of first refusal, first offer or similar right to any third party that paid or received, or would reasonably be likely to pay or receive, consideration of more than \$50,000,000 during the term of such TMUS Contract;

(F) TMUS Contracts that are requirements contracts or contain volume or purchase commitments that would reasonably be likely to involve consideration of more than \$100,000,000 in any 12-month period;

(G) any TMUS Contract that would reasonably be likely to involve consideration of more than \$100,000,000 in any 12-month period that contains any commitment to (1) provide wireless services coverage in a particular geographic area, (2) build out tower sites in a particular geographic area, or (3) pay for a specified number of minutes of roaming usage of a third party s network regardless of the amount of actual usage (except for those that are terminable, without penalty, on 12 months or less notice);

(H) stock purchase agreements and other TMUS Contracts relating to the pending acquisition, lease or disposition by TMUS or any of its Subsidiaries of any Equity Interest of TMUS or any of its Subsidiaries, except for any such stock purchase agreement or other TMUS Contract relating to the pending acquisition, lease or disposition by TMUS or any of its Subsidiaries of any Equity Interest in any Subsidiary of TMUS that is not wholly-owned, directly or indirectly, by TMUS for consideration of \$10,000,000 or less;

(iii) Prior to the date hereof, MetroPCS has been provided with complete and correct copies of each Material TMUS Contract listed on Schedule 3.2(o)(i) of the TMUS Disclosure Letter and each Restricted TMUS Contract listed on Schedule 3.2(o)(ii) of the TMUS Disclosure Letter, including amendments thereof and exhibits, annexes and schedules thereto. To the Knowledge of TMUS, as of the date hereof, each Material TMUS Contract and each Restricted TMUS Contract described in Sections 3.2(o)(ii)(A), (C), (F), (G) and (H) is in full force and effect and valid, binding and enforceable against the other parties thereto in accordance with its terms, subject to applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equity principles. None of TMUS, any of its Subsidiaries or, to the Knowledge of TMUS, any other Person is in breach or violation of, or default under, any Material TMUS Contract or Restricted TMUS Contract, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. To the Knowledge of TMUS, no event has occurred that would result in a breach of or default under, require any consent or other action by any Person under, or give rise to any penalty or right of termination, cancellation or acceleration of any right or obligation of TMUS or its Subsidiaries to a loss of any benefit to which TMUS or any of its Subsidiaries is entitled under (in each case, with or without notice or lapse of time, or both), any Material TMUS Contract or Restricted TMUS Contract, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS or its potch), any Material TMUS contract, except as would not, individually or in the aggregate, reasonably be likely to have a time, or both), any Material TMUS contract or Restricted TMUS contract, except as would not, individually or in the aggregate, reasonably be likely to

#### (p) Sufficiency and Ownership of Assets; Business.

(i) The assets owned, leased or licensed by TMUS and its Subsidiaries (including the TMUS Communications Licenses and any interest TMUS and its Subsidiaries have in the Tower Assets), together with the assets that are the subject of the Trademark License, constitute all the assets, properties and rights (A) necessary to conduct the Business as presently conducted by TMUS and its Subsidiaries in all material respects and (B) used to generate the results of TMUS and its Subsidiaries set forth in the TMUS Financial Statements, other than assets disposed of in the ordinary course of business. Neither TMUS nor any of its Subsidiaries owns, leases or licenses any assets or properties (other than contract rights) outside of the Territory.

(ii) All of the wireless telecommunications operator business of DT and its Affiliates in the Territory is operated by TMUS and its Subsidiaries and is included in the assets owned by TMUS and its Subsidiaries. There are no agreements between DT or any of its Subsidiaries (other than TMUS and its Subsidiaries), on the one hand, and any third party, on the other hand, with respect to the resale of, or acting as an agent in the sale of, wireless telecommunications services in the Territory, the existence or effects of which are material to TMUS and its Subsidiaries, taken as a whole.

### (q) Property.

(i) Each of TMUS and its Subsidiaries has good and marketable title to all Owned Real Property of TMUS material to TMUS and its Subsidiaries, taken as a whole, free and clear of all Encumbrances except TMUS Permitted Encumbrances. No parcel of such Owned Real Property is subject to any governmental decree or order to be sold or is being condemned, expropriated or otherwise taken by any public authority with or without payment of compensation therefore, nor, to the Knowledge of TMUS, has any such condemnation, expropriation or taking been proposed, threatened or noticed, in each case, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect.

(ii) Each of TMUS and its Subsidiaries has good and marketable leasehold title to all Leased Real Property of TMUS material to TMUS and its Subsidiaries, taken as a whole, free and clear of all Encumbrances except TMUS Permitted Encumbrances. No parcel of such Leased Real Property is subject to any governmental decree or order to be sold or is being condemned, expropriated or otherwise taken by any public authority with or without payment of compensation therefore, nor, to the Knowledge of TMUS, has any such condemnation, expropriation or taking been proposed, threatened or noticed, in each case, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. Except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect, all leases of such Leased Real Property and all amendments and modifications thereto are in full force and effect and valid, binding and enforceable against the other parties thereto in accordance with its terms, subject to applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equity principles. There exists no default under any lease of such Leased Real Property by TMUS, any of its Subsidiaries or, to the Knowledge of TMUS, any of its Subsidiaries or, to the Knowledge of TMUS, any of its Subsidiaries or, to the Knowledge of TMUS, any of its Subsidiaries or, to the Knowledge of TMUS, any of its Subsidiaries or, to the Knowledge of TMUS, any other Person party thereto, nor any event which, with notice or lapse of such Leased Real Property shall remain valid and binding in accordance with their terms following the Closing, except as would not, individually or in the aggregate, reasonably be likely to have a TMUS Material Adverse Effect. All leases of such Leased Real Property shall remain valid and binding in accordance with their terms following the Closin

(iii) There are no contractual or legal restrictions that preclude or restrict the ability to use any of TMUS s Owned Real Property or Leased Real Property (other than any Owned Real Property or Leased Real Property included in the Tower Assets) in any material respect for the current or contemplated use of such real property. There are no material latent defects or material adverse physical conditions affecting such Owned Real Property or Leased Real Property or Leased Real Property or Leased Real Property. TMUS and its Subsidiaries have, in all material respects, valid leasehold interests in, or other valid rights to use, all plants, warehouses, distribution centers, structures and other buildings on such Owned Real Property and Leased Real Property, which are adequately maintained and are in good operating condition and repair for the requirements of the business of TMUS and its Subsidiaries as currently conducted.

(iv) Except as would not, individually or in the aggregate, reasonably be likely to result in a TMUS Material Adverse Effect, (A) TMUS and its Subsidiaries have good and marketable title to or, in the case of leased assets, a valid leasehold interest in, free and clear of all Encumbrances (other than TMUS Permitted Encumbrances), all of the tangible personal property and assets (except for properties and assets disposed of in the ordinary course of business consistent with past practice) used in or necessary to conduct their businesses substantially as presently conducted and (B) each item of tangible personal property of TMUS and each of its Subsidiaries, or in which TMUS or any of its Subsidiaries owns an undivided interest, is in all material respects in good operating condition and repair for the requirements of the business of TMUS and its Subsidiaries as currently conducted, ordinary wear and tear excepted.

(r) <u>Related-Party Agreements</u>. As of the date hereof, there are no agreements between TMUS or any of its Subsidiaries, on the one hand, and DT and/or any of its Affiliates (other than TMUS and its Subsidiaries), on the other hand, that are material to the operations of the Business as presently conducted by TMUS and its

Subsidiaries, other than as set forth on Schedule 3.2(r) of the TMUS Disclosure Letter, and TMUS or one of its Affiliates has provided MetroPCS with complete and correct copies of all such agreements prior to the date hereof.

(s) <u>Prohibited Payments</u>. To the Knowledge of TMUS, none of TMUS or any of its Subsidiaries or any of their respective directors, officers, agents, employees or other Persons associated with them or acting on their behalf has (i) used any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expenses relating to political activity; (ii) made any direct or indirect unlawful payment to any foreign or domestic government official or employee from corporate funds; (iii) violated or is in violation of any provision of the Foreign Corrupt Practices Act of 1977; or (iv) made any unlawful bribe, rebate, payoff, influence payment, kickback or other payment.

3.3 <u>Representations and Warranties of MetroPCS</u>. Except as specifically identified in the notes to the MetroPCS Financial Statements or in the MetroPCS SEC Reports (excluding the MetroPCS Financial Statements) filed with the SEC after December 31, 2011 and prior to the date hereof (excluding, in each case, any disclosures set forth in any risk factor section or in any other section to the extent they are forward-looking statements or cautionary, predictive or forward-looking in nature) or in the corresponding sections of the disclosure letter delivered to DT by MetroPCS on or prior to entering into this Agreement (the MetroPCS Disclosure Letter ) (it being agreed that disclosure of any item in any part of the MetroPCS Disclosure Letter shall be deemed disclosure with respect to any other part to which the relevance of such item is reasonably apparent on its face notwithstanding the omission of a reference or cross-reference thereto), MetroPCS hereby represents and warrants to DT as follows:

(a) <u>Organization. Good Standing and Qualification</u>. Each of MetroPCS and its Subsidiaries is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite power and authority to own and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing in each jurisdiction where the ownership or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so qualified or in good standing or to have such power and authority would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. Prior to the date hereof, DT has been provided with complete and correct copies of each of MetroPCS s and its Subsidiaries Organizational Documents, and each as so delivered is in full force and effect.

#### (b) Capitalization.

(i) The authorized capital stock of MetroPCS consists solely of 1,000,000,000 shares of MetroPCS Common Stock, of which 367,457,662 shares were issued and outstanding (and of which 3,582,173 shares are MetroPCS Restricted Stock) as of the close of business on October 1, 2012, and 100,000,000 shares of preferred stock (<u>MetroPCS Preferred Stock</u>), of which no shares were issued and outstanding as of the close of business on October 1, 2012. All of the outstanding shares of MetroPCS Common Stock (A) have been duly authorized and validly issued, (B) are fully paid and nonassessable, and (C) were issued in compliance with all applicable Laws concerning the issuance of securities and not in violation of any preemptive rights, purchase option, call, right of first refusal or any similar right.

(ii) As of the close of business on October 1, 2012, MetroPCS had no shares of MetroPCS Common Stock or MetroPCS Preferred Stock reserved or otherwise subject to issuance, except for (A) 31,627,302 shares of MetroPCS Common Stock reserved for issuance pursuant to the MetroPCS Stock Plans for outstanding awards thereunder, (B) 12,345,015 shares of MetroPCS Common Stock otherwise reserved for future issuance pursuant to the MetroPCS Stock Plans, and (C) 1,000,000 shares of MetroPCS Preferred Stock reserved for future issuance pursuant to the MetroPCS Rights Agreement. All such shares of MetroPCS Common Stock reserved for or otherwise subject to issuance, when issued in accordance with the respective terms thereof, will be duly authorized and validly issued and will be fully paid, nonassessable and not in violation of any preemptive rights, purchase option, call, right of first refusal or any similar right.

Section 3.3(b)(ii) of the MetroPCS Disclosure Letter contains a complete and correct list of each award under the MetroPCS Stock Plans outstanding as of October 1, 2012, including, to the extent applicable, (1) the name of the holder of the award; (2) the number of shares of MetroPCS Common Stock subject to such award; (3) in the case of a MetroPCS Stock Option, the exercise price of such MetroPCS Stock Option; (4) the date on which such award was granted; (5) the extent to which such award is vested (and, in the case of any MetroPCS Stock Option, exercisable) as of October 1, 2012; (6) the dates and extent to which such award is scheduled to become vested (and, in the case of any MetroPCS Stock Option, exercisable) after October 1, 2012, including any events (including the consummation of the Transaction) that would result in any acceleration of such vesting or exercisability, as applicable; (7) in the case of a MetroPCS Stock Option, whether such MetroPCS Stock Option is an incentive stock option or a nonqualified stock option; and (8) in the case of a MetroPCS Stock Option, the date on which such MetroPCS is stock Option expires. Except as set forth in clauses (i) and (ii) of this Section 3.3(b), there are no Equity Interests of MetroPCS issued, authorized or outstanding.

(iii) MetroPCS has no Voting Debt issued or outstanding. Except as set forth in Section 3.3(b)(ii), as of the date hereof there are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements or commitments of any character under which MetroPCS is or may become obligated to issue or sell, or giving any Person a right to subscribe for or acquire, or in any way dispose of, any Equity Interests or Voting Debt of MetroPCS, or any securities or obligations exercisable or exchangeable for, or convertible into, any Equity Interests or Voting Debt of MetroPCS, and no securities or obligations evidencing such rights are authorized, issued or outstanding. No award under any MetroPCS Stock Plan has been granted in connection with the Transaction. None of the MetroPCS Stock Options currently outstanding has an exercise price below or deemed to be below fair market value on the date of grant. All grants of awards under MetroPCS Stock Plans currently outstanding were validly made and properly approved by the MetroPCS Board (or a duly authorized committee or subcommittee thereof) in compliance with applicable Laws and recorded on the consolidated financial statements of MetroPCS in accordance with GAAP, and, where applicable, no such grants involved any back dating, forward dating or similar practices.

(iv) Except with respect to the MetroPCS Rights Agreement and the MetroPCS Stock Plans and the related award agreements, and Restricted Stock that has not vested, there are no outstanding obligations of MetroPCS or any of its Subsidiaries or contracts to which MetroPCS or any of its Subsidiaries is bound (A) requiring the repurchase, redemption, acquisition or disposition of, or containing any right of first refusal with respect to, (B) requiring the registration for sale of, (C) applying voting restrictions to, or (D) otherwise restricting any Person from purchasing, selling, pledging or otherwise disposing of, any Equity Interests in MetroPCS or any MetroPCS Subsidiary. Except as set forth in Section 3.3(b)(ii), there are no phantom stock or similar rights providing economic benefits based, directly or indirectly, on the value or price of the Equity Interests of MetroPCS.

(v) Schedule 3.3(b)(v) of the MetroPCS Disclosure Letter lists all Indebtedness described in clauses (i) and (iii) of the definition of Indebtedness. There are no purchase money security interests for which liens have been filed, in each case created, issued, assumed, guaranteed or permitted to exist by each of MetroPCS and its Subsidiaries as of the date hereof, that are material to MetroPCS and its Subsidiaries, taken as a whole.

#### (c) Subsidiaries.

(i) A true and complete list of the Subsidiaries of MetroPCS as of the date hereof is set forth on Schedule 3.3(c)(i) of the MetroPCS Disclosure Letter, and such list sets forth, with respect to each such Subsidiary, as of the date hereof (A) its jurisdiction of organization or formation and (B) the direct or indirect ownership interest of MetroPCS in such Subsidiary, as well as the ownership interest of any other Person in such Subsidiary if it is not wholly-owned, directly or indirectly, by MetroPCS. MetroPCS does not own, directly or indirectly, any Equity Interests in any Person that is not a Subsidiary of MetroPCS where the value of such Equity Interests with respect to any such Person exceeds \$1,000,000.

(ii) All of the Equity Interests of each Subsidiary of MetroPCS owned beneficially or of record, directly or indirectly, by MetroPCS are owned free and clear of any Encumbrances (other than MetroPCS Permitted Encumbrances). No Subsidiary of MetroPCS has any Voting Debt issued or outstanding. There are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements or commitments of any character under which any Subsidiary of MetroPCS is or may become obligated to issue or sell, or giving any Person a right to subscribe for or acquire, or in any way dispose of, any Equity Interests or Voting Debt of any Subsidiary of MetroPCS or any securities or obligations exercisable or exchangeable for, or convertible into, any Equity Interests or Voting Debt of any Subsidiary of MetroPCS, and no securities or obligations evidencing such rights are authorized, issued or outstanding.

(iii) All of the outstanding Equity Interests of the Subsidiaries of MetroPCS have been duly authorized and are validly issued, fully paid and nonassessable. The outstanding Equity Interests of each Subsidiary of MetroPCS are not subject to any voting trust agreement or other contract, agreement or arrangement restricting or otherwise relating to the voting, dividend rights or disposition of such Equity Interests (other than MetroPCS Permitted Encumbrances). There are no phantom stock or similar rights providing economic benefits based, directly or indirectly, on the value or price of the Equity Interests of any Subsidiary of MetroPCS.

(d) <u>Authorization</u>. MetroPCS has all requisite power and authority and has taken all action necessary in order to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the Transaction, subject to (i) the approval of the MetroPCS Share Issuance by a majority of the votes cast on such matter at a stockholders meeting duly called and held for such purpose (the <u>MetroPCS Stockholders</u> <u>Meeting</u>): provided that the total votes cast on such matter represent over 50% in interest of all MetroPCS Common Stock entitled to vote on such matter at the MetroPCS Stockholders Meeting, (ii) the approval of the New MetroPCS Certificate by the holders of a majority of the outstanding shares of MetroPCS Common Stock entitled to vote on such matter at the MetroPCS Stockholders Meeting ((i) and (ii) together, the <u>MetroPCS Stockholder Approval</u>), and (iii) the filings and actions referred to in Section 3.3(e). The execution and delivery by MetroPCS of this Agreement, the performance of its obligations hereunder and the consummation by MetroPCS of the Transaction have been duly authorized by all necessary action of MetroPCS, subject to receipt of the MetroPCS Stockholder Approval. This Agreement has been duly executed and delivered by MetroPCS and, assuming the due authorization, execution and delivery of this Agreement by DT, constitutes the legal, valid and binding obligation of MetroPCS, enforceable against MetroPCS in accordance with its terms, except as limited by applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equitable principles.

#### (e) Governmental Filings; No Conflicts.

(i) Other than (A) the reports, filings, registrations, consents, approvals, permits, waivers, petitions for declaratory ruling, authorizations and/or notices (1) under the HSR Act, (2) with, by, from or to the FCC pursuant to the Communications Act, (3) pursuant to any applicable state or territorial public utility Laws and rules, regulations and orders of any PUCs or similar foreign public utility Laws and rules, regulations and orders of any PUCs or similar foreign public utility Laws and rules, regulations and orders of any regulatory bodies regulating telecommunications businesses set forth on Schedule 3.3(e) of the MetroPCS Disclosure Letter, or (4) under the Exchange Act, the Securities Act or state Blue Sky or foreign securities laws, including the filing with the SEC of a proxy statement in definitive form relating to the MetroPCS Stockholder Approval at the MetroPCS Stockholders Meeting (as amended or supplemented from time to time, the <u>Proxy Statement</u>), (B) the filing of the New MetroPCS Certificate with the Secretary of State of the State of Delaware, and (C) the filing with the NYSE of a supplemental listing application, subject to official notice of issuance, of the TMUS Stock Consideration, no material notices, reports or other filings are required to be made or effected by MetroPCS or its Subsidiaries with, nor are any material consents, registrations, approvals, permits or authorizations required to be obtained by MetroPCS

or its Subsidiaries from, any Governmental Entity in connection with the execution and delivery of this Agreement by MetroPCS, the performance of its obligations hereunder or the consummation of the Transaction.

(ii) The execution and delivery of this Agreement by MetroPCS, the performance of its obligations hereunder and the consummation of the Transaction will not constitute or result in (A) a breach or violation of, or a default under, the Organizational Documents of MetroPCS or of any of its Subsidiaries, (B) a breach or violation of, a termination (or right of termination) or default under, the creation or acceleration of any obligations under, the change of any rights of MetroPCS or any of its Subsidiaries under, or the creation of an Encumbrance (other than an Encumbrance set forth in clauses (i) through (v) of the definition of MetroPCS Permitted Encumbrance) on, any of the assets of MetroPCS or any of its Subsidiaries (with or without notice, lapse of time or both) pursuant to any agreement, lease, license, contract, note, mortgage, credit agreement, security agreement, indenture, arrangement or other obligation binding upon MetroPCS or any of its Subsidiaries, except for, with respect to the consummation of the Transaction, the MetroPCS Existing Finance Documents, or (C) conflict with, breach or violate any Law applicable to MetroPCS or by which its properties are bound or affected, except, in the case of clause (B) or (C) above, for any such breach, violation, termination, default, creation, acceleration or change that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

#### (f) MetroPCS SEC Reports; Financial Statements; Undisclosed Liabilities.

(i) MetroPCS has filed or furnished, as applicable, on a timely basis all MetroPCS SEC Reports since December 31, 2009. Each of the MetroPCS SEC Reports, at the time of its filing or being furnished complied, or if not yet filed or furnished, will comply, in all material respects with the applicable requirements of the Exchange Act, the Securities Act and the Sarbanes-Oxley Act, and any rules and regulations promulgated thereunder applicable to the MetroPCS SEC Reports. As of their respective dates (or, if amended prior to the date hereof, as of the date of such amendment), the MetroPCS SEC Reports did not, and any MetroPCS SEC Reports filed with or furnished to the SEC subsequent to the date hereof will not, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances in which they were made, not misleading.

(ii) Each of the audited and unaudited balance sheets and consolidated statements of income and comprehensive income, changes in stockholders equity and cash flows of MetroPCS and its consolidated Subsidiaries included in or incorporated by reference into the MetroPCS SEC Reports (including any related notes and schedules) (the <u>MetroPCS Financial Statements</u>) (A) have been prepared in accordance with GAAP applied on a consistent basis during the periods involved, except as may be noted therein or in the notes thereto, (B) present fairly, in all material respects, the consolidated financial position of MetroPCS and its consolidated Subsidiaries as of the dates thereof and the consolidated statements of income and comprehensive income, changes in stockholders equity and cash flows of MetroPCS and its consolidated Subsidiaries for the periods then ended and (C) accurately reflect in all material respects the books of account and other financial records of MetroPCS and its consolidated Subsidiaries.

(iii) Neither MetroPCS nor any of its Subsidiaries has any Liabilities except for (A) Liabilities reflected or reserved against on MetroPCS s consolidated audited balance sheet as of December 31, 2011 (or the notes thereto) and not heretofore paid or discharged, (B) Liabilities incurred since December 31, 2011 in the ordinary course of business consistent with past practice or (C) Liabilities that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

(iv) MetroPCS has designed and maintains a system of internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) sufficient to provide reasonable assurances regarding the reliability of financial reporting for MetroPCS and its Subsidiaries. MetroPCS (A) has designed and maintains disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) effective to ensure that information required to be disclosed by MetroPCS in the reports

that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC s rules and forms and is accumulated and communicated to MetroPCS s management (including MetroPCS s principal executive and principal financial officers, or Persons performing similar functions) as appropriate to allow timely decisions regarding required disclosure and (B) based on its most recent evaluation of internal control prior to the date hereof, has disclosed to MetroPCS s auditors and the audit committee of the MetroPCS Board (I) any significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect MetroPCS s ability to record, process, summarize and report financial information and (II) any fraud, whether or not material, that involves management or other employees who have a significant role in MetroPCS s internal control over financial reporting, except with respect to clause (A), those failures to design or maintain that would not, individually or in the aggregate, be material to MetroPCS and its Subsidiaries, taken as a whole. MetroPCS has made available to DT a summary of any such disclosure made by management to MetroPCS s auditors and to the audit committee since December 31, 2009. Since December 31, 2009, to the Knowledge of MetroPCS no attorney representing MetroPCS or any of its Subsidiaries, whether or not employed by MetroPCS or any of its Subsidiaries, has reported to MetroPCS s chief legal officer, chief executive officer, audit committee or the MetroPCS Board evidence of a material violation of securities laws, material breach of fiduciary duty or similar material violation of law, relating to periods after December 31, 2009.

(v) MetroPCS has made available to DT true and complete copies of all comment letters from the staff of the SEC since December 31, 2009 relating to MetroPCS SEC Reports and all written responses of MetroPCS thereto through the date hereof. As of the date hereof, there are no outstanding or unresolved comments in comment letters received from the SEC staff with respect to any MetroPCS SEC Reports. To the Knowledge of MetroPCS, there are no SEC inquiries or investigations, other governmental inquiries or investigations or internal investigations pending or threatened involving MetroPCS or any of its Subsidiaries, in each case regarding any accounting practices of MetroPCS or any of its Subsidiaries.

(g) <u>Litigation</u>. As of the date hereof, there is no civil, criminal or administrative action, suit, demand, claim, hearing or proceeding or investigation pending or, to the Knowledge of MetroPCS, threatened, against MetroPCS or any of its Subsidiaries, except those that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. Neither MetroPCS nor any of its Subsidiaries is a party to, or subject to the provisions of, any judgment, order, writ, injunction, decree or award of any Governmental Entity that would, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. No representation or warranty is made in this Section 3.3(g) with respect to Tax matters, which shall be governed exclusively by Section 3.3(h) (Employee Benefits) and 3.3(m) (Taxes), or environmental matters, which shall be governed exclusively by Section 3.3(l) (Environmental Matters).

#### (h) Employee Benefits.

(i) All benefit and compensation plans, contracts, agreements, policies or arrangements sponsored or contributed to by MetroPCS or any of its Subsidiaries, including those covering any of its past or present employees, officers or directors (or for which MetroPCS or any of its Subsidiaries could have any liability), including employee benefit plans within the meaning of Section 3(3) of ERISA, and employment agreements, collective bargaining agreements, deferred compensation, change of control, retention, stock option, stock purchase, restricted stock, stock appreciation, phantom share, stock based, incentive, severance and bonus plans (other than any immaterial benefit plans) (the <u>MetroPCS Benefit Plans</u>) in effect as of the date hereof are listed on Schedule 3.3(h) of the MetroPCS Disclosure Letter. True and complete copies of all MetroPCS Benefit Plans listed on Schedule 3.3(h) of the MetroPCS Disclosure Letter, and of all related material funding documents, actuarial and financial reports, government correspondence, summary plan descriptions, annual reports, IRS determination letters, and audit reports have been provided or made available to DT prior to the date hereof.

(ii) Each MetroPCS Benefit Plan was established and, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect, has been documented, maintained and administered in compliance in all respects with the terms thereof and the applicable requirements of ERISA, the Code, and any other applicable Law. Each MetroPCS Benefit Plan that is intended to be qualified under Section 401(a) of the Code has received a favorable determination or opinion letter from the IRS with respect to its qualified status under Section 401(a) of the Code or has pending or has time remaining in which to file an application for such determination from the IRS (or MetroPCS and its Subsidiaries are entitled to rely on a favorable opinion or advisory letter issued by the IRS in accordance with Revenue Procedure 2005-16 with respect to the qualified status of the plan document), and, to the Knowledge of MetroPCS, there is no fact or circumstance that exists that would, individually or in the aggregate, reasonably be likely to adversely affect or give rise to the revocation of such qualified status. All contributions required to be made under the terms of any MetroPCS Benefit Plan or applicable Law (including all employer contributions and employee salary reduction contributions) have been timely made or are reflected in the MetroPCS Financial Statements as of the dates thereof. All unfunded liabilities, if any, under the MetroPCS Benefit Plans are fully reflected in the MetroPCS Financial Statements. No event has occurred and no condition exists that would, individually or in the aggregate, reasonably be likely to subject MetroPCS or any of its Subsidiaries to any material Tax, fine, lien, penalty or other liability imposed by ERISA or the Code or other applicable Law in respect of any MetroPCS Benefit Plan.

(iii) Neither MetroPCS nor any ERISA Affiliate maintains or contributes to or has within the past six complete calendar years maintained or contributed to, or been required to contribute to, or otherwise has any direct or indirect liability with respect to, an employee pension benefit plan (within the meaning of Section 3(2) of ERISA) that is subject to Title IV of ERISA, is a multiemployer plan (within the meaning of Section 3(37) of ERISA) or is a multiple employer plan (within the meaning of Section 4063 of ERISA or Section 413(c) of the Code). Except as required by the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, or similar state or local law, or as is reflected on the MetroPCS Financial Statements, neither MetroPCS nor any ERISA Affiliate is obligated to provide any retiree health or life insurance benefits to any employee or former employees of MetroPCS or any ERISA Affiliate. Each MetroPCS Benefit Plan can be amended or terminated at any time without liability, other than liability for benefits accrued as of the date of any such amendment or termination.

(iv) Excluding routine, uncontested claims for benefits under any MetroPCS Benefit Plan and except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect, (A) there is no action, suit, demand, audit, claim, hearing, proceeding or investigation pending against or involving or, to the Knowledge of MetroPCS, threatened, against or involving any MetroPCS Benefit Plan before any court or arbitrator or any Governmental Entity, or federal, state or local official that would, individually or in the aggregate, reasonably be likely to subject MetroPCS or any of its Subsidiaries to a liability, except those first arising after the date hereof in the ordinary course of business, and (B) to the Knowledge of MetroPCS, there are no facts or circumstances existing that would, individually or in the aggregate, reasonably be likely to give rise to such actions, suits, demands, audits, claims, hearings or proceedings.

(v) No agreement, binding commitment or obligation exists to materially increase benefits under, or adopt any new, MetroPCS Benefit Plan. There is no provision of any MetroPCS Benefit Plan or other related contract or agreement, and there has been no amendment to or announcement by, MetroPCS or any of its Subsidiaries relating to, or change in employee participation or coverage under, any MetroPCS Benefit Plan that would increase materially the expense of maintaining such plan above the level of the expense incurred therefor for the most recent fiscal year.

(vi) Neither the execution of this Agreement nor the consummation of the Transaction will (whether alone or in connection with any other related event(s)): (A) entitle any employee of MetroPCS or any of its Subsidiaries to severance pay or any increase in severance pay (or other compensation or benefits) upon any termination of employment; (B) accelerate the time of payment or vesting or result in any payment or

funding (through a grantor trust or otherwise) of any compensation or benefits under, or increase the amount payable pursuant to, or result in the deemed satisfaction of vesting conditions, goals or any other requirements or conditions under, any of the MetroPCS Benefit Plans; (C) limit or restrict the right of MetroPCS or any of its Subsidiaries or, after the consummation of the Transaction, DT or any of its Subsidiaries (including MetroPCS and any of its Subsidiaries), to merge, amend or terminate any of the MetroPCS Benefit Plans (other than solely pursuant to applicable Law); or (D) result in the creation, increase, forgiveness, extension or modification of any loan to any employee, officer or director of MetroPCS or any of its Subsidiaries.

(vii) No MetroPCS Benefit Plan or other MetroPCS practice provides any Person with any amount of additional compensation if such individual is provided amounts subject to excise or additional taxes imposed under Section 409A or 4999 of the Code.

(viii) Each MetroPCS Benefit Plan that is a nonqualified deferred compensation plan (as defined under Section 409A(d)(1) of the Code) is, except as would not result in a material liability, in documentary compliance with Section 409A of the Code and the guidance provided thereunder and has been operated and administered in compliance in all material respects with Section 409A of the Code and the guidance provided thereunder.

#### (i) Compliance with Laws; Licenses.

(i) The business of MetroPCS and its Subsidiaries has been, and is being, conducted in compliance with all Laws, except for violations that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. No investigation or review by any Governmental Entity with respect to MetroPCS or any of its Subsidiaries is pending or, to the Knowledge of MetroPCS, threatened, nor has any Governmental Entity indicated an intention to conduct the same, except for such investigations or reviews that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. Each of MetroPCS and its Subsidiaries has obtained and is in compliance with all Licenses necessary to conduct its business as presently conducted, except those the absence of which would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect (the <u>MetroPCS Material Licenses</u>).

(ii) Schedule 3.3(i)(ii) of the MetroPCS Disclosure Letter sets forth a true and complete list, as of the date hereof, of (A) all MetroPCS Material Licenses and, to the extent not otherwise constituting MetroPCS Material Licenses, all Licenses issued or granted to MetroPCS or any of its Subsidiaries by the FCC and all leases for the use of wireless spectrum licensed by other FCC licensees to MetroPCS or any of its Subsidiaries (such licenses and leases, the <u>MetroPCS FCC Licenses</u>), all Licenses issued or granted to MetroPCS or any of its Subsidiaries by PUCs regulating telecommunications businesses (<u>MetroPCS State Licenses</u>), and all Licenses issued or granted to MetroPCS or any of its Subsidiaries by foreign Governmental Entities regulating telecommunications businesses (collectively with the MetroPCS Material Licenses, MetroPCS FCC Licenses and MetroPCS State Licenses, the <u>MetroPCS Communications Licenses</u> ); (B) all pending applications for issuance, grant, assignment or transfer of Licenses, or applications for leases of wireless spectrum, to MetroPCS or any of its Subsidiaries, that would be MetroPCS Communications Licenses if issued or granted; (C) all pending applications for assignment or transfer of Licenses, or applications for leases of wireless spectrum, by MetroPCS or its Subsidiaries to any Person (other than MetroPCS or its Subsidiaries); and (D) all pending applications by MetroPCS or any of its Subsidiaries for modification, extension or renewal of any MetroPCS Communications License; provided that Schedule 3.3(i)(ii) may exclude point-to-point microwave licenses, business radio licenses, experimental licenses and Section 214 certificates and pending applications regarding the same. Each of MetroPCS and its Subsidiaries is in compliance with its obligations under each of the MetroPCS FCC Licenses and the rules and regulations of the FCC, and with its obligations under each of the MetroPCS State Licenses, in each case, except for such failures to be in compliance with Licenses that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

MetroPCS and its Subsidiaries are not the subject of, and there are no pending nor, to the Knowledge of MetroPCS, threatened, proceedings, notices of violation, orders of forfeiture or complaints or investigations relating to the MetroPCS Communications Licenses before the FCC, the FAA, or any other Governmental Entity, in each case, except that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. The FCC actions granting all MetroPCS FCC Licenses, together with all underlying construction permits, have not been reversed, stayed, enjoined, annulled or suspended, and there is not pending or, to the Knowledge of MetroPCS, threatened any application, petition, objection or other pleading with the FCC, the FAA or any other Governmental Entity that challenges or questions the validity of or any rights of the holder under any such MetroPCS FCC License, in each case, except that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

(iii) MetroPCS holds the MetroPCS Communications Licenses, and the MetroPCS Communications Licenses are issued in the name of MetroPCS or one of its Subsidiaries. The MetroPCS Communications Licenses are in full force and effect, are granted without conditions, except for those conditions on the face of such MetroPCS Communications Licenses or conditions generally applicable to all similarly situated licenses of comparable spectrum, and are free and clear of all Encumbrances (other than MetroPCS Permitted Encumbrances) or any restrictions which might, individually or in the aggregate, limit the full operation of the MetroPCS Communications Licenses in any material respect.

(iv) All of the currently operating cell sites and microwave paths of MetroPCS and its Subsidiaries in respect of which a filing with the FCC was required have been constructed and are currently operated as represented to the FCC in currently effective filings, and modifications to such cell sites and microwave paths have been preceded by the submission to the FCC of all required filings, in each case, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

(v) All transmission towers owned or leased by MetroPCS and its Subsidiaries are (to the Knowledge of MetroPCS with respect to leased towers) obstruction-marked and lighted by MetroPCS or its Subsidiaries to the extent required by, and in accordance with, the FAA Rules, except that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. Appropriate notification to the FAA has been made for each transmission tower owned or leased by MetroPCS and its Subsidiaries to the extent required to be made by MetroPCS or any of its Subsidiaries by, and in accordance with, the FAA Rules, in each case, except that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

(vi) Neither MetroPCS nor any of its Subsidiaries holds any MetroPCS Communications Licenses through a partnership, joint venture or other Person that is not a Subsidiary of MetroPCS.

(vii) MetroPCS does not hold any License to offer, and does not offer, any services or products other than wireless telecommunications and wireless information services and products, and any ancillary services or products thereto. MetroPCS and its Subsidiaries do not conduct any business other than the Business.

(viii) MetroPCS and its Subsidiaries are fully qualified under the Communications Act and the rules and regulations of the FCC to hold the MetroPCS FCC Licenses generally. To the Knowledge of MetroPCS, there are no facts or circumstances relating to the qualifications of MetroPCS and its Subsidiaries that would prevent or materially delay the grant of any FCC Form 603 application (or other appropriate form) under the FCC Rules and the Communications Act with respect to the Transaction.

(ix) No representation or warranty is made in this Section 3.3(i) with respect to Tax matters, which shall be governed exclusively by Sections 3.3(h) (Employee Benefits) and 3.3(m) (Taxes), or environmental matters, which shall be governed exclusively by Section 3.3(l) (Environmental Matters).

(j) <u>Absence of Certain Changes</u>. Since December 31, 2011 and prior to the date hereof, other than expenses and capital expenditures incurred or made in accordance with the MetroPCS Business Plan, MetroPCS and its Subsidiaries have conducted their respective businesses only in, and have not engaged in any material transaction other than in accordance with, the ordinary course of such businesses, and there has not been any:

(i) Circumstance (including any adverse change with respect to any Circumstance existing on or prior to December 31, 2011) that, individually or in the aggregate, has had or would reasonably be likely to have a MetroPCS Material Adverse Effect;

(ii) merger or consolidation between MetroPCS or any of its Subsidiaries with any other Person, or any restructuring, reorganization or complete or partial liquidation or similar transaction, or the entry into any agreements or arrangements imposing material changes or restrictions on its assets, operations or businesses, except for any such transactions among wholly-owned Subsidiaries of MetroPCS;

(iii) acquisition of assets outside of the ordinary course of business consistent with past practice for consideration in excess of \$25,000,000 individually, other than in accordance with the MetroPCS Business Plan;

(iv) creation or incurrence of any Encumbrance (other than any MetroPCS Permitted Encumbrances) on (x) any MetroPCS FCC Licenses or (y) the other assets of MetroPCS or its Subsidiaries that are, individually or in the aggregate, material to MetroPCS or any of its Subsidiaries;

(v) loan, advance, guarantee or capital contribution to, or investment in, any Person (other than any of the foregoing to or on behalf of MetroPCS or any direct or indirect wholly-owned Subsidiary of MetroPCS and other than loans or advances to employees and contractors in the ordinary course of business consistent with past practices in an amount not to exceed \$250,000 individually);

(vi) material damage, destruction or other casualty loss with respect to any material asset, or MetroPCS Owned Real Property, MetroPCS Leased Real Property or property otherwise used by MetroPCS or any of its Subsidiaries, whether or not covered by insurance;

(vii) declaration, setting aside or payment of any non-cash distribution with respect to any Equity Interests of MetroPCS or any of its Subsidiaries (except for distributions by any direct or indirect wholly-owned Subsidiary of MetroPCS to MetroPCS or any other such Subsidiary of MetroPCS);

(viii) incurrence of any Indebtedness for borrowed money other than (x) from MetroPCS or any of its wholly-owned Subsidiaries or (y) in the ordinary course of business under MetroPCS s existing revolving credit facility under the MetroPCS Existing Finance Documents;

(ix) material change in any method of financial accounting or accounting practice by MetroPCS or any of its Subsidiaries, except for any such change required by changes in GAAP or applicable Law;

(x) increase in the compensation payable or to become payable to its officers or employees (except for increases in the ordinary course of business consistent with past practice);

(xi) fundamental change to any of the important elements of the network technologies or principal billing systems of MetroPCS and its Subsidiaries (excluding, for the avoidance of doubt, system upgrades, improvements and modernization, equipment replacement and similar matters, in each case within the same fundamental framework of such network technologies and billing systems); or

(xii) agreement to do any of the foregoing.

(k) <u>Insurance</u>. All Insurance Policies maintained by MetroPCS or any of its Subsidiaries, together with adequately capitalized self-insurance arrangements, provide adequate coverage for all normal risks incident to the business of MetroPCS and its Subsidiaries and their respective properties and assets, except for any such failures to maintain such Insurance Policies that would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. As of the date hereof, neither MetroPCS nor any of its Subsidiaries has received any written notice of cancellation of any material Insurance Policy.

(1) <u>Environmental Matters</u>. Except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect:

(i) since the date that is four years prior to the date hereof, MetroPCS and its Subsidiaries have been in compliance with all applicable Environmental Laws and have not incurred any Liabilities concerning any Environmental Laws with respect to the business of MetroPCS and its Subsidiaries;

(ii) there are no writs, injunctions, decrees, awards, orders or judgments outstanding, or any actions, suits, demands, claims, hearings, proceedings or investigations pending or, to the Knowledge of MetroPCS, threatened, relating to compliance with, or Liability under, any Environmental Law affecting the business of MetroPCS and its Subsidiaries, other than those first arising after the date hereof in the ordinary course of business;

(iii) to the Knowledge of MetroPCS, there has been no release, threatened release, contamination or disposal of Hazardous Substances at any property currently or formerly owned or operated in connection with the business of MetroPCS and its Subsidiaries (including in soils, groundwater, surface water, buildings or other structures) or at any third-party property, or from any waste generated by MetroPCS or any of its Subsidiaries or any legally responsible predecessor corporation thereof, that has given or would, individually or in the aggregate, reasonably be likely to give rise to any Liability under any Environmental Law for which MetroPCS or any of its Subsidiaries would incur or share Liability; and

(iv) there are no consent decrees, orders or similar agreements with any Governmental Entity imposing restrictions on the ownership, use or transfer of any real property relating to, or derived from, any Environmental Law, and there are no indemnification or other agreements with any third party (other than ordinary course provisions in leases of real property or in agreements for the acquisition or disposition of assets or businesses) relating to any Liability or potential Liability under any Environmental Law.

(m) <u>Taxes</u>. Except as would not have and would not reasonably be expected to have, individually or in the aggregate, a MetroPCS Material Adverse Effect:

(i)(A) All Tax Returns required to be filed by, or on behalf of, or with respect to, MetroPCS and each of its Subsidiaries have been timely filed (taking into account extensions) with the appropriate Taxing Authority and all such Tax Returns are true and complete, and (B) MetroPCS has, or has caused each of its Subsidiaries to, duly and timely pay all Taxes due and payable, including Taxes required to be withheld from amounts owing to any Person, except in each case of clauses (A) and (B), with respect to matters contested in good faith or for which adequate reserves have been established, in accordance with GAAP, in the most recent MetroPCS Financial Statements, as adjusted to reflect operations in the ordinary course of business since the date thereof.

(ii) All deficiencies or assessments made in writing as a result of any audit, examination or investigation by any Taxing Authority of Tax Returns of MetroPCS and its Subsidiaries that are due and payable have been fully paid, and no other audits, examination or investigations by any Taxing Authority relating to any Tax Returns of MetroPCS and its Subsidiaries are in progress. Neither MetroPCS nor any of its Subsidiaries has received written notice from any Taxing Authority of the commencement of any audit, examination or investigation not yet in progress. There is no action, suit, demand, claim or hearing or, to the Knowledge of MetroPCS, proceeding, relating to Taxes pending or, to the Knowledge of MetroPCS, threatened, against MetroPCS or any of its Subsidiaries.

(iii) Neither MetroPCS nor any of its Subsidiaries is a party to any Tax indemnification, Tax allocation or Tax sharing agreement pursuant to which MetroPCS or any of its Subsidiaries, as applicable, will have any obligation to make any payments after the Closing Date, other than (A) any agreements solely among MetroPCS and/or its Subsidiaries and (B) Tax provisions in loan agreements, leases, license agreement and other commercial agreements the principal purpose of which does not relate to Taxes. Neither MetroPCS nor any of Subsidiaries is or could be liable for Taxes of any Person (other than of a member of the

affiliated group for United States federal income tax purposes of which MetroPCS or any of its Subsidiaries is or was the common parent) (x) under Section 1.1502-6 of the Treasury Regulations (or any similar provision of state, local or foreign Law), (y) as a transferee or successor, or (z) otherwise, for any taxable period (or portion thereof) ending on or before the Closing Date for which the applicable statute of limitations (including extensions) is not closed.

(iv) In the past five years, MetroPCS has not received any IRS private letter ruling or entered into any closing agreements within the meaning of Section 7121 of the Code relating to or with respect to the income and/or assets of MetroPCS or any of its Subsidiaries. There are no pending requests by MetroPCS or any of its Subsidiaries for an IRS private letter ruling.

(v) Neither MetroPCS nor any of its Subsidiaries will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any (A) change in method of accounting for a taxable period ending on or prior to the Closing Date, (B) closing agreement (within the meaning of Section 7121 of the Code or any corresponding or similar provision of state, local, or non-United States income Tax law) entered into on or prior to the Closing Date,
(C) installment sale or open transaction disposition made on or prior to the Closing Date, or (D) prepaid amount received on or prior to the Closing Date.

(vi) There are no Encumbrances for Taxes upon any assets of MetroPCS or any of its Subsidiaries other than MetroPCS Permitted Encumbrances.

(vii) Within the preceding three years, no written claim has been received by MetroPCS or any of its Subsidiaries from a Taxing Authority in any jurisdiction where MetroPCS or any of its Subsidiaries does not file Tax Returns asserting that MetroPCS or any of its Subsidiaries is or may be subject to taxation by such jurisdiction.

(viii) Neither MetroPCS nor any of its Subsidiaries has granted any currently effective waiver, extension or comparable consent regarding the application of the statute of limitations with respect to any Taxes or Tax Return, nor has any request for any such waiver, extension or consent been made.

(ix) Within the preceding three years, neither MetroPCS nor any of its Subsidiaries has been a distributing or controlled corporation in a transaction intended or purported to be governed by Section 355 of the Code.

(x) Neither MetroPCS nor any of its Subsidiaries has participated in a listed transaction within the meaning of Section 1.6011-4(b)(2) of the Treasury Regulations.

(xi) Neither MetroPCS nor any of its Subsidiaries is a party to any agreement, contract, undertaking, commitment, arrangement or plan that would result, and neither the execution of this Agreement nor the consummation of the Transaction (whether alone or in connection with any other related event(s)) will result, separately or in the aggregate, in the payment of any excess parachute payment within the meaning of Code Section 280G.

#### (n) Labor Matters.

(i) None of MetroPCS or its Subsidiaries is party to or otherwise bound by, and, as of the date hereof, none of MetroPCS or its Affiliates are proposing, offering or negotiating to enter into or adopt, any labor and collective bargaining agreements, contracts or other agreements or understandings with a labor union or labor organization relating to, affecting, or in any way binding on MetroPCS and its Subsidiaries.

(ii) As of the date hereof, neither MetroPCS nor any of its Subsidiaries is the subject of any proceeding, nor is any proceeding pending, or to the Knowledge of MetroPCS, threatened asserting that it has committed any material unfair labor practice or seeking to compel it to bargain with any labor union or labor organization.

(iii) There has not been since December 31, 2009, and as of the date hereof, there is not pending or, to the Knowledge of MetroPCS, threatened any material labor strike, dispute, walk-out, work stoppage, slow-down, union activity, picketing, lockout or other similar occurrence by employees of MetroPCS or its Subsidiaries.

(iv) MetroPCS and its Subsidiaries have complied in all material respects with all Laws relating to labor and employment, including those relating to wages, hours, collective bargaining, meals and rest times, unemployment compensation, worker s compensation, equal employment opportunity, age and disability discrimination, immigration control, employee classification, information privacy and security, payment and withholding of taxes and continuation coverage with respect to group health plans.

(v) Since January 1, 2009, neither MetroPCS nor any of its Subsidiaries has engaged in any plant closing or mass layoff, as defined in the WARN Act, without complying in all material respects with the notice requirements of the WARN Act.

#### (o) Intellectual Property.

(i) All Intellectual Property owned or held exclusively by MetroPCS and its Subsidiaries (<u>MetroPCS Owned Intellectual Property</u>) is exclusively owned or held (beneficially and of record, where applicable) by MetroPCS or one of its Subsidiaries, free and clear of all Encumbrances (other than MetroPCS Permitted Encumbrances), and is not subject to any open source or similar license agreement or distribution model, or to any commitments to any standards-setting or similar organization, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. The MetroPCS Owned Intellectual Property is valid, subsisting and enforceable, and is not subject to any outstanding order, judgment, decree or agreement adversely affecting MetroPCS s or its Subsidiaries use of, or their rights to, such Intellectual Property, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

(ii) To the Knowledge of MetroPCS, MetroPCS and its Subsidiaries have sufficient rights to use all material Intellectual Property used in, or necessary for the conduct of, the Business as presently conducted, all of which rights shall survive the consummation of the Transaction unchanged in all respects material to MetroPCS and its Subsidiaries, taken as a whole. MetroPCS and its Subsidiaries have taken commercially reasonable measures to protect the MetroPCS Owned Intellectual Property, and to protect the confidentiality of all Trade Secrets that are owned, used or held for use by MetroPCS and its Subsidiaries. MetroPCS and each of its Subsidiaries maintains a policy requiring that upon their hire, employees of MetroPCS and its Subsidiaries execute confidentiality and intellectual property assignment agreements which prohibit such employees from disclosing MetroPCS as and its Subsidiaries. Trade Secrets and confidential information without the written approval of an officer of MetroPCS and which assign to MetroPCS all Intellectual Property rights developed by such employees during the course of their employment with MetroPCS or its Subsidiaries.

(iii) Neither MetroPCS nor any of its Subsidiaries has infringed, misappropriated or otherwise violated the Intellectual Property rights of any third party in the past six years, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. There is no litigation, opposition, cancellation, proceeding, objection or claim pending, asserted or, to the Knowledge of MetroPCS, threatened, against MetroPCS or any of its Subsidiaries concerning the ownership, validity, registrability, enforceability, infringement or use of, or licensed right to use, any Intellectual Property, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. To the Knowledge of MetroPCS, no Person is infringing, misappropriating or otherwise violating any MetroPCS Owned Intellectual Property right of MetroPCS or its Subsidiaries in any respect material to MetroPCS and its Subsidiaries, taken as a whole.

(iv) The material IT Assets used by MetroPCS or any of its Subsidiaries operate and perform as needed by MetroPCS and its Subsidiaries to adequately conduct their respective businesses as presently conducted

and, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect, the data therein have not been subject to unauthorized access by any Person.

#### (p) Contracts.

(i) Schedule 3.3(p)(i) of the MetroPCS Disclosure Letter lists all Material MetroPCS Contracts in effect as of the date hereof. The term <u>Material</u> <u>MetroPCS Contracts</u> means all of the following types of MetroPCS Contracts (other than MetroPCS Contracts solely among or between MetroPCS and/or its direct or indirect wholly-owned Subsidiaries, Organizational Documents of MetroPCS and its Subsidiaries, MetroPCS Benefit Plans or other agreements related to employee benefits and agreements related to labor matters to the extent that such items are provided for in Sections 3.3(h) (Employee Benefits) and 3.3(n) (Labor Matters), respectively):

(A) MetroPCS Contracts evidencing Indebtedness for borrowed money with a principal amount greater than \$35,000,000;

(B) each MetroPCS Contract for distribution, supply, inventory, purchase, license or advertising or similar agreement that is reasonably likely to involve consideration of more than \$100,000,000 in the aggregate in any 12-month period, other than any such contract that can be cancelled without penalty or further payment on 90 days or less notice;

(C) MetroPCS Contracts relating to the acquisition, lease or disposition by MetroPCS or any of its Subsidiaries of assets and properties for consideration in excess of \$50,000,000, or under which MetroPCS or any of its Subsidiaries has any indemnification obligations or any other ongoing obligations that would reasonably be likely to result in payments in excess of \$15,000,000;

(D) MetroPCS Contracts that are reasonably likely to involve consideration of more than \$100,000,000 in any 12-month period or involved consideration of more than \$100,000,000 in the aggregate during calendar year 2011 or \$200,000,000 in the aggregate over the term of such MetroPCS Contract;

(E) any MetroPCS Contract that would reasonably be likely to involve consideration of more than \$20,000,000 in any 12-month period that is an interconnection, bundling or similar agreement in connection with which the equipment, networks and services of MetroPCS or any of its Subsidiaries are connected to those of another service provider in order to allow their respective customers access to each other s services and networks;

(F) any MetroPCS Contract that would reasonably be likely to involve consideration of more than \$35,000,000 in any 12-month period that is an agency, dealer, reseller, franchise or other similar contract (except for those that are terminable, without penalty, on 90 days or less notice);

(G) roaming MetroPCS Contracts that would reasonably be likely to involve payment by or expense to MetroPCS or any of its Subsidiaries of more than \$50,000,000 in any 12-month period; and

(H) MetroPCS Contracts that would reasonably be likely to involve consideration of more than \$20,000,000 in any 12-month period pursuant to which MetroPCS or any of its Subsidiaries licenses Intellectual Property to or from any Person.

(ii) Schedule 3.3(p)(ii) of the MetroPCS Disclosure Letter lists all Restricted MetroPCS Contracts in effect as of the date hereof. The term <u>Restricted MetroPCS Contracts</u> means all of the following types of MetroPCS Contracts (other than MetroPCS Contracts solely among or between MetroPCS and/or its direct or indirect wholly-owned Subsidiaries, Organizational Documents of MetroPCS and its Subsidiaries, MetroPCS Benefit Plans or other agreements related to employee benefits and agreements related to labor matters to the extent that such items are provided for in Sections 3.3(h) (Employee Benefits) and 3.3(n) (Labor Matters), respectively):

(A) joint venture, partnership, limited partnership or limited liability company agreements relating to the formation, creation, operation, existence, management or control of any joint venture, partnership, limited partnership or limited liability company that is not wholly-owned, directly or indirectly, by MetroPCS;

(B) MetroPCS Contracts that purport to limit in any material respect either the type of business in which MetroPCS or any of its Subsidiaries, or DT, TMUS or their respective Affiliates (other than MetroPCS and/or its Subsidiaries) after the Closing (other than any MetroPCS Contracts between DT, TMUS or their respective Affiliates, on the one hand, and MetroPCS and its Subsidiaries, on the other hand), may engage or the manner or locations in which any of them may so engage in any business or purport to create any material exclusive relationship;

(C) MetroPCS Contracts that could require the disposition of any material operations or line of business of MetroPCS or any of its Subsidiaries;

(D) MetroPCS Contracts that grant most favored nation status to any third party that paid or received consideration of more than \$25,000,000 in any 12-month period;

(E) any MetroPCS Contract that grants any right of first refusal, first offer or similar right to any third party that paid or received, or would reasonably be likely to pay or receive, consideration of more than \$25,000,000 during the term of such MetroPCS Contract;

(F) MetroPCS Contracts that are requirements contracts or contain volume or purchase commitments that would reasonably be likely to involve consideration of more than \$50,000,000 in any 12-month period;

(G) any MetroPCS Contract that would reasonably be likely to involve consideration of more than \$50,000,000 in any 12-month period that contains any commitment to (1) provide wireless services coverage in a particular geographic area, (2) build out tower sites in a particular geographic area, or (3) pay for a specified number of minutes of roaming usage of a third party s network regardless of the amount of actual usage (except for those that are terminable, without penalty, on 12 months or less notice);

(H) stock purchase agreements and other MetroPCS Contracts relating to the pending acquisition, lease or disposition by MetroPCS or any of its Subsidiaries of any Equity Interest of MetroPCS or any of its Subsidiaries for consideration in excess of \$5,000,000, except for stock purchase agreements or other MetroPCS Contracts solely among MetroPCS and its Subsidiaries; and

(I) MetroPCS Contracts that contain requirements to provide services using code division multiple access ( $\underline{CDM}A$ ) or would otherwise prohibit or delay the transition or migration of MetroPCS s CDMA network or CDMA customers to a new technology or network;

(iii) Prior to the date hereof, TMUS has been provided with complete and correct copies of each Material MetroPCS Contract listed on Schedule 3.3(p)(i) of the MetroPCS Disclosure Letter and each Restricted MetroPCS Contract listed on Schedule 3.3(p)(ii) of the MetroPCS Disclosure Letter, including amendments thereof and exhibits, annexes and schedules thereto. To the Knowledge of MetroPCS, as of the date hereof, each Material MetroPCS Contract, and each Restricted MetroPCS Contract described in Sections 3.3(p)(ii)(A), (C), (F), (G), (H) and (I), is in full force and effect and valid, binding and enforceable against the other parties thereto in accordance with its terms, subject to applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equity principles. None of MetroPCS Contract or Restricted MetroPCS Contract, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. To the Knowledge of MetroPCS, no event has occurred that would result in a breach of or default under, require any consent or

other action by any Person under, or give rise to any penalty or right of termination, cancellation or acceleration of any right or obligation of MetroPCS or its Subsidiaries to a loss of any benefit to which MetroPCS or any of its Subsidiaries is entitled under (in each case, with or without notice or lapse of time, or both), any Material MetroPCS Contract or Restricted MetroPCS Contract, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

#### (q) Property.

(i) MetroPCS does not have any Owned Real Property as of the date hereof.

(ii) Each of MetroPCS and its Subsidiaries has good and marketable leasehold title to all Leased Real Property of MetroPCS material to MetroPCS and its Subsidiaries, taken as a whole, free and clear of all Encumbrances except MetroPCS Permitted Encumbrances. No parcel of such Leased Real Property is subject to any governmental decree or order to be sold or is being condemned, expropriated or otherwise taken by any public authority with or without payment of compensation therefore, nor, to the Knowledge of MetroPCS, has any such condemnation, expropriation or taking been proposed, threatened or noticed, in each case, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect. Except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect, all leases of such Leased Real Property and all amendments and modifications thereto are in full force and effect and valid, binding and enforceable against the other parties thereto in accordance with its terms, subject to applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equity principles. There exists no default under any lease of such Leased Real Property by MetroPCS, any of its Subsidiaries or, to the Knowledge of MetroPCS, any of its Subsidiaries or, to the Knowledge of MetroPCS, any of its Subsidiaries or, to the Knowledge of MetroPCS, any of its Subsidiaries or, to the Knowledge of Such Leased Real Property MetroPCS and there or both, would constitute a default thereunder by TMUS, any of its Subsidiaries or, to the Knowledge of MetroPCS Material Adverse Effect. All leases of such Leased Real Property shall remain valid and binding in accordance with their terms following the Closing, except as would not, individually or in the aggregate, reasonably be likely to have a MetroPCS Material Adverse Effect.

(iii) There are no contractual or legal restrictions that preclude or restrict the ability to use any of MetroPCS s Leased Real Property in any material respect for the current or contemplated use of such real property. There are no material latent defects or material adverse physical conditions affecting such Leased Real Property. MetroPCS and its Subsidiaries have, in all material respects, valid leasehold interests in, or other valid rights to use, all plants, warehouses, distribution centers, structures and other buildings on such Leased Real Property, which are adequately maintained and are in good operating condition and repair for the requirements of the business of MetroPCS and its Subsidiaries as currently conducted in all material respects.

(iv) Except as would not, individually or in the aggregate, reasonably be likely to result in a MetroPCS Material Adverse Effect, (A) MetroPCS and its Subsidiaries have good and marketable title to or, in the case of leased assets, a valid leasehold interest in, free and clear of all Encumbrances (other than MetroPCS Permitted Encumbrances), all of the tangible personal property and assets (except for properties and assets disposed of in the ordinary course of business consistent with past practice) used in or necessary to conduct their businesses substantially as presently conducted and (B) each item of tangible personal property of MetroPCS and each of its Subsidiaries, or in which MetroPCS or any of its Subsidiaries owns an undivided interest, is in all material respects in good operating condition and repair for the requirements of the business of MetroPCS and its Subsidiaries as currently conducted, ordinary wear and tear excepted.

(r) <u>Related-Party Agreements</u>. Schedule 3.3(r) of the MetroPCS Disclosure Letter sets forth a true and complete list of all agreements to which MetroPCS or any of its Subsidiaries are party or by which MetroPCS or

any of its Subsidiaries are bound as of the date hereof that are required to be reported in the MetroPCS SEC Reports under Section 404 of Regulation S-K promulgated under the Exchange Act that are not so reported. MetroPCS has provided DT with complete and correct copies of all such agreements prior to the date hereof.

(s) <u>Prohibited Payments</u>. To the Knowledge of MetroPCS, none of MetroPCS, any of its Subsidiaries or any of their respective directors, officers, agents, employees or other Persons associated with them or acting on their behalf has (i) used any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expenses relating to political activity; (ii) made any direct or indirect unlawful payment to any foreign or domestic government official or employee from corporate funds; (iii) violated or is in violation of any provision of the Foreign Corrupt Practices Act of 1977; or (iv) made any unlawful bribe, rebate, payoff, influence payment, kickback or other payment.

(t) <u>Brokers and Finders</u>. Neither MetroPCS nor any of its officers, directors or employees has employed any broker or finder for which MetroPCS or a Subsidiary of MetroPCS is not solely responsible for such broker s or finder s fees or incurred any Liability for any brokerage fees, commissions or finder s fees in connection with the Transaction for which MetroPCS or a Subsidiary of MetroPCS is not solely responsible and the terms of which have not been disclosed to DT prior to the date hereof.

#### (u) TMUS Stock Consideration.

(i) Assuming the accuracy of DT s representations and warranties set forth in Section 7.13, upon issuance, the TMUS Stock Consideration will be fully paid, nonassessable, issued in compliance with all applicable Laws concerning the issuance of securities and not in violation of any preemptive rights, purchase option, call, right of first refusal or any similar right granted by MetroPCS, and not be subject to any voting trust agreement or other contract, agreement or arrangement to which MetroPCS or any of its Affiliates (other than TMUS and its Subsidiaries) is a party restricting or otherwise relating to the voting, dividend rights or disposition of such TMUS Stock Consideration other than the Stockholder s Agreement.

(ii) At the Closing, MetroPCS will have sufficient authorized but unissued shares or treasury shares of MetroPCS Common Stock for MetroPCS to meet its obligation to deliver the TMUS Stock Consideration under this Agreement. Upon consummation of the Transaction, DT shall acquire good and valid title to the TMUS Stock Consideration, free and clear of all Encumbrances (other than any Encumbrances relating to transfers of securities under applicable Laws and any Encumbrances permitted, granted, or required by DT or its Affiliates).

(v) <u>Rights Agreement Amendment</u>. The MetroPCS Board has approved an amendment of the MetroPCS Rights Agreement, which amendment is in full force and effect, to provide that (i) neither DT nor any of its Affiliates or associates shall be deemed to be an Acquiring Person (as defined in the MetroPCS Rights Agreement) with respect to the TMUS Stock Consideration and (ii) no Flip In Event, Flip Over Event, Distribution Date or Stock Acquisition Date (in each case, as defined in the MetroPCS Rights Agreement) shall be deemed to have occurred as a result of the execution, delivery or performance of this Agreement, or the consummation of the Transaction.

(w) <u>Required Vote of MetroPCS Stockholders</u>. The only vote of holders of MetroPCS capital stock necessary to approve the Transaction is the MetroPCS Stockholder Approval.

(x) <u>Post-Closing Restructuring</u>. Neither MetroPCS nor any of its Subsidiaries has taken or agreed to take any action (other than any action required to be taken pursuant to this Agreement) or is aware of any fact or circumstance (other than (i) the fact of the Stock Purchase, (ii) the fact of the MetroPCS Finance Transactions and (iii)(x) with respect to the MetroPCS Merger, the fact of TMUS Merger and (y) with respect to the TMUS Merger, the fact of the MetroPCS Merger) that would prevent or impede, or would be reasonably likely to

prevent or impede, the MetroPCS Merger or the TMUS Merger from qualifying as a reorganization within the meaning of Section 368(a) of the Code. There are no material intercompany gains (within the meaning of Treasury Regulation Section 1.1502-13) with respect to the stock of MetroPCS HoldCo or MetroPCS OpCo that will be required to be included in taxable income as a result of the MetroPCS Merger or the TMUS Merger.

(y) <u>No Other Representations or Warranties</u>. Except for the representations and warranties contained in Sections 3.1, 3.2 and 7.13 and any representation contained in any certificate delivered pursuant to Section 5.2, MetroPCS acknowledges that neither DT, nor any Subsidiary of DT or any other Person on behalf of DT (including any Representative of DT), makes any express or implied representation or warranty with respect or relating to DT, any of its Subsidiaries, or any information provided to MetroPCS or any other Person, or MetroPCS is use of any such information, including any information, documents, projections, forecasts or other material made available to MetroPCS in certain data rooms or management presentations in expectation of the Transaction, and MetroPCS has not relied on such information or any other representation or warranty not set forth in this Agreement.

#### ARTICLE IV

#### **COVENANTS**

4.1 Interim Operations of TMUS. During the period from the date hereof and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Closing, except (1) as may be required by Law, (2) with MetroPCS s prior written consent, not to be unreasonably withheld, conditioned or delayed (and in no event to be delayed more than ten days following DT s written request for consent), (3) as otherwise expressly provided in this Agreement, or (4) as provided on Schedule 4.1 of the TMUS Disclosure Letter, TMUS shall, DT shall cause TMUS, and DT and TMUS shall cause each of TMUS s Subsidiaries, (i) to conduct its business in the ordinary course and, to the extent consistent therewith, use its commercially reasonable efforts to (A) preserve its business organizations intact, (B) maintain existing relations and goodwill with Governmental Entities, customers, suppliers, distributors, dealers, retailers, creditors, lessors, employees and business associates, and (C) keep available the services of its present employees, officers and agents; and (ii) not to:

(a) amend its Organizational Documents, unless such amendment would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction;

(b) merge or consolidate with any other Person or authorize, recommend, propose or announce an intention to adopt a plan of complete or partial liquidation, dissolution, consolidation, restructuring, recapitalization or any other reorganization, in each case other than transactions solely among TMUS and its Subsidiaries;

(c) acquire assets, unless (1) in the ordinary course of business consistent with past practice or in material compliance with the provisions for expenses and capital expenditures in the TMUS Business Plan or otherwise in compliance with Section 4.1(j), (2) from any other Person with a value or purchase price in the aggregate not exceeding \$100,000,000, (3) in acquisitions of spectrum licenses for consideration not exceeding \$100,000,000 in the aggregate or (4) pursuant to any agreement in effect on the date hereof for consideration not exceeding \$25,000,000 individually; provided that any transaction under (1), (2), (3) or (4) of this clause would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction;

(d) issue, sell, pledge, dispose of, grant, transfer, Encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, lease, license, guarantee or other Encumbrance of, any Equity Interests of TMUS or any of its Subsidiaries (other than the issuance of Equity Interests (i) by a wholly-owned Subsidiary of TMUS to TMUS or another wholly-owned Subsidiary or (ii) by TMUS to DT, Global or Holding), securities convertible or exchangeable into, or exercisable for, any such Equity Interests or any options, warrants or other rights of any kind to acquire any such Equity Interests or such convertible or exchangeable securities;

(e) enter into any agreement with respect to the voting of its Equity Interests;

(f) create or incur any Encumbrance (other than a TMUS Permitted Encumbrance or in the ordinary course of business consistent with past practice) on the assets of TMUS or any of its Subsidiaries that, individually or in the aggregate, is material to TMUS and its Subsidiaries, taken as a whole, or that would reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction;

(g) make any loans, advances, guarantees or capital contributions to or investments in any Person in excess of \$50,000,000, other than (x) any of the foregoing to or on behalf of TMUS or any of its direct or indirect wholly-owned Subsidiaries, (y) in the ordinary course of business consistent with past practice and which would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction, or (z) in compliance with Section 4.1(j);

(h) declare, set aside, make or pay any non-cash distributions or dividends, payable in stock, property or otherwise, with respect to any of its Equity Interests (except for non-cash distributions paid by any direct or indirect wholly-owned Subsidiary to TMUS or to any other direct or indirect wholly-owned Subsidiary), it being understood that TMUS and its Subsidiaries may make or pay any cash distributions or dividends, including to Holding;

(i)(i) incur any Indebtedness for borrowed money, or issue or sell any debt securities or warrants or other rights to acquire any debt security of TMUS or any of its Subsidiaries, in each case other than Indebtedness contemplated by Section 4.25, the DT Notes, the MetroPCS Existing Notes, Indebtedness under the MetroPCS Existing Finance Documents or any Hedge Agreements to which MetroPCS or any of its Subsidiaries is a party and which have been provided to DT prior to the date hereof, or the Permitted MetroPCS Notes, or (ii) amend, modify, supplement or waive the terms of any existing Indebtedness, debt securities or warrants or other rights to acquire debt securities of TMUS or any of its Subsidiaries, except in the ordinary course of business consistent with past practice (and in any case not in any manner that would increase the principal amount owed thereunder); provided, however, that none of the foregoing shall prohibit TMUS or any of its Subsidiaries from incurring any guarantee or support arrangement of obligations of TMUS or its wholly-owned Subsidiaries otherwise permitted hereunder or incurring, amending, modifying, supplementing, or waiving any Indebtedness (A) in the ordinary course of business consistent with past practice; (B) as contemplated by the TMUS Business Plan, or (C) solely among or between TMUS and/or its direct or indirect wholly-owned Subsidiaries;

(j) make or authorize any payment of, or accrual or commitment for, capital expenditures in excess of \$200,000,000 (or \$400,000,000 in the event of an increase in data demand in the Business significantly in excess of the demand anticipated on the date hereof) in the aggregate in any consecutive six-month period and incremental to any capital expenditures set forth in the TMUS Business Plan;

(k)(i) enter into any TMUS Contract that would be a Restricted TMUS Contract if in effect on the date hereof, (ii) amend or supplement any TMUS Contract that is not a Restricted TMUS Contract in a manner that would cause it to be a Restricted TMUS Contract if it had been so amended or supplemented as of the date hereof; <u>provided</u>, that nothing in these clauses (i) and (ii) shall prohibit TMUS from entering into any joint venture, partnership, limited partnership or limited liability company agreements relating to the formation, creation, operation, existence, management or control of any joint venture, partnership, limited partnership or limited partnership or limited liability company that is not wholly-owned, directly or indirectly, by TMUS if the sum of the initial capitalization thereof and binding capital commitments thereto does not exceed \$20,000,000 individually, or \$50,000,000 in the aggregate, (iii) amend, supplement, extend, or renew any Restricted TMUS Contract in a manner that would reasonably be expected to be material and adverse to TMUS and its Subsidiaries or, to the extent purporting to bind MetroPCS and its Affiliates (other than TMUS and its Subsidiaries) after the Closing, material and adverse to MetroPCS and its Affiliates, in each case, taking into account the terms of such amendment, supplement, extension or renewal as a whole, or (iv) fail to provide MetroPCS written notice, within

ten Business Days of entry, of the entry into any TMUS Contract that would be a Material TMUS Contract if in effect on the date hereof; <u>provided</u>, that nothing in this clause (v) shall require TMUS to violate any confidentiality or disclosure obligations it has with respect to such TMUS Contract or to provide written notice of, and/or disclose to MetroPCS any portion of any Material TMUS Contracts containing sales, pricing, marketing, customer care, commissions, or other related commercially sensitive information;

(l)(i) enter into any Intercompany Contract (A) involving aggregate consideration payable by either party thereto of greater than \$10,000,000 individually or \$50,000,000 in the aggregate or (B) on terms and conditions other than arm s length terms and conditions, or (ii) amend, extend, renew, modify or waive any Intercompany Contract in any manner that would result in TMUS or its Subsidiaries paying to the other parties thereto aggregate consideration per applicable period greater than \$10,000,000 individually or \$50,000,000 in the aggregate than provided for in such Intercompany Contract as of the date hereof or, if later, the date of such Intercompany Contract; provided, that nothing in this clause (l) shall prevent TMUS in the ordinary course of business to continue, extend, renew or enter into Intercompany Contracts on terms and conditions other than arm s length terms and conditions that are substantially equivalent (but in no event less favorable to TMUS and its Subsidiaries) in all material respects to the ones currently in force as of the date hereof;

(m)(i) make any changes with respect to material financial accounting policies or procedures, except as required (A) by changes in GAAP, international financial reporting standards or Regulation S-X of the SEC, (B) by a Governmental Entity or quasi-Governmental Entity (including the Financial Accounting Standards Board or any similar organization), or (C) by a change in applicable Law, or (ii) write up, write down or write off the book value of any of its assets, other than in the ordinary course of business consistent with past practice or as may be consistent with TMUS s financial accounting policies and procedures and GAAP (including as may be required by GAAP as a result of the announcement or pendency of the Transaction);

(n)(i) enter into any line of business other than the Business, (ii) except as currently conducted, engage in the conduct of any business that would require the receipt or transfer of a TMUS Communications License or any other License issued by any Governmental Entity authorizing operation or provision of any communication services or foreign country that would require the receipt or transfer of, or application for, a License to the extent such License would reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction, or (iii) enter into any business or operations outside of the Territory other than in support of business or operations within the Territory;

(o) apply for, request or file for any TMUS License the receipt of which would reasonably be likely to prevent, materially delay or materially impair the consummation of the Transaction;

(p) settle any litigation or other proceedings pending or threatened before a Governmental Entity against TMUS or its Subsidiaries, except (i) to the extent such settlement is subject to and not in excess of the reserves that relate to such litigation or proceedings set forth in TMUS s consolidated unaudited balance sheet as of June 30, 2012, or (ii) for a payment by TMUS or any of its Subsidiaries in an amount less than \$50,000,000 for any individual or group of related settlements (excluding any amounts that may be paid under existing Insurance Policies and any settlements permitted pursuant to clause (i)), without the imposition of ongoing restrictions (A) prior to the Closing, on TMUS or any of its Subsidiaries that are material to TMUS and its Subsidiaries, taken as a whole, or (B) after the Closing, on MetroPCS or any of its Subsidiaries that are material to MetroPCS and its Subsidiaries, taken as a whole;

(q) make or change any Tax election, change any method of Tax accounting, settle or finally resolve any controversy with respect to Taxes for an amount that exceeds the amount reserved with respect thereto in the most recent TMUS Financial Statements, or file any amended Tax Return, in each case, if such action would have a material and adverse effect on TMUS and its Subsidiaries, taken as a whole;

(r) transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon or allow to lapse or expire or otherwise dispose of (in each case except among TMUS and its wholly-owned Subsidiaries) any TMUS Communications Licenses or wireless spectrum other than (i) point-to-point microwave licenses, business radio licenses and experimental licenses, (ii) exchanges of spectrum licenses within the same FCC defined market area (<u>FMA</u>) that improve contiguity of TMUS s and its Subsidiaries spectrum within such FMA and do not worsen the expected contiguity between TMUS and MetroPCS and their respective Subsidiaries, (iii) exchanges of spectrum licenses, in which TMUS or its Subsidiaries would transfer TMUS Communications Licenses covering 300,000,000 Licensed MHz POPs or fewer in the aggregate and involving cash consideration of \$10,000,000 or less in the aggregate to any Person other than a party hereto or to TMUS or its Subsidiaries, which do not adversely affect existing TMUS operations or the expected benefits of the Transaction to TMUS and MetroPCS in any material respect, or (iv) in the ordinary course of business consistent with past practice; <u>provided</u> that any of the transaction under (i), (ii), (iii), or (iv) of this clause would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction;

(s) transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon or allow to lapse or expire or otherwise dispose of (in each case except among TMUS and its wholly-owned Subsidiaries), any Licenses (other than TMUS Communications Licenses), assets (other than wireless spectrum), operations, rights, businesses or interests therein of TMUS or its Subsidiaries, unless (i) in the ordinary course of business consistent with past practice, (ii) to any other Person with a value or purchase price in the aggregate not exceeding \$100,000,000, or (iii) pursuant to any agreement in effect on the date hereof for consideration not exceeding \$25,000,000 individually;

(t) other than as required pursuant to the terms and conditions of any TMUS Benefit Plan as in effect on the date hereof or as may be required by applicable Law, (i) terminate, establish, adopt or amend any TMUS Benefit Plan other than the adoption of annual TMUS Benefit Plans in the ordinary course of business consistent with past practice and amendments to plans (other than retention and severance plans) that in each case do not increase benefits upon the consummation of the Transaction or otherwise materially increase benefits or result in materially increased administrative costs, (ii) grant any salary or wage increase, other than in the ordinary course of business consistent with past practice (it being understood that increases in base or hourly salary and wages for employees by no more than 4.0% in the aggregate shall be deemed to be in the ordinary course of business consistent with past practice), (iii) pay aggregate bonus or incentive compensation other than in the ordinary course consistent with past practice, (iv) (x) grant any new compensation award, other than cash bonus awards and cash-based long term incentive compensation awards, in each case in amounts and on terms that are in the ordinary course of business consistent with past practice; provided, however, that no new awards shall be granted under the Phantom Share Plan, (y) amend the terms of outstanding compensation awards other than in a manner that does not increase the amounts payable or accelerate the vesting, payment or timing of any payment under such awards and in the ordinary course of business consistent with past practice, or (z) materially increase the compensation opportunity under any TMUS Benefit Plan, (v) pay any severance other than in the ordinary course of business consistent with past practice in connection with employees entering into and not revoking a release of claims against TMUS in connection with terminations of employment, (vi) take any action to increase or accelerate the vesting, payment or timing, or fund or secure the payment, of any amounts under any TMUS Benefit Plan, (vii) change any assumptions used to calculate funding or contribution obligations under any TMUS Benefit Plan, other than as required by GAAP, (viii) provide, forgive or modify any loans to directors, officers or employees of TMUS or any of its Subsidiaries; or (ix) take any of the actions described in Schedule 4.1(t) of the TMUS Disclosure Letter:

(u) transfer, sell, lease, license, divest or otherwise dispose of any transmission towers owned or leased by TMUS or any of its Subsidiaries (it being understood that the foregoing shall not prohibit a disposition of the Tower Assets, dispositions of transmission towers solely among TMUS and its Subsidiaries, the decommissioning or sale of towers in the ordinary course of business, in one or more transactions that are scheduled to close prior to the Closing Date);

(v) except with respect to definitive agreements that have been executed by the parties thereto prior to the date hereof, purchase, lease or otherwise acquire any wireless spectrum that would reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction or result in required divestitures of assets;

(w) make a fundamental change to the network technologies or principal billing systems of TMUS and its Subsidiaries (excluding system upgrades, improvements and modernization, equipment replacement and similar matters consistent with the TMUS Business Plan);

(x) authorize, or commit, resolve, announce, offer, agree or enter into an agreement to do or take, any of the foregoing actions or any actions inconsistent with the foregoing.

4.2 Interim Operations of MetroPCS. During the period from the date hereof and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Closing, except (1) as may be required by Law, (2) with DT s prior written consent, not to be unreasonably withheld, conditioned or delayed (and in no event to be delayed more than ten days following MetroPCS s written request for consent), (3) as otherwise expressly provided in this Agreement, or (4) as provided on Schedule 4.2 of the MetroPCS Disclosure Letter, MetroPCS shall, and shall cause each of its Subsidiaries, (i) to conduct its business in the ordinary course and, to the extent consistent therewith, use its commercially reasonable efforts to (A) preserve its business organizations intact, (B) maintain existing relations and goodwill with Governmental Entities, customers, suppliers, distributors, dealers, retailers, creditors, lessors, employees and business associates, and (C) keep available the services of its present employees, officers and agents; and (ii) not to:

(a) amend its Organizational Documents, unless such amendment would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction;

(b) merge or consolidate with any other Person or authorize, recommend, propose or announce an intention to adopt a plan of complete or partial liquidation, dissolution, consolidation, restructuring, recapitalization or any other reorganization, in each case other than transactions solely among MetroPCS and its Subsidiaries;

(c) acquire assets, unless (i) in the ordinary course of business consistent with past practice or in material compliance with the provisions for expenses and capital expenditures in the MetroPCS Business Plan or otherwise in compliance with Section 4.2(j), (ii) from any other Person with a value or purchase price in the aggregate not exceeding \$50,000,000 other than any acquisition of spectrum, (iii) in acquisitions of spectrum licenses for consideration not exceeding \$35,000,000 in the aggregate, or (iv) pursuant to any agreement in effect on the date hereof and whose value does not exceed \$10,000,000 individually; provided that any transaction under clauses (i), (ii), (iii) or (iv) of this clause (c) would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction;

(d) issue, sell, pledge, dispose of, grant, transfer, Encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, lease, license, guarantee or other Encumbrance of, any Equity Interests of MetroPCS or any of its Subsidiaries, securities convertible or exchangeable into, or exercisable for, any such Equity Interests or any options, warrants or other rights of any kind to acquire any such Equity Interests or such convertible or exchangeable securities; <u>provided</u>, <u>however</u>, that none of the foregoing shall prohibit (i) any pledge, grant or Encumbrance permitted or required under the MetroPCS Existing Finance Documents, (ii) the issuance of MetroPCS Common Stock upon the conversion or exercise of securities granted under the MetroPCS Benefit Plans and outstanding as of the date hereof according to their respective terms or the terms hereof (including to reflect the MetroPCS Reverse Stock Split, the Cash Payment and any cash in lieu of fractional shares to be paid pursuant to Section 2.1(f)(v)) or the issuance of MetroPCS Common Stock in respect of awards of MetroPCS Restricted Stock granted after the date hereof pursuant to clause (iv) of this Section 4.2(d), (iii) the issuance by a wholly-owned Subsidiary of MetroPCS of capital stock of such Subsidiary to such Subsidiary s

parent or another wholly-owned MetroPCS Subsidiary, (iv) (A) grants of MetroPCS Stock Options (with an exercise price determined in accordance with the applicable MetroPCS Benefit Plan) and MetroPCS Restricted Stock to officers and employees of MetroPCS and its Subsidiaries hired or promoted in each case on or after the date hereof in the ordinary course of business consistent with past practice, and (B) annual grants of MetroPCS Stock Options and MetroPCS Restricted Stock to officers, employees and directors in each case at such times, to such persons, and in such amounts as are in the ordinary course of business consistent with past practice, such grants under clauses (A) and (B) to be for an aggregate number of shares of MetroPCS Common Stock set forth on Schedule 4.2(d)(iv) of the MetroPCS Disclosure Letter, or (vi) the issuance of MetroPCS Common Stock in satisfaction of obligations under the MetroPCS Benefit Plans as of the date hereof in the ordinary course of business consistent with past practice;

(e) enter into any agreement with respect to the voting of its Equity Interests;

(f) create or incur any Encumbrance (other than a MetroPCS Permitted Encumbrance or in the ordinary course of business consistent with past practice) on the assets of MetroPCS or any of its Subsidiaries that, individually or in the aggregate, is material to MetroPCS and its Subsidiaries, taken as a whole, or that would reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction;

(g) make any loans, advances, guarantees or capital contributions to or investments in any Person in excess of 20,000,000, other than (x) any of the foregoing to or on behalf of MetroPCS or any of its direct or indirect wholly-owned Subsidiaries, (y) in the ordinary course of business consistent with past practice and which would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction, or (z) in compliance with Section 4.2(j);

(h) declare, set aside, make or pay (i) any cash distributions or dividends except for distributions or dividends by any direct or indirect wholly-owned Subsidiary of MetroPCS to MetroPCS or to any other direct or indirect wholly-owned Subsidiary of MetroPCS or (ii) any non-cash distributions or dividends, payable in stock, property or otherwise, with respect to any of its Equity Interests;

(i)(i) incur any Indebtedness for borrowed money, or issue or sell any debt securities or warrants or other rights to acquire any debt security of MetroPCS or any of its Subsidiaries, other than the Permitted MetroPCS Notes and in the ordinary course of business under MetroPCS s existing revolving credit facility under the MetroPCS Existing Finance Documents, or (ii) amend, modify, supplement or waive the terms of any existing Indebtedness, debt securities or warrants or other rights to acquire debt securities of MetroPCS or any of its Subsidiaries, except for MetroPCS Consent Offers or otherwise in the ordinary course of business consistent with past practice (and in any case not in any manner that would increase the principal amount owed thereunder); provided, however, that (a) the foregoing shall not prohibit MetroPCS or any of its Subsidiaries from obtaining the MetroPCS Consent Offers, and (b) with respect to clause (ii) above, and except with respect to the MetroPCS Existing Notes, none of the foregoing shall prohibit MetroPCS or any of its Subsidiaries from incurring any guarantee, security interests or other support arrangement of obligations of MetroPCS or its wholly-owned Subsidiaries otherwise permitted hereunder or required under the MetroPCS Existing Finance Documents, or incurring, amending, modifying, supplementing, or waiving any Indebtedness in a manner that would not reasonably be expected to impair or delay the issuance of DT Notes, impair or cause any default under the MetroPCS Existing Notes, or impose any post-Closing restrictions (other than any restrictions in connection with guarantees, security interests or other support arrangements by TMUS or its Affiliates of obligations of MetroPCS or its wholly-owned Subsidiaries otherwise permitted hereunder) on TMUS or its Affiliates (other than MetroPCS and its Subsidiaries as of the date hereof), and (A) in the ordinary course of business consistent with past practice, (including under the MetroPCS Existing Finance Documents), (B) as contemplated by the MetroPCS Business Plan, or (C) solely among or between MetroPCS and/or its direct or indirect wholly-owned Subsidiaries;

(j) make or authorize any payment of, or accrual or commitment for, capital expenditures in excess of \$70,000,000 (or \$135,000,000 in the event of an increase in data demand in the Business significantly in excess of the demand anticipated on the date hereof) in the aggregate in any consecutive six-month period and incremental to any capital expenditures set forth in the MetroPCS Business Plan;

(k) in each case except as required under the MetroPCS Existing Finance Documents, (i) enter into any MetroPCS Contract that would be a Restricted MetroPCS Contract in a manner that would cause it to be a Restricted MetroPCS Contract if it had been so amended or supplemented as of the date hereof; provided, that nothing in these clauses (i) and (ii) shall prohibit MetroPCS from entering into any joint venture, partnership, limited partnership or limited liability company agreements relating to the formation, creation, operation, existence, management or control of any joint venture, partnership, limited partnership or limited liability company that is not wholly-owned, directly or indirectly, by MetroPCS if the sum of the initial capitalization thereof and binding capital commitments thereto do not exceed \$10,000,000 individually, or \$50,000,000 in the aggregate, (iii) amend, supplement, extend, or renew any Restricted MetroPCS Contract in a manner that would reasonably be expected to be material and adverse to MetroPCS and its Subsidiaries or, to the extent purporting to bind DT and its Affiliates (other than TMUS and its Subsidiaries) after the Closing, material and adverse to DT and its Affiliates, in each case, taking into account the terms of such amendment, supplement, extension or renewal as a whole, or (iv) fail to provide DT written notice, within ten Business Days of entry, of the entry into any MetroPCS Contract that would be a Material MetroPCS Contract if in effect on the date hereof; provided, that nothing in this clause (iv) shall require MetroPCS to violate any confidentiality or disclosure obligations it has with respect to such MetroPCS Contract or to provide written notice of, and/or disclose to DT any portion of any Material MetroPCS Contracts containing sales, pricing, marketing, customer care, commissions, or other related commercially sensitive information;

(l)(i) make any changes with respect to material financial accounting policies or procedures, except as required (A) by changes in GAAP or Regulation S-X of the SEC, (B) by a Governmental Entity or quasi-Governmental Entity (including the Financial Accounting Standards Board or any similar organization), or (C) by a change in applicable Law, or (ii) write up, write down or write off the book value of any of its assets, other than in the ordinary course of business consistent with past practice or as may be consistent with MetroPCS s financial accounting policies and procedures and GAAP (including as may be required by GAAP as a result of the announcement or pendency of the Transaction);

(m)(i) enter into any line of business other than the Business, (ii) except as currently conducted, engage in the conduct of any business that would require the receipt or transfer of a MetroPCS Communications License or any other License issued by any Governmental Entity authorizing operation or provision of any communication services or foreign country that would require the receipt or transfer of, or application for, a License to the extent such License would reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction, or (iii) enter into any business or operations outside of the Territory other than in support of business or operations within the Territory;

(n) apply for, request or file for any MetroPCS License the receipt of which would reasonably be likely to prevent, materially delay or materially impair the consummation of the Transaction;

(o) settle any litigation or other proceedings pending or threatened before a Governmental Entity against MetroPCS or its Subsidiaries, except (i) to the extent such settlement is subject to and not in excess of the reserves that relate to such litigation or proceedings set forth in MetroPCS s consolidated unaudited balance sheet as of June 30, 2012, or (ii) for a payment by MetroPCS or any of its Subsidiaries in an amount less than \$12,500,000 for any individual or group of related settlements (excluding any amounts that may be paid under existing Insurance Policies and any settlements permitted pursuant to clause (i)), without the imposition of ongoing restrictions on MetroPCS or any of its Subsidiaries material to MetroPCS and its Subsidiaries taken as a whole;

(p) make or change any Tax election, change any method of Tax accounting, settle or finally resolve any controversy with respect to Taxes for an amount that exceeds the amount reserved with respect thereto in the most recent MetroPCS Financial Statements, or file any amended Tax Return, in each case, if such action would have a material and adverse effect on MetroPCS and its Subsidiaries, taken as a whole;

(q) transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon or allow to lapse or expire or otherwise dispose of (in each case except among MetroPCS and its wholly-owned Subsidiaries) any MetroPCS Communications Licenses or wireless spectrum other than (i) point-to-point microwave licenses, business radio licenses and experimental licenses, (ii) exchanges of spectrum licenses within the same FMA that improve contiguity of MetroPCS s and its Subsidiaries spectrum within such FMA but do not worsen the expected contiguity between MetroPCS and TMUS and their respective Subsidiaries, or (iii) exchanges of spectrum licenses, in which MetroPCS or its Subsidiaries would transfer MetroPCS Communications Licenses covering 150,000,000 Licensed MHz POPs or fewer in the aggregate to any Person other than a party hereto or to MetroPCS communications Licenses so transferred and (B) cash consideration of \$100,000,000 or less in the aggregate which exchange does not adversely affect existing MetroPCS operations or the expected benefits of the Transaction to TMUS and MetroPCS in any material respect; provided that any of the transactions under clauses (i), (ii) or (iii) of this clause (q) would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction;

(r) transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon or allow to lapse or expire or otherwise dispose of (in each case except among MetroPCS and its wholly-owned Subsidiaries), any Licenses (other than MetroPCS Communications Licenses), assets (other than wireless spectrum), operations, rights, businesses or interests therein of MetroPCS or its Subsidiaries, unless (i) in the ordinary course of business consistent with past practice, (ii) to any other Person with a value or purchase price in the aggregate not exceeding \$50,000,000, or (iii) pursuant to any agreement in effect on the date hereof for consideration not exceeding \$10,000,000 individually;

(s) other than as required pursuant to the terms and conditions of any MetroPCS Benefit Plan as in effect on the date hereof or as may be required by applicable Law, (i) terminate, establish, adopt or amend any MetroPCS Benefit Plan other than the adoption of annual MetroPCS Benefit Plans in the ordinary course of business consistent with past practice and amendments to plans (other than retention and severance plans) that in each case do not increase benefits upon the consummation of the Transaction or otherwise materially increase benefits or result in materially increased administrative costs, (ii) grant any salary or wage increase, other than in the ordinary course of business consistent with past practice (it being understood that increases in base or hourly salary and wages for employees by no more than 4.0% in the aggregate shall be deemed to be in the ordinary course of business consistent with past practice), (iii) pay aggregate bonus or incentive compensation other than in the ordinary course consistent with past practice, (iv) (x) grant any new compensation award, other than (A) cash bonus awards and cash-based long term incentive compensation awards, in each case in amounts and on terms that are in the ordinary course of business consistent with past practice, (B) cash awards to retain employees (excluding officers) for not more than the amount set forth on Schedule 4.2(s)(iv)(x)(B) of the MetroPCS Disclosure Letter in the aggregate and otherwise according to the terms and conditions of Schedule 4.2(s)(iv)(x)(B) of the MetroPCS Disclosure Letter, and (C) MetroPCS Stock Options and MetroPCS Restricted Stock to the extent permitted by clause (iv) of Section 4.2(d), (y) amend the terms of outstanding compensation awards other than (A) in a manner that does not increase the amounts payable or accelerate the vesting, payment or timing of any payment under such awards and in the ordinary course of business consistent with past practice, or (B) as otherwise specifically contemplated by Section 2.1(d)(iv)(y) hereof, or (z) materially increase the compensation opportunity under any MetroPCS Benefit Plan, (v) pay any severance other than in the ordinary course of business consistent with past practice in connection with employees entering into and not revoking a release of claims against MetroPCS in connection with terminations of employment, (vi) take any

action to increase or accelerate the vesting, payment or timing, or fund or secure the payment, of any amounts under any MetroPCS Benefit Plan, (vii) change any assumptions used to calculate funding or contribution obligations under any MetroPCS Benefit Plan, other than as required by GAAP, (viii) provide, forgive or modify any loans to directors, officers or employees of MetroPCS or any of its Subsidiaries; or (ix) take any of the actions described in Schedule 4.2(s) of the TMUS Disclosure Letter;

(t) transfer, sell, lease, license, divest or otherwise dispose of any transmission towers owned or leased by MetroPCS or any of its Subsidiaries (it being understood that the foregoing shall not prohibit dispositions of transmission towers solely among MetroPCS and its Subsidiaries, the decommissioning or sale of towers in the ordinary course of business in one or more transactions that are scheduled to close prior to the Closing Date);

(u) except with respect to definitive agreements that have been executed by the parties thereto prior to the date hereof, purchase, lease or otherwise acquire any wireless spectrum that would reasonably be expected to prevent, materially delay or materially impair the consummation of the Transaction or result in required divestitures of assets;

(v) make a fundamental change to the network technologies or principal billing systems of MetroPCS and its Subsidiaries (excluding system upgrades, improvements and modernization, equipment replacement and similar matters, in each case within the same fundamental framework of such network technologies and billing systems consistent with the MetroPCS Business Plan);

(w) take any action, which action would prevent or impede, or would be reasonably likely to prevent or impede, the MetroPCS Merger or TMUS Merger from qualifying as a reorganization within the meaning of Section 368(a) of the Code; or

(x) authorize, or commit, resolve, announce, offer, agree or enter into an agreement to do or take, any of the foregoing actions or any actions inconsistent with the foregoing.

#### 4.3 Proxy Statement.

(a) MetroPCS and DT shall cooperate to, and MetroPCS shall, as promptly as practicable following the date hereof, and in no event later than 45 calendar days following the date hereof (unless DT fails to furnish information relating to TMUS that is required by the Exchange Act and the rules and regulations promulgated thereunder to be included in the Proxy Statement, or delays furnishing such information to a time when it is no longer practicable for MetroPCS to meet such deadline, in which case such deadline shall be extended to the first practicable date following the furnishing of such information), prepare and file with the SEC the Proxy Statement. DT shall, and shall cause its Subsidiaries to, promptly furnish to MetroPCS all the information relating to them required by the Exchange Act and the rules and regulations promulgated thereunder to be included in the Proxy Statement and any other information reasonably requested by MetroPCS. MetroPCS shall use its reasonable best efforts, and DT shall cooperate with MetroPCS, to resolve all SEC comments with respect to the Proxy Statement as promptly as practicable after receipt thereof. MetroPCS shall use its reasonable best efforts to cause the Proxy Statement in definitive form to be mailed to the MetroPCS Stockholders as promptly as practicable following the clearance of the Proxy Statement by the SEC.

(b) DT hereby covenants and agrees with MetroPCS that the Proxy Statement (at the time it is first mailed to the MetroPCS Stockholders and at the time of the MetroPCS Stockholders Meeting) will not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading; <u>provided</u>, <u>however</u>, that this Section 4.3(b) shall apply only to information contained in the Proxy Statement that was supplied by DT or any of its Subsidiaries expressly for inclusion therein.

(c) MetroPCS hereby covenants and agrees with DT that the Proxy Statement (at the time it is first mailed to the MetroPCS Stockholders and at the time of the MetroPCS Stockholders Meeting) will not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading; <u>provided</u>, <u>however</u>, that this Section 4.3(c) shall not apply to information contained in the Proxy Statement that was supplied by DT or any of its Subsidiaries expressly for inclusion therein.

(d) If at any time prior to obtaining the MetroPCS Stockholder Approval, any information relating to a party hereto, or any of its respective Affiliates, officers or directors, should be discovered by such party that should be set forth in an amendment or supplement to the Proxy Statement so that such document would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, such party shall promptly notify the other parties hereto in writing thereof and MetroPCS will use its reasonable best efforts to file an appropriate amendment or supplement describing such information with the SEC and, to the extent required under applicable Law, disseminate such amendment or supplement to the MetroPCS Stockholders Meeting; provided that the delivery of such notice and the filing or dissemination of any such amendment or supplement shall not affect or be deemed to modify any representation or warranty made by any party hereunder or otherwise affect the remedies available hereunder to any party.

#### 4.4 MetroPCS Stockholders Meeting.

(a) Provided that there shall not have been a MetroPCS Adverse Recommendation Change permitted by Section 4.5(e), MetroPCS shall include in the Proxy Statement the recommendation of the MetroPCS Board that the MetroPCS Stockholders approve the New MetroPCS Certificate and the MetroPCS Share Issuance (the <u>MetroPCS Recommendation</u>) and use reasonable best efforts to (i) solicit from the MetroPCS Stockholders proxies in favor of the approval of the New MetroPCS Certificate and the MetroPCS Share Issuance and (ii) take all other actions necessary or advisable to secure such approval.

(b) MetroPCS agrees to duly call, give notice of, convene and hold the MetroPCS Stockholders Meeting in no event later than 45 Business Days following clearance of the Proxy Statement by the SEC; provided that (i) (A) MetroPCS may postpone or adjourn the MetroPCS Stockholders Meeting to a date no more than 15 days after its originally noticed date, and (B) MetroPCS shall postpone or adjourn the MetroPCS Stockholders Meeting to a date no more than 15 days after its originally noticed date at the request of DT, in each case of clauses (A) and (B) only to the extent reasonably required in order to solicit additional proxies so as to establish a quorum and obtain the MetroPCS Stockholder Approval, and (ii) MetroPCS may postpone or adjourn the MetroPCS Stockholders Meeting to a amended disclosure document that the MetroPCS Board has determined in good faith (after consultation with its outside legal counsel) is necessary or required to be filed and disseminated under applicable Laws.

#### 4.5 No Solicitation MetroPCS.

(a) Upon execution of this Agreement, MetroPCS and its Subsidiaries shall, and shall cause their respective directors and officers and shall use their reasonable best efforts to cause their other Representatives to, cease and terminate any and all existing activities, discussions or negotiations with any Person with respect to a MetroPCS Acquisition Proposal. MetroPCS shall promptly after the date hereof instruct each Person that has heretofore executed a confidentiality agreement relating to a MetroPCS Acquisition Proposal with or for the benefit of MetroPCS promptly to return to MetroPCS or destroy all information, documents, and materials relating to the MetroPCS Acquisition Proposal or to MetroPCS or its businesses, operations or affairs heretofore furnished by MetroPCS or any of its Representatives to such Person or any of its Representatives in accordance with the terms of any confidentiality agreement with such Person.

(b) Except as provided in Section 4.5(c), MetroPCS agrees that neither it nor any of its Subsidiaries shall, that it shall not authorize or permit any of its and their respective directors and officers to, and that it shall not authorize, and shall use its reasonable best efforts not to permit, any of its and their other respective Representatives to, directly or indirectly, (i) initiate, solicit or knowingly encourage or knowingly take or continue any other action to facilitate the submission of any inquiry, indication of interest, proposal or offer that constitutes, or would reasonably be expected to lead to, a MetroPCS Acquisition Proposal, (ii) participate in any discussions or negotiations regarding, or that would reasonably be expected to lead to, any MetroPCS acquisition Proposal (other than to inform a Person of the existence of this Section 4.5), (iii) furnish any non-public information or data regarding MetroPCS or any of its Subsidiaries to, or afford access to the properties, personnel, books and records of MetroPCS to, any Person (other than DT and its Subsidiaries) in connection with or in response to or in circumstances that would reasonably be expected to lead to, any MetroPCS Acquisition Proposal, (iv) take any action to make the provisions of any fair price, moratorium, control share acquisition, business combination or other similar anti-takeover statute or regulation (including any transaction under, or a third party becoming an interested stockholder under, Section 203 of the DGCL), or any restrictive provision of any applicable anti-takeover provision in MetroPCS s Organizational Documents or the MetroPCS Acquisition Proposal, or (v) resolve or agree to do any of the foregoing.

(c) Notwithstanding Section 4.5(b) or anything to the contrary contained in this Agreement, from the date hereof and prior to the receipt of the MetroPCS Stockholder Approval, if (i) MetroPCS or its Representatives receive a *bona fide*, unsolicited written MetroPCS Acquisition Proposal that did not result from a breach of this Section 4.5 and that the MetroPCS Board determines in good faith, after consultation with its outside legal counsel and financial advisors, constitutes, or is reasonably likely to result in, a MetroPCS Superior Proposal, (ii) the MetroPCS Board concludes in good faith, after consultation with its outside legal counsel, that the failure to take such action with respect to such MetroPCS Acquisition Proposal would be reasonably likely to be inconsistent with the MetroPCS Board s fiduciary obligations to MetroPCS and/or the MetroPCS Stockholders under applicable Law, and (iii) MetroPCS timely provides to DT in accordance with Section 4.5(f) the information required under Section 4.5(f) to be delivered by MetroPCS to DT, then MetroPCS may take the following actions, provided that MetroPCS first receives from the third party making such MetroPCS Acquisition Proposal (a <u>MetroPCS Qualified Bidder</u>) an executed confidentiality agreement): (x) furnish, or cause to be furnished, information to the MetroPCS Qualified Bidder and its Representatives; and (y) engage in discussions or negotiations with the MetroPCS Qualified Bidder and its Representatives; and (y) engage in discussions or negotiation has previously been provided to DT, all non-public written information that is provided to the MetroPCS Acquisition Proposal. Unless such information has previously been provided to DT.

(d) Except as otherwise provided in Section 4.5(e), neither the MetroPCS Board nor any committee of the MetroPCS Board may (i) withdraw or withhold, amend, modify or qualify in any manner adverse to DT the MetroPCS Recommendation or make any public announcement inconsistent with the MetroPCS Recommendation, or publicly propose to do any of the foregoing, (ii) approve, adopt, endorse, recommend, or take a neutral position (other than any factually accurate public statement by MetroPCS that solely describes MetroPCS s receipt of a MetroPCS Acquisition Proposal and the operation of this Agreement with respect thereto or any stop, look and listen communication or similar communication of the type contemplated by Rule 14d-9(f) under the Exchange Act) on any MetroPCS Acquisition Proposal or any inquiry or proposal that would reasonably be expected to lead to a MetroPCS Acquisition Proposal, (iii) following the date any MetroPCS Acquisition Proposal or any material modification thereto is first made public, sent or given to the stockholders of MetroPCS, fail to issue a press release that expressly reaffirms the MetroPCS Recommendation within 10 Business Days following DT s written request to do so (which request may only be made once with respect to any such MetroPCS Acquisition Proposal and each material modification thereto), (iv) fail to include the

MetroPCS Recommendation in the Proxy Statement (any action described in clause (i), (ii), (iii) or (iv), whether taken by MetroPCS, the MetroPCS Board or any committee thereof, being referred to as a <u>MetroPCS Adverse Recommendation Change</u>), or (v) cause or permit MetroPCS to enter into any MetroPCS Contract, letter of intent, memorandum of understanding, or agreement in principle regarding or providing for any MetroPCS Acquisition Proposal (other than a confidentiality agreement as contemplated by Section 4.5(c)) or requiring MetroPCS to abandon, terminate, delay or fail to consummate the Transaction.

(e) Notwithstanding any other provision hereof, including Section 4.5(d), at any time prior to the receipt of the MetroPCS Stockholder Approval, the MetroPCS Board may effect a MetroPCS Adverse Recommendation Change in response to an Intervening Event or in response to a MetroPCS Acquisition Proposal (provided that, if the MetroPCS Adverse Recommendation Change relates to a MetroPCS Acquisition Proposal (a MetroPCS Superior Proposal Adverse Recommendation Change), the MetroPCS Board shall have determined in good faith, after consultation with its outside legal counsel and financial advisors, that such MetroPCS Acquisition Proposal constitutes a MetroPCS Superior Proposal and such MetroPCS Superior Proposal is not withdrawn), if the MetroPCS Board determines in good faith, after consultation with MetroPCS s outside legal counsel, that failure to do so would reasonably be likely to be inconsistent with the MetroPCS Board s fiduciary obligations to MetroPCS and/or the MetroPCS Stockholders under applicable Law; provided, however, that such actions may only be taken (A) if MetroPCS shall have complied with this Section 4.5 (except for any immaterial violation of this Section 4.5 that is not reasonably related to such MetroPCS Adverse Recommendation Change), (B) at a time that is after the fifth Business Day following DT s receipt of written notice from MetroPCS that the MetroPCS Board is prepared to take such action (the MetroPCS Subsequent Determination Notice ), which notice will set forth MetroPCS s reason for delivery of the MetroPCS Subsequent Determination Notice, and (1) in the case of a MetroPCS Superior Proposal Adverse Recommendation Change, specify the terms of the MetroPCS Superior Proposal that is the subject thereof, including the identity of the Person making such MetroPCS Superior Proposal and (2) in the case of an Intervening Event, include a description of the Intervening Event (it being agreed that the delivery of a MetroPCS Subsequent Determination Notice, in each case, shall not in and of itself be deemed a MetroPCS Adverse Recommendation Change), and (C) if at the end of such five Business Day period, the MetroPCS Board determines in good faith, taking into account all amendments or revisions committed to by DT, (x) after consultation with its outside legal counsel, that failure to do so would be reasonably likely to be inconsistent with the MetroPCS Board s fiduciary obligations to MetroPCS and/or the MetroPCS Stockholders under applicable Law and (y) in the case of a MetroPCS Superior Proposal Adverse Recommendation Change, after consultation with its outside legal counsel and financial advisors, that the MetroPCS Acquisition Proposal remains a MetroPCS Superior Proposal. During any such five Business Day period following a MetroPCS Subsequent Determination Notice, DT shall be entitled to deliver to MetroPCS one or more counterproposals. Any material amendment to the MetroPCS Superior Proposal that is the subject of such MetroPCS Superior Proposal Adverse Recommendation Change, including any revision to price, shall require MetroPCS to deliver to DT a new MetroPCS Subsequent Determination Notice and again comply with the requirements of this Section 4.5(e) with respect to such revised MetroPCS Superior Proposal, except that the time period after DT s receipt of a MetroPCS Subsequent Determination Notice when MetroPCS may effect a MetroPCS Superior Proposal Adverse Recommendation Change shall be three Business Days.

(f) From and after the execution of this Agreement, MetroPCS shall notify DT promptly (but in any event within 24 hours) of the receipt of any MetroPCS Acquisition Proposal, and (i) if it is in writing, deliver to DT a copy of such MetroPCS Acquisition Proposal and any related draft agreements and other written material setting forth the terms and conditions of such MetroPCS Acquisition Proposal and (ii) if oral, provide to DT a detailed summary thereof. MetroPCS shall keep DT reasonably informed on a prompt and timely basis of the status and material details of any such MetroPCS Acquisition Proposal and with respect to any material change to the terms of any such MetroPCS Acquisition Proposal within 24 hours of any such material change.

(g) Nothing in this Section 4.5 shall be deemed to prohibit MetroPCS or its Subsidiaries from complying with Rule 14e-2, Rule 14d-9 or Item 1012(a) of Regulation M-A promulgated under the Exchange Act or to

prohibit MetroPCS from making any disclosure if the MetroPCS Board determines in good faith (after consultation with its outside counsel) that failure to do so may violate MetroPCS s disclosure requirements under applicable Law, nor shall any such action be deemed to constitute a breach of MetroPCS s obligations under this Agreement; provided, however, that any disclosure regarding a MetroPCS Acquisition Proposal (other than any factually accurate public statement by MetroPCS that solely describes MetroPCS s receipt of a MetroPCS Acquisition Proposal and the operation of this Agreement with respect thereto or any stop, look and listen communication or similar communication of the type contemplated by Rule 14d-9(f) under the Exchange Act) shall be deemed to be a MetroPCS Adverse Recommendation Change unless the MetroPCS Board expressly reaffirms the MetroPCS Recommendation in such disclosure; and provided, further, that nothing in this Section 4.5(g) shall permit MetroPCS to effect a MetroPCS Adverse Recommendation Change without complying with Section 4.5(e).

(h) MetroPCS agrees that its obligations pursuant to Sections 4.3 and 4.4(b) shall not be affected by the commencement, public proposal, public disclosure or communication to MetroPCS or its stockholders of any MetroPCS Acquisition Proposal, any determination that a MetroPCS Acquisition Proposal is a MetroPCS Superior Proposal, or by the occurrence of any MetroPCS Adverse Recommendation Change.

(i) For purposes hereof, <u>MetroPCS Superior Proposal</u> shall mean any *bona fide* written MetroPCS Acquisition Proposal (with all references to 20% in the definition of MetroPCS Acquisition Proposal being treated as references to 50% for these purposes) made by a third party that the MetroPCS Board determines in good faith, after consultation with its outside legal counsel and financial advisors, would be more favorable to the MetroPCS Stockholders from a financial point of view than the Transaction, taking into account (i) any proposal by DT in writing to amend or modify the terms hereof, (ii) the identity of the Person making such MetroPCS Acquisition Proposal, and (iii) the terms, conditions, timing, likelihood of consummation and legal, financial, and regulatory aspects of such MetroPCS Acquisition Proposal.

(j) For purposes hereof, <u>MetroPCS Acquisition Propos</u>al means any inquiry, proposal, offer, plan, arrangement or other expression or indication of interest with respect to any direct or indirect acquisition or purchase, in one transaction or a series of transactions, and whether through any merger, reorganization, consolidation, tender offer, self-tender, exchange offer, stock acquisition or issuance, asset acquisition, binding share exchange, business combination, recapitalization, liquidation, dissolution, joint venture or otherwise, of (i) assets or businesses of MetroPCS and its Subsidiaries that generate or represent 20% or more of the (A) net revenues or net income of MetroPCS and its Subsidiaries, taken as a whole, immediately prior to such transaction, (B) aggregate Licensed MHz POPs of the MetroPCS FCC Licenses immediately prior to such transaction, or (C) total assets (based on fair market value) of MetroPCS and its Subsidiaries, taken as a whole, immediately prior to such transaction, or (ii) 20% or more of any class of capital stock, Equity Interests, other equity securities, economic interests or voting power of MetroPCS, any of its Subsidiaries or any resulting parent company of MetroPCS; provided, however, that the term MetroPCS Acquisition Proposal shall not include the Transaction.

#### 4.6 No Solicitation DT.

(a) Upon execution of this Agreement, DT and its Subsidiaries shall, and shall cause their respective directors and officers and use their reasonable best efforts to cause their other Representatives to, cease and terminate any and all existing activities, discussions or negotiations with any Person with respect to a TMUS Acquisition Proposal. DT shall promptly after the date hereof instruct each Person which has heretofore executed a confidentiality agreement relating to a TMUS Acquisition Proposal with or for the benefit of DT or TMUS promptly to return to DT or TMUS, as applicable, or destroy all information, documents, and materials relating to TMUS Acquisition Proposal or to TMUS or its businesses, operations or affairs heretofore furnished by DT, TMUS or any of their Representatives to such Person or any of its Representatives in accordance with the terms of any confidentiality agreement with such Person.

(b) DT agrees that neither it nor any of its Subsidiaries shall, that it shall not authorize or permit any of its and their respective directors and officers to, and that it shall not authorize, and shall use its reasonable best efforts not to permit, any of its and their other respective Representatives to, directly or indirectly, (i) initiate, solicit or knowingly encourage or knowingly take or continue any other action to facilitate the submission of any inquiry, indication of interest, proposal or offer that constitutes, or would reasonably be expected to lead to, a TMUS Acquisition Proposal, (ii) participate in any discussions or negotiations regarding any TMUS Acquisition Proposal (other than to inform a Person of the existence of this Section 4.6), (iii) furnish any non-public information or data regarding DT or any of its Subsidiaries to, or afford access to the properties, personnel, books and records of DT or TMUS to, any Person (other than MetroPCS and its Subsidiaries) in connection with or in response to or in circumstances that would reasonably be expected to lead to, any TMUS Acquisition Proposal, (iv) take any action to make the provisions of any fair price, moratorium, control share acquisition, business combination or other similar anti-takeover statute or regulation (including any transaction under, or a third party becoming an interested stockholder under, Section 203 of the DGCL), or any restrictive provision of any applicable anti-takeover provision in TMUS s Organizational Documents, inapplicable to any Person other than MetroPCS and its Subsidiaries or to any transactions constituting or contemplated by a TMUS Acquisition Proposal, or (v) resolve or agree to do any of the foregoing.

(c) From and after the execution of this Agreement, DT shall notify MetroPCS promptly (but in any event within 24 hours) of the receipt of any TMUS Acquisition Proposal, and (i) if it is in writing, deliver to MetroPCS a copy of such TMUS Acquisition Proposal and any related draft agreements and other written material setting forth the terms and conditions of such TMUS Acquisition Proposal and (ii) if oral, provide to MetroPCS a detailed summary thereof. DT shall keep MetroPCS reasonably informed on a prompt and timely basis of the status and material details of any such TMUS Acquisition Proposal and with respect to any material change to the terms of any such TMUS Acquisition Proposal within 24 hours of any such material change.

(d) For purposes hereof, <u>TMUS Acquisition Proposal</u> means any inquiry, proposal, offer, plan, arrangement or other expression or indication of interest with respect to any direct or indirect acquisition or purchase, in one transaction or a series of transactions, and whether through any merger, reorganization, consolidation, tender offer, self-tender, exchange offer, stock acquisition or issuance, asset acquisition, binding share exchange, business combination, recapitalization, liquidation, dissolution, joint venture or otherwise, of (i) assets or businesses of TMUS or any of its Subsidiaries that generate or represent 20% or more of the (A) net revenues or net income of TMUS and its Subsidiaries, taken as a whole, immediately prior to such transaction, (B) aggregate Licensed MHz POPs of the TMUS FCC Licenses immediately prior to such transaction, or (C) total assets (based on fair market value) of TMUS and its Subsidiaries, taken as a whole, immediately prior to such transaction, or (C) total assets (based on fair market value) of TMUS and its Subsidiaries or any resulting parent of TMUS (excluding DT); provided, however, that the term TMUS Acquisition Proposal shall not include the Transaction.

#### 4.7 Access; Post-Signing Deliverables.

(a) Prior to the Closing, upon reasonable notice, and except as may otherwise be prohibited by applicable Law, each of TMUS and MetroPCS shall, and shall cause its Subsidiaries to (and DT shall cause TMUS to), afford the other party s Representatives reasonable access, during normal business hours throughout the period prior to the Closing, to its properties, books, contracts and records and, during such period, each of TMUS and MetroPCS shall, and shall cause its Subsidiaries to, furnish promptly to the other party and the other party s Representatives all information concerning, in the case of TMUS s or any of its Subsidiaries , and in the case of MetroPCS, its or any of its Subsidiaries , business, properties and personnel as the other party may reasonably request, including by making available to the other party online and/or at office locations substantially all information necessary (i) to confirm compliance with respect to the Business by TMUS or MetroPCS, as applicable, with the FCC Rules as soon as reasonably practicable after the date hereof but in no event later than 60 days prior to the Closing, (ii) for integration and planning purposes, and (iii) to confirm a party s

representations and warranties and compliance with the covenants in this Agreement; <u>provided</u>, that no investigation pursuant to this Section 4.7 shall affect or be deemed to modify any representation or warranty made by any party hereunder; <u>provided</u>, <u>further</u>, that the foregoing shall not require TMUS, MetroPCS or any of their respective Subsidiaries to permit any inspection or disclose any information that in the reasonable judgment of TMUS or MetroPCS, as applicable, would waive attorney-client or attorney work product privileges, or result in the disclosure of any Trade Secrets of third parties, violate any applicable Laws or violate any of its obligations with respect to confidentiality. All requests for information made pursuant to this Section 4.7 shall be directed to an executive officer of TMUS or MetroPCS, as applicable, or such Person as may be designated by any such executive officer, as the case may be. Notwithstanding the foregoing, none of TMUS, MetroPCS or any of their Subsidiaries shall be obligated to afford any other party or its Representatives any access to any properties, books, contracts, commitments, personnel or records relating to, or in respect of, any forward product plans, product specific cost information, pricing information, customer specific information, merchandising information provided or made available pursuant to this Section 4.7 shall be subject to the Confidentiality Agreement shall remain in full force and effect in accordance with its terms.

(b) Following the Closing, upon reasonable notice, and except as may otherwise be prohibited by applicable Law, MetroPCS shall cause TMUS and its Subsidiaries to afford DT s Representatives such reasonable access, during normal business hours during the period from the Closing until one year following the expiration of the statute of limitations applicable to DT s financial and tax reporting, to its books and records relating to the period prior to the Closing and, during such period, MetroPCS shall, and shall cause TMUS and its Subsidiaries to, furnish promptly to DT and DT s Representatives such information concerning TMUS s or any of its Subsidiaries historical financial performance, in each case, as DT may reasonably request for its financial reporting purposes; provided, that the foregoing shall not require MetroPCS, TMUS or any of its Subsidiaries to permit any inspection or disclose any information that in the reasonable judgment of MetroPCS would waive attorney-client privilege or attorney work product, or result in the disclosure of any Trade Secrets of third parties, violate any applicable Laws or violate any of its obligations with respect to confidentiality if, in the case of Trade Secrets or confidentiality obligations, MetroPCS, TMUS or its Subsidiaries, as applicable, shall have used reasonable efforts to obtain the consent of the applicable third party to such inspection or disclosure. DT shall reimburse TMUS and its Subsidiaries for all reasonable, out-of-pocket costs and expenses incurred by them in providing such access. All requests for information made pursuant to this Section 4.7(b) shall be directed to an executive officer of MetroPCS or such Person as may be designated by any such executive officer, as the case may be. Notwithstanding the foregoing, none of MetroPCS, TMUS or any of its Subsidiaries shall be obligated to afford DT or its Representatives any access to any properties, books, contracts, commitments, personnel or records relating to, or in respect of, any forward product plans, product specific cost information, pricing information, customer specific information, merchandising information or other similar competitively sensitive information. All information provided or made available pursuant to this Section 4.7(b) shall be subject to Section 4.22. No action pursuant to this Section 4.7(b) shall impair the rights of any party under Section 2.4.

(c) DT shall furnish to MetroPCS, to the extent and in the form delivered to DT in the ordinary course of business, promptly following such delivery, monthly and/or quarterly unaudited consolidated statements of operations and comprehensive income, changes in stockholders equity and cash flows of TMUS and its Subsidiaries. MetroPCS shall prepare and furnish to DT, to the extent and in the form prepared in the ordinary course of business, promptly after they become available, monthly and/or quarterly unaudited consolidated statements of operations and comprehensive income, stockholders equity and cash flows of MetroPCS and its Subsidiaries.

4.8 <u>Publicity</u>. The initial press release announcing the transaction shall be in a form agreed by DT and MetroPCS. Each of DT and MetroPCS agrees that it shall consult with the other before it or any of its Subsidiaries issues any press release or similar public statement regarding the Transaction, including by

providing a draft of such press release or public statement and providing the other with a reasonable period of time to review such draft, and neither DT nor MetroPCS nor any of their respective Subsidiaries shall issue such press release or similar public statement, except (a) with the prior written consent of the other party, not to be unreasonably withheld, conditioned or delayed, (b) to the extent required by Law (including any listing requirement), (c) in accordance with a mutually agreed communications plan, or (d) as otherwise expressly permitted hereunder.

4.9 Expenses. Except as otherwise specifically provided in this Agreement, DT, on the one hand, and MetroPCS, on the other hand, shall bear their respective expenses, costs and fees (including attorneys , auditors and financing fees, if any) in connection with the Transaction, including the preparation, execution and delivery of this Agreement and compliance herewith, whether or not the Transaction is effected; provided, that (a) DT shall be responsible for (i) any attorneys , tax advisors and financial advisors fees of TMUS and its Subsidiaries incurred in connection with the execution and delivery of this Agreement and the consummation of the Transaction and all such expenses, costs and fees (including attorneys and auditors fees, if any) payable by TMUS or any of its Subsidiaries shall be paid and satisfied in full prior to the Closing by DT or any of its Subsidiaries (other than TMUS and its Subsidiaries) (ii) any costs, charges and expenses (including costs and expenses of DT s counsel) incurred by the parties or their respective Subsidiaries in connection with issuing the DT Notes, (iii) any fees, costs, charges and expenses (including fees of credit rating agencies) incurred in connection with obtaining a credit rating pre-determination for MetroPCS and the MetroPCS Existing Notes giving effect to the Transaction, (iv) any HSR Act filing fees, and (v) any expenses, costs and fees payable to Ernst & Young LLP for services provided in connection with the financial statements (including pro forma financial statements) included in the Proxy Statement; provided, that, at the Closing, MetroPCS shall reimburse DT in full for the amount of such fees, costs, charges and expenses (including costs and expenses of counsel) incurred by MetroPCS in connection with the MetroPCS Finance Transactions, and (y) costs and expenses of MetroPCS s counsel incurred by MetroPCS or its Subsidiaries in connection with the DT Notes.

4.10 <u>Resignations</u>. DT shall use its reasonable best efforts to cause each director of TMUS and any officer or director of any of TMUS s Subsidiaries to resign in such capacity other than individuals who will continue as officers of TMUS or any of its Subsidiaries after the Closing, such resignations to be effective as of the Closing. MetroPCS shall use its reasonable best efforts to cause each director of MetroPCS and any officer or director of any of MetroPCS s Subsidiaries to resign in such capacity other than individuals who will continue as directors or officers of MetroPCS or any of its Subsidiaries after the Closing, such resignations to be effective as of the Closing.

#### 4.11 Filings; Other Actions; Notification.

(a) Subject to Sections 4.11(b) and 4.11(c), DT and MetroPCS shall cooperate with each other and use, and shall cause their respective Subsidiaries to use, their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, necessary, proper or advisable on its part under this Agreement and applicable Laws to consummate the Transaction as promptly as reasonably practicable, including (i) preparing and filing as promptly as reasonably practicable all documentation to effect all necessary notices, reports and other filings (including by filing promptly after the date hereof on a date agreed by the parties the notification and required form under the HSR Act and all applications and necessary and appropriate filings and any necessary and appropriate amendments thereto required to be filed with the FCC (including any petition for declaratory ruling regarding foreign ownership and any applications and filings pertaining to the transfer of the TMUS FCC Licenses pursuant to the Transaction that could be considered *pro forma* by the FCC), any PUCs or similar state or foreign regulatory bodies; <u>provided</u>, <u>however</u>, that the failure to file within such applicable periods will not constitute a breach of this Agreement); (ii) subject to the foregoing, obtaining as promptly as reasonably practicable all Governmental Consents; (iii) furnishing all information required or reasonably

requested for any application or other filing to be made pursuant to any applicable Laws in connection with the Transaction; (iv) keeping the other parties informed in all material respects of any material communication received by such party from, or given by such party to, any Governmental Entity and of any material communication received or given in connection with any proceeding by a private party, in each case relating to the Transaction; (v) (A) negotiating, proposing and/or agreeing to the sale, divestiture, license, disposition or hold separate of any asset and other actions, restrictions, limitations or conditions required to obtain any consents, registrations, approvals, permits or authorizations in connection with the Transaction, (B) conducting or agreeing to conduct its business post-Closing in any manner as necessary to obtain any consents, registrations, approvals, permits or authorizations in connection with the Transaction or (C) agreeing to any order, action or regulatory condition of any regulatory body, whether in an approval proceeding or another regulatory proceeding; and (vi) defending against the entry of any decree, order, or judgment that would restrain, prevent or delay the Closing, including defending any lawsuits or other legal proceedings, whether judicial or administrative, challenging this Agreement or the consummation of the Transaction. Without limiting the foregoing, prior to the Closing, DT and MetroPCS shall not, and DT shall cause Global and its Subsidiaries not to, and MetroPCS shall cause its Subsidiaries not to, knowingly take any action, or knowingly fail to take any action, that would reasonably be likely to materially delay or interfere with the parties ability to consummate the Transaction.

(b) Neither MetroPCS nor its Subsidiaries shall be permitted to agree to any actions, restrictions or conditions with respect to obtaining any consents, registrations, approvals, permits or authorizations in connection with the Transaction without the prior written consent of DT. Subject to applicable Laws relating to the exchange of information and to the extent reasonably practicable, DT and MetroPCS shall have the right to review in advance and each will consult the other on, all of the information relating to DT or MetroPCS, as the case may be, and any of their respective Subsidiaries that appears in any filing made with, or written materials submitted to, any third party and/or any Governmental Entity in connection with the Transaction. To the extent permitted by Law, each such party shall provide the other with copies of all material correspondence between it (or its advisors) and any Governmental Entity relating to the Transaction and, to the extent reasonably practicable and permitted by such Governmental Entity, all telephone calls and meetings with a Governmental Entity regarding the Transaction shall include Representatives of DT and MetroPCS. In exercising the foregoing rights, each of DT and MetroPCS shall act and DT shall cause TMUS to act reasonably and as promptly as practicable.

(c) Nothing in this Section 4.11 or otherwise in this Agreement shall require MetroPCS or DT to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets (including FCC Licenses), businesses or product lines of MetroPCS, DT, TMUS, any of their respective Subsidiaries, or any combination thereof, that would have a material adverse effect (for the avoidance of doubt, without giving effect to the carveouts (A) through (G) contained in the definition of TMUS Material Adverse Effect or MetroPCS Material Adverse Effect ) on the business, assets, liabilities, financial condition or results of operations of MetroPCS, TMUS and their respective Subsidiaries, taken as a whole (the effect of any such action, restriction, limitation or condition meeting these requirements, a <u>Regulatory Material Adverse Condition</u> ).

4.12 <u>Financial Working Group</u>. Promptly after the date hereof, DT and MetroPCS shall establish, and DT shall cause TMUS to designate the appropriate employees to participate with employees of MetroPCS in, a joint working group to prepare for the integration of TMUS and its Subsidiaries after the Closing into MetroPCS s internal control structure and procedures for financial reporting compliance with the requirements of Rule 404 of the Sarbanes-Oxley Act and MetroPCS s financial reporting structure.

4.13 Financing.

(a) DT and MetroPCS shall consult with each other and mutually cooperate in good faith to effect the MetroPCS Finance Transactions.

(b) Prior to or on the Closing Date, unless an alternative structure shall have been agreed by the Parties in good faith, (i) DT shall cause TMUS and its Subsidiaries to assign, and DT or one of its Subsidiaries designated by DT shall assume and procure the release of TMUS and its Subsidiaries from all obligations under, all Intercompany Indebtedness owed by TMUS and its Subsidiaries to DT or one of its Subsidiaries (other than TMUS and its Subsidiaries) (which has an aggregate principal amount of approximately \$14.4 billion as of the date hereof), including any accrued interest thereon and all other amounts payable by, and other obligations (including contingent obligations) of, TMUS and its Subsidiaries thereunder and all related documentation, in exchange for an obligation of TMUS to disburse to DT or one of its Subsidiaries designated by DT the principal amount of and all accrued interest on the Intercompany Indebtedness so assumed and all other amounts payable by, and other obligations (including contingent obligations) of, TMUS and its Subsidiaries thereunder and all related documentation; (ii) DT shall cause TMUS to issue and deliver to DT or one of its Subsidiaries designated by DT, and DT shall purchase or cause such designated Subsidiaries to purchase, notes in the aggregate principal amount of \$15,000,000 (together with any Additional DT Notes issued pursuant to Section 4.13(c) below, the <u>DT Notes</u>), which shall have, and be issued pursuant to an indenture containing, the terms set forth on Exhibit F and Exhibit G, respectively, and otherwise reasonably acceptable to DT and MetroPCS, in exchange for an obligation of DT or one of its Subsidiaries to disburse to TMUS an amount equal to the aggregate principal amount of such DT Notes; (iii) DT or one of its Subsidiaries designated by DT shall have an obligation to pay TMUS an amount equal to the excess, determined on an arm s length basis (as reasonably determined by TMUS and DT taking into account the respective interest rates, maturity profile and other relevant factors and supported by an investment bank fair market value analysis), of the fair market value of (A) the portion of the DT Notes having an aggregate principal amount equal to the amount of the Intercompany Indebtedness assumed by DT or one of its Subsidiaries over (B) the Intercompany Indebtedness assumed by DT or one of its Subsidiaries; and (iv) TMUS and DT or one of its Subsidiaries designated by DT shall set off the payment obligations described in clauses (i), (ii) and (iii) against each other and TMUS shall distribute as a dividend to Holding, prior to the Closing Date, the net receivable resulting from such setoff. As of 12:01 a.m., prevailing Eastern Time, on the Closing Date, there shall be no Intercompany Indebtedness outstanding owed by TMUS and its Subsidiaries to DT or one of its Subsidiaries (other than TMUS and its Subsidiaries), the DT Notes shall be issued and outstanding, and DT shall have no further obligation to disburse to TMUS all or any portion of the purchase price of the DT Notes.

(c) DT shall cause TMUS to issue and deliver to DT or one of its Subsidiaries designated by DT, and DT shall purchase or cause such designated Subsidiaries to purchase, additional notes (<u>Additional DT Notes</u>) which shall have, and be issued pursuant to an indenture containing, the terms, set forth on Exhibit G in accordance with and subject to the terms and conditions set forth on Exhibit I. In connection with the foregoing, TMUS shall pay to DT, in U.S. dollars in immediately available funds, such applicable fees as are set forth on Exhibit I (and, except as set forth on Exhibit I any such fees shall not be refundable for any reason whatsoever and shall be in addition to any other amounts payable to DT pursuant to this Agreement or otherwise in connection with the Transaction).

(d) DT and TMUS shall, and shall cause each issuer of DT Notes or any Affiliate of DT that holds any DT Notes to, enter into an agreement at or prior to the Closing reflecting the terms set forth on Exhibit J.

(e) On or prior to the Closing Date, DT shall cause TMUS to enter into and cause to become effective the TMUS Working Capital Facility with DT or one of its Subsidiaries designated by DT.

(f) From and after the date hereof until the Closing Date, and without limiting any other rights of MetroPCS or its Subsidiaries hereunder, MetroPCS shall be permitted to offer and sell the Permitted MetroPCS Notes on the terms described in the definition thereof.

#### 4.14 Regulatory Compliance.

(a) TMUS and MetroPCS shall, and shall cause their respective Subsidiaries to, use commercially reasonable efforts to (i) cure no later than the Closing any material violations and material defaults by any of them under any applicable rules and regulations of the FCC (<u>FCC Rules</u>) and the FAA Rules, (ii) comply in all material respects with the terms of their respective FCC Licenses and the FAA Rules, and (iii) file or cause to be filed with the FCC and the FAA all material reports and other material filings required to be filed under applicable FCC Rules and FAA Rules.

(b) During the period from the date hereof through the earlier of the Closing and the date this Agreement is terminated pursuant to the terms hereof, each of TMUS and MetroPCS shall, and shall cause its respective Subsidiaries to, use their reasonable best efforts to (i) take such actions as are reasonably necessary to maintain and preserve the Material TMUS Licenses or Material MetroPCS Licenses, respectively, except as expressly permitted under Section 4.1 or 4.2, respectively, and (ii) refrain from taking any action that would reasonably be expected to cause the FCC or any other Governmental Entity with jurisdiction over such party or any of its Subsidiaries to institute proceedings for the suspension, revocation or adverse modification of any TMUS Communications Licenses or MetroPCS Communications Licenses, respectively.

(c) During the period from the date hereof through the earlier of the Closing and the date this Agreement is terminated pursuant to the terms hereof, each of TMUS and MetroPCS shall, and shall cause its respective Subsidiaries to, take such steps as are necessary to renew any expiring TMUS Communications Licenses or MetroPCS Communications Licenses, respectively, including preparing and filing with the appropriate Governmental Entities all necessary applications in connection therewith as soon as reasonably practicable after the commencement of the period during which such applications may be made.

(d) During the period from the date hereof through the earlier of the Closing and the date this Agreement is terminated pursuant to the terms hereof, each of TMUS and MetroPCS shall, and shall cause its respective Subsidiaries to, continue to use reasonable best efforts to complete the relocation of incumbent licensees and the satisfaction of any cost-sharing obligations as required by Subpart L, Part 27 of Title 47 of the Code of Federal Regulations and applicable FCC orders and as such requirements apply to any of TMUS FCC Licenses or MetroPCS FCC Licenses, respectively.

#### 4.15 Further Action.

(a) Subject to the terms and conditions hereof, each of the parties shall use their reasonable best efforts to (i) take or cause to be taken all appropriate action, to do or cause to be done all things necessary, proper or advisable under applicable Law and to execute and deliver such documents and other papers, as may be required to carry out the provisions hereof and consummate the Transaction and (ii) refrain from taking or causing to be taken any action that would be reasonably expected to prevent, materially delay or materially impair the consummation of the Transaction.

(b) From time to time after the Closing, without additional consideration, each party will (or, if appropriate, will cause its Subsidiaries to) execute and deliver such further instruments and take such other action as may be necessary or reasonably requested by another party to make effective the Transaction.

4.16 <u>Intercompany Arrangements</u>. Following the date hereof, DT and MetroPCS shall agree, acting reasonably and in good faith, as to which Intercompany Contracts shall continue in effect following the Closing and which shall be terminated on or prior to the Closing Date, without any post-Closing Liability of TMUS or any of its Subsidiaries; provided, that nothing in this Section 4.16 shall require MetroPCS to agree to any action by TMUS or its Subsidiaries that is proscribed in Section 4.1(1).

4.17 <u>Customer Communications</u>. Subject to applicable Laws, DT and MetroPCS shall cooperate in developing language for a program of communications or notices relating to the Transaction to be sent to customers of MetroPCS, TMUS and their respective Subsidiaries on or after the date hereof and prior to the Closing. Each party shall not, and shall cause its Subsidiaries not to, send any communications or notices primarily relating to the Transaction to customers of TMUS, MetroPCS or their respective Subsidiaries on or after the date hereof and prior to the Closing other than (a) in accordance with any program of communications or notices adopted in writing by DT and MetroPCS, (b) with the prior written approval of the other party, not to be unreasonably withheld, conditioned or delayed, (c) as required by Law, or (d) as expressly permitted hereunder.

#### 4.18 Employee Matters.

(a) For at least the one-year period immediately following the Closing, MetroPCS shall, and DT shall cause MetroPCS to, continue to provide to those employees of MetroPCS and its Subsidiaries who are employed by MetroPCS or its Subsidiaries both immediately before the Closing and immediately after the Closing (<u>MetroPCS Employees</u>) compensation (other than equity compensation), severance pay and benefits, and other employee benefits that are substantially comparable, in the aggregate, to the compensation (other than equity compensation), severance pay and benefits, and other employee benefits provided by MetroPCS as of the date hereof. MetroPCS shall, and DT shall cause MetroPCS to, use its reasonable best efforts to (x) make a decision regarding whether to continue the employment of each MetroPCS Employee in a position and on terms of employment comparable to such MetroPCS Employee s position and terms of employment as of the Closing and (y) communicate such decision to each MetroPCS Employee and carry out such decision during the one year period immediately following the Closing. Notwithstanding the foregoing, nothing contained herein shall (i) be treated as the creation or an amendment of any particular compensation and benefit plan provided by DT, MetroPCS or any of their respective Subsidiaries, (ii) give any third party any right to enforce the provisions of this Section 4.18, or (iii) obligate DT, MetroPCS or any of their respective Subsidiaries to (A) maintain any particular compensation and benefit plan provided by MetroPCS or its Subsidiaries, or (B) retain the employment of any particular employee.

(b) Following the Closing, to the extent that any of the MetroPCS Benefit Plans are replaced, each MetroPCS Employee shall receive service credit under any new benefit plans to the extent credited under the MetroPCS Benefit Plans immediately prior to the Closing for all purposes of determining eligibility to participate, vesting and level of benefits for purposes of vacation, severance and paid time off (but not for benefit accrual purposes, for the purpose of qualifying a subsidized early retirement benefit, or as would result in a duplication of benefits) for the same purposes under the replacement employee benefit plans in which such MetroPCS Employee participates following the Closing. To the extent a MetroPCS Employee participates in a new welfare plan or arrangement (a Replacement Welfare Plan ) following the Closing, MetroPCS will, and DT will cause MetroPCS to, cause all (i) pre-existing condition limitations which otherwise would be applicable to such MetroPCS Employee and his or her covered dependents to be waived to the extent satisfied under a MetroPCS Benefit Plan in which such MetroPCS Employee participated and that is comparable to such Replacement Welfare Plan, in each case immediately prior to the Closing or, if later, immediately prior to such MetroPCS Employee s commencement of participation in such Replacement Welfare Plan, and (ii) participation waiting periods under each Replacement Welfare Plan that would otherwise be applicable to such MetroPCS Employee to be waived to the same extent waived or satisfied under the MetroPCS Benefit Plan in which such MetroPCS Employee participated and that is comparable to such Replacement Welfare Plan immediately prior to the Closing or, if later, immediately prior to such MetroPCS Employee s commencement of participation in such Replacement Welfare Plan. In addition, MetroPCS and its Subsidiaries will, and DT will cause MetroPCS to, honor or cause to be honored any expenditures incurred by MetroPCS Employees and their covered dependents in satisfying the deductible, co-payment and out-of-pocket maximums under the MetroPCS Benefit Plans during the portion of the applicable plan year that includes the Closing in satisfying any deductibles, co-payments or out-of-pocket maximums under any comparable Replacement Welfare Plans in which they are eligible to participate after the Closing for the portion of the applicable plan year that includes the Closing.

(c) DT, TMUS and MetroPCS shall cooperate in developing any written or material oral communications to the directors, officers or employees of MetroPCS, TMUS, or any of their respective Subsidiaries pertaining to the effect of the Transaction on compensation or benefit matters. TMUS and DT shall provide MetroPCS with any proposed employee communications relating to the effect of the Transaction on any employment, compensation or benefit matters and provide MetroPCS a reasonable opportunity to comment on the same. MetroPCS shall provide DT and TMUS with any proposed employee communications relating to the effect of the Transaction on any employment, compensation or benefit matters and provide DT and TMUS a reasonable opportunity to comment on the same. Without limiting the foregoing, for purposes of this paragraph, provision of a proposed employee communication at least three Business Days prior to its use shall be deemed to be a reasonable opportunity to comment.

#### 4.19 MetroPCS Directors and Officers; Name.

(a) MetroPCS shall, and DT shall cooperate to, take all actions necessary to cause the MetroPCS Board, effective as of the Closing, to consist of eleven individuals that MetroPCS and DT shall cooperate in good faith to identify prior to the Closing, consistent with the principles set forth on Exhibit K and Section 3.1 of the Stockholders Agreement.

(b) MetroPCS shall, and DT shall cooperate to, take all actions necessary to cause the executive officers of MetroPCS, effective as of the Closing, to be the individuals set forth on Exhibit K as MetroPCS officers (or their replacements as determined pursuant to the following sentences) and, to the extent not specified on Exhibit K, the individuals that MetroPCS and DT shall cooperate in good faith to identify prior to the Closing, and DT and TMUS shall, and MetroPCS shall cooperate to, take all actions necessary to cause the executive officers of TMUS, effective as of the Closing to be the individuals set forth on Exhibit K as TMUS officers (or their replacements as determined pursuant to the following sentences) and, to the extent not specified on Exhibit K, the individuals that MetroPCS and DT shall cooperate in good faith to identify prior to the Closing. Prior to the Closing, in the event that one or more of the executive officers set forth on Exhibit K are unwilling or unable to serve in that capacity, DT and MetroPCS will discuss in good faith a replacement for each such individual. If DT and MetroPCS are unable to reach agreement on any replacement for an individual specified on Exhibit K or on any officer not specified on Exhibit K, and not otherwise identified prior to Closing pursuant to this Section 4.19(b) to be an executive officer of MetroPCS effective as of the Closing, shall be deemed to not be an executive officer of MetroPCS following the Closing.

(c) On the Closing Date, MetroPCS shall, as part of the amendment of its certificate of incorporation, change its name to T-Mobile US, Inc. (or such other name as selected by DT prior to the Closing Date) and reflect such name in the New MetroPCS Certificate. On or promptly following the Closing Date, MetroPCS shall request the assignment of a new ticker symbol to be determined by DT.

(d) At or prior to the Closing, the MetroPCS Board shall pass such resolutions and take such other actions as may be necessary to effectuate the provisions of this Section 4.19.

4.20 <u>Transition Arrangements</u>. Promptly following the date hereof and subject to applicable Laws, MetroPCS and DT shall cooperate to develop a procedure such that as of no later than the Closing Date, (x) TMUS, MetroPCS and their respective Subsidiaries will not require in order to conduct the Business, or have access to, any IT Assets, networks or electronic data of DT and its Subsidiaries (other than TMUS and its Subsidiaries) and (y) DT and its Subsidiaries will not have access to any IT Assets, networks or electronic data of MetroPCS and its Subsidiaries (including TMUS and its Subsidiaries), in each case other than as provided in Intercompany Contracts with obligations remaining on the Closing Date.

4.21 Intellectual Property. DT hereby covenants not to assert against MetroPCS or its Subsidiaries any Intellectual Property owned by DT and used by TMUS or any of its Subsidiaries immediately prior to the Closing Date, provided that (i) MetroPCS does not use such Intellectual Property for any activities, products and services or for any other purposes other than those for which TMUS used the respective Intellectual Property immediately prior to the Closing Date, (ii) this covenant is limited to the territory of the United States, Puerto Rico and the territories and protectorates of the United States, and (iii) this covenant shall not be assignable and not extend to MetroPCS s legal successors and assignees. The covenant of the preceding sentence shall not apply to any Intellectual Property with respect to which TMUS and DT or, as the case may be, MetroPCS and DT have made a written agreement, including the Trademark License. Except as set forth in the preceding sentence, no license or other right is granted by this Section 4.21.

4.22 <u>Confidentiality</u>. For a period of three years following the later of the Closing and the date of disclosure, and five years following the Closing with respect to customer information, DT and each of its Subsidiaries shall treat as confidential and shall safeguard any and all confidential or proprietary information, knowledge and data about TMUS, its Subsidiaries and the Business by using the same degree of care, but no less than a reasonable standard of care, to prevent the unauthorized use, dissemination or disclosure of such information, knowledge and data as DT or its Subsidiaries used with respect thereto prior to the execution of this Agreement.

#### 4.23 Indemnification; Release.

(a) MetroPCS, DT and TMUS agree that all rights to exculpation, indemnification and advancement of expenses for acts or omissions occurring at or prior to the Closing, whether asserted or claimed prior to, at or after the Closing, now existing in favor of the current or former directors, officers or employees, as the case may be, of MetroPCS, TMUS or their respective Subsidiaries as provided in their respective Organizational Documents or in any agreement as in effect on the date hereof and which has prior to the date hereof been made available to the other parties hereto shall survive the Closing and shall continue in full force and effect to the extent provided in the following sentence. DT shall cause MetroPCS and TMUS to maintain in effect any and all exculpation, indemnification and advancement of expenses provisions of MetroPCS s, TMUS s and any of their respective Subsidiaries Organizational Documents or in any indemnification agreements of MetroPCS, TMUS or their respective Subsidiaries with any of their respective current or former directors, officers or employees, in each case in effect as of the date hereof and which has been provided to DT prior to the date hereof, for acts or omissions occurring on or prior to the Closing. DT, for and on behalf of itself and on behalf of its Affiliates, hereby acquits, releases and discharges each of the current or former directors and officers of TMUS and each of its Subsidiaries on or prior to the Closing.

(b) For six years after the Closing, MetroPCS (or a Subsidiary thereof) shall, and DT shall cause MetroPCS (or a Subsidiary thereof) to, maintain in effect for the benefit of the current and former directors, officers, fiduciaries, agents and employees of MetroPCS and its Subsidiaries an insurance and indemnification policy with an insurer with the same or better credit rating as the current carrier for MetroPCS that provides coverage for acts or omissions occurring at or prior to the Closing (the <u>D&O Tail Policy</u>) covering each such Person covered by the officers and directors liability insurance policies of MetroPCS (the <u>Insured Parties</u>) on terms with respect to coverage and advancement of expenses, and in amounts, no less favorable to the Insured Parties than those of MetroPCS s directors and officers insurance policy in effect on the date of this Agreement; <u>provided</u>, <u>however</u>, that MetroPCS shall not be required to pay an annual premium for the D&O Tail Policy in excess of 250% of the annual premium currently paid by MetroPCS for such coverage; and <u>provided</u>, <u>further</u>, that if any annual premium for such insurance coverage exceeds 250% of such annual premium, MetroPCS shall obtain as much coverage as practicable for a cost not exceeding such amount. MetroPCS s and DT s obligations under this Section 4.23(b) may be satisfied by MetroPCS, with the approval, not to be unreasonably withheld, conditioned or delayed, of DT, by purchasing a tail policy from an insurer with substantially the same or better credit rating as the current carrier for MetroPCS s existing directors and officers insurance policy, which (i) has an effective

term of six years from the Closing, (ii) covers each Insured Party for actions and omissions occurring at or prior to the Closing (including with respect to acts or omissions by directors or officers of MetroPCS or its Subsidiaries in their capacities as such arising in connection with the entry into, performance under, or the transactions contemplated by, this Agreement), and (iii) contains terms with respect to coverage and advancement of expenses, and in amounts, that are no less favorable to the Insured Parties than those of MetroPCS s directors and officers insurance policy in effect on the date of this Agreement. If such tail policy has been obtained by MetroPCS prior to the Closing, MetroPCS shall cause such policy to be maintained in full force and effect, for its full term, and cause all obligations thereunder to be honored by MetroPCS.

(c) In the event MetroPCS or TMUS or any of their respective successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity in such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any Person, then, and in either such case, proper provision shall be made so that the successors and assigns of MetroPCS or TMUS, as applicable, shall assume all of the obligations set forth in this Section 4.23.

(d) Effective as of the Closing, DT, for and on behalf of itself and on behalf of its Subsidiaries, hereby acquits, releases and discharges each of TMUS and its Subsidiaries from any and all Liabilities as of the Closing to DT or any of its Subsidiaries (other than TMUS and its Subsidiaries), except in respect of obligations under Intercompany Contracts or under this Agreement that are not terminated as of the Closing Date. DT shall cause its other Affiliates to use their reasonable best efforts to take, or cause to be taken, all appropriate action and to execute and deliver such documents and other papers, as may be required to effect the release set forth in this Section 4.23(d).

(e) Notwithstanding anything to the contrary contained in this Agreement, the current and former directors, officers and employees of MetroPCS, TMUS and their respective Subsidiaries are intended to be, and shall be, third party beneficiaries of this Section 4.23.

4.24 <u>MetroPCS Common Stock</u>. During the period from the date hereof to the Closing, DT shall not, and shall cause each of its Affiliates not to, directly or indirectly, alone or in concert with any other Person acquire, offer to acquire or agree to acquire Beneficial Ownership of any shares of MetroPCS Common Stock.

4.25 <u>Sale of Towers</u>. From and after the date hereof until the Closing Date, and without limiting any other rights of DT or TMUS hereunder, DT shall be permitted to cause TMUS to sell, lease, transfer, or otherwise dispose of the Tower Assets, including by selling or transferring an entity holding such Tower Assets (and no other material assets) (a <u>Tower Holdco</u>) on terms that TMUS has made available to MetroPCS prior to the date hereof (or other terms which, taken as a whole, have an equivalent or more favorable economic cost to TMUS and its Subsidiaries with respect the use of the Tower Assets by TMUS). On or prior to the Closing Date, any and all actual proceeds (net of fees and expenses other than Taxes) from such sale, lease, transfer or disposition (or, to the extent such proceeds are not received on or prior to the Closing Date, a note, contingent value right or other obligation or security for the amount of any reasonably expected proceeds), may be distributed to DT or any of its Subsidiaries. From and after the Closing Date, if any or all of the Tower Assets have not been so sold, leased, transferred or otherwise disposed of, MetroPCS shall, and cause its Subsidiaries (including TMUS and Subsidiaries) to, use their reasonable best efforts to sell, lease, transfer, or otherwise dispose of such Tower Assets on terms acceptable to DT in its sole discretion, and shall transfer any proceeds thereof to DT promptly after the receipt of such proceeds (net of fees and expenses other than Taxes), including pursuant to any note, contingent value right or other obligation or prior to the Closing Date.

4.26 <u>Notification of Certain Matters</u>. Each of TMUS and MetroPCS shall promptly advise the other party of any Circumstance (a) having or reasonably expected to have a TMUS Material Adverse Effect or MetroPCS Material Adverse Effect, respectively, or (b) that it believes would or would be reasonably likely to cause or constitute a breach of any of its representations, warranties or covenants contained in this Agreement such that it

would cause a failure of the applicable condition related to such breach in Article V if the Closing were otherwise to occur on the date the Circumstance occurs; <u>provided</u>, <u>however</u>, that no such notification shall affect the representations, warranties, covenants or agreements of the parties (or remedies with respect thereto) or the conditions to the obligations of the parties under this Agreement; <u>provided</u>, <u>further</u>, that a failure to comply with this Section 4.26 shall not constitute the failure of any condition set forth in Article V to be satisfied unless the underlying TMUS Material Adverse Effect or MetroPCS Material Adverse Effect, respectively, or breach would independently result in the failure of a condition set forth in Article V to be satisfied.

4.27 <u>Litigation</u>. Notwithstanding anything to the contrary set forth herein, each of DT and MetroPCS shall notify the other party promptly if and after it receives notice of any action or proceeding instituted or threatened in writing against it or any of its stockholders, directors, officers or Affiliates before any court or other Governmental Entity relating to, or seeking damages or other relief or discovery in connection with, this Agreement or the Transaction. Until the earlier of the termination of this Agreement in accordance with its terms or the Closing, to the extent legally permissible, each of DT and MetroPCS shall give the other party the opportunity to reasonably participate in the defense or settlement of any action or proceeding relating to this Agreement or the Transaction, and shall not settle any such action or proceeding without the other party s written consent, which will not be unreasonably withheld, conditioned or delayed.

4.28 <u>Anti-Takeover Statutes</u>. If any other potentially applicable anti-takeover or similar Law or provision in MetroPCS s or TMUS s governing documents is or becomes applicable to this Agreement or the Transaction, MetroPCS or DT, as applicable, shall, and shall cause its and its Subsidiaries directors, officers and Affiliates to, grant such approvals and take such other actions as may be required so that the Transaction may be consummated as promptly as practicable on the terms and subject to the satisfaction of conditions set forth in this Agreement.

4.29 <u>Control of Operations</u>. Without in any way limiting and subject to the parties rights and obligations under this Agreement, the parties understand and agree that nothing contained in this Agreement shall give any party, directly or indirectly, the right to exercise de facto or de jure control over the operations, licenses, spectrum or other assets of any other party prior to the Closing.

4.30 Listing of TMUS Stock Consideration. MetroPCS shall use its reasonable best efforts to cause the TMUS Stock Consideration to be approved for listing on the NYSE, subject to official notice of issuance, prior to the Closing.

#### ARTICLE V

#### CONDITIONS

5.1 <u>Conditions to Each Party</u> s <u>Obligation to Effect the Transaction</u>. The respective obligations of each party to effect the Transaction are subject to the satisfaction or waiver (if permissible under applicable Laws) at or prior to the Closing of each of the following conditions:

(a) MetroPCS Stockholder Approval. The MetroPCS Stockholder Approval shall have been obtained.

(b) <u>Required Regulatory Consents</u>. (i) The waiting period (and any extensions thereof) applicable to the consummation of the Transaction under the HSR Act shall have expired or been earlier terminated, and (ii) all Governmental Consents required to be obtained from the FCC in connection with the consummation of the Transaction shall have been granted by the FCC by Final Order, in each of cases (i) and (ii) without requiring DT or MetroPCS to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets (including FCC Licenses), business or product lines of MetroPCS, DT, TMUS, or any of their respective Subsidiaries, or any combination thereof, that would cause a Regulatory Material Adverse Condition; <u>provided</u>, <u>however</u>, that the Governmental Consents to be granted from the FCC in connection with the consummation of the Transaction shall not be required to have been granted by the FCC by

Final Order in the event that (A) the Governmental Consents required to be granted by the FCC in connection with the consummation of the Transaction shall have been granted but not pursuant to a Final Order, (B) neither party shall have appealed or sought reconsideration of the authorizations granted by the FCC in connection with the consummation of the Transaction, (C) all other conditions pursuant to this Article V shall have been met or waived (except for those conditions that, by their nature, cannot be satisfied until the Closing Date but would be capable of satisfaction if the Closing Date were the same day that the Governmental Consents required to be granted by the FCC were granted), and (D) either party would have the right to terminate this Agreement pursuant to Section 6.1(c) before such Governmental Consents granted by the FCC would become a Final Order.

(c) <u>Other Governmental Consents</u>. (i) All Governmental Consents (other than those described in Section 5.1(b) and clauses (ii) and (iii) of this Section 5.1(c)), including those required to be made or obtained (A) with or from any PUCs or similar state regulatory bodies in connection with the consummation of the Transaction, and (B) with or from any foreign regulatory bodies under any foreign antitrust, competition or similar Laws or any foreign public service or foreign public utility commissions or similar foreign regulatory bodies in connection with the consummation of the Transaction, shall have been made or obtained by Final Order, (ii) the Committee on Foreign Investment in the United States terminates its review under 31 C.F.R. Part 800 and, where applicable, investigation, without unresolved national security concerns with respect to the Transaction, and (iii) all Governmental Consents set forth on Schedule 5.1(c) of the MetroPCS Disclosure Letter shall have been obtained by Final Order, in each of cases (i) and (ii), except as would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the business, assets, liabilities, financial condition or results of operations of MetroPCS and its Subsidiaries, taken as a whole (after giving effect to the Transaction), or the ability of either party to consummate the Transaction, and in each of cases (i), (ii) and (iii), without requiring DT or MetroPCS to take, or cause to be taken, any action, or to agree to any restriction, limitation or condition, in each case with respect to any of the assets (including FCC Licenses), business or product lines of MetroPCS, DT, TMUS, or any of their respective Subsidiaries, or any combination thereof, that would cause a Regulatory Material Adverse Condition.

(d) <u>No Order</u>. No Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any law, statute, ordinance, rule, regulation, judgment, injunction, decree or other order (whether temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits consummation of the Transaction (collectively, an <u>Order</u>).

5.2 <u>Conditions to Obligations of MetroPCS</u>. The obligations of MetroPCS to effect the Transaction are also subject to the satisfaction or waiver by MetroPCS at or prior to the Closing of the following conditions:

(a) <u>Representations and Warranties</u>. (i) The representations and warranties of DT, Global, Holding or TMUS, as applicable, set forth in Sections 3.1(d) (Ownership of Global, Holding and TMUS Shares), 3.1(g)(i) (Ownership of MetroPCS Common Stock), 3.2(b) (Capitalization) and 3.2(c)(ii) (Subsidiaries) shall be true and correct (A) on the date hereof and (B) at the Closing (except to the extent that such representation and warranty speaks only as of a particular date, in which case such representation and warranty shall be true and correct as of such earlier date, and except where the failure of such representations and warranties to be true and correct as of the date hereof and at the Closing, or such particular date, as applicable, is *de minimis*); (ii) the other representations and warranties of DT, Global, Holding or TMUS, as applicable, set forth in this Agreement shall be true and correct (A) on the date hereof and (B) at the Closing (except to the extent that any such representation and warranty speaks only as of a particular date, in which case such representation and warranty shall be true and correct as of such earlier date); <u>provided</u>, <u>however</u>, that notwithstanding anything herein to the contrary, the condition set forth in this Section 5.2(a)(ii) shall be deemed to have been satisfied even if any representations and warranties of DT, Global, Holding or TMUS are not so true and correct unless the failure of such representations and warranties to be so true and correct (read for purposes of this Section 5.2(a)(ii) without any materiality or TMUS Material Adverse Effect qualification), individually or in the aggregate, has had or would reasonably be likely to have a TMUS Material Adverse Effect; and (iii) MetroPCS shall have received at the Closing a certificate signed on behalf of DT by an executive officer of DT to the effect that the condition set forth in this

Section 5.2(a) has been satisfied as to the representations and warranties of DT, Global, and Holding, and a certificate signed on behalf of TMUS by an executive officer of TMUS to the effect that the condition set forth in this Section 5.2(a) has been satisfied as to the representations and warranties of TMUS.

(b) <u>Performance of Obligations of DT, Holding and TMUS</u>. DT, Global, Holding and TMUS (i) shall have performed in all respects all obligations required to be performed by them under this Agreement pursuant to Sections 2.2(a) and 2.2(c) at the Closing, (ii) shall have made available the DT Notes (including Additional DT Notes, if any) and the TMUS Working Capital Facility in compliance with Sections 4.13(b), (c) and (e), and (iii) shall have performed in all material respects all other obligations required to be performed by them under this Agreement at or prior to the Closing (provided that, for the avoidance of doubt, the determination regarding whether such parties shall have performed such obligations in all material respects shall be made with respect to all such obligations in the aggregate for all such parties), and MetroPCS shall have received a certificate signed on behalf of DT, Global and Holding by an executive officer of DT, to the effect that the condition set forth in this Section 5.2(b) has been satisfied as to the obligations of DT, Global and Holding, and a certificate signed on behalf of TMUS by an executive officer of TMUS, to the effect that the condition set forth in this Section 5.2(b) has been satisfied as to the obligations of TMUS.

(c) <u>No TMUS Material Adverse Effect</u>. Since the date hereof, there shall not have occurred any Circumstance that, individually or in the aggregate, has had, or is reasonably likely to have, a TMUS Material Adverse Effect, and MetroPCS shall have received at the Closing a certificate signed on behalf of DT by an executive officer of DT to the effect that the condition set forth in this Section 5.2(c) has been satisfied.

(d) <u>Ancillary Agreements</u>. MetroPCS shall have received an executed counterpart of each of the Ancillary Agreements, signed by each party other than MetroPCS.

(e) <u>FIRPTA Certificate</u>. TMUS shall have issued to MetroPCS a certificate described in Treasury Regulation Section 1.1445-2(c)(3) to the effect that TMUS Shares are not United States real property interests within the meaning of Section 897(c)(1) of the Code; provided, that if TMUS fails to deliver such certificate, MetroPCS shall be permitted to withhold from the consideration otherwise payable to DT pursuant to Section 2.2 any amounts required to be withheld pursuant to Section 1445 of the Code.

5.3 <u>Conditions to Obligations of DT, Holding and TMUS</u>. The obligations of DT, Holding and TMUS to effect the Transaction are also subject to the satisfaction or waiver by DT at or prior to the Closing of the following conditions:

(a) <u>Representations and Warranties</u>. (i) The representations and warranties of MetroPCS set forth in Sections 3.3(b) (Capitalization), 3.3(c)(ii) (Subsidiaries) and 3.3(u) (TMUS Stock Consideration) shall be true and correct (A) on the date hereof and (B) at the Closing (except to the extent that any such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty shall be true and correct as of such date, and except where the failure of such representations and warranties to be true and correct as of the date hereof and at the Closing, or such particular date, as applicable, is *de minimis*); and (ii) the other representations and warranties of MetroPCS set forth in this Agreement shall be true and correct on the date hereof and at the Closing (except to the extent that any such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty expressly speaks as of a particular date, in which case such representation and warranty shall be true and correct as of such date); <u>provided</u>, <u>however</u>, that notwithstanding anything herein to the contrary, the condition set forth in this Section 5.3(a)(ii) shall be deemed to have been satisfied even if any representations and warranties of MetroPCS are not so true and correct unless the failure of such representations and warranties of MetroPCS to be so true and correct (read for purposes of this Section 5.3(a)(ii) without any materiality or MetroPCS Material Adverse Effect qualification), individually or in the aggregate, has had or would reasonably be likely to have a MetroPCS Material Adverse Effect; and (iii) DT shall have received at the Closing a certificate signed on behalf of MetroPCS by an executive officer of MetroPCS to the effect that the condition set forth in this Section

(b) <u>Performance of Obligations of MetroPCS</u>. MetroPCS (i) shall have performed in all respects all obligations required to be performed by it under this Agreement pursuant to Sections 2.1 and 2.2(b) at the Closing, and (ii) shall have performed in all material respects all other obligations required to be performed by it under this Agreement at or prior to the Closing (<u>provided</u> that, for the avoidance of doubt, the determination regarding whether MetroPCS shall have performed such obligations in all material respects shall be made with respect to all such obligations in the aggregate), and DT shall have received a certificate signed by an executive officer of MetroPCS to such effect.

(c) <u>No MetroPCS Material Adverse Effect</u>. Since the date hereof, there shall not have occurred any Circumstance that, individually or in the aggregate, has had, or is reasonably likely to have, a MetroPCS Material Adverse Effect, and DT shall have received at the Closing a certificate signed on behalf of MetroPCS by an executive officer of MetroPCS to the effect that the condition set forth in this Section 5.3(c) has been satisfied.

(d) <u>Ancillary Agreements</u>. DT shall have received an executed counterpart of each of the Ancillary Agreements, signed by each party other than DT.

(e) <u>MetroPCS Existing Credit Agreement</u>. DT shall have received a payoff letter reasonably acceptable to it with respect to the termination of the MetroPCS Existing Credit Agreement and all commitments or other extensions of credit thereunder and the satisfaction and discharge of all principal, premium, if any, interest, fees and other amounts then due or outstanding thereunder and the satisfaction, release and discharge of all security interests, mortgages, liens and other Encumbrances over MetroPCS and its Subsidiaries properties and assets over MetroPCS and its Subsidiaries properties and assets securing such obligations (and such payoff letter shall require the administrative agent under the MetroPCS Existing Credit Agreement to deliver all instruments necessary or desirable to evidence or effect the foregoing).

(f) <u>MetroPCS Existing Notes</u>. Except to the extent refinanced pursuant to a change of control offer on or prior to the Closing with Permitted MetroPCS Notes or Additional DT Notes, the MetroPCS Existing Notes shall remain outstanding, there shall be no event of default in respect of any of the MetroPCS Existing Notes, and the consummation of the Transaction alone shall not give rise to any fact, event, circumstance or effect that with notice or lapse of time would constitute an event of default in respect of any of the MetroPCS Existing Notes.

(g) Listing of MetroPCS Shares. The TMUS Stock Consideration shall have been approved for listing on the NYSE, subject to official notice of issuance.

### ARTICLE VI

#### TERMINATION

6.1 <u>Termination</u>. Notwithstanding anything herein to the contrary, this Agreement may be terminated and the Transaction may be abandoned at any time prior to the Closing, whether before or after the MetroPCS Stockholder Approval:

(a) by mutual written consent of MetroPCS and DT;

(b) by either MetroPCS or DT, by written notice to the other party, if any Governmental Entity of competent jurisdiction shall have issued a final and non-appealable Order or taken any other final and non-appealable action permanently enjoining, restraining, denying or otherwise prohibiting the consummation of Transaction; <u>provided</u> that the party seeking to terminate this Agreement shall have used its reasonable best efforts to have such Order lifted if and to the extent required by Section 4.11;

(c) by either MetroPCS or DT, by written notice to the other party, if the Transaction shall not have been consummated on or before October 3, 2013; provided, however, that if the conditions set forth in Sections 5.1(b) and 5.1(c) shall not have been satisfied by October 3, 2013, either party may extend the Termination Date from time to time, by written notice to the other party given prior to the Termination Date in effect prior to such notice, to a date not later than January 3, 2014 (such date as it may be extended from time to time pursuant to this Section 6.1(c), the <u>Termination Date</u> );

(d) by DT (provided that none of DT, Global, Holding and TMUS is then in material breach of any representation, warranty, covenant or other agreement herein), by written notice to MetroPCS, in the event that (i) assuming all conditions set forth in Article V (other than Sections 5.3(a) and (b)) were satisfied and the Closing were otherwise to occur on the date DT delivers such notice, a breach by MetroPCS of any representation, warranty, covenant or other agreement contained herein would result in a failure of a condition set forth in Section 5.3(a) or (b), and (ii) such breach (A) if curable, has not been cured within 30 calendar days following MetroPCS s receipt of such notice, or if the Termination Date is less than 30 calendar days from such notice, has not been or cannot reasonably be expected to be cured by the Termination Date or (B) is not curable;

(e) by MetroPCS (provided that MetroPCS is not then in material breach of any representation, warranty, covenant or other agreement herein), by written notice to DT, in the event that (i) assuming all conditions set forth in Article V (other than Section 5.2(a) and (b)) were satisfied and the Closing were otherwise to occur on the date MetroPCS delivers such notice, a breach by DT, Global, Holding or TMUS of any representation, warranty, covenant or other agreement contained herein would result in a failure of a condition set forth in Section 5.2(a) or (b), and (ii) such breach (A) if curable, has not been cured within 30 calendar days following DT s receipt of such notice, or if the Termination Date is less than 30 calendar days from such notice, has not been or cannot reasonably be expected to be cured by the Termination Date or (B) is not curable;

(f) by either DT or MetroPCS, by written notice to the other party, if the MetroPCS Stockholder Approval shall not have been obtained at the MetroPCS Stockholders Meeting, or at any adjournment or postponement thereof, at which a vote seeking the MetroPCS Stockholder Approval was taken; <u>provided</u>, <u>however</u>, that no party may terminate this Agreement pursuant to this Section 6.1(f) if such party has breached in any material respect any of its obligations under this Agreement in any manner that would reasonably be expected to cause the failure to obtain the MetroPCS Stockholder Approval at the MetroPCS Stockholders Meeting or at any adjournment or postponement thereof;

(g) by DT, by written notice to MetroPCS prior to receipt of the MetroPCS Stockholder Approval, if there shall have been a MetroPCS Adverse Recommendation Change, whether or not in compliance with Section 4.5;

(h) by DT, by written notice to MetroPCS, if since the date hereof there shall have been a MetroPCS Material Adverse Effect and such MetroPCS Material Adverse Effect is not curable or, if curable, (i) is not cured within 30 calendar days after such notice is given by DT to MetroPCS or (ii) if the Termination Date is less than 30 calendar days from such notice, has not been or cannot reasonably be expected to be cured by the Termination Date; or

(i) by MetroPCS, by written notice to DT, if since the date hereof there shall have been a TMUS Material Adverse Effect and such TMUS Material Adverse Effect is not curable or, if curable, (i) is not cured within 30 calendar days after such notice is given by MetroPCS to DT or (ii) if the Termination Date is less than 30 calendar days from such notice, has not been or cannot reasonably be expected to be cured by the Termination Date.

#### 6.2 Effect of Termination and Abandonment.

(a) In the event of termination of this Agreement and the abandonment of the Transaction pursuant to this Article VI, this Agreement (other than Section 4.9 (Expenses), this Section 6.2 (Effect of Termination and

Abandonment) and Article VII (Miscellaneous and General)) shall become void and of no effect with no liability on the part of any party hereto (or of any of its directors, officers, employees, agents, legal and financial advisors or other Representatives); <u>provided</u>, <u>however</u>, except as otherwise provided herein, no such termination shall relieve any party hereto of any liability or damages resulting from any willful breach of any material provision of this Agreement, the Confidentiality Agreement or any other agreement delivered in connection herewith prior to the termination of this Agreement.

(b) If (i) DT terminates this Agreement pursuant to Section 6.1(g), or (ii) DT or MetroPCS terminates this Agreement pursuant to Section 6.1(f) following (A) a material breach by MetroPCS of Section 4.3, 4.4 or 4.5 that is reasonably related to the failure to obtain the MetroPCS Stockholder Approval or (B) a MetroPCS Adverse Recommendation Change, MetroPCS shall make a cash payment to DT in the amount of \$150,000,000 (the <u>MetroPCS Termination Amount</u>) in immediately available funds, as directed by DT in writing, within two Business Days after such termination by DT or concurrently with or prior to a termination by MetroPCS.

(c) If DT or MetroPCS terminates this Agreement pursuant to Section 6.1(f) in circumstances other than those described in Section 6.2(b)(ii) or pursuant to Section 6.1(c) and (i) (A) at or prior to the termination of this Agreement, a Person or group shall have publicly made a MetroPCS Acquisition Proposal or a MetroPCS Acquisition Proposal shall have otherwise become publicly announced or shall have been communicated to MetroPCS, the MetroPCS Board or MetroPCS s management and shall not have been publicly withdrawn or rescinded (if publicly made or announced) prior to the MetroPCS Stockholders Meeting, and (B) no later than 12 months after the termination of this Agreement, MetroPCS enters into, publicly approves or submits to the MetroPCS Stockholders for approval, an agreement with respect to a MetroPCS Acquisition Proposal, or a MetroPCS Acquisition Proposal is consummated (which in each case need not be the same MetroPCS Acquisition Proposal as the MetroPCS Acquisition Proposal described in clause (A), or made by the same Person or group as the MetroPCS Acquisition Proposal described in clause (A)), or (ii) (x) at or prior to the termination of this Agreement, one or more Persons or groups shall have publicly made one or more MetroPCS Acquisition Proposals or one or more MetroPCS Acquisition Proposals shall have otherwise become publicly announced or shall have been communicated to MetroPCS, the MetroPCS Board or MetroPCS s management, all of which have been publicly withdrawn or rescinded (if publicly made or announced) prior to the MetroPCS Stockholders Meeting, and (y) no later than 12 months after the termination of this Agreement, MetroPCS enters into, publicly approves or submits to the MetroPCS Stockholders for approval, an agreement with respect to a MetroPCS Acquisition Proposal, or a MetroPCS Acquisition Proposal is consummated, which MetroPCS Acquisition Proposal is made by any of the same Persons or groups as the MetroPCS Acquisition Proposal(s) described in clause (x), then in either case of clauses (i) or (ii), MetroPCS will pay to DT, on the date of the consummation of the transaction in respect of such MetroPCS Acquisition Proposal described in clause (i)(B) or (ii)(y), the MetroPCS Termination Amount in immediately available funds, as directed by DT in writing (for purposes of this Section 6.2(c), all references in the term MetroPCS Acquisition Proposal to 20% or more shall be deemed to be references to more than 50%).

(d) If (i) DT or MetroPCS terminates this Agreement pursuant to Section 6.1(b) as a result of an Order or action under Regulatory Law or (ii) DT or MetroPCS terminates this Agreement pursuant to Section 6.1(c) and, in each case, at the time of such termination all of the conditions set forth in Article V (other than (A) the conditions set forth in Sections 5.1(b), 5.1(c), and 5.1(d), and (B) those other conditions that, by their nature, cannot be satisfied until the Closing Date but would be capable of satisfaction if the Closing Date were the date of such termination) have been satisfied or waived on or prior to the date of such termination, DT shall make a cash payment to MetroPCS in the amount of \$250,000,000 (the <u>DT Termination Amount</u>) in immediately available funds, as directed by MetroPCS in writing, within two Business Days after such termination by MetroPCS or concurrently with or prior to a termination by DT.

(e) Each of MetroPCS and DT acknowledges that, in the event of a termination of this Agreement as described in Section 6.2(b), 6.2(c), or 6.2(d), (i) Damages would be extremely difficult to calculate, (ii) the

provisions regarding the MetroPCS Termination Amount or the DT Termination Amount, as applicable, represent the parties best estimate of such Damages and as such are an integral part of the Transaction, and (iii) the MetroPCS Termination Amount or the DT Termination Amount, as applicable, is not a penalty, but rather is liquidated damages in a reasonable amount that will compensate and reimburse DT or MetroPCS, as applicable, in the circumstances in which the MetroPCS Termination Amount or the DT Termination Amount, as applicable, is required for the efforts, costs and expenses expended and opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the Transaction, which amount would otherwise be impossible to calculate with precision. Accordingly, and notwithstanding anything to the contrary in this Agreement other than the immediately preceding sentence, if this Agreement is terminated in accordance with its terms and such termination gives rise to the obligation of MetroPCS to pay the MetroPCS Termination Amount or DT to pay the DT Termination Amount, as applicable, the sole and exclusive remedy of the other party and its Subsidiaries and their respective officers, directors and Affiliates against such paying party and its Subsidiaries and their respective officers, directors and Affiliates for any Damages resulting from, arising out of, or incurred in connection with, this Agreement (including termination thereof) or any transactions ancillary hereto shall be the MetroPCS Termination Amount or the DT Termination Amount, as applicable, and no Person shall have, except as provided herein, any rights or claims against such paying party and its Subsidiaries and their respective officers, directors and Affiliates under this Agreement or otherwise, whether at law or equity, in contract, in tort or otherwise, and neither the paying party, nor any of its Subsidiaries nor their respective officers, directors or Affiliates shall have any further liability or obligation resulting from, arising out of, or incurred in connection with, this Agreement; provided, that nothing in this Section 6.2(e) shall apply to any party s rights to seek equitable remedies, including injunctive relief or specific performance, with respect to the surviving provisions of and obligations under this Agreement, or to limit DT s, Global s, Holding s, TMUS s or MetroPCS s rights with respect to any Damages incurred or suffered by DT, Global, Holding, TMUS or MetroPCS, as applicable, as a result of the willful breach by DT, Global, Holding, TMUS or MetroPCS, as applicable, of any provision of this Agreement, the Confidentiality Agreement or any other agreement delivered in connection herewith prior to the termination of this Agreement. In no event shall MetroPCS be required to pay the MetroPCS Termination Amount, or DT be required to pay the DT Termination Amount, on more than one occasion.

#### ARTICLE VII

#### MISCELLANEOUS AND GENERAL

7.1 <u>Survival</u>. None of the representations and warranties of the parties contained in this Agreement or in any instrument delivered pursuant to this Agreement shall survive the Closing. Except for any covenant or agreement that by its terms contemplates performance after the Closing, none of the covenants and agreements of the parties contained in this Agreement shall survive the Effective Time.

7.2 <u>Amendment: Waivers, Etc</u>. No amendment, modification or discharge of this Agreement, and no waiver hereunder, and no extension of time for the performance of any of the obligations hereunder, shall be valid or binding unless set forth in writing and duly executed by (a) MetroPCS where enforcement of the amendment, modification, discharge, waiver or extension is sought against MetroPCS or (b) DT where enforcement of the amendment, modification, discharge, waiver or extension is sought against MetroPCS or (b) DT where enforcement of the amendment, modification, discharge, waiver or extension is sought against DT, Holding or TMUS. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the party granting such waiver in any other respect or at any other time. The waiver by MetroPCS or DT of a breach of, or a default under, any of the provisions hereof, or to exercise any right or privilege hereunder, shall not be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. Except as expressly provided in this Agreement, the rights and remedies herein provided are cumulative and none is exclusive of any other, or of any rights or remedies that any party may otherwise have at law or in equity. Notwithstanding anything to the contrary

contained herein, after the MetroPCS Stockholder Approval is obtained, no amendment, modification, discharge or waiver of this Agreement or any portion hereof shall be made that by Law requires further approval by the MetroPCS Stockholders without obtaining such approval.

7.3 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each such counterpart (including any facsimile or electronic document transmission of such counterpart) being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

7.4 <u>Governing Law; Jurisdiction; Forum; Waiver of Trial by Jury</u>. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE WITHOUT REGARD TO ANY CONFLICT-OF-LAWS RULE OR PRINCIPLE THEREOF. Each party hereto agrees that it shall bring any action or proceeding in respect of any claim arising out of, or related to, this Agreement or the Transaction, exclusively in the Delaware Court of Chancery, New Castle County, or solely if that court does not have jurisdiction, a federal court sitting in the State of Delaware (the <u>Chosen Courts</u>), and solely in connection with claims arising under this Agreement or the Transaction (a) irrevocably submits to the exclusive jurisdiction of the Chosen Courts, (b) waives any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waives any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any party hereto, and (d) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 7.5. Each party hereto irrevocably designates C.T. Corporation as its agent and attorney-in-fact for the acceptance of service of process and making an appearance on its behalf in any such claim or proceeding and for the taking of all such acts as may be necessary or appropriate in order to confer jurisdiction over it before the Chosen Courts and each party hereto stipulates that such consent and appointment is irrevocable and coupled with an interest. EACH PARTY HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF, OR RELATING TO, THIS AGREEMENT OR THE TRANSACTION.

7.5 <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party to the others shall be in writing and delivered personally or sent by registered or certified mail, return receipt requested and postage prepaid, or by facsimile (providing confirmation of such facsimile transmission):

if to MetroPCS, to:

MetroPCS Communications, Inc.

2250 Lakeside Blvd.

Richardson, Texas 75082

Attention: Mark A. Stachiw

Melanie Stapp Klint

Fax: (866) 685-9618

with copies to (which shall not constitute notice):

Gibson, Dunn & Crutcher LLP

2100 McKinney Avenue, Suite 1100

Dallas, Texas 75201

Attention: Jeffrey A. Chapman

Robert B. Little

Fax: (214) 571-2900

and

Akin Gump Strauss Hauer & Feld LLP

1700 Pacific Avenue, Suite 4100

Dallas, Texas 75201

Attention: J. Kenneth Menges, Jr., P.C.

Fax: (214) 969-4343

if to DT, Global, Holding or TMUS, to:

c/o Deutsche Telekom AG

Friedrich-Ebert-Alle 140

53113 Bonn, Germany

Attention: General Counsel

Fax: +49-228-181-74008

with a copy to (which shall not constitute notice):

Wachtell, Lipton, Rosen & Katz

51 West 52nd Street

New York, New York 10019

Attention: Adam O. Emmerich

Fax: (212) 403-2000

or to such other Persons or addresses as may be designated in writing by the party to receive such notice as provided above, and any such notice, request, instruction or other document shall be deemed to have been given as of the date received by the addressee as provided above; <u>provided</u> that any such notice, request, instruction or other document received by facsimile transmission or otherwise at the addressee s location on any Business Day after 5:00 p.m. (addressee s local time) shall be deemed to have been received by such addresse at 9:00 a.m. (addressee s local time) on the next Business Day.

7.6 <u>Entire Agreement</u>. This Agreement (including any annexes and exhibits hereto), the TMUS Disclosure Letter, the MetroPCS Disclosure Letter, the Confidentiality Agreement and, when executed, the Ancillary Agreements, constitute the entire agreement, and supersede all other prior and contemporaneous agreements, understandings, undertakings, arrangements, representations and warranties, both written and oral, among the parties with respect to the subject matter hereof.

7.7 <u>Specific Performance</u>. The parties agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with the terms hereof and that, subject to Section 6.2(e), the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement or to enforce specifically the performance of the terms and provisions hereof in any of the Chosen Courts without any requirement to post bond, in addition to any other remedy to which they are entitled at law or in equity.

7.8 <u>No Third-Party Beneficiaries</u>. This Agreement is not intended to confer upon any Person other than the parties hereto any rights or remedies hereunder, except to the extent contemplated by Section 4.23. Notwithstanding the foregoing, following the Effective Time, the provisions of

# Table of Contents

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

Article II shall be enforceable by each holder of MetroPCS Common Stock or MetroPCS Stock Options as of the Effective Time solely to the extent necessary for any such holder to receive the cash and/or MetroPCS Common Stock, as applicable, to which it is entitled pursuant to Article II.

7.9 <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability or the other provisions hereof. If

any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision; and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

7.10 Interpretation. The table of contents and headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. The parties hereto have participated jointly in the negotiation and drafting of this Agreement and, in the event that an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

7.11 <u>Assignment</u>. This Agreement shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors, heirs, legal representatives and permitted assigns. Neither party may directly or indirectly assign any of its rights or delegate any of its obligations under this Agreement, by operation of Law or otherwise, without the prior written consent of the other party. Any purported direct or indirect assignment in violation of this Section 7.11 shall be null and void *ab initio*. Notwithstanding the foregoing, DT may assign any of its rights and/or delegate any of its obligations under this Agreement to one or more of its wholly-owned Subsidiaries (but no such assignment shall relieve DT of any of its obligations hereunder).

7.12 Limitation of Liability. Notwithstanding anything to the contrary contained in this Agreement, (a) neither DT, MetroPCS, their respective Subsidiaries or Affiliates, or any of their respective officers, directors, employees or attorneys shall be liable to any Person with respect to a claim under this Agreement for any punitive, indirect or exemplary damages or any damages that are not the reasonably foreseeable result of a breach hereof (other than to the extent such damages are paid to a third party), whether such claim is based on warranty, contract, tort (including negligence or strict liability) or otherwise, and (b) this Agreement shall not create or be deemed to create any liability or obligation on the part of any Representative, Affiliate or direct, indirect or beneficial owner or holder of any securities of any party hereto (unless itself a party hereto).

7.13 Securities Matters. Each of MetroPCS and DT represents and warrants to the other party as follows:

(a) <u>Experience</u>; <u>Risk</u>. Such party has such knowledge, sophistication and experience in financial and business matters that it is capable of evaluating the merits and risks of the receipt of TMUS Shares or the TMUS Stock Consideration, respectively, and of protecting its interests in connection herewith. DT has the ability to bear the economic risk of this investment, including complete loss of the investment.

(b) <u>Investment</u>. Such party is acquiring TMUS Shares or the TMUS Stock Consideration, respectively, for investment for its own account, not as a nominee or agent, and not with a view to, or for resale in connection with, any distribution thereof, and has no present intention of selling, granting any participation in or otherwise distributing the same. Such party understands that TMUS Shares or the TMUS Stock Consideration, respectively, have not been registered under the Securities Act, by reason of a specific exemption from the registration provisions of the Securities Act which depends upon, among other things, the *bona fide* nature of the investment intent and the accuracy of such party s representations as expressed in this Section 7.13.

(c) <u>Access to Information</u>. Such party acknowledges that, as of the date hereof, it has been afforded (i) the opportunity to ask such questions as it has deemed necessary of, and to receive answers from, Representatives of the other party concerning the terms and conditions of the Transaction and TMUS Shares or the TMUS Stock Consideration, respectively, and the merits and risks of investing in TMUS Shares or the TMUS Stock Consideration, respectively, and the merits and risks of investing in TMUS Shares or the TMUS Stock Consideration, respectively, and any such questions have been answered to such party s reasonable satisfaction;

(ii) access to information about TMUS or MetroPCS, respectively, and its financial condition, results of operations, business, properties, management and prospects sufficient to enable it to evaluate its investment; (iii) the opportunity to obtain such additional information that the other party possesses or can acquire without unreasonable effort or expense that is necessary to make an informed investment decision with respect to the investment and any such additional information has been provided to such party s reasonable satisfaction; and (iv) the opportunity to ask questions of management of the other party and any such questions have been answered to such party s reasonable satisfaction. Such party has sought such accounting, legal and tax advice as it has considered necessary to make an informed decision with respect to its acquisition of TMUS Shares or the TMUS Stock Consideration, respectively. Each party hereto acknowledges that no other party hereto nor any Affiliate or Representative of such other party has made any representation, express or implied, with respect to the accuracy, completeness or adequacy of any available information except or to the extent such information is covered by the representations and warranties contained herein.

(d) Accredited Investor. Such party is an accredited investor within the meaning of Regulation D, Rule 501(a), promulgated by the SEC.

(e) <u>Restricted Securities</u>; <u>Rule 144</u>. Such party understands that TMUS Shares and the TMUS Stock Consideration, respectively, are characterized as restricted securities under the United States federal securities laws inasmuch as they are being acquired from the other party in a transaction not involving a public offering and that under such laws and applicable regulations the TMUS Stock Consideration may be resold without registration under the Securities Act only in certain limited circumstances. Such party acknowledges that TMUS Shares or the TMUS Stock Consideration, respectively, must be held indefinitely unless a sale of such TMUS Shares or TMUS Stock Consideration, respectively, is subsequently registered under the Securities Act or an exemption from such registration is available. Such party is aware of the provisions of Rule 144 promulgated under the Securities Act which permit limited resale of shares purchased in a private placement or shares owned by certain Persons associated with TMUS subject to the satisfaction of certain conditions.

(f) <u>Legends</u>. DT understands and agrees that each certificate, if any, representing the TMUS Stock Consideration and any securities issued in respect thereof or in exchange therefor shall bear a legend in the following forms (in addition to any other legend required under applicable state and foreign securities laws) (and a comparable notation or other arrangement will be made with respect to any uncertificated TMUS Stock Consideration):

THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE BEEN ACQUIRED FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. SUCH SHARES MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR UNLESS THE ISSUER RECEIVES AN OPINION OF COUNSEL REASONABLY ACCEPTABLE TO IT STATING THAT SUCH SALE OR TRANSFER IS EXEMPT FROM THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS OF SAID ACT.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE PROVISIONS OF A STOCKHOLDER S AGREEMENT, DATED AS OF [ ], 201[ ], TO WHICH THE ISSUER AND CERTAIN OF ITS STOCKHOLDERS ARE PARTY, A COPY OF WHICH MAY BE INSPECTED AT THE PRINCIPAL OFFICE OF THE ISSUER OR OBTAINED FROM THE ISSUER WITHOUT CHARGE.

7.14 <u>Transfer Taxes</u>. Any Transfer Taxes imposed with respect to the Transaction shall be borne by MetroPCS; provided, that any German value added tax payable with respect to the Stock Purchase shall be borne by DT and its Subsidiaries (other than TMUS and its Subsidiaries). The party so required by applicable Law shall file all necessary Tax Returns and other documentation with respect to all Transfer Taxes, and, if required by the applicable Law, the other parties shall, and shall cause their Affiliates to join in the execution of any such Tax Returns and other documentation.

7.15 <u>Reliance of Other Parties</u>. This Agreement contains representations and warranties by each of the parties hereto that are the product of negotiations between the parties hereto and are for the sole benefit of the parties hereto. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance with Section 7.2. These representations and warranties have been qualified by certain disclosures that were made between the parties in the TMUS Disclosure Letter and the MetroPCS Disclosure Letter, which disclosures are not reflected in this Agreement itself. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the date of this Agreement or as of any other date.

7.16 <u>Effect of Breaches</u>. Notwithstanding anything to the contrary herein, for all purposes under this Agreement, a breach of any representation, warranty, covenant or agreement contained herein by DT, Global, Holding or TMUS shall be deemed to be a breach of such representation, warranty, covenant or agreement by each of DT, Global, Holding and TMUS.

[Remainder of page intentionally left blank.]

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the parties hereto as of the date first written above.

#### **Deutsche Telekom AG**

By: /s/ Timotheus Hottges Name: Timotheus Hottges Title: Member of the Deutsche Telekom

Board of Management, Finance

By: /s/ Dr. Thomas Kremer Name: Dr. Thomas Kremer Title: Member of the Deutsche Telekom

> Board of Management for Data Privacy, Legal Affairs and Compliance

#### **T-Mobile Global Zwischenholding GmbH**

- By: /s/ Axel Lützner Name: Axel Lützner Title: Authorized Signatory
- By: /s/ Vincente Vento Name: Vincente Vento Title: Authorized Signatory

#### **T-Mobile Global Holding GmbH**

- By: /s/ Axel Lützner Name: Axel Lützner Title: Authorized Signatory
- By: /s/ Vincente Vento Name: Vincente Vento Title: Authorized Signatory

#### T-Mobile USA, Inc.

By: /s/ John Legere Name: John Legere Title: Chief Executive Officer

#### **MetroPCS Communications, Inc.**

By: /s/ Roger D. Linquist Name: Roger D. Linquist Title: Chief Executive Officer and

Chairman of the Board

[Signature Page to Business Combination Agreement]

### <u>Exhibit A</u>

# FOURTH AMENDED AND RESTATED

### CERTIFICATE OF INCORPORATION

OF

### T-MOBILE US, INC.

#### Pursuant to Sections 242 and 245

#### of the Delaware General Corporation Law ( DGCL )

The undersigned, [Name], hereby certifies that,

ONE: He is the duly elected and acting [Title] of MetroPCS Communications, Inc. (the <u>Corporation</u>).

TWO: The name of the Corporation prior to the Effective Time is MetroPCS Communications, Inc..

THREE: The original certificate of incorporation of the Corporation was filed in the Office of the Secretary of State of the State of Delaware on March 10, 2004 pursuant to the DGCL.

FOUR: The directors and the stockholders of the Corporation, in accordance with Sections 242 and 245 of the DGCL, have duly adopted and approved this Fourth Amended and Restated Certificate of Incorporation.

The certificate of incorporation of the Corporation is hereby amended and restated to read in its entirety as follows:

#### ARTICLE I

#### CORPORATE NAME

The name of the Corporation from and after the Effective Time is T-Mobile US, Inc.

#### **ARTICLE II**

#### **REGISTERED ADDRESS AND AGENT**

The address of the registered office of the Corporation in the State of Delaware is 2711 Centerville Road, City of Wilmington, County of New Castle, 19808. The name of its registered agent at such address is Corporation Service Company.

#### **ARTICLE III**

#### NATURE OF BUSINESS

The nature of the business or purposes to be conducted or promoted is to engage in any lawful act or activity for which corporations may be organized under the DGCL.

#### ARTICLE IV

### AUTHORIZED CAPITAL STOCK

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

(A) <u>General</u>. The Corporation is authorized to issue two (2) classes of capital stock to be designated, respectively, Common Stock and Preferred Stock. The total number of shares which the Corporation is authorized to issue is One Billion One Hundred Million (1,100,000,000) shares. One Billion (1,000,000,000) shares shall be Common Stock, par value \$0.00001 per share, and One Hundred Million (100,000,000) shares shall be Preferred Stock, par value \$0.00001 per share.

(B) Upon the Effective Time, each share of Common Stock, par value \$0.0001 per share, issued and outstanding immediately before the Effective Time automatically shall be, without any action on the part of the Corporation or the holder thereof, reclassified as, and converted into, 0.5 of a validly issued, fully paid and non-assessable share of Common Stock, par value \$0.00001 per share (such reclassification and conversion, the <u>Reverse Stock Spl</u>it ), subject to the treatment of fractional share interests as described below. Any stock certificate that, immediately before the Effective Time, represented a share or shares of Common Stock shall, from and after the Effective Time, automatically and without the presentation of such certificate to the Corporation for exchange thereof, represent such share or shares of Common Stock that result from the Reverse Stock Split, subject to the elimination of fractional shares as described below. Notwithstanding the foregoing, each Common Stock holder that would otherwise be entitled to a fraction of a share of Common Stock as a result of the Reverse Stock Split after aggregating all fractions of shares of Common Stock to be received by such holder shall in lieu thereof be entitled to receive payment in cash (rounded up to the nearest whole cent, without interest and subject to applicable withholding taxes) from the Corporation s transfer agent in lieu of such fractional shares in accordance with Section 155 of the DGCL.

#### (C) Preferred Stock.

1. The Preferred Stock may be issued from time to time in one or more series and in such amounts as may be determined by the Board of Directors or by order or decree of a court of competent jurisdiction over the Corporation administering any applicable statute of the United States relating to plans of reorganization of corporations, subject to any qualifications, limitations and restrictions set forth elsewhere in this Article IV or in Article VIII. The voting powers, designations, preferences and relative, participating, optional or other special rights, if any, and the qualifications, limitations or restrictions thereof, if any, of each series of the Preferred Stock shall be such as are fixed by the Board of Directors or fixed by such court, the authority to do so being hereby expressly granted, and as are stated and expressed in a resolution or resolutions adopted by the Board of Directors or in an order or decree of such court providing for the issue of such series of Preferred Stock (herein called the <u>Certificate of Designation</u>). The Certificate of Designation as to any series shall, subject to any qualifications, limitations and restrictions set forth elsewhere in this Article IV or in Article VIII, (a) designate the series, (b) fix the dividend rate, if any, of such series, the payment dates for dividends on shares of such series and the date or dates, or the method of determining the date or dates, if any, from which dividends on shares of such series shall be cumulative, (c) fix the amount or amounts payable on shares of such series upon voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Corporation, and (d) state the price or prices or rates, and adjustments, if any, at which, and the time or times and the terms and conditions upon which, the shares of such series may be redeemed at the option of the Corporation or at the option of the holder or holders of shares of such series or upon the occurrence of a specified event, and state whether such shares may be redeemed for cash, property or rights, including securities of the Corporation or other equity securities; and such Certificate of Designation may (i) limit the number of shares of such series that may be issued, (ii) provide for a sinking fund for the purchase or redemption of shares of such series and specify the terms and conditions governing the operations of any such fund, (iii) grant voting rights to the holders of shares of such series, (iv) impose conditions or restrictions upon the creation of indebtedness of the Corporation or upon the issuance of additional Preferred Stock or other capital stock ranking on a parity therewith, or prior thereto, with respect to dividends or distribution of assets upon liquidation, (v) impose conditions or restrictions upon the payment of dividends upon, or the making of other distributions to, or the acquisition of, shares ranking junior to such series with respect to dividends or distributions of assets upon liquidation, (vi) state the time or times, the price or prices or the rate or rates of exchange and other terms, conditions and adjustments upon which shares of any such series may be made convertible into, or exchangeable for, at the option of the holder or the Corporation or upon the occurrence of a specified event, shares of any other class or classes or of any other series of Preferred Stock or any other class or classes of stock or other securities of the Corporation, and (vii) grant such other special rights and impose such qualifications, limitations or restrictions thereon as shall be fixed by the Board of Directors or such court, to the extent not inconsistent with this Article IV or Article VIII and to the full extent now or hereafter permitted by the laws of the State of Delaware.

- 2 -

2. Preferred Stock that is redeemed, purchased or retired by the Corporation shall assume the status of authorized and unissued Preferred Stock and may thereafter, subject to the provisions of any Certificate of Designation providing for the issue of any particular series of Preferred Stock, be reissued in the same manner as authorized and unissued Preferred Stock.

(D) <u>Common Stock</u>. All shares of Common Stock shall be identical except as expressly set forth in this Article IV. Each share of Common Stock shall have attributed to it the number of votes set forth in Section (E) below.

#### (E) Rights of Holders of Capital Stock.

1. <u>Holders of Preferred Stock</u>. Except as such rights may be specifically limited herein, the rights of holders of Preferred Stock shall be as set forth in any Certificate of Designation relating thereto.

2. <u>Holders of Common Stock</u>. The rights of holders of Common Stock shall be as set forth in this Section (E)(2), except with respect to such rights as are set forth in Section (F) of this Article IV.

(a) <u>Dividend Rights</u>. Subject to the prior rights of holders of all classes of stock at the time outstanding having prior rights as to dividends, the holders of the Common Stock shall be entitled to receive, when and as declared by the Board of Directors, out of any assets of the Corporation legally available therefor, such dividends, distributed ratably among the holders of the Common Stock in proportion to the number of shares of such Common Stock owned by each such holder, as may be declared from time to time by the Board of Directors.

(b) <u>Liquidation Preference</u>. Subject to the prior rights of holders of all classes of stock at the time outstanding having prior rights as to distributions in the event of liquidation, dissolution or winding up of the Corporation, and after any and all distributions are made in accordance therewith, in such event, either voluntary or involuntary, the remaining assets and funds of the Corporation legally available for distribution shall be distributed ratably among the holders of the Common Stock in proportion to the number of shares of such Common Stock owned by each such holder.

### 3. Redemption.

(a) Subject to Section (E)(3)(e) of this Article IV, if, at any time, a holder of shares of Common Stock or Preferred Stock acquires additional shares of Common Stock or Preferred Stock, or is otherwise attributed with ownership of such shares, that would cause the Corporation to violate (in each case, an <u>FCC Violation</u>) (A) any requirement of the Federal Communications Commission (FCC) regarding foreign ownership (collectively, <u>Foreign Ownership Requirements</u>) or (B) any other rule or regulation of the FCC applicable to the Corporation, then the Corporation may, at the option of the Board of Directors, redeem from the holder or holders causing such FCC Violation a sufficient number of shares of Common Stock or Preferred Stock to eliminate the FCC Violation by paying in cash therefor a sum equal to the Redemption Price. The <u>Redemption Price</u> (herein so called) shall equal such price as is mutually determined by such stockholders and the Corporation, or, if no mutually acceptable agreement can be reached, shall equal either (i) seventy-five percent (75%) of the fair market value of the Common Stock (the <u>Common Stock Fair Market Value</u>) or the Fair Market Value of the Preferred Stock, as applicable, where such holder caused the FCC Violation, or (ii) one hundred percent (100%) of the Common Stock Fair Market Value or the Fair Market Value of the Preferred Stock, as applicable, where the FCC Violation was caused by no fault of the holder; *provided, however*, that the determination of whether such party caused the FCC Violation shall be made, in good faith, by the disinterested members of the Board of Directors. As used in this Section (E)(3), the Common Stock Fair Market Value shall be determined as follows:

(i) if the Common Stock is publicly traded at the time of determination, the average of the closing prices for the Common Stock on all domestic securities exchanges on which the Common Stock may at

- 3 -

the time be listed, or, if there have been no sales of the Common Stock on any such exchange on such day, the average of the highest bid and lowest asked prices for the Common Stock on all such exchanges at the end of such day, or, if on any day the Common Stock is not so listed, the average of the representative bid and asked prices for the Common Stock quoted on the NASDAQ system as of the close of trading on such day, or if on any day such security is not quoted in the NASDAQ system, the average of the highest bid and lowest asked prices for the Common Stock on such day in the domestic over-the-counter market as reported by the Pink Sheets, LLC, or any similar successor organization, in each such case averaged over the 30-day period ending three days prior to the Redemption Date (as defined in Section (E)(3)(b) of this Article IV); and

(ii) if the Common Stock is not publicly traded at the time of determination then, the fair value of the Common Stock as determined in good faith by the disinterested members of the Board of Directors.

As used in this Section (E)(3), the Preferred Stock Fair Market Value shall mean the value determined by multiplying the Common Stock Fair Market Value by the number of shares of Common Stock into which the share of Preferred Stock is then convertible.

(b) At least five (5) but no more than thirty (30) days prior to any date on which Common Stock or Preferred Stock is to be redeemed (a <u>Redemption Date</u>), written notice shall be sent by mail, first class postage prepaid, overnight mail, facsimile, or electronic mail to each holder of record (at the close of business on the business day next preceding the day on which notice is given) of the shares of Common Stock or Preferred Stock to be redeemed, at the address last shown on the records of the Corporation for such holder, notifying such holder of the redemption to be effected, specifying the number of shares to be redeemed from such holder, the Redemption Date, the Redemption Price, the place at which payment may be obtained and calling upon such holder to surrender to the Corporation, in the manner and at the place designated, his, her or its certificate or certificates representing the shares to be redeemed (the <u>Redemption Notice</u>). Except as provided in Section (E)(3)(c) of this Article IV, on or after the Redemption Date, each holder of shares of Common Stock or Preferred Stock to be redeemed shall surrender to the Corporation the certificate or certificates representing such shares, in the manner and at the place designated in the Redemption Notice, and thereupon the Redemption Price of such shares shall be payable to the order of the person whose name appears on such certificate or certificates as the owner thereof and each surrendered certificate shall be canceled. In the event less than all the shares represented by any such certificate are redeemed, a new certificate shall be issued representing the unredeemed shares.

(c) From and after the Redemption Date, unless there shall have been a default in payment of the Redemption Price, all rights of the holders of shares of Common Stock or Preferred Stock designated for redemption in the Redemption Notice as holders of such shares of Common Stock or Preferred Stock (except the right to receive the Redemption Price without interest upon surrender of their certificate or certificates) shall cease with respect to such shares, and such shares shall not thereafter be transferred on the books of the Corporation or be deemed to be outstanding for any purpose whatsoever. If the funds of the Corporation legally available for redemption of shares of Common Stock or Preferred Stock on any Redemption Date are insufficient to redeem the total number of shares of Common Stock or Preferred Stock to be redeemed on such date, those funds which are legally available will be used to redeem the maximum possible number of such shares of Common Stock or Preferred Stock or Preferred Stock not redeemed shall remain outstanding and entitled to all the rights and preferences provided herein. At any time thereafter when additional funds of the Corporation are legally available for the redemption of shares of Common Stock or Preferred Stock, such funds will immediately be used to redeem the shares which the Corporation has become obligated to redeem on any Redemption Date but which it has not redeemed.

(d) Prior to effecting any such redemption, the Corporation shall provide any holder of Common Stock or Preferred Stock to be redeemed with reasonable prior written notice of the reason giving rise to the Corporation s redemption right and, if requested to do so by such holder, the Corporation shall reasonably

- 4 -

cooperate with such affected holder in arranging another method to minimize or eliminate the reason giving rise to the Corporation s redemption right, including, but not limited to and not in any particular order of priority, preparing and filing waiver requests with the FCC, developing alternative ownership structures, assisting with a sale of such holders interest in the Corporation, amending the Corporation s Certificate of Incorporation and obtaining FCC approvals for such transaction.

(e) The provisions of Sections (E)(3)(a) through (E)(3)(d) of this Article IV shall not apply to Deutsche Telekom AG (the <u>Stockholder</u>) or any of its subsidiaries, any acquisition of shares of Common Stock or Preferred Stock by the Stockholder or any of its subsidiaries, or any ownership of such shares otherwise attributed to the Stockholder or any of its subsidiaries, and the Corporation shall not have the authority under Sections (E)(3)(a) of this Article IV to redeem any shares of Common Stock or Preferred Stock beneficially owned, directly or indirectly, by the Stockholder or any of its subsidiaries, in each case notwithstanding anything to the contrary therein. In the event that any waivers or approvals are required from the FCC in order for the Stockholder or any of its subsidiaries to acquire or hold Common Stock or Preferred Stock, the Stockholder and its subsidiaries shall cooperate to secure such waivers or approvals and abide by any conditions related to such waivers or approvals.

#### (F) Voting Rights.

1. <u>Common Stock</u>. The holders of Common Stock shall have the right to vote on every matter submitted to a vote of the holders of capital stock of the Corporation other than any matter on which only the holders of one or more other classes or series of capital stock of the Corporation are entitled to vote separately as a class.

2. <u>Preferred Stock</u>. Except as specifically limited herein, the holders of Preferred Stock shall have such voting rights as shall be set forth in any Certificate of Designation relating thereto.

### ARTICLE V

### POWER TO AMEND BYLAWS

(A) <u>Board of Directors</u>. Except as otherwise provided in this Fourth Amended and Restated Certificate of Incorporation (as it may be amended from time to time, this <u>Certificate of Incorporation</u>) and subject to any additional requirements expressly set forth in the Bylaws of the Corporation, the Board of Directors is expressly authorized, upon the affirmative vote of a majority of the directors then serving, to make, adopt, alter, amend, and repeal from time to time the Bylaws of the Corporation and make from time to time new Bylaws of the Corporation (subject to the right of the stockholders entitled to vote thereon to adopt, alter, amend, and repeal Bylaws made by the Board of Directors or to make new Bylaws).

(B) <u>Stockholders</u>. The stockholders of the Corporation may adopt, alter, amend, or repeal Bylaws made by the Board of Directors or make new Bylaws upon the affirmative vote of the holders of shares having a majority of the aggregate voting power of all of the outstanding shares of the Corporation s capital stock then entitled to vote thereon, subject to any additional requirements in the Bylaws of the Corporation.

### ARTICLE VI

#### **BOARD OF DIRECTORS**

#### (A) Number, Election and Term of Directors.

1. The number of directors constituting the entire Board of Directors shall be fixed by, or in the manner provided in, the Bylaws of the Corporation, or as provided in accordance with any Certificate of Designation. All

- 5 -

of the directors of the Corporation shall be of one class and shall be elected annually. Each director shall hold office until the next annual meeting of stockholders and, the foregoing notwithstanding, shall serve until his successor shall have been duly elected and qualified or until his earlier death, resignation, retirement, disqualification or removal.

2. At all times when the ratio, expressed as a percentage, of (i) the number of votes entitled to be cast generally in the election of directors (<u>Votes</u>) by the Common Stock and any class of capital stock or other securities of the Corporation other than the Common Stock that are entitled to vote generally in the election of directors (the <u>Voting Securities</u>) Beneficially Owned by the Stockholder to (ii) the aggregate Votes entitled to be cast by all then-outstanding Voting Securities (such ratio, the <u>Stockholder Voting Percentage</u>) is ten percent (10%) or more, the Stockholder shall have the right to designate a number of individuals to be nominees for election to the Board of Directors (<u>Stockholder Designees</u>) equal to the Stockholder Voting Percentage multiplied by the total number of directors of the Board of Directors that the Corporation would have if there were no vacancies, rounded to the nearest whole number; provided, that the number of directors on the Board of Directors who are also officers, employees, directors or affiliates of the Stockholder shall not in any event exceed a number equal to the Stockholder Voting Percentage multiplied by the total number of Directors that the Corporation would have if there were no vacancies, rounded to the nearest whole number greater than zero. In addition, the Corporation shall cause any committee of the Board of Directors to include in its membership a number of Stockholder Designees then serving as directors on the Board of Directors equal to the Stockholder Voting Percentage multiplied by the total number of members that such committee would have if there were no vacancies on such committee, rounded up to the nearest whole number, except to the extent that such membership would violate applicable securities laws or stock exchange or stock market rules; provided, however, that no committee may consist solely of directors who are also officers, employees, directors or affiliates of the Stockholder. If at any time the Stockholder Voting Percentage is less than ten percent (10%), the Stockholder shall promptly cause all of the Stockholder Designees then serving as Directors to resign from the Board, and all of the rights of the Stockholder set forth in this Section (A)(2) of Article VI shall forever terminate.

3. If at any time the number of Stockholder Designees then serving as directors or as members of any committee of the Board of Directors exceeds the number of Stockholder Designees the Stockholder is entitled to designate to the Board of Directors or any committee thereof pursuant to this Section (A) of Article VI, the Stockholder shall cause the number of Stockholder Designees then serving as directors or as members of such committee of the Board of Directors representing such excess to resign immediately as directors or committee members, as applicable.

4. In the event of any change in the authorized number of directors, each director then continuing to serve as such shall nevertheless continue as a director until the expiration of his current term, or his prior death, resignation, retirement, disqualification or removal. In the event of any increase in the authorized number of directors as a result of which the Stockholder shall be entitled to designate one or more additional Stockholder Designees based upon the increased size of the Board of Directors and the then Stockholder Voting Percentage, (i) the Stockholder shall be entitled promptly to designate such Stockholder Designees, and (ii) the Corporation shall cause the prompt appointment or election of such Stockholder Designee(s) as director(s). If at any time the Stockholder Voting Percentage is less than ten percent (10%), all of the rights of the Stockholder set forth in this Section (A)(4) of Article VI shall forever terminate.

5. The Company and the Stockholder shall use their reasonable best efforts to cause at least three of the directors to be considered independent under the rules of the Securities and Exchange Commission, the New York Stock Exchange and any other or additional exchange on which the securities of the Corporation are listed, including for purposes of Rule 10A-3 promulgated under the Securities Exchange Act of 1934, as amended (or any successor rule thereto).

- 6 -

(B) <u>Vacancies</u>. Any vacancies of the Board of Directors resulting from death, resignation, retirement, disqualification, removal from office or other reason may be filled as provided in the Bylaws of the Corporation; <u>provided</u> that upon any such vacancy resulting in any Stockholder Designee ceasing to serve as a director at a time when the Stockholder has the right under Section (A)(2) of Article VI to designate a replacement Stockholder Designee, (i) the Stockholder shall be entitled promptly to designate a replacement Stockholder Designee to fill such vacancy, and (ii) the Corporation shall cause the prompt appointment or election of such replacement Stockholder Designee as a director.

(C) Written Ballots not Required. Elections of directors need not be by written ballot unless the Bylaws of the Corporation shall so provide.

(D) <u>Removal of Directors</u>. Subject to the rights of the holders of any class or series of Preferred Stock then outstanding, any director may be removed from office at any time, with or without cause, by the affirmative vote of the holders of at least a majority of the voting power of all of the outstanding shares of the Corporation s capital stock entitled to elect such director, voting separately as a class, at a duly organized meeting of stockholders or by written consent; <u>provided</u> that no Stockholder Designee may be removed under this Section (D) of Article VI without the prior written consent of the Stockholder.

#### (E) Corporate Opportunity Matters.

1. Except as set forth in Section (E)(2) of this Article VI, to the extent permitted by the DGCL, if any non-employee director (or any of his or her affiliates) acquires knowledge of a potential transaction or matter which may be a corporate opportunity in the same or similar activity or line of business as the Corporation, the Corporation shall have no interest or expectancy in being offered by such non-employee director any opportunity to participate in such corporate opportunity, any such interest or expectancy being hereby renounced, so that, as a result of such renunciation and without limiting the scope of such renunciation, such person (a) shall have no duty to communicate or present such corporate opportunity to the Corporation and (b) shall have the right to hold any such corporate opportunity for its (and its officers , directors , agents , stockholders or affiliates ) own account or to recommend, sell, assign or transfer such corporate opportunity to any individual, corporation (including not-for-profit), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, governmental entity or other entity of any kind or nature (each, a <u>Person</u>) other than the Corporation; *provided, however*, that the foregoing shall not preclude or prevent the Corporation from pursuing any corporate opportunity that may be presented to it by any means.

2. Notwithstanding the provisions of Section (E)(1) of this Article VI, the Corporation does not renounce any interest or expectancy it may have in any corporate opportunity that is offered to any non-employee director, if such opportunity is expressly offered to such non-employee director (or his or her affiliates) solely in, and as a direct result of, his or her capacity as a director of the Corporation.

### ARTICLE VII

### MEETINGS OF STOCKHOLDERS; ACTION WITHOUT A MEETING

(A) <u>Meetings of Stockholders</u>. Meetings of stockholders may be held within or without the State of Delaware, as the Bylaws of the Corporation may provide. Special meetings of the stockholders of the Corporation (i) may be called, for any purpose or purposes, by the chairman of the Board of Directors or the chief executive officer and (ii) shall be called by the secretary of the Corporation at the request of (a) a majority of the Board of Directors or (b) for so long as the Stockholder Voting Percentage is twenty-five percent (25%) or greater, the holders of not less than thirty-three and one third percent (33-1/3%) of the voting power of all of the outstanding voting stock of the Corporation entitled to vote generally for the election of directors, which such request must be in writing, shall state the purpose or purposes of the proposed meeting (which shall be included in the notice of such meeting) and shall include all information required to be delivered pursuant to the notice

- 7 -

requirements set forth in the Bylaws of the Corporation in order for nominations or business, as applicable, to be properly brought before a meeting by a stockholder. If at any time the Stockholder Voting Percentage is less than twenty-five percent (25%), the stockholders ability to request a special meeting of stockholders pursuant to the immediately preceding sentence shall forever terminate. Each special meeting shall be held within a reasonable time after being called in accordance with the preceding sentence. Advance notice of stockholder nominations for the election of directors and of business to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws of the Corporation. The books of the Corporation may be kept (subject to any provision contained in the statutes) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or in the Bylaws of the Corporation.

(B) <u>Stockholder Action by Written Consent</u>. For so long as the Stockholder Voting Percentage is twenty-five percent (25%) or greater, any action required or permitted to be taken at any annual or special meeting of stockholders may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing setting forth the action so taken shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted. If at any time the Stockholder Voting Percentage is less than twenty-five percent (25%), the stockholders ability to act by written consent pursuant to the immediately preceding sentence shall forever terminate, and, thereafter, (i) no action shall be taken by the stockholders of the Corporation except at an annual or special meeting of such stockholders and (ii) the stockholders may not take action by written consent.

#### ARTICLE VIII

#### CONSENTS

(A) In addition to any other vote, consent or approval required by this Certificate of Incorporation, the Bylaws of the Corporation, or applicable law, for so long as the Stockholder Voting Percentage is thirty percent (30%) or greater, the Corporation shall not, and shall cause its subsidiaries not to, take or agree to take any of the following actions, in each case without the prior written consent of the Stockholder, which consent the Stockholder may withhold in its sole discretion:

1. create, incur, issue, assume or otherwise become liable for (including through a merger, acquisition or otherwise) or refinance or guarantee any Indebtedness (as defined in the stockholder s agreement entered into on [\_\_\_\_], 201[\_\_] by the Stockholder and the Corporation (the <u>Stockholder s Agreement</u> )) (excluding any Permitted Debt (as defined in the Stockholder s Agreement)) that would result in the Corporation and its subsidiaries, on a consolidated basis, having or being liable for Indebtedness (as defined in the Stockholder s Agreement) in an aggregate principal amount that would result in the Debt to Cash Flow Ratio (as defined in the Stockholder s Agreement) for the Corporation s most recently ended four full fiscal quarters for which financial statements are available to be greater than 5.25 to 1.0 on a pro forma basis as if the additional Indebtedness had been incurred at the beginning of such four-quarter period;

2. take any action or enter into any transaction that would reasonably be expected to result in a breach of or default under any credit agreement, indenture, note, or similar instrument or security to which the Stockholder or any of its affiliates is a party or is bound;

3. acquire (including by way of merger, recapitalization, reorganization, liquidation or dissolution) any business, debt or equity interests, operations or assets of any Person, or make any investment in or loan to any Person, in any single transaction or series of related transactions (excluding the acquisition of products and equipment in the ordinary course of business), for consideration in excess of \$1,000,000;

- 8 -

4. sell, lease, transfer, Encumber (as defined in the Stockholder s Agreement) (other than Permitted Liens (as defined in the Stockholder s Agreement)) or otherwise dispose of (including by way of merger, recapitalization, reorganization, liquidation or dissolution) any division, business, or operations of the Corporation or any of its subsidiaries, or any equity interests of the Corporation or any of its subsidiaries, in any single transaction or series of related transactions, for consideration in excess of \$1,000,000,000;

5. change the size of the Board of Directors;

6. issue any equity or equity-linked securities or other Voting Securities of the Corporation or any of its subsidiaries, in any single transaction or series of related transactions, (i) constituting ten percent (10%) or more of the then outstanding shares of Common Stock (other than grants of incentive awards to officers or employees of the Corporation or its subsidiaries that are approved by the Board or the applicable committee thereof or issuances of securities to the Corporation or any of its wholly owned Subsidiaries) or (ii) for the purpose of redeeming or purchasing any indebtedness of the Corporation held by the Stockholder or its affiliates;

7. subject to Section (E)(3)(e) of Article IV, (i) repurchase or redeem any equity (or equity-based) securities of the Corporation or any of its non-wholly owned subsidiaries, or (ii) make any extraordinary or in-kind dividend with respect to any of the equity (or equity-based) securities of the Corporation or any of its subsidiaries, other than a dividend on a pro rata basis with respect to all stockholders of the Corporation, or a dividend to the Corporation or any of its wholly owned Subsidiaries; or

8. hire, or terminate without cause, its Chief Executive Officer, or agree to do so.

(B) In addition to any other vote, consent or approval required by this Certificate of Incorporation, the Bylaws of the Corporation, or applicable law, for so long as the Stockholder Voting Percentage is five percent (5%) or greater, the Corporation shall not amend or seek to amend this Certificate of Incorporation, the Bylaws of the Corporation or the Stockholder s Agreement (including, for the avoidance of doubt, the creation of any shareholder rights plan or other amendment intended to limit the Stockholder s ownership or acquisition of securities of the Corporation) in any manner that could limit, restrict or adversely affect the Stockholder or its rights thereunder without the prior written consent of the Stockholder, which consent may be withheld in its sole discretion.

(C) Notwithstanding anything contained herein to the contrary, if at any time the Stockholder Voting Percentage is less than thirty percent (30%), all of the obligation of the Corporation and rights of the Stockholder set forth in Section (A) of this Article VIII shall forever terminate.

### ARTICLE IX

#### INDEMNIFICATION

(A) The Corporation shall, to the fullest extent permitted by the DGCL in effect on the date of the effectiveness of this Certificate of Incorporation, and to such greater extent as the DGCL may thereafter permit, indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding whether civil, criminal, administrative or investigative by reason of the fact that he is or was a director, manager, member, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, manager, member, officer, employee or agent of another corporation, partnership, limited liability company, joint venture, trust or other enterprise, against expenses (including attorneys fees), judgments, liabilities, losses, fines and amounts paid in settlement actually and reasonably incurred by him in connection with such action, suit or proceeding.

- 9 -

(B) To the extent that a director, manager, member, officer, employee or agent of the Corporation has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in Section (A) of this Article IX, or in defense of any claim, issue or matter therein, he shall be indemnified against expenses (including attorneys fees) actually and reasonably incurred by him in connection therewith.

(C) Any indemnification under Section (A) of this Article IX (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that indemnification of the director, officer, employee or agent is proper in the circumstances because he has met the applicable standard of conduct set forth in the DGCL. Such determination shall be made (i) by the Board of Directors by a majority vote of a quorum consisting of directors who were not parties to such action, suit or proceeding, or (ii) if such a quorum is not obtainable, or, even if obtainable, a quorum of disinterested directors so directs, by independent legal counsel in a written opinion, or (iii) by the stockholders of the Corporation.

(D) Expenses (including attorneys fees) incurred by an officer, director, a manager of a Corporation limited liability company, or a member of a management committee of a Corporation limited liability company, in defending any civil, criminal, administrative or investigative action, suit or proceeding shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director, officer, manager or member to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized in this Article IX. Such expenses (including attorneys fees) incurred by other employees and agents may be so paid upon such terms and conditions, if any, as the Board of Directors deems appropriate.

(E) The indemnification and advancement of expenses provided by, or granted pursuant to, the other sections of this Article IX shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under any law, Bylaw, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in an official capacity and as to action in another capacity while holding such office.

(F) The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, manager, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, manager, member, employee or agent of another corporation, partnership, limited liability company, joint venture, trust or other enterprise against any liability asserted against him and incurred by him in any such capacity, or arising out of his status as such, whether or not the Corporation would have the power to indemnify him against such liability under the provisions of Section 145 of the DGCL.

(G) For purposes of this Article IX, references to the Corporation shall include, in addition to the resulting corporation, any constituent corporation, partnership, limited liability company, or joint venturer or other enterprise (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, managers, members, employees or agents so that any person who is or was a director, officer, manager, member, employee or agent of such constituent corporation, partnership, limited liability company, or joint venturer or other enterprise, or is or was serving at the request of such constituent corporation, partnership, limited liability company, or joint venturer or other enterprise as a director, officer, manager, member, employee or agent of another corporation, partnership, limited liability company, joint venture, trust or other enterprise, shall stand in the same position under the provisions of this Article IX with respect to the resulting or surviving corporation, partnership, limited liability company, or joint venture corporation, partnership, limited as to the resulting or surviving corporation, partnership, limited liability company, or joint venture corporation, partnership, limited liability company, or surviving corporatio

(H) For purposes of this Article IX, (i) references to other enterprises shall include employee benefit plans, including without limitation, any plan of the Corporation which is governed by the Employee Retirement Income Security Act of 1974, as amended from time to time (collectively, <u>Employee Benefit Plans</u>),

- 10 -

(ii) references to fines shall include any excise taxes assessed on a person with respect to an Employee Benefit Plan, (iii) references to serving at the request of the Corporation shall include any service as a director, officer, employee or agent of the Corporation which imposes duties on, or involves service by, such director, officer, employee or agent with respect to any Employee Benefit Plan, its participants or beneficiaries, and (iv) a person who acted in good faith and in a manner he reasonably believed to be in the interest of the participants and beneficiaries of an Employee Benefit Plan shall be deemed to have acted in a manner not opposed to the best interests of the Corporation as referred to in this Article IX.

(I) The indemnification and advancement of expenses provided by, or granted pursuant to, this Article IX shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director, officer, manager, member, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a person.

(J) The provisions of this Article IX shall be deemed to be a contract between the Corporation and each director, officer, employee or agent who serves in such capacity at any time while this Article IX is in effect. Any repeal or modification of this Article IX shall be prospective only, and shall not adversely affect any rights or obligations then existing with respect to any state of facts then or theretofore existing or any action, suit or proceeding theretofore or thereafter brought based in whole or in part upon any such state of facts, whether such state of facts was then known or later known.

### ARTICLE X

### LIMITATION OF LIABILITY

A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director s duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the DGCL, as the same exists or hereafter may be amended or replaced, or (iv) for any transaction from which the director derived any improper personal benefit. If the DGCL is amended after the date of filing this Certificate of Incorporation to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation, in addition to the limitation on personal liability provided herein, shall be limited to the fullest extent permitted by the DGCL as so amended. Any repeal or modification of this Article X shall be prospective only, and shall not adversely affect any limitation on the personal liability of a director of the Corporation existing at the time of such repeal or modification.

### ARTICLE XI

#### AMENDMENTS TO CERTIFICATE OF INCORPORATION

The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation in the manner now or hereafter prescribed by law and subject to Section C of Article VIII, and all rights and powers conferred herein on stockholders, directors and officers are subject to this reserved power; *provided that*, notwithstanding the fact that a lesser percentage may be specified by the DGCL, the affirmative vote of the holders of record of outstanding shares representing at least seventy-five percent (75%) of the voting power of all of the shares of capital stock of the Corporation then entitled to vote generally in the election of the Board of Directors, voting together as a single class, shall be required to amend, alter, change, repeal, or adopt any provision or provisions inconsistent with, Article IX and this Article XI of this Certificate of Incorporation unless such amendment, alteration, change, repeal or adoption of any inconsistent provision or provisions is adopted or authorized by the Board of Directors by the affirmative vote of at least seventy-five percent (75%) of all of the members of the Board of Directors.

- 11 -

### ARTICLE XII

### GOVERNING LAW; FORUM FOR ADJUDICATION OF DISPUTES

This Certificate of Incorporation and the internal affairs of the Corporation shall be governed by and interpreted under the laws of the State of Delaware, excluding its conflict of laws principles. Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim of breach of a fiduciary duty owed by any director, officer or other employee of the Corporation to the Corporation or the Corporation s stockholders, (iii) any action asserting a claim arising pursuant to any provision of the Delaware Corporation Law, this Certificate of Incorporation or the Bylaws of the Corporation, or (iv) any other action asserting a claim arising under, in connection with, and governed by the internal affairs doctrine.

#### ARTICLE XIII

#### EFFECTIVE TIME

This Certificate of Incorporation shall become effective as of [Time] on [Date] (the <u>Effective Time</u>).

\*\*\*\*

- 12 -

IN WITNESS WHEREOF, the Corporation has caused this Fourth Amended and Restated Certificate of Incorporation to be executed by the [Title] of the Corporation on this [] day of [], 201[].

# **MetroPCS Communications, Inc.**

By: [Name] [Title]

- 13 -

### <u>Exhibit B</u>

#### FIFTH AMENDED AND RESTATED BYLAWS

#### OF T-MOBILE US, INC.

#### **PREAMBLE**

In accordance with power conferred to the board of directors (the <u>Board of Directors</u>) of T-Mobile US, Inc., a Delaware corporation (the <u>Corporation</u>), in the Fourth Amended and Restated Certificate of Incorporation (<u>Certificate of Incorporation</u>) of the Corporation, the Board of Directors approved and adopted these Fifth Amended and Restated Bylaws (these <u>Bylaws</u>), effective as of []. These Bylaws are subject to, and governed by, the General Corporation Law of the State of Delaware (the <u>Delaware Corporation Law</u>) and the Certificate of Incorporation. In the event of a direct conflict between the provisions of these Bylaws and the mandatory provisions of the Delaware Corporation Law or the provisions of the Certificate of Incorporation, as the case may be, will be controlling.

#### ARTICLE I

#### OFFICES

1. The registered office of the Corporation shall be the registered office named in the Certificate of Incorporation or such other place as shall be determined by the Board from time to time.

2. The Corporation may also have offices at such other places both within and without the State of Delaware as the Board of Directors may from time to time determine or the business of the Corporation may require.

#### ARTICLE II

#### MEETINGS OF STOCKHOLDERS

1. All meetings of the stockholders for the election of directors shall be held at such time and place either within or without the State of Delaware as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting. Meetings of stockholders for any other purpose may be held at such time and place, within or without the State of Delaware, as shall be stated in the notice of the meeting or in a duly executed waiver of notice thereof.

2. The annual meeting of the stockholders for the election of directors and the transaction of such other business as may properly be brought before the meeting shall be held at such date, time and place as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting.

3. Written notice of the annual meeting stating the place, date and hour of the meeting shall be given to each stockholder entitled to vote at such meeting not fewer than ten (10) nor more than sixty (60) days before the date of the meeting.

4. The officer of the Corporation who has charge of the stock ledger of the Corporation shall prepare and make, at least ten (10) days before every meeting of stockholders a complete list of the stockholders entitled to vote at the meeting of stockholders (<u>provided</u>, <u>however</u>, if the record date for determining the stockholders entitled to vote is less than ten days before the meeting, the list shall reflect the stockholders entitled to vote as of the tenth day before the meeting date), arranged in alphabetical order, and showing the address of each stockholder and the number of shares of each class of capital stock registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for a period of at least ten (10) days prior to the meeting, either at a place within the city

where the meeting is to be held, which place shall be specified in the notice of the meeting, or, if not so specified, at the place where the meeting is to be held. The list shall also be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. Except as provided by applicable law, the stock ledger of the Corporation shall be the only evidence as to who are the stockholders entitled to examine the stock ledger and the list of stockholders entitled to vote in person or by proxy at any meeting of stockholders.

5. Special meetings of the stockholders, for any purpose or purposes, may be called as set forth in the Certificate of Incorporation.

6. Written notice of a special meeting stating the date, time and place of the meeting and the purpose or purposes for which the meeting is called, shall be given not fewer than ten (10) nor more than sixty (60) days before the date of the meeting, to each stockholder entitled to vote at such meeting.

7. Business transacted at any special meeting of stockholders shall be limited to the purposes stated in the notice.

8. At each meeting of the stockholders, the stockholders holding issued and outstanding capital stock of the Corporation having not less than a majority of the votes of the capital stock of the Corporation entitled to vote thereat, present in person or represented by proxy, shall constitute a quorum for the transaction of business except as otherwise provided by statute or by the Certificate of Incorporation. If, however, such quorum shall not be present or represented at any meeting of the stockholders, the stockholders entitled to vote thereat, present in person or represented by proxy, shall have the power to adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present or represented. A quorum, once established, shall not be broken by the subsequent withdrawal of enough votes to leave less than a quorum. At such adjourned meeting at which a quorum shall be present or represented any business may be transacted which might have been transacted at the meeting as originally notified. If the adjournment is for more than thirty (30) days, or, if after the adjournment a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting.

9. When a quorum is present at any meeting, unless otherwise required by applicable law, the Certificate of Incorporation, or these Bylaws, the election of directors and any advisory vote on the frequency of stockholder votes related to the compensation of executives required by Section 14A(a)(2) of the Securities Exchange Act of 1934, as amended (Exchange Act ), shall be decided by a plurality of the votes cast at a meeting in person or by proxy by the holders of stock entitled to vote therein. When a quorum is present at any meeting, unless otherwise required by applicable law, the Certificate of Incorporation, or these Bylaws, any matter, other than the election of directors and an advisory vote on the frequency of stockholder votes related to the compensation of executives required by Section 14A(a)(2) of the Exchange Act, brought before any meeting of stockholders shall be decided by the vote of the holders of a majority of the votes cast in person or by proxy in favor of such action by the holders of stock entitled to vote therein. For the avoidance of doubt, abstentions and, except as may be permitted pursuant to the rules of any exchange where the securities of the Corporation may be listed, broker non-votes, will not be counted as votes cast for such purposes.

10. Unless otherwise provided by applicable law, in the Certificate of Incorporation, or in these Bylaws, each stockholder shall at every meeting of the stockholders be entitled to one vote in person or by proxy for each share of the capital stock having voting power held by such stockholder, but no proxy shall be voted on after three years from its date, unless the proxy provides for a longer period. Such proxy shall be filed with the Secretary before or at the time of the meeting. A proxy shall be irrevocable if it states that it is irrevocable and if, and only as long as, it is coupled with an interest sufficient in law to support an irrevocable power. A stockholder may revoke any proxy which is not irrevocable by attending the meeting and voting in person or by delivering to the secretary of the Corporation a revocation of the proxy or a new proxy bearing a later date. Voting at meetings of stockholders need not be by written ballot.

- 2 -

11. The Board of Directors of the Corporation may adopt by resolution such rules and regulations for the conduct of the meeting of the stockholders as it shall deem appropriate. At every meeting of stockholders, the chief executive officer, or in his or her absence or inability to act, the secretary, or, in his or her absence or inability to act, the person whom the chief executive officer shall appoint, shall act as chairman of, and preside at, the meeting. The secretary or, in his or her absence or inability act, the person whom the chairman of the meeting shall appoint as secretary of the meeting, shall act as secretary of the meeting and keep the minutes thereof. Except to the extent inconsistent with such rules and regulations as adopted by the Board of Directors, the chairman of any meeting of the stockholders shall have the right and authority to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairman, are appropriate for the proper conduct of the meeting, without limitation, the following: (a) order of business for the meeting; (b) the determination of when the polls shall open and close for any given matter to be voted on at the meeting; (c) rules and procedures for maintaining order at the meeting and the safety of those present; (d) limitations on attendance at or participation in the meeting to stockholders of record of the Corporation, their duly authorized and consituted proxies or such other persons as the chairman of the meeting shall determine; (e) restrictions on entry to the meeting after the time fixed for the commencement thereof; (f) limitations on the time allotted to questions or comments by participants; and (g) policies and procedures with respect to the adjournment of such meeting.

12. Unless otherwise provided in the Certificate of Incorporation, the provisions of this Section 12 shall apply to the nominations of directors to the Board of Directors. Nominations of persons for election to the Board of Directors may be made in advance of any annual meeting of stockholders or any special meeting of stockholders at which directors are to be elected as provided in the notice of meeting delivered in accordance with Sections 3 or 6 of this Article II, respectively. Nominations for election to the Board of Directors must be made by the Board of Directors or by any Eligible Stockholder (as defined below). Nominations, other than those made by the Board of Directors of the Corporation, must be preceded by notification in writing for each nominee received by the secretary of the Corporation at the executive offices of the Corporation (i) in the case of an annual meeting, not later than the close of business on the ninetieth (90<sup>th</sup>) calendar day nor earlier than the close of business on the one hundred twentieth (120th) calendar day prior to the first anniversary of the preceding year s annual meeting, regardless of whether the party or parties seeking to make the nominations are seeking to include the nominees in management s proxy materials or in their own or other proxy materials; provided, however, that in the event that the date of the annual meeting is more than thirty (30) calendar days before or more than sixty (60) calendar days after such anniversary date, notification must be received by the secretary of the Corporation not earlier than the close of business on the one hundred twentieth (120th) calendar day prior to such annual meeting but not later than the close of business on the latter of the ninetieth (90th) calendar day prior to such annual meeting or the tenth (10th) calendar day following the calendar day on which public announcement of the date of such meeting is first made by the Corporation, and (ii) in the case of a special meeting, not later than the close of business on the sixtieth (60<sup>th</sup>) calendar day nor earlier than the close of business on the ninetieth (90<sup>th</sup>) calendar day prior to the date of such special meeting, or if the first public announcement of the date of such special meeting is less than seventy (70) days prior to the date of such special meeting, the tenth (10<sup>th</sup>) day following the day on which public announcement is first made of the date of the special meeting and, if applicable, of the nominees proposed by the Board of Directors to be elected at such meeting. Such notification shall contain the following:

(a) the written consent of each proposed nominee to serve as a director if so elected;

(b) the following information as to each proposed nominee:

(1) the name, age, citizenship, residence address, and business address of each proposed nominee;

(2) the principal occupation or employment, and the name, type of business and address of the Corporation or other organization in which such employment is carried on, of such proposed nominee and of each such person nominating such proposed nominee;

(3) the qualifications of such proposed nominee to serve as a director of the Corporation;

- 3 -

(4) the amount of stock of the Corporation owned of record and beneficially, either directly or indirectly, by each proposed nominee;

(5) a description of any arrangement or understanding of each proposed nominee and of each person proposing such nomination with each other or any other person regarding future employment or any future transaction to which the Corporation will or may be a party; and

(6) all information required by the Corporation s director questionnaire then in use by the Corporation for its directors and officers, a copy of which shall be available at the offices of the Corporation; and

(c) the following information with respect to the stockholder giving the notice and the beneficial owner, if any, on whose behalf the nomination is made, and as to each person, acting alone or in conjunction with one or more other persons as a partnership, limited partnership, syndicate or other group, who participates or is expected to participate in making such nomination or in organizing, directing or financing such nomination or solicitation of proxies to vote for the nominee:

(1) the name and address of such stockholder, as they appear on the Corporation s books, and such beneficial owner;

(2) the class and number of shares of the Corporation which are owned of record and beneficially by such stockholder and such beneficial owner;

(3) the voting rights of such stockholder; and

(4) the hedging and derivative positions of such stockholder, if any, in the Corporation s capital stock.

The term Eligible Stockholder means a stockholder of the Corporation (i) who is a stockholder of record on the date of the giving of the notice by such stockholder provided for in this Section 12 or Section 13 (as the case may be) and on the record date for the determination of stockholders entitled to vote at the applicable annual or special meeting of the stockholders, (ii) who is entitled to vote on the business proposed in such notice to be conducted at an annual meeting of the stockholders (in the case of Section 13) or for the election of directors to be elected at an annual or special meeting of stockholders (in the case of this Section 12) and (iii) who complies with the applicable procedures set forth in this Section 13 (as the case may be).

Public announcement of a stockholder meeting shall be deemed to occur upon disclosure of the date of such meeting in a press release reported by the Dow Jones News Service, Associated Press or comparable national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the Exchange Act.

13. At any meeting of the stockholders, only such business shall be conducted as shall have been brought before the meeting (a) pursuant to the Corporation s notice of meeting, (b) by or at the direction of the Board of Directors or (c) by any Eligible Stockholder who complies with the notice procedures set forth in these Bylaws. The procedures referred to in clause (c) of the immediately preceding sentence and described in the remainder of this Section 13 shall be the exclusive means for a stockholder to submit business (other than stockholder proposals properly submitted in accordance with Rule 14a-8 under the Exchange Act and included in the Corporation s notice of meeting) to be considered or acted upon at an annual meeting of stockholders. To be properly brought before a special meeting of the stockholders, business must be specified in the notice of meeting (or any supplement thereto), and stockholders shall not be entitled to submit business to be considered or acted upon at any special meeting except in accordance with any applicable procedures for calling a special meeting as set forth in the Certificate of Incorporation. For the avoidance of doubt, the nomination of directors and the election of directors are deemed separate and distinct items of business, and candidates eligible for election to the Board of Directors at any meeting will be limited to those candidates who have been nominated in accordance with the provisions of Section 12.

For business to be properly brought before an annual meeting by a stockholder pursuant to clause (c) of this Section 13, the stockholder must have given timely notice thereof in writing to the secretary of the Corporation and such business must otherwise be a proper matter for stockholder action. To be timely, a stockholder s notice must be delivered to or mailed and received at the principal executive offices of the Corporation (i) in the case of an annual meeting, not later than the close of business on the ninetieth (90<sup>th</sup>) calendar day nor earlier than the close of business on the one hundred twentieth (120th) calendar day prior to the first anniversary of the preceding year s annual meeting; provided, however, that in the event that the date of the annual meeting is more than thirty (30) calendar days before or more than sixty (60) calendar days after such anniversary date, notice to the Corporation to be timely must be so delivered not earlier than the close of business on the one hundred twentieth (120th) calendar day prior to such annual meeting but not later than the close of business on the latter of the ninetieth (90th) calendar day prior to such annual meeting or the tenth (10<sup>th</sup>) calendar day following the calendar day on which public announcement of the date of such meeting is first made by the Corporation, and (ii) in the case of a special meeting, not later than the close of business on the ninetieth (90<sup>th</sup>) calendar day nor earlier than the close of business on the one hundred twentieth (120<sup>th</sup>) calendar day prior to the date of such special meeting, or if the first public announcement of the date of such special meeting is less than one hundred (100) days prior to the date of such special meeting, the tenth (10<sup>th</sup>) day following the day on which public announcement is first made of the date of the special meeting. A stockholder s notice to the secretary shall set forth as to each matter the stockholder proposes to bring before the meeting (i) a precise description of the business desired to be brought before the meeting and the reasons for conducting such business at the meeting, (ii) the name and address, as they appear on the Corporation s books, of the stockholder proposing such business, and the name and address of the beneficial owner, if any, on whose behalf the proposal is made, (iii) the class and number of shares of capital stock of the Corporation which are owned of record and beneficially and of record by such stockholder of record and by the beneficial owner, if any, on whose behalf of the proposal is made, (iv) any material interest of such stockholder of record and the beneficial owner, if any, on whose behalf the proposal is made in such business, (v) the voting rights of such stockholder; and (vi) the hedging and derivative positions of such stockholder, if any, in the Corporation s stock. The preceding requirements must be met independently for each item of business that any stockholder proposes to bring before a meeting, regardless of whether the stockholder seeks to include the proposal in management s proxy materials or in their own or other proxy materials.

Notwithstanding anything in these Bylaws to the contrary, no business shall be conducted at a meeting except in accordance with the procedures set forth in this Section 13. The presiding officer of the meeting shall, if the facts warrant, determine and declare to the meeting that business was not properly brought before the meeting and in accordance with the procedures prescribed by this Section 13, and if such person should so determine, such person shall so declare to the meeting any such business not properly brought before the meeting shall not be transacted. Notwithstanding the foregoing provisions of this Section 13, a stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 13.

14. Failure to receive notice of any meeting shall not invalidate the meeting.

15. The Board of Directors, in advance of any meeting of stockholders, may, and shall if required by law, appoint one or more inspectors, who may be employees of the Corporation, to act at the meeting or any adjournment thereof and make a written report thereof. The Board of Directors may designate one or more persons as alternate inspectors to replace any inspector who fails to act. If no inspector or alternate is able to act at a meeting, the chairman of the meeting shall appoint one or more inspectors to act at the meeting. Each inspector, before entering upon the discharge of his or her duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of his or her ability. The inspectors shall (a) ascertain the number of shares outstanding and the voting power of each, (b) determine the shares represented at the meeting, the existence of a quorum and the validity of proxies and ballots, (c) count all votes and ballots, (d) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors and (e) certify their determination of the number of shares represented at the

- 5 -

meeting and their count of all votes and ballots. The inspectors may appoint or retain other persons or entities to assist the inspectors in the performance of their duties. Unless otherwise provided by the Board of Directors, the date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at a meeting shall be determined by the person presiding at the meeting and shall be announced at the meeting. No ballot, proxies, votes or any revocation thereof or change thereto, shall be accepted by the inspectors after the closing of the polls unless the Court of Chancery of the State of Delaware upon application by a stockholder shall determine otherwise. In determining the validity and counting of proxies and ballots cast at any meeting of stockholders, the inspectors may consider such information as is permitted by applicable law. No person who is a candidate for office at an election may serve as an inspector at such election.

### ARTICLE III

#### DIRECTORS

1. Except as provided in the Certificate of Incorporation, the number of directors which shall constitute the whole Board of Directors shall be determined by resolution of the Board of Directors; provided, however, that no decrease in the number of directors shall have the effect of shortening the term of an incumbent director. Except as provided in Section 2 of this Article and the Certificate of Incorporation, the directors shall be elected at the annual meeting of the stockholders and each director elected shall hold office until his successor is elected and qualified, unless he shall resign, die, become disqualified or disabled, or otherwise be removed. Directors need not be stockholders.

2. Except as provided in the Certificate of Incorporation, vacancies and newly created directorships resulting from any increase in the authorized number of directors may be filled by a majority of the votes of the directors then in office, though less than a quorum, or by a sole remaining director. The term of a director elected to fill a newly created directorship or other vacancy shall expire at the next annual election of directors, and such director shall hold office until his successor is duly elected and shall qualify, or until his earlier death, resignation, retirement, disqualification or removal. If there are no directors in office, then an election of directors may be held in the manner provided by applicable law.

3. The business and affairs of the Corporation shall be managed by or under the direction of its Board of Directors which may exercise all such powers of the Corporation and do all such lawful acts and things as are not by statute or by the Certificate of Incorporation or by these Bylaws directed or required to be exercised or done by the stockholders or by specific stockholders.

#### MEETINGS OF THE BOARD OF DIRECTORS

4. The Board of Directors of the Corporation may hold meetings, both regular and special, either within or without the State of Delaware. The Board of Directors may adopt such rules and procedures, not inconsistent with the Certificate of Incorporation, these Bylaws or applicable law, as it may deem proper for the conduct of its meetings and the management of the Corporation.

5. The first meeting of each newly elected Board of Directors shall be held at such time and place as shall be specified in a notice given as hereinafter provided for special meetings of the Board of Directors, or as shall be specified in a written waiver signed by all of the directors.

6. Regular meetings of the Board of Directors may be held without notice at such time and at such place as shall from time to time be determined by the Board of Directors.

7. Special meetings of the Board of Directors may be called by the chairman of the Board of Directors, the vice chairman of the Board of Directors, or the chief executive officer on two (2) days prior written notice to

- 6 -

each director by mail or forty-eight (48) hours prior notice to each director either personally or by facsimile, telegram or electronic mail; special meetings shall be called by the chairman of the Board of Directors, the vice chairman of the Board of Directors, the chief executive officer, or secretary of the Corporation in like manner and on like notice on the written request of a majority of the directors unless the Board of Directors consists of only one director, in which case special meetings shall be called by the chairman of the Board of Directors, the vice chairman of the sole director.

8. At all meetings of the Board of Directors a majority of all directors then serving in office shall constitute a quorum for the transaction of business and the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the Board of Directors, except as may be otherwise specifically provided by applicable law, the Certificate of Incorporation or these Bylaws. A quorum, once established, shall not be broken by the subsequent withdrawal or departure of directors to leave less than a quorum. If a quorum shall not be present at any meeting of the Board of Directors, the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present. Any business may be transacted at an adjourned meeting that might have been transacted at the meeting as originally called.

9. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if all members of the Board of Directors or committee, as the case may be, consent thereto in writing, and the writing or writings are filed with the minutes of proceedings of the Board of Directors or committee.

10. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, members of the Board of Directors, or any committee designated by the Board of Directors, may participate in a meeting of the Board of Directors, or any committee, by means of conference call, video conference, webcast or other means by which all persons participating in the meeting can hear or communicate with each other, and such participation in a meeting shall constitute presence in person at the meeting.

11. At each meeting of the Board of Directors, the chairman of the Board of Directors, or, in his or her absence, the vice chairman of the Board of Directors, and in his or her absence, the presiding director, or in his absence, a director in attendance at such meeting selected by the Board of Directors shall preside. The secretary shall act as secretary at each meeting of the Board of Directors. If the secretary is absent from any meeting of the Board of Directors, and all assistant secretary shall perform the duties of secretary at such meeting; and in the absence from any such meeting of the secretary and all assistant secretaries, the person presiding at the meeting may appoint any person to act as secretary of the meeting.

#### COMMITTEES OF THE BOARD OF DIRECTORS

12. The Board of Directors may, by resolution passed by a majority of all of the directors, designate one or more committees, each committee to consist of one or more of the directors of the Corporation, subject to the provisions of the Certificate of Incorporation.

Any such committee, to the extent provided in the resolution of the Board of Directors, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers which may require it; but no such committee shall have the power or authority in reference to submitting any actions to the stockholders which require stockholder approval (other than the election or removal of directors), amending the Certificate of Incorporation, adopting an agreement of merger or consolidation, recommending to the stockholders the sale, lease or exchange of all or substantially all of the Corporation s property and assets, recommending to the stockholders a dissolution of the Corporation or a revocation of a dissolution, amending these Bylaws; and, unless the resolution or the Certificate of Incorporation expressly so provide, no such

- 7 -

committee shall have the power or authority to declare a dividend or to authorize the issuance of capital stock. Such committee or committees shall have such name or names as may be determined from time to time by resolution adopted by the Board of Directors. Unless the Board of Directors or the applicable committee charter provides otherwise, at all meetings of such committee, a majority of the then authorized members of the committee shall constitute a quorum for the transaction of business, and the vote of a majority of the members of the committee present at any meeting at which there is a quorum shall be the act of the committee. A quorum, once established, shall not be broken by the subsequent withdrawal or departure of directors to leave less than a quorum. Each committee shall keep regular minutes of its meetings. Unless the Board of Directors or the committee charter provides otherwise and subject to the provisions of the Certificate of Incorporation, each committee designated by the Board of Directors may make, alter and repeal rules and procedures for the conduct of its business. In the absence of such rules and procedures each committee shall conduct its business in the same manner as the Board of Directors conducts its business pursuant to this Article III.

13. Each committee shall keep regular minutes of its meetings and report the same to the Board of Directors when required.

#### **COMPENSATION OF DIRECTORS**

14. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, the Board of Directors or designated committee thereof shall have the authority to fix the compensation of directors. The directors may be paid their expenses, if any, of attendance at each meeting of the Board of Directors and may be paid a fixed sum for attendance at each meeting of the Board of Directors or a stated salary as director payable in cash, stock, stock options, or other compensation or a combination thereof. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed like compensation and reimbursement of expenses for attending committee meetings.

#### **REMOVAL OF DIRECTORS**

15. Unless otherwise restricted or permitted by the Certificate of Incorporation or these Bylaws, any director or the entire Board of Directors may be removed, with or without cause, by the holders of a majority of shares entitled to vote at an election of directors at any meeting of stockholders or any adjournment thereof or by action taken by the stockholders by written consent without a meeting.

#### **RESIGNATIONS OF DIRECTORS OR COMMITTEE MEMBERS**

16. Any director or member of a committee may resign at any time. Such resignation shall be made in writing and shall take effect at the time specified therein, or if no time be specified, at the time of its receipt by the chairman of the Board of Directors or chief executive officer and the secretary of the Corporation. The acceptance of a resignation shall not be necessary to make it effective, unless expressly so provided in the resignation.

### FCC ELIGIBILITY DIRECTORS

17. The Corporation, to the extent necessary to comply with FCC reporting or disclosure requirements, shall obtain from each existing and proposed director information relating to the citizenship and foreign affiliations, if any, of the director and such other information regarding the director as is reasonable to ensure the Corporation is in compliance with applicable law.

- 8 -

### ARTICLE IV

#### NOTICES

1. Whenever, under the provisions of applicable law, the Certificate of Incorporation, or these Bylaws, notice is required to be given to any director or stockholder, it shall not be construed to mean personal notice, but to the extent permitted by law such notice may be given in writing, personally, by overnight mail, telegram, facsimile, or electronic mail or by mail, addressed to such director or stockholder, at his address as it appears on the records of the Corporation, with postage thereon prepaid, and such notice shall be deemed to be given when by United States mail, at the time when the same shall be deposited in the United States mail, and upon delivery if personally delivered, sent via telegram, overnight mail, facsimile, or electronic mail.

2. Whenever any notice is required to be given under the provisions of applicable law, the Certificate of Incorporation, or these Bylaws, a waiver thereof in writing, signed by the person or persons entitled to said notice, whether before or after the time stated therein, shall be deemed equivalent thereto. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting at the beginning of the meeting to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the stockholders, directors, or members of a committee need be specified in any written waiver of notice unless so required by the Certificate of Incorporation or these Bylaws.

#### ARTICLE V

#### **OFFICERS**

1. Subject to the restrictions of the Certificate of Incorporation, the officers of the Corporation shall be chosen by the Board of Directors and shall include a chief executive officer, president, chief financial officer, treasurer, and secretary and may include such additional officers as may from time to time be authorized by these Bylaws or the Board. Subject to the restrictions of the Certificate of Incorporation, the Board of Directors, and a presiding director of the Board of Directors. Subject to the restrictions of the Certificate of Incorporation, the Board of Directors, and a presiding director of the Board of Directors. Subject to the restrictions of the Certificate of Incorporation, the Board of Directors may also choose one or more officers, vice-presidents, a chief operating officer, and or more assistant secretaries and assistant treasurers. Any number of offices may be held by the same person, unless the Certificate of Incorporation or these Bylaws otherwise provide.

2. Subject to the restrictions of the Certificate of Incorporation, the Board of Directors, at its first meeting after each annual meeting of stockholders, shall choose a chief executive officer, president, chief financial officer, treasurer and secretary and may include such additional officers as may from time to time be authorized by these Bylaws or the Board of Directors.

3. Subject to the restrictions of the Certificate of Incorporation, the Board of Directors may appoint such other officers and agents as it shall deem necessary or appropriate who shall hold their offices for such terms and shall exercise such powers and perform such duties as shall be determined from time to time by the Board of Directors.

4. The salaries of all officers and agents of the Corporation shall be fixed by the Board of Directors.

5. Each officer of the Corporation shall hold office until such officer s successor is elected or appointed by the Board of Directors and shall qualify or until such officer s death, resignation or removal in the manner hereinafter provided. Any officer may resign at any time. Such resignation shall be made in writing and shall take effect at the time specified therein, or if no time be specified, at the time of its receipt by the chief executive officer or secretary. The acceptance of a resignation shall not be necessary to make it effective, unless expressly

- 9 -

so provided in the resignation. Any officer elected or appointed by the Board of Directors may be removed at any time by the affirmative vote of a majority of directors comprising the whole Board of Directors. Subject to the restrictions of the Certificate of Incorporation, any vacancy occurring in any office of the Corporation shall be filled by the Board of Directors.

### THE CHAIRMAN OF THE BOARD OF DIRECTORS

6. The chairman of the Board of Directors, if any, shall preside at all meetings of the Board of Directors and of the stockholders at which he shall be present. He shall have and may exercise such powers as are, from time to time, assigned to him by the Board of Directors and as may be provided by law. The Board of Directors may elect a presiding director of the Board of Directors and may elect one or more vice chairmen of the Board of Directors by majority vote.

7. In the absence of the chairman of the Board of Directors, the vice chairman of the Board of Directors, if any, or the presiding director of the Board of Directors (if the chairman of the Board of Directors and the vice chairman of the Board of Directors, if any, is not present) shall preside at all meetings of the Board of Directors and of the stockholders at which he shall be present. He shall have and may exercise such powers as are, from time to lime, assigned to him by the Board of Directors and as may be provided by law.

### THE CHIEF EXECUTIVE OFFICER, PRESIDENT, CHIEF OPERATING OFFICER

#### AND VICE-PRESIDENTS

8. The chief executive officer shall preside at all meetings of the stockholders; he shall have general and active management of the business of the Corporation and shall see that all orders and resolutions of the Board of Directors are carried into effect.

9. The president shall be the chief administrative officer of the Corporation and, in the absence of the appointment of a separate chief operating officer or in the event of his inability or refusal to act, the president shall perform the duties of the chief operating officer of the Corporation, and when so acting, shall have all the powers of, and be subject to, all the restrictions upon the chief operating officer. The president shall perform such duties as from time to time may be assigned to him by the Board of Directors or by the chief executive officer. In the absence of the chief executive officer or in the event of his inability or refusal to act, the president shall perform the duties of the chief executive officer, and when so acting, shall have all the powers of, and be subject to, all the restrictions upon the chief executive officer.

10. The chief operating officer shall be responsible for the day-to-day operations of the Corporation. The chief operating officer shall perform such duties as from time to time may be assigned to him by the Board of Directors or by the chief executive officer. In the absence of the president or in the event of his inability or refusal to act, the chief operating officer shall perform the duties of the president, and when so acting, shall have all the powers of, and be subject to, all the restrictions upon the president. In the absence of the chief executive officer and the president or in the event of their inability or refusal to act, the chief operating officer shall perform the duties of the chief executive officer and the president, and when so acting, shall have all the powers of and be subject to all the restrictions upon the president and chief executive officer.

11. The chief executive officer, the president, the chief operating officer, or any vice president shall execute bonds, mortgages and other contracts requiring a seal, under the seal of the Corporation, except where required or permitted by law to be otherwise signed and executed and except where the signing and execution thereof shall be expressly delegated by the Board of Directors to some other officer or agent of the Corporation.

12. In the absence of the chief executive officer, president and chief operating officer, or in the event of their inability or refusal to act, the vice-president, if any (or in the event there be more than one vice-president, the

- 10 -

vice-presidents in the order designated by the Board of Directors, or in the absence of any designation, then in the order of their election), shall perform the duties of the chief executive officer, president, and chief operating officer, and when so acting, shall have all the powers of, and be subject to, all the restrictions upon the chief executive officer, president, and chief operating officer. The vice-presidents shall perform such other duties and have such other powers as the Board of Directors, or the chief executive officer may from time to time prescribe.

#### **CHIEF FINANCIAL OFFICER**

13. The chief financial officer of the Corporation shall have responsibility for the general executive charge, management and control of the financial affairs and business of the Corporation and, jointly with the treasurer of the Corporation, shall have custody and control of all the funds and securities of the Corporation, and he shall have such other powers and duties as from time to time may be designated in these Bylaws or assigned to him by the Board of Directors. He shall perform all acts incident to the position of chief financial officer, subject to the control of the chief executive officer and the Board of Directors.

#### THE SECRETARY AND ASSISTANT SECRETARIES

14. The secretary shall attend all meetings of the Board of Directors, all meetings of committees of the Board of Directors, and all meetings of the stockholders and record all the proceedings of the meetings of the Corporation and of the Board of Directors in a book to be kept for that purpose and shall perform like duties for the standing committees when required. He shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board of Directors, and shall perform such other duties as may be prescribed by the Board of Directors or the chief executive officer. He shall have custody of the corporate seal of the Corporation and he, or an assistant secretary, shall have authority to affix the same to any instrument requiring it, and when so affixed, it may be attested by his signature or by the signature of such assistant secretary. The Board of Directors may give general authority to any other officer to affix the seal of the Corporation and to attest the affixing by his signature.

15. The assistant secretary, or if there be more than one, the assistant secretaries in the order determined by the Board of Directors (or if there be no such determination, then in the order of their election) shall, in the absence of the secretary or in the event of his inability or refusal to act, perform the duties and exercise the powers of the secretary and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

#### THE TREASURER AND ASSISTANT TREASURERS

16. The treasurer, jointly with the chief financial officer, shall have the custody of the corporate funds and securities and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and shall deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board of Directors. The treasurer shall disburse the funds of the Corporation as may be ordered by the Board of Directors, taking proper vouchers for such disbursements, and shall render to the chief executive officer and the Board of Directors, at its regular meetings, or when the Board of Directors so requires, an account of all transactions as treasurer and of the financial condition of the Corporation.

17. The assistant treasurer, or if there shall be more than one, the assistant treasurers in the order determined by the Board of Directors (or if there be no such determination, then in the order of their election) shall, in the absence of the treasurer or in the event of the treasurer s inability or refusal to act, perform the duties and exercise the powers of the treasurer and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

- 11 -

### ARTICLE VI

### CERTIFICATE OF STOCK

1. Upon written request, every holder of capital stock in the Corporation shall be entitled to have a certificate, signed by, or in the name of the Corporation by, the chairman or vice-chairman of the Board of Directors, or the president or a vice-president, and by the treasurer or an assistant treasurer, or the secretary or an assistant secretary of the Corporation, certifying the number of shares owned by him in the Corporation; provided that the Board of Directors may provide by resolution or resolutions that some or all of any class or series of capital stock shall be uncertificated shares that may be evidenced by a book-entry system maintained by the registrar of such stock. If shares are represented by certificates such certificates shall be in a form approved by the Board of Directors.

If the Corporation shall be authorized to issue more than one class of stock or more than one series of any class, the powers, designations, and preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or back of the certificate which the Corporation shall issue to represent such class or series of stock; provided that, except as otherwise provided in section 202 of the Delaware Corporation Law, in lieu of the foregoing requirements, there may be set forth on the face or back of the certificate which the Corporation shall issue to represent such class or series of stock, a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, and preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

2. Any of or all the signatures on the certificate may be facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if he were such officer, transfer agent or registrar at the date of issue.

### LOST CERTIFICATES

3. The Board of Directors may direct a new certificate or certificates to be issued in place of any certificate or certificates theretofore issued by the Corporation alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate of stock to be lost, stolen or destroyed. When authorizing such issue of a new certificate or certificates or uncertificated shares, the Board of Directors may, in its discretion and as a condition precedent to the issuance thereof, require the owner of such lost, stolen or destroyed certificate or certificates, or his legal representative, to advertise the same in such manner as it shall require and/or to give the Corporation a bond in such sum as it may direct as indemnity against any claim that may be made against the Corporation with respect to the certificate alleged to have been lost, stolen or destroyed.

4. The Board of Directors may appoint, or authorize any officer or officers to appoint, one or more transfer agents and one or more registrars.

#### TRANSFER OF STOCK

5. Stock of the Corporation shall be transferable in the manner prescribed by applicable law and in these Bylaws. Transfers of stock shall be made on the books of the Corporation only by the person named as the holder thereof on the stock records of the Corporation, by such person s attorney lawfully constituted in writing, and in the case of shares represented by a certificate upon the surrender of the certificate thereof, which shall be cancelled before a new certificate or uncertificated shares shall be issued. No transfer of stock shall be valid as against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by

- 12 -

an entry showing from and to whom transferred. Subject to Section 155 of the Delaware Corporation Law, to the extent designated by the president or any vice president or the treasurer of the Corporation, the Corporation may recognize the transfer of fractional uncertificated shares, but shall not otherwise be required to recognize the transfer of fractional shares. Upon surrender to the Corporation or the transfer agent of the Corporation of a certificate for shares duly endorsed or accompanied by proper evidence of succession, assignation or authority to transfer, it shall be the duty of the Corporation to issue a new certificate to the person entitled thereto, cancel the old certificate and record the transaction upon its books.

### FIXING RECORD DATE

6. In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix, in advance, a record date, which shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting nor more than sixty (60) days prior to any other action. If the Board of Directors so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board of Directors determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.

7. Subject to Section 213 of the Delaware Corporation Law, the record date for a determination of the stockholders entitled to consent to corporate action in writing without a meeting shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation by delivery to its registered office in Delaware (by hand or by certified or registered mail, return receipt requested), its principal place of business or to any officer or agent of the Corporation having custody of the book in which proceedings of meetings of stockholders are recorded.

#### **REGISTERED STOCKHOLDERS**

8. The Corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, and to vote as such owner, and to hold liable for calls and assessments a person registered on its books as the owner of shares and shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of the State of Delaware.

### ARTICLE VII

#### GENERAL PROVISIONS

#### DIVIDENDS

1. Dividends upon the capital stock of the Corporation, subject to applicable law and the provisions of the Certificate of Incorporation, if any, may be declared by the Board of Directors at any regular or special meeting, pursuant to law. Dividends may be paid in cash, in property, in shares of the capital stock of the Corporation or out of any other assets of the Corporation legally available therefor, subject to the provisions of the Certificate of Incorporation and applicable law.

- 13 -

2. Before payment of any dividend, there may be set aside out of any funds of the Corporation available for dividends such sum or sums as the Board of Directors from time to time, in its absolute discretion, deem proper as a reserve or reserves to meet contingencies, or for equalizing dividends, or for repairing or maintaining any property of the Corporation, or for such other purposes as the Board of Directors shall think conducive to the interest of the Corporation, and the Board of Directors may modify or abolish any such reserve in the manner in which it was created.

#### CHECKS

3. All checks or demands for money and notes of the Corporation shall be signed by such officer or officers or such other person or persons as the Board of Directors may from time to time designate.

#### FISCAL YEAR

4. The fiscal year of the Corporation shall be fixed by resolution of the Board of Directors. In the absence of such a resolution, the fiscal year of the Corporation shall be the calendar year.

#### SEAL

5. The Board of Directors may adopt a corporate seal having inscribed thereon the name of the Corporation, the year of its organization and the words Corporate Seal, Delaware . The seal may be used by causing it or a facsimile thereof to be impressed or affixed or reproduced or otherwise.

### FCC ELIGIBILITY STOCKHOLDERS

6. In order to enable the Corporation to establish that existing and proposed stockholders are eligible to be stockholders of the Corporation under applicable law, the officers of the Corporation, to the extent necessary, may request from each existing and proposed stockholder information relating to the citizenship and the extent, if any, of the foreign ownership of the stockholder, and such other information regarding the stockholder as is reasonable to ensure the Corporation is in compliance with applicable law.

#### **GOVERNING LAW; FORUM FOR ADJUDICATION OF DISPUTES**

7. These Bylaws and the internal affairs of the Corporation shall be governed by and interpreted under the laws of the State of Delaware, excluding its conflict of laws principals. Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim of breach of a fiduciary duty owed by any director, officer or other employee of the Corporation to the Corporation or the Corporation s stockholders, (iii) any action asserting a claim arising pursuant to any provision of the Delaware Corporation Law or the Corporation s Certificate of Incorporation or Bylaws, or (iv) any other action asserting a claim arising under, in connection with, and governed by the internal affairs doctrine.

#### ARTICLE VIII

#### AMENDMENTS

1. These Bylaws may be altered, amended or repealed only in accordance with the provisions of the Certificate of Incorporation.

- 14 -

<u>Exhibit C</u>

## STOCKHOLDER SAGREEMENT

## by and between

## DEUTSCHE TELEKOM AG

and

## METROPCS COMMUNICATIONS, INC.

DATED AS OF [ ], 201[ ]

### TABLE OF CONTENTS

		Page
Article I DEFINITIONS		1
Section 1.1	Definitions	1
Section 1.2	Other Definitional Provisions	5
	SENTATIONS AND WARRANTIES	5
Section 2.1	Representations and Warranties of the Company	5
Section 2.2	Representations and Warranties of the Stockholder	6
Article III CORPORATE GOVERNANCE		6
Section 3.1	Board Representation	6
Section 3.2	Specified Actions	8
Section 3.3	Organizational Documents Actions	9
Section 3.4	Debt Defaults	9
Section 3.5	Information	9
Section 3.6	Director Consent Rights	10
Article IV TRANSFERS OF COMMON STOCK		10
Section 4.1	Certain Acquisitions	10
Section 4.2	Certain Dispositions	11
Article V REGISTRATION RIGHTS		11
Section 5.1	Shelf Registration	11
Section 5.2	Demand Registration	11
Section 5.3	Registration Obligations	12
Section 5.4	Piggy-Back Registration	13
Section 5.5	Cutbacks	13
Section 5.6	Termination of Registration Obligation	14
Section 5.7	Registration Procedures	14
Section 5.8	Registration Expenses	18
Section 5.9	Indemnification: Contribution	18
Section 5.10	Indemnification Procedures	19
Section 5.11	Transferee Registration Rights	20
Article VI NON-COMPETITION		20
Section 6.1	Non-Competition	20
Section 6.2	Reasonable Scope	21
Article VII MISCELLANEOUS		21
Section 7.1	Injunctive Relief	21
Section 7.2	Assignment	21
Section 7.3	Amendments; Waiver	21
Section 7.4	Termination	22
Section 7.5	Notices	22
Section 7.6	Governing Law; Jurisdiction; Forum; Waiver of Trial by Jury	23
Section 7.7	Interpretation	23
Section 7.8	Entire Agreement; No Other Representations	23
Section 7.9	No Third-Party Beneficiaries	23
Section 7.10	<u>Severability</u>	23
Section 7.11	Counterparts	23

i

STOCKHOLDER S AGREEMENT, dated as of [ ], 201[ ] (this <u>Agreement</u>), by and between DEUTSCHE TELEKOM AG, an *Aktiengesellschaft* organized and existing under the Laws of the Federal Republic of Germany (the <u>Stockholder</u>), and METROPCS COMMUNICATIONS, INC., a Delaware corporation (the <u>Company</u>). Capitalized terms used herein and not otherwise defined shall have the respective meanings assigned to such terms in that certain Business Combination Agreement, dated as of October 3, 2012 (the <u>Business</u> <u>Combination Agreement</u>), by and among the Stockholder, the Company, T-Mobile Global Zwischenholding GmbH, T-Mobile Global Holding GmbH (<u>Holding</u>), and T-Mobile USA, Inc. (<u>TMUS</u>).

#### WITNESSETH:

WHEREAS, the Company and the Stockholder have entered into the Business Combination Agreement, pursuant to which, among other things, Holding desires to sell to the Company, and the Company desires to purchase from Holding, all of the issued and outstanding equity interests of TMUS in exchange for the issuance of a certain amount of shares of Common Stock (as defined below) to Holding and other consideration, all upon the terms and subject to the conditions set forth therein; and

WHEREAS, the Company and the Stockholder desire to establish in this Agreement certain rights and obligations in respect of the shares of common stock, par value \$0.0001 per share, of the Company (the <u>Common Stock</u>) received by Holding, and related matters concerning the Stockholder s relationship with and investment in the Company.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

#### ARTICLE I

#### DEFINITIONS

Section 1.1 Definitions. As used in this Agreement, the following terms shall have the meanings indicated below:

Acquiror Purchase Offer shall have the meaning set forth in Section 4.2(c).

<u>Affiliate</u> shall mean with respect to any Person, a Person that directly or indirectly through one or more intermediaries, Controls, is Controlled by, or is under common Control with such Person, <u>provided</u> that the Stockholder shall not be deemed to be an Affiliate of the Company and vice versa.

Affiliated Directors shall mean Directors who are also officers, employees, directors or Affiliates of the Stockholder .

Agreement shall have the meaning set forth in the Preamble.

<u>Beneficially Own</u> shall mean, with respect to any securities, (i) having beneficial ownership of such securities for purposes of Rule 13d-3 or 13d-5 under the Exchange Act (or any successor statute or regulation), (ii) having the right to become the Beneficial Owner of such securities (whether such right is exercisable immediately or only after the passage of time or the occurrence of conditions) pursuant to any agreement, arrangement or understanding, or upon the exercise of conversion rights, exchange rights, rights, warrants or options, or otherwise, or (iii) having an exercise or conversion privilege or a settlement payment or mechanism with respect to any option, warrant, convertible security, stock appreciation, swap agreement or other security, contract right or derivative position, whether or not currently exercisable, at a price related to the value of the

securities for which Beneficial Ownership is being determined or a value determined in whole or part with reference to, or derived in whole or in part from, the value of the securities for which Beneficial Ownership is being determined that increases in value as the value of the securities for which Beneficial Ownership is being determined increases or that provides to the holder an opportunity, directly or indirectly, to profit or share in any profit derived from any increase in the value of the securities for which Beneficial Ownership is being determined (excluding any interests, rights, options or other securities set forth in Rule 16a-1(c)(1)-(5) or (7) promulgated pursuant to the Exchange Act).

Blackout Period shall have the meaning set forth in Section 5.3(c).

Board shall mean, as of any date, the Board of Directors of the Company in office on that date.

Business Combination Agreement shall have the meaning set forth in the Preamble.

<u>Business Day</u> shall mean any day other than a Saturday, a Sunday, a federal holiday or a day on which banks in the City of New York or in Bonn, Germany are authorized or obligated by Law to close.

<u>Chosen Courts</u> shall have the meaning set forth in Section 7.6.

- <u>Claim Notice</u> shall have the meaning set forth in Section 5.10(a).
- <u>Claims</u> shall have the meaning set forth in Section 5.9(a).
- <u>Common Stock</u> shall have the meaning set forth in the Recitals.
- <u>Company</u> shall have the meaning set forth in the Preamble.
- <u>Company Information</u> shall have the meaning set forth in Section 3.5(b).
- <u>Competing Business</u> shall have the meaning set forth in Section 6.1.

<u>Control</u> shall mean the possession, direct or indirect, of the power to direct, or cause the direction of, the management and policies of a Person, whether through the ownership of voting securities, voting equity, limited liability company interests, general partner interests, or voting interests, by contract or otherwise.

<u>Debt to Cash Flow Ratio</u> shall have the meaning set forth in the indenture or other instrument governing the terms of the Stockholder Notes, as in effect on the date hereof.

Demand Registration Statement shall have the meaning set forth in Section 5.2.

Demand Request shall have the meaning set forth in Section 5.2.

Director shall mean any member of the Board.

EDGAR shall have the meaning set forth in Section 5.7(a)(ii).

Effective Period shall have the meaning set forth in Section 5.7(a)(iii).

<u>Encumbrance</u> shall mean any lien, pledge, charge, claim, encumbrance, hypothecation, security interest, option, lease, license, mortgage, easement or other restriction or third-party right of any kind, including any right of first refusal, tag-along or drag-along rights or restriction on voting, transferring, lending, disposing or assigning, in each case other than pursuant to this Agreement.

Exchange Act shall mean the Securities Exchange Act of 1934, as amended.

<u>Existing Registration Rights Agreement</u> shall mean the Registration Rights Agreement, effective as of April 24, 2007, by and among MetroPCS and the stockholders listed therein.

Holding shall have the meaning set forth in the Preamble.

<u>Indebtedness</u> shall have the meaning set forth in the indenture or other instrument governing the terms of the Stockholder Notes, as in effect on the date hereof.

<u>Indemnifying Party</u> shall have the meaning set forth in Section 5.10(a).

Lock-Up Period shall have the meaning set forth in Section 4.2(a).

<u>Maximum Number</u> shall have the meaning set forth in Section 5.5.

Non-Affiliated Directors shall mean any Directors who are not Affiliated Directors.

<u>NYSE</u> shall mean the New York Stock Exchange.

<u>Organizational Documents</u> shall mean, with respect to any Person, such Person s articles or certificate of association, incorporation, formation or organization, by-laws, limited liability company agreement, partnership agreement or other constituent document or documents, each in its currently effective form as amended from time to time.

Other Holder shall have the meaning set forth in Section 5.5.

<u>Permitted Debt</u> shall have the meaning set forth in the indenture or other instrument governing the terms of the Stockholder Notes, as in effect on the date hereof.

<u>Permitted Liens</u> shall have the meaning set forth in the indenture or other instrument governing the terms of the Stockholder Notes, as in effect on the date hereof.

<u>Person</u> shall mean any individual, corporation (including not-for-profit), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, governmental entity or other entity of any kind or nature.

<u>Piggy-Back Registration</u> shall have the meaning set forth in Section 5.4.

<u>Piggy-Back Reque</u>st shall have the meaning set forth in Section 5.4.

Piggy-Back Securities shall have the meaning set forth in Section 5.4.

Potential Default shall have the meaning set forth in Section 3.4.

<u>Proposed Acquiror</u> shall have the meaning set forth in Section 4.2(b).

<u>Proposed Acquisition</u> shall have the meaning set forth in Section 4.1(b).

<u>Proposed Sale</u> shall have the meaning set forth in Section 4.2(b).

<u>Purchaser Shares</u> shall mean the shares of Common Stock Beneficially Owned, as of the date of determination, by the Stockholder and any other securities issued in respect thereof or into which such shares of Common Stock shall be converted or exchanged in connection with stock dividends or distributions, combinations or any similar recapitalizations on or after the date hereof.

## - 3 -

<u>Registrable De</u>bt shall mean, at any time, notes, debentures or other debt securities of the Company or any of its Subsidiaries that are Beneficially Owned by the Stockholder.

Registrable Securities shall mean the Registrable Shares and the Registrable Debt.

<u>Registrable Securities Transfere</u>e shall have the meaning set forth in Section 5.11(a).

Registrable Shares shall mean, at any time, the Purchaser Shares that are Beneficially Owned by the Stockholder.

<u>Required Approval</u> shall have the meaning set forth in Section 4.1(b).

<u>Representatives</u> shall have the meaning set forth in Section 3.5(b).

<u>Rule 144</u> shall mean Rule 144 promulgated under the Securities Act or any similar rule or regulation hereafter adopted by the SEC having substantially the same effect as such rule.

<u>S-3 Eligible</u> shall have the meaning set forth in Section 5.1.

Stockholder shall have the meaning set forth in the Preamble.

Stockholder Designees shall have the meaning set forth in Section 3.1(b).

Stockholder Notes shall mean the DT Notes, as defined in the Business Combination Agreement.

Stockholder Purchase Offer shall have the meaning set forth in Section 4.1(b).

<u>Subsidiary</u> shall mean, with respect to any Person, any entity, whether incorporated or unincorporated, of which (i) voting power to elect a majority of the board of directors, management committee or others performing similar functions with respect to such other Person is held by the first mentioned Person and/or by any one or more of its Subsidiaries, (ii) a general partnership interest is held by such first mentioned Person and/or by any one or more of its Subsidiaries (where such first mentioned Person (A) does not Beneficially Own a majority of the general partnership interests or voting interests and (B) does not otherwise Control such entity, directly or indirectly, by contract, arrangement or otherwise), or (iii) at least 50% of the Equity Interests of such other Person is, directly or indirectly, owned or Controlled by such first mentioned Person and/or by any one or more of its Subsidiaries.

Takedown Prospectus Supplement shall have the meaning set forth in Section 5.1.

Takedown Request shall have the meaning set forth in Section 5.1.

<u>TMUS</u> shall have the meaning set forth in the Preamble.

<u>Transfer</u> shall mean any direct or indirect sale, transfer, assignment, pledge, hypothecation, mortgage, license, gift, creation of a security interest in or lien on, placement in trust (voting or otherwise), encumbrance or other disposition to any Person, including those by way of spin-off (such as through a dividend), hedging or derivative transactions or otherwise; <u>provided</u>, <u>however</u>, that any tender or exchange offer, merger (other than a merger by the Stockholder to effect a reorganization or recapitalization), amalgamation, plan of arrangement or consolidation or any similar transaction in which each holder of capital stock of the Stockholder (other than, if applicable, the Person proposing such transaction) disposes, sells, transfers or assigns or is offered the opportunity to dispose, sell, transfer or assign some or all of the capital stock of the Stockholder Beneficially Owned by each such holder or which otherwise results in the acquisition of some or all of the capital stock of the Stockholder Beneficially Owned by each such holder shall not be deemed to be the Transfer of any Purchaser Shares Beneficially Owned by the Stockholder.

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

#### Table of Contents

Votes shall mean the number of votes entitled to be cast generally in the election of Directors.

<u>Voting Percentage</u> of a Person shall mean, as of the date of determination, the ratio, expressed as a percentage, of (i) the Votes entitled to be cast by the holders of the Voting Securities Beneficially Owned by such Person to (ii) the aggregate Votes entitled to be cast by all holders of the then-outstanding Voting Securities.

<u>Voting Securities</u> shall mean, together, (i) the Common Stock and (ii) any class of capital stock or other securities of the Company other than the Common Stock that are entitled to vote generally in the election of Directors.

Section 1.2 Other Definitional Provisions. Unless the express context otherwise requires:

(a) the words hereof, herein, and hereunder and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement;

(b) the words date hereof, when used in this Agreement, shall refer to the date set forth in the Preamble;

(c) the terms defined in the singular have a comparable meaning when used in the plural, and vice versa;

(d) the terms defined in the present tense have a comparable meaning when used in the past tense, and vice versa;

(e) any references herein to Dollars and \$ are to United States Dollars;

(f) any references herein to a specific Section, Schedule, Annex or Exhibit shall refer, respectively, to Sections, Schedules, Annexes or Exhibits of this Agreement;

(g) wherever the word include , includes , or including is used in this Agreement, it shall be deemed to be followed by the words without limitation ;

(h) references herein to any gender includes each other gender; and

(i) the word or shall not be exclusive.

#### ARTICLE II

#### **REPRESENTATIONS AND WARRANTIES**

Section 2.1 <u>Representations and Warranties of the Company</u>. The Company represents and warrants to the Stockholder that, as of the date hereof:

(a) The Company is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware.

(b) The Company has all requisite power and authority and has taken all action necessary in order to execute and deliver this Agreement and to perform its obligations hereunder. The execution and delivery by the Company of this Agreement and the performance of its obligations hereunder have been duly authorized by all necessary action of the Company. This Agreement has been duly executed and delivered by the Company and, assuming the due authorization, execution and delivery of this Agreement by the Stockholder, constitutes the legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as limited by applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equitable principles.

(c) The execution and delivery of this Agreement by the Company and the performance of its obligations hereunder will not constitute or result in (i) a breach or violation of, or a default under, the Organizational Documents of the Company, (ii) a breach or violation of, a termination (or right of termination) or default under, the creation or acceleration of any obligations under, or the creation of an Encumbrance on any of the assets of the Company (with or without notice, lapse of time or both) pursuant to, any agreement, lease, license, contract, note, mortgage, indenture, arrangement or other obligation binding upon the Company, or (iii) conflict with, breach or violate any Law applicable to the Company or by which its properties are bound or affected, except, in the case of clause (ii) or (iii) above, for any breach, violation, termination, default, creation or acceleration that would not, individually or in the aggregate, reasonably be likely to impair in any material respect the ability of the Company to perform its obligations under this Agreement.

(d) The Company is a well-known seasoned issuer (as defined in Rule 405 promulgated under the Securities Act) eligible to register the Registrable Shares for resale by the Stockholder on a registration statement on Form S-3 under the Securities Act. The Company is subject to the reporting requirements of the Exchange Act.

Section 2.2 <u>Representations and Warranties of the Stockholder</u>. The Stockholder represents and warrants to the Company that, as of the date hereof:

(a) The Stockholder is an Aktiengesellschaft organized and existing under the Laws of the Federal Republic of Germany.

(b) The Stockholder has all requisite power and authority to execute and deliver this Agreement and to perform its obligations hereunder. The execution and delivery by the Stockholder of this Agreement and the performance of its obligations hereunder have been duly authorized by all necessary action of the Stockholder. This Agreement has been duly executed and delivered by the Stockholder and, assuming the due authorization, execution and delivery of this Agreement by the Company, constitutes the legal, valid and binding obligation of the Stockholder, enforceable against the Stockholder in accordance with its terms, except as limited by applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws affecting the enforcement of creditors rights generally or, as to enforceability, by general equitable principles.

(c) The execution and delivery of this Agreement by the Stockholder and the performance of its obligations hereunder will not constitute or result in (i) a breach or violation of, or a default under, the Organizational Documents of the Stockholder, (ii) a breach or violation of, a termination (or right of termination) or default under, the creation or acceleration of any obligations under, or the creation of an Encumbrance on any of the assets of the Stockholder (with or without notice, lapse of time or both) pursuant to, any agreement, lease, license, contract, note, mortgage, indenture, arrangement or other obligation binding upon the Stockholder, or (iii) conflict with, breach or violate any Law applicable to the Stockholder or by which its properties are bound or affected, except, in the case of clause (ii) or (iii) above, for any breach, violation, termination, default, creation or acceleration that would not, individually or in the aggregate, reasonably be likely to impair in any material respect the ability of the Stockholder to perform its obligations under this Agreement.

### ARTICLE III

#### CORPORATE GOVERNANCE

Section 3.1 Board Representation. At all times when the Stockholder s Voting Percentage is 10% or more:

(a) The Company and the Stockholder shall use their reasonable best efforts to cause at least three of the Directors to be considered independent under the rules of the SEC, the NYSE and any other or additional exchange on which the securities of the Company are listed, including for purposes of Rule 10A-3 promulgated under the Exchange Act (or any successor rule thereto).

- 6 -

(b) The Stockholder shall have the right to designate a number of individuals to be nominees for election to the Board (<u>Stockholder Designees</u>) equal to the Stockholder s Voting Percentage multiplied by the total number of Directors that the Company would have if there were no vacancies, rounded to the nearest whole number (and in any event not less than one), and the Company and the Stockholder shall use their reasonable best efforts to cause such Stockholder Designees to be elected to the Board; <u>provided</u> that the number of Directors who are Affiliated Directors shall not in any event exceed a number equal to the Stockholder s Voting Percentage multiplied by the total number of Directors that the Company would have if there were no vacancies, rounded to the nearest whole number greater than zero. If at any time the Stockholder s Voting Percentage is less than 10%, the Stockholder shall promptly cause all of the Stockholder Designees pursuant to this Article III shall forever terminate.

(c) The Company shall cause any committee of the Board to include in its membership a number of Stockholder Designees then serving as Directors equal to the Stockholder s Voting Percentage multiplied by the total number of members that such committee would have if there were no vacancies on such committee, rounded to the nearest whole number, except to the extent that such membership would violate the rules of the SEC, the NYSE and any other or additional exchange on which the securities of the Company are listed, or any other applicable securities Laws; provided, however, that no committee may consist solely of Affiliated Directors. If at any time the number of Stockholder Designees then serving as Directors or as members of any committee of the Board exceeds the number of Stockholder Designees the Stockholder is entitled to designate to the Board or any committee thereof pursuant to this Article III, the Stockholder shall cause the number of Stockholder Designees then serving as Directors or as members of such committee of the Board representing such excess to resign immediately as Directors or committee members, as applicable.

(d) Each Stockholder Designee shall not be prohibited or disqualified from serving as a Director pursuant to any rule or regulation of the SEC, the NYSE or any other or additional exchange on which securities of the Company are listed or by applicable Law. The Stockholder shall, and shall cause the Stockholder Designees to, timely provide the Company with accurate and complete information relating to the Stockholder and the Stockholder Designees that may be required to be disclosed by the Company under the Securities Act or the Exchange Act, including such information required to be furnished by the Company with respect to the Stockholder Designees in a proxy statement pursuant to Rule 14a-101 promulgated under the Exchange Act, and the nationality of such Stockholder Designee. In addition, at the Company is request, the Stockholder shall cause the Stockholder Designees to complete and execute the Company is director and officer questionnaire prior to being elected to the Board or standing for reelection at an annual meeting of stockholders or at such other time as may be reasonably requested by the Company.

(e) With respect to each meeting of stockholders of the Company at which Directors are to be elected, the Company shall provide the Stockholder with notice of such meeting not less than 120 days prior to the date thereof, and the Stockholder shall provide the Company with written notice of the names (together with all other information requested by the Company pursuant to Section 3.1(d)) of the Stockholder Designees to be nominated for election at such meeting not more than 30 days following the delivery of such notice. If the Stockholder shall fail to timely provide the Company with the names of that number of Stockholder Designees equal to the number of Stockholder Designees the Stockholder Designee is not qualified, available or eligible to stand for election, then the Stockholder may name an acceptable and available replacement Stockholder Designee and any such Stockholder Designees to be included as a nominee for election at such meeting of the proxy statement for such meeting. The Company shall cause the Stockholder Designees to be included in the slate of Directors approved and recommended by the Board for election at such meeting and shall use its reasonable best efforts to cause the election of each such Stockholder Designee, including soliciting proxies in favor of the election of such Stockholder Designees at such meeting.

- 7 -

(f) In the event the size of the Board is increased at any time and as a result of such increase, the Stockholder shall be entitled to designate one or more additional Stockholder Designees based upon the increased size of the Board and its then Voting Percentage pursuant to this Section 3.1,
(i) the Stockholder shall be entitled promptly to designate such Stockholder Designees, and (ii) the Company shall cause the prompt appointment or election of such Stockholder Designee(s) as Director(s).

(g) Upon the resignation, retirement, death or other removal (with or without cause) from office of any Stockholder Designee serving as a Director at a time when the Stockholder has the right under this Section 3.1 to designate a replacement Stockholder Designee, (i) the Stockholder shall be entitled promptly to designate a replacement Stockholder Designee and (ii) the Company shall cause the prompt appointment or election of such replacement Stockholder Designee as a Director.

Section 3.2 <u>Specified Actions</u>. In addition to any other vote, consent or approval required by the Company s Organizational Documents, this Agreement or applicable Law, for so long as the Stockholder s Voting Percentage is 30% or greater, the Company shall not, and shall cause its Subsidiaries not to, take or agree to take any of the following actions, in each case without the prior written consent of the Stockholder, which consent the Stockholder may withhold in its sole discretion:

(a) create, incur, issue, assume or otherwise become liable for (including through a merger, acquisition or otherwise) or refinance or guarantee any Indebtedness (excluding any Permitted Debt) that would result in the Company and its subsidiaries, on a consolidated basis, having or being liable for Indebtedness in an aggregate principal amount that would result in the Debt to Cash Flow Ratio for the Company s most recently ended four full fiscal quarters for which financial statements are available to be greater than 5.25 to 1.0 on a *pro forma* basis as if the additional Indebtedness had been incurred at the beginning of such four-quarter period;

(b) take any action or enter into any transaction that would reasonably be expected to result in a breach of or default under any credit agreement, indenture, note, or similar instrument or security to which the Stockholder or any of its Affiliates is a party or is bound;

(c) acquire (including by way of merger, recapitalization, reorganization, liquidation or dissolution) any business, debt or equity interests, operations or assets of any Person, or make any investment in or loan to any Person, in any single transaction or series of related transactions (excluding the acquisition of products and equipment in the ordinary course of business), for consideration in excess of \$1,000,000,000;

(d) sell, lease, transfer, Encumber (other than Permitted Liens) or otherwise dispose of (including by way of merger, recapitalization, reorganization, liquidation or dissolution) any division, business, or operations of the Company or any of its Subsidiaries, or any equity interests of the Company or any of its Subsidiaries, in any single transaction or series of related transactions, for consideration in excess of \$1,000,000,000;

(e) change the size of the Board;

(f) issue any equity or equity-linked securities or other Voting Securities of the Company or any of its Subsidiaries, in any single transaction or series of related transactions, (i) constituting 10% or more of the then outstanding shares of Common Stock (other than grants of incentive awards to officers or employees of the Company or its Subsidiaries that are approved by the Board or the applicable committee thereof or issuances of securities to the Company or any of its wholly-owned Subsidiaries) or (ii) for the purpose of redeeming or purchasing any indebtedness of the Company held by the Stockholder or its Affiliates;

(g)(i) except as required in the Organizational Documents, repurchase or redeem any equity (or equity-based) securities of the Company or any of its non-wholly owned Subsidiaries, or (ii) make any extraordinary or in-kind dividend with respect to any of the equity (or equity-based) securities of the Company or any of its Subsidiaries, other than a dividend on a *pro rata* basis with respect to all stockholders of the Company, or a dividend to the Company or any of its wholly owned Subsidiaries; or

- 8 -

(h) hire, or terminate without cause, its Chief Executive Officer, or agree to do so.

Section 3.3 <u>Organizational Documents Actions</u>. In addition to any other vote, consent or approval required by the Company's Organizational Documents, this Agreement or applicable Law, for so long as the Stockholder's Voting Percentage is 5% or greater, the Company shall not amend or seek to amend its Organizational Documents (including, for the avoidance of doubt, the creation of any shareholder rights plan or other amendment intended to limit the Stockholder's ownership or acquisition of securities of the Company) in any manner that could limit, restrict or adversely affect the Stockholder or its rights thereunder without the prior written consent of the Stockholder, which consent may be withheld in its sole discretion.

Section 3.4 <u>Debt Defaults</u>. The Company shall notify the Stockholder any time it is reasonably likely that the Company will default on any Indebtedness (as defined in the Business Combination Agreement) with a principal amount greater than \$75 million (a <u>Potential Default</u>). Thereupon, the Stockholder shall have the right, but not the obligation, to provide new debt financing to the Company up to the amount of the Indebtedness that is the subject of the Potential Default plus any applicable prepayment or other penalties, on the same terms and conditions as such Indebtedness (together with any waiver of the Potential Default). If Stockholder elects to provide the Company with new debt financing, the Company shall take any actions reasonably requested by the Stockholder (i) to prepare documentation reflecting the terms and conditions of the new debt financing; (ii) to repay the Indebtedness that is the subject of the Potential Default.

#### Section 3.5 Information.

(a) For so long as the Stockholder s Voting Percentage is 10% or greater, the Stockholder shall be entitled to the information and consultation rights set forth in this Section 3.5 with respect to the Company and its Subsidiaries, in addition to any other vote, consent or approval rights set forth herein, in the Company s Organizational Documents or otherwise: (i) the Stockholder shall be entitled to consult with the officers of the Company with respect to the Company s business and financial matters, including management s proposed annual operating plans, and, upon request, members of management will meet with representatives of the Stockholder at mutually agreeable times and places for such consultation, including to review progress in achieving said plans; provided that such consultation shall not unreasonably disrupt the normal operations of the Company or its Subsidiaries and the Stockholder shall be responsible for any out-of-pocket costs and expenses incurred by the Company in connection with respect to the business and properties of the Company and its Subsidiaries as the Stockholder may reasonably request; provided that such request must be made through the Company s chief financial officer or one or more individuals designated by such person, and in any event, if a Stockholder Designee is then serving as a Director, with all information provided to members of the Board; and (iii) the Stockholder shall be entitled to inspect all books and records and facilities and properties of the Company at reasonable times and intervals.

(b) Subject to the requirements of applicable Law, regulations and rules (including the regulations and rules of the NYSE and any other or additional exchange on which the securities of the Company are listed), the Stockholder shall, and shall cause its officers, directors, employees, accountants, counsel and consultants (<u>Representatives</u>) and the Stockholder Designees to, keep confidential all information and documents of the Company and its Affiliates obtained by the Stockholder and the Stockholder Designees (the <u>Company Information</u>) unless the Company Information (i) is or becomes publicly available other than as a result of a breach of this Section 3.5 by the Stockholder, including by way of actions taken by its Representatives or the Stockholder Designees; (ii) was within the possession of the Stockholder or the Stockholder Designees prior to being furnished such information by or on behalf of the Company on a non-confidential basis; <u>provided</u> that the source of such information was not known by the Stockholder, its Representatives or the Stockholder Designees to be bound by a confidentiality agreement with, or other contractual, legal or fiduciary obligation of confidentiality to, the Company or any of its Subsidiaries with respect to the Company Information; (iii) was

- 9 -

available to the Stockholder or the Stockholder Designees on a non-confidential basis from a source other than the Company, any of its Subsidiaries or any of its or their Representatives; <u>provided</u> that such source was not known to the Stockholder or the Stockholder Designees to be bound by a confidentiality agreement with, or other contractual, legal or fiduciary obligation of confidentiality to, the Company or any of its Subsidiaries with respect to such Company Information; or (iv) was independently developed by or on behalf of the Stockholder without violating any of the obligations under this Section 3.5. The Stockholder shall, and shall cause its controlled Affiliates, Representatives, and the Affiliated Directors to, comply with applicable law regarding insider trading in the Company s securities to the extent any of them is in possession of Company Information.

(c) The Stockholder hereby acknowledges that it is aware and will advise its Representatives who are informed as to the matters which are the subject of this Agreement, that the United States securities Laws prohibit any Person who is in possession of material, non-public information concerning the matters that are the subject of this Agreement from purchasing or selling securities of the Company or from communicating such information to any other Person under circumstances in which it is reasonably foreseeable that such Person is likely to purchase or sell securities.

(d) From the date hereof until the next subsequent filing of the Company s annual report on Form 10-K, the Stockholder agrees to use its commercially reasonable efforts to assist the Company with the integration of TMUS and its Subsidiaries into the Company s internal control structure and procedures for financial reporting compliance with the requirements of Rule 404 of the Sarbanes-Oxley Act of 2002.

#### Section 3.6 Director Consent Rights.

(a) In addition to any other vote, consent or approval required by the Company s Organizational Documents, this Agreement, applicable Law or otherwise, during the term of this Agreement, the Stockholder agrees not to, and shall cause the Stockholder Designees then serving as Directors not to, support, enter into or vote in favor of any transaction between, or involving both (A) the Company and (B) the Stockholder or an Affiliate of the Stockholder unless such transaction is approved by a majority of the Directors, which majority includes a majority of the Non-Affiliated Directors (it being understood that such approval or similar approvals in this Agreement shall not require any separate vote or consent of the Non-Affiliated Directors).

(b) From and after the Closing, the Stockholder agrees that the Non-Affiliated Directors shall direct and make any determinations with respect to the Company s post-Closing actions under Section 2.4 of the Business Combination Agreement.

### ARTICLE IV

#### TRANSFERS OF COMMON STOCK

Section 4.1 <u>Certain Acquisitions</u>. (a) The Stockholder shall not, and shall cause each of its Affiliates not to, directly or indirectly, alone or in concert with any other Person, acquire, offer to acquire or agree to acquire (including from the Company) Beneficial Ownership of any Common Stock that would cause the Stockholder s and its Affiliates Voting Percentage to exceed 80.1%, except in accordance with the following provisions of this Section 4.1.

(b) If the Stockholder or its Affiliates proposes to acquire Common Stock that would cause the Stockholder s Voting Percentage to exceed 80.1% (the <u>Proposed Acquisition</u>), the Stockholder or its Affiliates shall offer to acquire all of the then-outstanding Common Stock at the same price and on the same terms and conditions as the Proposed Acquisition (the <u>Stockholder Purchase Offer</u>). For the avoidance of doubt, the Stockholder Purchase Offer may contemplate a merger or other consolidation, a tender offer, or any other transaction that could permit the acquisition of all of the then-outstanding Common Stock. The Stockholder shall

- 10 -

not, and shall cause its Affiliates not to, consummate, in whole or in part, any Proposed Acquisition or Stockholder Purchase Offer unless such Stockholder Purchase Offer is either (i) accepted and approved by a majority of the Directors, which majority includes a majority of the Non-Affiliated Directors, or (ii) accepted or approved by holders of a majority of the Common Stock held by stockholders of the Company other than the Stockholder and its Affiliates (either of clause (i) or (ii), the <u>Required Approval</u>). The Stockholder may, in its sole discretion, withdraw any Stockholder Purchase Offer and terminate any Proposed Acquisition at any time.

Section 4.2 <u>Certain Dispositions</u>. (a) The Stockholder shall not Transfer any Registrable Shares during the 183-day period commencing at the Closing (the <u>Lock-Up Period</u>) other than Transfers approved by a majority of the Directors, which majority includes a majority of the Non-Affiliated Directors.

(b) Following the Lock-Up Period, the Stockholder and its Affiliates may freely Transfer their Common Stock subject to applicable Law, <u>provided</u> that if the Stockholder intends to Transfer any Common Stock to a third party (the <u>Proposed Acquiror</u>), as a result of which Transfer (to the knowledge of Stockholder following reasonable inquiry) the Proposed Acquiror s Voting Percentage would be greater than 30% (the <u>Proposed Sale</u>), then the Stockholder shall not effect such Proposed Sale other than in accordance with Section 4.2(c).

(c) No Proposed Sale shall be consummated unless the Proposed Acquiror shall contemporaneously make a binding offer to acquire all of the then-outstanding Common Stock of the Company, at the same price and on the same terms and conditions as the Proposed Sale (the <u>Acquiror</u> <u>Purchase Offer</u>). For the avoidance of doubt, the Acquiror Purchase Offer may contemplate a merger or other consolidation, a tender offer, or any other transaction that permits the acquisition of all of the then-outstanding Common Stock.

#### ARTICLE V

#### **REGISTRATION RIGHTS**

Section 5.1 <u>Shelf Registration</u>. As soon as reasonably practicable following the Closing, and in any event within 30 days thereof, the Company shall file, and shall thereafter use its commercially reasonable efforts to make and keep effective (including by renewing or refiling upon expiration), a shelf registration statement permitting the resale from time to time on a delayed or continuous basis pursuant to Rule 415 of the Securities Act by the Stockholder of the Registrable Securities, which registration statement shall be filed on (i) Form S-3, if the Company is then eligible to file a registration statement on Form S-3 (pursuant to the General Instructions to Form S-3) (<u>S-3 Eligible</u>), or (ii) any other appropriate form under the Securities Act for the type of offering contemplated by the Stockholder, if the Company is not then S-3 Eligible. Thereafter, the Company shall, as promptly as reasonably practicable following the written request of the Stockholder for a resale of Registrable Securities (a <u>Takedown Request</u>), file a prospectus supplement (<u>a Takedown Prospectus Supplement</u>) to such shelf registration statement filed under Rule 424 promulgated under the Securities Act with respect to resales of the Registrable Securities pursuant to the Stockholder s intended method of distribution thereof, and to the extent such Takedown Prospectus Supplement is not automatically effective upon filing, shall, subject to the terms of this Article V, use its commercially reasonable efforts to cause such prospectus supplement to be declared effective under the Securities Act promptly after the filing thereof. Each Takedown Request shall specify the Registrable Securities to be registered, their aggregate amount, and the intended method or methods of distribution thereof. The Stockholder agrees to provide the Company with such information in connection with a Takedown Request as may be reasonably requested by the Company to ensure that the Takedown Prospectus Supplement complies with the requirements of the Securiti

Section 5.2 <u>Demand Registration</u>. At any time after the end of the Lock-Up Period and at which time the shelf registration statement required pursuant to Section 5.1 shall not be available for the resale of the Registrable Securities, including if for any reason the Company shall be ineligible to maintain or use a shelf registration statement, the Company shall, as promptly as reasonably practicable following the written request of the

- 11 -

Stockholder for registration under the Securities Act of all or part of the Registrable Securities (a <u>Demand Request</u>), file a registration statement with the SEC (a <u>Demand Registration Statement</u>) with respect to resales of the Registrable Securities pursuant to the Stockholder s intended method of distribution thereof, and shall, subject to the terms of this Article V, use its commercially reasonable efforts to cause such Demand Registration Statement to be declared effective under the Securities Act promptly after the filing thereof; <u>provided</u> that such Demand Registration Statement shall be filed on (i) Form S-3, if the Company is then S-3 Eligible, or (ii) any other appropriate form under the Securities Act for the type of offering contemplated by the Stockholder, if the Company is not then S-3 Eligible. Each Demand Request shall specify the Registrable Securities to be registered, their aggregate amount, and the intended method or methods of distribution thereof. The Stockholder agrees to provide the Company with such information in connection with a Demand Request as may be reasonably requested by the Company to ensure that the Demand Registration Statement complies with the requirements of the Securities Act.

#### Section 5.3 Registration Obligations.

(a) Notwithstanding anything to the contrary set forth in Section 5.1 or Section 5.2, the Company shall not be obligated to prepare, file or cause a Demand Registration Statement or Takedown Prospectus Supplement to become effective:

(i) unless the expected proceeds from the sale of the Registrable Securities to be included in such Demand Registration Statement or Takedown Prospectus Supplement is \$100,000,000 or greater; and

(ii) in the case of Registrable Shares, within 90 days after the effective date of a Takedown Prospectus Supplement, a Demand Registration Statement or a registration statement in which the Stockholder participated pursuant to its piggy-back rights pursuant to Section 5.4 (provided that the number of Registrable Shares included in such Piggy-Back Registration was not less than 60% of the number of Registrable Shares requested to be registered by the Stockholder pursuant to the Piggy-Back Request related to such Piggy-Back Registration), in each case with respect to Registrable Shares.

(b) Any Takedown Request or Demand Request may be revoked by notice from the Stockholder to the Company at any time prior to the effective date of the corresponding Takedown Prospectus Supplement or Demand Registration Statement; <u>provided</u> that (i) the Stockholder reimburses the Company for all reasonable, out-of-pocket expenses incurred by the Company in connection with such Takedown Request or Demand Request, and (ii) the Stockholder shall not make another Takedown Request or Demand Request with respect to Registrable Shares during the 45 days following the date of a revocation with respect to Registrable Shares.

(c) Notwithstanding anything in this Agreement to the contrary, the Company shall be entitled to postpone and delay, for reasonable periods of time not in excess of 60 days, but in no event more than twice in any 12-month period (a <u>Blackout Period</u>), the filing or effectiveness of any Takedown Prospectus Supplement or Demand Registration Statement or the offer or sale of any Registrable Securities thereunder if one or more executive officers of the Company shall determine in good faith that any such filing or the offering or sale of any Registrable Securities thereunder would (i) impede, delay or otherwise interfere with any pending or contemplated material acquisition, disposition, corporate reorganization or other similar material transaction involving the Company, (ii) based upon advice from the Company s investment banker or financial advisor, materially and adversely impede, delay or otherwise interfere with any pending or contemplated financing, offering or sale of any class of securities by the Company and its stockholders, or (iv) have a material adverse effect on the Company; provided, that in the event that the Company proposes to register Common Stock, whether or not for sale for its own account, during a Blackout Period, the Stockholder shall have the right to exercise its rights under Section 5.4 with respect to such registration, subject to the limitations contained in this Agreement on the exercise of such rights. Upon notice by the Company to the Stockholder of any such determination, the Stockholder shall, except as required by applicable Law, including

- 12 -

any disclosure obligations under Section 13 of the Exchange Act, keep the fact of any such notice strictly confidential, and during any Blackout Period, promptly halt any offer, sale, trading or Transfer by it of any Common Stock for the duration of the Blackout Period set forth in such notice (or until such Blackout Period shall be earlier terminated in writing by the Company) and promptly halt any use, publication, dissemination or distribution of any prospectus or prospectus supplement covering such Registrable Securities for the duration of the Blackout Period set forth in such notice (or until such Blackout Period shall be earlier terminated in writing by the Company) and, if so directed by the Company, shall deliver to the Company any copies then in its possession of any such prospectus or prospectus supplement.

(d) The Stockholder agrees that any rights granted pursuant to this Article V shall be subject to any rights pursuant to the Existing Registration Rights Agreement to the extent such rights remain in effect after the Closing.

(e) In connection with any offering pursuant to a Takedown Prospectus Supplement or a Demand Registration Statement, the managing underwriter for such offering shall be selected by the Stockholder; <u>provided</u>, <u>however</u>, that the managing underwriter must be a nationally recognized investment banking firm.

Section 5.4 <u>Piggy-Back Registration</u>. If the Company at any time proposes or is required to register any Common Stock or Company debt securities under the Securities Act on its behalf or on behalf of any of its stockholders, on a form and in a manner that would permit registration of the Registrable Securities (other than in connection with dividend reinvestment plans, rights offerings or a registration statement on Form S-4 or S-8 or any similar successor form), the Company shall give the Stockholder prompt written notice of its intent to do so not less than 15 Business Days prior to the contemplated filing date for such registration statement. Upon the written request of the Stockholder (a <u>Piggy-Back Request</u>), given within five Business Days following the time that the Stockholder was given any such written notice (which request shall specify the number of Registration statement (a <u>Piggy-Back Registration</u>), subject to the provisions of this Section 5.4 and, in the case of a registration on behalf of any of the Company stockholders, subject to the rights of such stockholders, the number of Registrable Securities set forth in such Piggy-Back Request.

Section 5.5 <u>Cutbacks</u>. In the event that (x) the Company proposes or is required (other than pursuant to a Takedown Request or Demand Request) to register Common Stock or Company debt securities in connection with an underwritten offering, (y) the Stockholder has made a Piggy-Back Request with respect to such offering, and (z) a nationally recognized investment banking firm selected by the Company to act as managing underwriter thereof reasonably and in good faith shall have advised the Company, the Stockholder or any other holder of Common Stock or Company debt securities intending to offer Common Stock or Company debt securities in the offering, as applicable (each, an <u>Other Holder</u>) in writing that, in its opinion, the inclusion in the registration statement of some or all of the Common Stock or Company debt securities sought to be registered by the Company, the Stockholder or the Other Holder(s) would adversely affect the price or success of the offering, the Company shall include in such registration statement such number of shares of Common Stock or principal amount of Company debt securities as the Company is advised can be sold in such offering without such an effect (the <u>Maximum Number</u>) as follows and in the following order of priority:

(a) if such registration is by the Company for its own account, (i) *first*, such number of shares of Common Stock or principal amount of Company debt securities as the Company proposes to register for its own account, (ii) *second*, to the extent the number of shares of Common Stock or Company debt securities to be included in the registration pursuant to clause (i) is less than the Maximum Number, such number of Piggy-Back Securities as the Stockholder proposes to be included pursuant to a Piggy-Back Request, and (iii) *third*, to the extent the number of shares of Common Stock or Company debt securities to be included in the registration pursuant to clause (i) and (ii) *third*, to the extent the number of shares of Common Stock or Company debt securities to be included in the registration pursuant to clauses (i) and (ii) is less than the Maximum Number, such number of shares of Common Stock or principal amount of Company debt securities as all Other Holders request to be included for their own account, on a *pro rata* basis; or

- 13 -

(b) if such registration is pursuant to the demand registration rights of one or more Other Holders, (i) *first*, such number of shares of Common Stock or principal amount of Company debt securities as such Other Holder(s) and the Stockholder propose to be included on a *pro rata* basis, and (ii) *second*, to the extent the number of shares of Common Stock or Company debt securities to be included in the registration pursuant to clause (i) is less than the Maximum Number, such number of shares of the Company Common Stock or principal amount of Company debt securities as the Company requests to be included, on a *pro rata* basis.

Section 5.6 <u>Termination of Registration Obligation</u>. The obligation of the Company to register Registrable Securities pursuant to this Article V and maintain the effectiveness of any shelf registration statement filed pursuant to Section 5.1 and Section 5.2 shall terminate (a) solely with respect to Registrable Shares, on the first date on which the Stockholder Voting Percentage is less than 5% and (b) solely with respect to Registrable Debt, on the first date on which the Stockholder no longer beneficially owns any Registrable Debt.

Section 5.7 <u>Registration Procedures</u>. (a) In connection with each registration statement prepared pursuant to this Article V pursuant to which Registrable Securities will be offered and sold, and in accordance with the intended method or methods of distribution of the Registrable Securities as described in such registration statement, the Company shall:

(i) use its commercially reasonable efforts to, as promptly as reasonably practicable, prepare and file with the SEC a registration statement on an appropriate registration form of the SEC and cause such registration statement to become effective under the Securities Act promptly after the filing thereof, which registration statement shall comply as to form in all materials respects with the requirements of the applicable form and include all financial statements required by such form to be filed therewith; provided that before filing a registration statement or prospectus or any amendments or supplements thereto, the Company shall furnish to one counsel selected by the Stockholder draft copies of all such documents proposed to be filed at least five Business Days prior to such filing, which documents will be subject to the reasonable review and comment of the Stockholder and its agents and Representatives and the underwriters, if any, and the Company shall not file any amendment or supplement to a Takedown Prospectus Supplement or Demand Registration Statement to which the Stockholder or the underwriters, if any, shall reasonably object;

(ii) use its commercially reasonable efforts to, as promptly as reasonably practicable, furnish without charge to the Stockholder, and the underwriters, if any, at least one conformed copy of the registration statement and each post-effective amendment or supplement thereto (including all schedules and exhibits but excluding all documents incorporated or deemed incorporated therein by reference, unless requested in writing by the Stockholder or an underwriter, except to the extent such exhibits and schedules are currently available via the SEC s Electronic Data Gathering, Analysis and Retrieval System ( $\underline{EDGAR}$ )) and such number of copies of the registration statement and each amendment or supplement thereto (excluding exhibits and schedules) and the summary, preliminary, final, amended or supplemented prospectuses included in such registration statement as the Stockholder or such underwriter may reasonably request in order to facilitate the public sale or other disposition of the Registrable Securities being sold by the Stockholder (the Company hereby consents to the use in accordance with the U.S. securities Laws of such registration statement (or post-effective amendment thereto) and each such prospectus (or preliminary prospectus or supplement thereto) by the Stockholder and the underwriters, if any, in connection with the offering and sale of the Registrable Securities covered by such registration statement or prospectus);

(iii) use its commercially reasonable efforts to keep such registration statement effective until the date that is 45 days after the date such registration statement is initially declared effective (or such shorter period as shall terminate when all of the securities covered by the registration statement have been disposed or withdrawn, or if such registration statement relates to a firm commitment underwritten offering, such longer period as, in the opinion of counsel for the underwriters for such offering, a prospectus is required under the Securities Act to be delivered in connection with sales of Registrable Securities by an underwriter or dealer

- 14 -

(but not in excess of 90 days) (the <u>Effective Period</u>), prepare and file with the SEC such amendments, post-effective amendments and supplements to the registration statement and the prospectus as may be necessary to maintain the effectiveness of the registration for the Effective Period) and cause the prospectus (and any amendments or supplements thereto) to be filed with the SEC;

(iv) use its commercially reasonable efforts to, as promptly as reasonably practicable, register or qualify the Registrable Securities covered by such registration statement under such other securities or blue sky laws of such jurisdictions in the United States as are reasonably necessary, keep such registrations or qualifications in effect for so long as the registration statement remains in effect, and do any and all other acts and things which may be reasonably necessary to enable the Stockholder or any underwriter to consummate the disposition of the Registrable Securities in such jurisdictions; provided, however, that in no event shall the Company be required to (A) qualify to do business as a foreign corporation in any jurisdiction where it would not, but for the requirements of this subparagraph (iv), be required to be so qualified, (B) execute or file any general consent to service of process under the Laws of any jurisdiction, (C) take any action that would subject it to service of process in suits other than those arising out of the offer and sale of the securities covered by the registration statement, or (D) subject itself to taxation in any jurisdiction where it would not otherwise be obligated to do so, but for this subparagraph (iv);

(v) use its commercially reasonable efforts to, as promptly as reasonably practicable, cause all Registrable Shares covered by such registration statement, if any, to be listed (after notice of issuance) on the NYSE or on the principal securities exchange or interdealer quotation system on which the Common Stock is then listed or quoted;

(vi) use its commercially reasonable efforts to promptly notify the Stockholder and the managing underwriter or underwriters, if any, after becoming aware thereof, (A) when the registration statement or any related prospectus or any amendment or supplement thereto has been filed, and, with respect to the registration statement or any post-effective amendment, when the same has become effective, (B) of any request by the SEC or any U.S. state securities authority for amendments or supplements to the registration statement or the related prospectus or for additional information, (C) of the issuance by the SEC of any stop order suspending the effectiveness of the registration statement or the initiation of any proceedings for that purpose, (D) of the receipt by the Company of any notification with respect to the suspension of the qualification of the Registrable Securities for sale in any jurisdiction or the initiation of any proceeding for such purpose, or (E) within the Effective Period of the happening of any event or the existence of any fact which makes any statement in the registration statement or any post-effective amendment thereto, prospectus or any amendment or supplement thereto, or any document incorporated therein by reference untrue in any material respect or which requires the making of any changes in the registration statement or post-effective amendment thereto or any prospectus or amendment or supplement thereto, in light of the circumstances under which they were made, not misleading;

(vii) during the Effective Period, use its commercially reasonable efforts to obtain, as promptly as practicable, the withdrawal of any order enjoining or suspending the use or effectiveness of the registration statement or any post-effective amendment thereto or the lifting of any suspension of the qualification of any of the Registrable Securities for sale in any jurisdiction at the earliest date reasonably practicable;

(viii) use its commercially reasonable efforts to deliver promptly to the Stockholder and the managing underwriters, if any, copies of all correspondence between the SEC and the Company, its counsel or auditors and all memoranda relating to discussions with the SEC or its staff with respect to the registration statement (except to the extent such correspondence is currently available via EDGAR) and permit the Stockholder to do such investigation with respect to information contained in or omitted from the registration statement as it deems reasonably necessary for the purpose of conducting due diligence with respect to the Company; provided that any such investigation shall not interfere unreasonably with the Company s business;

- 15 -

(ix) use its commercially reasonable efforts to, as promptly as reasonably practicable, provide and cause to be maintained a transfer agent and registrar for all Registrable Shares covered by such registration statement not later than the effective date of such registration statement;

(x) use its commercially reasonable efforts to cooperate with the Stockholder and the managing underwriter or underwriters, if any, to facilitate the timely preparation and delivery of certificates representing the Registrable Securities to be sold under the registration statement in a form eligible for deposit with the Depository Trust Corporation not bearing any restrictive legends (other than as required by the Depository Trust Corporation) and not subject to any stop transfer order with any transfer agent, and cause such Registrable Securities to be issued in such denominations and registered in such names as the managing underwriters, if any, may request in writing or, if not an underwritten offering, in accordance with the instructions of the Stockholder, in each case at least two Business Days prior to any sale of Registrable Securities;

(xi) in the case of a firm commitment underwritten offering, use its commercially reasonable efforts to, as promptly as reasonably practicable, enter into an underwriting agreement customary in form and substance (taking into account the Company s prior underwriting agreements) for firm commitment underwritten secondary offerings of the nature contemplated by the applicable registration statement;

(xii) use its commercially reasonable efforts to, as promptly as reasonably practicable, obtain an opinion from the Company s counsel and a cold comfort letter from the Company s independent public accountants (and, if necessary, any other independent certified public accountants of any Subsidiary of the Company or of any business acquired by the Company for which financial statements and financial data is, or is required to be, included in the registration statement) in customary form and covering such matters as are customarily covered by such opinions and cold comfort letters in connection with an offering of the nature contemplated by the applicable registration statement;

(xiii) use its commercially reasonable efforts to, as promptly as reasonably practicable, provide to counsel to the Stockholder and to the managing underwriters, if any, and no later than the time of filing of any document which is to be incorporated by reference into the registration statement or prospectus (after the initial filing of such registration statement), copies of any such document;

(xiv) cause its officers to fully cooperate with the marketing of the Registered Securities covered by the registration statement, including, at the recommendation or request of the underwriters, making themselves available to participate in road-show, one-on-one, and other customary marketing activities in such locations (domestic and foreign) as recommended by the underwriter(s);

(xv) otherwise use its commercially reasonable efforts to comply with all applicable rules and regulations of the SEC and any applicable national securities exchange; and

(xvi) use its commercially reasonable efforts to comply with the requirements of Rule 144(c)(1) with respect to public information about the Company.

(b) In the event that the Company would be required, pursuant to Section 5.7(a)(vi)(E), to notify the Stockholder or the managing underwriter or underwriters, if any, of the happening of any event specified therein, the Company shall, subject to Section 5.3(c), as promptly as practicable, prepare and furnish to the Stockholder and to each such underwriter a reasonable number of copies of a prospectus supplemented or amended so that, as thereafter delivered to purchasers of Registrable Securities that have been registered pursuant to this Agreement, such prospectus shall not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading. The Stockholder agrees that, upon receipt of any notice from the Company pursuant to Section 5.7(a)(vi)(E), it shall, and shall use its reasonable best efforts to, cause any sales or placement agent or agents for the Registrable Securities and the underwriters, if any, to forthwith discontinue disposition of the Registrable Securities until such Person shall have received copies of such amended or supplemented prospectus and, if so directed by the Company, to destroy all copies, other than permanent file copies, then in its possession

- 16 -

of the prospectus (prior to such amendment or supplement) covering such Registrable Securities as soon as practicable after the Stockholder s receipt of such notice.

(c)(i) If requested by the managing underwriter for an underwritten offering (primary or secondary) of any equity securities of the Company, the Stockholder agrees not to effect any Transfer of any Registrable Shares, including any sale pursuant to Rule 144, and not to effect any Transfer of any other equity security of the Company (in each case, other than as part of such underwritten public offering) during the ten days prior to, and during the 90-day period (or such longer period as the Stockholder agrees with the underwriter of such offering) beginning on, the consummation of any underwritten public offering covered by a registration statement referred to in Section 5.4 if the Stockholder is permitted to include Registrable Shares thereunder.

(ii) The Company hereby agrees that if it shall previously have received a request pursuant to Section 5.1 or Section 5.2 for registration of Registrable Securities in an underwritten offering, and if such previous registration shall not have been withdrawn or abandoned, the Company, if requested by the managing underwriter for such underwritten offering, shall not Transfer to a third party or third parties any Common Stock, any other equity security of the Company or any security convertible into or exchangeable for any equity security of the Company until the earlier of (A) 90 days after the effective date of such registration statement and (B) such time as all of the Registrable Securities covered by such registration statement have been distributed; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, the Company may Transfer Common Stock or such other securities (1) as part of such underwritten offering, (2) pursuant to a registration statement on Form S-8 or Form S-4 under the Securities Act or any successor or similar form, (3) as part of a transaction under Rule 145 of the Securities Act, (4) in one or more private transactions that would not interfere with the method of distribution contemplated by such registration statement, or (5) if such Transfer was publicly announced or agreed to in writing by the Company prior to the date of the receipt of such request pursuant to Section 5.1.

(d) The Stockholder shall furnish to the Company in writing such information regarding the Stockholder and its intended method of distribution of the Registrable Securities as the Company may from time to time reasonably request in writing in order for the Company to comply with its obligations under all applicable securities and other Laws and to ensure that the prospectus relating to such Registrable Securities conforms to the applicable requirements of the Securities Act and the rules and regulations thereunder. The Stockholder shall promptly notify the Company of any inaccuracy or change in information previously furnished by the Stockholder to the Company or of the occurrence of any event, in either case as a result of which any prospectus relating to the Registrable Securities contains or would contain an untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, and promptly furnish to the Company any additional information required to correct and update any previously furnished information or required so that such prospectus shall not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements of a material fact or omit to state a material fact required to be stated therein or necessary to make the statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

(e) In the case of any underwritten offering of shares of Common Stock registered under a Takedown Prospectus Supplement or a Demand Registration Statement, or in the case of a registration under Section 5.4 if the Company has entered into an underwriting agreement in connection therewith, all shares of Common Stock to be included in such offering or registration, as the case may be, shall be subject to the applicable underwriting agreement and no Person may participate in such offering or registration unless such Person agrees to sell such Person s securities on the basis provided therein and completes and executes all questionnaires, indemnities, underwriting agreements and other documents (other than powers of attorney) which must be executed in connection therewith, and provides such other information to the Company or the underwriter as may be reasonably requested to offer or register such Person s Common Stock.



Section 5.8 <u>Registration Expenses</u>. The Stockholder shall bear all agent fees and commissions, underwriting discounts and commissions and fees and disbursements of its counsel and accountants in connection with any registration and sale of any Registrable Securities, including pursuant to Section 5.4. Except as otherwise provided in this Agreement, the Company shall bear all other fees and expenses in connection with any registration statement for the registration of any Registrable Securities, including all registration and filing fees, all printing costs and all fees and expenses of counsel and accountants for the Company.

Section 5.9 Indemnification; Contribution. (a) The Company shall, and it hereby agrees to, indemnify and hold harmless the Stockholder and its controlling Persons, if any, and each underwriter and its controlling Persons, if any, in any offering or sale of the Registrable Securities, including pursuant to Section 5.1, Section 5.2 or Section 5.4, against any losses, claims, damages or liabilities, actions or proceedings (whether commenced or threatened) in respect thereof and expenses (including actual out-of-pocket fees of counsel reasonably incurred) (collectively, \_Claims ) to which each such indemnified party may become subject, insofar as such Claims (including any amounts paid in settlement effected with the consent of the Company as provided herein), or actions or proceedings in respect thereof, arise out of, relate to, are in connection with, or are based upon an untrue statement or alleged untrue statement of a material fact contained in any registration statement, or any preliminary or final prospectus contained therein, or any amendment or supplement thereto, or any document incorporated by reference therein, or arise out of, relate to, are in connection with, or are based upon any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading, and the Company shall, and it hereby agrees to, reimburse periodically the Stockholder or any such underwriter for any actual out-of-pocket legal or other actual out-of-pocket expenses reasonably incurred by them in connection with investigating or defending any such Claims; provided, however, that the Company shall not be liable to any such Person in any such case to the extent that any such Claims arise out of, relate to, are in connection with, or are based upon an untrue statement or alleged untrue statement or omission or alleged omission made in such registration statement, or preliminary or final prospectus, or amendment or supplement thereto, in reliance upon information furnished to the Company by the Stockholder, any underwriter or any Representative of the Stockholder, expressly for use therein, or by the Stockholder s failure to furnish the Company, upon request, with the information with respect to the Stockholder, or any underwriter or Representative of the Stockholder, or the Stockholder s intended method of distribution, that is the subject of the untrue statement or omission.

(b) The Stockholder shall, and hereby agrees to, (i) indemnify and hold harmless the Company, its directors, officers, employees and controlling Persons, if any, and each underwriter, its partners, officers, directors, employees and controlling Persons, if any, in any offering or sale of Registrable Securities against any Claims to which each such indemnified party may become subject, insofar as such Claims (including any amounts paid in settlement as provided herein), or actions or proceedings in respect thereof, arise out of, relate to, are in connection with, or are based upon an untrue statement or alleged untrue statement of a material fact contained in any registration statement, or any preliminary or final prospectus contained therein, or any amendment or supplement thereto, or any document incorporated by reference therein, or arise out of, relate to, are in connection with, or are based upon any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading, in each case only to the extent that such untrue statement or alleged untrue statement or alleged omission or alleged omission furnished to the Company by the Stockholder expressly for use therein, and (ii) reimburse the Company for any actual out-of-pocket legal or other out-of-pocket expenses reasonably incurred by the Company in connection with investigating or defending any such Claim.

(c) The Stockholder and the Company agree that if, for any reason, the indemnification provisions contemplated by Section 5.9(a) or Section 5.9(b) are unavailable to or are insufficient to hold harmless an indemnified party in respect of any Claims referred to therein, then each indemnifying party shall contribute to the amount paid or payable by such indemnified party as a result of such Claims in such proportion as is appropriate to reflect the relative fault of the indemnifying party, on the one hand, and the indemnified party, on

- 18 -

the other hand, with respect to the applicable offering of securities. The relative fault of such indemnifying party and indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact relates to information supplied by such indemnifying party or by such indemnified party, and the parties relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. If, however, the allocation in the first sentence of this Section 5.9(c) is not permitted by applicable Law, then each indemnifying party shall contribute to the amount paid or payable by such indemnified party in such proportion as is appropriate to reflect not only such relative faults, but also the relative benefits of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The parties hereto agree that it would not be just and equitable if contributions pursuant to this Section 5.9(c) were to be determined by *pro rata* allocation or by any other method of allocation which does not take into account the equitable considerations referred to in the preceding sentences of this Section 5.9(c). The amount paid or payable by an indemnified party as a result of the Claims referred to above shall be deemed to include (subject to the limitations set forth in Section 5.10) any actual out-of-pocket legal or other out-of-pocket fees or expenses reasonably incurred by such indemnified party in connection with investigating or defending any such action, proceeding or claim. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

Section 5.10 Indemnification Procedures. (a) If an indemnified party shall desire to assert any claim for indemnification provided for under Section 5.9 in respect of, arising out of or involving a Claim against the indemnified party, such indemnified party shall notify the Company or the Stockholder, as the case may be (the <u>Indemnifying Party</u>), in writing of such Claim, the amount or the estimated amount of Damages sought thereunder to the extent then ascertainable (which estimate shall not be conclusive of the final amount of such Claim), any other remedy sought thereunder, any relevant time constraints relating thereto and, to the extent practicable, any other material details pertaining thereto (a <u>Claim</u> <u>Notice</u>) promptly after receipt by such indemnified party of written notice of the Claim; provided, however, that failure to provide a Claim Notice shall not affect the indemnified party shall deliver to the Indemnifying Party, promptly after the indemnified party shall deliver to the Indemnifying Party, promptly after the indemnified party shall deliver to the Indemnifying Party, promptly after the indemnified party shall deliver to the Indemnifying Party, promptly after the indemnified party shall deliver to the Indemnifying Party, promptly after the indemnified party shall deliver to the Indemnified party relating to the Claim; provided, however, that failure to provide any such copies shall not affect the indemnification obligations provided hereunder except to the indemnified party relating to the claim; provided, however, that failure to provide any such copies shall not affect the indemnification obligations provided hereunder except to the extent the Indemnified party relating to the extent the Indemnified party shall have been materially prejudiced as a result of such failure.

(b) If a Claim is made against an indemnified party, the Indemnifying Party will be entitled to participate in the defense thereof and, if it so chooses, to assume the defense thereof with separate counsel selected by the Indemnifying Party and reasonably satisfactory to the indemnified party. Should the Indemnifying Party so elect to assume the defense of a Claim, the Indemnifying Party will not be liable to the indemnified party for legal expenses subsequently incurred by the indemnified party in connection with the defense thereof, unless the Claim involves potential conflicts of interest or substantially different defenses for the indemnified party and the Indemnifying Party. If the Indemnifying Party assumes such defense, the indemnified party shall have the right to participate in the defense thereof and to employ counsel, at its own expense (except as provided in the immediately preceding sentence), separate from the counsel employed by the Indemnifying Party. The Indemnifying Party shall be liable for the actual out-of-pocket fees and expenses of counsel reasonably incurred by the two immediately preceding sentences. If the Indemnifying Party has not assumed the defense thereof and as otherwise contemplated by the two immediately preceding sentences. If the Indemnifying Party chooses to defend any Claim, the other party shall cooperate in the defense or prosecution thereof. Such cooperation shall include the retention and (upon the Indemnifying Party s request) the provision to the Indemnifying Party of records and information that are reasonably relevant to such Claim, and use of reasonable efforts to make employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Whether or not the Indemnifying Party shall have assumed the defense of a Claim, the indemnifying Party shall not admit any liability with respect to, or settle, compromise or discharge, such Claim

<sup>- 19 -</sup>

without the Indemnifying Party s prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed). The Indemnifying Party may pay, settle or compromise a Claim without the written consent of the indemnified party, so long as such settlement (i) includes an unconditional release of the indemnified party from all liability in respect of such Claim, (ii) does not subject the indemnified party to any injunctive relief or other equitable remedy, and (iii) does not include a statement or admission of fault, culpability or failure to act by or on behalf of any indemnified party.

Section 5.11 <u>Transferee Registration Rights</u>. Any transferee who acquires at least 5% of either the Registrable Shares or of the Registrable Debt pursuant to a transaction that is not registered under the Securities Act (each, a <u>Registrable Securities Transferee</u>) shall be entitled to enjoy the same registration and other rights pursuant to this Article V as does the Stockholder so long as the Registrable Shares or the Registrable Debt held by any such Registrable Securities Transferee may not be sold or disposed of pursuant to Rule 144 without volume limitations at the time when a Registrable Securities Transferee seeks to exercise its rights pursuant to this Agreement. Any Registrable Securities Transferee shall enjoy such right pursuant to this Section 5.11 if and to the extent the Company shall have received (x) written notice from the Stockholder stating the name and address of such Registrable Securities Transferee and identifying the amount of Registrable Shares or Registrable Debt with respect to which such rights under this Article V apply and (y) a written agreement from the Registrable Securities Transferee to be bound by all of the relevant terms of this Article V. In relation to any such Registrable Securities Transferee, the term Stockholder as used in this Article V shall, where appropriate, be deemed to refer to such Registrable Securities Transferee. After such Transfer, the Stockholder, the Company shall execute a registration rights agreement with such Registrable Securities Transferee or a proposed Registrable Securities Transferee substantially similar to the applicable sections of this Article V.

#### ARTICLE VI

#### NON-COMPETITION

Section 6.1 Non-Competition. The Stockholder agrees that (a) for the period commencing at the Closing and expiring on the date that is two years after the first date on which the Stockholder s Voting Percentage is less than 10%, neither the Stockholder nor any of its controlled Affiliates shall, either directly or indirectly, alone or with others, engage in (i) providing wireless telecommunications services through a facilities based network in the Territory, (ii) hold licenses from the FCC related to or necessary to provide such services, (iii) act as a reseller, dealer or distributor in the Territory of such services, or (iv) act as a mobile virtual network operator in the Territory, and (b) for the period commencing at the Closing and expiring on the first anniversary of the termination of the Trademark License in accordance with its terms, manufacture, market or distribute any products or services under, or use in any way, the trademark T-MOBILE in connection with any of the activities described in clauses (a)(i), (iii), (iii) or (iv), other than by the Company and its Affiliates in accordance with the terms of the Trademark License (each of (a) and (b), a <u>Competing Business</u>). The Stockholder further agrees that, during the applicable period set forth in clause (a) or (b), it will not own an interest in (whether as a stockholder, member or partner, but in each case excluding any such interest not exceeding 10% of any class of security), or manage, operate, or control, or act as or have the right to appoint a director of, any Person engaged in a Competing Business (other than the Company and its Subsidiaries) (it being understood that no ownership permitted by this sentence shall be considered to be a breach of any other part of this Section 6.1). For the avoidance of doubt, neither (x) the offering or provision of products or services (including software, apps, and over-the-top services) on, or the conducting of transactions through, mobile or wireless devices or platforms, nor (y) the resale of wireless telecommunications services ancillary to providing information technology outsourcing services shall in any event be deemed to be engaging a Competing Business. If the final judgment of a court of competent jurisdiction declares any term or provision of this Section 6.1 invalid or unenforceable, the parties hereto agree that the court making the determination of invalidity or unenforceability shall have the power



to and shall reform this Section 6.1 to reduce the time, geographic area and/or scope of activity, to delete specific words or phrases, and/or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified.

Section 6.2 <u>Reasonable Scope</u>. The Stockholder acknowledges and agrees that the agreements and covenants set forth in Section 6.1 are (i) necessary to protect the legitimate business interests of the Company, (ii) reasonable as to time, geographic area and scope of activity and do not impose a greater restraint on the activities of the Stockholder than is reasonably necessary to protect such legitimate interests of the Company, and (iii) reasonable in light of the consideration and other value provided, directly or indirectly, to the Stockholder by the Company pursuant to this Agreement and the Business Combination Agreement. The Stockholder hereby waives any and all rights to contest the validity of the agreements and covenants set forth in Section 6.1 on the ground of the reasonableness of the length of their term or the breadth of their geographic area or scope of activity.

#### ARTICLE VII

#### MISCELLANEOUS

Section 7.1 Injunctive Relief. Each party hereto acknowledges that it would be impossible to determine the amount of damages that would result from any breach of any of the provisions of this Agreement and that the remedy at law for any breach, or threatened breach, of any of such provisions would likely be inadequate and, accordingly, agrees that the other party shall, in addition to any other rights or remedies which it may have, be entitled to such equitable and injunctive relief as may be available from any court of competent jurisdiction to compel specific performance of, or restrain any party from violating, any of such provisions. In connection with any action or proceeding for injunctive relief, each party hereto hereby waives the claim or defense that a remedy at law alone is adequate and agrees, to the maximum extent permitted by Law, to have each provision of this Agreement specifically enforced against it, without the necessity of posting bond or other security against it, and consents to the entry of injunctive relief against it enjoining or restraining any breach or threatened breach of such provisions of this Agreement.

Section 7.2 <u>Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors, heirs, legal representatives and permitted assigns. Neither party may directly or indirectly assign any of its rights or delegate any of its obligations under this Agreement, by operation of law or otherwise, without the prior written consent of the other party other than (a) in connection with a change in control of the Stockholder or to any successor of the Company; (b) by the Stockholder in whole or in part, to one or more of its Subsidiaries so long as the Stockholder remains liable for its obligations as contained herein, or (c) in accordance with Section 5.11. Any purported direct or indirect assignment in violation of this Section 7.2 shall be null and void *ab initio*.

Section 7.3 <u>Amendments; Waiver</u>. No amendment, modification or discharge of this Agreement, and no waiver hereunder, and no extension of time for the performance of any of the obligations hereunder, shall be valid or binding unless set forth in writing and duly executed by (a) the Company where enforcement of the amendment, modification, discharge, waiver or extension is sought against the Company or (b) the Stockholder where enforcement of the amendment, modification, discharge, waiver or extension is sought against the Stockholder. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the party granting such waiver in any other respect or at any other time. The waiver by the Company or the Stockholder of a breach of, or a default under, any of the provisions hereof, or to exercise any right or privilege hereunder, shall not be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. Except as

- 21 -

expressly provided in this Agreement, the rights and remedies herein provided are cumulative and none is exclusive of any other, or of any rights or remedies that any party may otherwise have at law or in equity.

Section 7.4 <u>Termination</u>. Except as otherwise provided in this Agreement (including Section 5.6(b)), this Agreement shall terminate at any time after which the Stockholder s Voting Percentage is less than 5%: provided, however, that (a) the obligations under Section 3.2 shall terminate at any time after which the Stockholder s Voting Percentage is less than 30%, (b) the indemnity and contribution provisions contained in Section 5.9 and Section 5.10 shall remain operative and in full force and effect regardless of any termination of this Agreement; and (c) the provisions of Article VI and this Article VII shall survive any termination of this Agreement or any provision thereof. Nothing in this Agreement shall be deemed to release any party from any liability for any willful and material breach of this Agreement occurring prior to any termination hereof or to impair the right of a party to compel specific performance by the other party of its obligations under this Agreement.

Section 7.5 <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party to the others shall be in writing and delivered personally or sent by registered or certified mail, return receipt requested and postage prepaid, or by facsimile (providing confirmation of such facsimile transmission):

if to MetroPCS, to:

MetroPCS Communications, Inc.

2250 Lakeside Blvd.

Richardson, Texas 75082

Attention: Mark A. Stachiw

Melanie Stapp Klint

Fax: (866) 685-9618

with copies to (which shall not constitute notice):

Gibson, Dunn & Crutcher LLP

2100 McKinney Avenue, Suite 1100

Dallas, Texas 75201

Attention: Jeffrey A. Chapman

Robert B. Little Fax: (214) 571-2900

#### and

Akin Gump Strauss Hauer & Feld LLP

1700 Pacific Avenue, Suite 4100

Dallas, Texas 75201

Attention: J. Kenneth Menges, Jr., P.C.

Fax: (214) 969-4343

if to the Stockholder, to:

# Table of Contents

- Deutsche Telekom AG
- Friedrich-Ebert-Alle 140
- 53113 Bonn, Germany
- Attention: General Counsel
- Fax: +49-228-181-74008
- with a copy to (which shall not constitute notice):
- Wachtell, Lipton, Rosen & Katz
- 51 West 52nd Street
- New York, New York 10019
- Attention: Adam O. Emmerich
- Fax: (212) 403-2000

- 22 -

or to such other Persons or addresses as may be designated in writing by the party to receive such notice as provided above.

Section 7.6 Governing Law; Jurisdiction; Forum; Waiver of Trial by Jury.

(a) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE WITHOUT REGARD TO ANY CONFLICT-OF-LAWS RULE OR PRINCIPLE THEREOF. Each party hereto agrees that it shall bring any action or proceeding in respect of any claim arising out of, or related to, this Agreement or the Transaction, exclusively in the Delaware Court of Chancery, New Castle County, or solely if that court does not have jurisdiction, a federal court sitting in the State of Delaware (the <u>Chosen Courts</u>), and solely in connection with claims arising under this Agreement or the Transaction (a) irrevocably submits to the exclusive jurisdiction of the Chosen Courts, (b) waives any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waives any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any party hereto, and (d) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 7.5. Each party hereto irrevocably designates C.T. Corporation as its agent and attorney-in-fact for the acceptance of service of process and making an appearance on its behalf in any such claim or proceeding and for the taking of all such acts as may be necessary or appropriate in order to confer jurisdiction over it before the Chosen Courts and each party hereto stipulates that such consent and appointment is irrevocable and coupled with an interest.

# (b) EACH PARTY HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF, OR RELATING TO, THIS AGREEMENT.

Section 7.7 <u>Interpretation</u>. The table of contents and headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. The parties hereto have participated jointly in the negotiation and drafting of this Agreement and, in the event that an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

Section 7.8 Entire Agreement; No Other Representations. This Agreement and the Business Combination Agreement constitute the entire agreement, and supersede all other prior and contemporaneous agreements, understandings, undertakings, arrangements, representations and warranties, both written and oral, among the parties with respect to the subject matter hereof.

Section 7.9 <u>No Third-Party Beneficiaries</u>. Except as explicitly provided for in Section 5.9, Section 5.10 and Section 5.11, this Agreement is not intended to confer upon any Person other than the parties hereto any rights or remedies hereunder.

Section 7.10 <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision; and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

Section 7.11 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each such counterpart (including any facsimile or electronic document transmission of such counterpart) being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

- 23 -

IN WITNESS WHEREOF, the parties have caused this Agreement to be duly executed by their respective authorized officers as of the date first written above.

# **MetroPCS Communications, Inc.**

By:

Name: Title:

#### **Deutsche Telekom AG**

By: Name: Title:

By:

Name: Title: [Signature Page to Stockholder s Agreement]

## <u>Exhibit D</u>

License Agreement

between

#### **Deutsche Telekom AG**

Friedrich-Ebert-Allee 140, 53113 Bonn, Germany

hereinafter referred to as **DT** 

and

hereinafter referred to as the Licensee

DT and the Licensee hereinafter collectively referred to

as Parties or individually as Party

#### Preamble

 DT is a provider also via subsidiary companies in the whole area of telecommunication, information technology, multimedia, consulting and entertainment as well as the services connected with these areas and related areas in Germany and worldwide.
 DT is owner of various designations. These include trademarks, domain names and company names.

DT has developed a group branding strategy with a joint corporate identity, a joint corporate design, a standardized market appearance and a standardized trademark image. Several of the designations licensed under this Agreement represent the corporate identity of the DT group of companies.

- (2) The Licensee is a provider of wireless telecommunication, broadband and information services, including services and products connected to wireless telecommunication, broadband and information services and related areas in the Territory (as defined herein).
- (3) DT has a majority interest in the Licensee and it is intended that the Licensee shall offer wireless telecommunication, broadband and information, including services and products connected to wireless telecommunication, broadband and information and related areas by using DT s trademarks, domain names and corporate identity in the future. However, it is agreed by the Parties that DT s trademarks and, in particular, the T-Mobile Brand will not be the only brand used by the Licensee and its subsidiaries.

Now, therefore in consideration of the above promises and the terms and conditions contained herein the Parties, intending to be legally bound hereby, agree as follows:

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

# §1 Definitions

For the purposes of this Agreement, whenever used with capitalized initial letter, the following terms shall have the meaning assigned to them below or in any provision of this Agreement, wherein words defined in the singular form shall also apply to the plural correspondingly and vice-versa. In any event the term including shall mean including, but not limited to :

1.1 **Agreement** shall mean this license agreement including all Annexes thereto and all future amendments and supplements on which the Parties may agree from time to time.

- 1.2 **Affiliate** shall mean a legal entity which is under the applicable laws, directly or indirectly, controlling, controlled by or under joint control relative to another company, whereby control shall mean any of (A) direct or indirect ownership of more than 50% of the ownership interest in any other entity, (B) the right to exercise more than 50% of the voting rights in any other entity, or (C) the contractual right to designate more than half of the members of such entity s board of directors or similar executive bodies.
- 1.3 **BAC** shall have the meaning defined in § 5.11.
- 1.4 **Benchmark Analysis** shall have the meaning defined in § 7.4.
- 1.5 **Change of Control** shall have the meaning defined in § 14.3.
- 1.6 **Confidential Information** shall have the meaning defined in § 10.1.
- 1.7 **Date of the Agreement** shall have the meaning defined in § 14.1.
- 1.8 **Designations** shall mean jointly the Trademarks and the Domains.
- 1.9 **Domains** shall mean the Internet domain name registrations listed in **Annex 2**.
- 1.10 **DT Sell-Down** shall have the meaning defined in § 14.3.
- 1.11 **Licensed Products** shall mean all products (including their packaging) that are sold, leased, provided, used for or distributed (including through retailers, dealers, agents, data carriers and online) in connection with the Licensed Services, including (A) cellphones, smartphones, tablet PCs, personal digital assistants, portable media players and any other type of wireless devices and wireless telecommunication. broadband and information devices, (B) SIM cards for the aforementioned devices, (C) accessories for the aforementioned devices and (D) ancillary products for the aforementioned devices.
- 1.12 **Licensed Services** shall mean the marketing, selling and providing of wireless telecommunication, broadband and information services (including voice and data services) and all services ancillary thereto.
- 1.13 **License Fee** shall have the meaning defined in § 7.1.
- 1.14 **Negotiation Period** shall have the meaning defined in § 14.3.
- 1.15 **New License Fee** shall have the meaning defined in § 7.4.
- 1.16 **Revenue Report** shall have the meaning defined in § 7.2.

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

- 1.17 **Subsidiaries** shall have the meaning defined in § 3.2.
- 1.18 **Territory** shall mean the United States of America, Puerto Rico and the territories and protectorates of the United States to the extent the Trademarks in these territories and protectorates are subject to U.S. trademark law.
- 1.19 **Trademarks** shall mean the trademark registrations listed in **Annex 1**.
- 1.20 Utilization Regulations shall have the meaning defined in § 5.3.
- 1.21 **VAT** shall have the meaning defined in § 9.1.

#### § 2 License Grant

- 2.1 DT grants to the Licensee the limited, non-revocable (except as provided herein), and royalty-bearing license to use the Trademarks for the Licensed Services and the Licensed Products in the Territory, including:
  - (A) the offering and provision of the Licensed Services under the Trademarks and the advertising for, and marketing of, the Licensed Services under the Trademarks;

- 2 -

- (B) the application of the Trademarks to the Licensed Products (putting on the market either through retailers, dealers, agents, data carriers or online as download) and the advertising for, marketing of, and using the Licensed Products under the Trademarks;
- (C) the use of the Trademarks as the name of Licensee s stores and in the Licensee s business papers; and
- (D) the use of the Trademarks as part of Internet domains names for websites on which the Licensed Products and Licensed Services are promoted within the Territory;

in each case, subject to the conditions provided hereinafter.

- 2.2 DT grants to the Licensee the limited, non-exclusive, non-revocable (except as provided herein), and royalty-bearing license to use the Domains as Internet addresses for websites on which the Licensed Products and Licensed Services are promoted within the Territory and in the Licensee s business papers. The Licensee may indicate the Domains in its promotion of the Licensed Products and Licensed Services within the Territory as it is common in its business, to the extent that such use is lawful under the laws and regulations in the Territory.
- 2.3 The licenses granted in § 2.1 shall be (A) exclusive (even as to DT) with respect to the Trademarks listed in Annex 1.1 and Domains, provided however, that DT and its Affiliates shall be permitted to grant licenses to use such Trademarks in the Territory in connection with (i) multinational sponsoring activities (e.g., sports teams apparel, but, for the avoidance of doubt, not on Licensed Products) and (ii) multinational co-operation partners provided these co-operation partners do not offer Licensed Services directly or indirectly under the Trademarks listed in Annex 1.1 in the Territory and (B) non-exclusive with respect to the Trademarks listed in Annex 1.2.
- 2.4 For the purposes of § 2.1 and § 2.2 any use of the Designations in connection with the Internet shall not be deemed use outside the Territory, provided:
  - (A) said use does not actively target customers in countries outside the Territory (e.g., by content or choice of language, provided, however, that, for the avoidance of doubt, the Licensee may advertise in Spanish, or other languages, inside the Territory in order to target customers who speak Spanish, or such other language, inside the Territory for offerings of Licensed Products and Licensed Services delivered in the Territory); and
  - (B) the Licensee implements technically reasonable precautions in order to ensure that customers located outside the Territory are not able to access the Licensed Products and Licensed Services (e.g., by checking the address of customers, if available, or prohibiting customers located outside the Territory from accessing the Licensed Services, provided, however, that, for the avoidance of doubt, customers located inside the Territory who are temporarily outside the Territory may use the Licensed Products and access the Licensed Services through roaming agreements with the Licensee or third party carriers).
- 2.5 In addition, DT permits the Licensee free of charge to use the trademark T-Mobile as a company name. DT shall not assert prohibition claims against this insofar as the trademark T-Mobile is used in accordance with the provisions of this Agreement.
- 2.6 Any use of the Designations by the Licensee or its sub-licensees shall inure to the benefit of DT and qualifies as use by DT for the purposes of acquiring and maintaining rights in these Designations. The Licensee is obliged to transfer to DT all rights acquired by using the Designations with effect as of the day such right came into existence. The Licensee hereby in advance transfers to DT all rights resulting from the use of the Designations, as far as legally possible. Any trademark rights resulting from such use shall become Designations as of the date they come into existence. Each Party shall notify the other about any rights arising out of the use of any Designations under this Agreement to the best of its knowledge.

# - 3 -

2.7 The Parties may at any time agree on an amendment of the portfolio of licensed Designations. Such agreement has to be made in writing in order to be valid.

#### § 3 Sublicenses

- 3.1 Except for the rights granted in the following § 3.2 and § 3.3 the Licensee is only entitled to assign the licenses granted herein, including individual Designations, to third parties or to grant sub-licenses and/or licenses after DT has given its prior written consent to such assignment or sub-license.
- 3.2 The Licensee shall be entitled to enter into sub-license agreements with companies in which the Licensee holds more than 50% of the shares and the voting rights ( **Subsidiaries** ). Any sub-licensing by the Licensee to a Subsidiary shall only be effective if (A) the Licensee imposes on the respective Subsidiary its obligations vis-à-vis DT under this Agreement, and (B) the sub-license agreement provides that the sub-license terminates automatically if this Agreement is terminated, and (C) the respective agreement with the Subsidiary directly authorises DT as third-party beneficiary to enforce all claims of the Licensee against the Subsidiary (provided however that DT first attempts to enforce such claim against the Licensee as described below), and (D) the Licensee shall cause compliance by the sub-licensee with the contractual obligations by contractual provisions and contract management, and (E) the Licensee provides a copy of any such sub-licensing agreements to DT immediately after execution for information. Before DT asserts the rights granted to it vis-à-vis the sub-licensee against the sub-licensee, DT will notify the Licensee and attempt to assert such rights against the Licensee, taking into consideration the Licensee s interests and any further proceedings, for a period of 30 days. If DT and the Licensee are unable to resolve DT s claim after such period of time, then DT may assert such rights against the sub-licensee.
- 3.3 The Licensee shall be entitled to enter into sub-license agreements with affiliated and non-affiliated companies which offer Licensed Services and Licensed Products on the basis of the Principles for Sub-license Agreements with distribution partners set out in **Annex 3**.

#### § 4 Quality requirements and quality control

- 4.1 The Licensee shall use the Designations only to mark Licensed Services and Licensed Products conforming to the usual market standards for wireless telecommunication, broadband and information providers in the Territory and the Utilization Regulations (as described in § 5.3). The Utilization Regulations result from the guidelines and documentation as provided by the responsible departments and committees of DT or by the committees formed of representatives of different companies of DT Group and employed by DT for this purpose. If no Utilization Regulations are provided by DT, the Licensee is obliged to use as its quality standard similar products or services by national providers of wireless telecommunication, broadband and information services and products. If the Licensee is uncertain as to whether a specific product or service meets the above-mentioned Utilization Regulations, the Licensee shall obtain DT s binding opinion on the matter. For products that the Licensee obtains from DT, it is assumed that the above-mentioned Utilization Regulations are met.
- 4.2 For quality control purposes, the Licensee shall inform DT at reasonable intervals upon DT s request regarding the Licensee s current portfolio of Licensed Products and Licensed Services, which are bearing or offered under the Designations, and shall provide samples of Licensed Products free of charge in a quantity necessary for verification, as well as a representative sample of the contents of the Licensed Services. If DT does not object to the quality of the Licensed Products or Licensed Services concerned within 14 days of receipt of the representative samples of the Licensed Products or Licensed Services, the quality requirements specified in § 4.1 shall be deemed fulfilled in respect of these samples. The obligation to send samples shall not apply to products that the Licensee obtains from DT, or that were previously checked by

DT. For the purposes of quality control, DT is entitled to enter all premises of the Licensee upon prior written notice, during normal business hours and in a manner that does not disrupt Licensee s business operations, to check compliance with the quality requirements specified in § 4.1. DT shall cause the DT representatives to adhere to, whenever on the Licensee s premises, all applicable Licensee rules, regulations and policies.

#### § 5 Use of the Designations

- 5.1 The Licensee shall use the Designations only as registered. For any use of the Designations, the Licensee has to make a statement, as far as is reasonable, that they are registered trademarks or corresponding other designations of DT, if this is necessary in order to:
  - (A) meet the legal requirements in force for the respective Designation (if the law prescribes such an indication), or
  - (B) maintain legal protection of the trademark in accordance with the applicable law for the respective Designation, or to make this protection more efficient, wherein for such purposes the Licensee may use the Trademarks in connection with the symbol<sup>®</sup> instead of such statement where this is sufficient under the law in the applicable Territory to maintain the legal protection of the respective Designations.

If the Licensee is using Designations as Internet domain names, it is sufficient to make the required statement on the associated website. DT can waive the Licensee s obligation set forth in this § 5.1 in writing.

- 5.2 The Licensee shall not use the Designations in any manner that is inaccurate, distasteful, or that disparages DT or any Affiliate of DT. In particular, the Licensee shall not use the Designations in connection with any content which is pornographic, racist, glorifying violence or which contains political statements.
- 5.3 The Licensee shall comply with the guidelines on the use of the Designations published by DT and enclosed as Annex 4 (the Utilization Regulations ). The version of the Utilization Regulations that is valid at the signing of the Agreement is set forth in Annex 4. Otherwise, the respective current Utilization Regulations are available electronically in the CI-Net of DT s intranet and DT shall provide the Licensee access to such Utilization Regulations. DT is free to change the Utilization Regulations, upon notice to the Licensee. Changes to the Utilization Regulations shall become effective 90 days after transmission of a new version of the Utilization Regulations to the Licensee (provided DT indicates the changes made to the previous version of the Utilization Regulations by the new version of the Utilization Regulations) or, if the Parties agree on the new version of the Utilization Regulations, with immediate effect following agreement by the Parties. For materials that have already been produced (i.e., Licensed Products or advertising materials) the Licensee may continue to use such Licensed Products perpetually and such advertising materials for a transitional use-up period of 180 days after the new version of the Utilization Regulations applicable to the Designations at the time the materials were produced.
- 5.4 Where applicable, the Licensee is insofar as TV advertising campaigns (which shall include any promotional motion pictures, including motion pictures on the Intenet) are concerned obliged to adhere to the approval process as specified in the current version of DT s tool Campaign World available at https://campaignworld.t-mobile.de. In particular, the Licensee is obliged to upload the respective milestones of each and every TV campaign (currently Briefing, Concept, Creatives) as soon as possible. DT will provide the Licensee with its specific log-in data. If DT does not object to the TV advertising campaign concerned within 3 working days after receipt of the submission confirmation, the TV advertising campaign shall be deemed approved. DT is entitled to modify the provisions of the Campaign World at any time and to any extent. DT will notify the Licensee about any amendments on the websites of the Campaign World. Any amendments of the approval process as provided for in the Campaign World will not effect the

implementation of already approved advertisement campaigns. If the Utilization Regulations are amended after an advertising campaign was approved, the provisions of § 5.3 apply accordingly. Any other advertising campaign has to be submitted as further specified in § 5.5.

- 5.5 Each and every advertisement measure and each advertising campaign other than TV advertising campaigns has to be submitted on the Approval Tool available on DT s CI-Net and as described in Annex 5. Advertisement measures shall include any publications and presentations in print and online media, radio, MMS and SMS in connection with the Licensed Services and Licensed Products. The approval of the advertisement measures and advertising campaigns, other than TV advertising campaigns, submitted by the Licensee on the Approval Tool will be given by DT within 48 hours after the receipt of the submission conformation, wherein only working days shall be considered for this 48 hour period. If DT does not object to the advertisement measures or advertising campaign shall be deemed approved. DT is entitled to modify the provisions of the Approval Tool at any time and to any extent. DT will notify the License about any amendments on the websites of the Approval Tool. Any amendments of the approval process as provided for in the Approval Tool will not effect the implementation of already approved advertisement campaigns. If the Utilization Regulations are amended after an advertising campaign was approved, the provisions of § 5.3 apply accordingly.
- 5.6 The Licensee is permitted to develop, apply for, register, acquire and use designations other than the Designations that are subject to this Agreement (Own Designations), including trademarks, business designations and domain names. This also includes the acquisition of companies or affiliated companies that have their own trademarks, business designations and domain names. Own Designations must not be identical or confusingly similar to existing or registered designations of DT, unless DT agrees to such use or registration in advance in writing.

The Licensee is permitted to use its Own Designations in connection with any of the Designations that are subject to this Agreement upon prior written consent of DT. As of the Date of the Agreement, DT pre-approves the Licensee to use its Own Designations with the Designations as described in **Annex 6**. The Parties intend to promote the Licensed Products and Licensed Services by using the Own Designations and Designations together. The Parties shall meet and agree on guidelines for the combined use of the Designations with the Licensee s Own Designations promptly after execution of this Agreement.

- 5.7 If the Licensee acquires or obtains rights to use trademarks or other designations of third parties, the Licensee is solely responsible for the use of such trademarks or designations.
- 5.8 In the event the Licensee uses or publishes an advertisement measure, advertising campaign or TV or motion picture advertising campaign despite the fact that DT refused to give its approval pursuant to § 5.4 or § 5.5, and the Licensee fails to cure such breach within 5 working days after notice thereof, DT s sole and exclusive remedy shall be for the Licensee shall pay compensatory damages of (A) with respect to § 5.4, USD 100,000 (in words: one hundred thousand) and (B) with respect to § 5.5, USD 50,000 (in words: fifty thousand), in each case, unless the Licensee can prove lesser damages or DT can prove higher damages.
- 5.9 The Licensee shall keep record and provide a documentation of its use of the Designations to a reasonable extent and within the framework of the Licensee s business processes; and such documentation is to be provided to DT upon DT s request to enable DT to prove the use of the Designations by the Licensee.
- 5.10 The Licensee shall be fully responsible and liable for compliance with its obligations under this Agreement with respect to use of the Designations and advertising campaigns, and the Licensee shall also be liable in this respect for all acts and omissions of its sub-licensees.
- 5.11 The Parties shall establish a brand advisory committee (**BAC**) comprising two representatives of each Party. The object of the BAC is to review the use of the Designations and the implementation of the trademark Utilization Regulations by the Licensee, the handling and

# Table of Contents

implementation of the approval

- 6 -

processes set forth in § 5.4 and § 5.5, the market communication of the Licensee, the communication between the Parties and possible options to improve such communication, the use of Licensee s Own Designations with DT s Trademarks, and the potential of the license to possible further trademarks of DT.

Each Party shall appoint its representatives by written declaration, fax or by e-mail to the other Party. Each Party may change its representatives at any time by written declaration, fax or by e-mail to the other Party.

The BAC shall meet at least once per calendar half year, preferably in January and July. Meetings of the BAC shall be held by videoconference. Each Party shall be entitled to convene a meeting of the BAC with two weeks notice. The Party convening the meeting shall also set up the videoconference and provide dial-in information.

#### § 6 Exclusion and Limitation of Liability

- 6.1 DT represents and warrants that it has the right and authority to grant the license granted in § 2. With respect to the Trademarks listed in **Annex 1.1** and Domains, DT represents and warrants that Licensee s use of such Trademarks and Domains do not and will not infringe, violate or misappropriate any third party intellectual property rights under the condition that such use is in accordance with the terms of this Agreement and the Licensee permits DT to take over the Licensee s defence of such claim as provided in § 13.1. With respect to the Trademarks listed in **Annex 1.2**, DT shall not be liable if the use of such Trademarks infringe, violate or misappropriate any third party intellectual property rights.
- 6.2 With respect to the Trademarks listed in **Annex 1.1** and Domains, DT shall indemnify, defend and hold harmless and release the Licensee from any claims raised by third parties that are asserted due to the infringement, violation or misappropriation of any third party intellectual property rights by the Designations under the condition that Licensee s use of such Designations is in accordance with the terms of this Agreement and the Licensee permits DT to take over the Licensee s defence of such claim as provided in § 13.1.
- 6.3 With respect to the Trademarks listed in **Annex 1.2**, in the event DT elects to defend the Licensee against a third party claim under § 13, DT shall indemnify, defend and hold harmless and release the Licensee from any claims raised by third parties that are asserted due to the infringement, violation or misappropriation of any third party intellectual property rights by the Designations.
- 6.4 The Licensee shall indemnify, defend and hold harmless and release DT from any claims raised by third parties that are asserted due to advertising claims of the Licensee for the Licensed Services and Licensed Products offered under the Designations and/or due to anti-competitive use of the Designations.
- 6.5 In any event, the mutual liability of the Parties is limited to EUR 1 million per calendar year. This limitation does not apply to cases of intentional conduct or a Party s indemnification obligations under § 6.2, § 6.3 and § 6.4.

#### §7 License Fees

7.1 In consideration for the licenses granted hereunder, Licensee shall pay to DT a license fee in an amount equal to 0.25% of the Licensee s and its sub-licensee s net revenues generated by the Licensed Products and Licensed Services sold or offered under the Designations (License Fee).

For purposes of this § 7, the term net revenues shall mean all revenues generated by Licensee and its sub-licenses in connection with the sale of Licensed Products and the provision of Licensed Services, including all inbound national and international roaming revenue generated by subscribers of other mobile telecommunication, broadband and information providers due to such subscribers use of the Licensed Services sold or offered under the Designations. For the avoidance of doubt, net revenues excludes any revenues generated by Licensee and its sub-licenses in connection with (A) the sale of Licensed Products or

- 7 -

Licensed Services sold or offered solely under the Own Designations or (B) the sale of Licensed Products or Licensed Services under the Own Designations, wherein in the same advertising medium the Designation is used as the name of the company only (e.g., a non-prominent use of the company name T-Mobile in a legal disclaimer) and not otherwise used as a trademark (e.g., by using the T-Mobile logo or powered by T-Mobile or brought to you by T-Mobile ). Examples of such usage are set forth in **Annex 7**.

- 7.2 Within 4 weeks after the end of each quarter calendar year, the Licensee shall provide DT with a report showing all data necessary for the computation of the License Fee, wherein the Licensee shall provide detailed reports of its own revenues, the revenues of distributions partners and revenues of its sub-licensees with respect to the Licensed Services and Licensed Products ( **Revenue Report** ). Moreover, the Revenue Report shall specify the revenues with each individual Licensed Product and Licensed Service sold or offered under the Designations. The License Fee resulting from the Revenue Reports shall be invoiced to the Licensee.
- 7.3 The License Fee for each calendar quarter shall be due 6 weeks after the end of the respective calendar quarter and after the Licensee s receipt of a DT issued proper invoice complying with the applicable tax laws.
- 7.4 After five years the Parties shall meet and attempt to agree on an adjustment of the License Fee ( New License Fee ) to the then current commercial royalty rates applicable to similar licenses for the combined use of trademarks, corporate brands and domain names in the field of wireless telecommunication, broadband and information products and services in the Territory. For these purposes, DT shall provide until December 31, 2017 a benchmark analysis performed by an accounting firm or a market research institution and providing an analysis of the then current, average commercial royalty rates charged in the field of wireless telecommunication, broadband and information products and services in the Territory ( Benchmark Analysis ). On the basis of the Benchmark Analysis submitted by DT, the Parties shall determine the then current, average commercial royalty rates charged in the field of wireless telecommunication, broadband and information products and services in the Territory. If the Parties agreed to a New License Fee, the New License Fee determined by the Parties shall be applicable as of January 1, 2019.

If the Parties fail to agree to the New License Fee by March 31, 2018, the Parties shall jointly appoint an accounting firm that is specialized in analyzing royalty rates in the field of trademark licenses. No later than July 31, 2018, the appointed accounting firm shall make a determination of the then current average commercial royalty rates applicable to similar licenses for the combined use of trademarks, corporate brands and domain names in the field of wireless telecommunication, broadband and information products and services in the Territory. The accounting firm s determination shall be based on an independent benchmark analysis performed by the accounting firm. The New License Fee determined by the accounting firm shall be applicable as of January 1, 2019. Each of the Parties shall bear half of the costs of the accounting firm.

If the Parties fail to agree on an accounting firm as provided in the preceding sub-paragraph, the dispute between the Parties about the New License Fee applicable as of January 1, 2019 shall be referred to and finally determined by arbitration in accordance with the WIPO Arbitration Rules in force on the date of the commencement of the arbitration. The arbitral tribunal shall consist of one arbitrator, who must have experience in the field of trademark and licensing law, and who must be fluent in the language of the proceedings. The place of arbitration shall be New York, New York, United States. The language to be used in the arbitral proceedings shall be English.

7.5 Every further five years the Parties shall again review the License Fee and adjust the License Fee according to the procedure set forth in § 7.4, i.e. the Parties shall again determine a New License Fee that is applicable for the periods as of 1 January 2024, 2029 etc. For this procedure, the periods set forth in § 7.4 shall apply *mutatis mutandis*, wherein the years stated in § 7.4 shall be adjusted accordingly.

- 8 -

#### §8 Right to Audit

DT shall have the right to have the Licensee s books examined in reasonable intervals upon prior written notice (and no more than once every 12 months), and a last time one year after the termination of this Agreement, for conformance with the Revenue Reports by an independent financial auditor who is obligated to maintain secrecy even with respect to DT. The auditor shall be permitted to inform DT if there are discrepancies between the Revenue Reports and the findings of the auditor, in which case the auditor is permitted to disclose to DT all information required for complete and correct Revenue Reports for the period under review. The cost of such book inspection shall be borne by DT, unless the examination shows that the Licensee s Revenue Reports under review contained discrepancies to the disadvantage of DT of more than 10 percent of the License Fee owed for the period under review, in which case the costs of the audit shall be borne by the Licensee.

#### §9 Taxes

- 9.1 The agreed prices and the License Fee are net prices exclusive of sales tax/VAT (**VAT**). Any amounts of VAT that accrue are to be borne by the Licensee.
- 9.2 Since the Licensee has its principal place of business outside of Germany, the following provisions shall apply:
  - (A) All taxes, fees, and other charges incurred by the Licensee in connection with concluding and performing this Agreement are at the expense of the Licensee.
  - (B) The prices agreed are net prices. Incurred VAT or use and sales taxes shall be borne by the Licensee.
  - (C) The Licensee shall not be entitled to withhold taxes for DT s account from the contractually agreed payments to DT, in particular withholding taxes that may apply under the laws of the country where the Licensee s registered office is located. Any taxes withheld at source shall be paid by the Licensee. Taxes for DT s account may be withheld only if and to the extent that DT may offset the Licensee s accrued and withheld income and corporate tax liabilities according to the law of the country of residence of DT against its income or corporate tax liabilities or cause it to be deducted when DT s taxable income is determined. Upon request of the Licensee, DT shall notify the Licensee at the beginning of each calendar year if such an offset or such a deduction is possible. A tax deduction shall also not be to the Licensee s detriment, if the tax deduction is due only because of DT s culpable (intentional or negligent) failure to file or delay in filing any application for exemption from or reduction of the withholding tax provided for by the law of the respective country of residence of the Licensee.
- 9.3 The Licensee shall support DT in fulfilling tax obligations and formalities.
- 9.4 In the case of sub-licensing to a Subsidiary or any other third party located outside of Germany, the Licensee shall include the provisions contained in § 9.2 and § 9.3 in the sub license agreement in such way that the sub-licensee has the same obligations that apply to the Licensee under § 9.2 and § 9.3.

#### §10 Confidentiality

10.1 The Parties agree that (A) all information disclosed by the disclosing party to the receiving party, whether in oral form, visual form or in writing, including all specifications, formulas, prototypes, computer programs and any and all records, data, ideas, methods, techniques, processes and projections, plans, marketing and advertising campaign information, materials, financial statements, memoranda, analyses, notes, legal documents and other data and information (in whatever form), as well as improvements, patents (whether pending or duly

# Table of Contents

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

registered) and any know-how related thereto, (B) information learned by the receiving party from the disclosing party through the inspection of the disclosing party s property, that relates to disclosing party s products, designs, business plans, business opportunities, finances, research,

development, know-how, personnel, or third-party confidential information, (C) information regarding the other Party s business operations and (D) the terms and conditions of this Agreement, in each case, will be considered and referred to collectively in this Agreement as **Confidential Information**.

Notwithstanding the foregoing, a Party s obligation to keep Confidential Information confidential under § 10.2, shall not apply to information that:

- (i) was known to the recipient before disclosure by the other party,
- (ii) is or becomes publicly known without breach of the confidentiality obligations set forth in this Agreement, or
- (iii) is disclosed pursuant to the order or requirement of a court, administrative agency, or other governmental body; provided, however, that the recipient shall make reasonable efforts to provide prompt notice of such court order or requirement to the disclosing Party to enable the disclosing Party to seek a protective order or otherwise prevent or restrict such disclosure.
- 10.2 Each Party shall keep confidential the Confidential Information of the other Party during the term of this Agreement (including the conditions of this Agreement) in a manner at least consistent with the procedures that it uses to protect its own confidential information but in no event less than a reasonable degree of care. Each Party may make the Confidential Information of the other Party available only to employees for whom that knowledge is absolutely necessary for fulfilment of their duties in the context of implementing this Agreement, and undertakes to impose on those employees a written secrecy obligation extending even beyond the duration of their work for the respective Party.
- 10.3 The Licensee may reveal Confidential Information in order to maintain or to accomplish the rights granted herein, including any national trademark register requirements.
- 10.4 The Licensee is entitled to publish Confidential Information that is required for compliance with or efficient implementation of the rights granted, including any national trademark register requirements. The Licensee must notify DT of these requirements promptly and must collaborate with DT in order to limit the Confidential Information to be published as far as possible.
- 10.5 All Confidential Information, and any derivatives thereof is and shall remain the property of the Party disclosing such Confidential Information and no license or other rights to Confidential Information is granted or implied hereby to have been granted to the receiving Party, now or in the future.
- 10.6 The disclosing Party may decide to discontinue the disclosure of Confidential Information under this Agreement, at will, with or without cause, by giving written notice to the receiving Party, with immediate effect. Upon such termination or upon request of the disclosing Party, or upon the receiving Party s determination that it no longer has a need for such Confidential Information, or upon the termination of this Agreement, the receiving Party shall (A) return to the disclosing Party any information disclosed in any tangible form, and all copies thereof (on whatever physical, electronic or other media such information may be stored) containing any of the Confidential Information (and if such Confidential Information is stored in electronic form, it is to be immediately deleted); and (B) provide a certification, in writing, executed by an appropriate officer of the receiving Party, that it has retained no copies of the Confidential Information. The obligations set forth herein regarding confidentiality and use of Confidential Information shall survive any expiration or termination of this Agreement.

#### § 11 Maintaining and Defending the Designations

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

11.1 DT shall maintain the registration of the Designations at its own expense throughout the term of this Agreement and defend the Designations against any attacks by third parties, subject to the following § 11.2. DT has, however, no obligation to use the Designations.

- 10 -

11.2 DT is entitled not to maintain (entirely or partially) the protection of individual Designations or to cancel a Designation s registration, in particular if the maintenance of such registration of a Designation is legally impossible or contrary to significant economic interests of DT. If DT decides not to maintain (entirely or partially) the protection of individual Designations or to cancel a Designation s registration, DT shall provide the Licensee with notice of such decision prior to any such registration lapsing and afford the Licensee the opportunity to maintain such registration in the Licensee s name at Licensee s expense, assigning to Licensee for no additional consideration any such Designation to their registration lapsing, if requested by Licensee; provided, however DT shall not be obligated to assign a Designation to Licensee if the Designation that DT intends not to maintain or to cancel contains elements of another designation owned by DT.

#### § 12 Infringement of the Designations by Third Parties

- 12.1 If either Party discovers or otherwise learns that a third party is using a company name, designation, brand or trademark in the Territory in a manner likely to cause confusion with a Designation or that may infringe DT s or the Licensee s rights in such Designation (in each case, as reasonable determined by DT or the Licensee, as applicable), and/or registers it as a trademark, it shall promptly notify the other Party in writing.
- 12.2 DT alone shall be entitled to take action against the use and/or registration of a trademark, company name, brand or designation that could be confused with a Designation. DT shall also have full discretion whether to settle all claims that it may have against third parties for alleged infringement of the rights in the Designations (provided, however that DT shall not consent to the entry of any judgement or enter into any settlement of any such claim in the event such judgement or settlement imposes any liability, restriction or obligation on the Licensee without the Licensee s prior written consent).
- 12.3 The Licensee shall authorize DT to assert any and all claims of the Licensee against third parties, including claims for an injunction and damages, on grounds that the respective third party violates the rights of the Licensee to use the Designations. DT shall regularly consult with the Licensee with respect to such action. If DT asserts claims of the Licensee in its own name in court, it will inform the Licensee of this in regular intervals and the proceeds from the enforcement of the Licensee s claims less DT s costs incurred with respect to such enforcement, shall be for the Licensee s account. The Licensee shall have the right to be represented in connection with such action by its own legal counsel, at its own expense; provided, however, that such legal counsel shall act only in an advisory capacity.

#### § 13 Third-party attacks on use of the Designations

13.1 If action is taken against the Licensee by a third party to cease and desist and/or for compensatory damages due to the marking of Licensed Services and/or Licensed Products using Designations, or any other use of Designations in conformance with this Agreement, the Licensee shall promptly notify DT by e-mail or in writing. With respect to the Trademarks listed in **Annex 1.1** and Domains, DT shall defend the Licensee in accordance with § 6.2. With respect to the Trademarks listed in **Annex 1.2**, DT may at DT s option defend the Licensee in accordance with § 6.3. In those instances where DT defends the Licensee, (A) the Licensee hereby grants DT a power of attorney to defend the Licensee against any such claims in court and out of court (provided, however that DT shall not consent to the entry of any judgement or enter into any settlement of any such claim in the event such judgement or settlement imposes any liability, restriction or obligation on the Licensee without the Licensee is obliged to support DT to the best of its abilities in the described defence of the claims asserted against the Licensee and (C) the Licensee shall have the right to be represented in connection with such claim by its own legal counsel, at the Licensee s expense; provided, however, that such legal counsel shall act only in an advisory capacity.

- 13.2 The Licensee remains obligated to pay the agreed License Fee in the event of an attack by a third party against the use of the Designations. If DT does not defend Licensee against such action then the Licensee may defend itself against such action (A) at DT s cost and expense with respect to the Trademark listed in **Annex 1.1** and Domains and (B) at the Licensee s cost and expense with respect to the Trademark listed in **Annex 1.1** and Domains and (B) at the Licensee s cost and expense with respect to the Trademark listed in **Annex 1.1** and Domains and (B) at the Licensee to execute all documents to permit and enable the Licensee to conduct such defence, (ii) DT is obliged to support the Licensee to the best of its abilities in the described defence of the claims asserted against the Licensee and (iii) DT shall have the right to be represented in connection with such claim by its own legal counsel, at the DT s expense; provided, however, that such legal counsel shall act only in an advisory capacity.
- 13.3 DT shall inform the Licensee of attacks by third parties regarding the Designations in the Territory and will regularly provide information on the further progress of corresponding proceedings.

#### § 14 Duration and termination of the Agreement

- 14.1 This Agreement shall enter into force on [] ( **Date of the Agreement** ). This Agreement is in effect until December, 31 2018. The term of this Agreement shall be extended until December 31, 2023, unless the Licensee gives notice of termination until December 31, 2017. After December 31, 2023 this Agreement shall be extended by periods of five years each, unless it is terminated by the Licensee with a notice period of at least 12 months prior to its respective expiry date. Notwithstanding the foregoing, the Licensee may terminate this Agreement for convenience at any time upon written notice and such termination shall be effective on the first anniversary after the Licensee s notice of its exercise of such right.
- 14.2 Each Party shall be entitled to terminate this Agreement for cause if the other Party commits a material breach of any provision of this Agreement, including the Utilization Regulations in **Annex 4**, and fails to cure such breach within 30 days after notice thereof is given to the breaching Party by the non-breaching Party. DT shall have the right to terminate this Agreement for cause if the Licensee has ceased to genuinely use the Trademarks listed in **Annex 1.1** for a period of six consecutive months and in DT s reasonable discretion the Licensee s usage is insufficient to maintain the Trademark registration, provided that DT provides prior notice of such insufficient Trademark usage and Licensee fails to cure such insufficient usage within a period of 6 months after such notice.

In the event a Party terminates this Agreement for cause, this shall not affect the terminating Party s entitlement to compensation or damages, including claims for breach of contract.

14.3 In the event of (A) any transaction or event, immediately prior to which DT and its subsidiaries has more than 50% of the voting power of the outstanding shares of capital stock of the Licensee, and immediately after which DT and its subsidiaries has 50% or less of the voting power of the outstanding shares of capital stock of the Licensee (a **DT Sell-Down**), or (B) any transaction or event as a result of which or following which any third party or group owns or controls, directly or indirectly, beneficially or of record, 50% or more of the voting power of the outstanding shares of voting stock or of capital stock of the Licensee, or otherwise has the power to direct or cause the direction of the management and policies of the Licensee, whether through the ownership of voting securities, by contract, or otherwise (a Change of Control), whichever Party has notice of the transaction shall notify the other Party accordingly. As of the date on which such notice is given, the Parties shall in good faith negotiate for a period of up to 6 months ( Negotiation Period ) on the terms of a new license agreement taking into account the new shareholder structure of the Licensee and DT s need to sufficiently control an appropriate use of the Designations. If the Parties fail to reach an agreement on the terms and conditions of a new license agreement within the Negotiation Period, either Party shall have the right to terminate this Agreement, such right to be exercised within three months after the end of the Negotiation Period. Such termination by a Party shall become effective (i) in the case of a DT Sell-Down in the absence of a Change of Control, on the third anniversary of a Party s notice of its exercise of such right and (ii) in the case of a Change of Control (whether or not a DT Sell-Down shall also have occurred), on the second anniversary of DT s notice of its exercise of such right.

- 14.4 DT is permitted to assign its rights and obligations under this Agreement in full to an Affiliate upon notice to the Licensee.
- 14.5 After termination or expiration of this Agreement, the Licensee shall no longer be entitled to use the Designations, provided however that (A) with respect to Licensed Products manufactured prior to such termination or expiration, the Licensee shall be entitled to deliver such manufactured Licensed Products and render the Licensed Services with respect to such Licensed Products for a period of one year after termination or expiration and (B) with respect to advertising incorporating the Designations, the Licensee shall be entitled to continue using such advertising materials bearing the Designations for up to six months after the termination or expiration of this Agreement.
- 14.6 Upon termination or expiration of this Agreement, all documents and records provided to the Licensee by DT, including any and all copies and reproductions of the Designations, must be returned to DT.
- 14.7 Upon termination or expiration of this Agreement, the Licensee shall assign to DT free of charge all trademarks, Internet domains and any rights in such designations owned by the Licensee, in each case, if they contain any of the Designations, parts of the Designations or are confusingly similar to a Designation. DT may request such assignment within one month after the Licensee has informed DT about such rights and registrations. In the event of an assignment of rights, DT shall bear the reasonable, external costs of such assignment. The Licensee undertakes to procure that its Affiliates and sub-licensees also assign any such rights to DT, as necessary.

#### § 15 Jurisdiction and Venue, Choice of Law

- 15.1 This Agreement shall be governed and construed in accordance with the laws of Germany, excluding conflict of laws rules.
- 15.2 In case of any disputes arising in connection with this Agreement such dispute shall be referred to and finally determined by arbitration in accordance with the WIPO Arbitration Rules in force on the date of the commencement of the arbitration. The arbitral tribunal shall consist of three arbitrators (except for the proceedings mentioned in § 7.4, which shall consist of one arbitrator), who must have experience in the field of trademark and licensing law, and who must be fluent in the language of the proceedings. The place of arbitration shall be London, UK. The language to be used in the arbitral proceedings shall be English.

#### §16 Miscellaneous

- 16.1 Any changes and amendments to this Agreement, including any waiver of this clause, must be made in writing and signed by both parties in order to be valid.
- 16.2 This Agreement represents the entire understanding with respect to the subject matter of this Agreement and it supersedes all prior and contemporaneous agreements between the Parties concerning the subject matter of this Agreement.
- 16.3 This Agreement is binding and authorizes the Parties and their legal successors. The Parties undertake to impose their obligations under this Agreement also on their legal successors.
- 16.4 Should one of the provisions of this Agreement be or become invalid, the validity of the remaining provisions shall not be affected thereby. The Parties shall be required to replace the invalid provision by a valid one best achieving the economically desired result. The same applies in the event of a gap.

16.5 This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which taken together will constitute one single agreement between the Parties.

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

# Table of Contents

(Place, Date)	Deutsche Telekom AG Signed by: Position:
(Place, Date)	Deutsche Telekom AG Signed by: Position:
(Place, Date)	Licensee Signed by: Position:
(Place, Date)	Licensee Signed by: Position:

- 14 -

## Annex 1 Trademarks

# Annex 1.1 T-Mobile Trademarks

[NOTE: Parties to add schedule of T-Mobile registrations prior to closing]

# Annex 1.2 Other Trademarks

[NOTE: Parties to add schedule of other trademarks used by T-Mobile US and owned by DT prior to closing]

- 15 -

# Annex 2 Domains

[NOTE: Parties to add schedule of domain names prior to closing]

- 16 -

#### Annex 3 Principles for Distribution Partner Agreements

(1) The Licensee is permitted to grant to a distributor ( **Distributor** ) a non-exclusive, non-transferable and non-sub-licensable license for the offering and selling of the Licensed Products and Licensed Services provided by the Licensee.

For the avoidance of doubt, the Distributor must not use the Designations as the Distributor's company name, as any other business name for the Distributor, as an internet domain for the Distributor or as a component of the Distributor's company name, or as any other business name or internet domain of the Distributor. In particular, the Distributor is not allowed to label sales products with the Designations.

- (2) Any sub-license granted to a Distributor has to be subject to the following conditions:
  - (a) Distributor has to use the Designations only in their registered form and in accordance with the Utilization Regulations as provided for in **Annex 4** of this Agreement.
  - (b) Any use of the Designations by the Distributor qualifies as a use by DT for the purpose of acquiring or maintaining trademark rights or other rights to these Designations.
  - (c) The Licensee assumes no liability and provides no warranty that the use of the Designations does not violate third-party rights. The Distributor shall indemnify, defend and hold harmless the Licensee and DT from any and all any third-party claims that are caused by the Distributor s use of the Designations.
  - (d) Maintaining and defending the Designations shall be the exclusive domain of DT.
  - (e) The Distributor is not allowed to have the Designations registered as a trademark, as a company name, as an Internet domain, or as a component thereof.
  - (f) The Distributor authorises DT as third-party beneficiary to enforce all claims of the Licensee against the Distributor.
  - (g) The Distributor shall only use the Designations conforming to the highest quality and professional standards relevant in the Territory and having at least the same quality as the premium global providers of wireless telecommunication, broadband and information services. The Distributor acknowledges that DT is entitled to issue quality guidelines specifying such quality and professional standards. Such quality guidelines shall become binding on the Distributor 30 days after receipt of such guidelines in writing or in text form, including fax and e-mail.
  - (h) If the Distributor should learn that a third party uses the Designations and/or applies for their registration as a trademark, company name or Internet domain, it is obliged to inform DT immediately.
  - (i) The Licensee is entitled to terminate the Distribution Partner Agreement (insofar as the license provisions are concerned) at any time for good cause if the Distributor violates any of the above obligations or commits an act or omission that would entitle DT to terminate the License Agreement.

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

(j) If the Distribution Partner Agreement is terminated, the Distributor shall no longer be entitled to utilize the Designations.

## Annex 4 Utilization Regulations

The utilization regulations of DT, as valid at the time the Agreement is signed, are set forth as follows:

Guideline Basic Elements

Guideline Product Labelling

Guideline Device Packaging

Guideline Product Design

Guideline Electronic Media

Guideline Print Media

Guideline Forms

Guideline Giveaways

Guideline Literature

Guideline Distribution

Guideline Corporate Fashion

Guideline Shops

Guideline Co-operation Partners

Picture Tool

Brand Wizard

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

Brand Fit

A CD-ROM with utilization regulations will be delivered to the Licensee. Further guidelines are accessible for the Licensee under www.telekom-ci.net.

- 18 -

Annex 5 Approval Tool

- 19 -

## Annex 6 Pre-approved Co-Branding

[NOTE: Parties to fill-in prior to closing]

- 20 -

# Annex 7 Examples of combines Usage of Licensee s Company Name and Own Designations

Examples for which no License Fee is payable:

[NOTE: Parties to fill-in prior to closing]

Examples for which the License Fee is payable:

[NOTE: Parties to fill-in prior to closing]

- 21 -

## <u>Exhibit E</u>

## VOTING AND SUPPORT AGREEMENT

This Voting and Support Agreement (this <u>Agreement</u>) is made and entered into as of October 3, 2012, among Deutsche Telekom AG, an *Aktiengesellschaft* organized under the laws of Germany (<u>Deer</u>), and MADISON DEARBORN CAPITAL PARTNERS IV, L.P. (<u>Stockholder</u>), aDelaware limited partnership.

WHEREAS, Deer, certain subsidiaries of Deer, and MetroPCS Communications, Inc., a Delaware corporation (<u>Denali</u>) propose to enter into an Business Combination Agreement, to be dated as of or after the date hereof (the <u>Combination Agreement</u>), which provides, among other things, for Denali to effect a recapitalization and reverse split of its common stock, par value \$0.0001 per share (the <u>Common Stock</u>), to make a cash payment to Denali s stockholders as part of such recapitalization and reverse split, to issue shares of its Common Stock to Deer that will result in Deer s ownership of approximately 74% of the outstanding shares of Common Stock, and to amend the certificate of incorporation of Denali (collectively, the <u>Transaction</u>), upon the terms and subject to the conditions set forth in the Combination Agreement (capitalized terms used herein without definition shall have the respective meanings specified in the Combination Agreement);

WHEREAS, Deer and Stockholder are executing this agreement prior to or contemporaneously with the execution of the Combination Agreement;

WHEREAS, Stockholder owns shares of Common Stock (together with any other shares of capital stock of Denali acquired (whether beneficially or of record) by Stockholder after the date hereof and prior to the earlier of the Closing and the termination of all of Stockholder s obligations under this Agreement, including any shares of Common Stock acquired by means of purchase, dividend or distribution, or issued upon the exercise of any stock options to acquire Common Stock or warrants or the conversion of any convertible securities or otherwise, being collectively referred to herein as the <u>Securities</u>);

WHEREAS, receipt of stockholder approval of the issuance of shares of Common Stock to Deer and the amendment of the certificate of incorporation of Denali (the <u>Approval</u>) is a condition to the consummation of the Transaction; and

WHEREAS, as a condition to the willingness of Deer to enter into the Combination Agreement and as an inducement and in consideration therefor, Stockholder has agreed to enter into this Agreement.

NOW, THEREFORE, in consideration of the foregoing, the mutual covenants and agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, intending to be legally bound, the parties hereto agree as follows:

## ARTICLE I

#### VOTING; GRANT AND APPOINTMENT OF PROXY

Section 1.1 <u>Voting</u>. From and after the date hereof until the earlier of (a) the consummation of the Transaction, (b) the date of the Denali Stockholders Meeting and (c) the termination of the Combination Agreement pursuant to and in compliance with the terms therein (such earlier date, the <u>Expiration Date</u>), Stockholder irrevocably and unconditionally hereby agrees that at any meeting (whether annual or special and each adjourned or postponed meeting) of Denali s stockholders, however called, or in connection with any written consent of Denali s stockholders, the Stockholder (in such capacity and not in any other capacity) will (i) appear at such meeting or otherwise cause all of the Securities to be counted as present thereat for purposes of calculating a quorum and (ii) vote or cause to be voted (including by proxy or written consent, if applicable) all of the Securities, without regard to any Denali Adverse Recommendation Change:

(a) in favor of the Approval (and, in the event that the Approval is presented as more than one proposal, in favor of each proposal that is part of the Approval), and in favor of any other matter presented or proposed as to approval of the Transaction or any part or aspect thereof or any other transactions or matters contemplated by the Combination Agreement;

(b) against any Denali Acquisition Proposal, without regard to the terms of such Denali Acquisition Proposal, or any other transaction, proposal, agreement or action made in opposition to adoption of the Combination Agreement or in competition or inconsistent with the Transaction and the other transactions or matters contemplated by the Combination Agreement,

(c) against any other action, agreement or transaction, that is intended, that could reasonably be expected, or the effect of which could reasonably be expected, to materially impede, interfere with, delay, postpone, discourage or adversely affect the Transaction or any of the other transactions contemplated by the Combination Agreement or this Agreement or the performance by Stockholder of its obligations under this Agreement, including: (i) any extraordinary corporate transaction, such as a merger, consolidation or other business combination involving Denali or any of its Subsidiaries; (ii) a sale, lease or transfer of a material amount of assets of Denali or any of its Subsidiaries (other than the Transaction) or a reorganization, recapitalization or liquidation of Denali or any of its Subsidiaries; (iii) an election of new members to the board of directors of Denali, other than nominees to the board of directors of Denali who are serving as directors of Denali on the date of this Agreement or as otherwise provided in the Combination Agreement; (iv) any material change in the present capitalization or dividend policy of Denali or any amendment or other change to Denali s certificate of incorporation or bylaws, except if approved in writing by Deer; or (v) any other material change in Denali s corporate structure or business, except if approved in writing by Deer,

(d) against any action, proposal, transaction or agreement that would reasonably be expected to result in a breach in any respect of any covenant, representation or warranty or any other obligation or agreement of Denali contained in the Combination Agreement, or of Stockholder contained in this Agreement, and

(e) in favor of any other matter necessary or desirable to the consummation of the transactions contemplated by the Combination Agreement, including the Transaction (clauses (a) through (e), the <u>Required Votes</u>).

#### Section 1.2 Grant of Irrevocable Proxy: Appointment of Proxy.

1.2.1 From and after the date hereof until the Expiration Date, Stockholder hereby irrevocably and unconditionally grants to, and appoints, Deer and any designee thereof as Stockholder s proxy and attorney-in-fact (with full power of substitution), for and in the name, place and stead of Stockholder, to vote or cause to be voted (including by proxy or written consent, if applicable) its Securities, without regard to any Denali Adverse Recommendation Change, in accordance with the Required Votes.

- 2 -

1.2.2 Stockholder hereby represents that any proxies heretofore given in respect of the Securities, if any, are revocable, and hereby revokes such proxies.

1.2.3 Stockholder hereby affirms that the irrevocable proxy set forth in this <u>Section 1.2</u> is given in connection with the execution of the Combination Agreement, and that such irrevocable proxy is given to secure the performance of the duties of Stockholder under this Agreement. Stockholder hereby further affirms that the irrevocable proxy is coupled with an interest and, except as set forth in this <u>Section 1.2</u>, is intended to be irrevocable. If for any reason the proxy granted herein is not irrevocable, then Stockholder agrees, until the Expiration Date, to vote its Covered Securities in accordance with <u>Section 1.2.1(a)</u> through <u>Section 1.2.1(e)</u> above as instructed by Deer in writing. The parties agree that the foregoing is a voting agreement.

Section 1.3 <u>Restrictions on Transfers</u>. Stockholder hereby agrees that, from the date hereof until the Expiration Date, it shall not, directly or indirectly, (a) sell, transfer, assign, tender in any tender or exchange offer, pledge, encumber, hypothecate or similarly dispose of (by merger, by testamentary disposition, by operation of law or otherwise), either voluntarily or involuntarily, or to enter into any contract, option or other arrangement or understanding with respect to the sale, transfer, assignment, pledge, Lien, hypothecation or other disposition of (by merger, by testamentary disposition, by operation of law or otherwise), any Securities, (b) deposit any Securities into a voting trust or enter into a voting agreement or arrangement or grant any proxy or power of attorney with respect thereto that is inconsistent with this Agreement, or (c) agree (whether or not in writing) to take any of the actions referred to in the foregoing clause (a) or (b).

Section 1.4 <u>Inconsistent Agreements</u>. Stockholder hereby covenants and agrees that, except for this Agreement, it (a) shall not enter into at any time while this Agreement remains in effect, any voting agreement or voting trust with respect to its Covered Securities and (b) shall not grant at any time while this Agreement remains in effect a proxy, consent or power of attorney with respect to its Covered Securities.

## ARTICLE II

## NO SOLICITATION

Section 2.1 <u>Restricted Activities</u>. Prior to the Expiration Date, Stockholder (in its capacity as a stockholder of Denali) shall not, and shall cause his agents, advisors and other representatives (in each case, acting in their capacity as such to Stockholder, the <u>Stockholder Representatives</u>) not to, (a) initiate, solicit or knowingly encourage or knowingly take or continue any other action to facilitate the submission of any inquiry, indication of interest, proposal or offer that constitutes, or would reasonably be expected to lead to, a Denali Acquisition Proposal, (b) participate in any discussions or negotiations regarding, or that would reasonably be expected to lead to any Denali Acquisition Proposal (other than to inform a Person of the existence of this Section 2.1 and Section 4.5 of the Combination Agreement), (c) furnish any non-public information or data regarding Denali or any of its Subsidiaries to, or afford access to the properties, personnel, books and records of Denali to, any Person (other than Deer and its Subsidiaries) in connection with or in response to or in circumstances that would reasonably be expected to lead to, any Denali Acquisition Proposal, (d) take any action to make the provisions of any fair price, moratorium, control share acquisition, business combination or other similar anti-takeover statute or regulation (including any transaction under, or a third party becoming an interested stockholder under, Section 203 of the DGCL), or any restrictive provision of any applicable anti-takeover provision in Denali s Organizational Documents or the Denali Rights Agreement, inapplicable to any Person other than Deer and its Subsidiaries or to any transactions constituting or contemplated by a Denali Acquisition Proposal, or (e) resolve or agree to do any of the foregoing (the activities specified in clauses (a) through (e) being hereinafter referred to as the <u>Restricted Activities</u>).

Section 2.2 <u>Notification</u>. Stockholder (in its capacity as a stockholder of Denali) shall and shall cause the Stockholder Representatives to, immediately cease and terminate any and all existing activities, discussions or negotiations with any Person with respect to a Denali Acquisition Proposal. From and after the date hereof until

- 3 -

the Expiration Date, Stockholder shall as promptly as practicable (and in any event within 24 hours) (i) notify Deer of (x) any Denali Acquisition Proposal it receives in its capacity as a stockholder of Denali, (y) any request it receives in its capacity as a stockholder of Denali for non-public information relating to Denali or its Subsidiaries, other than requests for information not reasonably expected to be related to an Denali Acquisition Proposal, and (z) any inquiry or request for discussion or negotiation it receives in its capacity as a stockholder of Denali regarding an Denali Acquisition Proposal, (ii) if such Denali Acquisition Proposal, request or inquiry is in writing, deliver to Deer a copy of such Denali Acquisition Proposal, request or inquiry and any related draft agreements and other written material setting forth the terms and conditions of such Denali Acquisition Proposal, and (iii) if such Denali Acquisition Proposal, request or inquiry is oral, provide to Deer a detailed summary thereof. Stockholder shall keep Deer reasonably informed on a prompt and timely basis of the status and material details of any such Denali Acquisition Proposal and with respect to any material change to the terms of any such Denali Acquisition Proposal within 24 hours of any such material change. This <u>Section 2.2</u> shall not apply to any Denali Acquisition Proposal received by Denali.

Section 2.3 <u>Capacity</u>. Stockholder is signing this Agreement solely in his capacity as a stockholder of Denali and nothing contained herein shall in any way limit or affect any actions taken by any Stockholder Representative in his capacity as an officer or director of Denali, and no action taken in any such capacity as an officer or director shall be deemed to constitute a breach of this Agreement.

#### ARTICLE III

#### REPRESENTATIONS, WARRANTIES AND COVENANTS

#### OF STOCKHOLDER

Section 3.1 Representations and Warranties. Stockholder represents and warrants to Deer as follows: (a) Stockholder has full legal right and capacity to execute and deliver this Agreement, to perform Stockholder s obligations hereunder and to consummate the transactions contemplated hereby, (b) this Agreement has been duly executed and delivered by Stockholder and the execution, delivery and performance of this Agreement by Stockholder and the consummation of the transactions contemplated hereby have been duly authorized by all necessary action on the part of Stockholder and no other actions or proceedings on the part of Stockholder are necessary to authorize this Agreement or to consummate the transactions contemplated hereby, (c) this Agreement constitutes the valid and binding agreement of Stockholder, enforceable against Stockholder in accordance with its terms, (d) the execution and delivery of this Agreement by Stockholder does not, and the consummation of the transactions contemplated hereby and the compliance with the provisions hereof will not, conflict with or violate any Laws or agreement binding upon Stockholder or the Securities, nor require any authorization, consent or approval of, or filing with, any Governmental Entity, except for filings with the Securities and Exchange Commission by Stockholder, (e) Stockholder owns, beneficially and of record, or controls 30,568,184 shares of Common Stock and (f) except for such transfer restrictions of general applicability as may be provided under the Securities Act of 1933, as amended, and the blue sky laws of the various states of the United States, Stockholder owns, beneficially and of record, or controls all of its Securities (and any additional Securities acquired by it after the date hereof) free and clear of any proxy, voting restriction, adverse claim or other Lien (other than any restrictions created by this Agreement) and has sole voting power with respect to the Securities and sole power of disposition with respect to all of the Securities, with no restrictions on Stockholder s rights of voting or disposition pertaining thereto, and no person other than Stockholder has any right to direct or approve the voting or disposition of any of the Securities.

Section 3.2 Covenants. Stockholder hereby:

(a) agrees, prior to the Expiration Date, not to take any action that would make any representation or warranty of Stockholder contained herein untrue or incorrect or have or would reasonably be expected to have the effect of preventing, impeding or interfering with or adversely affecting the performance by Stockholder of his obligations under this Agreement;

- 4 -

(b) irrevocably waives, and agrees not to exercise, any rights of appraisal or rights of dissent from the Transaction that Stockholder may have with respect to the Securities;

(c) agrees to promptly notify Denali and Deer of the number of any new Securities acquired by Stockholder after the date hereof and prior to the Expiration Date. Any such Securities shall be subject to the terms of this Agreement as though owned by Stockholder on the date hereof;

(d) agrees to permit Denali to publish and disclose in the Proxy Statement Stockholder s identity and ownership of the Securities and the nature of the Stockholder s commitments, arrangements and understandings under this Agreement; and

(e) shall and does authorize Deer or its counsel to notify Denali s transfer agent that there is a stop transfer order with respect to all of its Securities (and that this Agreement places limits on the voting and transfer of such shares), provided that Deer or its counsel further notifies Denali s transfer agent to lift and vacate the stop transfer order with respect to the Securities no later than the Expiration Date.

#### ARTICLE IV

#### TERMINATION

This Agreement shall terminate and be of no further force or effect (i) upon the Expiration Date or (ii) on the fifth (5th) calendar day following the date of this Agreement should the Combination Agreement not be executed by all of the parties thereto as of such date. Notwithstanding the preceding sentence, this <u>Article IV</u> and <u>Article V</u> shall survive any termination of this Agreement. Nothing in this <u>Article V</u> shall relieve or otherwise limit any party of liability for willful breach of this Agreement.

#### ARTICLE V

#### MISCELLANEOUS

Section 5.1 <u>Expenses</u>. Deer, on the one hand, and Stockholder, on the other hand, shall bear their respective expenses, costs and fees (including attorneys, auditors and financing fees, if any) in connection with the preparation, execution and delivery of this Agreement and compliance herewith, whether or not the Transaction is effected.

Section 5.2 <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party to the others shall be in writing and delivered personally or sent by registered or certified mail, return receipt requested and postage prepaid, or by facsimile (providing confirmation of such facsimile transmission):

To Deer:

c/o Deutsche Telekom AG

Friedrich-Ebert-Alle 140

53113 Bonn, Germany

Attention: General Counsel

Fax: +49-228-181-74008

with copies to:

Wachtell, Lipton, Rosen & Katz

51 West 52nd Street

New York, New York 10019

Attention: Adam O. Emmerich, Esq.

Facsimile: 212-403-2000

Email: aoemmerich@wlrk.com

- 5 -

To Stockholder:

Madison Dearborn Capital Partners IV, L.P.

c/o Madison Dearborn Partners, LLC

Three First National Plaza, Suite 4600

Chicago, IL 60602

Attention: Michael P. Cole

Facsimile: 312-895-1226

Email: mcole@mdcp.com

with a copy to:

Madison Dearborn Partners, LLC

Three First National Plaza, Suite 4600

Chicago, IL 60602

Attention: Mark B. Tresnowski, Esq.

Facsimile: 312-895-1041

Email: mtresnowski@mdcp.com

or to such other Persons or addresses as may be designated in writing by the party to receive such notice as provided above.

Section 5.3 <u>Amendments; Waivers</u>. Any provision of this Agreement may be amended or waived if, and only if, such amendment or waiver is in writing and signed (i) in the case of an amendment, by Deer and Stockholder, and (ii) in the case of a waiver, by the party against whom the waiver is to be effective. No failure or delay by any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof, nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege.

Section 5.4 <u>Assignment</u>. No party to this Agreement may assign any of its rights or obligations under this Agreement, including by sale of stock, operation of law in connection with a merger or sale of substantially all the assets, without the prior written consent of the other party hereto; provided that Deer may assign its rights and obligations under this Agreement to a Subsidiary of Deer, so long as Deer remains liable for its obligations hereunder.

Section 5.5 <u>No Partnership</u>, <u>Agency</u>, <u>or Joint Venture</u>. This Agreement is intended to create, and creates, a contractual relationship and is not intended to create, and does not create, any agency, partnership, joint venture or any like relationship between the parties hereto.

Section 5.6 Entire Agreement. This Agreement and the Combination Agreement constitute the entire agreement, and supersede all other prior and contemporaneous agreements, understandings, undertakings, arrangements, representations and warranties, both written and oral, among the parties with respect to the subject matter hereof.

Section 5.7 <u>No Third-Party Beneficiaries</u>. This Agreement is not intended to confer upon any Person other than the parties hereto any rights or remedies hereunder.

Section 5.8 <u>Governing Law</u>; Jurisdiction; Waiver of Trial by Jury. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE WITHOUT REGARD TO ANY CONFLICT-OF-LAWS RULE OR

## Table of Contents

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

PRINCIPLE THEREOF. Each party hereto agrees that it shall bring any action or proceeding in respect of any claim arising out of, or related to, this Agreement or the Transaction, exclusively in the Delaware Court of Chancery, New Castle County, or solely if that court does not have jurisdiction, a federal court sitting in the State of Delaware (the

<u>Chosen Courts</u>), and solely in connection with claims arising under this Agreement or the Transaction (a) irrevocably submits to the exclusive jurisdiction of the Chosen Courts, (b) waives any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waives any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any party hereto, and (d) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 5.2. Deer irrevocably designates CT Corporation, and Stockholder irrevocably designates National Registered Agents, Inc., as its agent and attorney-in-fact for the acceptance of service of process and making an appearance on its behalf in any such claim or proceeding and for the taking of all such acts as may be necessary or appropriate in order to confer jurisdiction over it before the Chosen Courts and each party hereto stipulates that such consent and appointment is irrevocable and coupled with an interest. EACH PARTY HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF, OR RELATING TO, THIS AGREEMENT OR THE TRANSACTION.

Section 5.9 <u>Specific Performance</u>. The parties agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with the terms hereof and that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement or to enforce specifically the performance of the terms and provisions hereof in any of the Chosen Courts without any requirement to post bond, in addition to any other remedy to which they are entitled at law or in equity.

Section 5.10 Interpretation. (a) The words hereof, herein, and hereunder and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement; (b) the words date hereof, when used in this Agreement, shall refer to the date set forth in the Preamble; (c) the terms defined in the singular have a comparable meaning when used in the plural, and vice versa; (d) the terms defined in the present tense have a comparable meaning when used in the past tense, and vice versa; (e) any references herein to a specific Section or Article shall refer, respectively, to Sections or Articles of this Agreement; (f) wherever the word include, includes, or including is used in this Agreement, it shall be deemed to be followed by the words without limitation; (g) references herein to any gender includes each other gender; (h) the word or shall not be exclusive; (i) the headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof; and (j) the parties hereto have participated jointly in the negotiation and drafting of this Agreement and, in the event that an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

Section 5.11 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each such counterpart (including any facsimile or electronic document transmission of such counterpart) being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

Section 5.12 <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability or the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision; and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

[Signature Pages Follow]

- 7 -

IN WITNESS WHEREOF, the parties hereto have duly executed and delivered this Agreement as of the date and year first written above.

## MADISON DEARBORN CAPITAL

## PARTNERS IV, L.P.

By: Madison Dearborn Partners IV, L.P. Its: General Partner

By: Madison Dearborn Partners, LLC Its: General Partner

By: Its: Managing Director [Signature Page to Voting and Support Agreement]

## DEUTSCHE TELEKOM AG

By:

Name: Title:

By: Name: Title:

## <u>Exhibit F</u>

#### DT Notes Pricing Schedule<sup>1</sup>

#### DT Notes (Issued Regardless of Backstop)

TABLE I: PERMANENT NOTES							
Tenor	Year <sup>2</sup>	Amount <sup>3</sup>	Call Protection	Call Coupon Schedule <sup>4</sup>	Make-Whole Call <sup>5</sup>		
6 years	2019	\$1,250	NC-2.0	$\frac{1}{2}$ , $\frac{1}{4}$ , par thereafter	T+50bps		
7 years	2020	\$1,250	NC-3.0	$1/_{2}$ , $1/_{4}$ , par thereafter	T+50bps		
8 years	2021	\$1,250	NC-4.0	1/2, $1/4$ , par thereafter	T+50bps		
9 years				<sup>2</sup> <sup>4</sup> <sup>1</sup> / <sub>2</sub> ,			
	2022	\$1,250	NC-4.0	$\frac{1}{3}$ , $\frac{1}{6}$ , par thereafter	T+50bps		
10 years				$1/_{2}, 1/_{3}, 1/_{6}, \text{ par}$			
-	2023	\$1,250	NC-5.0	thereafter	T+50bps		
11 years				$1/_{2}$ , $1/_{3}$ , $1/_{6}$ , par			
-	2024	\$1,250	NC-5.0	thereafter	T+50bps		
					-		
Total		\$7,500					

			TA	BLE II: RESET M	NOTES			
Tenor	Year	Amount	Reset Date	Call Protection (before Reset)	Tenor after Reset	Call Protection (after Reset)	Call Coupon Schedule <sup>6</sup>	Make-Whole Call
6 years			Year 2 Reset					
	2019	\$ 1,250	Date <sup>7</sup>	NC-Life	4.0 years	NC-2.0	$1/_{2}$ , par thereafter	T+50 bps
7 years			Year 2 Reset				<sup>1</sup> / <sub>2</sub> ,	
	2020	\$ 1,250	Date	NC-Life	5.0 years	NC-2.0	$1/_{4}$ , par thereafter	T+50 bps
8 years			Year 2.5 Reset				$1/_{2}, 1/_{4}, \text{ par}$	
	2021	\$ 1,250	Date <sup>8</sup>	NC-Life	5.5 years	NC-2.5	thereafter	T+50 bps
9 years			Year 2.5 Reset				$1/_{2}$ , $1/_{4}$ , par	
	2022	\$ 1,250	Date	NC-Life	6.5 years	NC-2.5	thereafter	T+50 bps
10 years			Year 3 Reset				$1/_{2}$ , $1/_{4}$ , par	
	2023	\$ 1,250	Date <sup>9</sup>	NC-Life	7.0 years	NC-3.0	thereafter	T+50 bps
11 years			Year 3 Reset				$1/_{2}$ , $1/_{4}$ , par	
	2024	\$ 1,250	Date	NC-Life	8.0 years	NC-3.0	thereafter	T+50 bps
Total		\$ 7,500						

<sup>1</sup> The pricing of all Deer Notes listed in this Exhibit shall be determined by the attached Pricing Mechanism.

- <sup>2</sup> Assumes closing takes place in 2013.
- <sup>3</sup> All dollar amounts in this Exhibit are in millions.
- <sup>4</sup> Call price on first day of call period and anniversaries of that date.
- <sup>5</sup> T means Treasury Rate.
- <sup>6</sup> Call price on first day of call period and anniversaries of that date.
- <sup>7</sup> Year 2 Reset Date means the second anniversary of the Issue Date.
- <sup>8</sup> Year 2.5 Reset Date means the date that is six months after the second anniversary of the Issue Date.
- <sup>9</sup> Year 3 Reset Date means the third anniversary of the Issue Date.

# 1

#### Additional DT Notes (Issued if Backstop Used)<sup>10</sup>

#### TABLE III: PERMANENT NOTES

Tenor	Year	aximum Mount	Call Protection	Call Coupon Schedule <sup>11</sup>	Make-Whole Call
3 years <sup>12</sup>	2016	\$ 500	NC-Life		T+50 bps
5 years <sup>13</sup>	2018	\$ 1,250	NC-Life		T+50 bps
7 years <sup>14</sup>	2020	\$ 500	NC-4.0	$\frac{1}{2}$ , $\frac{1}{4}$ , par thereafter	T+50 bps
8 years	202115	\$ 500	NC-4.0	1/2, 1/4, par thereafter	T+50 bps
Total		\$ 2,750			

TABLE IV: RESET NOTES <sup>16</sup>							
Tenor		Year	Maximu	m Amount			
3 years <sup>17</sup>		2016	\$	500			
5 years <sup>18</sup>		2018	Ŧ				
3 years <sup>17</sup> 5 years <sup>18</sup> 7 years <sup>19</sup>		2020	\$	1.250			
8 years			Ф	1,250			
•							
			\$	500			
		202120	\$	500			
Total			\$	2,750			
10(4)			φ	2,750			

- <sup>10</sup> The Maximum Amount listed for Deer Notes in Tables III and IV reflect the full backstop and no commitment reductions. The allocation of Additional Deer Notes among the series of Deer Notes in Tables III and IV shall be determined, as described in Section 5 of Exhibit I (*Financing Backstop Terms and Conditions*).
- <sup>11</sup> Call price on first day of call period and anniversaries of that date.
- <sup>12</sup> Assumes transaction closes in 2013.
- <sup>13</sup> Assumes transaction closes in 2013.
- <sup>14</sup> Assumes transaction closes in 2013.
- <sup>15</sup> Assumes transaction closes in 2013.
- <sup>16</sup> The Reset Date, Call Protection, Call Coupon Schedule, and Make-Whole Call for each series of Additional Deer Notes in Table IV shall be as described on page 5.
- <sup>17</sup> Assumes transaction closes in 2013.
- <sup>18</sup> Assumes transaction closes in 2013.
- <sup>19</sup> Assumes transaction closes in 2013.
- <sup>20</sup> Assumes transaction closes in 2013.

#### Reset Date for Additional DT Notes in Table IV Issued on the Closing Date

## ( Initial Backstop Reset Notes<sup>21</sup>)

The Reset Date applicable to each series of Initial Backstop Reset Notes shall be determined as follows.

## A. The second anniversary of the Issue Date (the Year 2 Reset Date )

Each series of Initial Backstop Reset Notes described by the following method shall have its price reset on the Year 2 Reset Date, according to the Pricing Mechanism.

A-1. Determine the aggregate principal amount of all Initial Backstop Reset Notes outstanding on the Closing Date.

A-2. Select a set of series of Initial Backstop Reset Notes by adding the outstanding principal amount of each series of Initial Backstop Reset Notes as of the Closing Date, in order of maturity from earliest to latest, until the sum is greater than one third of the amount calculated in Step A-1.

A-3. From the set of series of Initial Backstop Reset Notes specified by Step A-2, remove the series with the longest maturity. The remaining series of Initial Backstop Reset Notes shall have their price reset on the Year 2 Reset Date.

#### B. Six months after the second anniversary of the Issue Date (the Year 2.5 Reset Date )

Each series of Initial Backstop Reset Notes described by the following method shall have its price reset on the Year 2.5 Reset Date, according to the Pricing Mechanism.

**B-1.** Determine the aggregate principal amount of all Initial Backstop Reset Notes as of the Closing Date that did not have their price reset on the Year 2 Reset Date.

**B-2.** Select a set of series of Initial Backstop Reset Notes by adding the principal amount as of the Closing Date of each series of Initial Backstop Reset Notes identified in Step B-1, in order of maturity from earliest to latest, until the sum is greater than one half of the amount calculated in Step B-1.

**B-3.** From the set of series of Initial Backstop Reset Notes specified by Step B-2, remove the series with the longest maturity. The remaining series of Initial Backstop Reset Notes shall have their price reset on the Year 2.5 Reset Date.

**B.4.** Notwithstanding the foregoing, if there are only two series of Initial Backstop Reset Notes identified in Step B-1, the Reset Date of the series with the earlier maturity shall be the Year 2.5 Reset Date and the Reset Date of the series with the later maturity shall be the Year 3 Reset Date.

#### C. The third anniversary of the Issue Date ( Year 3 Reset Date )

Each series of Initial Backstop Reset Notes that did not have its price reset on the Year 2 Reset Date or the Year 2.5 Reset Date shall have its price reset on the Year 3 Reset Date according to the Pricing Mechanism.

<sup>21</sup> Includes any Additional Deer Notes in Table IV issued in respect of the Change of Control Backstop on the Closing Date.

#### Reset Date for Additional DT Notes in Table IV Issued in Respect of Change of Control Backstop after the Closing Date

If any Initial Backstop Reset Notes were issued on the Closing Date, the Reset Date applicable to each series of Additional DT Notes in Table IV issued after the Closing Date shall be determined as follows: (i) If there is one such series of Additional DT Notes, the Reset Date of such series shall be the third anniversary of the Issue Date; and (ii) if there are two such series of Additional DT Notes, the Reset Date of the series of such notes with the earlier maturity shall be the second anniversary of the Issue Date and the Reset Date of the series of such notes with the later maturity shall be the third anniversary of the Issue Date.

If no Initial Backstop Reset Notes were issued on the Closing Date, the Reset Date applicable to each series of Additional DT Notes in Table IV issued after the Closing Date shall be determined as follows: (i) If there is one such series of Additional DT Notes, the Reset Date of such series shall be the third anniversary of the Issue Date; and (ii) if there are two such series of Additional DT Notes, the Reset Date of the series of such notes with the earlier maturity shall be six months after the second anniversary of the Issue Date and the Reset Date of the series of such notes with the later maturity shall be the third anniversary of the Issue Date.

4

#### Potential Terms of the Additional DT Notes in Table IV

Table V provides an overview of the terms of all potential series of Additional DT Notes given the tenor (as specified in Exhibit I (Backstop Terms and Conditions) and possible Reset Dates (as specified on pages 3-4 above).

	TABLE V: RESET NOTES POTENTIAL ADDITIONAL DT NOTES IF BACKSTOP USED						
Α	В	С	D Call Protection	E Tenor after	F Call Protection	G Call Coupon	H Make- Whole
Tenor	Year	Reset Date	(before Reset)	Reset	(after Reset)	Schedule <sup>22</sup>	Call
3 years		Year 2.0 Reset					
	2016	Date <sup>23</sup>	NC-Life	1.0 years	NC-Life		T+50bps
3 years	2016	Year 2.5 Reset Date <sup>24</sup>	NC-Life	0.5 years	NC-Life		T+50bps
5 years		Year 2.0 Reset					
	2018	Date	NC-Life	3.0 years	NC-Life		T+50bps
5 years		Year 2.5 Reset					
	2018	Date	NC-Life	2.5 years	NC-Life		T+50bps
5 years		Year 3.0 Reset					
	2018	Date <sup>25</sup>	NC-Life	2.0 years	NC-Life		T+50bps
7 years		Year 2.0 Reset				<sup>1</sup> / <sub>2</sub> ,	
	2020	Date	NC-Life	5.0 years	NC-2	$^{1}/_{4}$ , par thereafter	T+50bps
7 years		Year 2.5 Reset				$1/_{2}$ , $1/_{4}$ , par	
	2020	Date	NC-Life	4.5 years	NC-2	thereafter	T+50bps
7 years		Year 3.0 Reset					
	2020	Date	NC-Life	4.0 years	NC-2	$1/_{2}$ , par thereafter	T+50bps
8 years		Year 2.0 Reset				$1/_{2}, 1/_{4}, \text{ par}$	
	2021	Date	NC-Life	6.0 years	NC-3	thereafter	T+50bps
8 years		Year 2.5 Reset				$1/_{2}$ , $1/_{4}$ , par	
	2021	Date	NC-Life	5.5 years	NC-3	thereafter	T+50bps
8 years		Year 3.0 Reset				$1/_{2}$ , $1/_{4}$ , par	
	2021	Date	NC-Life	5.0 years	NC-2	thereafter	T+50bps

<sup>22</sup> Call price on first day of call period and anniversaries of that date.

<sup>23</sup> Year 2 Reset Date means the second anniversary of the Issue Date.

<sup>24</sup> Year 2.5 Reset Date means the date that is six months after the second anniversary of the Issue Date.

<sup>25</sup> Year 3 Reset Date means the third anniversary of the Issue Date.

5

## **Pricing Mechanism**

See attached.

6

#### Pricing Mechanism

#### Applicable Rates

The interest rate applicable to each series of Permanent Notes shall at all times be the Permanent Rate for such series.

From the Issue Date until the applicable Reset Date, the interest rate applicable to each series of Reset Notes shall be the Initial Rate for such series.

On and after the applicable Reset Date, the interest rate applicable to each series of Reset Notes shall be the Reset Rate for such series.

Permanent Rate means, for Permanent Notes of any series, a rate per annum equal to the sum of:

- (i) the Reference Yield as of the Issue Date, plus
- (ii) 187.5 bps, plus
- (iii) the Maturity Adjustment for DT Notes of such series as of the Issue Date (the sum of (i), (ii), and (iii), for purposes of this definition, the *Permanent Specified Rate*), plus
- (iv) the OID Amount applicable to DT Notes of such series based on the Permanent Specified Rate.

Initial Rate means, for Reset Notes of any series, a rate per annum equal to the sum of:

- (i) the Reference Yield as of the Issue Date, plus
- (ii) 100.0 bps, plus
- (iii) the Maturity Adjustment for DT Notes of such series as of the Issue Date (the sum of (i), (ii), and (iii), for purposes of this definition, the *Initial Specified Rate*), plus
- (iv) the OID Amount applicable to DT Notes of such series based on the Initial Specified Rate.

Reset Rate means, for Reset Notes of any series, a rate per annum equal to the sum of:

- (i) the Reference Yield as of the Reset Date, plus
- (ii) 100.0 bps, plus

(iii) the Maturity Adjustment for DT Notes of such series as of the Reset Date, plus

(iv) the amount that was included in the calculation of the Initial Rate of such series of DT Notes on the Issue Date pursuant to clause (iv) of the definition of Initial Rate .

*Maturity Adjustment* means, for any series of DT Notes as of any specified date, an amount (which may be negative) equal to the product of (i)(a) the Remaining Tenor of such series of DT Notes *minus* (b) the number eight (8) and (ii) 12.5 basis points.

*OID Amount* means, as to DT Notes of any series for any Permanent Specified Rate or Initial Specified Rate, the difference between (i) the effective yield to maturity that would be applicable to newly-issued notes having the same Remaining Tenor as such series of DT Notes, with a coupon equal to the Permanent Specified Rate or Initial Specified Rate (as applicable), that were issued with 200 bps of original issue discount, and (ii) such Permanent Specified Rate or Initial Specified Rate (as applicable), .

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

*Reference Yield* means, as of any date of determination, a rate per annum, as determined by the Calculation Agent in a commercially reasonable manner, equal to the sum of:

(i) the product of (a) the Index Weight and (b) the Average Adjusted Index Yield as of such date; *plus*,

1

(ii) the product of (a) the Comparable Bond Weight and (b) the Average Adjusted Comparable Bond Yield as of such date; *plus*,

(iii) the product of (a) the Company Bond Weight and (b) the Average Adjusted Company Bond Yield as of such date.

#### Notice of Rates

Within five business days of the Issue Date, Issuer will send a notice to each holder and the trustee, specifying the Permanent Rate and the Initial Rate, and setting forth in reasonable detail the manner in which each such rate was calculated. Within five business days of each Reset Date, Issuer will send a notice to each holder and the trustee, specifying the Reset Rate for such Reset Date, and setting forth in reasonable detail the manner in which each such rate was calculated.

2

#### Telecom HY Index Definitions

Average Adjusted Index Yield means, as of any date of determination, the arithmetic average of the Adjusted Index Yield for each Component Index as of such date.

*Index Weight* means: (i) if the number of Qualified Comparable Bonds is one or more and the number of Qualified Company Bonds is one or more, one-half; and (ii) if (a) the number of Qualified Comparable Bonds is zero and the number of Qualified Company Bonds is one or more, or (b) the number of Qualified Company Bonds is zero and the number of Qualified Company Bonds is one or more, two-thirds; and (iii) if the number of Qualified Comparable Bonds is zero and the number of Qualified Company Bonds is zero, one.

JPM HY Index means the JP Morgan Domestic Telecom High Yield Index ; provided, that (i) if the sponsor of such index discontinues calculation of such index, and such sponsor or another entity publishes a successor or substitute index that the Calculation Agent determines, in its sole discretion, to be comparable to such index (such comparable index, the Successor Index ), then the Calculation Agent shall substitute the Successor Index for such index for all purposes hereunder; and (ii) in the event that the sponsor of such index, the Calculation Agent determines that no suitable successor or substitute index is available for such index, the Calculation Agent will calculate the Daily Yield to Worst and Effective Yield to Worst Time for such index in accordance with the formula for and method of calculation last in effect as to such index prior to its discontinuance, but using only those securities with such weightings as comprised such index immediately prior to its discontinuance.

*CS HY Index* means the Credit Suisse US Liquid High Yield Telecom Index ; *provided*, that (i) if the sponsor of such index discontinues calculation of such index, and such sponsor or another entity publishes a successor or substitute index that the Calculation Agent determines, in its sole discretion, to be comparable to such index (such comparable index, the *CS Successor Index*), then the Calculation Agent shall substitute the CS Successor Index for such index for all purposes hereunder; and (ii) in the event that the sponsor of such index discontinues publication of such index and the Calculation Agent determines that no suitable successor or substitute index is available for such index, the Calculation Agent will calculate the Daily Yield to Worst and Effective Yield to Worst Time for such index in accordance with the formula for and method of calculation last in effect as to such index prior to its discontinuance, but using only those securities with such weightings as comprised such index immediately prior to its discontinuance.

*BAML HY Index* means the Bank of America Merrill Lynch US High Yield Telecommunications (HOTC) Index ; *provided*, that (i) if the sponsor of such index discontinues calculation of such index, and such sponsor or another entity publishes a successor or substitute index that the Calculation Agent determines, in its sole discretion, to be comparable to such index (such comparable index, the *BAML Successor Index*), then the Calculation Agent shall substitute the BAML Successor Index for such index for all purposes hereunder; and (ii) in the event that the sponsor of such index discontinues publication of such index and the Calculation Agent determines that no suitable successor or substitute index is available for such index, the Calculation Agent will calculate the Daily Yield to Worst and Effective Yield to Worst Time for such index in accordance with the formula for and method of calculation last in effect as to such index prior to its discontinuance, but using only those securities with such weightings as comprised such index immediately prior to its discontinuance.

3

#### **Comparable Bond Definitions**

Average Adjusted Comparable Bond Yield means, as of any date of determination, the arithmetic average of the Adjusted Bond Yields for each of the Qualified Comparable Bonds as of such date.

*Comparable Bond Weight* means: (i) if the number of Qualified Comparable Bonds is one or more and the number of Qualified Company Bonds is zero, one-third; (ii) if the number of Qualified Comparable Bonds is one or more and the number of Qualified Company Bonds is one or more, one-quarter; and (iii) if the number of Qualified Comparable Bonds is zero, zero.

Comparable Issuer means Sprint Nextel Corporation or any successor or assign.

Qualified Comparable Bonds means, as of any date of determination, each series of publicly-traded unsecured notes that:

(i) is issued by the Comparable Issuer;

(ii) has a minimum aggregate principal amount outstanding as of such date of at least \$1.0 billion;

(iii) has an Average Bond Price during the Calculation Period of not less than 800.00 or greater than 1200.00 (per \$1000.00 of principal amount);

(iv) for which the date that results in the yield to worst is not less than four (4) and not greater than ten (10) years from such date of determination;

(v) for which there are at least five Trading Days in the Calculation Period on which there is a reported trade of such Qualified Comparable Bond in an amount greater than \$500,000 of principal amount; *and* 

(vi) which was not subject, on any date during the Calculation Period, to a publicly announced tender offer, exchange offer, or any transaction that would result or has resulted in a change of control.

4

## Company Bond Definitions

Average Adjusted Company Bond Yield means, as of any date of determination, the arithmetic average of the Adjusted Bond Yields for each of the Qualified Company Bonds as of such date.

*Company Bond Weight* means: (i) if the number of Qualified Company Bonds is one or more and the number of Qualified Comparable Bonds is zero, one-third; (ii) if the number of Qualified Comparable Bonds is one or more and the number of Qualified Company Bonds is one or more, one-quarter; and (iii) if the number of Qualified Company Bonds is zero, zero.

Qualified Company Bonds means, as of any date of determination, each series of publicly-traded unsecured notes that:

(i) is issued by the Issuer;

(ii) has a minimum aggregate principal amount outstanding as of such date of at least \$1.0 billion;

(iii) has an Average Bond Price during the Calculation Period of not less than 800.00 or greater than 1200.00 (per \$1000.00 of principal amount);

(iv) for which the date that results in the yield to worst is not less than four (4) and not greater than (10) years from such date of determination;

(v) for which there are at least five Trading Days in the Calculation Period on which there is a reported trade of such Qualified Comparable Bond in an amount greater than \$500,000 of principal amount; *and* 

(vi) which was not subject, on any date during the Calculation Period, to a publicly announced tender offer, exchange offer, or any transaction that would result or has resulted in a change of control.

5

#### General Definitions

Adjusted Bond Yield means, for any Component Bond as of any date of determination:

(i) the Average Yield to Worst of such Component Bond as of such date, plus

(ii) an amount (which may be negative) equal to the product of (a)(1) the number of days until a date that is eight (8) years from such date of determination, minus (2) the number of days until the date that results in the yield to worst of such Component Bond, and (b) the fraction 12.5 basis points divided by three hundred sixty-five (365); *plus* 

(iii) an amount (which may be negative) equal to the product of (a)(1) the score set forth in the definition of Composite Rating for the Composite Rating of the applicable series of DT Notes, minus (2) the score set forth in the definition of Composite Rating for the Composite Rating of such Component Bond, and (b) 50 basis points.

Adjusted Index Yield means, for any Component Index as of any date of determination:

(i) the Average Yield to Worst of such Component Index as of such date, plus

(ii) an amount (which may be negative) equal to the product of (a)(1) the number of days until a date that is eight (8) years from such date of determination, minus (2) Effective Yield to Worst Time of such Component Index, and (b) the fraction 12.5 basis points divided by three hundred sixty-five (365).

Average Yield to Worst means, as to any Component on any date of determination, the arithmetic average of the Daily Yield to Worst of such Component on each Trading Day in the Calculation Period.

Average Bond Price means, as to any Component Bond on any date of determination, the arithmetic average of the Daily Bond Price of such Component on each Trading Day in the Calculation Period.

Bloomberg means Bloomberg Financial Markets.

Calculation Agent means [], or such other nationally recognized investment bank as is selected by DT.

*Calculation Period* means, as to any date of determination, the period that begins on the first Trading Day on or following the day that is 45 days prior to such date and ends on such date.

Component means any Component Bond or any Component Index.

Component Bond means any Qualified Comparable Bond or any Qualified Company Bond.

Component Index means the JPM HY Index, the CS HY Index, and the BAML HY Index.

6

*Composite Rating* means (i) if such Security is rated by either S&P or Moody s, but not both, the composite rating indicated on the chart below as corresponding to the rating given to such Security by S&P or Moody s; and (ii) if such Security is rated by both S&P and Moody s, the composite rating indicated on the chart below as corresponding to a score equal to the average (rounded up to the nearest integer) of the scores assigned on the chart below to such Security s S&P rating and such Security s Moody s rating.

Moody s	S&P	Score	Composite Rating
Baa1	BBB+	1	BBB+
Baa2	BBB	2	BBB
Baa3	BBB-	3	BBB-
Ba1	BB+	4	BB+
Ba2	BB	5	BB
Ba3	BB-	6	BB-
B1	B+	7	B+
B2	В	8	В
B3	В-	9	В-
Caal	CCC+	10	CCC+
Caa2	CCC	11	CCC
Caa3	CCC-	12	CCC-

Daily Yield to Worst means:

(i) as to any Component Bond on any date of determination, the arithmetic average of the yield to worst of each reported trade of such Component Bond in an amount greater than \$500,000 of principal amount on such date (or, if there are no reported trades of such Component Bond in an amount greater than \$500,000 of principal amount on such date, the arithmetic average of the yield to worst of each reported trade of such Component Bond in an amount greater than \$500,000 of principal amount on the next preceding Trading Day during the Calculation Period on which there was such a trade), determined by reference to the Pricing Source; and

(ii) as to any Component Index on any date of determination, the yield to worst of such Component Index for such date as reported by the Pricing Source.

*Daily Bond Price* means, as to any Component Bond on any date of determination, the arithmetic average of the price of each reported trade of such Component Bond in an amount greater than \$500,000 of principal amount on such date (or, if there are no reported trades of such Component Bond in an amount greater than \$500,000 of principal amount on such date, the arithmetic average of the price of each reported trade of such Component Bond in an amount greater than \$500,000 of principal amount on such date, the arithmetic average of the price of each reported trade of such Component Bond in an amount greater than \$500,000 of principal amount on the next preceding Trading Day during the Calculation Period on which there was such a trade), determined by reference to the Pricing Source.

*Effective Yield to Worst Time* means, as to any Component Index, the number of days to the date that results in the yield to worst implied by such Component Index, as reported by the Pricing Source, or if not reported by the Pricing Source, as calculated by the Calculation Agent in good faith.

Issue Date means, as to any series of DT Notes, the date on which such series of DT Notes were initially issued.

Moody s means Moody s Investors Service, Inc., and its successors.

*Pricing Source* means (i) with respect to any Component Index, the index sponsor; and (ii) with respect to any Component Bond, TRACE as provided by Bloomberg. Except where another source or method is specified, the

Pricing Source shall be used for all applicable purposes hereunder. If the Pricing Source for any Component is unavailable at any time for any reason, and an alternative source or method is not expressly provided herein, the Calculation Agent shall determine a suitable substitute source in good faith.

*Relevant Exchange* means, for any Component Bond, the primary exchange or quotation system on which such Component Bond is traded or quoted, as determined by the Calculation Agent.

*Reset Date* means the date that is [24] [30] [36] months after the Issue Date.

*Remaining Tenor* means, for any series of DT Notes as of any specified date, the number of days from such specified date to the Stated Maturity for such series of DT Notes, divided by three hundred sixty-five (365).

Security means any series of DT Notes or any Component Bond.

S&P means Standard & Poor s Rating Services, a division of The McGraw-Hill Companies, Inc., and its successors.

*Trading Day* means (i) with respect to any Component Index, any date on which the sponsor of such index reports an updated value for such index; and (ii) with respect to any Component Bond, a day, as determined by the Calculation Agent, on which the Relevant Exchange with respect to such Component Bond is scheduled to be open for trading for their respective regular trading sessions or on which such Component Bond is quoted, as applicable.

8

# Exhibit G (conformed to reflect the amendments made pursuant to the letter agreement, dated December 5, 2012, between MetroPCS Communications, Inc. and Deutsche Telekom AG)

#### Form of Description of Notes

You can find the definitions of certain terms used in this description of notes under the subheading Certain definitions below. In this description of notes, *Issuer* refers only to T-Mobile USA, Inc., a Delaware corporation, and not to any of its Subsidiaries and *Parent* refers only to MetroPCS Communications, Inc., a Delaware corporation, and not to any of its Subsidiaries.

Issuer will issue the notes as a series of debt securities under a new indenture among itself, Parent, the Subsidiary Guarantors and [Wells Fargo Bank, N.A.], as trustee. The terms of the notes include those stated in the indenture and those made part of the indenture by reference to the Trust Indenture Act of 1939, as amended.<sup>1</sup>

The obligations and covenants of Issuer described hereunder are only of Issuer and not of Parent, its direct parent company. Although Parent will be a guarantor of the notes, it and its Subsidiaries, except Issuer and its Restricted Subsidiaries, are generally not subject to any of the obligations and covenants described hereunder.

[The following description is a summary of the material provisions of the indenture. It does not restate the indenture in its entirety. We urge you to read the indenture in its entirety because it, and not this description of notes, defines your rights as a holder of the notes. We have filed a copy of the indenture as an exhibit to [\_\_\_\_], which has been incorporated by reference in this prospectus supplement. For more information on how you can obtain a copy of the indenture, see Incorporation of documents by reference. Certain defined terms used in this description of notes but not defined below under Certain definitions have the meanings assigned to them in the indenture.]

The registered holder of a note will be treated as the owner of it for all purposes. Only registered holders will have rights under the indenture.

#### Brief description of the notes and the note guarantees

The notes

The notes:

will be general unsecured obligations of Issuer;

will be *pari passu* in right of payment with all existing and future unsecured senior Indebtedness and other liabilities of Issuer, including Issuer s existing senior notes;

will be senior in right of payment to any future subordinated Indebtedness of Issuer to the extent that such future Indebtedness provides by its terms that it is subordinated to the notes; and

will be unconditionally guaranteed on a senior unsecured basis by the Guarantors.

However, the notes will be effectively subordinated to all borrowings under existing and future secured Indebtedness of Issuer or any Guarantor to the extent of the assets securing such Indebtedness and to all liabilities of any of Issuer s Subsidiaries that do not guarantee the notes to the extent of the assets of those Subsidiaries. [See Risk Factors Risks related to the notes The notes and the guarantees will be unsecured and effectively subordinated to our and the guarantors existing and future secured indebtedness and structurally subordinated to any future indebtedness and other liabilities of our non-guarantor subsidiaries. ]

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

The Form of the Indenture to be customary for offerings of this type, and to include customary provisions for debt securities offered under and exception from the securities laws that are subject to registration rights.

<sup>2</sup> NTD: Non-operative disclosure language has been retained and bracketed globally for reference.

```
A-189
```

#### The note guarantees

The notes will be guaranteed by Parent, all of Issuer s Domestic Restricted Subsidiaries that are Wholly-Owned Subsidiaries (other than Immaterial Subsidiaries), Issuer s Restricted Subsidiaries that guarantee any Specified Issuer Indebtedness, and any future Subsidiary of Parent that directly or indirectly owns equity interests of Issuer.

Each guarantee of the notes by a Guarantor:

will be a general unsecured obligation of that Guarantor;

will be *pari passu* in right of payment with all existing and future unsecured senior Indebtedness and other liabilities of that Guarantor, including its guarantee of Issuer s existing senior notes; and

will be senior in right of payment to any future subordinated Indebtedness of that Guarantor to the extent that such future Indebtedness provides by its terms that it is subordinated to its guarantee of the notes.

However, the guarantees will be effectively subordinated to all existing and future secured Indebtedness of the Guarantors to the extent of the assets securing such Indebtedness. [See Risk Factors Risks related to the notes The notes and the guarantees will be unsecured and effectively subordinated to our and the guarantors existing and future secured indebtedness and structurally subordinated to any future indebtedness and other liabilities of our non-guarantor subsidiaries. ]

Initially, all of Issuer s existing Domestic Restricted Subsidiaries will guarantee the notes. Under the circumstances described below under the subheading Certain covenants Additional note guarantees, one or more of Issuer s Subsidiaries together with certain newly created or acquired Subsidiaries in the future may not guarantee the notes. In the event of a bankruptcy, liquidation or reorganization of any of these non-guarantor Subsidiaries, the non-guarantor Subsidiaries will pay their trade creditors and holders of their debt and other obligations before they will be able to distribute any of their assets to Issuer.

Initially, all of Issuer s Subsidiaries will be Restricted Subsidiaries. However, under the circumstances described below under the caption Certain covenants Designation of restricted and unrestricted subsidiaries, Issuer will be permitted to designate certain of its Subsidiaries as Unrestricted Subsidiaries. Issuer s Unrestricted Subsidiaries will not be subject to many of the restrictive covenants in the indenture. Issuer s Unrestricted Subsidiaries will not guarantee the notes.

#### Principal, maturity and interest

Issuer will issue  $[]^3$  in aggregate principal amount of notes in this offering. Issuer may issue additional notes from time to time after this offering, and such additional notes may be issued either under the indenture or one or more supplemental indentures. Any issuance of additional notes is subject to all of the covenants in the indenture, including the covenant described below under the caption Certain covenants Incurrence of indebtedness and issuance of preferred stock. The notes and any additional notes subsequently issued under the indenture will be treated as a single series for all purposes under the indenture, including waivers, amendments, redemptions and offers to purchase. Issuer will issue notes in minimum denominations of \$2,000 and integral multiples of \$1,000. The notes will mature on [].<sup>4</sup>

Interest on the notes will accrue (i) at the rate of  $[]\%^5$  per annum to and including the date that is  $[]^6$  months after the Closing Date and (ii) thereafter at the rate per annum equal to  $[]\%,^7$  and will be payable semiannually

- <sup>3</sup> NTD: Amount of each series of DT Notes issued will be as set forth on the DT Note Pricing Schedule.
- <sup>4</sup> NTD: Tenor of each series of DT Notes will be as set forth on the DT Note Pricing Schedule.
- <sup>5</sup> NTD: Initial rate for each series of DT Notes to be determined as set forth on the DT Note Pricing Schedule.
- <sup>6</sup> NTD: Reset date for Reset Notes to be determined as set forth on the DT Note Pricing Schedule.
- <sup>7</sup> NTD: Reset rate for each series of DT Notes to be determined as set forth on the DT Note Pricing Schedule.

# - 2 -

in arrears on [	] and [	], commencia	ng on [
on the immediately preceding [		] and [	].

].8 Issuer will make each interest payment to the holders of record

Interest on the notes will accrue from the date of original issuance or, if interest has already been paid, from the date it was most recently paid. Interest will be computed on the basis of a 360 day year comprised of twelve 30-day months. If an interest payment date or the maturity date falls on a day that is not a business day, the related payment of principal or interest will be made on the next succeeding business day as if made on the date the payment was due, and no interest shall accrue for the intervening period.

Payments of principal of and interest on the notes issued in book-entry form or definitive form, if any, will be made as described below under the caption Same day settlement and payment.

The notes will initially be evidenced by one or more global notes deposited with a custodian for, and registered in the name of, Cede & Co., as nominee of The Depository Trust Company (DTC). Except as described below, beneficial interests in the global notes will be shown on, and transfers thereof will be effected only through, records maintained by DTC and its direct and indirect participants.<sup>9</sup> We do not intend to apply for the notes to be listed on any securities exchange or to arrange for the notes to be quoted on any quotation system.

### Methods of receiving payments on the notes

If a holder of a definitive note has given wire transfer instructions to Issuer and Issuer is the paying agent, Issuer will pay all principal, interest and premium, if any, on that holder s notes in accordance with those instructions until given written notice to the contrary. All other payments on the notes will be made at the office or agency of the paying agent and registrar within the City and State of New York unless Issuer elects to make interest payments by check mailed to the noteholders at their address set forth in the books and records of the registrar.

### Paying agent and registrar for the notes

The trustee will initially act as paying agent and registrar. Issuer may change the paying agent or registrar without prior notice to the holders of the notes, and Issuer or any of its Subsidiaries may act as paying agent or registrar.

### Transfer and exchange

Except as set forth below, the global notes may be transferred, in whole and not in part, only to another nominee of DTC or to a successor of DTC or its nominee, as described below under the caption Book-entry, delivery and form.

A holder of a definitive note may transfer or exchange notes in accordance with the provisions of the indenture. The registrar and the trustee may require a holder, among other things, to furnish appropriate endorsements and transfer documents in connection with a transfer of notes. Holders will be required to pay all taxes relating to, arising out of, or in connection with such transfer. Issuer will not be required to transfer or exchange any note selected for redemption. Also, Issuer will not be required to transfer or exchange any note for a period of 15 days before a selection of notes to be redeemed.

<sup>&</sup>lt;sup>8</sup> NTD: Payment and record dates for Permanent Notes to be the 1<sup>st</sup> and 15<sup>th</sup> day of the month falling 3 and 9 months after the Closing Date. Payment and record dates for Reset Notes to be the 1<sup>st</sup> and 15<sup>th</sup> day of the month falling 6 and 12 months after the Closing Date.

<sup>&</sup>lt;sup>9</sup> DT and TMUS will cooperate to take all actions reasonably required to submit the notes promptly to DTC to the extent that DTC registration is available.

### Note guarantees

The notes will be guaranteed by Parent, all of Issuer s Domestic Restricted Subsidiaries that are Wholly-Owned Subsidiaries (other than Immaterial Subsidiaries), all of Issuer s Restricted Subsidiaries that guarantee any Specified Issuer Indebtedness, and any future Subsidiary of Parent that directly or indirectly owns equity interests of Issuer. These Note Guarantees will be joint and several obligations of the Guarantors. The obligations of each Guarantor under its Note Guarantee will be limited as necessary to prevent that Note Guarantee from constituting a fraudulent conveyance under applicable law. [See Risk factors Risks related to the notes The guarantees may not be enforceable because of fraudulent conveyance laws. ]

A Guarantor (other than Parent) may not sell or otherwise dispose of all or substantially all of its assets to, or consolidate with or merge with or into (whether or not such Guarantor is the surviving Person) another Person, other than Issuer or another Guarantor, unless:

- (1) immediately after giving effect to that transaction, no Default or Event of Default exists; and
- (2) either:
  - (a) if it is not already a Guarantor, the Person acquiring the property in any such sale or disposition or the Person formed by or surviving any such consolidation or merger assumes all the obligations of that Guarantor under the indenture and its Note Guarantee pursuant to a supplemental indenture satisfactory to the trustee; or
  - (b) such sale or other disposition complies with the Asset Sale provisions of the indenture (it being understood that only such portion of the Net Proceeds as is or is required to be applied on or before the date of such release in accordance with the terms of the indenture needs to be so applied).

The Note Guarantee of a Subsidiary Guarantor will be released:

- in connection with any sale or other disposition of all or substantially all of the assets of that Guarantor (including by way of merger or consolidation) to a Person that is not (either before or after giving effect to such transaction) the Issuer or a Restricted Subsidiary of Issuer, if the sale or other disposition is not prohibited by the Asset Sale provisions of the indenture;
- (2) in connection with any issuance, sale or other disposition of Capital Stock of that Guarantor to a Person that is not (either before or after giving effect to such transaction) the Issuer or a Restricted Subsidiary of Issuer, if the issuance, sale or other disposition does not violate the Asset Sale or Restricted Investment provisions of the indenture, and the Guarantor ceases to be a Wholly-Owned Subsidiary of Issuer as a result of such sale or other disposition and does not guarantee any Specified Issuer Indebtedness;
- (3) if such Guarantor ceases to guarantee any Specified Issuer Indebtedness and is not a Wholly-Owned Subsidiary;
- (4) if Issuer designates any Restricted Subsidiary that is a Guarantor to be an Unrestricted Subsidiary in accordance with the applicable provisions of the indenture;
- (5) upon legal defeasance or satisfaction and discharge of the indenture as provided below under the captions Legal defeasance and covenant defeasance and Satisfaction and discharge ;

- (6) upon the liquidation or dissolution of such Guarantor provided no Default or Event of Default has occurred that is continuing; or
- (7) solely in the case of a Note Guarantee created pursuant to clause (b) of the covenant described below under the caption Additional note guarantees, upon the release or discharge of the Guarantee that resulted in the creation of such Note Guarantee pursuant to the covenant described below under the caption Additional note guarantees, except a discharge or release by or as a result of payment under such Guarantee.

See Repurchase at the option of holders Asset sales.

- 4 -

### **Optional redemption**

At any time prior to  $[ ]^{10}$ , Issuer may on any one or more occasions redeem up to 35% of the aggregate principal amount of notes issued under the indenture at a redemption price of  $[ ]^{11}$ % of the principal amount, plus accrued and unpaid interest to, but not including, the redemption date, with the net cash proceeds of one or more sales of Equity Interests (other than Disqualified Stock) of Issuer or contributions to Issuer s common equity capital made with the net cash proceeds of one or more sales of Equity Interests (other than Disqualified Stock) of Parent; *provided* that:

(1) at least 65% of the aggregate principal amount of notes issued under the indenture (excluding notes held by Issuer and its Subsidiaries) remains outstanding immediately after the occurrence of such redemption; and

(2) the redemption occurs within 180 days of the date of the closing of such sale of Equity Interests by Issuer or the date of contribution to Issuer s common equity capital made with net cash proceeds of one or more sales of Equity Interests of Parent.

On or after [  $]^{12}$ , Issuer may redeem all or a part of the notes upon not less than 10 nor more than 60 days notice, at the redemption prices (expressed as percentages of principal amount) set forth below plus accrued and unpaid interest on the notes redeemed, to the applicable redemption date, if redeemed during the twelve month period beginning on [ ] of the years indicated below, subject to the rights of holders of notes on the relevant record date to receive interest on the relevant interest payment date for periods prior to such redemption date:



Unless Issuer defaults in the payment of the redemption price, interest will cease to accrue on the notes or portions thereof called for redemption on the applicable redemption date.

At any time prior to [ ]<sup>14</sup>, Issuer may also redeem all or a part of the notes, upon not less than 10 nor more than 60 days prior notice sent electronically or mailed by first-class mail to each holder s registered address, at a redemption price equal to 100% of the principal amount of notes redeemed plus the Applicable Premium as of, and accrued and unpaid interest to, but not including, the date of redemption, subject to the rights of holders of notes on the relevant record date to receive interest due on the relevant interest payment date for periods prior to such date of redemption.

### Mandatory redemption

Issuer is not required to make mandatory redemption or sinking fund payments with respect to the notes.

### Repurchase at the option of holders

### Change of control triggering event

If a Change of Control Triggering Event occurs, each holder of notes will have the right to require Issuer to repurchase all or any part (equal to \$2,000 or an integral multiple of \$1,000) of that holder s notes pursuant to a

- <sup>10</sup> NTD: To be the date that is three years after the Closing Date.
- <sup>11</sup> NTD: To be par plus the coupon for the applicable series.
- <sup>12</sup> NTD: Redemption dates to be as set forth in the DT Note Pricing Schedule
- <sup>13</sup> NTD: Percentage for each tranche of DT Notes to be a specified percentage of the Reset Price for the applicable year, as set forth on the DT Note Pricing Schedule.
- <sup>14</sup> NTD: Redemption dates to be as set forth in the DT Note Pricing Schedule

# - 5 -

Change of Control Offer on the terms set forth in the indenture. In the Change of Control Offer, Issuer will offer a Change of Control Payment in cash equal to 101% of the aggregate principal amount of notes repurchased plus accrued and unpaid interest on the notes repurchased to, but not including, the date of purchase, subject to the rights of holders of notes on the relevant record date to receive interest due on the relevant interest payment date for periods prior to such repurchase date (the Change of Control Payment ). Within 30 days following any Change of Control Triggering Event, Issuer will send a notice (the *Change of Control Offer*) to each holder and the trustee describing the transaction or transactions and identify the ratings decline that together constitute the Change of Control Triggering Event and offering to repurchase notes on the Change of Control Payment Date specified in the notice, which date will be no earlier than 10 days and no later than 60 days from the date such notice is sent (the *Change of Control Payment Date*), pursuant to the procedures required by the indenture and described in such notice. Issuer will comply with the requirements of Rule 14e-1 under the Exchange Act and any other securities laws and regulations thereunder to the extent those laws and regulations are applicable in connection with the repurchase of the notes as a result of a Change of Control Triggering Event. To the extent that the provisions of any securities laws or regulations conflict with the Change of Control Triggering Event provisions of the indenture, or compliance with the Change of Control Triggering Event provisions of the indenture would constitute a violation of any such laws or regulations, Issuer will comply with the applicable securities laws and regulations and will not be deemed to have breached its obligations under the Change of Control Triggering Event provisions of the indenture by virtue of such compliance. In connection with the tender of any notes with respect to a Change of Control Triggering Event, the tendering holder shall provide good title to the notes, free and clear of all liens and encumbrances, and shall represent and warrant that such holder is presenting good title, free and clear of all liens and encumbrances, and such other representations and warranties as are customary.

On the Change of Control Payment Date, Issuer will, to the extent lawful:

- (1) accept for payment all notes or portions of notes properly tendered pursuant to the Change of Control Offer;
- (2) deposit with the paying agent an amount equal to the Change of Control Payment in respect of all notes or portions of notes properly tendered; and
- (3) deliver or cause to be delivered to the trustee the notes properly accepted together with an officers certificate stating the aggregate principal amount of notes or portions of notes being purchased by Issuer.

The paying agent will promptly make payment to each holder of notes properly tendered the Change of Control Payment for such notes, and the trustee will promptly authenticate and mail (or cause to be transferred by book entry) to each holder, a new note equal in principal amount to any unpurchased portion of the notes surrendered, if any; *provided* that each new note will be in a principal amount of \$2,000 or an integral multiple of \$1,000 in excess thereof. Issuer will publicly announce the results of the Change of Control Offer on or as soon as practicable after the Change of Control Payment Date.

The provisions described above that require Issuer to make a Change of Control Offer following a Change of Control Triggering Event will be applicable whether or not any other provisions of the indenture are applicable. Except as described above with respect to a Change of Control Triggering Event, the indenture does not contain provisions that permit the holders of the notes to require, or otherwise provide, that Issuer repurchase or redeem the notes in the event of a takeover, recapitalization or similar transaction.

Notwithstanding the foregoing, Issuer will not be required to make a Change of Control Offer upon a Change of Control Triggering Event if (1) a third party makes the Change of Control Offer in the manner, at the times and otherwise in compliance with the requirements set forth in the indenture applicable to a Change of Control Offer made by Issuer and purchases all notes properly tendered and not withdrawn under the Change of Control Offer, or (2) notice of redemption has been given pursuant to the indenture as described above under the caption Optional redemption, unless and until there is a default in payment of the applicable redemption price.

- 6 -

A Change of Control Offer may be made in advance of a Change of Control Triggering Event, and conditioned upon such Change of Control Triggering Event, if a definitive agreement has been executed for a transaction that would constitute a Change of Control at the time of making of the Change of Control Offer.

In the event that holders of not less than 90% of the aggregate principal amount of the outstanding notes accept a Change of Control Offer and Issuer purchases all of the notes held by such holders, Issuer will have the right, upon not less than 10 nor more than 60 days prior notice, given not more than 30 days following the purchase pursuant to the Change of Control Offer described above, to redeem all of the notes that remain outstanding following such purchase at a redemption price equal to the Change of Control Payment plus, to the extent not included in the Change of Control Payment, accrued and unpaid interest on the notes that remain outstanding, to, but not including, the date of redemption (subject to the right of holders of record on the relevant record date to receive interest due on an interest payment date that is on or prior to the redemption date).

[The definition of Change of Control includes a phrase relating to the direct or indirect sale, lease, transfer, conveyance or other disposition of all or substantially all of the properties or assets of Issuer and its Restricted Subsidiaries taken as a whole. Although there is a limited body of case law interpreting the phrase substantially all, there is no precise established definition of the phrase under applicable law. Accordingly, the ability of a holder of notes to require Issuer to repurchase its notes as a result of a sale, lease, transfer, conveyance or other disposition of less than all of the assets of Issuer and its Restricted Subsidiaries taken as a whole to another Person or group may be uncertain.]

### Asset sales

Issuer will not, and will not permit any of its Restricted Subsidiaries to, consummate an Asset Sale unless:

- (1) Issuer (or the Restricted Subsidiary, as the case may be) receives consideration at the time of the Asset Sale at least equal to the Fair Market Value of the assets or Equity Interests issued or sold or otherwise disposed of;
- (2) at least 75% of the consideration received by Issuer or such Restricted Subsidiary in the Asset Sale and all other Asset Sales since the date of the indenture is in the form of cash, Cash Equivalents or Replacement Assets or a combination thereof. For purposes of this provision, each of the following will be deemed to be cash:
  - (a) any liabilities, as shown on Issuer s most recent consolidated balance sheet (or as would be shown on Issuer s consolidated balance sheet as of the date of such Asset Sale), of Issuer or any Restricted Subsidiary (other than contingent liabilities and liabilities that are by their terms subordinated to the notes or any Note Guarantee) that are assumed by the transferee of any such assets pursuant to a novation agreement that releases Issuer or such Restricted Subsidiary from further liability; and
  - (b) any securities, notes or other obligations received by Issuer, or any such Restricted Subsidiary, from such transferee that are converted by Issuer or such Restricted Subsidiary into cash, Cash Equivalents or Replacement Assets within 90 days after such Asset Sale, to the extent of the cash, Cash Equivalents or Replacement Assets received in that conversion.

Notwithstanding the foregoing, the 75% limitation referred to above shall be deemed satisfied with respect to any Asset Sale in which the cash, Cash Equivalents or Replacement Assets portion of the consideration received therefrom, determined in accordance with the foregoing provision on an after-tax basis, is equal to or greater than what the after-tax proceeds would have been had such Asset Sale complied with the aforementioned 75% limitation.

Within 365 days after the receipt of any Net Proceeds from an Asset Sale, Issuer or a Restricted Subsidiary may apply an amount equal to such Net Proceeds:

(1) to purchase Replacement Assets; or

# - 7 -

(2) to prepay, repay, defease, redeem, purchase or otherwise retire Indebtedness and other Obligations under a Credit Facility or Indebtedness secured by property that is subject to such Asset Sale and, if the Indebtedness repaid is revolving credit Indebtedness, to correspondingly reduce commitments with respect thereto.

Notwithstanding the foregoing, if within 365 days after the receipt of any Net Proceeds from an Asset Sale, Issuer or a Restricted Subsidiary enters into a binding written agreement committing Issuer or such Restricted Subsidiary, subject to customary conditions, to an application of funds of the kind described in clause (1) above, Issuer or such Restricted Subsidiary shall be deemed not to be in violation of the preceding paragraph so long as such application of funds is consummated within 545 days of the receipt of such Net Proceeds.

Pending the final application of any Net Proceeds of an Asset Sale, Issuer may temporarily reduce revolving credit borrowings or otherwise use the Net Proceeds in any manner that is not prohibited by the indenture.

An amount equal to any Net Proceeds from Asset Sales that are not applied or invested as provided in the third paragraph of this covenant will constitute *Excess Proceeds*. When the aggregate amount of Excess Proceeds exceeds \$100.0 million, within 20 days thereof, Issuer shall apply the entire aggregate amount of unutilized Excess Proceeds (not only the amount in excess of \$100.0 million) to make an offer (an *Asset Sale Offer*) to all holders of notes and all holders of other Indebtedness that is *pari passu* with the notes containing provisions requiring the Issuer to make an offer to purchase or redeem with the proceeds of sales of assets to purchase the maximum principal amount of notes and purchase or redeem such other *pari passu* Indebtedness that may be purchased or redeemed out of the Excess Proceeds. The offer price in any Asset Sale Offer will be equal to 100% of the principal amount of the notes and such other *pari passu* Indebtedness, and will be payable in cash. If any Excess Proceeds remain after consummation of an Asset Sale Offer, Issuer and its Restricted Subsidiaries may use those Excess Proceeds for any purpose not otherwise prohibited by the indenture. If the aggregate principal amount of notes and other *pari passu* Indebtedness tendered into such Asset Sale Offer exceeds the amount of Excess Proceeds, the trustee will select the notes and Issuer will select such other *pari passu* Indebtedness to be purchased or redeemed on a *pro rata* basis. Upon completion of each Asset Sale Offer, the amount of Excess Proceeds will be reset at zero.

Issuer will comply with the requirements of Rule 14e-1 under the Exchange Act and any other securities laws and regulations thereunder to the extent those laws and regulations are applicable in connection with each repurchase of notes pursuant to an Asset Sale Offer. To the extent that the provisions of any securities laws or regulations conflict with the Asset Sale provisions of the indenture, or compliance with the Asset Sale provisions of the indenture would constitute a violation of any such laws or regulations, Issuer will comply with the applicable securities laws and regulations and will not be deemed to have breached its obligations under the Asset Sale provisions of the indenture by virtue of such compliance.

[The agreements governing Issuer's other Indebtedness contain, and future agreements may contain, prohibitions of certain events, including events that would constitute a Change of Control or an Asset Sale and may prohibit repurchases of or other prepayments in respect of the notes. The exercise by the holders of the notes of their right to require Issuer to repurchase the notes upon a Change of Control Triggering Event or an Asset Sale could cause a default under these other agreements, even if the Change of Control Triggering Event or Asset Sale itself does not, due to the financial effect of such repurchases or other prepayments on Issuer. In the event a Change of Control Triggering Event or Asset Sale occurs at a time when Issuer is prohibited from purchasing notes, Issuer could seek the consent of the holders of such Indebtedness to the purchase of notes or could attempt to refinance the borrowings that contain such prohibition. If Issuer does not obtain a consent or repay those borrowings, Issuer will remain prohibited from purchasing notes. In that case, Issuer's failure to purchase tendered notes would constitute an Event of Default under the indenture that could, in turn, constitute a default under the other Indebtedness. Finally, Issuer's ability to pay cash to the holders of notes upon a repurchase may be limited by

- 8 -

Issuer s then existing financial resources. See Risk Factors Risks related to the notes Our senior secured credit facility, the indenture governing the notes and the indentures governing our existing senior notes include restrictive covenants that limit our operating flexibility. ]

### Selection and notice

If less than all of the notes are to be redeemed at any time, the trustee will select notes for redemption on a pro rata basis unless otherwise required by law or applicable stock exchange or depository requirements.

No notes of \$2,000 or less can be redeemed in part. Notices of redemption will be sent electronically or mailed by first class mail at least 10 but not more than 60 days before the redemption date to each holder of notes to be redeemed at its registered address, except that redemption notices may be sent more than 60 days prior to a redemption date if the notice is issued in connection with a defeasance of the notes or a satisfaction and discharge of the indenture. Except as otherwise set forth in the provisions described under the captions holders Change of control triggering event, notices of redemption may not be conditional.

If any note is to be redeemed in part only, the notice of redemption that relates to that note will state the portion of the principal amount of that note that is to be redeemed. If in definitive form a new note in principal amount equal to the unredeemed portion of the original note will be issued in the name of the holder of notes upon cancellation of the original note. Except to the extent that a notice of redemption is conditional as permitted in the provisions described under the captions Repurchase at the option of holders Change of control triggering event, notes called for redemption become due on the date fixed for redemption. On and after the redemption date, interest ceases to accrue on notes or portions of notes called for redemption.

### Certain covenants

### Changes in covenants when notes rated investment grade

If on any date following the Closing Date:

- (1) the notes are rated Investment Grade by two out of the three Rating Agencies; and
- (2) no Default or Event of Default shall have occurred and be continuing (other than with respect to the covenants specifically listed under the following captions),

then, beginning on that day, the covenants specifically listed under the following captions in this prospectus supplement will cease to apply and will not be later reinstated even if the ratings of the notes should subsequently decline:

- (1) Repurchase at the option of holders Asset sales ;
- (2) Restricted payments ;
- (3) Incurrence of indebtedness and issuance of preferred stock ;
- (4) Dividend and other payment restrictions affecting subsidiaries ;
- (5) Transactions with affiliates ;

## (6) Designation of restricted and unrestricted subsidiaries ; and

(7) clauses (3) (to the extent that a Default or Event of Default exists by reason of one or more of the covenants specifically listed in this paragraph) and (4) of the covenant described below under the caption Merger, consolidation or sale of assets.
 [There can be no assurance that the notes will ever achieve an Investment Grade rating.]

- 9 -

### **Restricted** payments

Issuer will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly:

- (1) declare or pay (without duplication) any dividend, or make any other payment or distribution, on account of Issuer s or any of its Restricted Subsidiaries Equity Interests (including any payment in connection with any merger or consolidation involving Issuer or any of its Restricted Subsidiaries) or to the direct or indirect holders of Issuer s or any of its Restricted Subsidiaries Equity Interests in their capacity as such (other than dividends or distributions payable in Equity Interests (other than Disqualified Stock) of Issuer and other than dividends or distributions payable to Issuer or a Restricted Subsidiary of Issuer);
- (2) purchase, redeem or otherwise acquire or retire for value (including in connection with any merger or consolidation involving Issuer) any Equity Interests of Issuer or any direct or indirect parent of Issuer;
- (3) make any payment on or with respect to, or purchase, redeem, defease or otherwise acquire or retire for value any Subordinated Indebtedness (excluding any intercompany Indebtedness between or among Issuer and any of its Restricted Subsidiaries), except a payment of interest or principal at the Stated Maturity thereof; or

(4) make any Restricted Investment

(all such payments and other actions set forth in clauses (1) through (4) above being collectively referred to as *Restricted Payments* ),

unless, at the time of and after giving effect to such Restricted Payment:

- (1) no Default or Event of Default has occurred and is continuing or would occur as a consequence of such Restricted Payment;
- (2) Issuer would, at the time of such Restricted Payment and after giving pro forma effect thereto as if such Restricted Payment had been made at the beginning of the applicable four-quarter period, have been permitted to incur at least \$1.00 of additional Indebtedness pursuant to the Debt to Cash Flow Ratio test set forth in the first paragraph of the covenant described below under the caption Incurrence of indebtedness and issuance of preferred stock ; and
- (3) such Restricted Payment, together with the aggregate amount of all other Restricted Payments made by Issuer and its Restricted Subsidiaries since the Closing Date (excluding Restricted Payments permitted by clauses (2), (3), (4), (5), (6), (7), (8), (9), (11), (12), (13), (14) and (15) of the next succeeding paragraph), is less than the sum, without duplication, of:
  - (a) 100% of Issuer s Consolidated Cash Flow for the period (taken as one accounting period) from and after the Closing Date to the end of Issuer s most recently ended fiscal quarter for which internal financial statements are available at the time of such Restricted Payment, less the product of 1.4 times Issuer s Consolidated Interest Expense for the same period; *plus*
  - (b) 100% of the aggregate net cash proceeds, and the Fair Market Value of any property other than cash, received by Issuer after the Closing Date as a contribution to its common equity capital or from the issue or sale of Equity Interests of Issuer (other than Disqualified Stock) or from the issue or sale of convertible or exchangeable Disqualified Stock or convertible or exchangeable debt securities of Issuer that have been converted into or exchanged for such Equity Interests (other than Equity Interests (or Disqualified Stock or debt securities) sold to a Subsidiary of Issuer); *plus*

(c) to the extent that any Restricted Investment that was made after the Closing Date is sold for cash or Cash Equivalents, or otherwise is liquidated or repaid for cash or Cash Equivalents, an amount equal to such cash and Cash Equivalents; *plus* 

- 10 -

- (d) to the extent that any Unrestricted Subsidiary of Issuer designated as such after the Closing Date is redesignated as a Restricted Subsidiary after the Closing Date, the Fair Market Value of Issuer s Investment in such Subsidiary as of the date of such redesignation; other than to the extent such Investment constituted a Permitted Investment; *plus*
- (e) 100% of any cash dividends or cash distributions, and the Fair Market Value of any property other than cash, actually received directly or indirectly by Issuer or a Restricted Subsidiary of Issuer that is a Guarantor after the Closing Date from an Unrestricted Subsidiary of Issuer, in each case, to the extent that such dividends, cash distributions or other property were not otherwise included in the Consolidated Net Income of Issuer for such period and other than to the extent such Investment constituted a Permitted Investment; *minus*
- (f) the aggregate amount of any Net Equity Proceeds taken into account for purposes of incurring Indebtedness pursuant to clause
   (14) of the definition of Permitted Debt set forth below under the caption Incurrence of indebtedness and issuance of preferred stock ; plus
- (g) the amount that would be calculated immediately prior to the occurrence of the Merger and Closing Date pursuant to sub-clause (G) of clause (3) of the second paragraph of Section 4.07(a) of the Existing Notes Indenture with respect to the MetroPCS Existing Notes (as defined in the Business Combination Agreement), as in effect as of the date of the issuance of the notes.<sup>15</sup>

So long as no Default has occurred and is continuing or would be caused thereby, the preceding provisions will not prohibit:

- the payment of any dividend or the consummation of any irrevocable redemption within 60 days after the date of declaration of the dividend or giving of the redemption notice, as the case may be, if at the date of declaration or notice, the dividend or redemption payment would have complied with the provisions of the indenture;
- (2) the making of any Restricted Payment in exchange for, or out of the net cash proceeds of the substantially concurrent sale (other than to a Subsidiary of Issuer) of, Equity Interests of Issuer (other than Disqualified Stock) or from the substantially concurrent contribution of common equity capital to Issuer; *provided* that the amount of any such net cash proceeds that are utilized for any such Restricted Payment will be excluded from clause (3)(b) of the preceding paragraph; *provided further* that any Net Equity Proceeds (x) used for making a Restricted Investment pursuant to clause (10) of this paragraph or (y) taken into account for purposes of incurring Indebtedness pursuant to clause (14) of the definition of Permitted Debt set forth below under the caption Incurrence of indebtedness and issuance of preferred stock, may not also be used to make a Restricted Payment pursuant to this clause (2);
- (3) the repurchase, redemption, defeasance or other acquisition or retirement for value of Subordinated Indebtedness of Issuer or any Subsidiary Guarantor with the net cash proceeds from a substantially concurrent incurrence of Permitted Refinancing Indebtedness;
- (4) the payment of any dividend (or, in the case of any partnership or limited liability company, any similar distribution) by a Restricted Subsidiary of Issuer to the holders of its Equity Interests on a *pro rata* basis;
- (5) the repurchase, redemption or other acquisition or retirement for value of any Equity Interests of Parent, Issuer, any Restricted Subsidiary of Issuer or any direct or indirect parent of Issuer held by any current or former officer, director, employee or consultant of Parent, Issuer or any of its Restricted Subsidiaries pursuant to any equity subscription agreement, stock option agreement, shareholders agreement or similar agreement; *provided* that the aggregate price paid for all such repurchased, redeemed, acquired or retired Equity Interests may not exceed \$50.0 million in any fiscal year; *provided further*, that such amount in any fiscal year may be increased by an amount equal to (a) the net cash proceeds from the sale of Equity

<sup>15</sup> Purpose is to start with an initial basket equal to the corresponding basket under the Existing Notes as of the Closing Date.

Interests of Parent to current or former members of management, directors, consultants or employees that occurs after the Closing Date plus (b) the net cash proceeds of key man life insurance policies received by Parent or its Restricted Subsidiaries after the Closing Date; *provided further*, that such amount in any fiscal year shall be reduced by the amount of Indebtedness incurred in such fiscal year pursuant to clause (22) of the second paragraph of the covenant described below under the caption Incurrence of indebtedness and issuance of preferred stock ;

- (6) the repurchase, redemption or other acquisition or retirement of Equity Interests deemed to occur upon the exercise or exchange of stock options, warrants or other similar rights to the extent such Equity Interests represent a portion of the exercise or exchange price of those stock options, warrants or other similar rights, and the repurchase, redemption or other acquisition or retirement of Equity Interests made in lieu of withholding taxes resulting from the vesting, exercise or exchange of stock options, warrants or other similar rights;
- (7) the declaration and payment of regularly scheduled or accrued dividends to holders of any class or series of Disqualified Stock of Issuer or any Restricted Subsidiary of Issuer issued on or after the Closing Date in accordance with the Debt to Cash Flow Ratio test described below under the caption Incurrence of indebtedness and issuance of preferred stock ;
- (8) Permitted Payments to Parent;
- (9) the repurchase, redemption or other acquisition or retirement for value of any Equity Interests of Parent to the extent necessary to comply with law or to prevent the loss or secure the renewal or reinstatement of any FCC License held by Issuer or any of its Subsidiaries;
- (10) Restricted Investments in an amount equal to 100% of the aggregate amount of any Net Equity Proceeds, less the aggregate amount of any Net Equity Proceeds (x) used for making a Restricted Payment pursuant to clause (2) of this paragraph or (y) taken into account for purposes of incurring Indebtedness pursuant to clause (14) of the definition of Permitted Debt set forth below under the caption Incurrence of indebtedness and issuance of preferred stock ;
- (11) payments made to DT or its Subsidiaries from the Proceeds of the Towers Transaction;
- (12) the repurchase, redemption or other acquisition or retirement for value of any Subordinated Indebtedness pursuant to the provisions similar to those described under Repurchase at the option of holders Change of control and Repurchase at the option of holders Asset sales ; provided that all notes tendered by the holders of the notes in connection with a Change of Control Offer or Asset Sale Offer, as applicable, have been repurchased, redeemed or otherwise acquired for value;
- (13) Restricted Payments in connection with the Cash Payment, as defined in the Business Combination Agreement;
- (14) the making of cash payments in connection with any conversion of Convertible Debt in an aggregate amount since the date of the Closing Date not to exceed the sum of (a) the principal amount of such Convertible Debt plus (b) any payments received by Issuer or any of its Restricted Subsidiaries pursuant to the exercise, settlement or termination of any related Permitted Bond Hedge Transactions; and

(15) other Restricted Payments in an aggregate amount since the Closing Date not to exceed \$375.0 million. The amount of all Restricted Payments (other than cash) will be the Fair Market Value on the date of the Restricted Payment of the asset(s) or securities proposed to be transferred or issued by Issuer or such Restricted Subsidiary, as the case may be, pursuant to the Restricted Payment.

### Incurrence of indebtedness and issuance of preferred stock

Issuer will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, create, incur, issue, assume, guarantee or otherwise become directly or indirectly liable, contingently or otherwise, with respect to

(collectively, *incur*) any Indebtedness (including Acquired Debt), and Issuer will not issue any Disqualified Stock and will not permit any of its Restricted Subsidiaries to issue any shares of Preferred Stock; *provided*, *however*, that Issuer may incur Indebtedness (including Acquired Debt) or issue Disqualified Stock and the Subsidiary Guarantors may incur Indebtedness (including Acquired Debt) or issue Preferred Stock, if the Debt to Cash Flow Ratio for Issuer s most recently ended four full fiscal quarters for which internal financial statements are available immediately preceding the date on which such additional Indebtedness is incurred or such Disqualified Stock or such Preferred Stock is issued, as the case may be, would have been no greater than 6.0 to 1, determined on a pro forma basis (including a pro forma application of the net proceeds therefrom), as if the additional Indebtedness had been incurred or the Disqualified Stock or the Preferred Stock had been issued, as the case may be, at the beginning of such four-quarter period.

The first paragraph of this covenant will not prohibit the incurrence of any of the following items of Indebtedness (collectively, *Permitted Debt*), nor will it prohibit Issuer s Restricted Subsidiaries from issuing the following types of Preferred Stock:

- (1) the incurrence by Issuer and any Subsidiary Guarantor of (a) additional Indebtedness under Credit Facilities, provided that giving effect to such incurrence, the aggregate principal amount (with letters of credit being deemed to have a principal amount equal to the maximum potential liability of Issuer and its Restricted Subsidiaries thereunder) of all Indebtedness under Credit Facilities then outstanding under this paragraph (1), together with any Indebtedness incurred pursuant to the following clause (b), does not to exceed the greater of (x) \$9.0 billion and (y) 150% of the Consolidated Cash Flow of the Issuer and its Subsidiaries for the most recently ended four full fiscal quarters for which financial statements are available, calculated on a pro forma basis in the manner described in the definition of *Debt to Cash Flow Ratio* and (b) without duplication, all Indebtedness incurred to renew, refund, refinance, replace, defease or discharge any Indebtedness incurred pursuant to the foregoing clause (a); *provided, however*, that the maximum amount permitted under this clause (1) shall not be deemed to limit additional Indebtedness under the Credit Facilities to the extent that the incurrence of such additional Indebtedness is permitted pursuant to any of the other provisions of this covenant;
- (2) the incurrence by Issuer and its Restricted Subsidiaries of any Existing Indebtedness;
- (3) the incurrence by Issuer and the Subsidiary Guarantors of Indebtedness represented by the notes to be issued on the Closing Date and the related Note Guarantees;
- (4) the incurrence by Issuer or any of its Restricted Subsidiaries of Indebtedness represented by Capital Lease Obligations, mortgage financings or purchase money obligations, in each case, incurred for the purpose of financing (whether prior to or within 270 days after) all or any part of the purchase price or cost of design, construction, installation or improvement of property, plant or equipment or the Capital Stock of any Person owning such assets used in the business of Issuer or any of its Restricted Subsidiaries, in an aggregate principal amount at any time outstanding, including all Permitted Refinancing Indebtedness incurred to renew, refund, refinance, replace, defease or discharge any Indebtedness incurred pursuant to this clause (4), not to exceed the greater of (x) \$2.5 billion and (y) 5.0% of Issuer s Total Assets, at the time of any such incurrence pursuant to this clause (4);
- (5) the incurrence by Issuer or any of its Restricted Subsidiaries of Permitted Refinancing Indebtedness in exchange for, or the net proceeds of which are used to renew, refund, refinance, replace, defease or discharge any Indebtedness (other than intercompany Indebtedness) that was permitted by the indenture to be incurred under the first paragraph of this covenant or clauses (2), (3), (4), (5), (13), (14), (15), (25) or (26) of this paragraph;
- (6) the incurrence by Issuer or any of its Restricted Subsidiaries of intercompany Indebtedness between or among Parent, Issuer and any of its Restricted Subsidiaries and any Guarantors; *provided*, *however*, that:

- (a) if Issuer or any Subsidiary Guarantor is the obligor on such Indebtedness and the payee is not Issuer or a Guarantor, such Indebtedness must be expressly subordinated to the prior payment in full in cash of all Obligations then due with respect to the notes, in the case of Issuer, or the Note Guarantee, in the case of a Subsidiary Guarantor; and
- (b)(i) any subsequent issuance or transfer of Equity Interests that results in any such Indebtedness being held by a Person other than Parent, Issuer or a Restricted Subsidiary of Issuer, or a Guarantor and (ii) any sale or other transfer of any such Indebtedness to a Person that is not either Parent, Issuer or a Restricted Subsidiary of Issuer, or a Guarantor,

will be deemed, in each case, to constitute an incurrence of such Indebtedness by Issuer or such Restricted Subsidiary, as the case may be, that was not permitted by this clause (6);

- (7) the issuance by any of Issuer s Restricted Subsidiaries to Issuer or to any of its Restricted Subsidiaries of shares of Preferred Stock; *provided, however*, that:
  - (a) any subsequent issuance or transfer of Equity Interests that results in any such Preferred Stock being held by a Person other than Parent, Issuer or a Restricted Subsidiary of Issuer or a Guarantor; and
  - (b) any sale or other transfer of any such Preferred Stock to a Person that is not either Parent, Issuer or a Restricted Subsidiary of Issuer, or a Guarantor,

will be deemed, in each case, to constitute an issuance of such Preferred Stock by such Restricted Subsidiary that was not permitted by this clause (7);

- (8) the incurrence by Issuer or any of its Restricted Subsidiaries of Hedging Obligations (other than for speculative purposes);
- (9) the guarantee by Issuer or any of the Subsidiary Guarantors of Indebtedness of Issuer or a Restricted Subsidiary of Issuer that was permitted to be incurred by another provision of this covenant; *provided* that if the Indebtedness being guaranteed is subordinated to or *pari passu* with the notes, then the guarantee shall be subordinated or *pari passu*, as applicable, to the same extent as the Indebtedness guaranteed;
- (10) the incurrence by Issuer or any of its Restricted Subsidiaries of Indebtedness in respect of workers compensation claims, self-insurance obligations, bankers acceptances, deposits, performance bonds, completion bonds, bid bonds, appeal bonds and surety bonds, indemnity bonds, specific performance or injunctive relief bonds or similar bonds or obligations in the ordinary course of business, and any Guarantees or letters of credit functioning as or supporting any of the foregoing;
- (11) the incurrence by Issuer or any of its Restricted Subsidiaries of Indebtedness arising from (a) the honoring by a bank or other financial institution of a check, draft or similar instrument drawn against insufficient funds, so long as such Indebtedness is covered within five business days of notice to Issuer or any of its Restricted Subsidiaries, (b) in respect of netting, overdraft protection and other arrangements arising under standard business terms of any bank at which Issuer or any Restricted Subsidiary maintains an overdraft, cash pooling or other similar facility or arrangement or (c) in respect of the financing of insurance premiums in the ordinary course of business, *provided* that the aggregate principal amount of Indebtedness incurred pursuant to clauses (11)(b) and (c) shall not, at any time outstanding exceed \$250 million;
- (12) the incurrence by Issuer or any of its Restricted Subsidiaries of Indebtedness in respect of letters of credit required to be issued in connection with any Permitted Joint Venture Investment;

- (13) the incurrence by Issuer or any of its Restricted Subsidiaries of Indebtedness for relocation or clearing obligations relating to Issuer s or any of its Restricted Subsidiary s FCC Licenses in an aggregate principal amount (or accreted value, as applicable), including all Permitted Refinancing Indebtedness incurred to renew, refund, refinance, replace, defease or discharge any Indebtedness incurred pursuant to this clause (13), at any time outstanding not to exceed \$400 million;
- (14) the incurrence by Issuer or any of its Restricted Subsidiaries of Contribution Indebtedness;

- 14 -

- (15) the incurrence by Issuer or any of its Restricted Subsidiaries of Indebtedness (including Acquired Debt or Indebtedness) used to finance an acquisition of or a merger with another Person, *provided* that, Issuer or the Person formed by or surviving any such consolidation or merger (if other than Issuer or a Restricted Subsidiary), on the date of such transaction after giving pro forma effect thereto and any related financing transactions as if the same had occurred at the beginning of the applicable four-quarter period, would either (a) be permitted to incur at least \$1.00 of additional Indebtedness pursuant to the Debt to Cash Flow Ratio test set forth in the first paragraph of this covenant or (b) have a Debt to Cash Flow Ratio no greater than the Debt to Cash Flow Ratio of Issuer immediately prior to such transaction;
- (16) the incurrence by Issuer or any of its Restricted Subsidiaries of Indebtedness arising from agreements providing for indemnification, adjustment of purchase price or similar obligations, or Guarantees or letters of credit, surety bonds or performance bonds securing any obligations of Issuer or any of its Restricted Subsidiaries pursuant to such agreements, in any case incurred in connection with the disposition of any business, assets or Restricted Subsidiary (other than Guarantees of Indebtedness incurred by any Person acquiring all or any portion of such business, assets or Restricted Subsidiary for the purpose of financing such acquisition), so long as the amount does not exceed the gross proceeds actually received by Issuer or any Restricted Subsidiary thereof in connection with such disposition;
- (17) the incurrence by Issuer or any Restricted Subsidiary of Indebtedness constituting reimbursement obligations with respect to letters of credit issued in the ordinary course of business; *provided* that, upon the drawing of such letters of credit, such obligations are reimbursed within 30 days following such drawing;
- (18) the incurrence by Issuer or any Restricted Subsidiary of Indebtedness to the extent that the net proceeds thereof are promptly deposited to defease or to satisfy and discharge the notes;
- (19) [reserved];
- (20) the incurrence by Issuer or any of the Subsidiary Guarantors of additional Indebtedness in an aggregate principal amount (or accreted value, as applicable) at any time outstanding, including all Permitted Refinancing Indebtedness incurred to renew, refund, refinance, replace, defease or discharge any Indebtedness incurred pursuant to this clause (20), not to exceed the greater of (x) \$1.0 billion and (y) 2.0% of the Issuer s Total Assets as of the time of incurrence;
- (21) the incurrence by Issuer or any Restricted Subsidiary of Indebtedness arising in connection with endorsement of instruments for deposit in the ordinary course of business;
- (22) the incurrence by Issuer or any Restricted Subsidiary of Indebtedness evidenced by promissory notes subordinated to the notes and the Note Guarantees issued to current or former employees or directors of Parent, Issuer or any Subsidiary (or their respective spouses or estates) in lieu of cash payments for Capital Stock being repurchased from such Persons, not to exceed, in any twelve-month period, an amount equal to the amount of Restricted Payments that could be made during such twelve-month period pursuant to clause (5) of the third paragraph under the covenant described above under the caption Restricted payments, less the amount of Restricted Payments that have been made during such twelve-month period pursuant to such clause;
- (23) the incurrence by Issuer or any Restricted Subsidiary of Indebtedness consisting of take-or-pay obligations contained in supply agreements entered into in the ordinary course of business;
- (24) to the extent that deposits with, or payments owed to, the FCC in connection with the auction or licensing of Governmental Authorizations are deemed to be Indebtedness, the incurrence by Issuer or any Restricted Subsidiary of such Indebtedness;

- (25) Indebtedness incurred in connection with the Towers Transaction; and
- (26) the incurrence by Restricted Subsidiaries that are not Guarantors of Indebtedness; provided, however, that the aggregate principal amount (or accreted value, as applicable) of all Indebtedness incurred under this

- 15 -

clause (26), when aggregated with the principal amount (or accreted value) of all other Indebtedness then outstanding and incurred pursuant to this clause (26), including all Permitted Refinancing Indebtedness incurred to renew, refund, refinance, replace, defease or discharge any Indebtedness incurred pursuant to this clause (26), does not exceed \$250 million as of the time of incurrence.
Issuer will not incur, and will not permit any Subsidiary Guarantor to incur, any Indebtedness (including Permitted Debt, but excluding Indebtedness permitted by clause (6) above) that is contractually subordinated in right of payment to any other Indebtedness of Issuer or such Subsidiary Guarantor unless such Indebtedness is also contractually subordinated in right of payment to the notes and the applicable Note Guarantee on substantially identical terms; *provided, however*, that no Indebtedness will be deemed to be contractually subordinated in right of payment to any other Indebtedness of Issuer or any Subsidiary Guarantor solely by virtue of such Indebtedness being unsecured or by virtue of such Indebtedness being unsecu

For purposes of (x) determining compliance with this Incurrence of indebtedness and issuance of preferred stock covenant, in the event that an item of proposed Indebtedness meets the criteria of more than one of the categories of Permitted Debt described in clauses (1) through (25) above, or is entitled to be incurred pursuant to the first paragraph of this covenant, Issuer will be permitted to classify all or a portion of such item of Indebtedness on the date of its incurrence, or later reclassify all or a portion of such item of Indebtedness, in any manner that complies with this covenant and (y) determining the amount of Indebtedness that may be incurred pursuant to clause (1)(a)(y) of Permitted Debt the Issuer may elect, pursuant to an officers certificate delivered to the Trustee, to treat all or any portion of the commitment under any Indebtedness (and any refinancing with respect thereto) as being incurred at such time, in which case any subsequent incurrence of Indebtedness under such commitment or refinancing, as the case may be, shall not be deemed, for purposes of this calculation, to be an incurrence at such subsequent time. The accrual of interest, the accretion or amortization of original issue discount, the payment of interest on any Indebtedness in the form of additional Indebtedness with the same terms, the reclassification of Preferred Stock as Indebtedness due to a change in accounting principles or the application thereof, and the payment of dividends on Disgualified Stock in the form of additional shares of the same class of Disgualified Stock will not be deemed to be an incurrence of Indebtedness or an issuance of Disgualified Stock for purposes of this covenant. Notwithstanding any other provision of this covenant, the maximum amount of Indebtedness that Issuer or any Restricted Subsidiary may incur pursuant to this covenant shall not be deemed to be exceeded solely as a result of fluctuations in exchange rates or currency values, and in no event shall the reclassification of any lease or other liability as indebtedness due to a change in accounting principles after the Closing Date be deemed to be an incurrence of Indebtedness. In determining the amount of Indebtedness outstanding under one of the clauses above, the outstanding principal amount of any particular Indebtedness of any Person shall be counted only once and any obligation of such Person or any other Person arising under any guarantee, Lien, letter of credit or similar instrument supporting such Indebtedness shall be disregarded so long as it is permitted to be incurred by the Person or Persons incurring such obligation.

The amount of any Indebtedness outstanding as of any date will be:

- (1) the accreted value of the Indebtedness, in the case of any Indebtedness issued with original issue discount;
- (2) in the case of Hedging Obligations, the termination value of the agreement or arrangement giving rise to such obligations that would be payable by such Person at such time;
- (3) the principal amount of the Indebtedness, in the case of any other Indebtedness; and
- (4) in respect of Indebtedness of another Person secured by a Lien on the assets of the specified Person, the lesser of:
  - (a) the Fair Market Value of such assets at the date of determination; and
  - $(b) \quad \mbox{the amount of the Indebtedness of the other Person.}$

### Liens

Issuer will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, create, incur, assume or suffer to exist any Lien securing Indebtedness upon any asset now owned or hereafter acquired, except Permitted Liens, unless the notes are equally and ratably secured (except that Liens securing Indebtedness that is contractually subordinated to the notes shall be expressly subordinate to any Lien securing the notes to at least the same extent that such Indebtedness is subordinate to the notes).

### Dividend and other payment restrictions affecting subsidiaries

Issuer will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, create or permit to exist or become effective any consensual encumbrance or restriction on the ability of any Restricted Subsidiary to:

- (1) pay dividends or make any other distributions on its Capital Stock to Issuer or any of its Restricted Subsidiaries, or pay any Indebtedness owed to Issuer or any of its Restricted Subsidiaries;
- (2) make loans or advances to Issuer or any of its Restricted Subsidiaries; or

(3) sell, lease or transfer any of its properties or assets to Issuer or any of its Restricted Subsidiaries. However, the preceding restrictions will not apply to encumbrances or restrictions existing under or by reason of:

- (1) agreements or instruments governing Existing Indebtedness, Equity Interests and Credit Facilities as in effect on the Closing Date and any amendments, restatements, modifications, renewals, increases, supplements, refundings, replacements or refinancings of those agreements or instruments; *provided* that the amendments, restatements, modifications, renewals, increases, supplements, refundings, replacements or refinancings are (in the good faith judgment of Board of Directors of Issuer or a senior financial officer of Issuer) not materially more restrictive, taken as a whole, with respect to such dividend and other payment restrictions than those contained in those agreements or instruments on the Closing Date
- (2) agreements or instruments governing Credit Facilities not in effect on the Closing Date so long as either (a) the encumbrances and restrictions contained therein do not impair the ability of any Restricted Subsidiary of Issuer to pay dividends or make any other distributions or payments directly or indirectly to Issuer in an amount sufficient to permit Issuer to pay the principal of, or interest and premium, if any, on the notes, or (b) the encumbrances and restrictions contained therein are no more restrictive, taken as a whole, than those contained in the indenture;
- (3) the indenture, the notes and the Note Guarantees;
- (4) applicable law, rule, regulation or order;
- (5) agreements or instruments with respect to a Person acquired by Issuer or any of its Restricted Subsidiaries as in effect at the time of such acquisition (except to the extent such Indebtedness or Capital Stock was incurred in connection with or in contemplation of such acquisition) or as may be amended, restated, modified, renewed, extended, supplemented, refunded, replaced or refinanced from time to time (so long as the encumbrances and restrictions in any such amendment, restatement, modification, renewal, extension, supplement, refunding, replacement or refinancing are, in the good faith judgment of Issuer s Board of Directors or a senior financial officer of Issuer, not materially more restrictive, taken as a whole, than those in effect on the date of the acquisition), which encumbrance or restriction is not applicable to any Person, or the properties or assets of any Person, other than the Person, or the property or assets of the Person, so

# Table of Contents

acquired; *provided* that, in the case of agreements or instruments governing Indebtedness, such Indebtedness was permitted by the terms of the indenture to be incurred;

(6) customary non-assignment provisions in contracts and licenses entered into in the ordinary course of business and customary contractual restrictions on transfers of all or substantially all assets of a Person;

- 17 -

- (7) any instrument governing any secured Indebtedness or Capital Lease Obligation that imposes restrictions on the assets securing such Indebtedness or the subject of such lease of the nature described in clause (3) of the preceding paragraph;
- (8) any agreement for the sale or other disposition of a Restricted Subsidiary that imposes restrictions of the nature described in clauses (1) and/or (3) of the preceding paragraph on the Restricted Subsidiary pending the sale or other disposition;
- (9) Permitted Refinancing Indebtedness; *provided* that the restrictions contained in the agreements governing such Permitted Refinancing Indebtedness are not materially more restrictive, taken as a whole, than those contained in the agreements governing the Indebtedness being refinanced;
- (10) Liens permitted to be incurred under the provisions of the covenant described above under the caption Liens that limit the right of the debtor to dispose of the assets subject to such Liens;
- (11) provisions limiting the disposition or distribution of assets or property in partnership and joint venture agreements, asset sale agreements, sale-leaseback agreements, stock sale agreements and other similar agreements, which limitation is applicable only to the assets that are the subject of such agreements;
- (12) restrictions on cash or other deposits or net worth imposed by customers, suppliers or landlords or required by insurance, surety or bonding companies, in each case, under contracts entered into in the ordinary course of business;
- (13) restrictions in other Indebtedness, Disqualified Stock or Preferred Stock incurred or issued in compliance with the covenant described under the caption Incurrence of indebtedness and issuance of preferred stock ; *provided* that such restrictions, taken as a whole, are, in the good faith judgment of Issuer s Board of Directors or a senior financial officer of Issuer, not materially more restrictive than those contained in the existing agreements referenced in clauses (1) and (3) above;
- (14) the issuance of Preferred Stock by a Restricted Subsidiary of Issuer or the payment of dividends thereon in accordance with the terms thereof; *provided* that issuance of such Preferred Stock is permitted pursuant to the covenant described above under the caption Incurrence of indebtedness and issuance of preferred stock and the terms of such Preferred Stock do not expressly restrict the ability of such Restricted Subsidiary to pay dividends or make any other distributions on its Capital Stock (other than requirements to pay dividends or liquidation preferences on such Preferred Stock prior to paying any dividends or making any other distributions on such other Capital Stock); and
- (15) any agreement or instrument with respect to Indebtedness incurred, or Preferred Stock issued, by any Restricted Subsidiary, *provided* that the restrictions contained in the agreements or instruments governing such Indebtedness or Preferred Stock (a) either (i) apply only in the event of a payment default or a default with respect to a financial covenant in such agreement or instrument or (ii) will not materially affect Issuer s ability to pay all principal, interest and premium, if any, on the notes, as determined in good faith by Issuer s Board of Directors or a senior financial officer of Issuer, whose determination shall be conclusive; and (b) are not materially more disadvantageous to the holders of the notes than is customary in comparable financings; and
- (16) any agreement or instrument entered into or assumed by the Issuer or any of its Subsidiaries in connection with the Merger, in each case, as such agreements or instruments may be amended, restated, modified, renewed or replaced from time to time; provided that the amendments, restatements, modifications, renewals, and replacements are (in the good faith judgment of Issuer's Board of Directors or a senior financial officer of the Issuer) not materially more restrictive, taken as a whole, with respect to such encumbrances and restrictions than those agreements or instruments as in effect as of the Closing Date; and

(17) restrictions arising from the Towers Transaction.

- 18 -

### Merger, consolidation or sale of assets

Issuer will not: (1) consolidate or merge with or into another Person (whether or not Issuer is the surviving corporation); or (2) directly or indirectly sell, assign, lease, transfer, convey or otherwise dispose of all or substantially all of the properties or assets of Issuer and its Restricted Subsidiaries taken as a whole, in one or more related transactions, to another Person, unless:

- (1) either: (a) Issuer is the surviving corporation; or (b) the Person formed by or surviving any such consolidation or merger (if other than Issuer) or to which such sale, assignment, lease, transfer, conveyance or other disposition has been made is a corporation, limited liability company or partnership organized or existing under the laws of the United States, any state of the United States or the District of Columbia; *provided* that if such Person is not a corporation, such Person immediately causes a Subsidiary that is a corporation organized or existing under the laws of the United States or the District of Columbia to be added as a co-issuer of the notes under the indenture;
- (2) the Person formed by or surviving any such consolidation or merger (if other than Issuer) or the Person to which such sale, assignment, lease, transfer, conveyance or other disposition has been made expressly assumes, by a supplemental indenture, executed and delivered to the trustee, in form reasonably satisfactory to the trustee, the payment of the principal of and any premium and interest on the notes and the performance or observance of every covenant of the indenture on the part of Issuer to be performed or observed;
- (3) immediately after such transaction, no Default or Event of Default exists; and
- (4) Issuer or the Person formed by or surviving any such consolidation or merger (if other than Issuer), or to which such sale, assignment, lease, transfer, conveyance or other disposition has been made would, on the date of such transaction after giving pro forma effect thereto and any related financing transactions as if the same had occurred at the beginning of the applicable four-quarter period, either (a) be permitted to incur at least \$1.00 of additional Indebtedness pursuant to the Debt to Cash Flow Ratio test set forth in the first paragraph of the covenant described above under the caption Incurrence of indebtedness and issuance of preferred stock or (b) have a Debt to Cash Flow Ratio no greater than the Debt to Cash Flow Ratio of Issuer immediately prior to such transaction.

Upon any merger or consolidation, or any sale, transfer, assignment, lease, conveyance or other disposition of all or substantially all of the properties or assets of Issuer and its Restricted Subsidiaries, taken as a whole, in accordance with the first paragraph of this covenant, the successor Person formed by the consolidation or into which Issuer is merged or to which the sale, transfer, assignment, lease, conveyance or other disposition is made, will succeed to and be substituted for Issuer, and may exercise every right and power of Issuer under the indenture with the same effect as if the successor had been named as Issuer therein. When the successor assumes all of Issuer s obligations under the indenture, Issuer will be discharged from those obligations.

This Merger, consolidation or sale of assets covenant will not apply to:

- a merger of Issuer with a direct or indirect Subsidiary of Parent solely for the purpose of reincorporating Issuer in another jurisdiction in the United States so long as the amount of Indebtedness of the Issuer and its Restricted Subsidiaries is not increased thereby;
- (2) any consolidation or merger, or any sale, assignment, transfer, conveyance, lease or other disposition of assets between or among Issuer and its Restricted Subsidiaries; or

(3) the Merger. *Transactions with affiliates* 

Issuer will not, and will not permit any of its Restricted Subsidiaries to, make any payment to, or sell, lease, transfer or otherwise dispose of any of its properties or assets to, or purchase any property or assets from, or enter

into or make or amend any transaction, contract, agreement, understanding, loan, advance or guarantee with, or for the benefit of, any Affiliate of Issuer (each, an *Affiliate Transaction*), in any one or series of related transactions involving aggregate payments or consideration in excess of \$50.0 million, unless:

- (1) the Affiliate Transaction is on terms that, taken as a whole, are no less favorable to Issuer or the relevant Restricted Subsidiary than those that would have been obtained in a comparable transaction by Issuer or such Restricted Subsidiary with an unrelated Person; and
- (2) Issuer delivers to the trustee:
  - (a) with respect to any Affiliate Transaction or series of related Affiliate Transactions involving aggregate consideration in excess of \$100.0 million, an officers certificate certifying that such Affiliate Transaction complies with this covenant; and
  - (b) with respect to any Affiliate Transaction or series of related Affiliate Transactions involving aggregate consideration in excess of \$250.0 million, a resolution of the Board of Directors of Issuer set forth in an officers certificate certifying that such Affiliate Transaction complies with this covenant and that such Affiliate Transaction has been approved by a majority of the disinterested members of the Board of Directors of Issuer.

The following items will not be deemed to be Affiliate Transactions and, therefore, will not be subject to the provisions of the prior paragraph:

- (1) any employment agreement, employee benefit plan, agreement or plan relating to employee, officer or director compensation or severance, officer or director indemnification agreement or any similar arrangement entered into by Issuer, any of its Restricted Subsidiaries or Parent existing on the Closing Date, or entered into thereafter in the ordinary course of business, and any indemnitees or other transactions permitted or required by bylaw, statutory provisions or any of the foregoing agreements, plans or arrangements and payments pursuant thereto;
- (2) transactions between or among Parent, Issuer and/or its Restricted Subsidiaries;
- (3) transactions with a Person (other than an Unrestricted Subsidiary of Issuer) that is an Affiliate of Issuer solely because Issuer owns, directly or through a Restricted Subsidiary, an Equity Interest in, or controls, such Person;
- (4) [reserved];
- (5) any issuance of Equity Interests (other than Disqualified Stock) of Issuer to, or receipt of any capital contribution from, any Affiliate of Issuer;
- (6) transactions in connection with any Permitted Joint Venture Investment;
- (7) any Permitted Investments or Restricted Payments that do not violate the provisions of the indenture described above under the caption Restricted payments;

- (8) any agreement listed (x) on Schedule 3.2(r) Related-Party Agreements to the TMUS Disclosure Letter to the Business Combination Agreement and (y) under the section entitled Transactions with Related Persons and Approval in the proxy statement of Parent filed with the SEC under cover of Schedule 14A on April 16, 2012, and any amendments to, replacements of, or orders pursuant to such agreements so long as any such amendments, replacements, or orders, taken as a whole, are not (in the good faith judgment of Issuer s Board of Directors or a senior financial officer of Issuer) more disadvantageous to Issuer or to the holders of the notes in any material respect than the original contracts, agreements or understandings as in effect on the Closing Date;
- (9) transactions with customers, clients, suppliers, or purchasers or sellers of goods or services, in each case in the ordinary course of business and otherwise in compliance with the terms of the indenture, *provided* that in the good faith determination of Issuer s Board of Directors or a senior financial officer of Issuer, such

- 20 -

transactions are on terms, taken as a whole, not materially less favorable to Issuer or the applicable Restricted Subsidiary than those that could reasonably be expected to be obtained in a comparable transaction at such time on an arm s length basis from a Person that is not an Affiliate of Issuer;

- (10) issuances, purchases or repurchases of notes or other Indebtedness of Issuer or its Restricted Subsidiaries or solicitations of amendments, waivers or consents in respect of notes or such other Indebtedness, if such issuance, purchase, repurchase or solicitation is approved by a majority of the disinterested members of the Board of Directors of Issuer;
- (11) reasonable payments made for any financial advisory, financing, underwriting, placement or syndication services approved by Issuer s Board of Directors or a senior financial officer of Issuer in good faith;
- (12) amendments, extensions, replacements and other modifications of transactions with Affiliates otherwise permitted by the indenture, provided that in the good faith determination of the Issuer s Board of Directors or a senior financial officer of the Issuer, such amendments, extensions, replacements or other modifications, taken as a whole, are no less favorable in any material respect to Issuer or the applicable Restricted Subsidiary than the transaction or transactions being amended, extended, replaced or modified;
- (13) (i) the Business Combination Agreement and any Ancillary Agreements, as defined in the Business Combination Agreement, in each case, as the same may be amended, modified, supplemented or replaced from time to time on terms that, taken as a whole, in the good faith determination of the Issuer s Board of Directors or a senior financial officer of the Issuer, are not materially less favorable to the Issuer or the applicable Restricted Subsidiary than those of the agreement being amended, modified, supplemented or replaced, (ii) transactions or agreements relating to the DT Notes (as defined in the Business Combination Agreement) and the TMUS Working Capital Facility (as defined in the Business Combination Agreement), each as may be amended, modified, or supplemented from time to time, and any indebtedness incurred in connection with the refinancing of the foregoing, on terms that, taken as a whole, in the good faith determination of the Issuer s Board of Directors or a senior financial officer of the Issuer, are not materially less favorable to the Issuer than those of the DT Notes or TMUS Working Capital Facility, as applicable, and (iii) transactions between the Issuer and its Restricted Subsidiaries, on the one hand, and any Designated Tower Entities that have been designated as Unrestricted Subsidiaries, on the other hand, in connection with the Towers Transaction; and
- (14) so long as 100% of the notes are held by Permitted Holders, any transactions between the Issuer or any Restricted Subsidiary, on the one hand, and DT or any Subsidiary thereof (other than Parent, Issuer or any Subsidiary thereof), on the other hand, and thereafter any agreements evidencing such transactions.

### **Business activities**

Issuer will not, and will not permit any of its Restricted Subsidiaries to, engage in any business other than Permitted Businesses, except to such extent as would not be material to Issuer and its Restricted Subsidiaries taken as a whole.

### Additional note guarantees

If (a) Issuer or any of Issuer s Domestic Restricted Subsidiaries acquires or creates another Domestic Restricted Subsidiary that is a Wholly-Owned Subsidiary and is neither a Designated Tower Entity nor an Immaterial Subsidiary after the Closing Date or (b) any Subsidiary of Issuer guarantees any Specified Issuer Indebtedness of Issuer after the Closing Date or (c) Parent or any Subsidiary of Parent acquires or creates a Subsidiary that directly or indirectly owns equity interests of Issuer, then such Person will become a Guarantor and execute a supplemental indenture and deliver an opinion of counsel reasonably satisfactory to the trustee within 10 business days after the date on which it was acquired or created or guarantees such Specified Issuer Indebtedness, as applicable, or reasonably promptly thereafter.

### Designation of restricted and unrestricted subsidiaries

The Board of Directors of Issuer may designate any Restricted Subsidiary to be an Unrestricted Subsidiary if that designation would not cause a Default. If a Restricted Subsidiary is designated as an Unrestricted Subsidiary, (i) the aggregate Fair Market Value of all outstanding Investments owned by Issuer and its Restricted Subsidiaries in the Subsidiary designated as an Unrestricted Subsidiary will be deemed to be an Investment made as of the time of the designation and will reduce the amount available for Restricted Payments under the covenant described above under the caption Restricted payments or under one or more clauses of the definition of Permitted Investments, as determined by Issuer in its discretion, and (ii) any Guarantee by Issuer or any Restricted Subsidiary thereof of any Indebtedness of the Restricted Subsidiary being so designated will be deemed to be an incurrence of Indebtedness by Issuer or such Restricted Subsidiary (or both, if applicable) at the time of such designation. That designation will only be permitted if the Investment and/or incurrence of Indebtedness would be permitted at that time and if the Restricted Subsidiary otherwise meets the definition of an Unrestricted Subsidiary. The Board of Directors of Issuer may redesignate any Unrestricted Subsidiary to be a Restricted Subsidiary if that redesignation would not cause a Default.

Any designation of a Subsidiary of Issuer as an Unrestricted Subsidiary will be evidenced to the trustee by filing with the trustee a certified copy of a resolution of the Board of Directors giving effect to such designation and an officers certificate certifying that such designation complied with the preceding conditions and was permitted by the covenant described above under the caption Restricted payments. The Board of Directors of Issuer may at any time designate any Unrestricted Subsidiary to be a Restricted Subsidiary of Issuer; *provided* that such designation will be deemed to be an incurrence of Indebtedness by a Restricted Subsidiary of Issuer of any outstanding Indebtedness of such Unrestricted Subsidiary, and such designation will only be permitted if (1) such Indebtedness is permitted under the covenant described under the caption Incurrence of indebtedness and issuance of preferred stock, calculated on a pro forma basis as if such designation had occurred at the beginning

of the four-quarter reference period; and (2) no Default would be in existence following, and as a result of, such designation.

Notwithstanding the foregoing, Issuer may at any time and from time to time designate any Designated Entity, by written notice to the trustee, as an Unrestricted Subsidiary, and any such Subsidiary shall upon such notice immediately be designated and deemed an Unrestricted Subsidiary, without any further action by Issuer (and, for the avoidance of doubt, shall not require delivery of a resolution of the Board of Directors or of an officers certificate) (each, a Specified Unrestricted Subsidiary Designation ). The aggregate Fair Market Value of all outstanding Investments owned by Issuer and its Restricted Subsidiaries in such Designated Entities so designated as Unrestricted Subsidiaries will, as calculated and to the extent permitted by clause (18) of the definition of Permitted Investments, be deemed to be an Investment made as of the time of such Specified Unrestricted Subsidiary Designation under such clause (18), and not reduce the amount available for Restricted Payments under the covenant described above under the caption Restricted payments .

#### Payments for consent

Issuer will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, pay or cause to be paid any consideration to or for the benefit of any holder of notes for or as an inducement to any consent, waiver or amendment of any of the terms or provisions of the indenture or the notes unless such consideration is offered to be paid and is paid to all holders of the notes that consent, waive or agree to amend in the time frame set forth in the solicitation documents relating to such consent, waiver or amendment.

#### Reports

Whether or not required by the rules and regulations of the SEC, so long as any notes are outstanding, Parent will file a copy of each of the reports referred to in clauses (1) and (2) below with the SEC for public availability

- 22 -

within the time periods (including all applicable extension periods) specified in the SEC rules and regulations applicable to such reports (unless the SEC will not accept such a filing):

(1) all quarterly and annual financial reports that would be required to be contained in a filing with the SEC on Forms 10-Q and 10-K if Parent were required to file such reports, including a Management s Discussion and Analysis of Financial Condition and Results of Operations and, with respect to the annual information only, a report on the annual financial statements by its certified independent accountants; and

(2) all current reports that would be required to be filed with the SEC on Form 8-K if Parent or Issuer were required to file such reports; *provided* that the availability of the foregoing reports on the SEC s EDGAR service (or successor thereto) shall be deemed to satisfy Issuer s delivery obligations to the trustee and any holder of notes.

All such reports will be prepared in all material respects in accordance with all of the rules and regulations applicable to such reports; *provided* that, if neither Parent nor Issuer is required under the rules and regulations of the SEC to file such reports with the SEC for public availability, such reports need not be prepared in accordance with all of the rules and regulations applicable to such reports and shall only be required to include the information or disclosure that would be required by such form to the extent that, and in the same general style of presentation as, the same or substantially similar information or disclosure is also included in this prospectus supplement. Each annual report on Form 10-K will include a report on Parent s consolidated financial statements by Parent s certified independent accountants. The Issuer will at all times comply with TIA §314(a).

If the SEC will not accept Parent s or Issuer s filings for any reason, Parent or Issuer will post the reports referred to in the preceding paragraphs on its website, on intralinks.com or another website (which may be password protected, provided that Issuer makes reasonable efforts to notify the holders of the notes of the password and other information required to access such reports on its website) within the time periods that would apply if Parent were required to file those reports with the SEC (including all applicable extension periods). If (i) Issuer has designated any of its Subsidiaries as Unrestricted Subsidiaries or (ii) the combined operations of Parent and its Subsidiaries, excluding the operations of Issuer and its Restricted Subsidiaries and excluding cash and Cash Equivalents, would, if held by a single Unrestricted Subsidiary of Issuer, constitute a Significant Subsidiary of Issuer, then the quarterly and annual financial information required by the preceding paragraphs will include a reasonably detailed presentation, either on the face of the financial statements or in the footnotes thereto, and in Management s Discussion and Analysis of Financial Condition and Results of Operations, of (A) in the case of (i) above, the financial condition and results of operations of Issuer and its Restricted Subsidiaries separate from the financial condition and results of operations of Issuer and its Restricted Subsidiaries separate from the financial condition and results of operations of the Unrestricted Subsidiaries separate from the financial condition and results of operations of Parent and its other Subsidiaries; *provided however*, that the requirements of this paragraph shall not apply if Parent or Issuer files with the SEC the reports referred to in clauses (1) and (2) of the first paragraph of this covenant, and any such report contains the information required in this paragraph.

In addition, Issuer and the Guarantors agree that, for so long as any notes remain outstanding, if at any time they are not required to file with the SEC the reports required by the preceding paragraphs, they will furnish to the holders of notes and prospective investors, upon their request, the information required to be delivered pursuant to Rule 144A(c)(2) under the Securities Act.

#### Events of default and remedies

Each of the following is an Event of Default :

(1) default for 30 days in the payment when due of interest on the notes;

- 23 -

- (2) default in the payment when due (at maturity, upon redemption or otherwise) of the principal of, or premium, if any, on, the notes;
- (3) failure by Issuer for 120 days after notice to Issuer by the trustee or the holders of at least 25% in aggregate principal amount of the notes then outstanding voting as a single class to comply with the provisions described under the caption Reports ;
- (4) failure by Issuer or any of its Restricted Subsidiaries for 30 days after notice to Issuer by the trustee or the holders of at least 25% in aggregate principal amount of the notes then outstanding voting as a single class to comply with the provisions described under the captions Repurchase at the option of holders Change of control triggering event or Repurchase at the option of holders Asset sales (in each case other than a failure to purchase notes that will constitute an Event of Default under clause (2) above), or Certain covenants Merger, consolidation or sale of assets ;
- (5) failure by Issuer or any of its Restricted Subsidiaries for 90 days after notice to Issuer by the trustee or the holders of at least 25% in aggregate principal amount of the notes then outstanding voting as a single class to comply with any of the other agreements in the indenture;
- (6) default under any mortgage, indenture or instrument under which there may be issued or by which there may be secured or evidenced any Indebtedness for money borrowed by Issuer or any of its Restricted Subsidiaries that is a Significant Subsidiary (or any Restricted Subsidiaries that together would constitute a Significant Subsidiary) (or the payment of which is guaranteed by Issuer or any of its Restricted Subsidiaries that would constitute a Significant Subsidiary), whether such Indebtedness or Guarantee now exists, or is created after the Closing Date, if that default:
  - (a) is caused by a failure to pay principal of, or interest or premium, if any, on, such Indebtedness prior to the expiration of the grace period provided in such Indebtedness on the date of such default (a *Payment Default*); or

(b) results in the acceleration of such Indebtedness prior to its express maturity; and, in each case, the principal amount of any such Indebtedness, together with the principal amount of any other such Indebtedness under which there has been a Payment Default or the maturity of which has been so accelerated, aggregates \$100.0 million or more, in each case for so long as such failure or acceleration is continuing;

- (7) failure by Issuer or any of its Restricted Subsidiaries that is a Significant Subsidiary (or any Restricted Subsidiaries that together would constitute a Significant Subsidiary) to pay or discharge final judgments entered by a court or courts of competent jurisdiction aggregating in excess of \$100.0 million (to the extent not covered by insurance), which judgments are not paid, discharged or stayed for a period of 60 consecutive days following entry of such final judgment or decree during which a stay of enforcement of such final judgment or decree, by reason of pending appeal or otherwise, is not in effect;
- (8) except as permitted by the indenture, any Note Guarantee is held in any judicial proceeding to be unenforceable or invalid or ceases for any reason to be in full force and effect, or any Guarantor, or any Person acting on behalf of any such Guarantor, denies or disaffirms its obligations under its Note Guarantee; and

(9) certain events of bankruptcy or insolvency described in the indenture with respect to Issuer or any of its Restricted Subsidiaries that is a Significant Subsidiary or any group of Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary.
 In the case of an Event of Default arising from certain events of bankruptcy or insolvency, with respect to Issuer, any Restricted Subsidiary of Issuer that is a Significant Subsidiary or any group of Restricted Subsidiaries of Issuer that, taken together, would constitute a Significant Subsidiary of Issuer that is a Significant Subsidiary or any group of Restricted Subsidiaries of Issuer that, taken together, would constitute a Significant Subsidiary all outstanding notes will become due and payable immediately without further action or notice. However, the effect of such

provision may be limited by

- 24 -

applicable laws. If any other Event of Default occurs and is continuing, the trustee or the holders of at least 25% in aggregate principal amount of the then outstanding notes may declare all the notes to be due and payable immediately.

Subject to certain limitations, holders of a majority in aggregate principal amount of the then outstanding notes may direct the trustee in its exercise of any trust or power. The trustee may withhold from holders of the notes notice of any continuing Default or Event of Default if it determines that withholding notice is in their interest, except a Default or Event of Default relating to the payment of principal, interest or premium, if any.

Subject to the provisions of the indenture relating to the duties of the trustee, in case an Event of Default occurs and is continuing, the trustee will be under no obligation to exercise any of the rights or powers under the indenture at the request or direction of any holders of notes unless such holders have offered to the trustee indemnity or security satisfactory to it against any loss, liability or expense. Except to enforce the right to receive payment of principal, premium, if any, or interest when due, no holder of a note may pursue any remedy with respect to the indenture or the notes unless:

- (1) such holder has previously given the trustee notice that an Event of Default is continuing;
- (2) holders of at least 25% in aggregate principal amount of the then outstanding notes have requested the trustee to pursue the remedy;
- (3) such holders have offered the trustee security or indemnity satisfactory to it against any loss, liability or expense;
- (4) the trustee has not complied with such request within 60 days after the receipt of the request and the offer of security or indemnity; and
- (5) holders of a majority in aggregate principal amount of the then outstanding notes have not given the trustee a direction inconsistent with such request within such 60-day period.

The holders of a majority in aggregate principal amount of the then outstanding notes by written notice to the trustee may, on behalf of the holders of all of the notes, rescind an acceleration or waive any existing Default or Event of Default and its consequences under the indenture except a continuing Default or Event of Default in the payment of interest or premium, if any, on, or the principal of, the notes.

In the case of any Event of Default occurring by reason of any willful action (or inaction) taken (or not taken) by or on behalf of Issuer with the intention of avoiding payment of the premium that Issuer would have had to pay if Issuer then had elected to redeem the notes pursuant to the optional redemption provisions of the indenture, an equivalent premium will also become and be immediately due and payable to the extent permitted by law upon the acceleration of the notes.

Issuer is required to deliver to the trustee annually a statement regarding compliance with the indenture. Upon becoming aware of any Default or Event of Default, Issuer is required to deliver to the trustee a statement specifying such Default or Event of Default.

#### No personal liability of directors, officers, employees and stockholders

No director, officer, member, manager, partner, employee, incorporator or stockholder of Issuer or any Guarantor, as such, will have any liability for any obligations of Issuer or the Guarantors under the notes, the indenture, the Note Guarantees, or for any claim based on, in respect of, or by reason of, such obligations or their creation. Each holder of notes by accepting a note waives and releases all such liability. The waiver and release are part of the consideration for issuance of the notes. The waiver may not be effective to waive liabilities under the federal securities laws.

#### Legal defeasance and covenant defeasance

Issuer may at any time, at the option of its Board of Directors evidenced by a resolution set forth in an officers certificate, elect to have all of its obligations discharged with respect to the outstanding notes and all obligations of the Guarantors discharged with respect to their Note Guarantees (*Legal Defeasance*) except for:

- (1) the rights of holders of outstanding notes to receive payments in respect of the principal of, or interest or premium, if any, on, such notes when such payments are due from the trust referred to below;
- (2) Issuer s obligations with respect to the notes concerning issuing temporary notes, registration of notes, mutilated, destroyed, lost or stolen notes and the maintenance of an office or agency for payment and money for security payments held in trust;
- (3) the rights, powers, trusts, duties and immunities of the trustee, and Issuer s and the Guarantors obligations in connection therewith; and
- (4) the Legal Defeasance and Covenant Defeasance provisions of the indenture.

In addition, Issuer may, at its option and at any time, elect to have the obligations of Issuer and the Guarantors released with respect to the provisions of the indenture described above under Repurchase at the option of holders and under the caption Certain covenants (other than the covenant described under the caption Certain covenants Merger, consolidation or sale of assets, except to the extent described below) and the limitation imposed by clause (4) under the caption Certain covenants Merger, consolidation or sale of assets (such release and termination being referred to as *Covenant Defeasance*), and thereafter any omission to comply with such obligations or provisions will not constitute a Default or Event of Default. In the event Covenant Defeasance occurs in accordance with the indenture, the Events of Default described under clauses (3) through (7) under the caption Events of default and remedies and the Event of Default described under clause (9) under the caption Events of default and remedies of Issuer), in each case, will no longer constitute an Event of Default.

In order to exercise either Legal Defeasance or Covenant Defeasance:

- (1) Issuer must irrevocably deposit with the trustee, in trust, for the benefit of the holders of the notes, cash in U.S. dollars, non-callable Government Securities, or a combination of cash in U.S. dollars and non-callable Government Securities, in amounts as will be sufficient, in the opinion of a nationally recognized investment bank, appraisal firm or firm of independent public accountants, to pay the principal of, or interest and premium, if any, on, the outstanding notes on the stated date for payment thereof or on the applicable redemption date, as the case may be, and Issuer must specify whether the notes are being defeased to such stated date for payment or to a particular redemption date;
- (2) in the case of Legal Defeasance, Issuer must deliver to the trustee an opinion of counsel reasonably acceptable to the trustee (which opinion of counsel may be subject to customary assumptions, qualifications and exclusions) confirming that (a) Issuer has received from, or there has been published by, the Internal Revenue Service a ruling or (b) since the Closing Date, there has been a change in the applicable U.S. federal income tax law, in either case to the effect that, and based thereon such opinion of counsel will confirm that, the holders of the outstanding notes will not recognize income, gain or loss for U.S. federal income tax purposes as a result of such Legal Defeasance and will be subject to U.S. federal income tax on the same amounts, in the same manner and at the same times as would have been the case if such Legal Defeasance had not occurred;
- (3) in the case of Covenant Defeasance, Issuer must deliver to the trustee an opinion of counsel reasonably acceptable to the trustee confirming that the holders of the outstanding notes will not recognize income, gain or loss for U.S. federal income tax purposes as a result of such Covenant Defeasance and will be subject to U.S. federal income tax on the same amounts, in the same manner and at the same times as would have been the case if such Covenant Defeasance had not occurred;

# - 26 -

- (4) no Default or Event of Default has occurred and is continuing on the date of such deposit (other than a Default or Event of Default resulting from the borrowing of funds, or the imposition of Liens in connection therewith, to be applied to such deposit, or a Default or Event of Default that will be cured by such Covenant Defeasance or Legal Defeasance) and the deposit will not result in a breach or violation of, or constitute a default under, any material instrument to which Issuer or any Guarantor is a party or by which Issuer or any Guarantor is bound;
- (5) such Legal Defeasance or Covenant Defeasance will not result in a breach or violation of, or constitute a default under, any material agreement or instrument (other than the indenture) to which Issuer or any of its Subsidiaries is a party or by which Issuer or any of its Subsidiaries is bound;
- (6) Issuer must deliver to the trustee an officers certificate stating that the deposit was not made by Issuer with the intent of preferring the holders of notes over the other creditors of Issuer with the intent of defeating, hindering, delaying or defrauding any creditors of Issuer or others;
- (7) Issuer must deliver to the trustee an officers certificate, stating that all conditions precedent relating to the Legal Defeasance or the Covenant Defeasance have been complied with; and
- (8) Issuer must deliver to the trustee an opinion of counsel (which opinion of counsel may be subject to customary assumptions, qualifications and exclusions), stating that all conditions precedent set forth in clauses (2), (3) and (5) of this paragraph, as applicable, have been complied with; *provided* that the opinion of counsel with respect to clause (5) of this paragraph may be to the knowledge of such counsel. Amendment, supplement and waiver

Except as provided in the next two succeeding paragraphs, the indenture or the notes or the Note Guarantees may be amended or supplemented with the consent of the holders of at least a majority in aggregate principal amount of the notes then outstanding (including consents obtained in connection with a purchase of, or tender offer or exchange offer for, notes), and any existing Default or Event of Default or compliance with any provision of the indenture or the notes or the Note Guarantees may be waived with the consent of the holders of a majority in aggregate principal amount of the then outstanding notes (including consents obtained in connection with a purchase of, or tender offer for, notes). Without the consent of each holder of notes affected, an amendment, supplement or waiver may not (with respect to any notes held by a non-consenting holder):

- (1) reduce the principal amount of notes whose holders must consent to an amendment, supplement or waiver;
- (2) reduce the principal of or change the fixed maturity of any note or alter the provisions with respect to the redemption of the notes (other than provisions relating to the covenants described above under the caption Repurchase at the option of holders );
- (3) reduce the rate of or change the time for payment of interest on any note;
- (4) waive a Default or Event of Default in the payment of principal of, or interest or premium, if any, on, the notes (except a rescission of acceleration of the notes by the holders of at least a majority in aggregate principal amount of the then outstanding notes and a waiver of the payment default that resulted from such acceleration);
- (5) make any note payable in money other than that stated in the notes;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

- (6) make any change in the provisions of the indenture relating to waivers of past Defaults or the rights of holders of notes to receive payments of principal of, or interest or premium, if any, on, the notes;
- (7) waive a redemption payment with respect to any note (other than a payment required by one of the covenants described above under the caption Repurchase at the option of holders );
- (8) release any Guarantor from any of its obligations under its Note Guarantee or the indenture, except in accordance with the terms of the indenture; or

- 27 -

(9) make any change in the preceding amendment and waiver provisions.

Notwithstanding the preceding, without the consent of any holder of notes, Issuer, the Guarantors and the trustee may amend or supplement the indenture, the notes or the Note Guarantees:

- (1) to cure any ambiguity, defect or inconsistency;
- (2) to provide for uncertificated notes in addition to or in place of certificated notes;
- (3) to provide for the assumption of Issuer s or a Guarantor s obligations to holders of notes and Note Guarantees in the case of a merger or consolidation or sale of all or substantially all of Issuer s or such Guarantor s assets, as applicable;
- (4) to effect the release of a Guarantor from its Note Guarantee and the termination of such Note Guarantee, all in accordance with the provisions of the indenture governing such release and termination;
- (5) to add any Guarantor or Note Guarantee or to secure the notes or any Note Guarantee;
- (6) to make any change that would provide any additional rights or benefits to the holders of notes or that does not adversely affect the legal rights under the indenture of any such holder [in any material respect]<sup>16</sup>;
- (7) to comply with requirements of the SEC in order to effect or maintain the qualification of the indenture under the Trust Indenture Act;
- (8) to conform the text of the indenture, the Note Guarantees, or the notes to any provision of this description of notes to the extent that such provision in this description of notes was intended to be a verbatim recitation of a provision of the indenture, the Note Guarantees, or the notes, in each case, as [conclusively]<sup>17</sup> evidenced by an officers certificate;
- (9) to provide for the issuance of additional notes in accordance with the limitations set forth in the indenture as of the Closing Date; or

(10) to allow any Guarantor to execute a supplemental indenture and/or a Note Guarantee with respect to the notes. The consent of the holders of the notes is not necessary under the indenture to approve the particular form of any proposed amendment or waiver. It is sufficient if such consent approves the substance of the proposed amendment or waiver.

#### Satisfaction and discharge

The indenture will be discharged and will cease to be of further effect as to all notes issued thereunder, when:

#### (1) either:

(a)

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

all notes that have been authenticated, except lost, stolen or destroyed notes that have been replaced or paid and notes for whose payment money has been deposited in trust and thereafter repaid to Issuer, have been delivered to the trustee for cancellation; or

- (b) all notes that have not been delivered to the trustee for cancellation have become due and payable by reason of the sending of a notice of redemption or otherwise or will become due and payable within one year and Issuer or any Guarantor has irrevocably deposited or caused to be deposited with the trustee as trust funds in trust solely for the benefit of the holders, cash in U.S. dollars, non-callable Government Securities, or a combination of cash in U.S. dollars and non-callable Government Securities, in amounts as will be sufficient, without consideration of any reinvestment of interest, to pay and discharge the entire Indebtedness on the notes not delivered to the trustee for cancellation for principal, premium, if any, and accrued interest to the date of maturity or redemption;
- <sup>16</sup> Subject to acceptance by the trustee.
- <sup>17</sup> Subject to acceptance by the trustee.

- 28 -

- (2) no Default or Event of Default has occurred and is continuing on the date of such deposit (other than a Default or Event of Default resulting from the borrowing of funds, or the imposition of any Liens in connection therewith, to be applied to such deposit, or a Default or Event of Default that will be cured by such discharge);
- (3) such deposit will not result in a breach or violation of, or constitute a default under, any material agreement or instrument (other than the indenture) to which Issuer or any Guarantor is a party or by which Issuer or any Guarantor is bound;
- (4) Issuer or any Guarantor has paid or caused to be paid all sums payable by it under the indenture; and
- (5) Issuer has delivered irrevocable instructions to the trustee under the indenture to apply the deposited money toward the payment of the notes at maturity or on the redemption date, as the case may be.

In addition, Issuer must deliver to the trustee (a) an officers certificate, stating that all conditions precedent set forth in clauses (1) through (5) above have been satisfied, and (b) an opinion of counsel (which opinion of counsel may be subject to customary assumptions and qualifications), stating that all conditions precedent set forth in clauses (3) and (5) above have been satisfied; *provided* that the opinion of counsel with respect to clause (3) above may be to the knowledge of such counsel.

#### Governing law

The indenture, the notes and the Note Guarantees will be governed by the laws of the State of New York.

#### Concerning the trustee

[We maintain ordinary banking relationships with Wells Fargo Bank, N.A. and its affiliates. Wells Fargo Bank, N.A. serves as syndication agent and Wells Fargo Securities, LLC, an affiliate of Wells Fargo Bank, N.A., serves as joint lead arranger and joint book-running manager under our senior secured credit facility.]

If the trustee becomes a creditor of Issuer or any Guarantor, the indenture limits the right of the trustee to obtain payment of claims in certain cases, or to realize on certain property received in respect of any such claim as security or otherwise. The trustee will be permitted to engage in other transactions; however, if it acquires any conflicting interest when a Default is continuing it must eliminate such conflict within 90 days of the date such conflict arises, apply to the SEC for permission to continue as trustee (if the indenture has been qualified under the Trust Indenture Act) or resign.

The holders of a majority in aggregate principal amount of the then outstanding notes will have the right to direct the time, method and place of conducting any proceeding for exercising any remedy available to the trustee, subject to certain exceptions. The indenture provides that in case an Event of Default occurs and is continuing, the trustee will be required, in the exercise of its power, to use the degree of care of a prudent man in the conduct of his own affairs.

Subject to such provisions, the trustee will be under no obligation to exercise any of its rights or powers under the indenture at the request of any holder of notes, unless such holder has offered to the trustee security and indemnity satisfactory to it against any loss, liability or expense.

#### Book-entry, delivery and form

Except as set forth below, the notes will be issued in registered, global form ( *Global Notes* ). The Global Notes will be deposited upon issuance with the trustee as custodian for DTC and registered in the name of DTC or its nominee, in each case, for credit to an account of a direct or indirect participant in DTC as described below.

Except as set forth below, the Global Notes may be transferred, in whole and not in part, only to another nominee of DTC or to a successor of DTC or its nominee. Beneficial interests in the Global Notes may not be exchanged for definitive notes in registered certificated form (*Certificated Notes*) except in the limited circumstances described below. See Exchange of global notes for certificated notes. Except in the limited circumstances described below. See in the Global Notes will not be entitled to receive physical delivery of such notes in certificated form.

Transfers of beneficial interests in the Global Notes will be subject to the applicable rules and procedures of DTC and its direct or indirect participants (including, if applicable, those of Euroclear System (*Euroclear*) and Clearstream Banking, S.A. (*Clearstream*)), which may change from time to time.

## [Depository procedures

The following description of the operations and procedures of DTC, Euroclear and Clearstream are provided solely as a matter of convenience. These operations and procedures are solely within the control of the respective settlement systems and are subject to changes by them. Issuer takes no responsibility for these operations and procedures and urges investors to contact the system or their participants directly to discuss these matters.

DTC has advised Issuer that DTC is a limited-purpose trust company created to hold securities for its participating organizations (collectively, the *Participants*) and to facilitate the clearance and settlement of transactions in those securities between the Participants through electronic book-entry changes in accounts of its Participants. The Participants include securities brokers and dealers (including the underwriters), banks, trust companies, clearing corporations and certain other organizations. Access to DTC s system is also available to other entities such as banks, brokers, dealers and trust companies that clear through or maintain a custodial relationship with a Participant, either directly or indirectly (collectively, the *Indirect Participants*). Persons who are not Participants may beneficially own securities held by or on behalf of DTC only through the Participants or the Indirect Participants. The ownership interests in, and transfers of ownership interests in, each security held by or on behalf of DTC are recorded on the records of the Participants and Indirect Participants.

DTC has also advised Issuer that, pursuant to procedures established by it:

- (1) upon deposit of the Global Notes, DTC will credit the accounts of the Participants designated by the underwriters with portions of the principal amount of the Global Notes; and
- (2) ownership of these interests in the Global Notes will be shown on, and the transfer of ownership of these interests will be effected only through, records maintained by DTC (with respect to the Participants) or by the Participants and the Indirect Participants (with respect to other owners of beneficial interest in the Global Notes).

Investors in the Global Notes who are Participants may hold their interests therein directly through DTC. Investors in the Global Notes who are not Participants may hold their interests therein indirectly through organizations (including Euroclear and Clearstream) which are Participants. Euroclear and Clearstream will hold interests in the Global Notes on behalf of their participants through customers securities accounts in their respective names on the books of their respective depositories, which are Euroclear Bank S.A./N.V., as operator of Euroclear, and Citibank, N.A., as operator of Clearstream. All interests in a Global Note, including those held through Euroclear or Clearstream, may be subject to the procedures and requirements of DTC. Those interests held through Euroclear or Clearstream may also be subject to the procedures and requirements of such systems. The laws of some states require that certain Persons take physical delivery in definitive form of securities that they own. Consequently, the ability to transfer beneficial interests in a Global Note to such Persons will be limited to that extent. Because DTC can act only on behalf of the Participants, which in turn act on behalf of the Indirect Participants, the ability of a Person having beneficial interests in a Global Note to pledge such interests to Persons that do not participate in the DTC system, or otherwise take actions in respect of such interests, may be affected by the lack of a physical certificate evidencing such interests.]

Except as described below, owners of interests in the Global Notes will not have notes registered in their names, will not receive physical delivery of notes in certificated form and will not be considered the registered owners or holders thereof under the indenture for any purpose.

Payments in respect of the principal of, and interest and premium, if any, on a Global Note registered in the name of DTC or its nominee will be payable to DTC in its capacity as the registered holder under the indenture. Under the terms of the indenture, Issuer and the trustee will treat the Persons in whose names the notes, including the Global Notes, are registered as the owners of the notes for the purpose of receiving payments and for all other purposes.

Consequently, neither Issuer, the trustee nor any agent of Issuer or the trustee has or will have any responsibility or liability for:

(1) any aspect of DTC s records or any Participant s or Indirect Participant s records relating to or payments made on account of beneficial ownership interests in the Global Notes or for maintaining, supervising or reviewing any of DTC s records or any Participant s or Indirect Participant s records relating to the beneficial ownership interests in the Global Notes; or

(2) any other matter relating to the actions and practices of DTC or any of its Participants or Indirect Participants. [DTC has advised Issuer that its current practice, upon receipt of any payment in respect of securities such as the notes (including principal and interest), is to credit the accounts of the relevant Participants with the payment on the payment date unless DTC has reason to believe that it will not receive payment on such payment date. Each relevant Participant is credited with an amount proportionate to its beneficial ownership of an interest in the principal amount of the relevant security as shown on the records of DTC. Payments by the Participants and the Indirect Participants to the beneficial owners of notes will be governed by standing instructions and customary practices and will be the responsibility of the Participants or the Indirect Participants or Instructions from DTC or its nominee for all purposes.

Transfers between the Participants will be effected in accordance with DTC s procedures, and will be settled in same-day funds, and transfers between participants in Euroclear and Clearstream will be effected in accordance with their respective rules and operating procedures.

Subject to compliance with the transfer restrictions applicable to the notes described herein, cross-market transfers between the Participants, on the one hand, and Euroclear or Clearstream participants, on the other hand, will be effected through DTC in accordance with DTC s rules on behalf of Euroclear or Clearstream, as the case may be, by their respective depositaries; however, such cross-market transactions will require delivery of instructions to Euroclear or Clearstream, as the case may be, by the counterparty in such system in accordance with the rules and procedures and within the established deadlines (Brussels time) of such system. Euroclear or Clearstream, as the case may be, will, if the transaction meets its settlement requirements, deliver instructions to its respective depositary to take action to effect final settlement on its behalf by delivering or receiving interests in the relevant Global Note in DTC, and making or receiving payment in accordance with normal procedures for same-day funds settlement applicable to DTC. Euroclear participants and Clearstream participants may not deliver instructions directly to the depositories for Euroclear or Clearstream.

DTC has advised Issuer that it will take any action permitted to be taken by a holder of notes only at the direction of one or more Participants to whose account DTC has credited the interests in the Global Notes and only in respect of such portion of the aggregate principal amount of the notes as to which such Participants or Participants

- 31 -

has or have given such direction. However, if there is an Event of Default under the notes, DTC reserves the right to exchange the Global Notes in certificated form, and to distribute such notes to its Participants.

Although DTC, Euroclear and Clearstream have agreed to the foregoing procedures to facilitate transfers of interests in the Global Notes and the Global Notes among participants in DTC, Euroclear and Clearstream, they are under no obligation to perform or to continue to perform such procedures, and may discontinue such procedures at any time. None of Issuer, the trustee and any of their respective agents will have any responsibility for the performance by DTC, Euroclear or Clearstream or their respective participants or indirect participants of their respective obligations under the rules and procedures governing their operations.]

#### Exchange of global notes for certificated notes

A Global Note is exchangeable for Certificated Notes if:

- (1) DTC (a) notifies Issuer that it is unwilling or unable to continue as depositary for the Global Notes or (b) has ceased to be a clearing agency registered under the Exchange Act and, in either case, Issuer fails to appoint a successor depositary within 120 days after the date of such notice; or
- (2) Issuer, at its option, notifies the trustee in writing that it elects to cause the issuance of the Certificated Notes; or
- (3) there has occurred and is continuing a Default or Event of Default with respect to the notes and DTC has notified Issuer and the trustee of its desire to exchange the Global Notes for Certificated Notes.

In addition, beneficial interests in a Global Note may be exchanged for Certificated Notes upon 30 days prior written notice given to the trustee by or on behalf of DTC in accordance with the indenture. In all cases, Certificated Notes delivered in exchange for any Global Note or beneficial interests in Global Notes will be registered in the names, and issued in any approved denominations, requested by or on behalf of the depositary (in accordance with its customary procedures).

#### Exchange of certificated notes for global notes

Certificated Notes may not be exchanged for beneficial interests in any Global Note unless the transferor first delivers to the trustee a written certificate (in the form provided in the indenture) to the effect that such transfer will comply with the appropriate transfer restrictions applicable to such notes.

#### Same day settlement and payment

Issuer will make payments in respect of the notes represented by the Global Notes (including principal, premium, if any, and interest) by wire transfer of immediately available funds to the accounts specified by DTC or its nominee. Issuer will make all payments of principal, interest and premium, if any, with respect to Certificated Notes by wire transfer of immediately available funds to the accounts specified by the holders of the Certificated Notes or, if no such account is specified, by mailing a check to each such holder s registered address. [The notes represented by the Global Notes are expected to be eligible to trade in DTC s Same-Day Funds Settlement System, and any permitted secondary market trading activity in such notes will, therefore, be required by DTC to be settled in immediately available funds. Issuer expects that secondary trading in any Certificated Notes will also be settled in immediately available funds.]

[Because of time zone differences, the securities account of a Euroclear or Clearstream participant purchasing an interest in a Global Note from a Participant will be credited, and any such crediting will be reported to the relevant Euroclear or Clearstream participant, during the securities settlement processing day (which must be a business day for Euroclear and Clearstream) immediately following the settlement date of DTC. DTC has

advised Issuer that cash received in Euroclear or Clearstream as a result of sales of interests in a Global Note by or through a Euroclear or Clearstream participant to a Participant will be received with value on the settlement date of DTC but will be available in the relevant Euroclear or Clearstream cash account only as of the business day for Euroclear or Clearstream following DTC s settlement date.]

#### **Certain definitions**

Set forth below are certain defined terms used in the indenture. Reference is made to the indenture for a full disclosure of all defined terms used therein, as well as any other capitalized terms used herein for which no definition is provided.

Acquired Debt means, with respect to any specified Person:

(1) Indebtedness of any other Person existing at the time such other Person is merged with or into or became a Subsidiary of such specified Person, whether or not such Indebtedness is incurred in connection with, or in contemplation of, such other Person merging with or into, or becoming a Restricted Subsidiary of, such specified Person; and

(2) Indebtedness secured by a Lien encumbering any asset acquired by such specified Person. The term Acquired Debt does not include Indebtedness of a Person that is redeemed, defeased, retired or otherwise repaid at the time of, or immediately upon, consummation of the transactions by which such Person becomes a Restricted Subsidiary or acquires such asset, as the case may be.

*Affiliate* of any specified Person means any other Person directly or indirectly controlling or controlled by or under direct or indirect common control with such specified Person. For purposes of this definition, *control*, as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through the ownership of voting securities, by agreement or otherwise. For purposes of this definition, the terms *controlling*, *controlled by* and *under common control with* have correlative meanings.

Applicable Premium means, with respect to any note on any redemption date, the greater of:

- (1) 1.0% of the principal amount of the note; or
- (2) the excess of:
  - (a) the present value at such redemption date of (i) the redemption price of the note at [ ]<sup>18</sup> (such redemption price being set forth in the table appearing above under the caption Optional redemption ), plus (ii) all required interest payments due on the note through such date (excluding accrued but unpaid interest to the redemption date), computed using a discount rate equal to the Treasury Rate as of such redemption date plus 50 basis points; over

(b) the principal amount of the note, if greater. *Asset Acquisition* means:

(1) an Investment by Issuer (or any predecessor thereto) or any of its Restricted Subsidiaries in any other Person pursuant to which such Person shall become a Restricted Subsidiary or shall be merged into or consolidated with Issuer or any of its Restricted Subsidiaries but only if (x) such Person s primary business constitutes a Permitted Business and (y) the financial condition and results of operations of such Person are not already consolidated with those of Issuer and its Restricted Subsidiaries immediately prior to such Investment, or <sup>18</sup> NTD: Date to reflect first call date of applicable series of DT Notes.

- 33 -

(2) an acquisition by Issuer (or any predecessor thereto) or any of its Restricted Subsidiaries of the property and assets of any Person, other than Issuer or any of its Restricted Subsidiaries, that constitute all or substantially all of a division, operating unit or line of business of such Person but only (x) if the property and assets so acquired constitute a Permitted Business and (y) the financial condition and results of operations of such Person are not already consolidated with those of Issuer and its Restricted Subsidiaries immediately prior to such acquisition.

For the avoidance of doubt, the Merger shall be deemed to be an Asset Acquisition.

Asset Disposition means the sale or other disposition by Issuer or any of its Restricted Subsidiaries other than to Issuer or another Restricted Subsidiary of (1) all or substantially all of the Capital Stock owned by Issuer or any of its Restricted Subsidiaries of any Restricted Subsidiary or any Person that is a Permitted Joint Venture Investment or (2) all or substantially all of the assets that constitute a division, operating unit or line of business of Issuer or any of its Restricted Subsidiaries.

Asset Sale means:

- (1) the sale, lease, conveyance or other disposition of any assets or rights; *provided* that the sale, lease, conveyance or other disposition of all or substantially all of the assets of Issuer and its Restricted Subsidiaries taken as a whole will be governed by the provisions of the indenture described above under the caption Repurchase at the option of holders Change of control triggering event and/or the provisions described above under the caption Certain covenants Merger, consolidation or sale of assets and not by the provisions of the Asset Sale covenant; and
- (2) the issuance of Equity Interests in any of Issuer s Restricted Subsidiaries or the sale by Issuer or any Restricted Subsidiary thereof of Equity Interests in any of its Restricted Subsidiaries.

Notwithstanding the preceding, none of the following items will be deemed to be an Asset Sale:

- (1) any single transaction or series of related transactions that involves assets having a Fair Market Value of less than \$100.0 million;
- (2) a sale, lease, conveyance or other disposition of assets or Equity Interests between or among Issuer and/or its Restricted Subsidiaries;
- (3) an issuance or sale of Equity Interests by a Restricted Subsidiary of Issuer to Issuer or to a Restricted Subsidiary of Issuer;
- (4) the sale, lease, sub-lease, conveyance or other disposition of (a) assets, products, services or accounts receivable in the ordinary course of business, (b) equipment or other assets pursuant to a program for the maintenance or upgrading of such equipment or assets, or (c) any sale, conveyance or other disposition of damaged, worn-out, uneconomic or obsolete assets in the ordinary course of business;
- (5) the sale, conveyance or other disposition of cash or Cash Equivalents;
- (6) a surrender or waiver of contract rights or settlement, release or surrender of contract, tort or other claims in the ordinary course of business or a grant of a Lien not prohibited by the indenture;
- (7) a Restricted Payment that does not violate the covenant described above under the caption Certain covenants Restricted payments ;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

- (8) arms-length sales, leases or sub-leases (as lessor or sublessor), sale and leasebacks, assignments, conveyances, transfers or other dispositions of assets or rights to a Person that is a Permitted Joint Venture Investment;
- (9) licenses and sales of intellectual property or other general intangibles (other than FCC Licenses) in the ordinary course of business;

- (10) a Permitted Investment;
- (11) dispositions of assets to the ISIS Joint Venture;
- (12) one or more sales, conveyances, leases, subleases, licenses, contributions, or other dispositions, assignments or transfers made as part of, or in connection with, the Towers Transaction; or

#### (13) the settlement or early termination of any Permitted Bond Hedge Transaction.

Asset Sale Offer has the meaning assigned to that term in the provision described under the caption Repurchase at the option of holders Asset sales .

*Beneficial Owner* has the meaning assigned to such term in Rule 13d-3 and Rule 13d-5 under the Exchange Act, except that (a) in calculating the beneficial ownership of any particular person (as that term is used in Section 13(d)(3) of the Exchange Act), such person will be deemed to have beneficial ownership of all securities that such person has the right to acquire by conversion or exercise of other securities, whether such right is currently exercisable or is exercisable only after the passage of time and (b) in the case of a group pursuant to Rule 13d-5(b)(1) of the Exchange Act which group includes one or more Permitted Holders (or one or more Permitted Holders is deemed to share Beneficial Ownership with one or more other persons of any shares of Capital Stock), (i) such group shall be deemed not to have Beneficial Ownership of any shares held by such Permitted Holder and (ii) any person (other than such Permitted Holder) that is a member of such group (or sharing such Beneficial Ownership) shall be deemed not to have Beneficial Ownership of any shares beneficial Ownership). The terms Beneficially Owns and Beneficially Owned have a corresponding meaning.

Board of Directors means:

- (1) with respect to a corporation, the board of directors of the corporation or any committee thereof duly authorized to act on behalf of such board;
- (2) with respect to a partnership, the Board of Directors of the general partner of the partnership;
- (3) with respect to a limited liability company, the managing member or members or any controlling committee of managing members thereof; and
- (4) with respect to any other Person, the board or committee of such Person serving a similar function.

*Business Combination Agreement* means that certain Business Combination Agreement, dated as of October 3, 2012, as amended from time to time, by and among Deutsche Telekom AG, T-Mobile Global Zwischenholding GmbH, T-Mobile Global Holding GmbH, T-Mobile and MetroPCS Communications, Inc.

*Capital Lease Obligation* means, at the time any determination is to be made, the amount of the liability in respect of a capital lease that would at that time be required to be capitalized on a balance sheet prepared in accordance with GAAP, and the Stated Maturity thereof shall be the date of the last payment of rent or any other amount due under such lease prior to the first date upon which such lease may be prepaid by the lessee without payment of a penalty.

Capital Stock means:

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

- (1) in the case of a corporation, corporate stock;
- (2) in the case of an association or business entity, any and all shares, interests, participations, rights or other equivalents (however designated) of corporate stock;
- (3) in the case of a partnership or limited liability company, partnership interests (whether general or limited) or membership interests, respectively; and

- 35 -

- (4) any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, the issuing Person, but excluding from all of the foregoing any debt securities convertible into Capital Stock, whether or not such debt securities include any right of participation with Capital Stock. Cash Equivalents means:
- (1) United States dollars;
- (2) securities issued or directly and fully guaranteed or insured by the United States government or any agency or instrumentality of the United States government (provided that the full faith and credit of the United States is pledged in support of those securities) having maturities of not more than two years from the date of acquisition;
- (3) demand deposits, certificates of deposit and eurodollar time deposits with maturities of six months or less from the date of acquisition, bankers acceptances with maturities not exceeding one year and overnight bank deposits, in each case, with any domestic commercial bank having capital and surplus in excess of \$500.0 million and a Thomson Bank Watch Rating of B or better;
- (4) repurchase obligations with a term of not more than seven days for underlying securities of the types described in clauses (2) and (3) above entered into with any financial institution meeting the qualifications specified in clause (3) above;
- (5) commercial paper having one of the two highest ratings obtainable from a Rating Agency at the date of acquisition and, in each case, maturing within one year after the date of acquisition;
- (6) securities issued and fully guaranteed by any state, commonwealth or territory of the United States, or by any political subdivision or agency or instrumentality thereof, rated at least A by a Rating Agency at the date of acquisition and having maturities of not more than two years after the date of acquisition;
- (7) auction rate securities rated at least AA- or Aa3 by a Rating Agency at the time of purchase and with reset dates of one year or less from the time of purchase;
- (8) investments, classified in accordance with GAAP as current assets of Issuer or any of its Restricted Subsidiaries, in money market funds, mutual funds or investment programs registered under the Investment Company Act of 1940, at least 90% of the portfolios of which constitute investments of the character, quality and maturity described in clauses (1) through (7) of this definition;
- (9) in the case of any Person that is operating outside the United States or anticipates operating outside the United States within the next 12 months, any substantially similar investment to the kinds described in clauses (1) through (7) of this definition rated at least P-2 by Moody s or A-2 by S&P or the equivalent thereof; and

 <sup>(10)</sup> deposits or payments made to the FCC in connection with the auction or licensing of Governmental Authorizations that are fully refundable.
 *Change of Control* means the occurrence of any of the following:

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

- (1) the direct or indirect sale, lease, transfer, conveyance or other disposition (other than by way of merger or consolidation), in one or a series of related transactions, of all or substantially all of the properties or assets of Issuer and its Restricted Subsidiaries taken as a whole to any person (as that term is used in Section 13(d) of the Exchange Act) other than any such disposition to a Restricted Subsidiary or a Permitted Holder;
- (2) the adoption of a plan relating to the liquidation or dissolution of Issuer;

- 36 -

- (3) the consummation of any transaction (including any merger or consolidation), the result of which is that any person (as defined above), other than a Permitted Holder, becomes the Beneficial Owner, directly or indirectly, of more than 50% of the Voting Stock of Parent (or its successor by merger, consolidation or purchase of all or substantially all of its assets or its equity), measured by voting power rather than number of shares;
- (4) during any period of 12 consecutive months, a majority of the members of the Board of Directors or other equivalent governing body of Issuer or Parent cease to be composed of individuals (i) who were members of that Board of Directors or equivalent governing body on the first day of such period, (ii) whose election or nomination to that Board of Directors or equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that Board of Directors or equivalent governing body, (iii) whose election or nomination to that Board of Directors or other equivalent governing body was approved by individuals referred to in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of that Board of Directors or equivalent governing body, or (iv) in the case of Issuer, whose election or nomination to that Board of Directors or equivalent governing body was approved by Parent; or
- (5) the Issuer ceases to be a direct or indirect Wholly-Owned Subsidiary of Parent. Change of Control Offer has the meaning assigned to that term in the indenture governing the notes.

*Change of Control Triggering Event* means the occurrence of a Change of Control that is accompanied or followed by a downgrade by one or more gradations (including gradations within ratings categories as well as between ratings categories), or withdrawal of the rating of the notes within the Ratings Decline Period by at least two of the Rating Agencies, as a result of which the rating of the notes on any day during such Ratings Decline Period is below the rating by each such Rating Agency in effect immediately preceding the first public announcement of the Change of Control (or occurrence thereof if such Change of Control occurs prior to public announcement), *provided* that in making the relevant decision(s) referred to above to downgrade or withdraw such ratings, as applicable, the relevant Rating Agency announces publicly or confirms in writing during such Ratings Decline Period that such decision(s) resulted, in whole or in part, from the occurrence (or expected occurrence) of such Change of Control or the announcement of the intention to effect such Change of Control; provided, further that no Change of Control Triggering Event shall be deemed to occur if at the time of the applicable downgrade the rating of the notes by at least two out of the three Rating Agencies is Investment Grade.

Closing Date means the date on which the Merger occurs.

*Consolidated Cash Flow* means, with respect to any specified Person for any period, the Consolidated Net Income of such Person for such period plus, without duplication:

- (1) provision for taxes based on income or profits of such Person and its Restricted Subsidiaries for such period, to the extent that such provision for taxes was deducted in computing such Consolidated Net Income; *plus*
- (2) the Consolidated Interest Expense of such Person and its Restricted Subsidiaries for such period, to the extent that such Consolidated Interest Expense was deducted in computing such Consolidated Net Income; *plus*
- (3) depreciation, amortization (including non-cash impairment charges and any write-off or write-down or amortization of intangibles but excluding amortization of ordinary course prepaid cash expenses that were paid in a prior period) and other non-cash expenses or charges (excluding any such non-cash expense to the extent that it represents an ordinary course accrual of or reserve for cash expenses in any future period or amortization of any ordinary course prepaid cash expense that was paid in a prior period) of such Person and its Restricted Subsidiaries for such period to the extent that such depreciation, amortization and other non-cash expenses or charges were deducted in computing such Consolidated Net Income; *plus*

- (4) any net after-tax extraordinary, nonrecurring or unusual gains or losses or income, expenses or charges (including all fees and expenses relating thereto), including (a) any fees, expenses and costs relating to the Towers Transaction, (b) any fees, expenses or charges related to any sale or offering of Equity Interests of such Person or Parent, any acquisition or disposition or any Indebtedness, in each case that is permitted to be incurred hereunder (in each case, whether or not successful), or the offering, amendment or modification of any debt instrument, including the offering, any amendment or other modification of the notes, *provided* that Consolidated Cash Flow shall not be deemed to be increased by more than \$250.0 million in any twelve-month period pursuant to this clause (b), (c) any premium, penalty or fee paid in relation to any repayment, prepayment or repurchase of Indebtedness, (d) any fees or expenses relating to the Transactions and the issuance and sale of the Permitted MetroPCS Notes , as defined in the Business Combination Agreement, and (e) restructuring charges, integration costs (including retention, relocation and contract termination costs) and related costs and charges, *provided* such costs and charges under this clause (e) shall not exceed \$300.0 million in any twelve-month period, plus, for the first four years after the Closing date, up to an additional \$300 million in any twelve-month period related to the Transactions); *plus*
- (5) New Market losses, up to a maximum aggregate amount of \$300 million in any twelve-month period; minus
- (6) non-cash items increasing such Consolidated Net Income for such period, other than the accrual of revenue in the ordinary course of business, in each case, on a consolidated basis and determined in accordance with GAAP.

Notwithstanding the preceding, the provision for taxes based on the income or profits of, and the depreciation and amortization and other non-cash expenses of, a Restricted Subsidiary of Issuer that is not a Subsidiary Guarantor will be added to Consolidated Net Income to compute Consolidated Cash Flow of Issuer only to the extent that a corresponding amount would be permitted at the date of determination to be dividended to Issuer by such Restricted Subsidiary without prior governmental approval (that has not been obtained), and without direct or indirect restriction pursuant to the terms of its charter and all agreements, instruments, judgments, decrees, orders, statutes, rules and governmental regulations applicable to that Restricted Subsidiary or its stockholders. For the avoidance of doubt, with respect to any period prior to the Merger, the Issuer shall be deemed to be MetroPCS Wireless, Inc., and calculations Consolidated Cash Flow of the Issuer for any period prior to the Closing Date for purposes of calculating the Debt to Cash Flow Ratio shall be on a pro forma basis as described in the last paragraph of the definition of *Debt to Cash Flow Ratio*.

*Consolidated Indebtedness* means, with respect to any Person as of any date of determination, the sum, without duplication, of (i) the total amount of Indebtedness of such Person and its Restricted Subsidiaries, plus (ii) the total amount of Indebtedness of any other Person, to the extent that such Indebtedness has been Guaranteed by the referent Person or one or more of its Restricted Subsidiaries, plus (iii) the aggregate liquidation value of all Disqualified Stock of such Person and all Preferred Stock of Subsidiaries of such Person, in each case, determined on a consolidated basis in accordance with GAAP.

Consolidated Interest Expense means, with respect to any Person for any period, the sum of, without duplication:

- (1) the consolidated interest expense of such Person and its Subsidiaries for such period, whether paid or accrued (including amortization of debt issuance costs or original issue discount, non-cash interest payments, the interest component of any deferred payment obligations, the interest component of all payments associated with Capital Lease Obligations, commissions, discounts and other fees and charges incurred in respect of letter of credit or bankers acceptance financings, and net of payments (if any) pursuant to Hedging Obligations); *plus*
- (2) the consolidated interest expense of such Person and its Restricted Subsidiaries that was capitalized during such period; plus

- 38 -

- (3) any interest expense on that portion of Indebtedness of another Person that is guaranteed by such Person or one of its Restricted Subsidiaries or secured by a Lien on assets of such Person or one of its Restricted Subsidiaries (whether or not such Guarantee or Lien is called upon); *plus*
- (4) the product of (a) all dividend payments on any series of Preferred Stock of such Person or any of its Restricted Subsidiaries; times (b) a fraction, the numerator of which is one and the denominator of which is one minus the then current combined federal, state and local statutory tax rate of such Person, expressed as a decimal;

in each case, on a consolidated basis and in accordance with GAAP; excluding, however, any amount of such interest of any Restricted Subsidiary of the referent Person if the net income of such Restricted Subsidiary is excluded in the calculation of Consolidated Net Income pursuant to clause (2) of the definition thereof (but only in the same proportion as the net income of such Restricted Subsidiary is excluded from the calculation of Consolidated Net Income pursuant to clause (2) of the definition thereof). Notwithstanding the foregoing, if any lease or other liability is reclassified as indebtedness or as a Capital Lease Obligation due to a change in accounting principles or the application thereof after the Closing Date, the interest component of all payments associated with such lease or other liability shall be excluded from Consolidated Interest Expense.

*Consolidated Net Income* means, with respect to any specified Person for any period, the aggregate of the Net Income of such Person and its Restricted Subsidiaries for such period, on a consolidated basis, determined in accordance with GAAP; *provided* that:

- the positive Net Income of any Person that is not a Restricted Subsidiary or that is accounted for by the equity method of accounting will be included only to the extent of the amount of dividends or similar distributions paid in cash to the specified Person or a Restricted Subsidiary of the Person;
- (2) solely for the purpose of determining the amount available for Restricted Payments under clause 3(A) of the second paragraph of the covenant described above under the caption Certain covenants Restricted Payments the Net Income of any Restricted Subsidiary that is not a Guarantor will be excluded to the extent that the declaration or payment of dividends or similar distributions by that Restricted Subsidiary of that Net Income is not at the date of determination permitted without any prior governmental approval (that has not been obtained) or, directly or indirectly, by operation of the terms of its charter or any agreement, instrument, judgment, decree, order, statute, rule or governmental regulation applicable to that Restricted Subsidiary or its stockholders;
- (3) the effect of a change in accounting principles or in the application thereof (including any change to IFRS and any cumulative effect adjustment) will be excluded;
- (4) unrealized losses and gains attributable to Hedging Obligations, including those resulting from the application of the Financial Accounting Standards Board (FASB) Accounting Standards Codification (ASC) 815, will be excluded; and
- (5) any non-cash compensation charge or expense realized from grants of stock, stock appreciation or similar rights, stock option or other rights to officers, directors and employees, will be excluded.

*Contribution Indebtedness* means, Indebtedness in an aggregate principal amount at any one time outstanding, including all Permitted Refinancing Indebtedness incurred to renew, refund, refinance, replace, defease or discharge such Indebtedness, not to exceed 150% of the aggregate amount of all Net Equity Proceeds.

*Convertible Debt* means Debt of the Issuer (which may be Guaranteed by the Guarantors) permitted to be incurred hereunder that is either (a) convertible or exchangeable into common stock of Parent (and cash in lieu of fractional shares) and/or cash (in an amount determined by reference to the price of such common stock) or (b) sold as units with call options, warrants or rights to purchase (or substantially equivalent derivative transactions) that are exercisable for common stock of Parent and/or cash (in an amount determined by reference to the price of such common stock).

# - 39 -

*Credit Facilities* means, one or more debt facilities, capital leases, purchase money financings or commercial paper facilities, providing for revolving credit loans, term loans, receivables financing (including through the sale of receivables to such lenders or to special purpose entities formed to borrow from such lenders against such receivables), capital leases, purchase money debt, debt securities or letters of credit, in each case, as amended, restated, modified, renewed, refunded, replaced (whether upon or after termination or otherwise) or refinanced (including, in each case, by means of sales of debt securities to institutional investors) in whole or in part from time to time.

*Debt to Cash Flow Ratio* means, with respect to any Person as of any date of determination, the ratio of (a) the Consolidated Indebtedness of such Person as of such date to (b) the Consolidated Cash Flow of such Person for the four most recent full fiscal quarters ending immediately prior to such date for which internal financial statements are available.

For purposes of making the computation referred to above:

- (1) pro forma effect shall be given to Asset Dispositions and Asset Acquisitions (including the Merger and including giving pro forma effect to any related financing transactions and the application of proceeds of any Asset Disposition) that occur during such four-quarter period or subsequent to such four quarter period but on or prior to the date on which the Debt to Cash Flow Ratio is to be calculated as if they had occurred and such proceeds had been applied on the first day of such four-quarter period;
- (2) pro forma effect shall be given to asset dispositions and, asset acquisitions (including giving pro forma effect to any related financing transactions and the application of proceeds of any asset disposition) that have been made by any Person that has become a Restricted Subsidiary of Issuer or has been merged with or into Issuer (including MetroPCS Wireless, Inc.) or any Restricted Subsidiary during such four-quarter period or subsequent to such four quarter period but on or prior to the date on which the Debt to Cash Flow Ratio is to be calculated and that would have constituted Asset Dispositions or Asset Acquisitions had such transactions occurred when such Person was a Restricted Subsidiary, as if such asset dispositions or asset acquisitions were Asset Dispositions or Asset Acquisitions that occurred on the first day of such four-quarter period;
- (3) to the extent that the pro forma effect of any transaction is to be made pursuant to clause (1) or (2) above, such pro forma effect shall be determined in good faith on a reasonable basis by a responsible financial or accounting officer of the specified Person, as if the subject transaction(s) had occurred on the first day of the four-quarter reference period and Consolidated Cash Flow for such reference period shall be calculated without giving effect to clause (3) of the proviso set forth in the definition of Consolidated Net Income;
- (4) the Consolidated Cash Flow attributable to discontinued operations, as determined in accordance with GAAP, and operations or businesses disposed of (without duplication of clauses (1) and (2) above) prior to the date on which the Debt to Cash Flow Ratio is to be calculated, shall be excluded;
- (5) any Person that is a Restricted Subsidiary on the date on which the Debt to Cash Flow Ratio is to be calculated will be deemed to have been a Restricted Subsidiary at all times during such four-quarter period; and
- (6) any Person that is not a Restricted Subsidiary on the date on which the Debt to Cash Flow Ratio is to be calculated will be deemed not to have been a Restricted Subsidiary at any time during such four-quarter period.

For the avoidance of doubt, for any period commencing prior to the date that is four fiscal quarters after the fiscal quarter during which the Closing Date occurs, the Debt to Consolidated Cash Flow Ratio shall be calculated giving pro forma effect to the Transactions as if the Transactions had occurred on the first day of the four quarter reference period.

Default means any event that is, or with the passage of time or the giving of notice or both would be, an Event of Default.

Designated Entity means (i) Iowa Wireless Services LLC, a Delaware limited liability company, or (ii) any Designated Tower Entity.

*Designated Tower Entity* means any entity established solely or primarily for the limited purpose of holding wireless communications sites, towers, and related contracts, equipment, improvements, real estate, and other assets, and performing other activities incidental thereto or in connection with the Towers Transaction. For the avoidance of doubt, T-Mobile USA Tower LLC and T-Mobile West Tower LLC are each Designated Tower Entities.

*Disqualified Stock* means any Capital Stock that, by its terms (or by the terms of any security into which it is convertible, or for which it is exchangeable, in each case, at the option of the holder of the Capital Stock), or upon the happening of any event, matures or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise, or redeemable at the option of the holder of the Capital Stock, in whole or in part, on or prior to the date that is 91 days after the date on which the notes mature; *provided* that any class of Capital Stock of such Person that, by its terms, requires such Person to satisfy in full its obligations with respect to the payment of dividends or upon maturity, redemption (pursuant to a sinking fund or otherwise) or repurchase thereof or otherwise by the delivery of Capital Stock, so long as such Person satisfies its obligations with respect thereto solely by the delivery of Capital Stock. Notwithstanding the preceding sentence, any Capital Stock that would constitute Disqualified Stock solely because the holders of the Capital Stock if the terms of such Capital Stock provide that Issuer may not repurchase or redeem any such Capital Stock pursuant to such provisions unless such repurchase or redeemed to be outstanding at any time for purposes of the indenture will be the maximum amount that Issuer and its Restricted Subsidiaries may become obligated to pay upon the maturity of, or pursuant to any mandatory redemption provisions of, such Disqualified Stock, exclusive of accrued dividends.

Domestic Restricted Subsidiary means any Restricted Subsidiary that is not a Foreign Subsidiary.

DT means Deutsche Telekom AG, an Aktiengesellschaft organized and existing under the laws of the Federal Republic of Germany.

*Equity Interests* means Capital Stock and all warrants, options or other rights to acquire Capital Stock (but excluding any debt security that is convertible into, or exchangeable for, Capital Stock).

*Existing Indebtedness* means Indebtedness of Issuer and its Subsidiaries (other than Indebtedness under the notes) in existence on the Closing Date, until such amounts are repaid. For the avoidance of doubt, Existing Indebtedness includes the DT Notes, and the Permitted MetroPCS Notes and the TMUS Working Capital Facility as such terms are defined in the Business Combination Agreement and in each case actually issued.

*Fair Market Value* means the value that would be paid by a willing buyer to an unaffiliated willing seller in a transaction not involving distress or necessity of either party, determined in good faith by Issuer s Board of Directors or a senior officer of the Issuer, which determination shall be conclusive.

*FCC* means the United States Federal Communications Commission and any successor agency that is responsible for regulating the United States telecommunications industry.

- 41 -

FCC Licenses means all licenses or permits now or hereafter issued by the FCC.

Fitch means Fitch Inc., a Subsidiary of Fimalac, S.A.

*Foreign Subsidiary* means any Subsidiary of Issuer other than a Subsidiary organized under the laws of the United States or any state of the United States or the District of Columbia, or any direct or indirect subsidiary thereof.

*GAAP* means generally accepted accounting principles as in effect on the Closing Date. At any time, Issuer may elect to apply IFRS accounting principles in lieu of GAAP and, upon any such election, references herein to GAAP (or parts of the Accounting Standards Codification or ASC )) shall thereafter be construed to mean IFRS (except as otherwise provided in the indenture); *provided* that any such election, once made, shall be irrevocable; *provided further*, that any calculation or determination in the indenture that requires the application of GAAP for periods that include fiscal quarters ended prior to Issuer s election to apply IFRS shall remain as previously calculated or determined in accordance with GAAP. Issuer shall give notice of any such election made in accordance with this definition to the trustee and the holders of notes.

*Governmental Authorization* means any permit, license, authorization, plan, directive, consent, permission, consent order or consent decree of or from any governmental authority.

*Guarantee* means a guarantee other than by endorsement of negotiable instruments for collection in the ordinary course of business, direct or indirect, in any manner including by way of a pledge of assets or through letters of credit or reimbursement agreements in respect thereof, of all or any part of any Indebtedness (whether arising by virtue of partnership arrangements, or by agreements to keep-well, to purchase assets, goods, securities or services, to take or pay or to maintain financial statement conditions or otherwise).

Guarantors means each of:

- (1) Parent;
- (2) Issuer s direct and indirect Domestic Restricted Subsidiaries (other than Immaterial Subsidiaries) that are Wholly-Owned Subsidiaries on the Closing Date;
- (3) Issuer s direct and indirect Restricted Subsidiaries that guarantee any Specified Issuer Indebtedness;
- (4) Any future Subsidiary of Parent that directly or indirectly owns equity interests of Issuer; and
- (5) any other Subsidiary of Parent that executes a Note Guarantee in accordance with the provisions of the indenture either (a) as required pursuant to the covenant described above under the caption Certain covenants Additional note guarantees or (b) because Parent or Issuer, in its sole discretion, causes such Subsidiary to do so,

and their respective successors and assigns, in each case, until the Note Guarantee of such Person has been released in accordance with the provisions of the indenture.

Hedging Obligations means, with respect to any specified Person, the obligations of such Person under:

(1) interest rate swap agreements (whether from fixed to floating or from floating to fixed), interest rate cap agreements and interest rate collar agreements;

# Edgar Filing: MILLER PETROLEUM INC - Form 10-K

- (2) other agreements or arrangements designed to manage interest rates or interest rate risk; and
- (3) other agreements or arrangements designed to protect such Person against fluctuations in currency exchange rates or commodity prices,

- 42 -

and any guarantee in respect thereof.

*IFRS* means the international accounting standards promulgated by the International Accounting Standards Board and its predecessors, as adopted by the European Union, as in effect from time to time.

*Immaterial Subsidiary* means any Subsidiary of the Issuer that at any time has less than \$100.0 million in Total Assets; *provided*, that the aggregate Total Assets of all Immaterial Subsidiaries shall not at any time exceed \$300.0 million.

Indebtedness means, with respect to any specified Person, without duplication,

- (a) any indebtedness of such Person (excluding accrued expenses and trade payables), whether or not contingent:
- (1) in respect of borrowed money;
- (2) evidenced by bonds, notes, debentures or similar instruments or letters of credit (or reimbursement agreements in respect thereof);
- (3) in respect of banker s acceptances;
- (4) representing Capital Lease Obligations;
- (5) representing the balance deferred and unpaid of the purchase price of any property or services due more than six months after such property is acquired or such services are completed; or
- (6) representing any Hedging Obligations; and

(b) any financial liabilities recorded in respect of the upfront proceeds received in connection with the Towers Transaction, in each case, if and only to the extent any of the preceding items (other than letters of credit and Hedging Obligations) would appear as a liability upon a balance sheet of the specified Person prepared in accordance with GAAP. In addition, the term Indebtedness includes all Indebtedness of others secured by a Lien on any asset of the specified Person (whether or not such Indebtedness is assumed by the specified Person) and, to the extent not otherwise included, the Guarantee by the specified Person of any Indebtedness of any other Person. Notwithstanding the foregoing, in no event shall the reclassification of any lease or other liability as indebtedness for any purpose under the indenture. The amount of any Indebtedness shall be determined in accordance with the last paragraph of the covenant described above under the caption Certain covenants Incurrence of indebtedness and issuance of preferred stock.

*Investments* means, with respect to any Person, all direct or indirect investments by such Person in other Persons (including Affiliates) in the forms of loans (including Guarantees), advances (excluding commission, travel, entertainment, drawing accounts and similar advances to directors, officers and employees made in the ordinary course of business and excluding the purchase of assets, equipment, property or accounts receivables created or acquired in the ordinary course of business) or capital contributions, and purchases or other acquisitions for consideration of Indebtedness, Equity Interests or other securities. If Issuer or any Restricted Subsidiary of Issuer sells or otherwise disposes of any Capital Stock of any direct or indirect Restricted Subsidiary of Issuer such that, after giving effect to any such sale or disposition, such Person is no longer a Restricted Subsidiary of Issuer, Issuer will be deemed to have made an Investment on the date of any such sale or disposition equal to the Fair Market Value of Issuer s Investments in such Restricted Subsidiary that were not sold or disposed of in an amount determined as provided in the final paragraph of the covenant described above under the caption Certain covenants Restricted payments. The acquisition by Issuer or any Subsidiary of

# Table of Contents

- 43 -

Issuer of a Person that holds an Investment in a third Person will be deemed to be an Investment by Issuer or such Subsidiary in such third Person in an amount equal to the Fair Market Value of the Investments held by the acquired Person in such third Person in an amount determined as provided in the final paragraph of the covenant described above under the caption Certain covenants Restricted payments as of the date the acquisition of the acquired Person is consummated. Except as otherwise provided in the indenture, the amount of an Investment will be determined at the time the Investment is made and without giving effect to subsequent changes in value.

#### Investment Grade means

- (1) with respect to Moody s (or any successor company acquiring all or substantially all of its assets), a rating of Baa3 (or its equivalent under any successor rating category of Moody s) or better;
- (2) with respect to S&P (or any successor company acquiring all or substantially all of its assets), a rating of BBB- (or its equivalent under any successor rating category of S&P) or better;
- (3) with respect to Fitch (or any successor company acquiring all or substantially all of its assets), a rating of BBB- (or its equivalent under any successor rating category of Fitch) or better; and
- (4) if any Rating Agency ceases to exist or ceases to rate the notes for reasons outside of the control of Issuer, the equivalent investment grade credit rating from any other nationally recognized statistical rating organization within the meaning of Rule 15c3-1(c)(2)(vi)(F) under the Exchange Act selected by Issuer as a replacement agency.

ISIS Joint Venture means Amended and Restated LLC Agreement of JVL Ventures, LLC dated October 1, 2010, as amended.

*Lien* means, with respect to any asset, any mortgage, lien, pledge, charge, security interest or encumbrance of any kind in respect of such asset, whether or not filed, recorded or otherwise perfected under applicable law, including any conditional sale or other title retention agreement and any lease in the nature thereof.

*Merger* means the merger of MetroPCS Wireless, Inc. with and into the Issuer with Issuer as the surviving Person, pursuant to the Business Combination Agreement.

Moody s means Moody s Investors Service, Inc.

*Net Equity Proceeds* means the net cash proceeds received by Issuer since the Closing Date as a contribution to its common equity capital or from the issue or sale of Equity Interests of Issuer (other than Disqualified Stock).

*Net Income* means, with respect to any specified Person, the net income (loss) of such Person, determined in accordance with GAAP and before any reduction in respect of Preferred Stock accretion or dividends, excluding however:

- any gain (or loss), together with any related provision for taxes on such gain (or loss) realized in connection with : (a) dispositions of assets (other than in the ordinary course of business); or (b) the extinguishment of any Indebtedness of such Person or any of its Restricted Subsidiaries; and
- (2) any extraordinary gain (or loss), together with any related provision for taxes on such extraordinary gain (or loss).

*Net Proceeds* means the aggregate cash proceeds received by Issuer or any of its Restricted Subsidiaries in respect of any Asset Sale (including any cash received upon the sale or other disposition of any non-cash consideration received in any Asset Sale, but excluding any items deemed to be cash pursuant to clause (2)(a) of the covenant described above under the caption Repurchase at the option of holders Asset sales ), net of all

- 44 -

costs relating to such Asset Sale, including (a) legal, accounting and investment banking fees, finder s fees, sales commissions, employee severance costs, and any relocation expenses incurred as a result of the Asset Sale, (b) taxes paid or payable as a result of the Asset Sale, in each case, after taking into account any available tax credits or deductions and any tax sharing arrangements, (c) amounts required to be applied to the repayment of Indebtedness, other than Indebtedness under a Credit Facility, secured by a Lien on the asset or assets that were the subject of such Asset Sale, (d) all distributions and other payments required to be made to minority interest holders in Restricted Subsidiaries as a result of such Asset Sale and (e) any amounts to be set aside in any reserve established in accordance with GAAP or any amount placed in escrow, in either case for adjustment in respect of the sale price of such properties or assets or for liabilities associated with such Asset Sale and retained by Issuer or any of its Restricted Subsidiaries until such time as such reserve is reversed or such escrow arrangement is terminated, in which case Net Proceeds shall include only the amount of the reserve so reversed or the amount returned to Issuer or its Restricted Subsidiaries from such escrow arrangement, as the case may be.

*New Market Losses* means, for any period, to the extent such losses were deducted in computing such Consolidated Net Income during the applicable period, an amount equal to any extraordinary loss plus any net loss (without duplication) realized by the Issuer or any of its Restricted Subsidiaries incurred in connection with construction, launch and operations in any New Market for such period, so long as such net losses are incurred on or prior to the fourth anniversary after the initial commencement of commercial operations in the applicable New Market.

*New Markets* means the collective reference to any wireless telephone markets other than the metropolitan areas of Las Vegas, Nevada; Los Angeles, San Francisco and Sacramento California; Detroit, Michigan; Dallas/Fort Worth, Texas; Tampa/Sarasota, Orlando, Miami and Jacksonville, Florida; Atlanta, Georgia; Philadelphia, Pennsylvania; New York, New York; Boston, Massachusetts; and Hartford, Connecticut.

Non-Recourse Debt means Indebtedness:

- as to which neither Issuer nor any of its Restricted Subsidiaries (a) provides credit support of any kind (including any undertaking, agreement or instrument that would constitute Indebtedness), subject to customary bad-boy exceptions, (b) is directly or indirectly liable as a guarantor or otherwise, or (c) constitutes the lender;
- (2) no default with respect to which (including any rights that the holders of the Indebtedness may have to take enforcement action against an Unrestricted Subsidiary) would permit upon notice, lapse of time or both any holder of any other Indebtedness of Issuer or any of its Restricted Subsidiaries to declare a default on such other Indebtedness or cause the payment of the Indebtedness to be accelerated or payable prior to its Stated Maturity; and
- (3) as to which the lenders have been notified in writing that they will not have any recourse to the stock or assets of Issuer or any of its Restricted Securities.

*Note Guarantee* means the Guarantee by each Guarantor of Issuer s obligations under the indenture and the notes, executed pursuant to the provisions of the indenture.

*Obligations* means any principal, interest, penalties, fees, indemnifications, reimbursements, cash collateral obligations, damages and other liabilities payable under the documentation governing any Indebtedness.

*Permitted Bond Hedge Transaction* means any call or capped call option (or substantively equivalent derivative transaction) on the Parent s common stock purchased by the Issuer in connection with the issuance of any Convertible Debt; *provided* that the purchase price for such Permitted Bond Hedge Transaction, does not exceed the net cash proceeds received by the Issuer from the sale of such Convertible Debt issued in connection with the Permitted Bond Hedge Transaction.

- 45 -

*Permitted Business* means those businesses in which Issuer and its Subsidiaries were engaged on the Closing Date, or any business similar, related, incidental or ancillary thereto or that constitutes a reasonable extension or expansion thereof, or any business reasonably related to the telecommunications industry, and the acquisition, holding or exploitation of any license relating to the delivery of those services.

Permitted Holder means (i) DT and (ii) any direct or indirect Subsidiary of DT.

Permitted Investments means:

- (1) any Investment in Issuer or in any Restricted Subsidiary of Issuer;
- (2) any Investment in Cash Equivalents;
- (3) any Investment by Issuer or any Restricted Subsidiary of Issuer in a Person, if as a result of such Investment:
  - (a) such Person becomes a Restricted Subsidiary of Issuer; or
  - (b) such Person is merged, consolidated or amalgamated with or into, or transfers or conveys substantially all of its assets to, or is liquidated into, Issuer or a Restricted Subsidiary of Issuer;
- (4) any Investment made as a result of the receipt of non-cash consideration from an Asset Sale that was made pursuant to and in compliance with the covenant described above under the caption Repurchase at the option of holders Asset sales ;
- (5) any acquisition of assets or Capital Stock solely in exchange for the issuance of Equity Interests (other than Disqualified Stock) of Issuer or Equity Interests of Parent;
- (6) any Investments received in compromise or resolution of (A) obligations of trade creditors or customers that were incurred in the ordinary course of business of Issuer or any of its Restricted Subsidiaries, including pursuant to any plan of reorganization or similar arrangement upon the bankruptcy or insolvency of any trade creditor or customer; or (B) litigation, arbitration or other disputes with Persons who are not Affiliates;
- (7) Investments represented by Hedging Obligations;
- (8) loans or advances to employees made in the ordinary course of business of Issuer or any Restricted Subsidiary of Issuer in an aggregate principal amount not to exceed \$50.0 million at any one time outstanding;
- (9) any payment on or with respect to, or purchase, redemption, defeasement or other acquisition or retirement for value of (i) the notes, or (ii) any Indebtedness that is *pari passu* with the notes;

(10)

advances and prepayments for asset purchases in the ordinary course of business in a Permitted Business of Issuer or any of its Restricted Subsidiaries;

- (11) Investments existing on the Closing Date, including Investments held by MetroPCS Wireless, Inc., Issuer and their Subsidiaries immediately prior to the Merger;
- (12) Investments in the ISIS Joint Venture having an aggregate Fair Market Value (measured on the date each such Investment was made and without giving effect to subsequent changes in value), when taken together with all other Investments made pursuant to this clause (12) since the Closing Date that are at that time outstanding, not to exceed \$300.0 million;
- (13) Permitted Bond Hedge Transactions which constitute Investments;
- (14) (a) Permitted Joint Venture Investments, and (b) other Investments in any Person other than an Affiliate of Issuer (excluding any Person that is an Affiliate of Issuer solely by reason of Parent s ownership, directly or indirectly, of Equity Interests, or control, of such Person or which becomes an Affiliate as a result of such

- 46 -

Investment), to the extent such Investment under (a) or (b) has an aggregate Fair Market Value (measured on the date each such Investment was made and without giving effect to subsequent changes in value), when taken together with all other Investments made pursuant to this clause (14) that are at the time outstanding, not to exceed 12.5% of the Issuer s Total Assets on the date of such Investment;

- (15) Investments in a Person primarily engaged in a Permitted Business having an aggregate Fair Market Value (measured on the date each such Investment was made and without giving effect to subsequent changes in value), when taken together with all other Investments made pursuant to this clause (15) since the Closing Date that are at that time outstanding, not to exceed \$250.0 million;
- (16) guarantees permitted under Certain Covenants Incurrence of Indebtedness and Issuance of Preferred Stock ; and
- (17) deposits or payments made with the FCC in connection with the auction or licensing of Governmental Authorizations.
- (18) any Investment deemed made from time to time pursuant to Certain Covenants Designation of restricted and unrestricted subsidiaries in connection with a Specified Unrestricted Subsidiary Designation, in an amount equal to the aggregate Fair Market Value of all outstanding Investments owned by the Issuer and its Restricted Subsidiaries in the Subsidiaries designated as Unrestricted Subsidiaries pursuant to such Specified Unrestricted Subsidiary Designation, but only to the extent not in excess of the aggregate Fair Market Value of all outstanding Investments owned by T-Mobile and its Restricted Subsidiaries in such designated Subsidiaries as of the Closing Date (for this purpose, it shall be assumed, as regards to Investments in any Designated Tower Entity, that all wireless communications sites, towers, and related contracts, equipment, improvements, real estate, and other assets of T-Mobile and its subsidiaries subject to the Towers Transaction that are contemplated to be transferred to the Designated Tower Entities in accordance with the terms of the Towers Transaction, as contemplated in the Tower Transaction Agreements as in effect as of the Closing Date, had been transferred to the Designated Tower Entities, whether or not all such transfers have in fact then taken place, but disregarding any transfers of assets not part of the Towers Transaction Agreements as in effect as of the Closing Date); and
- (19) any other Investments made in connection with the Towers Transaction, as contemplated in the Tower Transaction Agreements as in effect as of the Closing Date.

Notwithstanding any other provision to the contrary, no Permitted Investment shall be deemed to be a Restricted Payment.

*Permitted Joint Venture Investment* means, with respect to any specified Person, Investments in any other Person engaged in a Permitted Business of which at least 40% of the outstanding Capital Stock of such other Person is at the time owned directly or indirectly by the specified Person.

Permitted Liens means:

- (1) Liens securing Indebtedness and other Obligations under Credit Facilities and/or securing Hedging Obligations related thereto permitted by clauses (1), (8) and (20) of the second paragraph of the covenant entitled Certain covenants Incurrence of indebtedness and issuance of preferred stock, *provided* that any secured Permitted Refinancing Indebtedness incurred in respect of Indebtedness or other Obligations previously secured pursuant to this clause (1) will be treated as Indebtedness secured pursuant to this clause (1) in making any determination as to whether additional Indebtedness or other Obligations may be secured pursuant to this clause (1);
- (2) Liens in favor of Issuer or the Guarantors;

- (3) Liens on property of a Person existing at the time such Person becomes a Restricted Subsidiary or is merged with or into or consolidated with Issuer or any Subsidiary of Issuer; *provided* that such Liens were in existence prior to the contemplation of such merger or consolidation and do not extend to any assets (other than improvements thereon, accessions thereto and proceeds thereof) other than those of the Person that becomes a Restricted Subsidiary or is merged into or consolidated with Issuer or the Subsidiary;
- (4) Liens on property (including Capital Stock) existing at the time of acquisition of the property by Issuer or any Subsidiary of Issuer; *provided* that such Liens were in existence prior to, and not incurred in contemplation of, such acquisition;
- (5) (a) bankers Liens, rights of setoff or similar rights and remedies as to deposit accounts or other funds maintained with a depositary institution, and (b) Liens, deposits (including deposits with the FCC) or pledges to secure the performance of bids, tenders, trade or governmental contracts, leases, licenses, statutory obligations, surety or appeal bonds, performance bonds or other obligations of a like nature incurred in the ordinary course of business;
- (6) Liens to secure Indebtedness (including Capital Lease Obligations) permitted by clause (4) of the second paragraph of the covenant entitled Certain covenants Incurrence of indebtedness and issuance of preferred stock covering only the assets (including the proceeds thereof, accessions thereto and upgrades thereof) acquired with or financed by such Indebtedness;
- (7) Liens existing on the Closing Date (including Liens existing on the assets of MetroPCS Wireless, Inc. and its Subsidiaries);
- (8) Liens for taxes, assessments or governmental charges or claims that are not yet delinquent or that are being contested in good faith by appropriate proceedings; *provided* that any reserve or other appropriate provision as is required in conformity with GAAP has been made therefor;
- (9) Liens imposed by law or contract, such as carriers, warehousemen s, suppliers, vendors, construction, repairmen s, landlord s and mechanics Liens or other similar Liens, in each case, incurred in the ordinary course of business;
- (10) survey exceptions, encumbrances, easements or reservations of, or rights of others for, licenses, rights-of-way, sewers, electric lines, telegraph and telephone lines and other similar purposes, or zoning or other restrictions as to the use of real property that were not incurred in connection with Indebtedness and that do not in the aggregate materially adversely affect the value of said properties or materially impair their use in the operation of the business of such Person;
- (11) Liens arising by reason of a judgment, attachment, decree or court order, to the extent not otherwise resulting in an Event of Default, and any Liens that are required to protect or enforce any rights in any administrative, arbitration or other court proceedings in the ordinary course of business;
- (12) Liens created for the benefit of (or to secure) the notes (or the Note Guarantees);
- (13) Liens to secure any Permitted Refinancing Indebtedness permitted to be incurred under the indenture; provided, however, that:
  - (a) the new Lien shall be limited to all or part of the same property and assets that secured or, under the written agreements pursuant to which the original Lien arose, could secure the original Lien (plus improvements and accessions to such property and assets and proceeds or distributions of such property and assets and improvements and accessions thereto); and

(b) the Indebtedness secured by the new Lien is not increased to any amount greater than the sum of (x) the outstanding principal amount, or, if greater, committed amount, of the Indebtedness being renewed, refunded, refinanced, replaced, defeased or discharged and (y) an amount necessary to pay any fees and expenses, including premiums, related to such renewal, refunding, refinancing, replacement, defeasance or discharge;

- 48 -

- (14) (a) Liens contained in purchase and sale agreements or lease agreements limiting the transfer of assets pending the closing of the transactions contemplated thereby or the termination of the lease, respectively, (b) spectrum leases or other similar lease or licensing arrangements contained in, or entered into in connection with, purchase and sale agreements, and (c) Liens relating to deposits or escrows established in connection with purchase and sale agreements;
- (15) Liens that may be deemed to exist by virtue of contractual provisions that restrict the ability of Issuer or any of its Subsidiaries from granting or permitting to exist Liens on their respective assets;
- (16) Liens in favor of the trustee as provided for in the indenture on money or property held or collected by the trustee in its capacity as trustee;
- (17) Liens on cash or Cash Equivalents securing (a) workers compensation claims, self-insurance obligations, unemployment insurance or other social security, old age pension, bankers acceptances, performance bonds, completion bonds, bid bonds, appeal bonds, indemnity bonds, specific performance or injunctive relief bonds, surety bonds, public liability obligations, or other similar bonds or obligations, or securing any Guarantees or letters of credit functioning as or supporting any of the foregoing, in each case incurred in the ordinary course of business or (b) letters of credit required to be issued for the benefit of any Person that controls a Permitted Joint Venture Investment to secure any put right for the benefit of the Person controlling the Permitted Joint Venture Investment;
- (18) Liens arising from Uniform Commercial Code financing statement filings regarding operating leases entered into in the ordinary course of business covering only the property under lease (plus improvements and accessions to such property and proceeds or distributions of such property and improvements and accessions thereto);
- (19) any interest or title of a lessor, licensor or sublicensor in the property subject to any lease, license or sublicense entered into in the ordinary course of business;
- (20) Liens on cash or Cash Equivalents on deposit to secure reimbursement obligations under letters of credit incurred in the ordinary course of business;
- (21) Liens on and pledges of the Equity Interests of any Unrestricted Subsidiary or any Person that is a Permitted Joint Venture Investment owned by Issuer or any Restricted Subsidiary to the extent securing Non-Recourse Debt or other Indebtedness of such Unrestricted Subsidiary or Person;
- (22) Liens arising under operating agreements, joint venture agreements, partnership agreements, contracts for sale and other agreements arising in the ordinary course of business that are customary in the Permitted Business, and applicable only to the assets that are the subject of such agreements or contracts;
- (23) Liens securing Hedging Obligations;
- (24) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods in the ordinary course of business;
- (25) Liens encumbering reasonable customary initial deposits and margin deposits and similar Liens attaching to commodity trading accounts or other brokerage accounts incurred in the ordinary course of business and not for speculative purposes;

- (26) Liens upon specific items of inventory or other goods and proceeds of any Person securing such Person s obligations in respect of bankers acceptances issued or created for the account of such Person to facilitate the purchase, shipment or storage of such inventory or other goods;
- (27) Liens securing any arrangement for treasury, depositary or cash management services provided to Issuer or any of its Restricted Subsidiaries in the ordinary course of business;
- (28) Liens with respect to obligations that do not exceed at any time the greater of (x) \$500 million and (y) 1.0% of Issuer s Total Assets at such time;

- 49 -

- (29) Liens encumbering deposits made to secure obligations arising from statutory, regulatory, contractual or warranty requirements; and
- (30) Liens, if any, incurred in connection with the Towers Transaction. *Permitted Payments to Parent* means, without duplication as to amounts:
- (1) payments to Parent to permit Parent to pay reasonable accounting, legal, investment banking fees and administrative expenses of Parent when due; and
- (2) for so long as Issuer is a member of a group filing a consolidated or combined tax return with Parent, payments to Parent in respect of an allocable portion of the tax liabilities of such group that is attributable to Issuer and its Subsidiaries (*Tax Payments*). The Tax Payments shall not exceed the lesser of (i) the amount of the relevant tax (including any penalties and interest) that Issuer would owe if Issuer were filing a separate tax return (or a separate consolidated or combined return with its Subsidiaries that are members of the consolidated or combined group), taking into account any carryovers and carrybacks of tax attributes (such as net operating losses) of Issuer and such Subsidiaries from other taxable years and (ii) the net amount of the relevant tax that Parent actually owes to the appropriate taxing authority.

*Permitted Refinancing Indebtedness* means any Indebtedness of Issuer or any of its Restricted Subsidiaries, any Disqualified Stock of Issuer or any Preferred Stock of any Restricted Subsidiary issued (a) in exchange for, or the net proceeds of which are used to, extend the maturity, renew, refund, refinance, replace, defease, discharge or otherwise retire for value, in whole or in part, or (b) constituting an amendment, modification or supplement to or a deferral or renewal of ((a) and (b) above, collectively, a *Refinancing*), any other Indebtedness of Issuer or any of its Restricted Subsidiaries (other than intercompany Indebtedness), any Disqualified Stock of Issuer or any Preferred Stock of a Restricted Subsidiary in a principal amount or, in the case of Disqualified Stock of Issuer or Preferred Stock of a Restricted Subsidiary, liquidation preference, not to exceed (after deduction of reasonable and customary fees and expenses incurred in connection with the Refinancing) the lesser of:

- (1) the principal amount or, in the case of Disqualified Stock or Preferred Stock, liquidation preference, of the Indebtedness, Disqualified Stock or Preferred Stock so Refinanced (plus, in the case of Indebtedness, the amount of accrued interest and premium, if any paid in connection therewith), and
- (2) if the Indebtedness being Refinanced was issued with any original issue discount, the accreted value of such Indebtedness (as determined in accordance with GAAP) at the time of such Refinancing;

in each case, except to the extent that any such excess principal amount (or accreted value, as applicable) would be then permitted to be incurred by other provisions of the covenant described above under the caption Certain Covenants Incurrence of Indebtedness and Issuance of Preferred Stock; provided, that such excess principal amount of Indebtedness shall be deemed to be incurred under such other provision.

Notwithstanding the preceding, no Indebtedness, Disqualified Stock or Preferred Stock will be deemed to be Permitted Refinancing Indebtedness, unless:

- such Indebtedness, Disqualified Stock or Preferred Stock has a final maturity date or redemption date, as applicable, later than the final maturity date or redemption date, as applicable, of, and has a Weighted Average Life to Maturity equal to or greater than the Weighted Average Life to Maturity of, the Indebtedness, Disqualified Stock or Preferred Stock being Refinanced;
- (2) if the Indebtedness, Disqualified Stock or Preferred Stock being Refinanced is contractually subordinated in right of payment to the notes, such Indebtedness, Disqualified Stock or Preferred Stock is contractually subordinated in right of payment to, the notes, on terms at least as favorable to the holders of notes as those contained in the documentation governing the Indebtedness, Disqualified Stock or Preferred

Stock being Refinanced at the time of the Refinancing; and

- 50 -

(3) such Indebtedness or Disqualified Stock is incurred or issued by Issuer or such Indebtedness, Disqualified Stock or Preferred Stock is incurred or issued by the Restricted Subsidiary who is the obligor on the Indebtedness being Refinanced or the issuer of the Disqualified Stock or Preferred Stock being Refinanced, or a Restricted Subsidiary of such obligor or issuer.

*Person* means any individual, corporation, partnership, joint venture, association, joint-stock company, trust, unincorporated organization, limited liability company or government or other entity.

*Preferred Stock* means, with respect to any Person, any Capital Stock of such Person that has preferential rights to any other Capital Stock of such Person with respect to dividends or payments upon liquidation.

*Rating Agency* means each of Moody s, S&P, Fitch and, if any of Moody s, S&P or Fitch ceases to exist or ceases to rate the notes for reasons outside of the control of Issuer, any other nationally recognized statistical rating organization within the meaning of Rule 15c3-1(c)(2)(vi)(F) under the Exchange Act selected by Issuer as a replacement agency.

*Ratings Decline Period* means the period that (i) begins on the earlier of (a) the date of the first public announcement of the occurrence of a Change of Control or of the intention by Issuer or a shareholder of Issuer, as applicable, to effect a Change of Control or (b) the occurrence thereof and (ii) ends 90 days following consummation of such Change of Control; *provided* that such period shall be extended for so long as the rating of the notes, as noted by the applicable Rating Agency, is under publicly announced consideration for downgrade by the applicable Rating Agency.

*Replacement Assets* means: (i) capital expenditures with respect to any assets, (ii) other assets that will be used or useful in a Permitted Business, (iii) all or substantially all of the assets of a Permitted Business, (iv) Voting Stock of any Person engaged in a Permitted Business that, when taken together with all other Voting Stock of such Person owned by Issuer and its Restricted Subsidiaries, constitutes a majority of the Voting Stock of such Person and such Person will become a Restricted Subsidiary on the date of the acquisition thereof or (v) deposits or payments to acquire FCC Licenses.

*Reset Rate* means [ <sup>19</sup>].

Restricted Investment means an Investment other than a Permitted Investment.

Restricted Subsidiary of a Person means any Subsidiary of the referenced Person that is not an Unrestricted Subsidiary.

S&P means Standard & Poor s Ratings Group.

*Significant Subsidiary* means any Restricted Subsidiary that as of the end of the most recent fiscal quarter for which financial statements are available, would be a significant subsidiary as defined in Article 1, Rule 1-02 of Regulation S-X, promulgated pursuant to the Securities Act, as such Regulation is in effect on the Closing Date.

Specified Issuer Indebtedness means any Indebtedness of Issuer in a principal amount of \$250 million or more.

*Specified Unrestricted Subsidiary Designation* has the meaning assigned to such term in the provision described under the heading Certain Covenants Designation of restricted and unrestricted subsidiaries .

Stated Maturity means, with respect to any installment of interest or principal on any series of Indebtedness, the date on which the payment of interest or principal was scheduled to be paid in the documentation governing

<sup>19</sup> NTD: Reset Rate to be determined as set forth in the DT Note Pricing Schedule.

- 51 -

such Indebtedness as of the Closing Date, and will not include any contingent obligations to repay, redeem or repurchase any such interest or principal prior to the date originally scheduled for the payment thereof.

Subordinated Indebtedness means :

- (1) with respect to the Issuer, any Indebtedness of such Issuer which is by its terms subordinated in right of payment to the notes; and
- (2) with respect to any Guarantor, any Indebtedness of such Guarantor which is by its terms subordinated in right of payment to the such Guarantor s Guarantee of the notes.

Subsidiary means, with respect to any specified Person:

- (3) any corporation, association or other business entity of which more than 50% of the total voting power of shares of Capital Stock entitled (without regard to the occurrence of any contingency and after giving effect to any voting agreement or stockholders agreement that effectively transfers voting power) to vote in the election of directors, managers or trustees of the corporation, association or other business entity is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person (or a combination thereof); and
- (4) any partnership (a) the sole general partner or the managing general partner of which is such Person or a Subsidiary of such Person or (b) the only general partners of which are that Person or one or more Subsidiaries of that Person (or any combination thereof).
   Subsidiary Guarantors means, collectively, the Guarantors that are Subsidiaries of Issuer.

*Total Assets* means the consolidated total assets of a Person and its Subsidiaries as set forth on the most recent balance sheet of such Person prepared in accordance with GAAP.

Towers Transaction means the transactions contemplated by the Towers Transaction Agreements.

*Towers Transaction Agreements* means: (i) the Master Agreement, dated as of September 28, 2012 (as the same may be amended, modified, or supplemented from time to time), among T-Mobile, Crown Castle International Corp., a Delaware corporation, and certain Subsidiaries of T-Mobile; and (ii) each of the other transaction documents entered into in connection therewith or contemplated thereby, as they may be amended, modified or supplemented from time to time.

*Transactions* means (i) the Merger, (ii) the offering of the notes, all other DT Notes , the Permitted MetroPCS Notes and the incurrence of the TMUS Working Capital Facility, as defined in the Business Combination Agreement, (iii) the refinancing of Existing Indebtedness on or prior to the Closing Date, (iv) the Cash Payment and the MetroPCS Reverse Stock Split, each as defined in the Business Combination Agreement, and (v) all other transactions consummated in connection therewith.

<sup>20</sup> NTD: To match first call date for the applicable series of Notes.

- 52 -

*Unrestricted Subsidiary* means any Subsidiary of Issuer that is designated by the Board of Directors of Issuer as an Unrestricted Subsidiary pursuant to a resolution of the Board of Directors, but only to the extent that:

- (1) except as permitted by the covenant described above under the caption Certain covenants Transactions with affiliates, such Subsidiary is not party to any agreement, contract, arrangement or understanding with Issuer or any Restricted Subsidiary of Issuer unless the terms of any such agreement, contract, arrangement or understanding are, taken as a whole, no less favorable to Issuer or such Restricted Subsidiary than those that might be obtained at the time from Persons who are not Affiliates of Issuer;
- (2) such Subsidiary does not hold any Liens on any property of Parent, Issuer or any of its Restricted Subsidiaries; and

(3) such Subsidiary has not guaranteed or otherwise directly or indirectly provided credit support for any Indebtedness of Issuer or any of its Restricted Subsidiaries, except to the extent that such guarantee or credit support would be released upon such designation. *Voting Stock* of any specified Person as of any date means the Capital Stock of such Person that is at the time entitled to vote in the election of the Board of Directors of such Person.

Weighted Average Life to Maturity means, when applied to any Indebtedness at any date, the number of years obtained by dividing:

- (1) the sum of the products obtained by multiplying (a) the amount of each then remaining installment, sinking fund, serial maturity or other required payments of principal, including payment at final maturity, in respect of the Indebtedness, by (b) the number of years (calculated to the nearest one-twelfth) that will elapse between such date and the making of such payment; by
- (2) the then outstanding principal amount of such Indebtedness.

*Wholly-Owned Subsidiary* of any specified Person means a Subsidiary of such Person, all of the outstanding Capital Stock or other ownership interests of which (other than directors qualifying shares) will at the time be owned by such Person or by one or more Wholly-Owned Subsidiaries of such Person. Except if expressly otherwise specified, Wholly-Owned Subsidiary means a Wholly-Owned Subsidiary of the Issuer.

- 53 -

## <u>Exhibit H</u>

# REVOLVING CREDIT FACILITY

Summary of Terms and Conditions

I.	PARTIES			
	Borrower:	T-Mobile USA, Inc., a Delaware corporation (the <i>Borrower</i> ).		
	Guarantors:	All obligations of the Borrower under the Facility (as defined below) will be unconditionally guaranteed by MetroPCS Communications, Inc. ( <i>Parent</i> ) and all of (a) Borrower's existing and future (i) Domestic Restricted Subsidiaries (as defined in the Description of DT Notes (the <i>DT Notes DON</i> ) attached as Exhibit G to the Business Combination Agreement) that are Wholly-Owned Subsidiaries (as defined in the DT Notes DON) except for Immaterial Subsidiaries (as defined in the DT Notes DON) and (ii) Restricted Subsidiaries that guarantee any Specified Issuer Indebtedness (as defined in the DT Notes DON) and (b) any entity that becomes a parent entity of the Borrower.		
	Initial Lender:	Deutsche Telekom AG, an Aktiengesellschaft organized and existing under the laws of the Federal Republic of Germany, or one or more of its affiliates (the <i>Lender</i> ).		
II.	TYPE AND AMOUNT OF FACILITY			
	Type and Amount:	Senior unsecured revolving credit facility (the <i>Facility</i> , and the loans thereunder, the <i>Loans</i> ) in the amount of \$500 million (the <i>Commitment</i> ).		
	Availability:	The Facility shall be available on a revolving basis for borrowings during the period from the Closing Date to but excluding the fifth anniversary thereof (the <i>Commitment Termination Date</i> ).		
	Maturity:	All amounts outstanding under the Facility will be due and payable on the Commitment Termination Date.		
	Purpose:	The proceeds of the Loans shall be used for working capital and other general corporate purposes of the Borrower and its subsidiaries.		
III.	CERTAIN PAYMENT PROVISIO	<u>DNS</u>		
	Upfront Fee:	50 bps on the Commitment, payable within one business day after the Closing Date.		
	Undrawn Commitment Fee:	An amount per annum on the undrawn portion of the Commitment, payable quarterly in arrears determined in accordance with the pricing grid below, based on the then-applicable Debt to Cash Flow Ratio (as defined in the DT Notes DON).		
	Interest Rate:	Each Loan under the Facility shall bear interest at a rate per annum equal to the Eurodollar Rate or the Base Rate (in each case as defined in the Existing Credit Agreement), at the Borrower s option, in each case <u>plus</u> the Applicable Margin.		

As used herein, *Applicable Margin* means a rate per annum determined in accordance with the following pricing grid (expressed as basis points per annum), based on the then-applicable Debt to Cash Flow Ratio (as defined in the DT Notes DON), provided that the Applicable Margin for Base Rate Loans shall be 100 basis points less than the applicable Adjusted LIBOR Margin set forth below: Pricing Grid

	Debt to Cash Flow	Undrawn	Adjusted LIBOR
Category	Ratio	Commitment Fee	Margin
1	£ 1.50	25.0	250
2	> 1.50 and £ 2.50	37.5	275
3	>2.50	50.0	300

Optional Prepayments and Loans under the Facility may be prepaid without premium or penalty by the Borrower (but subject to **Commitment Reductions:** the payment of breakage costs, if any, for prepayments of any Eurodollar Loan on a day other than the last day of the interest period therefor) and commitments may be reduced at the option of the Borrower, in each case in minimum amounts of \$10 million. IV. CERTAIN CONDITIONS Initial Conditions: The availability of the Facility shall be conditioned upon satisfaction of the following conditions precedent (the date on which such conditions shall be satisfied and the Facility shall become effective being herein referred to as the *Closing Date* ): (a) The Borrower shall have executed and delivered definitive financing documentation for the Facility reasonably satisfactory to the Lender, which shall be substantially similar to the existing Third Amended and Restated Credit Agreement dated as of March 17, 2011, among MetroPCS Wireless, Inc., JP Morgan Chase Bank, N.A., as administrative agent, and the lenders from time to time party thereto (as amended, the Existing Credit Agreement ), except as modified hereby or otherwise to reflect the changes contemplated herein, including with adjustments to give effect to the unsecured nature of the facility and the deletion of the term facility, the swingline commitment, and the letter of credit commitment (the Credit Documentation ). (b) (i) The material accuracy of all representations and warranties in the Credit Documentation and (ii) the absence of any default or event of default as of the Closing Date, and a senior officer of the Borrower shall have delivered to the Lender a certificate to that effect. (c) The Lender shall have received such customary legal opinions and closing certificates, together with customary certified documentation covering valid corporate existence, good standing, due authorization and incumbency, as it may reasonably request. (d) The Lender shall have received all fees and expenses required to be paid by the Borrower in connection with the Facility on or before the Closing Date. (e) The Transaction (as defined in the Business Combination Agreement) shall have been consummated in accordance with the terms of the Business Combination Agreement and substantially simultaneously with the Closing Date.

2

On-Going Conditions:	The making of each extension of credit shall be conditioned upon (a) the accuracy in all material respects of all representations and warranties in the Credit Documentation, (b) there being no default or event of default in existence at the time of, or after giving effect to the making of, such extension of credit and (c) compliance with the financial maintenance covenant as of the last day of the most recently ended fiscal quarter for which financial statements have been delivered, giving pro forma effect to the relevant extension of credit.
V. CERTAIN DOCUMENTATION MA	ATTERS
	The Credit Documentation shall contain representations, warranties, covenants, events of default substantially similar to those in the Existing Credit Agreement, except as set forth herein or as the parties may agree.
Collateral:	None.
Financial Covenants:	
Debt to Cash Flow:	The Borrower shall not permit the Debt to Cash Flow Ratio to exceed 4.0 to 1.0 as of the last day of any fiscal quarter. Covenant to apply only when revolving loans are outstanding as of the relevant quarter-end date and (as provided above under On-Going Conditions ) as a condition to any draw.
Representations and Warranties:	
	Substantially the same as the Existing Credit Agreement (with the \$50 million materiality threshold applicable to certain representations to be raised to \$100 million).
Affirmative Covenants:	Substantially the same as the Existing Credit Agreement (with the \$50 million materiality threshold applicable to certain affirmative covenants to be raised to \$100 million).
Negative Covenants:	Substantially the same as the Existing Credit Agreement, with changes to reflect the baskets and exceptions specified in the DT Notes DON.
Events of Default:	Substantially the same as the Existing Credit Agreement (with the \$50 million materiality threshold applicable to certain events of default to be raised to \$100 million and the cross-default provision to be revised to provide for cross-payment-default and cross-acceleration only).
Change of Control:	Substantially the same as the definition of Change of Control under the DT Notes DON.
Voting:	Substantially the same as the Existing Credit Agreement, <i>provided</i> that there shall be no restrictions on voting by affiliates of the Borrower.
Assignments and Participations:	
	Substantially the same as the Existing Credit Agreement, <i>provided</i> that there shall be no restrictions on loans or commitments being held by affiliates of the Borrower. Credit Documentation to provide that in the event Lender elects to assign any Loans or Commitments to any other lenders, the Lender and Borrower will cooperate on amending the documentation to permit such assignments.
Yield Protection:	Substantially the same as the Existing Credit Agreement.
Expenses and Indemnification:	
	Substantially the same as the Existing Credit Agreement.
Governing Law and Forum:	
	State of New York.

## <u>Exhibit I</u>

#### Financing Backstop Terms and Conditions

Capitalized terms not otherwise defined herein have the meanings set forth in the Business Combination Agreement to which this Exhibit is attached (the BCA or Business Combination Agreement ).

#### 1. Credit Facility Backstop

- a. Purpose: To backstop the payoff of the MetroPCS Existing Credit Agreement.
- b. Commitment amount: Up to \$2.5 billion
- c. Commitment fee: 1.5% of the commitment, payable within one business day after the Closing Date.
- d. Commitment termination: The Credit Facility Backstop will terminate on the earliest of (i) the date that the MetroPCS Existing Credit Agreement is paid in full, which date shall be no later than date the TMUS Merger is consummated (the **Merger**, and such date the **Closing Date**), (ii) such commitment being reduced to zero pursuant to the commitment reduction terms below and (iii) the date the Business Combination Agreement is terminated.
- e. Fee reductions:
  - i. 100 bps on the amount of commitment reduced within 4.5 months after pro forma financials are available; and

ii. 50 bps on the amount of commitment reduced between 4.5 and 7.5 months after pro forma financials are available; *provided*, that for purposes of clauses i. and ii. above, pro forma financials shall be deemed to be available after the Proxy Statement is filed with the SEC; and, *provided further*, that the Credit Facility Backstop commitment shall only be reduced and the commitment fee shall only be rebated to the extent MetroPCS prepays amounts outstanding under the MetroPCS Existing Credit Agreement.

- f. Terms:
  - i. TMUS to issue to DT and DT to purchase from TMUS (in accordance with Section 4.13(c) of the BCA), on the Closing Date, an amount of Additional DT Notes equal to the amount required to pay off and terminate the MetroPCS Existing Credit Agreement.
  - ii. Any Additional DT Notes issued to fund the Credit Facility Backstop will have principal amounts, pricing, tenor and prepayment information for such series of notes to be issued as described in Exhibit F to the BCA (the **DT Note Pricing Schedule**), determined in accordance with Section 5.
- 2. Change of Control Backstop

- a. Purpose: To backstop an amendment to the change of control provisions in the MetroPCS Existing Notes to provide that there will be no change of control put right in connection with the Transaction (the **Proposed Waiver** ).
- b. Commitment amount: Up to \$2.0 billion
- c. Commitment termination: Commitment terminated immediately upon the earliest of (i) consummation of the Proposed Waiver,
   (ii) the expiration of the time that any holder of MetroPCS Existing Notes would have the right to put its notes in connection with the MetroPCS Merger (the Change of

1

**Control Put Right Exercise Date**), (iii) such commitment being reduced to zero pursuant to the commitment reduction terms below, and (iv) the date the Business Combination Agreement is terminated.

- d. Commitment fee: 1.50% of the commitment, payable within one business day after the Closing Date.
- e. Fee reductions:
  - i. 100 bps on the amount of commitment reduced within 3 months after the earlier of (a) receipt of ratings by either Moody s or S&P and (b) 30 days after the signing date of the BCA (the **Rebate Start Date** ).
  - ii. 50 bps on the amount of commitment reduced between 3 and 6 months after the Rebate Start Date.
- f. Terms:
  - i. TMUS to issue to DT and DT to purchase from TMUS (in accordance with Section 4.13(c) of the BCA), an amount of Additional DT Notes equal to the amount required to satisfy any put obligations in the event that the Proposed Waiver is not obtained and any bondholders exercise any put right upon consummation of the Transaction, on the date such put obligation is required to be satisfied.
  - ii. Any Additional DT Notes issued to fund the Change of Control Backstop will have principal amounts, pricing, tenor and prepayment information for such series of notes to be issued as described in the DT Note Pricing Schedule, determined in accordance with Section 5.
- 3. New Note Backstop
  - a. Purpose: To backstop the issuance by MetroPCS OpCo or MetroPCS HoldCo of up to \$1.0 billion in Permitted MetroPCS Notes to be used as provided in the definition thereof (the **New Note Funding Amount** ).
  - b. Commitment Amount: \$1 billion
  - c. Commitment fee: 1.50% of the commitment amount, payable within one business day after the Closing Date.
  - d. Commitment termination: Commitment terminated immediately upon the earliest of (i) such commitment being reduced to zero pursuant to the commitment reduction terms below, (ii) funding of the New Note Funding Amount on the Closing Date, and (iii) and (iii) the date the Business Combination Agreement is terminated.
  - e. Fee reductions:
    - i. 100 bps on amount of commitment reduced prior to 9 months after the signing date of the BCA.

- ii. 50 bps on amount of commitment reduced between 9 and 12 months after signing date of the BCA.
- f. Terms:
  - i. TMUS to issue to DT and DT to purchase from TMUS (in accordance with Section 4.13(c) of the BCA), on the Closing Date, an amount of Additional DT Notes equal to the amount of the commitment, if any, remaining on the Closing Date.
  - ii. Any Additional DT Notes issued will have principal amounts, pricing, tenor and prepayment information for such series of notes to be issued as described in the DT Note Pricing Schedule, determined in accordance with Section 5.

2

- 4. Commitment Reduction
  - a. Outstanding commitments under the Credit Facility Backstop, Change of Control Backstop and New Note Backstop shall be reduced on a dollar for dollar basis by the cash proceeds received by Parent or any subsidiary of Parent (including the Issuer) from the issuance of debt securities after the date the Business Combination Agreement is entered into until such time as all of the outstanding commitments under such backstop facilities are terminated in full (the **Take-Out Proceeds**).

b. The Take-Out Proceeds shall reduce the outstanding commitments under the backstops in reverse order of maturity of the Additional DT Notes that would be issued to fund such backstop facilities; provided, that if the Take-Out Proceeds are received prior to the Change of Control Put Right Exercise Date, then the outstanding commitments under the backstops shall be reduced without regard to the notes that would be issued to fund the Change of Control Backstop Facility.

5. Allocation of commitments to Additional DT Notes on and after the Closing Date

- a. The amount and tenor of the Additional DT Notes received by DT in exchange for satisfying its commitments under the backstop facilities on and after the Closing Date shall be determined as follows.
  - i. On the Closing Date, DT shall receive Additional DT Notes of the tenors set forth on page 2 of the DT Notes Pricing Schedule, in order of maturity from earliest to latest, allocated equally among Permanent Notes and Reset Notes, up to the total amount of outstanding commitments on the Closing Date (after giving effect to the issuance and use of proceeds of any Permitted MetroPCS Notes issued on or prior to the Closing Date), excluding the amount of the Change of Control Backstop commitment that will not be funded on the Closing Date.
  - ii. Additionally, to the extent that the Change of Control Backstop commitment remains outstanding on the Closing Date because a Proposed Waiver has not been obtained as to any amount of MetroPCS Existing Notes and the Change of Control put obligation is not to be funded on the Closing Date, the maturity corresponding to such MetroPCS Existing Notes (e.g., 2018 for the existing MetroPCS OpCo  $7^{7}/_{8}\%$  notes due 2018) shall be skipped in like amount and allocated as if they were funded in determining the amount of Additional DT Notes of each tenor to be issued to DT on the Closing Date according to Section 5.a.i. above.
  - iii. To the extent any remaining Change of Control Backstop commitment is funded by DT after the Closing Date, DT shall receive Additional DT Notes of the tenors set forth on page 2 of the DT Notes Pricing Schedule (with tenors determined from the Issue Date), in order of maturity from earliest to latest, allocated equally among Permanent Notes and Reset Notes, up to the total amount of the Change of Control Backstop so funded, but skipping the amounts and tenors of Additional DT Notes previously issued pursuant to Sections 5.a.i. and 5.a.ii.

3

## <u>Exhibit J</u>

#### Provisions to be in Noteholder Agreement between DT and the Issuer

#### Definitions

- DT means Deutsche Telekom AG, an Aktiengesellschaft organized and existing under the laws of the Federal Republic of Germany.
- DT Notes has the meaning specified in the Business Combination Agreement.
- DT Entities means DT or any of its Subsidiaries (other than MetroPCS or any of its Subsidiaries).
- DT Redemption Event means the occurrence of any of the following:
  - (1) a Change of Control;
  - (2) so long as DT owns, directly or indirectly, more than 50% of the Voting Stock of the Issuer immediately prior to such transaction, the consummation of any transaction (including, without limitation, any merger or consolidation), the result of which is that DT ceases to own, directly or indirectly, more than 50% of the voting stock of Issuer (or its successor by merger, consolidation or purchase of all or substantially all of its assets or its equity), measured by voting power rather than number of shares; or
  - (3) the consummation of any transaction (including, without limitation, any merger or consolidation), the result of which is that the owners directly or indirectly of the voting stock of the Issuer immediately prior to such consummation cease to be the owners, directly or indirectly, of more than 50% of the voting stock of Issuer (or its successor by merger, consolidation or purchase of all or substantially all of its assets or its equity), measured by voting power rather than number of shares.

Notwithstanding the foregoing, no Specified Change of Control shall constitute a DT Redemption Event.

*MetroPCS* means MetroPCS Communications, Inc., a Delaware corporation.

*Specified Change of Control* means any direct or indirect transfer of any common stock of MetroPCS by any DT Entity to a third party the result of which (to the knowledge of DT following reasonable inquiry) is the occurrence of a DT Redemption Event, other than in connection with a transaction in which all holders of common stock of MetroPCS are required to or are entitled to participate on the same terms.

TMUS means T-Mobile USA, Inc., a Delaware corporation.

Terms used but not otherwise defined have the meaning set forth in the indenture governing the DT Notes, which terms shall be included in the Noteholder Agreement.

#### DT Redemption Event

If a DT Redemption Event occurs, any DT Entity which holds any DT Notes will have the right to require Issuer to repurchase all or any part of such DT Notes for cash equal to 101% of the aggregate principal amount of DT Notes repurchased plus accrued and unpaid interest on the DT Notes repurchased to, but not including, the date of purchase (the *DT Redemption Event Payment*).

#### Specified Change of Control

Notwithstanding anything to the contrary in the indenture, in case of any Change of Control Triggering Event based upon a Change of Control that is a Specified Change of Control, no DT Entity will tender any DT Notes into the related Change of Control Offer.

If a DT Entity transfers any DT Notes to any third party after the earlier of announcement or consummation of a transaction that constitutes a Specified Change of Control, and before the Change of Control Offer that the Issuer is required to make in connection therewith expires, the transferee of such DT Notes shall agree that it will not tender any such DT Notes to the Issuer in any such Change of Control offer.

#### Merger, consolidation or sale of assets

The Issuer will cause any person with which it consolidates or merges, or directly or indirectly transfers all or substantially all of its assets to expressly assume the obligations under the Noteholder Agreement.

#### Amendments

For so long as DT holds, directly or indirectly, a majority in principal amount of any series of the DT Notes, the Issuer shall not agree to any waiver or amendment of any provision or term of such series of DT Notes or the indenture or other instrument governing such series of DT Notes without the prior written consent of DT, which may be withheld in the DT s sole discretion.

After such time as DT no longer holds, directly or indirectly, a majority in principal amount of any series of the DT Notes, the Issuer shall not agree to any waiver or amendment of any provision or term of such series of the DT Notes or the indenture or other instrument governing such series of DT Notes without the consent of holders of at least 50.1% of the aggregate principal amount of such series of DT Notes.

#### Restrictions on optional redemption of DT Notes

For so long as any DT Entity holds, directly or indirectly, any DT Notes, Issuer shall not (a) issue any equity interests of Issuer for the purpose of redeeming or otherwise purchasing any DT Notes, or (b) use the proceeds of any sale of equity interests of Issuer or contributions to Issuer s common equity capital made with the proceeds of one or more sales of equity interests of MetroPCS or any other person to directly or indirectly redeem or otherwise repurchase any DT Notes.

#### Mutual cooperation regarding DTC

For so long as any DT Entity holds, directly or indirectly, any DT Notes, Issuer, MetroPCS, and DT shall take all actions reasonable required to submit the DT Notes for electronic book-entry delivery and settlement through DTC on the earliest possible date, including, if necessary, through the use of one or more DTC Participants, provided that the out of pocket costs of such actions shall be paid by TMUS.

- 2 -

## <u>Exhibit K</u>

#### **Denali Directors**

Invitations to continue as a member of the Denali Board following the Closing have been extended to each of James Perry (as one of the three independent directors) and Roger D. Linquist.

In the event that either or both of Messrs. Perry and Linquist are unable or unwilling to continue to serve on the Denali Board following the Closing, Deer and Denali will discuss in good faith a replacement for each such individual from the members of the Denali Board as of the date hereof; provided that if Deer and Denali are unable to reach agreement on any replacement, Denali shall designate such replacement from among the members of the Denali Board as of the date hereof.

Deer and Denali will also discuss in good faith the identification of such other persons as may be mutually agreeable to serve as members of the Denali Board following the Closing; provided that if Deer and Denali are unable to reach agreement on any one or more such persons, Deer shall designate such persons; provided that such persons shall be chosen such that at least three members of the entire Denali Board are qualified to be independent within the meaning of applicable SEC and NYSE rules and regulations and shall not be Affiliates of Deer.

#### **Denali Officers**

John Legère, Chief Executive Officer

Braxton Carter, Chief Financial Officer

James Alling, Chief Operating Officer, T-Mobile USA

Thomas Keys, Chief Operating Officer, MetroPCS

# **EVERCORE GROUP L.L.C.**

October 2, 2012

The Special Committee of the Board of Directors

MetroPCS Communications, Inc.

2250 Lakeside Boulevard

Richardson, TX 75082

Members of the Special Committee:

We understand that MetroPCS Communications, Inc., a Delaware corporation (the Company ), proposes to enter into a Business Combination Agreement, dated as of the date hereof (the Business Combination Agreement ), with Deutsche Telekom AG, an Aktiengesellschaft organized in Germany ( DT ), T-Mobile Global Zwischenholding GmbH, a Gesellschaft mit beschrankter Haftung organized in Germany and a direct wholly-owned subsidiary of DT (Global), T-Mobile Global Holding GmbH, a Gesellschaft mit beschrankter Haftung organized in Germany and a direct wholly-owned subsidiary of Global (Holding), and T-Mobile USA, Inc., a Delaware corporation and direct wholly-owned subsidiary of Holding (T-Mobile), pursuant to which, among other things and subject to certain adjustments set forth in the Business Combination Agreement, (i) the Company will effect a recapitalization and reverse stock split of its common stock, par value \$0.0001 per share (the Company Common Stock ), whereby each existing share of Company Common Stock will receive 0.5 shares of common stock, par value \$0.00001 per share ( New Company Common Stock ), (the Reverse Stock Split ), (ii) immediately following the effective time of the Reverse Stock Split, the Company will pay its stockholders a cash payment in the aggregate amount of \$1,500,000,000 (the Cash Payment ), (iii) the Company will acquire from Holding all of the capital stock of T-Mobile (the T-Mobile Shares ) in consideration of the issuance by the Company to Holding (or its designee) of shares of Company Common Stock equal to 74% of the fully-diluted shares of New Company Common Stock outstanding immediately following the Cash Payment (on a grossed-up basis to take into account the number of shares of New Company Common Stock so issued to Holding or its designee) (such shares of New Company Common Stock, the Business Combination Consideration; the closing of the purchase and sale of the T-Mobile Shares by the Company is referred to as the Closing and the acquisition by the Company of T-Mobile is referred to as the Business Combination ). We further understand that pursuant to the Business Combination Agreement, DT will purchase from T-Mobile at their face amount (w) \$15,000,000 of senior unsecured notes to be issued by T-Mobile (the DT Notes ) in connection with the settlement of approximately \$14,400,00,000 in certain intercompany obligations existing between DT and its subsidiaries (other than T-Mobile and its subsidiaries), on the one hand, and T-Mobile and its subsidiaries, on the other hand, and the creation of a receivable of T-Mobile from Holding in the amount by which the note proceeds exceed the amount of such intercompany indebtedness, as adjusted for the difference in the discounted present values of the intercompany obligations, on the one hand, and principal amount of DT Notes equal to the principal amount of such intercompany obligations, on the other hand, which receivable is being distributed by T-Mobile to Holding immediately prior to the Closing (x) up to an additional \$1,000,000,000 of senior unsecured notes to the extent that the Company does not obtain such indebtedness from other parties on agreed terms prior to the Closing, (y) up to an additional \$2,500,000,000 of senior secured notes as required to repay amounts outstanding under the Company s existing credit agreement at Closing to the extent that the Company does not obtain such indebtedness from other parties on agreed terms prior to the Closing and (z) up to an additional \$2,000,000 of senior unsecured notes as required to redeem outstanding senior notes of the Company to the extent that holders thereof have not agreed to provide a waiver of the occurrence of a change of a control trigger under such senior notes (the note purchases described in the foregoing clauses (w), (x), (y) and (z) being referred to as the DT Financing), and that the Company will pay certain commitment fees to DT in connection with the DT Financing at the Closing. The terms and conditions of the of the transactions contemplated by the Business Combination Agreement are more fully set forth in the Business Combination Agreement and terms used herein and not defined shall have the meanings ascribed thereto in the Business Combination Agreement.

EVERCORE GROUP L.L.C. 55 EAST 52ND STREET NEW YORK, NY 10055 TEL: 212.857.3100 Fax: 212.857.3101

October 2, 2012

Page 2

The Special Committee has asked us, assuming payment of the Cash Payment, whether in our opinion the Company s issuance of the Business Combination Consideration to Holding in consideration for the T-Mobile Shares is fair, from a financial point of view, to the Company and its stockholders (other than DT and its affiliates).

In connection with rendering our opinion, we have, among other things:

- (i) reviewed certain publicly available business and financial information relating to the Company and T-Mobile that we deemed to be relevant;
- (ii) reviewed certain non-public historical financial statements and other non-public historical financial and operating data relating to the Company prepared and furnished to us by the management of the Company;
- (iii) reviewed certain non-public historical financial statements and other non-public historical financial and operating data relating to T-Mobile prepared by the management of DT and T-Mobile and furnished to us by the Company,
- (iv) reviewed certain non-public projected financial data relating to the Company prepared and furnished to us by the management of the Company,
- (v) reviewed certain non-public projected financial data relating to T-Mobile prepared by the management of DT and T-Mobile
   (the T-Mobile Management Projected Financial Data ) furnished to us by the Company as well as certain adjustments thereto
   made at the direction of the Company s management (as so adjusted, the Adjusted T-Mobile Projected Financial Data );
- (vi) reviewed certain non-public projected operating data relating to the Company prepared and furnished to us by the management of the Company;
- (vii) reviewed certain non-public projected operating data relating to T-Mobile prepared by the management of DT and T-Mobile
   (the T-Mobile Management Projected Operating Data ) furnished to us by the Company as well as certain adjustments thereto
   made at the direction of the Company s management (as so adjusted, the Adjusted T-Mobile Projected Operating Data );
- (viii) discussed the past and current operations, financial projections and current financial condition of the Company and T-Mobile with the management of the Company (including their views on the risks and uncertainties of achieving such projections);
- discussed the past and current operations, financial projections and current financial condition of the Company and T-Mobile with the management of DT and T-Mobile (including their views on the risks and uncertainties of achieving such projections);

- (x) reviewed the amount and timing of the cost savings and operating synergies estimated by the Company management to result from the Business Combination (the Estimated Synergies );
- (xi) reviewed the amount, timing and use of certain tax attributes of the combined company as estimated by management of the Company;
- (xii) reviewed the reported prices and the historical trading activity of the Company Common Stock;
- (xiii) compared the financial performance of the Company and its stock market trading multiples with those of certain other publicly traded companies that we deemed relevant;
- (xiv) reviewed certain valuation multiples relating to the Business Combination with those of certain other transactions that we deemed relevant;
- (xv) reviewed a draft of the Business Combination Agreement dated October 1, 2012; and

B-2

October 2, 2012

Page 3

(xvi) performed such other analyses and examinations and considered such other factors that we deemed appropriate. For purposes of our analysis and opinion, we have assumed and relied upon, without undertaking any independent verification of, the accuracy and completeness of all of the information publicly available, and all of the information supplied or otherwise made available to, discussed with, or reviewed by us, and we assume no liability therefor. With respect to the projected financial and operating data relating to the Company, we have assumed that they have been reasonably prepared on bases reflecting the best available estimates and good faith judgments of management of the Company as to the matters covered thereby. We express no view as to any projected financial or operating data relating to the Company or the assumptions on which they are based. With respect to T-Mobile Management Projected Financial Data and the T-Mobile Management Projected Operating Data, we have assumed that they have been reasonably prepared on bases reflecting the best available estimates and good faith judgments of management of DT and T-Mobile as to the matters covered thereby. With respect to Adjusted T-Mobile Projected Financial Data and the Adjusted T-Mobile Projected Operating Data, we have assumed that the adjustments which Company management has directed us to make to the T-Mobile Management Projected Financial Data and the T-Mobile Management Projected Operating Data have been reasonably determined by Company management on bases reflecting the best available estimates and good faith judgments of management of the Company as to the matters covered thereby. We express no view as to any projected financial or operating data relating to T-Mobile (whether prepared by management of DT or T-Mobile, or as adjusted at the direction of management of the Company) or the assumptions on which they are based. We have also assumed that the Estimated Synergies are reasonably obtainable, were estimated on bases reflecting the best currently available estimates and good faith judgments of the future competitiveness, operating and regulatory environments and related financial performance of the Company and T-Mobile and will be realized in the amounts and at the times indicated thereby. We express no view as to the Estimated Synergies or the assumptions on which they are based. With respect to the amount, timing and use of the tax attributes of the combined company estimated by management of the Company, we have assumed that estimates are reasonable and will be realized, although we express no view as to such estimates or the assumptions on which they are based. For purposes of our analysis and opinion, at your request, we have relied on the projections prepared by the management of the Company with respect to projected financial and operating data of the Company, including Expected Synergies and estimates as to the amount, timing and use of tax attributes of the combined company, and on the Adjusted T-Mobile Projected Financial Data and the Adjusted T-Mobile Projected Operating Data.

For purposes of rendering our opinion, we have assumed, in all respects material to our analysis, that the executed Business Combination Agreement will be the same as the draft dated October 1, 2012 and reviewed by us, that the representations and warranties of each party contained in the Business Combination Agreement are true and correct, that each party will perform all of the covenants and agreements required to be performed by it under the Business Combination Agreement and that all conditions to the consummation of the transactions contemplated by the Business Combination Agreement will be satisfied without material waiver or modification thereof. We have also assumed with your consent that the terms of the DT Financing (including any fees payable to DT in connection therewith) are on terms no less favorable to the Company than could be obtained from an unaffiliated third party lender. We have further assumed that all governmental, regulatory or other consents, approvals or releases necessary for the consummation of the Business Combination will be obtained without any material delay, limitation, restriction or condition that would have an adverse effect on the Company, T-Mobile or the consummation of the Business Combination or materially reduce the benefits to the Company of the Business Combination.

We have not made nor assumed any responsibility for making any physical inspection, independent valuation or appraisal of the assets or liabilities of the Company or T-Mobile, nor have we been furnished with

B-3

October 2, 2012

Page 4

any such valuation or appraisals, nor have we evaluated the solvency or fair value of the Company or T-Mobile under any state or federal laws relating to bankruptcy, insolvency or similar matters. In addition, we have assumed that the outcome of any current and pending litigation affecting the Company or T-Mobile will not be material to our analysis. Our opinion is necessarily based upon information made available to us as of the date hereof and financial, economic, market and other conditions as they exist and as can be evaluated on the date hereof. It is understood that subsequent developments may affect this opinion and that we do not have any obligation to update, revise or reaffirm this opinion.

We have not been asked to pass upon, and express no opinion with respect to, any matter other than, assuming payment of the Cash Payment, whether the Company s issuance of the Business Combination Consideration to Holding in consideration for the T-Mobile Shares is fair, from a financial point of view, to the Company and its stockholders (other than DT and its affiliates). We do not express any view on, and our opinion does not address, any other term or aspect of the Business Combination Agreement or the transactions contemplated thereby or any term or aspect of any other agreement or instrument contemplated by the Business Combination Agreement, including, without limitation, (i) the fairness of the Reverse Stock Split or of the Cash Payment or any other consideration to be received in connection therewith by, the holders of Company Common Stock or any other securities, creditors or constituencies of the Company, (ii) the terms of the DT Financing (including the amount and nature of the fees payable to DT in connection therewith) or (iii) the fairness of the amount or nature of any compensation to be paid or payable to any of the officers, directors or employees of the Company, or any class of such persons, whether relative to the Reverse Stock Split, the Cash Payment, the Business Combination Consideration or otherwise.

We have assumed that any modification to the structure of the transaction will not vary in any respect material to our analysis. Our opinion does not address the relative merits of the transactions contemplated by the Business Combination Agreement as compared to other business or financial strategies that might be available to the Company, nor does it address the underlying business decision of the Company to engage in the Business Combination or any other transaction contemplated by the Business Combination Agreement, including the Reverse Stock Split and the Cash Payment. In arriving at our opinion, we were not authorized to solicit, and did not solicit, interest from any third party with respect to the acquisition of any or all of the Company Common Stock or any business combination or other extraordinary transaction involving the Company. This letter, and our opinion, does not constitute a recommendation to the Special Committee, the Board of Directors or to any other persons in respect of the Business Combination Agreement. We express no opinion herein as to the price at which shares of Company Common Stock will trade at any time. We are not legal, regulatory, accounting or tax experts and have assumed the accuracy and completeness of assessments by the Company, T-Mobile and their respective advisors with respect to legal, regulatory, accounting and tax matters.

We will receive a fee for our services upon the rendering of this opinion. The Company has also agreed to reimburse our expenses and to indemnify us against certain liabilities arising out of our engagement. During the two year period prior to the date hereof, no material relationship existed between Evercore Group L.L.C. and its affiliates and DT, T-Mobile or the Company, pursuant to which compensation was received by Evercore Group L.L.C. or its affiliates as a result of such a relationship. We may provide financial or other services to DT, T-Mobile or the Company in the future and in connection with any such services we may receive compensation.

In the ordinary course of business, Evercore Group L.L.C. or its affiliates may actively trade the securities, or related derivative securities, or financial instruments of the Company, DT or T-Mobile and their respective affiliates, for its own account and for the accounts of its customers and, accordingly, may at any time hold a long or short position in such securities or instruments.

October 2, 2012

Page 5

This letter, and the opinion expressed herein is addressed to, and for the information and benefit of, the Special Committee, in its capacity as such, in connection with its evaluation of the proposed Business Combination. The issuance of this opinion has been approved by an Opinion Committee of Evercore Group L.L.C. This opinion may be disclosed, referred to and communicated in its entirety to the Board of Directors of the Company, in its capacity as such, in connection with the Board of Directors evaluation of the proposed Business Combination and the receipt of the Special Committee s recommendation, if any.

This opinion may not be disclosed, quoted, referred to or communicated (in whole or in part) to any third party for any purpose whatsoever except with our prior written approval, except the Company may reproduce this opinion in full in any document that is required to be filed with the U.S. Securities and Exchange Commission and required to be mailed by the Company to its stockholders relating to the transactions contemplated by the Business Combination Agreement; provided, however, that all references to us or our opinion in any such document and the description or inclusion of our opinion therein shall be subject to our prior consent with respect to form and substance, which consent shall not be unreasonably withheld or delayed.

Based upon and subject to the foregoing, assuming payment of the Cash Payment, it is our opinion that, as of the date hereof, the Company s issuance of the Business Combination Consideration to Holding in consideration for the T-Mobile Shares is fair, from a financial point of view, to the Company and its stockholders (other than DT and its affiliates).

Very truly yours,

EVERCORE GROUP L.L.C.

By: /s/ Michael J. Price Michael J. Price Senior Managing Director

B-5

#### SPECIAL MEETING OF STOCKHOLDERS OF

]

[

## IMPORTANT NOTICE REGARDING THE AVAILABILITY OF PROXY MATERIALS FOR THE SPECIAL MEETING:

The Notice of Meeting, Proxy Statement and Proxy Card

are available at http://www.amstock.com/ProxyServices/ViewMaterial.asp?CoNumber=15247

Please sign, date and mail

your proxy card in the

envelope provided as soon

as possible.

i Please detach along perforated line and mail in the envelope provided. i

¢

#### THE BOARD OF DIRECTORS RECOMMENDS A VOTE FOR PROPOSALS 1, 2, 3 AND 4.

# PLEASE SIGN, DATE AND RETURN PROMPTLY IN THE ENCLOSED ENVELOPE. PLEASE MARK YOUR VOTE IN BLUE OR BLACK INK AS SHOWN HERE x

1. To approve the issuance of shares of MetroPCS Communications, Inc. common stock in connection with the transaction contemplated by the Business Combination Agreement, dated October 3, 2012, among Deutsche Telekom AG, T-Mobile Global Zwischenholding GmbH, T-Mobile Global Holding GmbH, T-Mobile USA, Inc., and MetroPCS Communications, Inc.	FOR 	AGAINST 	ABSTAIN 
2. To approve the Fourth Amended and Restated Certificate of Incorporation of MetroPCS Communications, Inc. in connection with the transaction contemplated by the Business Combination Agreement, dated October 3, 2012, among Deutsche Telekom AG, T-Mobile Global Zwischenholding GmbH, T-Mobile Global Holding GmbH, T-Mobile USA, Inc., and MetroPCS Communications, Inc.			

3. To cast a non-binding, advisory vote on the compensation that may be paid or become payable to MetroPCS Communications, Inc. s named executive officers based on, or otherwise relating to, the transaction contemplated by the Business Combination Agreement, dated October 3, 2012, among Deutsche Telekom AG, T-Mobile Global Zwischenholding GmbH, T-Mobile USA, Inc., and MetroPCS Communications, Inc.

4. To approve the continuation, adjournment or postponement of the special meeting, if necessary or appropriate to solicit additional proxies if there are insufficient votes at the time of the special meeting to approve both the stock issuance proposal (Proposal #1) and the new certificate of incorporation proposal (Proposal #2)

The shares will be voted as recommended by the Board of Directors UNLESS otherwise indicated in which case they will be voted as marked.

••

..

••

••

..

To change the address on your account, please check the box at right and indicate your new address in the address space above. Please note that changes to the registered name(s) on the account may not be submitted via this method.

 Signature of Stockholder
 Date:
 Signature of Stockholder
 Date:

 Note:
 Please sign exactly as your name or names appear on this Proxy. When shares are held jointly, each holder should sign. When signing as executor, administrator, attorney, trustee or guardian, please give full title as such. If the signer is a corporation, please sign full corporate name by duly authorized officer, giving full title as such. If signer is a partnership, please sign in partnership name by authorized person.

••

#### SPECIAL MEETING ADMISSION TICKET

#### SPECIAL MEETING OF STOCKHOLDERS OF

[

1

10:00 A.M., CDT

[Bank of America Theater

at the Eisemann Center

2351 Performance Drive

Richardson, TX 75082]

At the Special Meeting, stockholders will vote upon the proposals outlined in the Notice of Special Meeting of Stockholders of MetroPCS Communications, Inc. We look forward to your participation.

Upon arrival please present this Admission Ticket, together with a valid government-issued picture identification to enter the Special Meeting. This Admission Ticket only admits the stockholder identified on the reverse side and is non-transferable.

#### SPECIAL MEETING OF STOCKHOLDERS

#### THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS

You are encouraged to specify your choices by marking the appropriate boxes ON THE REVERSE SIDE. If the proxy card is signed, dated and returned with no specific direction, your shares will be voted in accordance with the Board of Directors recommendations as stated in the Proxy Statement. The proxies cannot vote your shares unless you sign and date and return this card.

The undersigned stockholder of MetroPCS Communications, Inc. (MetroPCS) acknowledges receipt of a Notice of Special Meeting of Stockholders and the accompanying Proxy Statement. The undersigned hereby revokes any proxy heretofore given with respect to such Special Meeting and hereby appoints Messrs. J. Braxton Carter, Vice Chairman and CFO, and Thomas C. Keys, President and COO, with full authority

## Table of Contents

and power to act, which may be exercised by either one or both of them, with power of substitution, as proxies and attorneys-in fact and hereby authorizes them to represent and to vote all shares of common stock of MetroPCS (the Common Stock ) which the undersigned is entitled to vote at the Special Meeting of Stockholders of MetroPCS, and any continuation(s), adjournment(s) or postponement(s) thereof, as indicated on the reverse side, which shall be held in the [Bank of America Theater at the Eisemann Center located at 2351 Performance Drive, Richardson, Texas 75082] at 10:00 a.m. (CDT) on [ ] (the Special Meeting ).

#### (Continued and to be marked, signed and dated on the reverse side)

14475

	SPECIAL MEETING OF STOCKHOLDERS OF			
	ſ		]	
	PROXY VO	TING INST	TRUCTION	NS
<b><u>INTERNET</u></b> - Access <b>www.voteproxy.co</b> Have your proxy card available when you ad		n-screen instru	ctions.	
<b><u>TELEPHONE</u></b> - Call toll-free <b>1-800-PROX</b> States or <b>1-718-921-8500</b> from foreign cour and follow the instructions. Have your prox	ntries from any touch-	-tone telephone	e	OMPANY NUMBER
Vote online/phone until 11:59 PM EDT the	day before the meetin	ng.	А	CCOUNT NUMBER
<b>MAIL</b> - Sign, date and mail your proxy care possible.	d in the envelope prov	vided as soon a	15	
IN PERSON - You may vote your shares ir Meeting.	n person by attending	the Special		

## IMPORTANT NOTICE REGARDING THE AVAILABILITY OF PROXY MATERIALS FOR THE SPECIAL MEETING:

The Notice of Meeting, Proxy Statement and Proxy Card

are available at http://www.amstock.com/ProxyServices/ViewMaterial.asp?CoNumber=15247

i Please detach along perforated line and mail in the envelope provided IF you are not voting via telephone or the Internet. i

#### THE BOARD OF DIRECTORS RECOMMENDS A VOTE FOR PROPOSALS 1, 2, 3 AND 4.

#### PLEASE SIGN, DATE AND RETURN PROMPTLY IN THE ENCLOSED ENVELOPE. PLEASE MARK YOUR VOTE IN BLUE OR

#### BLACK INK AS SHOWN HERE x

		FOR	AGAINST	ABSTAIN
	1. To approve the issuance of shares of MetroPCS Communications, Inc. common stock in connection with the transaction contemplated by the Business Combination Agreement, dated October 3, 2012, among Deutsche Telekom AG, T-Mobile Global Zwischenholding GmbH, T-Mobile Global Holding GmbH, T-Mobile USA, Inc., and MetroPCS Communications, Inc.			
	2. To approve the Fourth Amended and Restated Certificate of Incorporation of MetroPCS Communications, Inc. in connection with the transaction contemplated by the Business Combination Agreement, dated October 3, 2012, among Deutsche Telekom AG, T-Mobile Global Zwischenholding GmbH, T-Mobile Global Holding GmbH, T-Mobile USA, Inc., and MetroPCS Communications, Inc.			
	3. To cast a non-binding, advisory vote on the compensation that may be paid or become payable to MetroPCS Communications, Inc. s named executive officers based on, or otherwise relating to, the transaction contemplated by the Business Combination Agreement, dated October 3, 2012, among Deutsche Telekom AG, T-Mobile Global Zwischenholding GmbH, T-Mobile USA, Inc., and MetroPCS Communications, Inc.			
	4. To approve the continuation, adjournment or postponement of the special meeting, if necessary or appropriate to solicit additional proxies if there are insufficient votes at the time of the special meeting to approve both the stock issuance proposal (Proposal #1) and the new certificate of incorporation proposal (Proposal #2)			
ease check the in the address	The shares will be voted as recommended by the Board indicated in which case they will be voted as marked.	of Directo	ors UNLESS	otherwise

To change the address on your account, ple box at right and indicate your new address space above. Please note that changes to the registered name(s) on the account may not be submitted via this method.

Date:

Signature of Stockholder

Signature of Stockholder

Date:

Please sign exactly as your name or names appear on this Proxy. When shares are held jointly, each holder should sign. When signing as Note: executor, administrator, attorney, trustee or guardian, please give full title as such. If the signer is a corporation, please sign full corporate name by

duly authorized officer, giving full title as such. If signer is a partnership, please sign in partnership name by authorized person.